Southampton

University of Southampton Research Repository

Copyright © and Moral Rights for this thesis and, where applicable, any accompanying data are retained by the author and/or other copyright owners. A copy can be downloaded for personal non-commercial research or study, without prior permission or charge. This thesis and the accompanying data cannot be reproduced or quoted extensively from without first obtaining permission in writing from the copyright holder/s. The content of the thesis and accompanying research data (where applicable) must not be changed in any way or sold commercially in any format or medium without the formal permission of the copyright holder/s.

When referring to this thesis and any accompanying data, full bibliographic details must be given, e.g.

Thesis: Author (Year of Submission) "Full thesis title", University of Southampton, name of the University Faculty or School or Department, PhD Thesis, pagination.

Data: Author (Year) Title. URI [dataset]

UNIVERSITY OF SOUTHAMPTON

FACULTY OF HUMANITIES

Archaeology

Volume [1] of [1]

A Multifactorial Approach to the Estimation of Sex using the Facet Joints of the Spine

DOI: https://doi.org/10.5258/SOTON/D1228

by

Carolyn Felton

Thesis for the degree of Doctor of Philosophy

February 2019

UNIVERSITY OF SOUTHAMPTON

ABSTRACT

FACULTY OF ART AND HUMANITIES

Archaeology

Thesis for the degree of Doctor of Philosophy

A MULTIFACTORIAL APPROACH TO THE ESTIMATION OF SEX USING THE FACET JOINTS OF THE SPINE

Carolyn Louise Felton

This research aims to demonstrate how "form related to function" can be used to explain differences in the degree of zygapophyseal (facet) joint sexual dimorphism seen in populations with differing lifestyles. Whilst sex can be estimated with a reasonable degree of accuracy from the facet joints of the spine in populations with a high degree of sexual dimorphism, it can be problematic in samples that do not. Bone adapts to reflect the physiological strain placed upon it during life. Analysis of the relationship between bony adaptation to occupational stress and facet morphology identifies extrinsic factors that correlate to changes in facet size and to a lesser degree sagittal angle. Extrinsic factors are external biocultural influences on bone morphology (e.g. nutrition and physical activity). The number of discriminant functions that can be derived from a population with a low degree of sexual dimorphism is increased when these factors are included, increasing the potential to estimate sex. The strength of correlation and prevalence of extrinsic factors can also be used to measure the degree of physical activity undertaken by individuals and is indicative of gendered division of labour in the population under study.

Analysis of facet size and sagittal angle and the relationship and prevalence of extrinsic factors related to physical activity from skeletal material were examined from three contrasting samples. Two were from 18th century London with differing socio-economic status and the third a composite sample from three 5th -7th century Anglo-Saxon cemeteries located in southeast England. A comparative study of facet size and angle identified inter-sample differences in the degree of sexual dimorphism. Further comparison of differences in lifestyle as evidenced by activity patterns was carried out to assess the impact of extrinsic factors on facet remodelling. In particular, this thesis focuses on evidence of the gendered division of labour as manifest by

femoral robusticity, humeral directional asymmetry, vertebral osteophytosis, and osteoarthritis with analysis of the individual diagnostic criteria of eburnation, pitting and osteophytes. A distinct difference in prevalence of these factors was observed in the Anglo-Saxon sample and most obviously in females when compared with the 18th century samples, indicating that there was a difference in intensity of activity undertaken by this group in comparison with the other samples in this study. This suggests that the lack of facet joint sexual dimorphism observed in Anglo-Saxons is attributable to the degree of physical activity undertaken by these females and the subsequent remodelling of the facets as a functional adaptation to the mechanical loading they were subjected to.

This research demonstrates that for some populations, sex can be estimated with reasonable accuracy from vertebral facet dimensions but for less dimorphic samples, inclusion of extrinsic factors related to physical activity when deriving discriminant functions increases the opportunity to estimate sex. Furthermore, analysis of inter-sample prevalence rates for extrinsic factors provides supporting evidence of different levels of physical activity between the samples.

Table of Contents

Table	e of Co	ontents		i
Table	e of Ta	bles		vii
Table	e of Fig	gures		xiii
Acad	emic ⁻	Thesis: I	Declaration of Authorship	xvii
Ackn	owled	lgemen	ts	xix
Abbr	eviati	ons		хх
Chap	ter 1	Intro	duction	1
Chap	ter 2	Verte	ebral Structure	7
2.1	Str	ucture o	of a typical vertebra	7
2.2	De	velopme	ent of the vertebral column	7
	2.2.1	Mese	enchymal stage	8
	2.2.2	Cartil	aginous phase	10
	2.2.3	Ossifi	ication phase	10
2.3	An	atomy c	of the adult vertebral column	11
	2.3.1	Cervi	cal region	13
	2.3.2	Thora	acic region	14
	2.3.3	Thora	acic region	14
	2.3.4	Lumb	par vertebrae	16
	2.3.5	Interv	vertebral joints	17
		2.3.5.1	Joints of the vertebral bodies (Interbody Joints)	17
		2.3.5.2	Joints of the Vertebral Arches	18
	2.3.6	Curva	atures of the Spinal Column	21
	2.3.7	Crani	al and Caudal Shifts	24
		2.3.7.1	Occipito/atlantal shifts	24
		2.3.7.2	Cervico-thoracic shifts	24
		2.3.7.3	Thoraco-lumbar shifts	25
		2.3.7.4	Lumbo-sacral shifts	26
Chap	ter 3	Sex,	Age and Vertebral Morphology	29
3.1	Sex	and Fa	cet Size	29
	3.1.1	Sex a	nd Facet Joint Orientation	36
	_			

3.2	Age	and fac	et size	41
	3.2.1	Age ar	nd Osteophytosis	42
	3.2.2	Summ	ary	43
Chap	ter 4	Bioard	chaeology and Activity	44
4.1	Bio	archaeo	logy	44
4.2	Bor	ne Plastio	city and Remodeling	46
	4.2.1	Plastic	city	46
	4.2.2	Mode	ls of Bone Functional Adaptation	
		4.2.2.1	Bone Modeling	49
		4.2.2.2	Bone Remodeling	50
	4.2.3	Mecha	anical Stress and Bone Remodeling	54
	4.2.4	Remo	deling of articular surfaces	55
4.3	Rec	onstruc	tion of Lifestyle Using Markers of Occupational Stress	58
	4.3.1	Activit	ty Related Changes	58
	4.3.2	Marke	ers of Occupational Stress	59
	4.3.3	Limita	tions in the Use of MOS as Markers of Activity	60
	4.3.4	Activit	ty and Degenerative Joint Disease	62
		4.3.4.1	Definition of Degenerative Joint Disease	62
		4.3.4.2	Primary and Secondary Osteoarthritis	63
		4.3.4.3	The Process of Degeneration in the Spine	64
		4.3.4.4	Causes of Osteoarthritis	64
	4	4.3.4.5	Physical Activity and OA	67
	4	4.3.4.6	Specific activity and OA	68
		4.3.4.7	Limitations to the use of OA as a marker of physical activity	74
Chap	ter 5	Mate	rials and Methods	77
5.1	Gre	at Chest	terford	78
	5.1.1	Histor	ical Background of Great Chesterford	78
	5.1.2	Archa	eological Background of Great Chesterford	78
	5.1.3	Tapho	nomic Alterations at Great Chesterford	79
5.2	Buc	kland Ce	emetery	79
	5.2.1	Histor	ical Background to Buckland Cemetery	79
	5.2.2	Archa	eological Background to Buckland	80

	5.2.3	Tapho	pnomic alterations at Buckland	80
5.3	St	Anne's R	oad	82
	5.3.1	Histor	rical background to St Anne's Road	82
	5.3.2	Archa	eological Background to St Anne's Road.	82
	5.3.3	Tapho	pnomic alterations at St Ann's Road	82
5.4	St	Bride's C	rypt and St Brides Lower Cemetery	83
	5.4.1	Histor	rical Background to St Bride's Crypt and Lower Cemetery	83
	5.4.2	Archa	eological Background to St Bride's Crypt and Lower Cemetery	83
	5.4.3	Tapho	onomic alterations at St Bride's Crypt and Lower Cemetery	84
5.5	Inc	lusion Cr	riteria	84
5.6	Exc	clusion C	riteria	84
5.7	Me	ethod		85
	5.7.1	Metho	ods of Analysis	85
	5.7.2	Demo	graphic Data - Sex	86
		5.7.2.1	Estimation of Sex from Archaeologically Derived Samples	86
		5.7.2.2	Morphological Variation in Skeletal Structure	87
		5.7.2.3	Metric Variation in Skeletal Structure	88
		5.7.2.4	Methods used to estimate sex.	91
	5.7.3	Demo	graphic Data- Age	94
		5.7.3.1	Age Estimation from Archaeologically Derived Samples	94
		5.7.3.2	Methods used to assess age	97
	5.7.4	Data (Collection Methods for Facet Size and Angle	98
		5.7.4.1	Facet Size	98
		5.7.4.2	Facet Angle	101
	5.7.5	Osteo	arthritis	102
	5.7.6	Diagn	ostic Criteria	102
		5.7.6.1	Eburnation	104
		5.7.6.2	Pitting	105
		5.7.6.3	Osteophytes	107
		5.7.6.4	Vertebral body osteophytosis	108
		5.7.6.5	Prevalence Rates	108
	5.7.7	Extrin	sic Factors	109

	5	5.7.7.1	Femoral Robusticity	109
	5	5.7.7.2	Directional Asymmetry	111
	5	5.7.7.3	Sexual Dimorphism Index (SDI)	113
	5.7.8	Data N	Aanagement and Statistical Analysis	113
	5	5.7.8.1	Missing Data	113
	5	5.7.8.2	Descriptive Statistics	114
	5	5.7.8.3	Bonferroni Adjustment	114
	5	5.7.8.4	Statistical analysis of the relationship between sex, facet size and angle	114
	5	5.7.8.5	Correlation Analysis	114
	5	5.7.8.6	Age groups v actual age	115
	5.7.9	Assess	ment of the roles of predictor variables in assessing sex	115
	5	5.7.9.1	Discriminant function analysis	115
5.8	Pilo	t Study.		119
	5.8.1	Analys	sis of Intra-observer Reliability	119
	5	5.8.1.1	Intra-observer Error	119
	5	5.8.1.2	Continuous Data	122
Chap	oter 6	Result	5	125
Chap 6.1	o ter 6 Sam	Result	s	125 125
Chap 6.1 6.2	o ter 6 Sam Inte	Result nple Den er-sample	s nography e comparison of predictor variables	125 125 126
Chap 6.1 6.2	oter 6 Sam Inte 6.2.1	Result nple Den er-sample Inter-s	rography e comparison of predictor variables sample comparison of femoral robusticity	125 125 126 126
Chap 6.1 6.2	oter 6 Sam Inte 6.2.1 6.2.2	Result aple Den r-sample Inter-s Inter-s	ss	125 125 126 126 127
Chap 6.1 6.2	oter 6 Sam Inte 6.2.1 6.2.2 6.2.3	Result nple Den er-sample Inter-s Inter-s Inter-s	nography e comparison of predictor variables sample comparison of femoral robusticity sample comparison of humeral directional asymmetry sample comparison of prevalence rates	125 125 126 126 127 127
Chap 6.1 6.2	oter 6 Sam Inte 6.2.1 6.2.2 6.2.3	Result nple Den er-sample Inter-s Inter-s Inter-s 5.2.3.1	nography e comparison of predictor variables sample comparison of femoral robusticity sample comparison of humeral directional asymmetry sample comparison of prevalence rates Crude Prevalence Rates	125 125 126 126 127 127 127
Chap 6.1 6.2	oter 6 Sam Inte 6.2.1 6.2.2 6.2.3	Result nple Den er-sample Inter-s Inter-s 5.2.3.1 5.2.3.2	nography e comparison of predictor variables sample comparison of femoral robusticity sample comparison of humeral directional asymmetry sample comparison of prevalence rates Crude Prevalence Rates True Prevalence Rates	125 125 126 126 127 127 127 128
Chap 6.1 6.2	oter 6 Sam Inte 6.2.1 6.2.2 6.2.3 6 6	Result nple Den er-sample Inter-s Inter-s 5.2.3.1 5.2.3.2 Preval	nography e comparison of predictor variables sample comparison of femoral robusticity sample comparison of humeral directional asymmetry sample comparison of prevalence rates Crude Prevalence Rates True Prevalence Rates ence rates of Osteoarthritis.	125 125 126 126 127 127 127 128 137
Chap 6.1 6.2	oter 6 Sam Inte 6.2.1 6.2.2 6.2.3 6 6.2.4 6.2.5	Result nple Den r-sample Inter-s Inter-s 5.2.3.1 5.2.3.2 Preval Inter-s	nography e comparison of predictor variables sample comparison of femoral robusticity sample comparison of humeral directional asymmetry sample comparison of prevalence rates Crude Prevalence Rates True Prevalence Rates ence rates of Osteoarthritis sample comparison of OA prevalence rates.	125 125 126 126 127 127 127 127 137
Chap 6.1 6.2	oter 6 Sam Inte 6.2.1 6.2.2 6.2.3 6.2.4 6.2.5 Face	Result nple Den er-sample Inter-s Inter-s 5.2.3.1 5.2.3.2 Preval Inter-s et size	nography e comparison of predictor variables sample comparison of femoral robusticity sample comparison of humeral directional asymmetry sample comparison of prevalence rates Crude Prevalence Rates True Prevalence Rates ence rates of Osteoarthritis sample comparison of OA prevalence rates.	125 125 126 126 127 127 127 127 137 137 137 140
Chap 6.1 6.2	oter 6 Sam Inte 6.2.1 6.2.2 6.2.3 6 6.2.4 6.2.5 Face 6.3.1	Result nple Den er-sample Inter-s Inter-s 5.2.3.1 5.2.3.2 Preval Inter-s et size Facet S	sample comparison of predictor variables	125 125 126 126 127 127 127 127 137 137 137 140
Chap 6.1 6.2	oter 6 Sam Inte 6.2.1 6.2.2 6.2.3 6.2.4 6.2.5 Face 6.3.1 6.3.2	Result nple Den er-sample Inter-s Inter-s 5.2.3.1 5.2.3.2 Preval Inter-s et size Facet s	rs	125 125 126 126 127
Chap 6.1 6.2	oter 6 Sam Inte 6.2.1 6.2.2 6.2.3 6.2.4 6.2.5 Face 6.3.1 6.3.2 6.3.3	Result nple Den er-sample Inter-s Inter-s 5.2.3.1 5.2.3.2 Preval Inter-s et size Facet s Statist	rs	125 125 126 126 127 127 127 127 127 127 127 142 140 141 144
Chap 6.1 6.2	oter 6 Sam Inte 6.2.1 6.2.2 6.2.3 6.2.4 6.2.5 Face 6.3.1 6.3.2 6.3.3 6.3.4	Result nple Den er-sample Inter-s Inter-s 5.2.3.1 5.2.3.2 Preval Inter-s et size Facet s Statist Stepw	rs	125 125 126 126 127 127 127 127 127 127 127 142 140 141 144 145

	6.3.6	Age G	roups v Actual Age	151
	6.3.7	Relatio	onship between facet size and sample and temporal patterning	152
	6.3.8	Effect	of sample and temporal variation on the accuracy of sex estimation us	ing DFA
		equati	ions	154
	6.3.9	Relatio	onship between facet size and predictor variables	156
		6.3.9.1	Correlation between femoral robusticity and facet size	156
		6.3.9.2	Correlation between humeral and facet size directional asymmetry	162
		6.3.9.3	Diagnostic criteria for osteoarthritis as predictor variables	165
		6.3.9.4	Correlation between eburnation score and facet size.	165
		6.3.9.5	Correlation between pitting score and facet size	169
		6.3.9.6	Correlation between osteophyte score and facet size	172
		6.3.9.7	Correlation between vertebral osteophytosis score and facet size	175
6.4	Fac	et Angle		178
	6.4.1	Facet	Angle Metric Analysis	178
	6.4.2	Sexua	l dimorphism index for facet angle	179
	6.4.3	Statist	ical analysis of the relationship between sex and facet angle	182
	6.4.4	Stepw	ise discriminant function analysis for facet angle	183
	6.4.5	Relatio	onship between Age and Facet Angle	187
	6.4.6	Age G	roups v actual age	188
	6.4.7	Relatio	onship between facet angle and sample and temporal patterning	190
	6.4.8	Relatio	onship between facet angle and predictor variables	191
		6.4.8.1	Correlation between femoral robusticity and facet angle	191
		6.4.8.2	Correlation between humeral and facet angle asymmetry	197
		6.4.8.3	Correlation between eburnation score and facet angle	200
		6.4.8.4	Correlation between pitting score and facet angle	202
		6.4.8.5	Correlation between osteophyte score and facet angle	205
		6.4.8.6	Correlation between vertebral osteophytosis score and facet angle	208
6.5	Ste	pwise di	scriminant function with extrinsic variables	211
	6.5.1	Deriva	tion of discriminant function equations for all parameters	211
	6.5.2	Compa	arison of percentage correct classification using stepwise discriminant	function
		analys	is (DFA) with and without predictor variables	218
Chap	ter 7	Discus	ssion	223
7.1	Sun	nmarv o	f Results – Sexual Dimorphism	223
7.2	Sun	nmary o	f Results Extrinsic Factors	231
		,		

7.	2.1	Age Related Changes	231
7.	2.2	Sample and Temporal Patterning	
7.	2.3	Femoral Robusticity	235
7.	2.4	Humeral Asymmetry	235
7.	2.5	Osteoarthritis	236
7.	2.6	Vertebral Body Osteophytosis	
7.	2.7	DFA using Extrinsic Factor Variables	241
7.	2.8	General Discussion	242
Chapter	r 8	Conclusion	247
	_		
8.1	Futur	re Directions	
Abbrev	iation	s used in appendices	253
Append	dix A M	Measure of intra-observer agreement	257
Append Append	dix A M dix B F	Neasure of intra-observer agreement	257 289
Append Append Append	dix A M dix B F dix C F	Measure of intra-observer agreement Results of facet size analysis Results of facet angle analysis	257 289 383
Append Append Append Append	dix A M dix B F dix C F dix D 1	Measure of intra-observer agreement Results of facet size analysis Results of facet angle analysis Frue prevalence rates	257 289 383 475
Append Append Append Append Append	dix A M dix B F dix C F dix D 1 dix E E	Measure of intra-observer agreement Results of facet size analysis Results of facet angle analysis True prevalence rates Extrinsic variable summary	257
Append Append Append Append Append	dix A M dix B F dix C F dix D T dix E E dix F S	Measure of intra-observer agreement Results of facet size analysis Results of facet angle analysis True prevalence rates Extrinsic variable summary	257

Table of Tables

Table 3-1 Measurements of C1 used by Marino (1995) (author's image)
Table 3-2 Measurements of C2 (Wescott, 2000) 32
Table 3-3 Facet orientation of typical vertebrae (author's image)
Table 4-1 Summary of Bertram and Swartz Response to Wolff's Tenets 53
Table 5-1 Summary Table of Skeletal Remains Included in this Study Study
Table 5-2 Cranial, Pelvic and Femoral Features used to Determine Adult Sex
Table 5-3 Use of Femoral Head Diameter to Estimate Sex
Table 5-4 Probabilistic Sex Diagnosis Database (DSP) Measurement Description
Table 5-5 Age at Death Estimation Methods Used in This Study 97
Table 5-6 Age Categories used to divide the population into life phases 98
Table 5-7 Eburnation Scores 105
Table 5-8 Pitting Scores106
Table 5-9 Osteophyte Scores 109
Table 5-10 Osteophytosis Scores (Säger, 1969)
Table 5-11 Definition of Variables 117
Table 5-12 Great Chesterford Demographic121
Table 5-13 Fleiss' Kappa Level of Agreement for Intra-Observer Error 121
Table 5-14 Summary of Count of Facets Showing Significant Difference in Repeated Measures122
Table 6-1 Summary of cemetery origin and age distribution of skeletons used in this study126
· · · · · · · · · · · · · · · · · · ·
Table 6-2 Percentage crude prevalence rates for eburnation, pitting, osteophytes and
Table 6-2 Percentage crude prevalence rates for eburnation, pitting, osteophytes and osteophytosis 127
Table 6-2 Percentage crude prevalence rates for eburnation, pitting, osteophytes and osteophytosis 127 Table 6-3 Summary of Kruskal-Wallace analysis of inter-sample differences in true prevalence of

Cable 6-4 Summary of Mann-Whitney U test for differences in true prevalence between samplepairs135
Table 6-5 Median scores for true prevalence of male and female pitting scores
able 6-6 Crude prevalence rates for osteoarthritis137
able 6-7 Kruskal-Wallace test for inter-sample variation in true prevalence rates of OA 139
able 6-8 Mann-Whitney U Test for Inter-Sample Variation in True Prevalence
able 6-9 Median scores for Male and Female True Prevalence OA
able 6-10 Mann-Whitney U test for inter-sex differences in true prevalence of OA140
able 6-11 Summary of number of facets analysed for each sample
able 6-12 Count of data outliers by spinal region and sample141
able 6-13 Maximum and minimum standard deviation values for facet size
Table 6-14 Count of sexual dimorphism index (SDI) results by sample
Table 6-15 Percentage frequency of facets with statistically significant differences in facet sizebetween sexes144
able 6-16 Stepwise discriminant function analysis for facet size
analysis
Table 6-18 Percentage frequency of correlation classifications between facet size and actual ageat death
Table 6-19 Percentage frequency of correlation classifications between facet size and actual ageat death by spinal region152
Table 6-20 Percentage frequency of significant inter-sample differences in facet size using Kruska
wallace test
able 6-21 Percentage frequency of significant Mann-Whitney U test results for facet size between sample groups by spinal region
Table 6-22 Percentage accuracy of correct classification achieved after discriminant functionsubstitution

Table 6-23 Summary of femora available for correlation study
Table 6-24 Percentage frequency of correlation classifications between femoral robusticity and facet size 157
Table 6-25 Percentage frequency of correlation classifications between femoral robusticity andfacet size by spinal region
Table 6-26 Percentage frequency of superior and inferior facets with matching correlation classification 160
Table 6-27 Percentage frequency of ipsilateral and contralateral matching correlations femoralrobusticity and facet size161
Table 6-28 Number of humeri included in correlation 162
Table 6-29 Percentage frequency of correlation classifications between humeral and facet sizedirectional asymmetry163
Table 6-30 Percentage frequency of correlation classification between humeral and facet sizeasymmetry by spinal region
Table 6-31 Percentage frequency of correlation classifications between ebtot (total eburnation score) and facet size
Table 6-32 Percentage frequency of correlations between ebtot (total eburnation score) and facetsize by spinal region
Table 6-33 Percentage frequency of correlation classifications between pitot and facet size .169
Table 6-34 Percentage frequency of correlation classifications between pitot and facet size by spinal region 170
Table 6-35 Percentage frequency of correlation classifications between ostot and facet size 173
Table 6-36 Summary of count of correlation classification between ostot and facet size by spinal region 174
Table 6-37 Count of correlation classification between cvostot (total score for vertebralosteophytosis) and facet size
Table 6-38 Percentage frequency of correlations between cvostot and facet size by spinal region

Table 6-39 Percentage of data outliers by spinal region and sample 179	
Table 6-40 Maximum and minimum standard deviation values for facet angle179	
Table 6-41 Count of SDI Results by Sample 180	
Table 6-42 Percentage frequency of statistically significant results, facet angle with sex 182	
Table 6-43 Stepwise discriminant function analysis of facet angle 184	
Table 6-44 Percentage frequency of statistically significant results for facet angle with age group	,
Table 6-45 Percentage frequency of correlation classification between facet angle and actual age	e
at death, St Bride's Crypt189	
Table 6-46 Percentage frequency of correlation classification between facet angle and actual age	е
at death by spinal region189	
Table 6-47 Percentage frequency of significant results from Kruskal Wallis Test for Inter-Sample	
Differences in Facet Angle190	
Table 6-48 Percentage frequency of statistically significant differences in facet angle between	
sample groups by spinal region191	
Table 6-49 Percentage frequency of correlation classifications between femoral robusticity and	
facet angle192	
Table 6-50 Percentage frequency of correlation classifications between femoral robusticty and	
facet angle by spinal region193	
Table 6-51 Percentage Frequency of superior and inferior facets with the same classification of	
correlation	
Table 6-52 Percentage frequency of ipsilateral and contralateral matching correlations femoral	
robusticity and facet angle199	
robusticity and facet angle	
robusticity and facet angle	
robusticity and facet angle	

Table 6-55 Percentage frequency of correlation classification between ebtot (total eburnation
score)and facet angle200
Table 6-56 Percentage frequency of correlation classification between ebtot and facet angle by
spinal region201
Table 6-57 Percentage frequency of correlation classification between pitot (total pitting score)
and facet angle202
Table 6-58 Percentage frequency of correlation classifications between pitot (total pitting score)
and facet angle by spinal region203
Table 6-59 Count of correlation classifications between ostot (total osteophyte score) and facet
angle205
Table 6-60 Percentage frequency of correlation between ostot (total osteophyte score) and facet
angle by spinal region206
Table 6-61 Percentage frequency of correlation classification between cvostot (total vertebral
body osteophytosis score) and facet angle
Table 6-62 Percentage frequency of correlations between cvostot (total score vertebral
ostophytosis) and facet angle by spinal region
Table 6-63 Stepwise discriminant function analysis with extrinsic variables 213
Table 6-64 Comparison of percentage of correct classification of sex identification using stepwise
DFA219
Table 6-65 Summary of highest percentages for correct classification 222

Table of Figures

Figure 2-1 Common anatomical features shared by vertebrae (author's image)8
Figure 2-2 The three primordial growth layers (Moore and Persaud, 2003)9
Figure 2-3 Somite differentiation into sclerotome (Moore and Persaud, 2003)9
Figure 2-4 Lateral view of curvatures of the spine (author's image)12
Figure 2-5 Cervical vertebra13
Figure 2-6 Typical thoracic vertebra
Figure 2-7 Typical lumbar vertebra17
Figure 2-8 Zygapophyseal joint
Figure 2-8 Schematic diagram of a typical synovial joint19
Figure 2-10 Shape and orientation of superior facets
Figure 2-11 Developmental curvatures of the spine22
Figure 2-12 Measurement of kyphotic Cobb angle24
Figure 2-13 Grades of cervical rib26
Figure 2-14 Variations in vertebrae and their relationship to ribs27
Figure 3-1 Measurements of C1 used by Marino (1995)30
Figure 3-2 Measurements from C7 and T12 (Amores et al., 2014)
Figure 3-3 Measurement of facet angle
Figure 4-1 Range of degenerative processes in the spine (author's image)65
Figure 5-1 Map of southern England showing cemetery locations (author's image)78
Figure 5-2 Overall plan of Buckland (Long Hill) Anglo-Saxon cemetery showing areas excavated in 1951-3 and 1994: from Parfitt and Anderson (2012)
Figure 5-3 Variation in facet border definition T10, B282B99
Figure 5-4 Margins of facet border for measurement T10 B282B100

Figure 5-5 Illustration of position of facet height and width measurements C6, GC9 (author's
image)100
Figure 5-6 Differences between C5 left and right inferior facet size GC100 (author's image) . 101
Figure 5-7 Measurement of facet sagittal angle102
Figure 5-8 Facet surface degenerative changes of the left superior facet of T6 from FAO901454 (author's image)
Figure 6-1 Inter-sample crude prevalence rates128
Figure 6-2 Inter-sample variation in the percentage of facets with eburnation
Figure 6-3 Inter-sample variation in the percentage of facets with pitting
Figure 6-4 Inter-sample variation in percentage of facets with osteophytes
Figure 6-5 Inter-sample variation in the percentage of vertebral bodies with osteophytosis . 132
Figure 6-6 Inter-sample variation in true prevalence of OA138
Figure 6-7 Sexual dimorphism index scores for facet size143
Figure 6-8 Percentage frequency of significant differences in facet size by sample and spinal region
Figure 6-9 Percentage frequency of strong correlations between femoral robusticity and facet size
Figure 6-10 Percentage frequency of matching strong correlations between femoral robustucty
and facet size in left and right superior and inferior facets
Figure 6-11 Percentage frequency of strong correaltions between humeral and facet size directional asymmetry
Figure 6-12 Percentage frequency of strong correlations between ebtot (total eburnation score)
and facet size169
Figure 6-13 Percentage frequency of strong correlations between pitot (total pitting score) and facet size
Figure 6-14 Percentage frequency of strong correlation classifications between ostot (total
osteophyte score) and facet size

Figure 6-15 Percentage frequency of strong correlations between CVOS (total vertebral body		
osteophytosis score) and facet size180		
Figure 6-16 Sexual dimorphism index scores for facet angle181		
Figure 6-17 Percentage frequency of significant differences in facet angle with sex by sample and spinal region		
Figure 6-18 Percentage frequency of strong correlations between femoral robusticity and facet		
angle by spinal region194		
Figure 6-19 Percentage frequency of matchin correlations between femoral robusticity and facet angle		
Figure 6-20 Percentage frequency of strong correlations between humeral and facet angle		
asymmetry200		
Figure 6-21 Percentage frequency of strong correlations between ebtot (total eburnation score) and facet angle202		
Figure 6-22 Percentage frequency of strong correlations between pitot (total pitting score) and		
facet angle205		
Figure 6-23 Percentage frequency of strong correlations between ostotot (total osteophyte score)		
and facet angle208		
Figure 6-24 Percentage frequency of strong correlations betwen cvostot (total vertebral body osteophytosis score and facet angle)211		

Academic Thesis: Declaration of Authorship

I, Carolyn Louise Felton declare that this thesis and the work presented in it are my own and has been generated by me as the result of my own original research.

A Multifactorial Approach to the Estimation of Sex using the Facet Joints of the Spine.

I confirm that:

- This work was done wholly or mainly while in candidature for a research degree at this University;
- 2. Where any part of this thesis has previously been submitted for a degree or any other qualification at this University or any other institution, this has been clearly stated;
- 3. Where I have consulted the published work of others, this is always clearly attributed;
- 4. Where I have quoted from the work of others, the source is always given. With the exception of such quotations, this thesis is entirely my own work.
- 5. I have acknowledged all main sources of help;
- 6. Where the thesis is based on work done by myself jointly with others, I have made clear
- 7. None of this work has been published before submission

Signed:	
-	
Date:	
Date.	

Acknowledgements

I have many people to thank for their help and support, without which I would not have got this far.

Firstly, I would like to thank my supervisors Professor Joanna Sofaer and Assistant Professor Sonia Zakrzewski for their unending patience and support. I am truly indebted to you.

I would like to thank Dr Andrew Richardson (Canterbury Archaeological Trust), Joe Seaman (Eastbourne Heritage) and Dr Rebecca Redfern and Jelena Bekvalac for giving me permission to access the Buckland, St Anne's Cemetery and St. Bride's collections.

Finally I would like to thank my husband David Felton for his patience and generosity without which I would never have been able to pursue my academic dream; my daughter Victoria, son James and daughter-in-law Carrie for being understanding of the constraints on my time and for providing me with delightful grandchildren, which are the best antidote to thesis stress. My final thanks go to my late parents, Jo and Terry Torpey, who brought me up to believe that if you have sufficient ambition, nothing is beyond reach. Abbreviations

Abbreviations

- AA absolute asymmetry
- AS Anglo-Saxon
- AP antero-posterior
- C cervical
- DA directional asymmetry
- DJD degenerative joint disease
- GC Great Chesterford
- HDB average diaphyseal diameter
- I inferior
- Lt left
- L lumbar
- ML medio-lateral
- MOS marker of occupational stress
- OA osteoarthritis
- R right
- S superior
- SBC St Bride's Crypt
- SBL St Bride's Lower
- T thoracic
- ZA zygapophyseal joint

Chapter 1 Introduction

Chapter 1 Introduction

"To measure is to know" (William Kelvin 1824-1907).

Human skeletal remains can provide unique insights into the past and the present, insights that cannot be gained elsewhere. This is achieved using bioarchaeological analysis to reconstruct the biological profile of the individuals under study by determination of sex, age and health of a skeleton.

Sex is the key parameter when creating a demographic profile (Scheuer, 2002; Bruzek and Murail, 2006; Iscan and Steyn, 2013). Correct classification of sex increases the accuracy of estimation of other biological attributes such as stature and body mass (Ruff et al., 2005; Sládek et al., 2015). The sex of a skeleton can be estimated from comparison of features that demonstrate sexual dimorphism. Sexual dimorphism is defined as the systematic difference in form between individuals of different sex in the same species (Plavcan and van Schaik, 1994) and is a measure of anatomical differences between males and females determined at time of conception and enhanced in subsequent physiological development (Armelagos, 1998:1). There are many factors involved in the development of the adult human skeleton. Bones can change shape in response to intrinsic or extrinsic factors. Intrinsic factors arise from within the body systematically and are under genetic constraint, e.g. hormone levels, (Gilsanz et al., 1997; Plavcan, 2001). Traits that are under genetic control seem to appear earlier than those affected more by the environment (extrinsic factors) (Moore, 2013). Extrinsic factors are introduced from outside the body e.g. nutrition, biomechanics of activity including the gendered division of labour (Charisi et al., 2011), body mass (Gilsantz et al., 1997), and functional wear of the bones (Acsádi and Nemeskéri ,1970). Load-bearing bones exhibit postnatal plasticity, enabling them to functionally adapt to their biological and cultural environment as a result of external stimuli. These adaptations manifest as subtle nuances in morphological change, recorded as an osteobiography and can be observed during detailed analysis of bone.

The pelvis is considered to display the greatest degree of sexual dimorphism and is the most reliable element for estimation of sex in adults (Bruzek and Murail, 2006; Işcan and Steyn, 2013), especially when used in conjunction with cranial morphology (Meindl et al., 1985). However, it is problematic if these elements are damaged or missing from archaeological contexts, resulting in the need for methods to estimate sex from other bones that demonstrate dimorphism (Cox and Mays, 2000). Vertebrae are suitable elements for estimation of sex because they exhibit sexual dimorphism in size, even before puberty (Moore, 2013).

1

Chapter 1 Introduction

There have been a number of studies that report the use of vertebral measurements to accurately classify the sex of human skeletons (MacLaughlin and Oldale, 1992; Marino, 1995; Wescott, 2000; Pastor, 2005; Yu et al., 2008; Marlow and Pastor, 2011; Hou et al., 2011; Amores et al., 2013). These studies have developed discriminant functions (DF) from vertebral measurements and achieved up to 90% correct classification. This implies that certain vertebral measurements are the most effective univariate predictors of sex for non-pelvic elements (Spradley and Jantz, 2011). Although these results are very encouraging, there is lack of consistency in the methods used. Individual or small groups of vertebrae are used in these studies rather than the whole spine. Whilst population specificity of the DF is mentioned, there is very little comment of the degree of dimorphism seen in the samples under study and no attempt to control for functional adaptation of the vertebrae to mechanical loading.

Previous vertebral studies can be criticised as they have continually emphasised the use of simple linear measurements and indices to estimate sex, which limits structural interpretation with little consideration of functional correlates and the relevance of findings to more general issues of biological adaptation (Lovejoy et al., 1982). This criticism can be extended to metric sexual dimorphism studies in general, as it has been noted that many are based on the development of statistical techniques to determine sex without any consideration of functional factors that could have led to those differences (Ruff, 1987).

The gendered allocation of labour can lead to differing male and female activity patterns with the resulting biomechanical stress effects manifesting as different degrees of sexual dimorphism (Pomeroy and Zakrzewski, 2009). Lifestyle-related factors are thought to influence vertebral size with the effects appearing more prominent in females than males (Oura et al., 2016; Oura et al., 2017). As there is a link between vertebral size and lifestyle-related factors, this thesis aims to analyse changing morphology of the facet joints to define a novel method of sex estimation and also to gain an understanding of the effects of extrinsic (biocultural) factors on facet size and angle. This analysis addresses the primary debate of this research which is to demonstrate that recognition of not only biological differences but also the effects of biomechanical differences may lead to a better understanding of sexual dimorphism in humans and improve sex discrimination from the skeleton. This approach is an important adjunct to the field of sex estimation as it evaluates the effect of extrinsic influences on degree of sexual dimorphism and adds explanation for the degree of dimorphism seen in a sample when compared to samples from dissimilar backgrounds. It is a response to the criticisms of Lovejoy et al. (1982) and Ruff (1987) that functional adaptation is not included in metric sex assessment and tests to see if that criticism is valid.

Vertebral facet joints were considered to be an appropriate variable for this study because changes in facet joint size and angle of orientation are related to the biomechanical requirements of the joint (movement and weight bearing) (Moore and Dalley, 2006). Facet joints form the posterior weight-supporting pillar of the spine and do not have any direct muscle attachments; therefore, changes in size or angle are not influenced by the exertional pull of muscles (Moore and Petty, 2005). From a methodological point of view, facet joints are readily identifiable and tend to have delineated boundaries of the articular surface allowing for ease of measurement.

To address the issues outlined above, this thesis takes a methodological approach to test the accuracy with which facet joint size and angle of orientation can be used to estimate sex for the whole spine. This novel approach also evaluates the relationship between extrinsic factors as a measure of biomechanical loading with facet morphology and the effect on degree of correct sex classification by inclusion of these factors into discriminant functions.

To facilitate this, data have been collected from three contrasting samples (See Chapter 5):

1) St Bride's Crypt, a documented set of skeletons of known sex and age from 18th-19th century London. These individuals were interred in the crypt of St Bride's Church in Fleet Street, London and were of high socio-economic status.

2) St Bride's Lower Cemetery, also from 18th-19th century London and found in Farringdon Road. The individuals were from the same parish as those of St Bride's Crypt but were of lower socioconomic status. The sex and age of these individuals was unknown prior to analysis.

3) Anglo-Saxon; this sample is comprised of three separate cemeteries from the 5th-7th century (Great Chesterford, Essex; Buckland, Kent and St Anne's, East Sussex). These individuals are from rural locations. The sex and age of these individuals was also unknown prior to analysis.

Discriminant functions (DF) were derived for each sample using facet size and angle and the percentage of correct classification of sex were compared. DF's derived from St Bride's Lower and Anglo-Saxon samples in this study were substituted into the St Bride's Crypt control sample to determine the degree of population specificity.

Discriminant functions (DF) that included scores from the extrinsic factors were also derived for each sample and the percentage for correct classification of sex were compared to the results of using DF derived from facet size and angle alone.

Inter-sample differences were further explored by comparing the difference in crude and true prevalence rates for eburnation, pitting, osteophytes, vertebral osteophytosis and osteoarthritis (as markers of differing degree of activity). The calculation of crude prevalence rate identified the

Chapter 1 Introduction

number of individuals with a specific variable but was not a measure of the degree with which that variable occurs in each individual. There is a risk of over- or under-representation because this method assumes that all bones are equally preserved, which is generally not the case for archaeologically derived skeletons. True prevalence rates were also calculated to identify the number of facets affected. Both crude and true prevalence rates were calculated as recommended by Waldron (2007).

This study was developed to answer two key questions:

- Does vertebral facet size and sagittal angle demonstrate sexual dimorphism? In the case
 of incomplete skeletal remains, damaged or co-mingled depositions, the vertebrae may
 be the only bones available for analysis. Therefore, an extension to the number of existing
 vertebral parameters that can be used to estimate sex increases the opportunity to
 accurately profile human remains.
- 2) Does the inclusion of extrinsic factors relating to physical activity in DFs increase the degree of accuracy for sex estimation? In a sample that is not very sexually dimorphic, it is not possible to generate discriminant functions for sex estimation. Inclusion of extrinsic factors that measure biomechanical differences leading to functional adaptation of the facets is hypothesized to increase the number of functions that can be derived, leading to greater effectiveness in estimating sex.

Overall this thesis demonstrates that to successfully consider sexual dimorphism of the facets, the lifeway of the individual should be taken into consideration. There is wide natural variation in and between samples, which may be explained by different physical activity patterns or lifestyle. A multifactorial approach to the estimation of sex gives a much broader insight into the lives of the individuals under examination whilst creating a novel approach using the facets of the spine to create discriminant functions.

Outline of Chapters

This thesis is organised into eight chapters including this introduction (Chapter 1).

Chapter 2 provides an overview of the anatomy of the spine and vertebrae beginning with development of the spine from embryo to adult. It describes typical and atypical vertebrae for each spinal region and the effects of caudal and cranial shifts of transitional vertebrae.

Chapter 3 explores the relationship between sex, age and vertebral morphology and reviews the existing literature.

Chapter 4 begins with a disciplinary history of bioarchaeology and describes biocultural theory, emphasising the origin and traditional application of this model. A review of bone plasticity, models of bone functional adaptation and the effects of mechanical stress on bone follow this. The chapter continues with an overview of markers of occupational stress in bioarchaeological contexts and discusses the effects that may be observed in bone. The complexities of relating patterns of degeneration to specific activities are also discussed.

Chapter 5 provides details on the materials and methods used in this study. The first part of the chapter provides information about the skeletal samples (materials). A brief history of each skeletal sample is included. The methods section identifies problems relating to the estimation of sex and age from skeletal material, details the parameters measured or scored and the method used, rationale for choice of method and choice of statistical method. The chapter ends with results from intra-observer reliability assessment from the pilot study of the methods used.

Chapter 6 presents the results of descriptive demographic statistics and basic metric data analysis followed by the results of statistical analysis to evaluate each hypothesis in turn

Chapter 7 is a discussion of the results obtained in Chapter 6

Chapter 8 is the conclusion to the thesis and identifies limitations to the study and areas for further research.

Chapter 2 Vertebral Structure

An understanding of the anatomy and physiology of the component parts of the spine is integral to this thesis as the questions posed in this study are related to its structure and function. This chapter begins with a description of the basic structure of a typical vertebra, followed by development of the vertebral column from embryo to adult. It continues with an overview of the structure and function of the cervical, thoracic and lumbar spinal regions. The chapter ends with identification of cranial and caudal shifts and the impact this has on vertebral level identification.

2.1 Structure of a typical vertebra

A typical vertebra (Figure 2-1) (overleaf) consists of an anterior body and a posterior arch (Schwartz, 1995; Moore and Dalley, 2006). These structures join to form the vertebral foramen, which contains the spinal cord surrounded by the meninges (Steele and Bramblett, 1988; Snell, 2004; White et al., 2012). The vertebral arch is formed from two pedicles at the sides and two laminae that join posteriorly (Tucker, 1990; Snell, 2004; Moore and Dalley, 2006). There are seven processes that arise from the vertebral arch, four articular (left and right superior and inferior facets) (Schwartz, 1995; Adams et al., 2006), two transverse processes (Moore and Petty, 2005, Moore and Dalley, 2006) and a single spinous process (Schwartz, 1995; Snell, 2004).

2.2 Development of the vertebral column

The spine is a complex and vital structure, serving as a pillar to support upper body weight whilst protecting the spinal cord. Embryonic development of the spine depends on a cascade of events that allow for the formation of musculoskeletal and neural components. The cascade begins very early in the life of the embryo with the primitive streak, well-defined germ layers and the notochord developing in the third week of gestation (Kaplan, 2005). The notochord and somites are the most significant structures responsible for development of the future vertebral column (O'Rahilly, 1996).

The embryonic spinal cord develops in three key stages that blend into each other rather than being distinct (mesenchymal, cartilaginous and ossification stages) (Oliver and Middleditch, 1999).

Chapter 2 Vertebral Structure



Figure 2-1 Common anatomical features shared by vertebrae (author's image)

2.2.1 Mesenchymal stage

Two weeks after conception, the embryo forms three distinct layers (see Figure 2-2.) (overleaf);

1) The ectoderm (forms the dermis and the central and peripheral nervous system) (Moore and Persaud, 2003:51). It is in contact with the amniotic fluid surrounding the embryo and foetus during gestation

2) The mesoderm (forms the connective tissues, organ linings and blood related organs) (Vernon-Roberts, 1988)

3) The endoderm (forms primitive soft tissue, notochord, lungs and digestive system) (Moore and Persaud, 2003). The notochord is an embryonic midline structure arising in the endoderm and forming the longitudinal midline axis, around which the vertebral bodies are organised (Snell, 2004:951). The notochord contains a central canal (see Figure 2-3) (overleaf).



Figure 2-2 The three primordial growth layers (Moore and Persaud, 2003)

In the mesenchymal stage, the mesoderm differentiates into three main areas; paraxial, intermediate and lateral mesoderm (Moore and Persaud, 2003). The vertebral column centres on the notochord and originates from two rods of paraxial mesoderm that form around the notochord as it extends below the base of the skull (Barnes, 2012)

The paraxial mesoderm segments into 42-44 pairs of somites (blocks of tissue) by the 5th week after conception (Rawls and Fisher, 2010). They develop in a craniocaudal fashion and will eventually be instrumental in the formation of the bones of the head, vertebrae and other bony structures of the thorax and associated musculature. The somites are transient structures that differentiate into a ventro-medial part (sclerotome) and a dorso-lateral part (dermomyotome) (Snell, 2004). The sclerotomal cells are responsible for the formation of the spine and the dermomyotomes form the muscle cells and overlying dermis of the skin (O'Rahilly, 1996).



Figure 2-3 Somite differentiation into sclerotome (Moore and Persaud, 2003)

In the 4th week of gestation, the cells of the sclerotome compartment of the somites migrate ventro-medially to surround the notochord (Figure 2-3), increasing in cell number, density and expression of extracellular matrix proteins which cause the mesenchyme to condense into a mesenchymal core and become the centrum of the vertebral body (Tam and Trainor, 1994). The

Chapter 2 Vertebral Structure

body of the vertebra develops from the caudal part of one sclerotome and the cranial part of the next sclerotome (Rawls and Fisher, 2010). Dorsal and lateral outgrowths from the mesenchymal vertebral body lead to the formation of the rest of the vertebra (Oliver and Middleditch, 1999) and the mesenchymal cells between the vertebral bodies become the intervertebral discs (Vernon-Roberts, 1988). The cells that initially migrated to adjacent to the neural tube, rather than the notochord, develop into neural arches, which serve to protect the spinal cord, vessels and nerve roots (Moore and Persaud, 2003). The neural arch consists of two pedicles and left and right halves of the laminae (Snell, 2004). The other processes associated with the posterior vertebral arch include the spinous process, transverse processes and articular processes, including facet joints) (Oliver and Middleditch, 1999).

2.2.2 Cartilaginous phase

This is a short phase and occurs at about the sixth week of development when two centres of chondrification appear in the centrum of each vertebral body and one in each of the lamellae of the neural arch (Scheuer and Black, 2004:188). The chondrification centres in the neural arches expand to complete neural arches posteriorly and a spinous process develops at the point of arch completion (Oliver and Middleditch, 1999). These chondrification centres also expand laterally to form the transverse processes and anteriorly to join with the centrum (Rawls and Fisher, 2010).

2.2.3 Ossification phase

Primary ossification centres form from about the 9th week of development after blood vessels grow into the developing vertebrae (Scheuer and Black, 2004). Ossification begins at T12, shifting cranially, and L1, shifting caudally (Schmorl and Junghanns, 1971:3). It occurs in the centrum and on each side of the vertebral arch (Moore and Persaud, 2003). By the seventh month of gestation, primary ossification centres can be seen in all areas (Oliver and Middleditch, 1999). Intervertebral discs form from the notochord and surrounding sclerotomes and the notochord begins to disintegrate (Vernon Roberts, 1988).

Secondary ossification centres appear during adolescence (Snell, 2004). They first appear in the cartilage covering the superior and inferior endplates of the vertebral bodies and form the epiphyseal end plates (Rawls and Fisher, 2010). Secondary ossification centres also appear at the tip of each transverse process and spinous process and ossification progresses until by the 25th year of life, the secondary ossification centres have fused with the rest of the vertebra (Snell, 2004)
2.3 Anatomy of the adult vertebral column

The spine is a segmented flexible structure extending from the cranium to the pelvis and consists of a series of 33 vertebrae that are divided into sections generally comprising seven cervical, twelve thoracic, five lumbar, five fused sacral and four fused coccygeal bones (Schwartz, 1995; Oliver and Middleditch, 1991; Levangie and Norkin, 2005; White et al., 2012). Figure 2-4 illustrates a lateral view of the spinal regions showing the anatomical regions and direction of curvature. There is disagreement on the number of individuals showing variation in the relative number of vertebrae. White et al.(2012) and Barnes (2012) consider it occurs in 10% of individuals and O'Rahilly (1986) in 5-12%. None of these papers commented on the sample profile from which their data was obtained.

The spine has a number of functions:

- 1) Protects the spinal cord and spinal nerves (Moore and Dalley, 2006; White et al., 2012)
- 2) Supports the weight of the body superior to the pelvis (Steele and Bramblett, 1988; Oliver and Middleditch, 1991)
- Provides a point of articulation for the ribs and supports the thoracic cage (Tucker, 1990; Schwartz, 1995; Snell, 2004).
- 4) Has flexibility to allow movement and support locomotion (Levangie and Norkin, 2005; Adams et al., 2006)
- 5) Is a site for muscle and ligament attachment (Gunn, 1992; Trew and Everett, 2005)
- Has elasticity to enable it to act as a shock absorber (Tucker, 1990; Oliver and Middleditch, 1991; Moore and Dalley, 2006).

The living adult is reported to have a mean length of the vertebral column of 72-75 cm with approximately a quarter of that length formed by the intervertebral discs (Schwartz, 1995; Snell, 2004; White et al, 2012). Vertebrae generally display common anatomical features with regional variations that are related to function (serial homology) (Moore and Petty, 2005).

Chapter 2 Vertebral Structure



Figure 2-4 Lateral view of curvatures of the spine (author's image)

The next three sections will discuss the spinal regions and the vertebral anatomy specific to that region.

2.3.1 Cervical region

There are generally seven vertebrae in the cervical region. C1 (atlas) and C2 (axis) and C7 are considered atypical vertebrae whereas C3-C6 are typical for the region (Figure 2-5) (Gosling et al., 1996; Snell, 2004).

The bodies of typical cervical vertebrae are roughly rectangular, wider in the medio-lateral aspect when compared with the antero-posterior aspect (Moore and Dalley, 2006). Joints of Luschka (uncovertebral joints) are found at the anterior periphery of the body by small uncinate processes and are formed between the uncinate process inferiorly and uncovertebral articulation superiorly (Williams, 2000). The pedicles project postero-laterally and are short and round whilst the laminae are long and thin (Moore and Petty, 2005; Adams et al., 2006). These combine with the body to form a large, triangular vertebral foramen (McMinn et al., 1993; Snell, 2004).

The superior articular facets face posteriorly and the inferior facets face anteriorly (Oliver and Middleditch, 1999). The facets are small and angled at approximately 45^o from the horizontal plane (Manaster and Osborne, 1987). Anterior and posterior tubercles can be found on the transverse process, which also contains the transverse foramen allowing the passage of the vertebral artery (Schwartz, 1995; White et al., 2012). The spinous processes are short (C3-C5) and bifid (C3-C6). The spinous process of C7 is long and prominent (Moore and Dalley, 2006).

C1 is an atypical vertebra as it does not have a body or spinous process (McMinn et al., 1993; Snell, 2004) but exists as a bony ring comprised of two wedge shaped lateral masses connected by anterior and posterior arches (Schwartz, 1995; Gosling et al., 1996, Oliver and Middleditch, 1996). The atlas vertebra articulates with the occiput superiorly (atlanto-occipital joint) and axis inferiorly (atlanto-axial joint) via articular surfaces on the superior and inferior aspects (Gunn, 1992; Williams, 2000). The superior articular facet of C1 can vary in its structure and may consist of a single oval facet or two discrete facets (Donlon, 2000).

C2 is also an atypical cervical vertebra. Unlike C1, this vertebra has a body and spinous process, however the superior surface of the body is represented by a projection from the anterior surface of the vertebra (the odontoid peg or dens). This represents the body of C1 when the two vertebrae articulate (Snell, 2004; White et al., 2012).

C7 is also considered to be an atypical cervical vertebra due to the presence of a long non-bifid spinous process making it easily identifiable as the *vertebra prominens* (Gosling et al., 1996; Gunn, 1992; Levangie and Norkin, 2005). The vertebral foramen is smaller, and the transverse processes are larger than in C3-C6 (Snell, 2004; Moore and Dalley, 2006).

13





2.3.3 Thoracic region

The twelve thoracic vertebrae increase in size from cranial to caudal (William, 2000). The bodies are heart shaped (Steele and Bramblett, 1988; Moore and Dalley, 2006) and have short, posteriorly facing pedicles with short laminae (Williams, 2000). The vertebral foramen is smaller than those of the cervical and lumbar vertebrae and is circular in shape (Gunn, 1992; Williams, 2000). The superior facets are thin, flat and face posteriorly and laterally, whilst the inferior joints face anteriorly, superiorly and medially, with the exception of the twelfth thoracic vertebra where the inferior facets face laterally in the transition to lumbar vertebrae (Snell, 2004; Moore and Dalley, 2006). The plane of the facets lies on an arc centred on the vertebral body (Oliver and Middleditch, 1999; Levangie and Norkin, 2005). In the thoracic region, the left superior facets are generally longer, narrower and more sagitally orientated than the right (Masharawi et al., 2008). Vertebral asymmetry is discussed in section 5.7.7.2. The transverse processes are thick, strong and extend postero-laterally. Their length shortens from T1-T12 (Schwartz, 1995; Moore and Dalley, 2006) The spinous processes are long and slope postero-inferiorly (Steele and Bramblett, 1988; White et al., 2012). T1-T12 have costal facets on the vertebral bodies, which articulate with the head of the rib (McMinn et al., 1993; Gosling et al., 1996). They manifest as demi-facets with the superior component at the root of the pedicle (Williams, 2000) and the inferior demi-facet at the lower border of the vertebral body (Williams, 2000). This pattern is seen on mid- thoracic vertebrae with variation at the upper and lower vertebrae.



Figure 2-6 Typical thoracic vertebra (T9 from GC81) (author's image)

T1 has a complete costal facet inferior to the superior margin of the body and also a demi-facet at the perimeter of the inferior margin of the body (Schwartz, 1995; Snell, 2004). There is also a "butting facet" associated with the superior facet and located on the laminae which serves to limit the downward displacement of the inferior articular surface of C2 (Levangie and Norkin, 2005; Moore and Petty, 2005). T10-T12 (and sometimes T9) also have complete costal facets on the sides of the vertebral body but unlike T1, do not have a demi-facet at the inferior margin of

Chapter 2 Vertebral Structure

the body (McMinn et al., 1993; Scheuer and Black, 2000; Moore and Dalley, 2006). The transition from the thoracic type-vertebra to the lumbar-type can occur anywhere between T10/11 and T12/L1 (Pal and Routal, 1999). Transition at T11/12 is the more frequently seen than at T12/L1. This occurs when there has been a cranial shift at the thoraco-lumbar junction and the transitional facets seen usually on T12 can be seen on T10 or T11 (Pal and Routal, 1999; Barnes, 2012). It is more common for there to be a slow transition (gradual change over a number of levels) from thoracic to lumbar facet with 93% of forty-one spinal columns from skeletons of Indian origin, demonstrating this trend, in contrast with 7% of the spines examined demonstrating an abrupt shift (Pal and Routal, 1999). Cranial shifts at the thoraco-lumbar junction can also lead to small or absent rib facets on T12 due to hypoplasia or agenesis of the ribs (Barnes 2012).

2.3.4 Lumbar vertebrae

A typical lumbar vertebra has a massive body when compared to the thoracic and cervical regions. When viewed from above, the body appears kidney shaped (Snell, 2004; Moore and Dalley, 2006). The pedicles are short, thick and are directed in a posterior direction to join with short, thick and inferiorly directed laminae (Steele and Bramblett, 1988, McMinn et al., 1993; White et al., 2012). The vertebral foramen is triangular in shape being smaller than in the cervical and larger than in the thoracic regions (Gosling et al., 1996; Snell, 2004). The superior facets face medially and posteriorly and have mamillary processes on the posterior surfaces (Moore and Dalley, 2006; Adams et al., 2006). The inferior articular facets are directed antero-laterally (or laterally) (Gunn, 1992; Snell, 2004; White et al., 2012). The transverse processes are long and slender with an accessory process on the posterior surface of the base of each transverse processes (Steele and Bramblett, 1988; Moore and Dalley, 2006). In contrast, the spinous processes are short and sturdy, quadrangular in shape and project posteriorly (Gosling et al., 1996; Snell, 2004).

The fifth lumbar vertebra articulates with the first sacral vertebra. Anomalies can occur here when there is a cranial shift of the lumbo-sacral border (Barnes, 2012). L5 can be assimilated into the sacrum (sacralisation) (Pal and Routal, 1999). The inverse can also be seen in the case of a caudal shift, where the first sacral segment shows lumbar characteristics, which can, in severe cases lead to non-fusion of the first to second sacral segment (Barnes, 2012). These lumbar transition vertebrae are reported to be a common occurrence with prevalence varying between 4% -30% (Elster, 1989; Bron et al., 2007).

The information above is an anatomic description. When considering the morphology of vertebrae, it should be remembered that there is considerable variation between individuals. Some individuals have vertebrae that are nearer the ancestral end of the range of shape variation, i.e. they bear a closer resemblance to chimpanzee and orang-utan vertebrae than typically

anatomically modern humans which predisposes the individual to intervertebral disc herniation and the formation of Schmorl's nodes (Plomp, 2015).



Figure 2-7 Typical lumbar vertebra (L2 from GC81) (author's image)

2.3.5 Intervertebral joints

The vertebrae articulate with each other by means of cartilaginous joints between the bodies and synovial joints between the articular processes (Levangie and Norkin, 2005).

2.3.5.1 Joints of the vertebral bodies (interbody joints)

These cartilaginous joints (symphyses) are formed between the hyaline covered vertebral end plates of two adjacent vertebrae and a fibrocartilaginous intervertebral disc joined by ligaments (Snell, 2005; Moore and Dalley, 2006). There are no intervertebral discs between the first two cervical vertebrae or between the fused bones of the sacrum and coccyx (Snell, 2004). The interbody joints of the lower cervical region are saddle joints thereby allowing motion in only two planes (Levangie and Norkin, 2005).

The anterior and posterior longitudinal ligaments act to hold the vertebrae firmly together, whilst allowing movement to take place (Snell, 2004; Levangie and Norkin 2005). The flexible nature of the intervertebral discs allows movement of the vertebral bodies with a range of movements taking place at the interbody joints which include: gliding (anterior to posterior), allowing the bodies to move during flexion and extension of the spine; distraction and compression, where the

Chapter 2 Vertebral Structure

bodies can move apart of closer together and rotation (anterior to posterior and laterally) (Oliver and Middleditch, 1991).

The intervertebral disc and anterior part of the vertebrae (i.e. the vertebral body and intervertebral disc) act as load bearing structures whilst the zygapophyseal joints provide posterior load support (Benoist, 2003).

2.3.5.2 Joints of the vertebral arches

These joints are termed the zygapophyseal joint (and commonly referred to as facet joints) (Moore and Dalley, 2006). The major function of the facet joint is to protect the discs from shear forces and/or excessive flexion and axial rotation whilst facilitating controlled movements in order to prevent damage to the discs (Adams and Hutton, 1980).



Figure 2-8 Zygapophyseal (facet) joint (author's image)

Facet joints are diarthrodial (synovial) joints, which are free moving articulations found in the axial and appendicular skeleton. They differ from synarthrodial joints (e.g. coronal sutures, symphysis pubis) in that there is no connective soft tissue that directly connects adjacent bony surfaces; instead, they are connected to one another by means of a joint capsule that encloses the joint, allowing flexibility of movement (Levangie and Norkin, 2005). All diarthrodial joints are constructed in a similar way and have a joint capsule that is composed of two layers, a joint cavity enclosed by the joint capsule, synovial tissue that lines the inner surface of the capsule, synovial fluid within the joint cavity and hyaline cartilage that lines the surface of the enclosed bones (Allan, 1998). Figure 2-9 illustrates a typical synovial joint.

The synovial zygapophyseal joints are formed between the superior and inferior articular processes of the vertebrae (Moore and Petty, 2005). In the living, the articular surfaces are covered in hyaline cartilage and surrounded by capsular ligament (Kalichman and Hunter, 2007). The joints are supported and stabilized by the supraspinous, interspinous, intertransverse ligaments, ligamentum flavum and ligamentum nuchae (McMinn et al., 1993; Moore and Dalley, 2005).



Figure 2-9 Schematic diagram of a typical synovial joint (author's image)

The cervical facet joints are true synovial joints with lax joint capsules that allow for a large range of movement but act as a limiter at the end of the available range (Levangie and Norkin, 2005). The joints contain fibroadipose meniscoids (Oliver and Middleditch, 1999), which are firmly attached to the fibrous capsule surrounding the joint (Mercer and Bogduk, 1993; Bogduk, 1997). These inclusions are thought to protect the joint surfaces during flexion and extension by evening out the undulations of the articular surface and increasing surface area for load transmission (Trew and Everett, 2005). Uhrenholt et al. (2008) identified the presence of folds in the synovial membrane of cervical facet joints that have a protective and lubricative function and compared their function with that of menisci.

Rotation of up to 40° is possible in the lower cervical spine due to the approximately 45° angle of the facet joints to the horizontal plane (Manaster and Osborn, 1987; Panjabi et al., 1993). The angle of the facet joints acts to limit rotation in the cervical spine in order to prevent over-rotation and potential injury to the spinal cord (Oliver and Middleditch (1999). Yogandan et al. (2003) made an anatomic study of cervical spine facet morphology using data collected from 6 cadavers. It was reported that facet morphology varied with vertebral level, with differences seen between the upper and lower cervical spine and between males and females, in particular females had a thinner layer of articular cartilage covering the surface of the articulation than males. This could lead to exposure of the underlying subchondral bone during mechanical stress.

Chapter 2 Vertebral Structure

This is an important issue for this study as levels of degenerative change on the facet surfaces are recorded. The reduced degree of articular cartilage may account for higher levels of degeneration in female cervical spines, as the variation in morphology would alter the biomechanical behaviour of the spine when subject to external loading. Yogandan et al. (2003) also noted that the cervical facet joint surfaces were more planar than in the thoracic region, this is in contrast with Oliver and Middleditch (1999) who report the opposite, describing the thoracic region joints as plane synovial joints. Other sources also describe the thoracic joints as planar (Snell, 2004; Moore and Dalley, 2006). Planar joints are a type of synovial gliding joint between two articular surfaces that are flat or nearly flat, permitting the joint surfaces to glide over each other in the direction of the plane of the joint (Moore and Dalley, 2006).

Thoracic region facet joints also have fibroadipose meniscoids present (Oliver and Middleditch, 1999) and the joint capsules are tauter than in the cervical and lumbar regions limiting the range of movement (Levangie and Norkin, 2005; Moore and Petty, 2005). Flexion and extension are limited by the shape and orientation of the thoracolumbar facets, whilst adding additional stability on flexion (Holdsworth, 1972). In the upper thoracic spine, the angle between the planes of the joints and the horizontal is approximately 60°. In the mid thoracic region, the angles become closer to vertical and the lower thoracic segments the joints lie more in the sagittal plane (Oliver and Middleditch, 1999).

The facet joints of the lumbar region are true synovial joints with fibroadipose meniscoids present (Levangie and Norkin, 2005). The superior facet shape approximates between a 'C' and a 'J' shape, which is thought to offer protection against forward displacement of the superior vertebra (Bogduk, 1997). This morphology also causes the lumbar vertebrae to sustain considerable amounts of load when the spine is in extension (Adams and Hutton, 1981). The joint capsules are laxer than in the thoracic region but more taut than in the cervical region (Moore and Petty, 2005). The articular surfaces are almost vertical in the upper lumbar spine, becoming more oblique at L4/5 and L5/S1 enabling the lower joints to resist more force (Yang and King, 1984).

The changes in angle of the plane and shape of the joint surface in each region of the spine are related to the biomechanical requirements of the joint (movement and weight bearing). Figure 2-10 presents examples of facet orientation in the cervical, thoracic and lumbar regions. The changes in angle and orientation can be clearly seen when the vertebrae are observed superiorly.



Figure 2-10 Shape and orientation of superior facets demonstrating regional change (from GC100) (author's image)

2.3.6 Curvatures of the spinal column

A lateral view of the curvatures of the adult spine can be seen in Figure 2-11. The curves of the spinal column can be described as primary or secondary. The thoracic and sacral curves are primary i.e. developed *in utero* due to the flexed foetal position (Moore and Petty, 2005). The

Chapter 2 Vertebral Structure

primary curve of the foetal vertebral column in the foetus is concave posteriorly (Levangie and Norkin, 2005). These curves are retained through life due to height differences between the anterior and posterior parts of the vertebrae (Moore and Dalley, 2006). The secondary curves in the cervical and lumbar regions develop in infancy as a result of extension from the flexed foetal position (Oliver and Middleditch, 1999).



Figure 2-11 Developmental curvatures of the spine (modified from Moore and Dalley, 2006:514)

A cervical lordosis develops when the child begins to hold its head upright, usually beginning at the end of pregnancy and becoming more pronounced at about three to four months of age (Comin et al., 1995; Snell, 2004) and the lumbar lordosis occurs at about 12-18 months of age when the child gains the ability to support itself in the upright position and begins to walk (Moore and Dalley, 2006). The lumbar curvature does not fully develop until the child is about two years old and fully ambulatory (Steele and Bramblett, 1988).

In the adult (See Figure 2-11), four curves are clearly visible in the normal spine. The cervical and lumbar curves are convex anteriorly (lordotic) (Gosling et al., 1996; Steele and Bramblett, 1988; Snell, 2004), whilst those of the thoracic and sacral regions are concave anteriorly (kyphotic) (Gosling et al., 1996; Steele and Bramblett, 1988; Snell, 2004). The degree of kyphosis in the sacral region is less pronounced in females than males, causing less projection of the coccyx into the pelvic outlet leading to larger size of the birth canal (Oliver and Middleditch, 1999; Snell, 2004; Moore and Dalley, 2006). Degeneration of the intervertebral discs in the elderly leads to reduction of lordosis in the cervical and lumbar regions and the C-shaped curve seen *in utero* become apparent again (Snell, 2004; Levangie and Norkin, 2005). Disc degeneration causes approximation of the vertebral bodies at the anterior margin due to reduced disc height; this in turn reduces the degree of lordosis (Oliver and Middleditch, 1999).

The kyphosis seen in the thoracic region may be caused by vertebral body morphology (Panjabi et al., 1991). In order to investigate the contribution of the intervertebral discs to the shape of the thoracic spine, Goh et al. (1999) measured the Cobb Angle (Figure 2-12) of 93 individuals (35 female and 58 male) using data collected from lateral spine radiographs and mid-sagittal CT films. The Cobb angle is the preferred clinical measure of sagittal deformity and can be defined as the angle formed at the point of intersection between a line drawn parallel to the superior endplate of the vertebra at the top of the curve to be measured and a line drawn parallel to the inferior endplate of the vertebra at the bottom of the curve (McAlister and Shackleford, 1975).



Figure 2-12 Measurement of kyphotic Cobb angle (Moore and Dalley 2006).

Goh et al. (1999) indicate that there is a relationship between degree of kyphosis and the shape of the vertebral bodies but less so for the relationship between kyphosis and the intervertebral

Chapter 2 Vertebral Structure

discs. Overall, they noted a trend towards a more noticeable anterior wedging of the vertebral body and intervertebral discs in the mid-thoracic region. The shape of the vertebral bodies and the degree of curvature in the thoracic spine is important because the curved vertebral column is more able to resist compressive loads than an uncurved spine and the curves provide additional flexibility (Levangie and Norkin, 2005). Increased mechanical loading causes passive compression of the intervertebral discs (limited by the zygapophyseal joints and longitudinal ligaments), and dynamic increased curvature (limited by the muscles; erector spinae, multifidus, semispinalis thoracis) that are antagonistic to the movement (Kapanji, 1972).

2.3.7 Cranial and caudal shifts

During development of the vertebrae, most spines are formed with 7 cervical, 12 thoracic and 5 lumbar vertebrae, however, shifts in the differentiating characteristics in the border vertebrae between two regions can occur. Border shifting occurs during development of the sclerotomes as they separate in cranial and caudal halves, with each half joining its neighbouring half (Barnes, 2012). These shifts happen most commonly at the transition zones of the occipito/atlantal, the cervico-thoracic, thoraco-lumbar and lumbosacral junctions.

2.3.7.1 Occipito/atlantal shifts

Cranial shifts occur when the base of the occiput and atlas bones shift upwards (Levangie and Norkin, 2005) leading to fusion of the atlas to the occipital bones. This is a very rare occurrence (0.14 to 0.75% incidence) (Guebert et al., 1987). A cranial shift can also affect development of the dens of the axis leading to presentation as a separate bone, with in extremely rare cases, complete agenesis of the dens (Barnes, 2008).

Caudal shifts in this area are a more common occurrence and can be expressed uni- or bilaterally (Kulkarni and Ramesh, 2012). They can cause the axis to become distorted as it assimilates into the skull base (Moore and Petty, 2005) and can also develop transverse foramina and/or demonstrate a complete anterior arch whilst the posterior arch is often incomplete (Barnes, 2012).

2.3.7.2 Cervico-thoracic shifts

Cranial shifts in this region primarily affect the anterior costal portion of the transverse process of C7 as it is incorporated into the rib-bearing thoracic region (Barnes, 2012). The variations in cervical rib development due to cranial shift can be seen in Figure 2-13. Caudal shifts in this region can lead to the reduction in size of the transverse process and rib of T1 Moore and Petty, 2005).

Cervical ribs are the most common bony anomaly associated with the cervico-thoracic region. They are most commonly seen on C7, although there is evidence that they can extend as high as C4 (Brain et al., 1967; Schmorl and Junghanns, 1971). Barnes (2012) explains this anomaly as the cranial shift of the cervico-thoracic border where the anterior costal portion of the C7 transverse process is incorporated into the rib bearing thoracic region. Cranial and caudal shifts are further described in section 2.2.6. There are different degrees of cervical rib development (see Figure 2-13). The reverse can also happen where there is a caudal shift of the cervico-thoracic junction (Pal and Routal, 1999). This may manifest as shortened abnormally shaped ribs and reduced transverse process size in C7 (Barnes, 2012).



Figure 2-13 Grades of cervical rib (modified from Barnes, 2012:62)

2.3.7.3 Thoraco-lumbar shifts

Cranial shifting in this region leads to a reduction in size or elimination of the 12th rib (Barnes, 2012). The facet joints on T12 are thought to designate the changing border between the thoracic and lumbar section of the spine in many anatomy text books (Snell, 2004; Levangie and Norkin, 2005), however Haeusler et al. (2014) report that between 40%-70% of modern humans are characterised by a change in vertebral articular facet orientation at T11 rather than T12. The broad range of affected modern humans is due to population variation (genetics).

Caudal shifts at the thoraco-lumbar junction can lead to the transverse processes transforming into lumbar ribs, or rib-like projections that are still attached to the vertebrae (Asher et al., 2011). The transitional facets from T12 can shift downwards to L1 (Barnes, 2012).

Chapter 2 Vertebral Structure

2.3.7.4 Lumbo-sacral shifts

Cranial shifts in this region can cause L5 to be sacralised and incorporated into the sacrum. This can be complete or incomplete (Bornstein and Peterson, 1966) whereas caudal shifts can cause S1 to take on the characteristics of a lumbar vertebra, sometimes completely separating from the sacrum to manifest as a 6th lumbar vertebra (Asher et al., 2011).

In summary, caudal and cranial shifts can change the characteristics of vertebrae at the regional junctions of the spine. This can cause confusion when identifying vertebrae and create measurements outside the normal range for facet size and angle (Pal and Routal, 1999). Vertebrae that were clearly transitional were excluded from this study.

Figure 2-14 overleaf illustrates the variation seen in rib distribution and also sacralisation and lumbarisation due to cranial and caudal shift.

This chapter has presented the structural, functional and developmental anatomy of the spine. Identification of vertebrae is facilitated by a good knowledge of regional differences in structure but hampered by the effects of cranial and caudal shifts. Vertebrae that were clearly transitional as a result of cranial or caudal shifts were excluded from this study as the variation in facet orientation and size would confound the results. Skeletons with other than normal numbers of vertebrae were also excluded (i.e. 7 cervical, 12 thoracic and 5 lumbar vertebrae).



Figure 2-14 Variations in vertebrae and their relationship to ribs (modified from Barnes, 2012)

- A) Normal arrangement of vertebrae
- B) Demonstrates the effect of a cranial shift with a cervical rib present, that can be seen articulating with C7, a diminished 12th rib articulating with T12 and an incomplete sacralisation of L5
- C) An example of a caudal shift, with increased size of the 12th rib and a small lumbar rib. In this situation, it is normal to see larger transverse processes at L4 and smaller than average transverse processes at L5. S1 is partially separated from the sacrum (lumbarisation)

The previous chapter presented an overview of the anatomy and development of the spine from embryo to adult. This chapter builds on that information by considering the relationship between sex, age and changes in vertebral morphology. It includes a review of the literature pertaining to sexual dimorphism of vertebrae from a clinical and archaeological perspective.

3.1 Sex and facet size

Many osteological elements demonstrate sexual dimorphism in size as well as shape. Sexual dimorphism can be defined as the body size, body shape, behavioural and rate/timing of development differences seen between males and females (Frayer and Wolpoff, 1985; Moore, 2013). It is the result of the complex inter-action of a series of genetic and environmental factors linked to the growth and development of individuals (Cocilovo et al. 2013). It would not be unreasonable to expect to be able to identify sexual dimorphism in the vertebral column given that males generally have larger bodies than females and males tend to develop greater muscle mass leading to increased development of the skeletal system (Cocilovo et al., 2013). However, there have been few clinical and bioarchaeological studies on the suitability of the vertebral column to be used to estimate the sex of a skeleton, possibly because the spine does not carry a sex-specific role.

Early clinical studies of vertebrae were predominantly focussed on comparative measurements without consideration of sexual variation (Aeby, 1879; Anderson, 1883; Cunningham, 1886; Dwight, 1907; Cyriax, 1920; Francis, 1955a) or were descriptive (Putti, 1927; Stewart, 1952; Francis, 1955b). Later studies recognised the possibility of estimating the sex of an individual by assessment of sexual dimorphism in bones and studies of metric variation in the vertebrae were made with varying degrees of success (Marino, 1995; Pastor, 2005). Most studies focussed on a single vertebral level or region of the spine (i.e. cervical, thoracic or lumbar). These studies are reviewed below.

Steele and Bramblett (1988) considered that the C1 vertebra had no previously described anatomical markers that could be considered visually useful as sex or ethnicity indicators, however, Marino (1993) demonstrated that estimation of sex could be reliably determined from C1 (atlas) using metric assessment and discriminant function equations (See Table 3-1, overleaf). He hypothesised that since the superior facets of C1 articulate with the occipital condyles of the cranium and form a load-bearing region, this would manifest as morphological differences between the sexes. He created seven regression and seven discriminant function equations,

concluding that sex could be accurately determined with 60-85% accuracy using data collected from 100 C1 vertebrae from the Terry Collection. When comparing his results with known sex samples from the Hammon-Todd collection, he achieved a success rate of 60-77%.

Measurement	Superior	Inferior
Maximum length of left and right facets		
Maximum width of left and right facets		
Maximum distance between lateral edges of facets		
Maximum vertebral foramen height and width		

Table 3-1 Measurements of C1 used by Marino (1995) (author's image)

Swenson (2013) replicated and expanded Marino's method in an unpublished Master's Thesis. She achieved 75.7-89.1% accuracy for sex estimation. She explained the difference in accuracy as being due to the use of three ancestral groups (Caucasian, African American and Hispanic) whilst Marino used two (Caucasian and African American). She argues that the introduction of a third group into the study led to increased variation and thus greater misclassification particularly of the Hispanic group. These individuals tend to be more gracile and there was a higher likelihood that males could be identified as females when looking at all three groups. Marino (1997) found that some measurements differed between the two groups and that the overall measurements were slightly greater in Caucasians that in African Americans. However, the outcome supports Marino's hypothesis that C1 can be used to determine the sex of an individual with reasonable accuracy, with the caveat from Marino:

"Before any reliance is placed on this technique within forensic applications, more work with larger and more geographically and temporally diverse populations should be conducted" (Marino, 1995:132).

Sexual dimorphism has also been demonstrated in C2 for eight measurements made by Wescott (2000) (Table 3-2 overleaf) He collected his data from a sample of 200 males and 200 females from the Hamman-Todd and Terry anatomical collections and generated five discriminant function equations using a stepwise discriminant analysis that selected the variables that had the greatest amount of discriminating ability. Using a cross validation procedure, it was found that an accuracy of 89% for the "White" sample, 81% for the "Black" sample and a range of 81.7-83.4% accuracy across the whole sample could be achieved depending on the function used. He identified the maximum length of C2, maximum length and width of the superior facet, length of the vertebral foramen and maximum height of C2 to be the most discriminating measurements.

Marlow and Pastor (2011) blind tested Wescott's method to validate its ability to accurately determine the sex of individuals from the Spitalfields collection (of known sex). They were able to validate Wescott's claims that C2 can be accurately used to determine the sex of human skeletal remains however, they noted a decrease in overall classification accuracy when compared to Wescott's results and ascribe this to population variability. They noted that the maximum distance between superior facets was the best discriminating measurement, followed by the maximum length of C2, maximum width of vertebral foramen and odontoid process sagittal diameter.

Schaffler et al. (1992) were also able to demonstrate sexual dimorphism using the dens of C2. Their study was of 120 individuals (males/females, whites and blacks) from the Hamman-Todd collection. They made seven measurements relating specifically to the dens as part of a clinical study. They identified that the height of the dens, the anterior and posterior height of the vertebral body and the antero-posterior diameter of the dens were dimorphic with the relative dimensions consistently skewed towards higher values in males and lower values in females. This research was investigating sex differences in morphology from a clinical viewpoint and not as a means of estimating the sex of an individual and therefore no measure of accuracy was

calculated. The dens was estimated to be 5-10% larger in males than females.

Table 3-2 Measurements of C2 (Wescott, 2000)



Medina (2011) took 14 measurements from C2 and achieved a percentage of correct classification of 84.2% using individuals from a Columbian sample. He reported that the sagittal maximum body diameter, the maximum length of the superior facet, the length of the vertebral foramen, the maximum width of the superior facet and the maximum distance between superior facets were the most useful measurements for sex estimation.

Bethard and Seet (2013) published a further validation of Wescott's method. They tested the applicability of the method against a modern American sample and again achieved comparable results, reiterating that C2 is a sexually dimorphic bone and effective predictor of sex. They noted that for their population they achieved a degree of accuracy of 86.7%, this is in contrast with Marlow and Pastor's (2011) results that achieved an accuracy of 77.6% when tested against the Spitalfields population. This supports the need to develop population specific methods of sex estimation as identified by Spradley and Jantz (2011).

Gama et al. (2015) developed Wescott's method further and used 13 dimensions from C2 to develop logistic regression models to estimate sex with a similar degree of accuracy (86.7 to 89.7%) for a Coimbra skeletal sample. They found that the most discriminating variables were the maximum width of C2, the sagittal body diameter, the maximum length of C2 and the maximum width of the right superior facet.

Haugen (1994) focused on the spinous process as part of her study on sex differences in the cervical spine. She used four measurements of the spinous process to assess vertebral sexual dimorphism from C2-C7 (C1 does not have a spinous process). Vertebral body height was also measured. All measurements were found to be sexually dimorphic with a 68-95% accuracy rate in estimation of sex, with slightly higher accuracy for females than males. She noted that female vertebrae were proportionally smaller than male vertebrae.

Several researchers have attempted to identify sexual dimorphism of the vertebral foramen in the cervical spine, with potentially conflicting results. Hashimoto and Tak (1977) and Hukuda and Kojima (2002) failed to identify differences in measurement of the anteroposterior diameter of Japanese male and female vertebral foramina. These results are in contrast with those of Payne and Spillane (1957). They identified small variations in the anteroposterior diameter of the cervical vertebral foramina in a sample of males and female from the U.K. but did not include detail on the statistical significance of the differences, making the significance of the differences difficult to interpret. However, Tatarek (2005) (using African American and Caucasian samples) and Lim and Wong (2004) (using a Chinese sample) demonstrated a statistically significant difference in the anteroposterior diameter of the vertebral foramen and identified the difference as being sexually dimorphic. The data for these studies was collected from different populations, which may have led to the contrasting outcomes. It has been identified that there is a need to identify population specific methods of sex estimation (Spradley and Jantz, 2011). The studies described above demonstrate that different populations may give different results for the same measurement, supporting the argument of Spradley and Jantz (2011). Kajanoja (1966) and Williams (1987) have both reported that discriminant functions for sex determination developed for one population can lead to incorrect sex determination of 32-48% of individuals from a different population.

Amores et al. (2014) investigated sexual dimorphism in C7 and T12 from San Jose cemetery in Granada, Spain using a sample of 121 individuals documented for sex, age and cause of death. The aim of the study was to measure the degree of sexual dimorphism in a southern Spanish population and to establish the accuracy of C7 and T12 for sex estimation. The measurements taken are listed in Figure 3-1. Unlike many of the other studies of sexual dimorphism of the

33

vertebrae, both right and left measurements were taken. The majority of studies discussed in this section took measurements from the left side only, as this is considered to be the standard side to use during skeletal analysis (Buikstra and Uberlaker, 1994) and measurements are only taken from the right side in the absence of the left. This does not allow for the effects of vertebral asymmetry and the potential effects of hand preference to be taken into account (See section 5.7.7). Amores et al (2014) did not find any significant asymmetry in their measurements but did find significant differences between males and females in terms of size of most measurements with the exception of the length of the superior and inferior facets of C7 and the inferior facets of T12. They identified an accuracy rate of 65.5-70.1% for univariate analysis but this increased to 80% when combined in a multivariate function. They emphasised the need to apply discriminant function equations in populations with similar characteristics.



Figure 3-1 Measurements from C7 and T12 (Amores et al., 2014).

Studies of the vertebral body dimensions in the cervical spine have reported differences in vertebral body height and anteroposterior measurements that relate to sex differences (Katz et al., 1973; Liguoro et al., 1994; Kwon, 2004). The sex differences seen in vertebral body measurements can be seen at other levels of the spine as identified below.

Bastir et al. (2014) investigated sexual dimorphism of the thoracic spine in relation to respiratory function using data collected from a skeletal collection curated at the School of Legal Medicine,

Madrid. The sample was of known sex and age at death. They assessed size and shape of the first 10 thoracic vertebrae using 3-D geometric morphometric analysis and identified that there was significant evidence for sexual dimorphism in size and shape of the vertebrae with transverse processes orientated dorsally in males and ventrally in females, the neural canal is relatively larger and more circular in females, the facets for articulation with the rib-tubercles are slightly more superiorly orientated in females leading to greater cranio-caudal inclination of the ribs in females. In the lower thoracic region only, the female spinous processes have a more horizontal (superior) orientation. The difference in regional factors may be related to the greater respiratory capacities seen in males (Bellamare et al., 2003) and the increased lordotic obstetric adaptation in females proposed by Whitcome et al. (2007) and Masharawi et al. (2010).

The posterior transverse diameter, the anteroposterior diameter and the anterior transverse diameter of vertebral bodies of the thoraco-lumbar transition area (T11, T12, L1) from 207 adults from Spitalfields Cemetery in London 18th century was assessed for sexual dimorphism by MacLauglin and Oldale (1992). They were able to demonstrate sexual variation with all measurements achieving a minimum reliability of accuracy of 70% and the anterior transverse measurement of T12 achieving 87% accuracy for correct classification of males and females. Pastor (2005) also investigated T12 and L1 vertebral levels and compared samples from the Spitalfields and Terry collections and demonstrated that out of 12 measurements, 8 showed significant sex variation allowing for accuracy levels in sex prediction of between 76-91%. Eisenstein (1983) achieved similar results in identifying sexual variation of the mid-sagittal and transverse diameters of the vertebral foramen and body. Yu et al. (2008) and Hou et al. (2012) investigated the suitability of T12 for the estimation of sex. Yu et al. (2008) took 33 linear measurements and derived two ratios from 102 T12 vertebrae listed on the Digital Korean database (a collection of 3D models of whole skeletons of Korean individuals created from computed tomography of donated cadavers). They identified 23 sexually dimorphic measurements and were able to predict sex with 62.7-85.3% accuracy using discriminant function equations. They noted that the coronal dimensions of the vertebral body (superior and inferior coronal diameter of endplate and superior and inferior maximum coronal diameter of endplate) were the most sexually dimorphic features of T12. Hou et al. (2012) gathered similar data from the T12 of 141 three dimensionally reconstructed vertebrae. They acknowledged that that metric elements of humans are population specific and their aim was to develop population specific functions for sex determination based on T12 in a contemporary north-eastern Chinese population. Of their 30 linear measurements, 28 were found to be sexually dimorphic. They were able to achieve 56.4-90.1% accuracy in predicting sex when using univariate discriminant function equations and when using four variables (superior maximum sagittal diameter of endplate,

35

inferior length of whole vertebra, distance between superior articular processes and the ratio of anterior/posterior height of the vertebral body) achieved an accuracy of 94%.

Zheng et al. (2012) identified 23 linear measurements and two ratios that provided a predictive accuracy of 57.1%-86.8% for L1 from a sample of 210 individuals from a Chinese population. Their measurements were from CT images rather than dry bones. They argue that this method is of similar degree of accuracy as from dry bone measurements, citing the work of Reid et al. (2008) who demonstrated that CT is as accurate as callipers and more accurate than computerised radiography. The highest precision of accuracy was calculated from the upper end plate width, the left pedicle height and middle end plate width.

A clinical magnetic resonance imaging (MRI) study of two cohorts of living individuals was used by Oura et al. (2018) to estimate sex using vertebral body measurements of L4 from a Finnish sample. The two cohorts consisted of a younger group (20-30 years) comprising 375 individuals and an older group (46+) comprising 1363 individuals. Data were collected from MRI scans, but individuals with degeneration or osteoporosis were excluded. They were able to achieve accuracies of 86.4% for the younger cohort and 82.8% for the older cohort.

There are a number of clinical studies that provide data relating to average zygapophyseal joint measurements across the whole spine, but they do not differentiate their results in terms of the age and sex of individuals being studied (Panjabi et al., 1993; Mahato, 2011). Masharawi et al. (2005) noted that facet size is greater in males than females, regardless of vertebral level and emphasised the importance of separating the data by sex and age in order to prevent misleading data being collected which increases difficulty in cross study comparisons. None of these studies, however, consider the influence of pregnancy of the size of the facet joints. Whitcombe et al. (2007) report that lumbar spine facet surface area is $14 \pm 3\%$ larger relative to vertebral body size in females. They consider that this is due to the redistribution of vertebral load during pregnancy.

In summary, a variety of different measurements have been used in sexual dimorphism studies of the vertebrae, with few focussing on the facet joints and most studies investigating specific spinal regions or vertebral levels. Although there is awareness of the population specificity of discriminant and logistic regression equations, there is less awareness of secular trends. The literature provides good evidence that the sex of an individual can be estimated from vertebrae.

3.1.1 Sex and facet joint orientation

The definition of facet angle is unclear and lacking consistency in the published literature. The literature reveals that measurement of the angle in the sagittal plane is referred to as the angle of

36

inclination (Tulsi and Hermanis, 1992), disc-facet angle (Boyle et al., 1996), the sagittal angle plane (Panjabi et al., 1993) or the transverse angle (Masharawi et al., 2008).

For the purposes of this study, the sagittal facet angle is defined as the angle between a line drawn across the lateral and medial borders of the facet and a line drawn between the junction of the lamellae and the centrum of the vertebral body i.e. along the sagittal plane (angles A and B in Figure 3-2).



Figure 3-2 Measurement of facet angle (L2 from GC100, author's image)

The orientation of the facets of typical vertebrae for each region is illustrated in Table 3-3 (overleaf). In the cervical region C1 and C2 are atypical, C1 having no vertebral body or spinous process, C2 also having no vertebral body but having an odontoid process.

As can be seen from the diagrams in Table 3-3, there is a change in orientation from the cervical to the lumbar region. The vertebral level of change between the cervical and thoracic region can occur anywhere between C4 and T1 with C6 being the most common site of transition (Pal and Routal, 2001). The level of change in angle between the thoracic and lumbar region usually occurs at T11 but can be T12 or T10 (Davis, 1955; Shinohara, 1997; Singer et al., 1988). Most of differences in transition area can be explained by natural variation ,but rarely there is a caudal or cranial shifts at the cervico-thoracic and thoracolumbar junctions, where the vertebral segment takes on the characteristics of the region above or below it (Barnes 2012). These shifts can lead to

extreme variation e.g. cervical or lumbar vertebrae having ribs attached, or mild variation e.g. mild cranial variation can cause a shift in the facet orientation of the transitional vertebrae giving rise to effects described in section 2.3.7. Spines that clearly demonstrated cranial or caudal shifts were excluded from this study.

Table 3-3 Facet orientation of typical vertebrae (author's image)

General Characteristics of Facet Orientation	Diagram of Typical Vertebra
Cervical Region: superior facets face superiorly and medially whilst inferior joints face anteriorly and laterally (Panjabi et al., 1993).	
Thoracic Region: Superior facets face posteriorly, superiorly and laterally whilst inferior facets face anteriorly, superiorly and medially (Moore and Dalley, 2006).	
Lumbar Region: Superior facets face postero-medially, whilst inferior facets face postero-laterally	

During the 1950's it was noted that there was very little published accurate information about the orientation of the facet joints at all levels of the human spinal column (Stewart, 1952) and indeed very little published on vertebral morphology. Overton and Grossman (1952) examined variations of the articular facets by measuring facet diameter, sagittal and vertical angles between C2 and C3 from 36 specimens. They noted that variations of angle of inclination from the vertical, between left and right sides were not a rare occurrence and also that variation in size between the left and right facets was observed in 44.2% of the specimens. Francis (1955) undertook a systematic study of the facet joints of 284 sets of cervical vertebrae from young adults where age,

sex and ethnicity were known. The data were collected from a documented skeletal collection curated at the Western Reserve University U.S.A. His aim was to record the natural variations that occur. He took various measurements, including largest diameters of the facet joints in planes parallel to the sagittal and frontal planes and described their orientation. His sample excluded vertebrae with signs of pathology such as degeneration. His work can be criticised in that some of the vertebrae used had suffered post-mortem damage and he stated that some of his measurements were approximate. Francis' (1955) research was aimed at identifying variation in cervical facet joints between males and females of American "White" and "Negroid" ethnicity and was not a study of sexual dimorphism. He reported that the facets from male vertebrae were generally bigger than those of females, but the results were based on descriptive statistics and the significance of the results not statistically analysed. This limits the validity of the reported results.

Although clinical studies are interested in collecting data on facet angle for morphometric studies, there are few published papers directly comparing male and female results. Many studies use male skeletons only (Pal and Routal, 2001; Masharawi et al., 2004; Masharawi et al., 2007). The reason for this is not justified within the research but it may be an attempt to control for possible differences in sex.

Boyle et al. (1996) undertook a study of the morphological changes that occur in the cervicothoracic junction of the spine, to provide information on the morphology of the region and quantify the facet orientation in both the sagittal and transverse planes, in order that anatomical variations and the effects of degeneration and trauma could be identified. Significant differences between the sexes were identified when measuring sagittal facet angles of the cervico-thoracic spine from the disarticulated skeletons of 26 male and 25 female individuals. This result conflicts with results reported by Milne (1991), who found significant differences in linear but not in angular measurements of the cervical spine. Milne hypothesised that the orientation of the facet joints with respect to the disc-facet angle in the cervical spine is greater in the upper cervical region, to allow for different degrees of rotation in the cervical spine. He did not know the age of his samples but obtained separate results for males and females.

Both Milne (1991) and Boyle's (1996) research was carried out on the skeletal collection held at the University of Western Australia. It is not clear whether the same specimens were used. They both used a zygapophyseal endplate protractor to measure the disc facet angle. Milne's (1991) work focussed on the upper cervical spine whereas Boyle (1996) focussed on the lower cervical spine at the cervico-thoracic junction. Neither researcher commented on the age at death of their

39

samples. There is evidence that facet joints reorientate with age (discussed in section 3-2) and this may account for the conflicting results achieved by these two studies.

There is little published literature on sex differences in facet orientation in the thoracic spine. Masharawi et al. (2004) published a comprehensive database of zygapophyseal joint orientation in the thoraco-lumbar vertebrae using three-dimensional measurement of 240 vertebral columns from the Hammon-Todd Collection. They measured longitudinal and transverse angles in groups classified by age, ethnicity and sex. They found no significant difference between groups in the orientation of the thoraco-lumbar facets but did note that facet size was independent of age and that males had statistically significantly larger facets than females. They did not take body size as a confounding factor into account. The thoracic vertebrae demonstrated asymmetry in orientation but they considered this to be anatomical variation. They noted that the facets at the thoraco-lumbar junction could be antero-posteriorly orientated as seen in the rest of the thoracic spine, or postero-medially orientated as seen in the lumbar spine. They recorded these differences as being natural variation and did not make any comment on thoraco-lumbar shifts as a possible reason for the variation.

Tulsi and Hermanis (1993) found a small but significant statistical difference in the sagittal angle and facet depth of the facets of the lumbar region in a clinically based study of 64 male and 48 female disarticulated spinal columns. They identified asymmetry of less than 10⁰ (i.e. considered symmetrical) in 91% of all vertebrae with the exception of L3, where asymmetry was found in 69% of male vertebrae and 71% of female vertebrae. They raised an interesting limitation to their research in that there is little or no information on the thickness or configuration of articular cartilage in living subjects and the effect this may have on the morphology of the facet surface. Archaeological samples do not tend to have articular cartilage remaining. This is advantageous in that the bony surface of the facet can be visibly examined; however, it limits the translation of results from archaeological studies into clinical research.

Whitcombe et al. (2007) report that that facet angles at L2, L3 and L4 are considerably more coronally orientated in females ($13 \pm 5\%$ when compared to males). Masharawi et al. (2010) also report a similar finding. This allows for increased resistance to shearing forces and increased lumbar lordosis caused by the weight of the developing baby.

A criticism of some of the existing research into facet orientation discussed above is that samples were not always controlled for age. It has been reported that the facets reorientate with age (Adams and Roughly, 2006), however this has been mostly reported in the lumbar region of the spine. A number of studies have identified reorientation of the lumbar facet joints with age (Love et al., 1999; Cohen and Raja, 2007; Wang and Yang, 2009). Difficulties in comparison of cross study results occur when age has not been factored into the research method and widely different methods of measurement are used. Furthermore, the effects of activity related markers on vertebrae, in particular the relationship between activity, size and angle were not taken into consideration in any of the studies described above. The orientation of the facet surfaces changes according to the functional and mechanical requirement of each vertebral level (Williams, 2000).

3.2 Age and facet size

There is evidence that vertebral morphology varies with age (Kim, 2013). To be able to test the relationship between age and facet size and angle, the age of the individuals must be estimated with a degree of precision. This is not a simple process. This section will review the brief literature related to the relationship between age and facet size and angle. The difficulties of aging an individual and the limitations of using age groups are discussed in section 5.7.3 in Chapter 5.

The literature suggests that the age at which an activity began and ceased, the frequency and intensity, duration of the activity as well as the type of mechanical load applied to the body and the particular bony element under load are all factors that may affect bone morphology in relation to age. There are contrasting reports on the effects of aging on spinal morphology. A number of studies have reported small but significant changes in spinal dimensions when correlated to age; overall shape of the lumbar body, in particular endplate transverse diameter and minimum transverse diameters increasing in size (Erikson, 1976), vertebral body anterior and posterior height and middle body breadth (Jankauskas, 1994), cervical foramen diameter (Humphries et al. 1998), body cross sectional area (Mosekilde and Mosekilde, 1990), lumbar neural canal width (Tatarek, 2001), sagittal and transverse body diameter and pedicle height (Rühli et al. 2005), anterior, middle and posterior vertebral body heights (Whitmarsh et al. 2012) and spinal canal diameter and vertebral body height (Kim, 2013). The results of these studies are in contrast to the study of vertebral heights by Black et al. (1991) who measured vertebral height dimensions from the lateral spinal x-ray of 2992 women aged between 65-70 years of age They compared their results to a similar study of premenopausal women and found that there was no significant difference in dimensions in non-fractured vertebrae.

The influence of age on vertebral facet joint orientation has been a point of debate with contrasting results being published. Wang and Jang (2009) and Jentzsch et al. (2013) identified variation in lumbar facet orientation that was linked to age, with the facets becoming more sagittal in orientation. Wang and Jang (2009) carried out a study of facet orientation with age and identified a negative correlation between coronal orientation in the facets between L4 and L5 and age, with the facets becoming more sagittally orientated. They considered this to place older

individuals at increasing risk of degenerative spondylolisthesis. Their study involved assessing CT scans from 159 men and 141 women divided into 6 age groups (1: <30 years, 2: 31-40 years, 3: 41-50 years, 4: 51-60 years, 5: 61-70 years and 6: >70 years). They did not identify any statistically significant sex-related differences in coronal orientation. Jentzsch et al. (2003) retrospectively analysed CT scan images from 418 males and 202 females in their study of facet orientation and facet joint arthritis in the lumbar spine. Whilst they differentiated between males and females for osteoarthritis prevalence, the data were combined for the sexes for results for facet orientation. They reported increased coronal orientation of the facet joints at L5/S1 in participants \leq 40 years of age and increased sagittal orientation at the same vertebral level in individuals ≥41 years of age. They did not find any statistically significant changes at other spinal levels in the lumbar spine. In contrast, Masharawi et al. (2004) and Cubuk et al. (2009) found that there was no correlation between age and facet orientation in the lumbar spine. Wang and Yang (2009), Jentzsch et al. (2003), Masharawi et al. (2004) and Çubuk et al. (2009) all collected their data from measurements of CT and x-ray images from clinical experiments on living participants. There are few if any published studies on the effects of age on the cervical and thoracic facet joints. The focus of clinical studies is on the lumbar region due to the links between facet orientation and spondylolisthesis and degenerative joint disease.

3.2.1 Age and osteophytosis

Degenerative change in the form of osteophytosis (osteophytic growth around the perimeter of the vertebral body as a compensation of the effects of degeneration of the intervertebral disc) is the most common general indicator of age in in the vertebrae (Stewart, 1966) It is visible as osteophyte formation around the edge of the articular surface of the vertebral body. Males and females show similar patterns of age-related changes in the development of vertebral body osteophytes but with females showing greater variability (Snodgrass, 2004). Snodgrass (2004) results indicated that the changes, measured on a 0-4 scale after the method described by Stewart (1973), could not be reliably used to estimate age at death, but the presence of extensive osteophytosis could help identify a vertebra from an older individual. Listi and Manheim (2012) expanded on the work done by Snodgrass (2004) to investigate the use of vertebral osteoarthritis and osteophytosis to estimate age at death, scoring the vertebral bodies and facets for degenerative change in 104 individuals from the Bass Collection. They achieved results indicating that the degree of osteophytosis is correlated with age, however the relationship is too weak to enable any more than a general estimate of age at death. Watanabe and Terazawa (2006) disagreed with the results of Snodgrass (2004) and Listi and Mannheim (2012) when they analysed the relationship between osteophyte formation and age in a Japanese population. They

formulated an 'osteophyte formation index' based on grading scores of 0-3 (no osteophyte to large osteophytes) at the anterior and bilateral surfaces of the superior and inferior margins of the vertebral body. They took the maximum score for each spinal level to use in their analysis and averaged the maximum values at each region (cervical, thoracic and lumbar) to define their 'osteophyte formation index'. Their analysis demonstrated a significant correlation between the osteophyte formation index and age at every level of the spine. This method requires data to be collected from the entire spine and therefore would not always be valid as an ageing tool in archaeological samples with poor preservation.

The increased presence of osteophytosis on the vertebral body is related to age and changes the morphology of the vertebral body but would not be accurate tool for the estimation of age at death. The evidence described above indicates that the relationship between osteophyte score and age is weak and would provide a general estimate of age when individual vertebrae are considered. The osteophyte formation index of Watanabe and Terazawa (2006) requires a complete set of data from the spine, something that would be problematic in many archaeologically derived samples.

3.2.2 Summary

In summary, there is plenty of evidence for the existence of sexually dimorphic traits in vertebrae; however, few bioarchaeological studies focus on the zygapophyseal joints in vertebral sexual dimorphism studies. Even fewer consider the sagittal angle of orientation of the facet joint as a potentially dimorphic feature. Whilst it is known that there is sexual dimorphism of angle in the lumbar spine due to the effects of pregnancy, little is known about differences between the sexes, of facet angle in other areas of the spine.

Chapter 4 Bioarchaeology and Activity

The previous chapter reviewed the literature pertaining to vertebral sexual dimorphism and the effects of aging on vertebral dimensions. Whilst it is clear that vertebrae are sexually dimorphic, all the studies reviewed excluded facets that showed signs of degeneration. Degeneration may be used as a marker of differing physical activity. Bones undergo functional adaptation as a result of the stresses and strains they are continuously subjected to. An understanding of the relationship between degree of degeneration with facet size and angle may help to explain inter-sample/population differences in facet size and angle within the samples chosen from this study as they are from differing socio-economic groups and temporal periods, i.e. differing lifestyles and activity patterns. It is therefore useful to describe the biocultural model that is an important bioarchaeological approach to creating an understanding of lifestyle. This discussion is followed by a review of the evidence that biocultural adaptation can be preserved in human bone as physiological stress markers.

This chapter continues with a discussion of bone plasticity, reviewing models of bone functional adaptation before considering biocultural features that can affect bone morphology such as diet, disease and age at death. The final section will review the use of markers of occupational stress in the reconstruction of lifestyle, in particular degenerative changes, and ends with an overview of osteoarthritis including classification, diagnostic criteria, aetiology and the relevance of vertebral osteoarthritis to bioarchaeology. Osteoarthritis of the facet joints is a particularly useful pathology when comparing populations in this study, as differences in distribution may reflect differences in culturally based activities (Jurmain and Kilgore, 1995). Robson Brown et al. (2008) consider that the diagnostic markers for degenerative joint disease are predictors of mechanical loading in life. Therefore, comparison of distribution and degree of severity of osteoarthritis will allow for comparison of physical activity and lifestyle of the samples under study and may be related to bone remodelling of the facet joints

4.1 Bioarchaeology

Bioarchaeology is an interdisciplinary endeavour focused primarily on questions of quality of life, behaviour and lifestyle, biological relatedness and population history through the study of human skeletal remains within their archaeological context. This is achieved by the integration of paleodemography, paleopathology, biological anthropology, archaeological theory and method, medical science and other related disciplines (Larsen, 1997; Knüsel 2000; Pearson and Buikstra, 2006). The importance of bioarchaeology in the understanding of the past has only been realised in fairly recent times, with the term "bioarchaeology" deriving independently in the United Kingdom and United States in the 1970's. The two derivations had distinct definitions. Clark (1972) first used the word in the title of his study "*Starr Carr: A Case Study in Bioarchaeology*" to describe inferences from the study of faunal remains from archaeological contexts. Clark re-worded the definition in 1973 by describing bioarchaeology as:

"The archaeology concerned first and foremost with life" (p. 464) and

"The archaeology of how men occupied territories and maintained life" (p. 466).

After being influenced by the "New Archaeology" proposed by Binford (1962), Jane Buikstra independently invented the term "Bioarchaeology" in 1977 in reference to the merging of physical anthropological methods with archaeology. She describes it simply as:

"The contextual analysis of human populations from archaeological sites" (p. 67).

She identified the need to use this new integrated approach of interdisciplinary and cross-cultural research tools to generate and solve research questions about how past populations lived in contrast to the descriptive skeletal studies previously used that only focused on individuals and the osteological and paleopathological aspects of medical and forensic examination. The fundamental need for an integrated approach is supported in the following quotation:

".... we cannot take an empiricist view and assume that the osteological data speak for themselves.... as the body is simultaneously biological, representational and material". (Sofaer, 2006:11).

This integrated approach can identify life experiences that affect human biology but may not be apparent in the archaeological or historical evidence (Swedlund and Herring, 2003; Sofaer, 2006). It adds emphasis to the importance of analysing the social process underlying biological variation such as chronological and physical aging, the sexual division of labour and access to adequate nutrition.

The main approach to bioarchaeology is through bioculturalism. The biocultural approach in bioarchaeology is described as problem orientated, examining the interrelationships between humans and their larger social, cultural and physical environments, (i.e. population based) (Buikstra, 2006). It provides a method for cultural comparison and can be applied to individuals and populations throughout time and place (Armelagos, 2008). It is explicit in the emphasis between interaction of humans with their physical, social and cultural environments (Dufour,

Chapter 4 Bioarchaeology and Activity

2006), with biology and culture being dialectically entwined (Levins and Lewontin, 1895).

Biocultural adaptation is preserved in the composition of the bones as human variability that develops as a function of the body's response to environmental factors and is also recorded in the context of the deposition of the remains (DiGangi and Moore 2013). Recognition of physiological stress marks on the skeleton has been a landmark in biocultural study, becoming an important area for research (Buikstra, 1991). Studying human remains from a biocultural perspective can help to contribute to an understanding of the life and health of that individual in the context of a larger society by the emulation of the real world in order to understand the environmental and cultural context of the population under investigation (Buikstra, 1977:82; Goodman and Leatherman, 1998:228).

There has been much criticism of the biocultural approach and the emphasis it places on the analysis of the social processes that affect biological variation, particularly the "adaptionist" program and the tendency to naturalise social process (Orlove, 1980; Singer, 1996). Segal and Yanagisako (2005) mounted a challenge against bioculturalism, arguing that there should be increased specialisation and separation between the sub disciplines to limit biological and adaptationist interpretation over socioculturalism. Zuckerman and Armelagos (2011) countered their criticism arguing that the separation and individualisation of methods suggested by Segal and Yanagisako is a misinterpretation of the basic principles of the biocultural approach and is a retrograde step in the development of bioarchaeology.

Buikstra (1991) identified that the study of physiological stress marks on the body is important in biocultural studies of archaeological skeletal remains. The next section will discuss evidence left on the skeleton from the effect of the functional adaptation of bone to external stimuli due to their inherent plasticity.

4.2 Bone plasticity and remodelling

This section explores the ability of bone to remodel in response to environmental change. It begins with an overview of the concept of bone plasticity and then reviews models of bone remodelling and ends with discussion of the effect of remodelling on articular surfaces.

4.2.1 Plasticity

Plasticity allows the bones to record the various roles held by an individual during life and provides a link between the osteological study of populations and individual skeletons. The effects of bone plasticity are manifest as the varied overall morphology of skeletons in the archaeological
record and can act as an indicator of skeletal health, disease and life course (Agarwal and Beauchesne, 2013). Levins and Lewontin (1985) viewed human variability as a responsiveness to environmental factors that both mediate and produce each other i.e. biology (genetics) and culture are held as dialectically intertwined. Individuals do not have homogeneity of life from birth to death, but their life course is defined by their sex, status, occupation and environment, all of which will result in variation in the underlying skeleton (Larsen, 1997; Judd and Redfern, 2012).

Numerous non-mechanical factors such as age- or sex-related differences as well as changes elicited by biomechanical demands are thought to have an effect on bones (Meyer et al., 2011; Milella et al. 2012). The most productive studies of human health and disease in the past have examined patterns of pathology "by age, sex and environmental setting" (Armelagos and Gerven, 2003:59). In order to do this, it is necessary to be able to accurately estimate the sex and age at death of individuals. The sex of an individual may predicate on certain behaviours or abilities and may be used to test for social role when reconstructing behaviour (Grauer and Stuart-Macadam, 1998). The effect of life and lifestyle on bone morphology should be considered when estimating the sex of an individual using the non-sex role specific bones (i.e. bones that are not related to physical differentiation relative to reproduction e.g. pelvis) such as the upper and lower limbs, similarly the sex of an individual should be considered when analysing lifestyle. It is thought that differing activity patterns between the sexes can have an effect on sexual dimorphism of bone dimensions. There are a number of studies that have demonstrated significant difference in long bone diaphyseal shape between males and females (Ruff, 1987, 1992, 2000; Bridges, 1989, 1991, 1993; Stock and Pfeiffer, 2001; Weiss, 2003; Wescott, 2006; Carlson et al., 2007; Wanner et al., 2007). Ruff (2000) reported changes in lower limb diaphyseal cross-section through time, as a reflection of differences in physical activity between male and females and used the patterns of sexual dimorphism identified to provide an understanding of the gendered division of labour within populations. Stock and Pfeiffer (2004) identified that sex differences in upper limb robustness and diaphyseal shape could be used as a measure of subsistence patterns and labour intensification. In a study by Pomeroy and Zakrzewski (2009) significant differences in the shape of lower limb diaphyses in a medieval Muslim population from Éjica, Spain were observed which supported documentary and osteological evidence of a marked gender difference in activity. These examples support the understanding that functional adaptation of bone to external stimuli can be used to identify gendered division of labour.

However, Thomas (2014) warns that intra-population sexual dimorphism that has been interpreted as resulting from the gendered division of labour may actually be a reflection of intrinsic dimorphism rather than cultural or population-specific idiosyncrasies. He recommends

that the sample to be analysed should be of assessable sex and age at death and should come from individuals subject to similar environmental influences.

The concept of bone plasticity in relation to activity is fundamental to bioarchaeology (Skerry, 2006). Interest in bone morphology developed due to the realisation that bone is a plastic material that functionally adapts to its environment particularly in response to physical activity. Early definitions of plasticity were based on the concept that human morphology appeared to be malleable during growth and development (Bogin, 1995).

Lasker (1969) redefined the concept of plasticity by separating behavioural, short termphysiological and developmental plasticity from "genetic" adaptation leading to the identification of three modes of adaptation:

- Natural selection where the selection of genotypes directly influences the genetic constitution of the population, e.g. population differences in height as a result of genetic differences between those populations (Beall and Steegman, 2000).
- Adaptive plasticity where an individual adapts to the immediate environment in a behavioural response e.g. professional tennis players demonstrate high levels of humeral asymmetry which corresponds with the timing and duration of their participation (Jones et al., 1997).
- 3) Developmental or ontogenic adaptation. Ontogenetic modification is a plastic response to growth and development that occurs in response to the environment and is not reversible. Baker (1984) considers it to be a "biological or cultural trait which aids the biological functioning of a population in a given environment" e.g. increased lower/upper limb bone length and strength proportions in adult humans occurs as an adaptation to bipedalism (Ruff, 2003b).

The morphology of each individual is formed by a combination of their genetic make-up and variation in development as a result of plasticity as an effect of environment and behaviour (Wells and Stock, 2007).

One of the best-documented effects of adaptive plasticity is the influence of habitual activity on skeletal morphology. Roberts (1995:1) defines adaptive plasticity as "the capability of being moulded" and describes it as a functional adaptation to the environment by both soft tissues and the skeleton, where the environment encompasses natural and cultural phenomena. Sofaer (2006:77) considers that the inherent plasticity of tissue is key to the body's materiality, which is brought into being as the body records the accumulated experience of the life course of skeletal responses to culturally defined activities in conjunction with biological change. Martin (2013) agrees with this and comments that the body is defined by the social and cultural influences that

shape the expression of the physical characteristics and not simply the effects of genes and biology.

The understanding of skeletal response to activity is based on the understanding that bone responds to mechanical stress by stimulating the deposition of new bone in areas of strain caused by mechanical loading. This deposition causes trabecular and cortical bone hypertrophy, which counteracts the loading (Shaw and Stock, 2009).

Functional adaptation is a key theme in this thesis. The next section will review four models of bone remodeling (functional adaptation) to demonstrate the development of theories relating to this subject.

4.2.2 Models of bone functional adaptation

During life, the skeleton is exposed to many continuously recurring loads that lead to changes in morphology (modelling) (Prescher, 1998). The plasticity of bone allows it to adapt to external forces from the mechanical environment acting upon it by modelling and remodelling its internal and external structure in the direction of functional stress as a response to mechanical stimulus (Pearson and Lieberman, 2004; Ruff et al., 2006: Adams et al., 2006). There are various theories used to explain the process of bone functional adaptation which are discussed after a brief description of the concepts of bone modelling and remodelling which are described below.

4.2.2.1 Bone modelling

Bone modelling is a description of the process whereby bones are shaped or reshaped by the independent action of osteoblasts and osteoclasts. In contrast the activities of osteoclasts and osteoblasts are coupled in bone remodelling (Langdahl et al., 2016). Modelling defines skeletal development and growth and is responsible for the shaping of bones with adaptation to strain in adults leading to modelling of bone (Taddei et al., 2009). Physical activity can stimulate bone modelling (Kontilainen et al., 2002). It is also controlled by other factors such as modelling-based bone formation seen in the ribs which are not axially loaded (Ominsky et al., 2015). It is thought that it is controlled by genetic factors in combination with environmental factors such as physical strain and probably hormonal factors as it has been demonstrated that the parathyroid hormone (PTH) and inhibition of sclerostin can stimulate modelling-based bone formation (Lindsay et al. 2006; Ominsky et al. 2014).

4.2.2.2 Bone remodelling

In contrast to bone modelling, remodelling renews the adult skeleton approximately every 10 years (Manolagas, 2000). Remodelling is characterised by four stages:

- 1) Activation stage osteoclasts are recruited
- 2) Resorption phase osteoclasts resorb bone
- 3) Reversal phase- osteoclasts undergo apoptosis and osteoblasts are recruited
- 4) Formation phase osteoblasts lay down new bone matrix that subsequently mineralizes (Langdahl et al., 2016).

Dempster et al. (2002) define bone remodelling as a process where osteoclasts and osteoblasts work sequentially in the same bone remodelling unit (BMU). It is most prominent on cancellous bone surfaces and 80% of bone remodelling takes place here (cancellous bone only comprises 20% of bone) (Seeman, 2013). Cortical remodelling increases in importance with age as cancellous bone is lost and remodelling activity in both compartments increases with cortical bone remodelling at the periosteal and endocortical surfaces (Bliziotes et al., 2006).

Remodelling has several functions:

- Replacement of old and damaged bones with new bone. This maintains the mechanical strength of bone however, excessive remodelling and repair poses a risk to bone strength as it destabilizes bone and introduces stress concentrators (Dempster, 1997). Excessive strain on a bone can cause microdamage which can lead to targeted remodelling to remove the damaged bone and increase the volume of the surrounding undamaged tissue leading to a cycle of damage and repair (Allen and Burr, 2008).
- Calcium homeostasis. Clinical evidence indicates that individuals with low calcium intakes are more prone to adult bone loss as bone acts as a reservoir for calcium (Langdahl et al., 2016).
- Maintenance of acid/base balance and release of growth factors embedded in bone (Moore and Dalley, 2006).
- 4) Provides a reservoir of labile mineral (short-term homeostasis) (Dempster, 2006)
- Only mechanism by which old, dying or dead osteocytes can be replaced (Dempster, 2006).

The next section discusses theories of bone modelling and remodelling.

1) Wolff's Law

Wolff's Law states:

"The form of a bone being given, the bone elements place or displace themselves in the direction of the functional pressure and increase or decrease their mass to reflect the amount of functional pressure" (Wolff, 1892)

Wolff was a 19th Century German orthopaedic surgeon who applied principles of mechanics to identify this law of transformation and used it to describe the ability of bone to resist stress by its response to mechanical forces acting upon it (modelling). Bone adaptation can take the form of increased or decreased mass depending on whether there is an increase or decrease in stress upon it. If the stress is removed and inactivity occurs, there is resorption of bone until it returns to the original strain levels (Ruff et al., 2006).

Wolff's Law (1892) is comprised of three major tenets: bone must be strong enough for support but light enough for locomotion, the trabeculae in the bone align themselves along the direction of principle strain, and that this is accomplished by self-regulating mechanisms as the bone responds to mechanical loads (Pearson and Lieberman, 2004). Wolff's Law has been widely used within bioarchaeology to explain the adaptation of bone to mechanical stress. However following criticism of unquestioning acceptance of Wolff's Law, the term "functional adaptation" is considered by some to be more appropriate (Bertram and Swartz, 1991; Carter et al., 1991; Cowin, 2004; Pearson and Lieberman, 2004).

There has been considerable criticism of the tenets proposed by Wolff. Pearson and Lieberman (2004) state that Wolff's Law is not a "law" and that it is over simplistic because it fails to consider the multiple processes that occur simultaneously and that these processes should be considered separately. They question the extent of genetic versus environmental influences on bone shape but consider that in the immature skeleton, the primary influence on cortical bone modelling is mechanical strain. This is supported by evidence from other research that has shown that epiphyses and articular surfaces respond to mechanical forces during growth (Frost, 1979, 1994, 1999; Hamrick, 1999; Plochocki and Organ, 2003).

Bertram and Swartz (1991) undertook a critical literature review to provide an analysis of some of the primary evidence to support Wolff's law and its mode of action. They considered that there was lack of control for factors that could potentially affect the outcome of many of the experiments leading to confused interpretation of the results. They identified that it is impossible to determine the extent of mechanical influence on bone form until all other factors are taken

into consideration. To this effect, they highlighted areas of Wolff's Law where other factors may influence the outcome. These are listed in Table 4-1.

Wolff's law is commonly cited in reference to bone modeling as a response to mechanical stimuli. Bertram and Swartz's (1991) critique of Wolff's tenets supports the proposition from many researchers that the phrase '*bone functional adaptation*' is a more accurate representation of the bone modeling process. Wolff's Law applies specific mathematical rules governing the modeling process in bone (where bone is modeled as being a solid, homogenous and isotropic substance subjected to statically applied forces) and is based on the assumption that there is a simple cause and effect sequence between mechanical loading and bone shape. However, this is now identified as being based on engineering and biomechanical misconceptions (Cowin, 2001) and does not accurately reflect or predict bone remodelling and therefore is a less suitable model. There are a number of reasons for it to be a less suitable model e.g. the function of bone is to provide support and protection for the internal structures of the body, it serves to store calcium, it is a site of production of red blood cells as well as being involved in movement (Currey, 1984). The Wolff model can be criticised for not considering these factors or other factors such as the effects of hormones, nutritional status, age and genetics (Lieberman, 1996).

Bertram and Swartz's Response
This is true only in growing bone
Sensitivity to bone resorption is dependent on age (in youth, bone
resorption can be reversed, with age bone resorption can lead to
osteoporosis). Rates of modelling and remodelling are not consistent for all
locations within the body with a "load threshold" triggering response.
Modelling occurs in areas of bone that are undergoing physical strain whilst
remodelling is a balanced process that occurs throughout the skeleton and
particularly in areas of microdamage to effect repair of the bone.
This does not account for the systemic bone hypertrophy seen in some
individuals. Atrophy is not the reverse of hypertrophy. The localised
response of bones in terms of apposition and resorption reflect the
biomechanics of the whole bone

Table 4-1 Summary of Bertram and Swartz response to Wolff's tenets

2) Frost's mechanostat model

This model is built upon the hypothesis that a negative feedback, homeostatic loop regulates bone modelling (Frost, 1987, 1990a, 1990b, 2003). Frost (1987) considers that the effects of mechanical strain will only induce changes in bone strength when a particular threshold is exceeded and that the range for optimal strain is genetically determined. No bone response will be seen if the mechanical load remains in the range of customary strain. Deposition to strengthen bone occurs when this range is exceeded, increasing the strength of the bone and thereby reducing the level of strain to within the customary range. Conversely, if there is a reduction in strain below the customary strain level, bone resorption occurs, and equilibrium returns. Frost defines this mechanism as the mechanostat. A weakness of Frost's model is that it does not fully consider the modulation effect of non-mechanical factors (including hormones and other humeral agents) on the mechanostat effect.

3) Turner's mechanotransduction model

Turner (1998) refined Frost's mechanostat model to create an understanding of localised responses to loading at a cellular level in bone. He describes a process of mechanotransduction whereby mechanical loads are sensed by osteocytes which release signals that activate or inhibit the action of osteoblasts and/or osteoclasts (Klein-Nulend et al. 2013). This process offers an explanation for structural change at the tissue level (Rauch and Schönau, (2001).

The mechanotransduction model identifies three rules that govern bone regulation (Turner, 1998):

a) Bone responds to dynamic rather than static loads (the magnitude of the loading, the loading frequency and the strain rate are all important factors in the determination of bone remodelling response) (Rubin and Lanyon, 1984; Forwood and Turner, 1995; Turner et al., 1995).

b) Short durations of loading are sufficient to stimulate a response; however, osteocytes can become desensitised to the leading stimulus if the loading duration increases (Rubin and Lanyon, 1994), with mechanosensitivity beginning to decline soon after the stimulus is initiated. Robling et al. (2006) were able to measure a 95% decrease in mechanosensitivity in rats after only 20 loading cycles. Brief resting between cycles of loading allows the osteocytes to recover sensitivity and has been shown to maximise the osteogenic effects of mechanical loading in the rat (Robling et al., 2001). This model may translate to humans.

c) A routine loading pattern reduces osteocyte response. Structural changes occur as a response to abnormal strain and not normal loading.

Turner's (1998) mechanotransduction model presents a good explanation for cellular level response to mechanical loading but does not identify the actual events that convert mechanical stress to a biochemical event. It also fails to identify whether specific stresses (e.g. stretch versus shear) will lead to different responses (Chen, 2008).

4) Carter and Beaupré's mechanical model

Carter and Beaupré (2001) developed the models proposed by Frost (1990a, 1990b) and Turner (1998) by incorporating the influencing effects of developmental factors. Their model identified biological and mechanobiological components in bone growth. Both components are necessary for normal bone formation, but the relative importance of the two factors changes over the course of development. The biological component recognises the intrinsic growth processes controlled by genes, hormones and other metabolic factors, whilst the mechanobiological component factors in the changes that occur in bone as a response to mechanical loading. With increasing age, the importance of the mechanobiological influence increases and the biological response decreases. A weakness of this paradigm is that it was modelled on healthy adults with the assumption that biological influences are constant and that it only represents net changes in bone without consideration of the separate activities of osteoblasts and osteoclasts (Hernandez et al., 2000).

The second to fourth models of bone remodelling presented here all have roots that date back to the work of Wolff. They each add further explanation to the effect that mechanical loading has on tissue formation in bone. Vertebral sexual dimorphism is thought to result in part from population specific activity (Gilsantz et al., 1997). The next section discusses the effect of mechanical loading (physical activity) on bone morphology.

4.2.3 Mechanical stress and bone remodelling

Physical activity is known to influence bone through dynamic biomechanical loading, especially when performed on a regular basis (Turner and Robling, 2003). There are strong links between mechanical stress from physical activity and changes in bone morphology and dimensions leading to increased cross-sectional size and bone density in both long and irregular bones (Mosekilde, 2000; Kontulainen et al., 2003; Martyn-St James and Carroll, 2010; Bolam et al., 2013). The converse effect has also been reported with low levels of physical activity being correlated to diminished vertebral size (Gilsanz et al., 1994). Bones respond to loading in a site-specific manner (Kontulainen et al., 2003). Nilsson et al. (2014) suggest that the morphological mechanisms that occur due to mechanical stress-induced changes in bone size are an increase in cortical bone thickness (prevalent during the growth period) and a decrease in endosteal bone loss (prevalent

in the aging population). Ruff et al. (2006) agree with this and report that increased mechanical loading mainly affects internal bone remodelling, with growth towards the medullary cavity. Ruff (2005) reports that estimation of adult activity from the external dimensions of a long bone is fraught with difficulty unless the internal dimensions of the bone are also considered. Mechanical loading triggers an increase in osteoblastic activity leading to change in morphology (Lieverse et al., 2011). This manifests as changes in robusticity of long bones (defined as the thickness of the shaft to its length (Martin and Saller, 1957) and quantified by cross-sectional cortical geometry (Ruff, 2000). The process of bone adaptation is complicated and not fully understood (Pearson and Lieberman, 2004), however, there is still a considerable body of evidence for a direct link between mechanical loading and the geometric properties of long bone diaphyses (Robling et al., 2000; Daly et al., 2004; Warden et al., 2005; Ruff, et al., 2006) and vertebral size (Gilsanz et al., 1994; Mosekilde and Mosekilde, 1990; Mosekilde, 2000).

The relationship between mechanical loading and functional adaptation of bone has been used to identify changes in behaviour in populations, with an increased robusticity considered to develop as a consequence of strenuous physical activity (Bridges et al., 2000; Steckel and Rose, 2002; Stock and Pfeiffer, 2004), enabling interpretation of patterns of prehistoric behaviour by measuring skeletal robusticity (Stock and Pfeiffer, 2001, 2004; Stock et al., 2011). The effects of bone remodelling on articular surfaces as a result of adaptive plasticity are reviewed in the next section.

4.2.4 Remodelling of articular surfaces

The effect of physical activity on remodeling of the facet joints is a key focus within this thesis as this may relate to the degree of sexual dimorphism seen. This section reviews the effects of remodelling of articular surface areas as a response to physical activity.

Within the context of bioarchaeology, bone remodelling has been considered to be a good indicator of skeletal response to repeated activity-induced stress in humans, with bone plasticity recording the body's response to social and environmental factors as an osteological record of life experiences (Sofaer, 2006:77). Bones respond to physical stress by remodelling their internal and external structure in the direction of the functional stress (Ruff et al., 2006). From a biomechanical perspective, the response of an articular surface to loading can have important behavioural implications in relation to the amount of force a joint undergoes in an individual's lifetime. However, there is conflicting evidence for the argument that activity and loading can affect remodelling and asymmetry in articular surfaces. Ruff (1992) identified that the cross-sectional area of the femoral midshaft from a population from Peco Pueblo increased significantly over time but the size of the femoral head did not follow the same pattern. The femoral head

correlated most closely with the body mass of the individual at age 18, whilst the femoral midshaft measurement correlated most strongly with the mass of the individual at the time of study. This led Ruff (1992) to theorise that articular surfaces are under mechanical constraint and do not demonstrate the same plasticity as long bone diaphyses. This result was confirmed by Lieberman et al. (2001), who investigated the effect of mechanical loading to articular surfaces in the developing skeleton to measure the phenotypic plasticity of the articular surface. He did this by comparing two groups of young sheep. One group was exercised on a treadmill and the others were non-exercised. There were no significant differences in body mass between the two groups.

He failed to find a significant difference in articular surface area once control for body size was undertaken. The exercised sheep demonstrated a significant increase on cross sectional size of the diaphysis with the most pronounced differences occurring in the juvenile group. This study confirmed Ruff's findings that the articular surfaces are highly constrained in their ability to macroscopically remodel.

Plochocki and Organ (2003) were able to offer a contrast to the conclusions drawn by Lieberman et al. (2001) and Ruff (1988) when they found levels of asymmetry on the articular surfaces of humeri, radii, femorae and tibiae from two archaeological sites in Missouri. Their data demonstrated a right-sided dominance in every joint dimension of the upper body except the distal radial articulation. They concluded that limb articular facets are adapted to the mechanical environment and that biomechanical factors are important in the development and maintenance of articular morphology.

Meyer et al. (2011) argued against the method used by Lieberman et al. (2001) saying that sheep do not provide an appropriate model to assess the effect of activity on bone in humans, as unlike humans, sheep are quadruped and use all for limbs for locomotion. They agree with Knüsel et al., (1997) that upper limbs in humans portray a clearer picture of the effects of increased strain on bone when measuring asymmetry due to the effects of differences in recruitment of right and left arms in the performance of occupational activities, as they are free from confounding factors related to locomotion. Bilateral asymmetry of the upper limb (particularly the humerus) can also be used to measure hand preference (Stirland, 1993; Steele and Mays, 1995; Steele, 2000; Blackburn and Knüsel, 2006). The lower limbs show less asymmetry as there is similar recruitment in the act of bipedal locomotion (Auerbach and Ruff, 2006). Bipedalism is also linked to significant increase in the cephalad to caudad size of vertebral bodies and associated facet joints (Nakatsukasa et al., 1994; Whitcombe et al., 2007). The degree to which the adaptation is reflected in facet asymmetry is not clear from the existing literature The macroscopic effects of bone remodelling and the challenges of interpretation have been identified above. However, not all bone remodelling is macroscopic; the next section reviews the microscopic changes that also occur. Barak et al. (2011) were able to identify microscopic changes in the articular surface within the tarsal joint as a result of remodelling. Their study used a sheep model to test the effects of induced changes in tarsal joint loading on corresponding adjustments in trabecular orientation in the cortical bone of the joints. Their research indicated that trabecular orientation is very sensitive to loading and effects were observed after a short duration of experimentation (34 days), although they comment that their study was subject to limitations of small sample size and duration. This result strongly supports Wolff's hypothesis (discussed in section 4.2.2) that trabecular bone structure can adjust to follow the principle directions of stress caused by the effect of external loads upon a joint.

The relationship between the degree of physical loading (as assessed by the degree of degenerative change undergone by the joint surface, see section 4.3.4) and the size and orientation of the facet will be measured as part of this research to test whether this influences the degree of sexual dimorphism in the facets. If bone alters in response to the loading it undergoes during activity, then the changes undergone by the articular surface area may provide information about the effect that behavioural variations may have on load-induced degeneration of the joints (Bridges, 1991; Spector et al., 1996; Kerrigan et al., 1998). Although Plochocki and Organ (2003) considered that limb articular facets responded to mechanical loading, there is significant evidence is to the contrary. Research has demonstrated that articular loading in adults can have a profound effect on subchondral bone and trabecular orientation inferior to the joint surface but that these changes do not affect the external articular dimensions (Pauwels, 1987; Poss, 1984; Radin et al., 1982, 1984; Rafferty and Ruff, 1994; Ruff, 2000; Barak et al., 2011; DeSilva and Devlin, 2012; Su et al., 2013). From this it could be suggested that different loading patterns will affect the trabecular architecture of zygapophyseal joints at different vertebral levels with varied degrees of orientation. This is an area for further research and beyond the remit for this study.

Although the studies described above all link changing bone morphology with mechanical loading, other factors also need to be taken into account. The relationship between sex, age and bone morphology (focussing specifically on the vertebrae) are factors for consideration within this study.

4.3 Reconstruction of lifestyle using markers of occupational stress

The previous section reviewed literature pertaining to articular surface remodeling and eluded to the use of articular degenerative change as a measure of degree of mechanical loading/physical activity undertaken by an individual. Inter-sample differences in facet size and orientation may be linked to differing activity levels and therefore the use of degenerative changes as a measure of activity may add insight into variations observed. The use of markers of occupational stress (MOS) including the role of degenerative change is reviewed in this section. Each marker is reviewed from a bioarchaeological perspective to demonstrate their use in providing information about physical activity in past populations.

4.3.1 Activity related changes

Knüsel (2000) considered that research into activity-related change is a developing area in human osteology. However, interest in the study of linking skeletal markers to activity is thought to have developed from within the context of study of the impact of occupation on the body (Buikstra and Beck, 2006). Early theories linking disease and occupational stress were considered from as early as the fifteenth century in work by Paracelsus (1493-1541). He identified a link between respiratory disease and occupation in miners (Kennedy, 1989). Ramazzi published the first systematic work on industrial medicine in 1700. He realised that recognition of an individual's occupation was a source for understanding problems of health (Kennedy, 1989). In 1887 William Arbuthnot Lane observed:

"When we find a condition of the skeleton differing from the normal and obviously not the result of disease, I think we are justified in concluding that the variation must have come from the performance of some purpose or function in addition to those normally performed during the life of that individual alone" (Lane, 1887: 586).

The interesting part of this quote is *"in addition to"*. Lane is identifying an individual as being different from their contemporaries as a result of the effects of physical activity on their skeleton that are not undertaken by others in the same subset of population.

When attempts have been made to create a theoretical link between physical activity and changes in bone morphology, emphasis has been placed on the plasticity of skeletal material as an adaptive response to the external forces of the mechanical environment acting upon it (Rubin et al., 1990; Goodship and Cunningham, 2001; Preuschoft, 2004). The emphasis on plasticity influenced a number of early anthropologists, including Boas, Martin and Weidenreich (Buikstra and Beck, 2006). It was soon realised that information from the skeleton itself was not enough to

identify specific activities but that combining evidence from archaeological contexts and human remains is an important tool in the reconstruction of general patterns and levels of activity in historic populations. This is still a dominant paradigm today (al-Oumaoui et al., 2004; Molnar, 2010, Cardoso and Henderson 2010).

4.3.2 Markers of occupational stress

Markers of occupational stress (MOS) have been used to explore inter-sample differences in prevalence and severity of stress markers to gain information about the arrival of agriculture (Marchi et al., 2006; Lieverse et al., 2007), sexual division of labour (Sofaer Deverenski, 2000) and economic intensification (Klaus et al., 2009). The successful use of MOS as activity indicators relies on the presumption that there is a close association between repetitive activity and skeletal change. Repetitive activities can lead to bone remodeling as an effect of mechanical loading and this is thought to be a good indicator of lifestyle in archaeological samples (Kennedy, 1989).

A number of different methods and procedures for the assessment of MOS, have been used to study behaviour and lifestyle of past people (Larsen, 1997; Cox and Mays, 2000; Katzenberg and Saunders, 2000; Buikstra and Beck, 2006, Jurmain et al., 2012). One such method is the examination of gross pathological changes in the skeleton, which involves study of bony changes such as enthesophytes, osteoarthritis, spondylolysis, os acromiale and osteochondritis dessicans (Ponce, 2010). These MOS are general indicators of joint use and can document long-term physical stresses, habitual movements and general motions on the body of an individual (Larsen, 1997; Knüsel, 2000).

Jurmain (1990) severely criticized the use of MOS and has questioned their validity, due in part to the difficulty in quantifying alterations in the bones. It is not easy to decide whether changes observed are caused by repeated activity or idiopathic. It is also difficult to identify a primary cause for the changes seen (Jurmain et al., 2012; Weiss et al., 2012). There are a number of confounding factors that can affect the expression of repeated activity on bone. These include trauma and micro trauma, robusticity, age, sex, genetic predisposition or poor manual handling techniques (Jurmain, 1990). Experimental control for the external factors that can affect the prevalence and severity of bone changes leads to improved comparison within and between sample populations increasing the validity of activity-related studies (Ostendorf Smith, 2013). It should be remembered that whilst bioarchaeological analysis can provide a great deal of information about bones, without context this information would be limited (Buikstra and Beck, 2006). Comparative population based descriptive studies have been published in the latter part of the 20th century (Pearson and Buikstra, 2006) but more work on MOS in past and present populations is needed to help identify which activities are responsible for the development of the

features observable in skeletal populations (Buikstra and Beck, 2006) and to identify further MOS that will support or refute existing results. The use of MOS as markers of activity is subject to a number of limitations and these are discussed in section 4.3.3

4.3.3 Limitations in the use of MOS as markers of activity

It should be recognized that there are limitations to the use of MOS in the reconstruction of the activity patterns of past populations:

- 1) The inconsistent use of scoring systems to record MOS makes cross-study comparison challenging. For example, at least three methods for recording enthesopathies have been recorded. Angel et al. (1987) used a scoring system with stages from "absent" to "showing strong development". Crubézy (1988) measured entheseal change with a four-stage scoring system based on the dimensions of each exostosis at entheseal sites. Vilotte et al. (2010) identified a methodological issue when they noted that entheses could be a useful tool for the identification of occupational stress provided that the correct entheses are measured using appropriate methods. Hawkey (1988) used a classification system that examined ossification exostoses, robusticity markers (where "robusticity" refers to "hypertrophy of the bone in the form of robust muscle attachment" (Hawkey, 1988:324)) and stress lesions. The identification of these methodological and theoretical discrepancies led to the Coimbra Workshop in Musculoskeletal Stress Markers in 2009. A continuously updated website (http://www.us.pt/en/cia/msm/) was established as a result of the workshop. It allows all researchers in musculo-skeletal markers (MSM) to contribute to more uniform and reliable analyses in the reconstruction of past activities (Santos et al., 2011). To further assure conformity of method, an entheseal-scoring workshop using this method was held at University of Sheffield in January 2015.
- 2) In order for bone morphology to be modified, the bone needs to be put through a series of movements that are energetic and carried out for a long period probably on a daily basis (Molleson, 2007). Many studies on MOS make the assumption that the most commonly assessed bony alterations are a result of habitual occupation. Jurmain (1999) placed emphasis on the point that the many varied activities (habitual and exceptional) that occur during a lifetime need to be considered in MOS studies. Merbs (1983) added to this argument by saying that although bodies are exposed to a variety of activities on a daily basis, there are certain activities that are performed repetitively, especially those linked to survival, e.g. in food preparation and hunting and these would leave the greatest evidence of adaptation in the skeleton. Luttman et al., (1988) identified that, in order to be able to compare activities,

there needs to be greater definition and characterisation of a specific activity, such as the length of time it is performed, the number of times the action is performed within a given time span and also how much overload is placed upon the body and on which part of the body.

- 3) External factors such as ethnicity, sex, age, diet and genetics may affect the way in which an activity is performed and how the activity affects the body as well as the type, duration and range of activities undertaken. Jurmain (1999) identified a number of questions that should be considered in order to try and understand the effect these external factors may have on activity. These included information on who did what behaviour and for how long; how old was the individual when they started the activity; how much individual variation is there between the participants in a specific activity and finally how was a specific activity undertaken?
- 4) There is a risk of circular argument in the interpretation of activity patterns from the skeleton. Relating skeletal changes to a "presumed" behavioural cause and then using the changes to support the argument that the behaviour caused the changes is a circular argument. The classic example of this is seen in Angel's (1966) discussion on the existence of "atlatl elbow". Ethnographical evidence demonstrates the existence of atlatls, osteological evidence identified the presence of osteoarthritis in the elbow joints, therefore the assumption has been made that throwing atlatls causes degenerative change in the elbow joint. Jurmain (1999) considers this circular approach to be in part, caused by lack of consideration of available clinical data, which identifies aetiology related to degenerative joint disease onset. Larsen (1997) contrasts this by suggesting that clinical data should be reviewed in bioarchaeological study, but that it is not always possible to make a direct comparison between the two sources of data as clinical data does not always include the nuances of subtle change visible in the dry bone sample. Clinical data does provide a baseline for bioarchaeological research and its usefulness should not be overlooked. Clinical studies can contribute to the general understanding of how habitual occupational or recreational activity causes change in the musculoskeletal system (Bird, 1990), but in clinical research, the emphasis is on soft-tissue damage and pathology, which limits the use this resource can have for bioarchaeologists (Kennedy, 1989).

The use of multiple markers of occupational stress to study activity patterns can diminish the risk of circular argument (Hawkey, 1998; Roberts and Manchester, 2005). Watts (2010) used multiple non-specific indicators of stress to identify which period of growth and maturity was most

affected by the stressor. Al-Oumaoui et al., (2004) used multiple musculoskeletal stress markers to assess activities carried out by populations from the Iberian Peninsula according to their sex, environment and culture and found that their results coincided with the available historical and archaeological data. Sofaer Deverenski, (2000) used the distribution and severity of degenerative change on the zygapophyseal joints and osteophytosis of the vertebral bodies to identify gender specific activities. Jurmain (1999) argued that the use of a single marker or a combination of markers does not give more rigour to the interpretation of activity patterns if the data are insufficient and that greater precision is not achieved by using lots of weak data, highlighting the importance of good and appropriate data collection.

4.3.4 Activity and degenerative joint disease

This section reviews the link between activity and degenerative joint disease. It begins by identifying the problems associated with clinical and archaeological terminology for degenerative joint disease then discusses classification of osteoarthritis and the multifactorial aetiology of the disease. This is followed by an overview of the degeneration cascade and process behind joint degeneration.

4.3.4.1 Definition of degenerative joint disease

There is a lack of consistency in the terminology used in clinical and archaeological literature when discussing degenerative joint disease and osteoarthritis. Clinically, osteoarthritis is defined as a "non-inflammatory degenerative joint disease occurring mostly in older persons" (Taylor, 1988:1197). Berkow (1992) adds to this definition by describing the condition as being characterised by "degeneration of the articular cartilage, hypertrophy of the bone at the margins and changes in the synovial membrane" Jurmain (1999) presents the archaeological nomenclature of osteoarthritis to the debate by discussing the arguments for naming the condition either osteoarthritis or degenerative joint disease; "osteoarthritis" is thought to imply an inflammatory condition (Hough, 1993) and degenerative joint disease is considered to imply a manifest, passive process associated with old age (Dieppe, 1987). Lozada (2011) considers that term degenerative joint disease used as a descriptive name for osteoarthritis to be inappropriate because secondary non-specific inflammatory changes may also affect joints as well as the effects of "wear and tear". The terms degenerative arthrosis, degenerative arthritis or osteoarthrosis are also commonly found in the clinical literature but are not commonly seen in the paleopathological literature (Resnick, 2002). Waldron (2012) considers that osteoarthritis should not be considered a degenerative condition because it is a disease of joint articular cartilage and the changes that can be seen in bones are secondary to articular cartilage breakdown. However, Bick (1948)

identified that neither term was more significant than the other, but it was important that there was consensus of agreement on which term to use. The term osteoarthritis (OA) will be used in this study as it is commonly used in clinical and bioarchaeological literature.

The next section presents classification into primary and secondary OA as a means of differentiating type of OA and the process of degeneration in the spine. It continues with a discussion of causation factors.

4.3.4.2 Primary and secondary osteoarthritis

The classification scheme of Mitchell and Cruess (1977) is frequently used to classify osteoarthritis as either primary (idiopathic, developing in previously undamaged joints in the absence of an apparent triggering event) or secondary (some known causative factor, disease or predisposing state, such as anatomic abnormalities, trauma and inflammatory and metabolic disorders) (Berkow, 1992; Arden and Nevitt, 2006). However, the spectrum of joint involvement and severity, and the etiological factors involved are considered to be too complex to be involved in two simple divisions (Doherty et al., 1983). Doherty et al.'s (1983) research into the role of primary OA in the hands of a patient having an effect on the onset and severity of secondary OA in knees of elderly patients invalidates the clear distinction between the two subsets and adds emphasis to the multifactorial aetiology of OA.

From a clinical perspective, primary osteoarthritis is seen typically in middle-aged females and is described in the clinical literature as being symmetrical and polyarticular, affecting the main weight-bearing joints and axial skeleton (Fergusson, 1987). Joints displaying degenerative change are considered to have primary OA when it occurs in the absence of any underlying predisposing factor (Moskowitz, 1993). It can be seen in the distal (Heberden's nodes) and proximal interphalangeal joints (Bouchard's nodes), the first metacarpal joint, cervical and lumbar spine, first metatarso-phalangeal joint, hip, knee, intervertebral discs and facet joints of the spine (Berkow, 1992; Kellgren and Moore, 1952).

Secondary osteoarthritis refers to degenerative disease in a synovial joint resulting from a predisposing factor, e.g. trauma that causes damage to the articular cartilage and may or may not also have caused damage to the underlying subchondral bone. It can occur in relatively young individuals (Sharma, 2001). Secondary osteoarthritis can develop following a number of pre-existing disorders. These include, for example, metabolic causes such as acromegaly and ochronosis, anatomic anomalies such as congenital hip dislocation, leg length inequality and hypermobility syndromes, traumatic events such as major joint trauma, joint fracture and chronic injury due to occupation or activity or inflammatory disorders such as septic arthritis (Arden and

Nevitt, 2006; Wright, 1990). Some of these disorders, e.g. ochronosis (hyperpigmentation) would not be visible in an osteoarchaeological sample due to the lack of preservation of the soft tissues.

4.3.4.3 The process of degeneration in the spine

It is thought that there are three phases in the spinal degenerative cascade that are linked to the process of degeneration in the spine (Kirkaldy-Willis, 1985). The first phase is when changes in the anterior vertebral elements (e.g. decreased height of the intervertebral disc) cause reduced stability between vertebrae and the disco-ligamentous structures become less supportive leading to degenerative changes (eburnation) in the vertebral body and facet joints (Figure 4-1). This progresses to the second phase, decrease in disc height continues and the vertebral segments become less stable, leading to loosening of the facet capsule and ligaments with a greater degree of degenerative change occurring. The third phase is identified by further degeneration and restabilisation of the spine by the formation of osteophytes and fibrosis (Suri et al, 2010).



Figure 4-1 Range of degenerative processes in the spine (author's image)

4.3.4.4 Causes of osteoarthritis

The exact aetiology of OA is yet to be determined (Greenberg et al., 2006; Berenbaum, 2013).

One possible cause is the failure of one or more factors in the interaction of mechanical, biologic,

biochemical and enzymatic feedback loops which can lead to the development of OA in a joint (Berkow 1992). OA can be considered to be a multifactorial condition due to many mechanisms that can initiate the cellular changes leading to the development of OA representing a sustained imbalance between cartilage synthesis and degradation (Moskowitz et al, 2004), leading to disruption of the integrity of the articular cartilage (Rogers et al., 1987). OA aetiology is a combination of intrinsic factors found in cartilage and extrinsic factors such as age, sex, hormonal changes, obesity, genetic predisposition, ethnicity, mechanical factors, physical activity and occupation (Merbs, 1983, 2001; Waldron, 1997; Sofaer Deverenski, 2000; Kahl and Smith, 2000, Solano, 2002; Weiss, 2005a, 2005b; Weiss and Jurmain, 2007). In summary, there is no one direct cause for OA. Mechanical influences are implicated as predisposing factors, and these are discussed next from a clinical and archaeological perspective.

Whilst moderate mechanical loading through a joint appears to be necessary for the maintenance of healthy cartilage (matrix synthesis), abnormal loading is thought to increase the risk of developing OA (Griffin and Guilak, 2005) and the mechanical environment caused by strenuous physical activity is considered to be a primary causal factor in the development of OA (Larsen, 1997); however, the development of OA is a lengthy process and can take 15-20 years (Poole, 1999). The clinical literature suggests that there are a number of mechanical factors such as gross malalignment or abnormal geometry within the joints, which can increase the risk of abnormal loading (Goodfellow and Mitsou, 1977; Bullough, 1981) as can the lack of proprioceptive and nociceptive receptors (Jordan et al., 2000). (Proprioception is joint position sense and nociception is the ability to sense pain.) Malalignment can occur when there is excessive valgus or varus strain through a joint (e.g. the knee). The strain increases local stresses inside the joint and causes OA progression due to a cycle of further damage to cartilage and bone in the compartment subjected to increased loading (Sharma, 1995; Sharma et al., 2001; Felson, 2004).

Another cause of malalignment is hypermobility (caused by ligamentous laxity) within a joint (Bird et al., 1978; Fergusson, 1987). Reduced muscle strength or muscle fatigue has also been recorded as a factor in joint malalignment leading to cartilage degeneration and OA (Fisher and Prendergast, 1997; Thorstensson et al., 2004). Conflicting studies have shown that both weaker and increased quadriceps strength in the setting of knee joint malalignment can be associated with increased risk of deterioration in a joint with OA. Baker et al. (2004) identified that patients diagnosed using radiography as having knee OA demonstrated significantly weaker quadriceps strength than patients without OA. In contrast, Sharma et al. (2003) found that patients with greater quadriceps strength are at greater risk of existing knee progression within the context of joint malalignment and ligamentous laxity. Reduced motor power (e.g. caused by poliomyelitis) appears to reduce the risk of OA onset even though the joint can be moved passively and may be

unstable (Glynn et al., 1966). Muscle fatigue caused by repetitive movements can increase the vulnerability of a joint to OA. This situation can be seen an occupational risk for workers using joints repeatedly for hours (Nuki, 1999). Abnormal geometry can be caused by congenital abnormality (e.g. congenital hip dysplasia (Tanzer and Noiseux, 2004), leg length anomalies (Golightly et al., 2008), joint injury earlier in life such as a meniscal tear in the knee or rupture of an anterior cruciate ligament (Englund and Lohmander, 2004) or tear of the hip labrum (Tanzer and Noiseux, 2004). Joints are subject to a process of continual remodeling throughout life (Johnson, 1962) (See section 4.2.3). This can change the profile of the joint surfaces allowing the two articular areas to become more congruent. However, this may cause loading in areas that already maximally bearing strain (Bullough, 1981). It has been hypothesized that abnormalities in the remodeling process may be involved in the development and progression of OA and is thought that this risk is increased in the abnormal joint (Bullough and Jagannath, 1983).

The effects of age, weight and mechanical loading have been discussed as causes of OA in the bioarchaeological literature (Merbs, 1983, 2001; Waldron, 1997; Sofaer Deverenski, 2000; Kahl and Smith, 2000; Solano, 2002; Weiss, 2005a, 2005b). The role of heavy mechanical loading as a cause of OA is an area that has been subject to intense scrutiny and has led to the conclusion that severe OA scores on joints are the result of continued use of specific muscles and joints in daily and repetitive tasks (Weiss and Jurmain, 2007) and as such has led some researchers to consider that this condition is ideal for the reconstruction of activity patterns. It is extremely difficult to link patterns of OA distribution in the body to specific activities, however, OA can be considered to be a general indicator of joint loading and can be used to identify joints under long-term stress or used for habitual movements (Knüsel, 2000). Some paleopathologists (Waldron, 1994; Jurmain, 2012) consider that research in this area is limited due to the lack of consideration of external factors (such as environmentally associated stress) reflected by variable lifestyles that can affect the onset of OA. Waldron (1994) argued that until the role of all non-mechanical influences (e.g. nutrition, hormones, age, genetics, disease) and biomechanical responses (e.g. magnitude of stress, rate of stress, stress interval etc.) in bone remodeling are fully understood, patterns of activity cannot be satisfactorily quantified. Martin et al. (2013) also encourage researchers to incorporate as much contextual information as possible when comparing activity related differences between sexes and populations to increase the validity of the results, rather than just identifying significant differences in order to try to create an understanding of why the differences occur

The relationship between repetitive physical activity and OA is of particular interest to this study. Movement is the essential prerequisite for the development of OA; joints that do not move do not develop OA (Waldron, 2012). As discussed above, there are also a number of other important factors that can precipitate the onset of OA (age, sex, hormone levels, obesity, genetic predisposition and mechanical factors). There have been many studies investigating the effect of occupationally related repetitive movements in relation to the development of OA. Weiss and Jurmain's (2007) review of general studies of occupation/epidemiological data and studies relating to specific activities data failed to demonstrate a definitive relationship between increased levels of activity and prevalence of OA. However, 22 of the 41 studies reviewed did show a significant relationship between movement and prevalence of OA. The studies chosen for inclusion in their review were selected based on content, rather than a representative sample of the available literature. The selected studies looked at individuals in specific risk groups relating to mechanically stressful activities. Their review predominantly included studies that examined prevalence of OA in relation to activity but did not consider the severity and extent of OA present. This pattern has continued in activity-related/OA research, e.g. eburnation was scored simply as absent or present in Molnar et al. (2011). Other studies have scored the individual diagnostic criteria for OA (pitting, eburnation, osteophytic growth) (Sofaer Deverenski, 2000). The inclusion of a high level of discrimination in recording of the individual criteria for OA allows for potential distinction in aetiology between samples under investigation.

4.3.4.5 Physical activity and OA

Waldron (2009) considers that after dental disease, osteoarthritis is the most frequently seen skeletal pathological condition. Macroscopically, it can be diagnosed on dry bone using the method described in section 5.7.6. The primary value of OA in bioarchaeological study relies upon the assumption that there is a relationship between the distribution and severity of OA and movement of the joints caused by repetitive mechanically loaded activities which allows for differentiation of lifestyle amongst and between populations (Ostendorf Smith, 2013).

The field of bioarchaeology has held an interest in the relationship between physical activity and patterns of OA distribution in the body as a source of information about behaviour since the 1960's when Angel (1966) first used the term "atlatl elbow" to explain patterns of OA seen in the radio-humeral joints of skeletons from Early Horizon burials at Tranquillity, California. Angel's work was one of the first to identify the potential link between osteoarthritis and specific behaviour when he attributed the distinct pattern of OA to throwing actions related to the use of the atlatl (a spear thrower) in males. This link between an activity and pathology was influential in the paleopathological study of activity markers. Bioarchaeologists have built upon this background by examining alterations in skeletal morphology as a response to habitual and repetitive physical activity (Havelkova et al., 2010; Lessa, 2011) with many bioarchaeologists believing that the presence of OA indicates patterns of use or "overuse" of the joints of a skeleton

due to heavy workload and demanding physical activities (Angel, 1966; Jurmain, 1977; Merbs, 1983; Bridges, 1991; Bridges, 1992; Larsen et al., 1995; Jurmain, 1999; Sofaer Deverenski, 2000). However, Jurmain (1999) urges caution when using OA as an activity marker. Angel's sample was very small (13 elbows). The small sample size did not allow for assessment of differences between age groups; the individuals' onset of OA is age related and Angel did not control for this factor. The term "Atlatl elbow" to describe a pattern of degenerative change (eburnation) of the capitulum of the humerus is actually based upon limited evidence. The distribution pattern of OA is not a clear link to specific physical activity. Whilst greater severity and extent of OA can infer a more physical lifestyle, it is difficult to interpret OA distribution patterns and link them to a specific activity due to its inherent multifactorial aetiology leading researchers to question the reliability of using OA as a marker of activity (Weiss, 2005; Weiss and Jurmain, 2007; Waldron, 2009). Jurmain (1977) was a keen supporter of using OA as an MOS, however further research and understanding of the complexity of OA caused him to become much less certain as to its validity. Jurmain (1999) identified that mechanical loading plays only a small part in the development of OA and other factors must be considered. Archaeological and ethnographical evidence as used by Angel (1966) to support the theory behind activity based MOS is not always valid. The evidence does not provide detail on the frequency or duration that a specific activity was performed by an individual. This evidence also generally fails to document the age that an individual began to undertake tasks that involved mechanical loading. This lack of supporting evidence makes it unlikely that general or specific activities can be inferred from MOS (Jurmain, 1999:138-9). In summary, Jurmain emphasised that there is not a straightforward relationship between OA and activity and many other factors need to be taken into consideration. Larsen (1997:164) states:

"...although articular pathology relating to activity offers important insight into behavioural characteristics of human populations in a general sense, the identification of specific activities or occupations from individual remains may not always be possible".

4.3.4.6 Specific activity and OA

There have been many clinical studies relating to the study of occupation/sporting activity and OA in modern populations, which identify areas of the body at risk from certain repetitive movements. Occupations that require fine finger movements and pincer grip increase the risk of hand OA (Hadler et al., 1978); those that require repetitive movements involving kneeling, knee bending, squatting or the carrying of heavy loads increase the risk of development of knee and hip OA (Thelin, 1990; Felson et al., 1991; Croft et al., 1992; Cooper et al., 1994; Coggon et al., 2000). Miners perform repeated kneeling and squatting actions and carry heavy weights, thereby

predisposing them to OA of the knee (McMillan and Williams, 2005). Farmers and farm workers appear to be predisposed to a higher frequency of OA of the hip than the general population due to their work requiring them to perform activities that involve carrying heavy weights (which can involve a flexed trunk when manoeuvring animals), digging and shovelling (Schouten et al., 2002). Sporting activities, such as running that involve moderate activity do not appear to increase the risk of hip or knee OA (Buckwalter and Lane, 1997). Vignon et al., (2006) systematically reviewed OA of the hip and knee and the relationship with sport and concluded that all sport could present a risk for the development of OA in the hip and knee even if moderately performed, but this risk is related to training intensity, duration of training, pre-existing trauma to the hip or knee and increased body mass index for each individual. Elite and professional sports participants are considered to be at much higher risk of the development of OA due to the effect of impact forces and torsional forces to the joints (Vingard et al., 1993; Kujala et al., 1994; Kujala et al., 1995; Spector et al., 1996). The number of hours of exposure to intense sporting activity and long-term exposure over years of high intensity activity is considered to be a high risk factor for OA (Vingard et al., 1993).

Early osteological emphasis when considering OA focussed on the spine. Elliot-Smith and Wood-Jones (1910) reported the presence of "spondylitis" as being a common occurrence in the spines of Ancient Nubians. Dorland's Medical Dictionary (Taylor, 1988) defines spondylitis as inflammation of the vertebrae. When one or more vertebrae are involved, it is termed spondyloarthritis. This term covers the arthritic complaints including rheumatoid and osteoarthritis. It is likely that Elliot-Smith and Wood-Jones were reporting on OA changes in the spines of the human remains they were studying. There was no attempt to link the findings to activity. The identification of osteoarthritic changes within the spine was a precursor to study of activity related change.

Pre-1970, many studies continued to focus on the spine and mostly looked at patterns of distribution of vertebral osteophytosis (Hooton, 1930; Stewart, 1947; Anderson, 1963; Bourke, 1967; Chapman, 1972; Swedborg, 1974). Swedborg's study used 1126 spines from a Polish medieval population and is notable for the rigour with which analysis of vertebral osteophytosis and facet joint OA was performed. The main aim of these studies was the recording of degenerative changes within the spine rather than attempting to link them to activity.

Post-1970, the bioarchaeological focus altered and the emphasis shifted towards identifying activities that could be linked to OA distribution. Merbs (1983) used data collected from a Canadian Inuit population (41 males and 50 females) to explore the relationship between known or inferred specific activities (obtained from ethnographical resources) performed by the Inuit and

selected pathological changes seen in their skeletons. He identified 20 activity patterns that had the potential to leave MOS. Some of these activities were common to all populations (e.g. bipedalism), but some were Inuit specific. He identified that female Inuits have a distinctive pattern of distribution of OA in the lower thoracic spine and linked this to the carrying of heavy objects on their backs. In addition, Merbs linked high levels of vertebral osteophytosis in males to sledding accidents. He noted the importance of sex and side differences. An interesting outcome of this study was the unexpected linking of elements of an activity with particular patterns of distribution of MOS. His data were derived from a Canadian Inuit population and he correlated OA found in the shoulder, elbow, wrist, hand, spine, ribs, hip, knee, ankle, foot and temporomandibular joint with specific activities identified from ethnographic sources e.g. paddling a kayak and throwing a harpoon placed stress upon the shoulders and elbows; the action of sewing or hammering, flaking and splitting hard materials such as bone or stone was compatible with OA seen in the hands and fingers; using teeth to soften skins and boots with the teeth could be manifest as OA in the temporomandibular joint. Although Merbs supported his findings with information provided by ethnographic evidence, caution should be applied regarding the activities and occupations suggested from the archaeological evidence of a small sample.

Lai and Lovell (1992) carried out a similar study using three male skeletons from Alberta, Canada, dating to the Fur Trade Period. They acknowledge the difficulties identified by Jurmain in linking specific activities to skeletal characteristics but argue that their sample met the criteria for the study of MOS due to the presence of OA, muscle insertion robusticity, enthesophytes and accessory articular facets. They found OA in the cervical spines of the skeletons and attributed the patterns of degenerative change seen to the use of tumplines to carry heavy loads. Although Merbs (1983) and Lai and Lovell (1992) were able to support their results with evidence gleaned from ethnographic and historical accounts, the sample size used in their research was moderate in the case of Merb's (1983) and small in Lau and Lovell's (1992) study and therefore caution should be used when considering activity MOS and the activities relating to them within these contexts.

Bridges (1994) also found similar results in the lower vertebrae of remains from the Pickwick Basin area of north-western Alabama (USA). Bridges suggested that the changes observed might be attributable to the extra strain placed on the spine by using a tumpline around the forehead or a body band around the upper torso.

The study by Sofaer Deverenski (2000) is a further example of degenerative changes in the spine being linked to the carrying of heavy weights. She analysed the spines from individuals from Ensay

and Wharram Percy in the UK and was able to demonstrate a correlation between degrees of arthritic change and activity, with differences being seen between sexes and samples. Her results were supported by ethnographical and historical documents that identified a gendered division of labour. The Ensay women carried large heavy creels strapped across their chests, with the weight of the creel resting on the low back. This was thought to cause a distinctive pattern of degeneration in the spine not seen in the males from Ensay nor the Wharram Percy population. She also noted that articular facet remodeling could be correlated to the age at death, the sex and population of the sample, whereas the presence of osteophytes and pitting could be considered to be the product of different causes e.g. different physical stresses placed upon the body due to differences in lifestyle.. This study identifies the importance of considering the diagnostic criteria for OA (eburnation, pitting, osteophytes) separately as not all are implicated in changes caused by repeated activity.

Stirland (1991) further questioned the interpretation of activity from skeletal remains using ethnographic and historic accounts but realised the benefit to population study, if some occupations or repetitive activities can be identified using MOS. The skeletons retrieved from Henry VIII's flagship, the Mary Rose offered an exceptional opportunity for research into activity related markers. There was documentary evidence for this population and they were controlled for environment (in that they were all crew members of one ship) and time period, which makes them ideal for this type of study. Although there was no crew list from the ship, (apart from three crew members that were known by name), there was a list of the occupations performed by the crew. Thus, the human remains from the Mary Rose fit Merb's (1983) criteria for activity related research, as there are a limited number of known specialised occupations; there is good skeletal preservation and also a known date of death. Analysis of the remains showed that the area of the body showing most pathological change was the vertebral column. The presence of symptoms indicative of high levels of stress in young men from the gun deck was attributed to hauling cannons into and out of the gun ports. This was further supported when Stirland and Waldron (1997) compared the pathological changes seen in the vertebrae from comingled remains of young men from the Mary Rose with those of the spines of mature or older adults from a medieval cemetery (St Mary Fybridgegate, Norwich). They noted that the spines of the younger men from the Mary Rose differed less than would be expected from the older men from Norwich. Stirland and Waldron question whether this lack of differential implies that there were accelerated ageing changes in the Mary Rose spines due to the occupations of the crew that would have involved much heavy work.

Rojas-Sepulveda et al. (2008) observed a high prevalence of vertebral OA in a Pre-Columbian Muisca series from Columbia. They examined vertebrae from 83 individuals and found that 83%

of individuals and 32% of vertebrae had at least one of the manifestations of OA (lipping, osteophytes, eburnation, pitting). Changes could be seen in the youngest cohort (15-30 years). This evidence was used to conclude that high levels of activity began in childhood and that this may have accelerated the aging process seen on the spines.

The previous studies refer to prevalence or distribution of OA in relation to activity. The prevalence of facet joint OA in a population is dependent upon the age and sex of the population being studied (Srikanth, 2005), however these categories are not always included in the information given in published literature. Most studies on vertebral OA focus on osteophytosis of the vertebral body (VOS). Regular patterning of involvement of VOS (with highest levels being seen in the lumbar region and lowest levels of involvement in the thoracic region) is considered to be highly relevant in the study of spinal OA (Jurmain, 1999). This patterning is supported by a number of studies (Bridges, 1992, 1994; Lovell, 1994). Stewart (1966) explained the patterning as being due to increased stress being placed upon certain intervertebral discs as a result of curvature of the spine caused by bipedalism. Facet OA follows similar distribution patterns as VOS with highest levels being reported at C2-C5, C7/T1, T9/T10 and L5/S1 (Inglemark et al., 1959; Kilgore, 1984; Bridges, 1994; Knüsel et al., 1997). These observations were made across different populations. Inglemark et al. (1959) collected their data from 215 skeletons from a medieval cemetery at Aebelholt in Denmark; Kilgore (1984) used a medieval Nubian population; Bridges (1994) analysed individuals from prehistoric South Eastern United States and Knüsel et al. (1994) remains from a Medieval monastic cemetery of the Gilbertine Priory of St Andrew, Fishergate. They all recorded that the changes became more pronounced with advancing age. The prevalence of OA increases with age in both clinical (Felson et al., 1995; Jordan et al., 2000; Loeser and Shakoor, 2003; Zukowski et al., 2012) and archaeological samples (Waldron, 1992; Jurmain, 1977). However, it is not possible to establish a relationship between age of onset and degree of joint destruction because many joints can remain unaffected in very old individuals (Bullough, 2004).

It is difficult to make categorical statements about the differences in prevalence of OA in males and females without taking age into consideration. Generally speaking, clinical evidence indicates that older women are more likely to have OA than men and have greater severity of the disease (Srikanth et al., 2005), however, prior to the age of 50 men tend to have a higher prevalence (Felson, 2004). The literature suggests that females have a higher incidence of knee and certain types of hand OA (Olivieria et al., 1995; Felson and Zhang, 1998; Sowers, 2001; Wilder et al. 2006) and males over 50 are more at risk of developing cervical spine OA (Srikanth, 2005). The agerelated rise in OA prevalence in post-menopausal women suggests a possible link with sex hormone levels, particularly oestrogen deficiency (Arden and Nevitt, 2006). Waldron (1996) studied the distribution of OA in the hands of subjects from archaeological sites across England (including Allington Avenue, Ashtead, Brighton Hill south, Farringdon Street, Great Chesterford, Kellington, Merton Priory, Red Cross Way, Royal Mint, Southgate Street, Spitalfields and Ulwell). The samples were categorised as medieval or post-medieval in period. Individuals were only included if a complete set of hand bones was present. A positive diagnosis of OA was made when eburnation was observed on a joint surface. This study also identified differences between males and females, with males having predominantly unifocal disease, whereas it was predominantly multifocal in females, and 64% of females and 58% of males having at least one joint showing eburnation. Waldron (1996) identified a potential source of bias in this study in that although OA was identified on the hands of many skeletons, these individuals had to be excluded from the study as their hands were incomplete. Waldron (1991) also conducted a similar study on the Georgian and Early Victorian population interred in the crypts of Christ Church Spitalfields in East London (1729-1869) and again found that when considering the finger joints, women were more likely to be affected than men (16% of females v 9% of males).

The archaeological literature identifies that the prevalence of OA can fluctuate with males usually having a higher prevalence than females for weight bearing joints in the under 50's, with the converse being seen as women become over 50 (Bridges, 1992). The association between sex, age and OA means that age profiles must always be considered when examining differences in OA in archaeological populations

In the past, the sex differences that occur in the distribution and prevalence of OA have been considered to be related to activity patterns (Jurmain, 1977). However, it should not be assumed that sex differences are cultural and hormones, body size, genetic predisposition and joint structure should also be taken into consideration (Weiss and Jurmain, 2007).

This section is concluded with reference to Bridges' (1992) bibliographic review of prehistoric OA in Pre-Columbian America. She observed that the knee demonstrated the highest prevalence of OA followed by the elbow. Furthermore, no correlation of specific pattern of OA with any particular subsistence economy was found because prehistoric groups activities range from hunter/gatherer to agriculture and fishing and all demonstrate a high prevalence of knee and elbow OA. Thus, it is not possible to infer specific activities from the presence or absence of OA in a joint, however, it is possible to infer different levels of activity. The above examples show that, although the validity of using OA as a marker of occupational stress has been questioned, it is still used to measure or compare activity levels between populations. Differences in distribution and severity of OA in the spine can be a useful tool when comparing activity levels between

individuals or populations. Although some studies (e.g. Merbs, 1983) were successful in linking OA distribution to specific activities, it is generally concluded that this is not ideal practice.

4.3.4.7 Limitations to the use of OA as a marker of physical activity

The extent to which OA can be used as an indicator of activity has been subjected to debate (Jurmain, 1999; Waldron, 1991). This is in part due to the realization that it is an age-related disease (Rogers et al., 1987) with other factors such as sex, ancestry, weight and movement being involved in its aetiology (Waldron, 1994). Knüsel et al. (1997) attempted to use the distribution and severity of degenerative joint disease in the spine to identify differing activity patterns between two known populations from a 13th-14th century medieval priory cemetery at St Andrew, Fishergate, York. They were unable to identify any significant differences and concluded that the vertebral column is not the best focus of study for occupational markers in skeletal collections due to the fact that biological constraints (curvature of the spine in response to bipedalism) obscure the expression of the markers. However, their study was a comparative analysis of the pattern and severity of OA at the facet joints and vertebral bodies and remodelling was scored as complementary to osteophytic growth and not separately. In contrast Sofaer Deverenski (2000) was able to identify intra-and inter-site differences in the frequency and distribution of identifiable changes in the spine in respect of OA of the facets and osteophytosis of the vertebral bodies. In her study remodelling was separately defined from the presence of osteophytes.

There is some evidence that degenerative change can be seen in younger adults (18-25 years of age) when they have been subjected to the stress of repeated loads upon the joints (Bridges, 1991). These are atypical degenerative changes as the pattern of degeneration occurs much earlier in life than expected. OA found in only one location in a younger adult is possibly related to an old healed traumatic injury to that joint, but activity related degenerative change is seen in multiple locations (Martin, 2013). Given that evidence of OA in relation to repetitive physical activity, can be seen across age groups (young adult to old adult), it is worthy of continued use as a marker of occupational stress.

On reviewing the literature, it is noticeable that there is great variation in the definition, diagnosis and recording of OA between researchers. Bridges (1992) identified areas of concern, including non-standard data-recording protocols and absence of statistical testing. This limits the use of OA as a marker of occupational stress in bioarchaeological studies as the lack of methodological consistency limits cross-study comparison and reduces the academic benefit of existing studies. It is difficult to apply clinical studies of risk factors to bioarchaeological studies of OA as clinicians use factors such as pain and swelling in the affected joints, crepitus, joint space narrowing, marginal osteophytes and sclerosis (seen on radiographs) to define OA, whereas paleopathologists use marginal osteophytes and eburnation to make their diagnosis (Waldron, 2009).

There are inherent difficulties in comparing population prevalence rates for OA within the literature from different data sets as varying methods of diagnosis are used and data is presented differently (Roberts and Manchester, 2005). The lack of a systematic scoring system for OA within the bioarchaeological context is a major limitation to research. This leads to extreme difficulty in comparing studies and consequently differences in prevalence of OA between similar populations may not accurately reflect population-specific patterns of distribution and severity. Inter-study comparison may highlight the diversity of classification and scoring method used rather than a true comparison of prevalence.

Jurmain (1999) noted that there was a lack of a single diagnostic and scoring system for OA in bioarchaeology and that eburnation should be used as the diagnostic criteria for the presence of OA. Waldron (2009) recommended that, in order to make a diagnosis of OA on a joint surface, eburnation or at least two other features such as marginal osteophytes, sclerosis, pitting or changes in joint contour must be present. The need for a standardised data collection method was recognised and lead to Buikstra and Ubelaker (1994) producing the Data Collection Standards Manual. This standardizes the recording of the presence and degree of severity of lipping, pitting and eburnation for all joint surfaces.

Another attempt to standardise diagnosis and recording is the Global History of Health Project (<u>http://global.sbs.ohio-state.edu/</u>). The aim of the project is to assess global levels of health from human remains using standardised methods. This method relies on vertebral body osteophytosis as a diagnostic for OA of the spine (Steckel and Rose, 2002) and is therefore lacking the specificity of diagnosis recommended by Waldron (2009).

There has been scepticism of the adoption of standardised methods of recording and diagnosis. Cohen and Crane-Kramer (2007) consider that it would be impossible to get all bioarchaeologists to work from the same standardised methods. Their opinion is that it would be better to get researchers from similar areas to adopt the same methods in order to reduce the risk of error from environmental, genetic and inter-observer error. They do not feel that this would work globally.

In conclusion, until there is a standardised method for diagnosing and recording OA in human remains, inter-site comparison will always be problematic. Further research could be directed towards methodological standardisation.

This section has reviewed the use of MOS in the comparison of differing activity levels between samples. The role of OA as a MOS has been given special attention as the degree of severity and distribution of OA throughout the spine is measurable and can provide information about the degree of but not type of activity undertaken by the samples under investigation in this study.

In summary, this chapter has identified the difficulties in using OA as a marker of occupational stress. OA has been used to compare degrees of physical activity between populations but cannot be used to infer a specific activity. Inter-sample variation of the degree and severity of OA (as scored using the methods described in Section 5.7.9) will be used to assess the degree of physical activity undertaken. This information will help to redress Ruff's (1987) criticism of sexual dimorphism studies in that there has been great focus on statistical techniques for differentiating sexes but little consideration of functional aspects. By considering the relationship between OA score and facet morphology and structure and function are explored.

Chapter 5 Methods

This chapter discusses the human skeletal sample used in this study and the methods used to resolve the research questions introduced in Chapter 1. The Materials section will provide detail about the archaeological sample used in this study with detail about the context of the cemetery and the sample composition, whilst in the Methods section, the processes to estimate sex, age, measure facet angle and size, degree of osteoarthritis, lower limb robusticity and humeral directional asymmetry (previously linked to asymmetry of distribution of degenerative disease in the spine: Stirland and Waldron, 1997) are described.

Studies have indicated populational variation in sexual dimorphism of the vertebra (Zheng et al., 2012; Hou et al., 2012; Amores et al., 2014). To reduce the effect of this confounding factor, all archaeologically derived cemetery populations chosen for this study came from the southeast of England (see Figure 5-1). The aim of this strategy for was to minimise inter-sample differences in sexual dimorphism. However it is unlikely that activity patterns were uniform across the populations sampled and that there may be considerable differences in activity and diet between temporally dispersed samples. The differing activity and diet patterns will be controlled for by individual analysis of each sample followed by inter-sample comparison.

The skeletal samples chosen for this study were selected because they were recorded in the site report as having a significant number of well-preserved adult burials of both sexes that represented a wide age range and were available for macroscopic analysis. The study will examine the sexual dimorphism of each vertebral level, which allowed for inclusion of skeletons with an incomplete complement of cervical, thoracic and lumbar vertebrae.

The Great Chesterford collection was used for the pilot study to test for intra-rater reliability of the method. The skeletons from this site were initially analysed in 2013 to test the method used to measure the facets and for the author to gain experience in vertebral level recognition and measuring techniques. The collection was re-measured in 2014 and 2015 providing data to test for intra observer reliability.

Chapter 5 Materials and Methods





5.1 Great Chesterford

5.1.1 Historical background of Great Chesterford

Great Chesterford (TL 501435) is located in Essex. The town lies south of Cambridge, on the east bank of the River Cam and was a strategically important site, straddling the entrance to the Fens through a gap in the low chalk hills as well as a number of significant route ways and the Iron Age tribal boundary between the Trinovantes and the Catuvellauni (Evison, 1994). The actual position of the Anglo-Saxon settlement in Great Chesterford is unknown. The town had its origins in the late Iron Age and the site of the Roman town in can be identified from traces in field boundaries, from earlier records and excavations (Brinson, 1963). The site of the Roman town and its environs are scheduled as an Ancient Monument.

5.1.2 Archaeological background of Great Chesterford

Commercial gravel extraction in the county produced firm evidence of an Anglo-Saxon cemetery near the Roman town of Great Chesterford in 1952, although recorded discoveries (notably a cruciform brooch) from as early as 1819 suggested that there was a cemetery in this area (Jones, 1980). Smith (1903) commented that cemeteries of the Anglo-Saxon period are rare in Essex. Jones (1980) considered that this statement was still true 80 years later. Excavation of the cemetery was carried out over several seasons between 1953 and 1955 on behalf of the Inspectorate of Ancient Monuments and reported by Evison (1994). The area permitted for excavation was extremely limited for each season due to the need to keep the ground level for the convenience of the gravel extraction contractors. To date, only the west and south limits of the cemetery have been found. The excavation area is thought to be a section cut through the centre of the cemetery. A total of 161 inhumation graves, 33 cremation graves, 2 horse graves and 2 dog burials were disinterred. It is thought that this only represents a fraction of the number of burials at this site. It is thought that further graves have been destroyed as a result of gravel digging, deep ploughing and the routing of the M11 motorway through the northern part of the cemetery in 1977 (Evison, 1994).

The skeletons were fairly well preserved with 88 adults and 83 sub-adults (42 males, 63 females and 66 not assigned a sex). A group of males were buried, some without weapons and some without grave goods. Evison (1994) suggests that these may be the burials of non-local merchants (buried without weapons) and slaves (buried without goods). It is thought that most of the excavated graves date from the 6th Century, however, it is considered probable that 5th and 7th Century graves were in the destroyed areas (Evison, 1994). The remains are curated at the University of Southampton

5.1.3 Taphonomic alterations at Great Chesterford

Waldron (1994b) noted that the remains from Great Chesterford had been stored badly, with boxes containing multiple elements rather than discrete skeletons. There was evidence of postmortem breakage of a substantial number of bones with two or more parts of the broken bones being stored in separate boxes. The bones were reconstructed into discrete skeletons when the inhumation numbers were visible.

5.2 Buckland Cemetery

5.2.1 Historical background to Buckland Cemetery

Buckland (TR310430) can be found in Kent. The cemetery is situated on Long Hill on the east bank of the river Dour and is separated from Castle Hill by the Dour Valley. Buckland is in close proximity to the Roman port of *Portus dubris*, built at the mouth of the river Dour, the only significant break in almost 20km of chalk cliffs. There is widespread evidence of Roman settlement in the countryside around Dover and also native settlements and associated cemeteries (Philp, 1989; Parfitt, 2002:394). Chapter 5 Materials and Methods

5.2.2 Archaeological background to Buckland

There have been two major campaigns of archaeological excavation at Buckland (see Figure 5-2). The first campaign was undertaken between 1951 and 1953 led by Vera Evison. It is thought that the cemetery was first disturbed by the building of the Dover-Deal railway and Evison found evidence of grave disturbance during her excavations. The building of a new housing estate at Long Hill led to the further discovery of a number of graves containing human remains and the Inspectorate of Ancient Monuments mounted a rescue excavation. In total, 171 graves from AD 475-750 were excavated. These remains are curated at the Natural History Museum.

In 1994 development of another housing development at Castle Hill was planned. This estate was situated on the slopes below the original cemetery and separated from it by the Dover-Deal railway cutting. The proximity of this site to the important Anglo-Saxon cemetery excavated by Evison led to a one-day evaluation by South-Eastern Archaeological Services on behalf of the Heritage Conservation Group of Kent County Council to check for the presence of outlying graves. Twelve graves were identified during this evaluation necessitating more extensive excavations to locate, record and remove any further graves before the site was destroyed by construction activities. Canterbury Archaeological Trust performed the excavations in July 1994, leading to the discovery of a total of 244 graves. The remains from this campaign are curated at the offices of the Canterbury Archaeological Trust in Kent and were used in this study. The site is now fully developed as a housing estate.

5.2.3 Taphonomic alterations at Buckland

Trevor Anderson evaluated the lower Buckland cemetery remains (Anderson, 2012). His report does not contain details of the methods used to identify sex and age group, although he does write that the analysis was based on osteological evidence, rather than by considering the grave goods. Bone preservation is very poor in this cemetery limiting the number of spines available to this study. This is a common finding in Anglo-Saxon cemeteries (Anderson, 1990; Hirst, 1985; Philp, 1973; Waldron, 1994).



Figure 5-2 Overall plan of Buckland (Long Hill) Anglo-Saxon cemetery showing areas excavated in 1951-3 and 1994: from Parfitt and Anderson (2012)

Chapter 5 Materials and Methods

5.3 St Anne's Road

5.3.1 Historical background to St Anne's Road

St Anne's Road cemetery is sited in Upperton, (OV603603) a suburb of Eastbourne East Sussex. St Anne's is a multi-period site on the upper southwest-facing slope of St Anne's Hill facing the river Bourne and the Bourne valley. The site lies on the margins of the South Downs adjacent to a trackway across the hill ridge.

5.3.2 Archaeological background to St Anne's Road.

This area has been subject to archaeological study since the Victorian Period with numerous excavation campaigns taking place. There is evidence of settlement from the Bronze Age onwards with a Bronze Age Barrow ditch being identified during excavations by Spurrell (1882) and evidence of Iron Age settlement and field systems. The first evidence for the presence of an Anglo-Saxon cemetery was unearthed at the end of the 19th Century during the construction of a school (Spurrell, 1882). In 1991 a series of archaeological trial trenching was carried out prior to redevelopment of the site as a housing estate. This fieldwork confirmed the existence of an important and extensive Anglo-Saxon cemetery with 27 inhumations and three cremations being excavated. East Sussex County Council requested further excavation in 1992 in an area south of the 1991 excavation area intended as a new college annex. A further 42 graves were identified but not examined intrusively. The college decided not to proceed with the extension scheme as extensive excavation works would have been required and the trench was refilled.

A further series of excavations were carried out in 1997 and 1998 by Archaeology South East, which led to the site being scheduled under the Ancient Monuments and Archaeological Areas Act, 1979 in recognition of the importance of the site. It is thought that there are still a large number of undisturbed graves in the area.

5.3.3 Taphonomic alterations at St Ann's Road

The preservation of the skeletal remains is poor with skeletal elements being fragmentary or degraded, although most skeletal elements are represented. After excavation in 1991, the remains were stored in poor conditions in a garage, which led to further deterioration. Of the 192 excavated individuals, 150 were adult and 42 were infant or juvenile. The remains are curated at Eastbourne Town Hall
5.4 St Bride's Crypt and St Brides Lower Cemetery

5.4.1 Historical background to St Bride's Crypt and Lower Cemetery

St Brides Church is situated in Fleet Street, London and is one of the oldest in the city. The church is built on the site of a Roman building which is thought to be the remains of a small mausoleum in a late Roman cemetery (Grimes, 1968). Christian worship took place at the site as early as the 7th Century. The church was destroyed in the Great Fire of London in 1666 and rebuilt by Sir Christopher Wren on top of the previous 6 churches that had stood on this site. It is still an active church today.

The Church sustained bomb damage during the Second World War, which led to excavations in the 1950s by Professor Grimes. The excavation revealed a large number of skeletal remains from a medieval charnel house and individuals interred in the crypt. It is thought that the charnel house contained the remains of as many as 7000 individuals. The crypt contains the remains of middleclass Londoners interred in the late 18th to early 19th century. 227 of these were interred in coffins with metal plates associated with each internment. These metal plates were inscribed with the name, age at death and date of death (Scheuer and Bowman, 1995). In 1854 the crypts were sealed by an Act of Parliament and left undisturbed until damaged in the Second World War.

St Bride's Lower Cemetery was founded in response to overcrowding in the St Bride's Churchyard and the burials are thought to date from AD 1770-1849 (Miles and Conheeney, 2005). It is situated at 75-82 Farringdon Street, London EC4. St Bride's Fleet Street and Lower St Bride's Churchyard are part of the same parish population but represent different echelons of society. It is thought that this population contained individuals from Bridewell workhouse and Fleet prison as well as other individuals of low socioeconomic status. The parish records contain detailed information about these individuals but as they were of low status, they were not buried with coffin plates making individual identification impossible.

5.4.2 Archaeological background to St Bride's Crypt and Lower Cemetery

Post War excavations of St Bride's church in the 1950's by Professor W. F. Grimes led to the identification of a crypt as part of the Chapel. This excavation was at the behest of the then curate Rev C.M. Armitage (Milne and Reynolds, 1994). The remains of the 227 individuals from the crypt were analysed and recorded by the Centre for Human Bioarchaeology. The remains are curated at the Church. The excavations at St Bride's Lower Cemetery led to the recovery of 606 individuals, of which 544 were analysed by the Centre for Human Bioarchaeology based at the Museum of London. The remains are curated at the Museum.

5.4.3 Taphonomic alterations at St Bride's Crypt and Lower Cemetery

Generally, bone preservation is good in both sets of remains with high levels of skeletal completeness.

5.5 Inclusion criteria

Adult skeletons only are included in this study as the estimation of sex in a non-adult is complex (Mays and Cox, 2000). Skeletons from the cohorts without known age at death were classified as adult when they exhibited epiphyseal fusion of the long bones (with the exception of the medial clavicle which fuses in the early twenties, (Langley-Shirley and Jantz, 2010)), closure of the spheno-occipital synchondrosis, Ascádi and Nemeskéri (1970), dental wear and antemortem loss (Miles, 1963; Brothwell, 1981), altered morphology of the pubic symphysis (Todd, 1920; Brooks and Suchey, 1990) and morphological change in the auricular surface (Lovejoy et al 1985),

An important inclusion criterion is the state of preservation of the skeletal material. The bones needed to be in a good state of preservation for estimation of sex, age, asymmetry and measurement of zygapophyseal angle and size to be undertaken. Table 5-1 summarises the numbers of skeletons from each cemetery that are suitable for inclusion in this study.

Cemetery	No of skeletons excavated	No of skeletons with sufficient preservation to be included in the study
Great Chesterford	167	20
Buckland	244	8
St Anne's	192	25
St Bride's Crypt	224	67
St Bride's Lower	544	60

Table 5-1 Summary table of skeletal remains included in this study

5.6 Exclusion criteria

Skeletal material was excluded from this study if on examination, it was deemed to be from a non-adult, if there was insufficient preservation of the bones necessary to collect the appropriate data and also if there was evidence of a pathology that would affect normal activity e.g. fracture, amputation of a limb, diseases affecting the spine e.g. tuberculosis, diffuse idiopathic skeletal

hyperostosis, scoliosis (idiopathic or acquired) or other pathology that would affect the normal functioning of the body. Transitional vertebrae (see section 2.3.7) were also excluded to limit the natural variation of the data.

5.7 Method

This section describes and discusses the methods used to collect and analyse the data. The difficulties of estimating sex and age using morphological and metric methods are identified. The methods chosen for use in this study are presented in detail. Observational and metric methods were used to estimate the sex and age at death of individuals without coffin plates. Robusticity, asymmetry, the presence of osteophytosis and the severity and extent of OA of the zygapophyseal joints are all estimated using standard osteological techniques which are discussed within the text.

5.7.1 Methods of analysis

This data were collected from Great Chesterford Cemetery Essex, Buckland Cemetery Kent, St Anne's Cemetery, East Sussex and St Bride's Church, Fleet Street, London. The sex, age and in some cases occupation was known for the individuals from St Bride's Church crypt. This is of great value in this study as age related/occupational changes can be accurately measured and provide a reference sample. Bocquet-Appel and Masset (1982) identified the importance of using reference sample for age related studies. They identified that no one reference collection is ideal but must have several characteristics, i.e. the sex and age-at death of the individuals are known, and that the collection contains a good representation of the population of interest. There is no Anglo-Saxon reference collection with this information available and therefore the St Bride's sample will be used in this study.

All the data collected from each sample was initially recorded on paper forms that were specifically designed for this study. No standard recording form exists for recording morphological changes in vertebrae and therefore an appropriate form was designed for recording information related to OA score, sagittal vertebral facet angle, facet height and width. The form also included sections for recording of data relating to sex, age and osteometrics (see Appendix E). The recording form section for estimating sex, age and long bone measurements was designed around the template from pre-existing recording standards sheets (Buikstra and Ubelaker, 1994; Brickley and McKinley, 2004) and the section of the recording form for facet measurement and OA scoring was designed specifically for this project. Macroscopic examination was used to collect data from the skeletons. Current bioarchaeological methods as described in the following sections were

85

chosen for sexing and aging techniques because they are commonly used in British research and would create a data set that would be comparable with other studies. Hard copy data was transferred to Excel spreadsheets and imported into SPSS for data analysis. The statistical methods used are described in 5.7.8.

5.7.2 Demographic data - sex

The majority of the individuals from the cemeteries used in this study had sex and age estimated as part of the post-excavation process, however, it was decided to reassess these data by applying the same standard sex and age methods to all of the skeletons from undocumented burials (i.e. age at death and sex not discernible from coffin plate) with the aim of preventing inter-observer error.

5.7.2.1 Estimation of sex from archaeologically derived samples

The difficulties in correctly estimating sex and the various methods that can be used are described in this section. The sex of an individual can be inferred using morphological and/or metric techniques. Spradley and Jantz (2011) emphasise the distinction between assessment of sex using subjective visual analysis of morphological features and estimation of sex using metric data and recommend the use of metric sex estimation (supported by evidence of a greater number of recently published papers using discriminant function analysis of metric data when compared to the number of papers using visual assessment techniques). However, Stewart (1979) recommends combining methods and using metric data to validate visual analysis. Adams and Byrd (2002) support this concept but warn that visual sex assessment is affected by subjectivity and has higher inter- and intra-observer errors than metric methods (Adams and Byrd, 2002). In contrast, Milner and Boldsen (2012a) consider that, in relatively complete skeletons, an experienced osteologist would be as accurate using morphological as metric methods but add the caveat that quantitative methods are more certain and it should be remembered that not all bone dimensions are equally sexually dimorphic. Thus, discrimination should be used when deciding what measurements are appropriate for each element. Morphological and metric variation can best be seen in structures related to reproduction. The zygapophyseal joints have no reproductive specific role and therefore may lack dichotomous variation. The differences that may be observed may be due to females being generally smaller than males. The effect of physical activity in particular the sexual division of labour may also play a part in differential zygapophyseal joint size.

The next two sections will discuss morphological and metric variation in more detail.

5.7.2.2 Morphological variation in skeletal structure

When using morphological variation to assess the sex of a skeleton, some bones are more sexually dimorphic than others, particularly the pelvis and skull. Pelvic shape differences are related to parturition and locomotion and are thought to be the most reliable features for sex determination (Brothwell, 1981) with females having a wider pelvic inlet, wider sub-pubic cavity and wider greater sciatic notch than males. High levels of accuracy have been reported when assessing morphological analysis of the pelvis using known sex reference collections; Bruzek (2002) reported a 95% success rate when using the Paris and Coimbra collections. His method utilized eleven pelvic traits from the pre-auricular surface, greater sciatic notch, the composite arch between the sciatic notch and auricular surface, the inferior pelvis and ischiopubic proportions. His results showed that there was considerable variation in the contribution of each trait towards a correct assessment of sex but using multiple variables a greater level of accuracy could be achieved.

Morphological variation in the skull using sex indicators outlined in Buikstra and Ubelaker (1994), i.e. nuchal crest, mastoid process, supra-orbital margin, supra-orbital ridge and mental eminence, tend to be less accurate than pelvic features when estimating sex. Williams and Rogers (2006) achieved an 80-90% accuracy rate using the dimorphic features of the skull. Generally, the skull indicators are larger and more robust in males but in some populations this can be problematic as ancestry can affect some cranial traits making sex estimation more difficult (Garvin et al., 2014). Increased robustness in females in certain populations can lead to misidentification (Walrath et al., 2004; Walker, 2008). The subjective nature of morphological analysis can also lead to errors of classification. Buikstra and Uberlaker (1994:16) state:

"Estimates of sex therefore can be difficult if the observer is not familiar with the overall pattern of variability within the population from which the sample is drawn".

The methods used to estimate the sex of an individual recognise, to a degree, the morphological variation that can occur in different samples by providing a continuum from male to female by including the classifications; typical male, probable male, unknown, probable female, typical female (Sofaer, 2006). This classification system allows recognition that there can be considerable variation within a given trait and an increased dispersion of measurements in both sexes produces an increase in the overlap of corresponding distributions and hence a lower degree of dimorphism leading to greater difficulty in dichotomous characterisation (Marini et al., 1999).

5.7.2.3 Metric variation in skeletal structure

Incomplete, fragmented or commingled remains are frequently encountered in bioarchaeological contexts and the typical anatomically dimorphic features of the pelvis or skull may be absent or damaged making qualitative methods for sex estimation problematic. As an alternative, the quantitative assessment of metric traits can be used. Metric estimation of sex is based on differences in size and/or shape in male and female bones and is based on taking measurements of various dimensions of skeletal material.

There are several strengths in the use of metric methods. The methods rely on standard osteometric points and are considered by some to be more repeatable than the reliance on subjective judgement against a scale of expression for the less sexually dimorphic bones leading to lower intra- and inter- observer error (Adams and Byrd, 2002; Spradley and Jantz, 2011). McKeown and Schmidt (2013) emphasise the importance of fully understanding the parameters of the measurement to be taken and the need for accuracy of observation, as minor errors in data collecting can lead to a significant effect on the validity of the results calculated. Konigsberg and Hens (1998) consider that metric variables have higher levels of simplicity and consistency in their recording and as continuous variables can be analysed using powerful statistical methods. However, there is conflicting argument on the validity of metric methods over morphologic al methods. Ramsthaler et al. (2007) compared metric and morphological methods for the determination of sex from the crania of reference collections from Frankfurt and Mainz (Germany). They used Fordisc software (discussed in the section below) to estimate the sex from the metric data and found that the average accuracy for both sexes was 86% (metric) vs. 94% (morphological). They explain their discrepancy as a weakness of Fordisc because their sample may not match the software reference sample drawn from the US American forensic database for ancestry, age distribution and socio-economic status.

Sexual dimorphism in size can be estimated in a number of different ways.

a) Single measurements: Single measurements capture variation in size and some measurements (e.g. humeral or femoral head diameters) can identify a reasonable separation of the sexes (Milner and Boldsen, 2012a). Unlike multiple measurements, single measurements do not capture a sense of shape, only size. There are practical reasons for using single measurement methods for sex estimation; the data are quick and easy to collect, they can be a reasonable estimator of sex and are suitable for fragmented or incomplete skeletons where the number of measurable options is reduced. One of the key issues is that, if there is an overrepresentation of one sex in the skeletal sample with one sex dominating the other, the probability that a bone of a certain size is male or female can be affected (Milner and Boldsen, 2011). In summary, the single measurement

method is suitable for a speedy estimation of sex, particularly in fragmented remains in populations that have an even distribution of males and females.

b) Simple ratios: These represent bone shape in contrast to single measurements described above that represent bone size. In comparative studies of sexual size dimorphism, simple ratios can be transformed into indices to reduce the bias that can occur when size alone is used as an indicator of sex (Arsuaga and Carretero, 1994) e.g. sexual dimorphism index (SDI). This index is usually calculated as a ratio of mean values of a specific measurement from males and females (see section 5.7.7.3). However, there has been some criticism of the use of indices:

i). Ratios do not tend to have normal distribution and therefore violate the assumption of normality required for parametric statistical tests (Atchley et al., 1976),

ii). Ratios do not remove the effects of size and require the assumption that the numerator and denominator are collinear with an intercept of zero (Jackson et al., 1990).

iii). Ratios increase problems related to measurement error leading to wide confidence intervals (Sokal and Rohlf, 1995).

Smith (1999) summarised the issues relating to the use of ratios/indices in comparative studies relating to sexual size dimorphism (SSD) and compared 25 data sets from the literature to duplicate a number of statistical procedures commonly used. He concluded that there was no compelling reason for ratios not to be used in the analysis of SSD and that the logarithmic ratio of male/female and linear scale ratios are equally appropriate for use in sexual dimorphism studies. In summary, ratios can be of limited use when estimating the sex of a skeleton, as a set of indices generated from a sample will not indicate which individuals do not fit into the expected range.

c) Complex multivariate analyses; within-group differences can be maximised using statistical tests such as discriminant function analysis (DFA) or logistic regression. These statistical processes classify the sex of unknown individuals and calculate the probability that they can be ascribed to a group e.g. sex or ancestry, by examining variability between all groups. They can be performed using statistical software. Studies have indicated that DFA equations are population specific (Introna et al., 1998; Slaus et al., 2003; Sumati and Phatak, 2010). Thieme and Schull (1957) noted that the applicability of DFA was dependent on the degree of sexual dimorphism between populations and the degree to which sexual dimorphism is the same within populations. They reported that inter-population differences within a single sex is often more pronounced than intra-population differences. This led to the development of population-specific discriminant functions for sex estimation among various populations (Murphy, 2002; Spradley et al., 2008)

DFA is considered to be an objective statistical technique for sex determination (Hsiao et al., 1996). The method selects the minimum number of traits that yield maximum discriminant effectiveness (Isçan and Helmer, 1993) and uses posterior probability and typicality probability to calculate the classification probabilities (Albanese et al., 2008). It can only be used to place continuous variables into two categories (e.g. males and females) and is suitable for analysis of facet size and angle. DFA assumes that the sample is normally distributed for the trait under examination (Field, 2013). It is a commonly used statistic in studies of skeletal sexual dimorphism. However, many of the studies make no comment on the normality of distribution of the data used.

Logistic regression is robust against violations of normality and homoscedasticity and has fewer restrictive assumptions (Field 2013). It allocates an individual of unknown sex to a set of *a priori* defined groups (male or female) and generates posterior probabilities of group membership. It can be used to place continuous and categorical variables into defined groups. It is therefore useful to use logistic regression for all data that is not amenable to DFA.

One of the most noted examples of multivariate analyses can be seen in the computer program FORDISC (Jantz and Ousley, 2005). This program is able to provide the probability that an individual is male of female or belongs to a particular demographic group by comparing standard cranial and postcranial measurements from an unknown individual against a large database of known individuals. The database is compiled from measurements collected from individuals of known ethnic origin and sex that were born after 1900 from the US American Forensic Database. The error risk due to population specificity is reduced with this program because data was collected from a broad geographic sample; however, there are limitations to its use. As with all statistical programs, data can only be placed into the categories available from the software (Simmons and Haglund, 2005). Ramsthaler et al. (2007) identified that Fordisc cannot be used as a single method for the estimation of sex of recent skeletal remains from Europe as the database is not compatible. Milner and Boldsen (2012) advise that the study skeletons should come from the same population as the reference population for there to be any validity in sex estimation. FORDISC 3.0 has been programmed so that classification will not be allowed if the post probability is too low, reducing the risk of misclassification (Morris, 2010).

Probabilistic Sex Diagnosis (DSP) is another computer program that can be used to determine the sex of an individual. The database consists of os coxae metric data collected from 2040 adult specimens from 12 different reference populations (Murail et al., 2005). The authors claim close to 100% accuracy in sex estimation. The sample populations are of known age at death and sex from different areas of the world (Europe, Africa, North America and Asia). They date from the

18th to early 20th century. This database is used in this study as the sample populations are more closely matched to the study population for location but do not match closely for period. In order to reduce error, multiple methods of sex estimation including DSP are used in the study.

There are limitations to the use of bioarchaeological databases because the reliability of sexing methods decreases when unrelated population samples are compared (Van Gerven and Oakland, 1973; Calcagno, 1981). Secular changes and differential activity patterns are thought to lead to variation in skeletal metric and non-metric features and hence differences in sexually dimorphic features in different populations (Işcan, 2005; Case and Ross, 2007).

A number of manuals defining the most utilised methods for the estimation of sex and age at death have been published (Ascadi and Nemeskeri, 1970; Bass, 1995; Buikstra and Ubelaker, 1994; Krogman and Işcan, 1986). However, these methods do not account for population specific variation in human skeletal remains. This has led to many regional studies relating to these variables, leading to better intra- and inter-populational variability comparisons e.g. Hanihara, 1952; Konigsberg and Frankenberg, 1992; Plato et al., 1994; Jackes, 2000; Hoppa and Vaupel, 2002; Hou et al., 2012; Nikita, 2012.

5.7.2.4 Methods used to estimate sex.

Good bone preservation is necessary for accurate sex estimation in adults (Milner and Boldsen, 2012). Typical accuracy for sex estimation in adults from morphological traits is 90-95% when using the pelvis and 80% when using the skull (Krogman and Işcan, 1986). In this study, a multifactorial approach was adopted when assessing the sex of an individual, with pelvic morphology being given the highest priority (Cox, 2000). Data for sex determination was recorded using the macroscopic assessment methods of Buikstra and Ubelaker (1994) and derived from the sexually dimorphic features of the skull, pelvis and femoral head. In order to reduce the risk of error between populations, all features were seriated before final assessment of the sex of an individual (White et al., 2012). Table 5-2 overleaf lists the features used in this study to ascribe a sex to an individual

Skeletal Element	Morphological/metric feature	Reference
Skull	Nuchal crest	Ascadi and Nemeskéri (1970)
	Mastoid process	
	Supra-orbital margin	
	Supra-orbital ridge	
	Mental eminence	
Pelvis	Ventral arc	Phenice (1970)
	Sub pubic cavity	
	Ischiopubic ramus bridge	
	Greater sciatic notch	Walker (2005)
Femur	Femoral head diameter	Stewart (1974)

Table 5-2 Cranial, pelvic and femoral features used to determine adult sex

Ubelaker and Volt (2002) identified Phenice's method to be the most accurate method for sex estimation and therefore this method was given the highest priority. In the absence of the pubic bones, the morphology of the sciatic notch was given the next priority level (Walker, 2005). In the absence of a pelvic girdle, cranial morphology or morphometrics were used to assess sex. High levels of accuracy (92.5-99%) can be achieved using skeletal measurements (Giles, 1970; Cox, 2000). The measurements taken can indicate a male or female individual depending on the robustness or gracility of the bones being measured (Bass, 1995). Many factors can affect skeletal robusticity (genetics, environment, nutrition, maturation rate, physical activity and health) and females could display male attributes or vice versa (Wolfe and Gray, 1982). In this study, Stewart's (1974) method in respect of femoral head diameter, was used to sex bones from individuals where bones with sexually dimorphic features were absent or unrecordable due to poor bone preservation. He identified that individuals with femoral head dimensions between 44 and 46 mm could not be sexed with any accuracy but measurements below 43 mm and above 47 mm can be identified as females and males respectively with 90% accuracy. This method was used to assign sex to Buckland B282B and B309.

Table 5-3 Use of femoral head diameter to estimate sex

Measurement	Males	Females	Reference
Femoral proximal/distal head diameter	>48 mm	<43 mm	Stewart (1974)

When using morphometric data to assess the sex of an individual, age was taken into account as older post-menopausal women can develop "male" cranial features which could potentially lead them to being identified as male (Walker, 1995).

To further increase the probability that sex has been estimated correctly, pelvic measurements were taken and assessed using the Probabilistic Sex diagnosis database [DSP] (Murail et al. 2005). This tool uses discriminant function analysis to calculate the individual probability that an individual is either male or female by comparing innominate measurements against a global reference sample. A minimum of four measurements is required for probability calculation. The measurements are listed in Table 5-4. The excel spreadsheet and measurement details are available from: http://www.paceau.u-bordeaux1.fr/publication/dsvp1/html. The pelvic evidence is considered to be more accurate in cases where the skull and pelvis provide conflicting diagnostic signs (Cox, 2000).

Individuals that were not sex distinctive due to anomalous features or absent diagnostic bones were excluded from this study. In the event of an anomalous assessment (probably male, probably female or unknown) the overall picture for that individual were reviewed from all the features analysed in conjunction with the position of the feature in the seriation rank. If it is still problematic to ascribe a sex to that individual, they were excluded from this study.

In order to be able to fully consider the effects of sexual dimorphism, this study includes data from 180 individual skeletons analysed by the author or of known sex.

Variable	Brief Definition	Reference
PUM	Minimum distance from the superior and medial point of the pubic symphysis and nearest point of acetabular rim at lunate surface	Bräuer 1988
SPU	Pubic breadth between the most lateral acetabular point and the medial aspect of the pubis. The measurement should be perpendicular to the major axis of the os pubis	Gaillard 1960
DCOX	Maximum height of os coxae measured from the inferior border of the os coxae to the most superior margin of os coxae	Bräuer 1988
IIMT	Distance from the posterior inferior iliac spine to the anterior border of the greater sciatic notch. The axis of measurement should be perpendicular to the anterior border	Bräuer 1988
ISMM	Distance from the most anterior inferior point of the ischial tuberosity to the furthest point on the acetabular border	Schulter-Ellis et al. 1983
SCOX	Distance between the anterior superior iliac spine and the posterior superior iliac spine	Bräuer 1988
SS	Distance between the anterior inferior iliac spine and the deepest point on the sciatic notch	Gaillard 1960
SA	Distance between the anterior inferior iliac spine and the auricular point (defined as the intersection between the arcuate line with the auricular surface)	Gaillard 1960
SIS	Distance between the lateral border of the acetabulum and the mid point of the anterior portion of the sciatic notch	Bräuer 1988
VEAC	Maximum vertical diameter of the acetabulum	Bräuer 1988

Table	5-4 Pr	obabilistic	sex diagno	osis databa	ase (DSP)	measurement	description
	-						

5.7.3 Demographic data- age

5.7.3.1 Age estimation from archaeologically derived samples

Knowing the age at death of a particular individual is essential when assessing the biological profile of a subject and to enable valid comment to be made about human characteristics that are subject to age-related change e.g. activity, genetics, health and overall lifestyle (Calce, 2012). Chronological age does not have external factors affecting it; it is purely a measure of time that has passed. In contrast, physiological age refers to the aging processes that occur in the body (growth, development and the effects of aging). Physiological changes are not synchronous and

can vary from individual to individual (e.g. menarche and menopause) (Gerdhem and Obrant, 2004). Over time, the effects of environment and activity cause a divergence in physiological and chronological age increasing error into age calculations particularly in the assessment of old individuals (Uhl, 2013). Low reliability in age estimation methods can result from the fact that chronological age and physiological age only rarely coincide (Campanacho et al., 2012). Age at death estimates assess the physiological age not the chronological age of an individual and broad age ranges are required for each age-related change in order to account for external factors (Santos, 1996).

There is considerable criticism about estimate accuracy, precision and reliability when ascribing an age at death to an individual with little consensus amongst osteologists as to which ageing method is the least unreliable (Hoppa and Vaupel, 2002; Milner, 2008). Mays, (1998: 50) stated

"At present, the lack of a wholly satisfactory technique for estimating age at death in adult skeletons from archaeological sites is one of the most thorny problems facing human osteoarchaeology."

The use of multiple methods for assessment of age of death are recommended to address the issues identified by Mays, (1998) (Buikstra, 2006). Lovejoy et al., (1985) developed the original multifactorial method. Milner and Boldsen (2012) use this method to enable a more accurate assessment of age at death using a Transition Analysis system. They used the pubic symphysis, sacro-iliac joint and cranial sutures in their procedure and found that the three areas did not perform equally well when evaluated against 252 known-age modern males and females. The cranial suture assessment performed least accurately and from this result Milner and Boldsen stated:

"It is hardly worth bothering with sutures as their positive effect is, at best, negligible, and their inclusion often worsens estimates". (p. 107).

In contrast, the use of changes in the pubic symphysis worked best when estimating age, however, there is a consistently low reliability in results, particularly when the methods are applied to a different population than the population used to develop the method (Santos, 1996). Occupation and/or physical activity can both be a factor in the degenerative changes seen in the os pubis (Scheuer, 2002) and may be a factor in the variation seen in rate of ageing in this bone. Heavy mechanical loading can lead to increased degeneration and apparent ageing of the bone. Scheuer's study (2002) captures a general sense of a population's age-at death distribution but is not particularly sensitive to estimate an individual's age.

Age estimation is more problematic for adult skeletons than for juvenile remains because juvenile ageing methods rely on more highly constrained developmental processes, whereas adult age estimates are derived from degenerative characteristics that accompany the aging process, e.g. changes in the pubic symphysis and sacro-iliac joint may occur at different ages in individuals and are only weakly associated with age (Milner and Boldsen, 2012). Degenerative changes that accompany advancing age can be seen as changes in the pelvis and these variations are used in the methods of Suchey-Brooks (Brook and Suchey, 1990), Todd (1920) and Lovejoy et al. (1985). There are many factors that impact upon the degeneration that is observed with increasing physiological age (e.g. nutrition, activity, diet, disease, stress) (Roberts and Manchester, 2005).

Any estimation of age is in effect a probability statement of accuracy and precision as there is no simple method to define age. Different methods give different determinations of age (White and Folkens, 2005). The multifactorial method for aging adult skeletons outlined by Buikstra and Ubelaker (1994) as appropriate for a British population was used in this study, as there is no one definitive method for estimation of age. The use of commonly used methods allows for cross study comparison. Lovejoy et al. (1985) recommend the seriation of skeletons for each indicator in the sample under investigation to maximise the information from biological age and to reduce observer error, however, Konigsberg and Frankenberg (2002) consider that seriation creates a problem of lack of independence and can complicate decision making when transitioning from one age to another. In order to increase the reliability of age at death assessment in this study, each population was considered separately, and all remains were seriated by the skeletal part being assessed (e.g. pubic symphysis). After seriation, the individual remains were placed in the age interval they most closely resembled according to the seriation sequence for each age interval after the method of Bedford et al. (1993).

Early researchers such as Hrdlička thought that it was sufficient for an anthropologist to place a skeleton into sub-adult, adult or senile categories by using observation of multiple features associated with ageing, such as pelvic auricular surfaces and cranial suture closure, however despite continued use of such an approach, there are limitations associated with this. The use of age categories can be problematic as it is necessary to assume that all individual age estimates have the same degree of error (Boldsen et al., 2002). A possible solution to this problem is to use broad categories that encompass considerable morphological variation although this carries a risk of blending age information.

5.7.3.2 Methods used to assess age

In this study, multiple aging methods were used to estimate age at death, which individuals being placed into three categories (young adult, middle adult and old adult). The methods used are described overleaf.

Buikstra and Ubelaker (1994) identified use of the multifactorial methods listed below as being appropriate for a British population: the minimum and maximum age at death for each individual from undocumented populations was estimated for each of the methods listed in Table 5-5. Lovejoy et al. (1985) recommended the use of seriation series for each indicator under investigation to maximise the information that could be gathered about biological age at death and to reduce observer error. A seriation sequence was established for each indicator of skeletal age at death by comparing the degrees of variation of each indicator and arranging the samples in ascending age order.

Skeletal element	Morphological/metric feature	Source
Skull	Dental Wear	Miles (1963) and Brothwell (1981)
Pelvis	Pubic symphysis	Brooks and Suchey (1990)
	Auricular surfaces	Lovejoy et al (1985)

Table 5-5 Age at de	ath estimation	methods used	l in this study
Table J-J Age at ac	attricstiniation	methous used	a ini tinis study

Once the maximum and minimum age determined for each individual was determined, the individual was included into a broad age category (Table 5-6) as described by Buikstra and Ubelaker (1994). The use of broad age range categories is recommended due to the difficulties of accurately estimating the age at death from skeletal material to less than a 15-year range (Cox, 2000).

A random sample from the documented St Bride's population was also assessed for age. (A random number generator computer program was used to identify the individuals to be assessed) and the accuracy of estimated age and actual age compared.

The dental wear methods of Miles (1963) and Brothwell (1981) were used for this study. The decision to use dental wear methods was supported by Mays (1988) who considers that these are one of the most accurate methods for ageing adult material.

Schwartz (1995:195) lists a number of methods that have been developed to assess age from the symphysis pubis, however, the majority of published research rely on the Todd and Suchey-

Brooks methods (McKern and Stewart, 1957; Gilbert and Kern, 1973; Meindl et al., 1985). Testing of these methods against reference populations has indicated that they are capable of producing accurate results with the Suchey-Brooks method (Brooks and Suchey, 1990) showing greater accuracy than the Todd method (Mays, 2010). The Suchey-Brooks method uses separate scoring systems for males and females making it appropriate to use for this study.

One of the advantages of using Lovejoy et al. (1985) method for estimating age at death using the auricular surface of the ilium is that it has been demonstrated to discriminate well between the middle adult and old adult age groups (Bedford et al., 1993). This discrimination is important, as the samples under study need to be allocated to an appropriate age range group with as much accuracy as possible.

Table 5-6 Age categories used to divide the population into life phases

(After Buikstra and Ubelaker, 1994; Cox, 2000).

Category	Age Range (Years)
Young Adults	21-35
Middle Adult	36-45
Old Adult	46+

The age range was recorded in the SPSS database as an independent variable. Population demographic profiles were drawn from the age range (Chamberlain, 2006). The age ranges were used to test for age-related variability in facet orientation and size both within and between age cohorts (Roberts and Connell, 2004). In order to be able to fully consider age related changes, this study only includes data from skeletal material aged by the author and from known age individuals.

5.7.4 Data collection methods for facet size and angle

5.7.4.1 Facet size

Facet joint articular surfaces may undergo functional adaptation as a response to use which may be correlated with the amount of stress placed upon it (Dhall, 1984). Panjabi et al. (1993) and Patel et al. (2007) used width and height measurements and width/height ratios to compare facets size at different vertebral levels. They assumed the facet surface area was circular and used the formula :

AREA (A) =
$$\pi \times (width \times height) \div 4$$

Their calculation did not take facet depth into account and assumes that the surface of the facet is flat. Facet shape varies widely with vertebral level, therefore the use of simplistic mathematical formulae to calculate area would not be appropriate for accurate area comparison. The assumption that the facet is flat is also assumed by Buikstra and Ubelaker (1994). They multiplied facet height measurement with facet width. This method was used in this study facets are more rectangular/oval than circular in shape and it was considered that this calculation would allow for greater accuracy of analysis. The calculation is not a true measure of surface area as the facet depth is not included, but does provide a basis for comparison between the facets.

The definition of the facet edge varied between vertebrae as illustrated in Figure 5-3. Clearly defined facet edges (as seen on the left facet) were used where possible to take measurements. In the absence of a defined facet margin, (as seen on the right facet), measurements were taken from the surface area that appeared to be involved in articulation (outlined in Figure 5-4). As all data was collected by one observer the same criteria were applied to all measurements minimising error.



Figure 5-3 Variation in facet border definition T10, B282B



Figure 5-4 Margins of facet border for measurement T10 B282B

A Mitutoyo digital caliper (calibrated to 0.01 mm, measurement taken to 0.1 mm) was used to take width and height measurements from each of the four facets on each vertebra after the method of Patel et al. (2007). The two measurements were taken at right angles to each other at the greatest point in each direction. Figure 5-5 illustrates the measurements taken to assess the height and width of the right inferior facet of C5 from GC14. Each facet was recorded separately as it is not uncommon for be variance in size to be seen. This anomaly is illustrated seen in Figure 5-6, which illustrates the differences between the inferior facets of C6 from GC14.



Figure 5-5 Illustration of position of facet height and width measurements(C6 from GC9 (author's image)



Figure 5-6 Differences between C5 left and right inferior facet size (C5 from GC100 (author's image)

5.7.4.2 Facet angle

The link between facet joint orientation and OA has been a focus of clinical research (Fujiwara et al., 2001, Kalichman and Hunter, 2007; Kalichman et al., 2009; Çubuk et al., 2009; Wang and Yang, 2009). Fujiwara et al. (2001) demonstrated a significant association between sagittal orientation and OA in the lumbar facet joints. Resnick and Niwayama (1981) consider that the remodeling process of OA in the lumbar region of the spine alters the morphological features of the facet joint resulting in sagittal orientation.

Most clinical research has focused on the lumbar spine, as this area is responsible for many episodes of back pain. In many studies, simple covariates such as age, sex and body mass index (BMI) have not been examined and the relationship between facet orientation and OA, whilst adjusting for these covariates, has not been clearly defined (Kalichman, 2009). The majority of the clinical studies have used CT or MRI imaging from in vivo studies to assess the extent and severity of OA. Jurmain (1999) considers the presence of eburnation to be pathognomic for OA. Eburnation is not visible in medical scans and so the diagnosis of OA is different in clinical and archaeological samples. The use of dry bone archaeological samples facilitates the study of articular surfaces and the measurement of size and angle that is not possible in the living subject.

The use of goniometers or modified protractors is one of the main methods used by other studies to measure the sagittal facet angle (Mahato, 2011; Ebraheim et al., 2008; Patel et al., 2004; Boyle et al., 1996). A transparent finger goniometer was used to measure the sagittal angle. The vertical arm of the goniometer is lined up with two points, on the vertebra: 1) the junction between the two laminae and 2) the midpoint of the vertebral body (which is found by marking the midpoint between the widest, and deepest measurements of the vertebral body). The angle subtending

this line is found by looking through the goniometer and lining up the calibration line on the moveable arm with the lateral and medial edges of the facet being examined (See Figure 5-7). The sagittal angle for each facet is recorded separately as it is not uncommon for asymmetries to exist in the vertebrae (Van Roy et al., 1997) and facet tropism can occur with variable angles between the facets (Putti, 1927). This method can be time consuming but with practice produces good intra-rater reliability (See Section 5.8.1).



Figure 5-7 Measurement of facet sagittal angle

5.7.5 Osteoarthritis

5.7.6 Diagnostic criteria

Joint space narrowing (thought to be caused by loss of articular cartilage), marginal osteophyte formation and the presence of subchondral cysts and sclerosis are considered to be important indicators of the presence of OA in joints examined by clinical radiography (Rogers et al., 1990). Paleopathological diagnostic criteria for OA in dry bone samples have been selectively derived from clinical radiographical methods (Mays, 2002).

Bourke (1967), Rogers et al. (1987), Jurmain and Kilgore (1995) and Weiss and Jurmain (2007) consider that the most reliable diagnostic criteria for the paleopathological determination of OA, in a dry bone sample, is by the consideration of changes to the joint margin (i.e. the development of osteophytes) in conjunction with changes to the articular surface of the joint (pitting and eburnation) (See Figure 5-8). The more severe osseous changes (pitting, eburnation and osteophyte formation) are more readily identifiable rather than the subtle manifestations of initial pathogenic changes within the joint leading Sokoloff (1987:8) to state,

"It seems sensible to bypass the initial event and recognize the disintegration of the osteoarticular surface as the distinctive feature of OA".



Figure 5-8 Facet surface degenerative changes of the left superior facet of T6 from FAO901454 (author's image)

There was no evidence of hyaline cartilage on the facet surfaces in all samples, therefore examination of the bones for evidence and extent of osteoarthritic change was limited to macroscopic examination of the subchondral surface.

Each facet was scored for the absence or degree of presence and severity of eburnation, pitting and osteophytes to maximize the degree of discrimination between all the facets. For each of the scoring methods the superior and inferior facet surfaces were assessed separately. One observer examined the vertebrae. The observer was blind as to the sex and age of the individual being examined in order to reduce potential bias.

Rogers and Waldron (1995) describe the progression of osteoarthritis as beginning with osteophytic growth then the development of subchondral porosity and finally eburnation of the joint surface occurs. There is lack of consensus as to which observations definitively indicate OA (Duncan, 1979; Jurmain, 1999), however, eburnation is often considered to be the definitive indicator of degeneration of a joint surface and is thus considered to be the determinant of the diagnosis of OA (Rogers and Waldron, 1995; Arcini, 1999; Molnar et al., 2011).

Brickley and McKinley (2004) recommend that presence of porosity, osteophytes and eburnation should be recorded separately as an increase in the extent of one feature may not necessarily be paralleled by an increase in the extent of another (Atkinson, 1985). They suggest using the grading system of Säger (1969) to record the presence of each change and that the system of Buikstra and Ubelaker (1994:123) should be used to record the severity of the changes by considering the area of the facet surface affected by the change. The presence of eburnation is considered to be firm evidence of OA with pitting and osteophytes considered diagnostic when manifest at the same time in the absence of eburnation (Rogers and Waldron 1995). Rogers and Waldron's (1995) methodology was considered to be the most appropriate for this study as it covers a range of bony changes, which are indicative of OA. The prevalence of OA in each facet was calculated following the method described in 5.7.6.5

5.7.6.1 Eburnation

Eburnation is considered to be pathognomic of degenerative joint disease and is synonymous with sclerosis (Rogers et al., 1987; Bridges, 1992). Rogers and Waldron (1995:44) wrote:

"In practice little will be lost by restricting the diagnosis to include only those joints with eburnation".

Many researchers use this parameter since this criterion is consistently used and frequently referenced in bioarchaeological literature. Bridges (1996) disagrees with this approach and recommends the use of a more evaluative approach using multiple diagnostic features for OA. For the purposes of this study, eburnation, pitting and osteophytes were recorded. Joint surface was recorded as positive for OA in the presence of eburnation or when there is a combination of pitting and osteophytes. The presence of pitting or osteophytes on their own was not considered positive (See sections 4.4.3.2 and 5.7.6.3 for a description of the diagnostic criteria for pitting and osteophytes).

During the degenerative process, the articular cartilage covering the joint surface can break down. Lajeunesse (2002) identified three physiological phases in this process, whereby normal cell metabolism and signalling to promote cartilage repair and loss are affected by alterations to the cartilage biochemistry. As the cartilage begins to erode the subchondral bone is gradually exposed. Continued compressive use of this degenerating surface causes bone on bone contact to occur and the surfaces become polished (eburnation) (Ortner, 2003). Further use of the joint leads to the development of indentations or grooves parallel to the line of motion of the joint being formed (Jurmain, 1999). Eburnation indicates that a joint was still moving even though the articular cartilage had worn away (Rogers et al., 1987; Rogers and Waldron, 1995). Eburnation can be seen when the surface of the bone is converted to a dense smooth layer as a result of bone-to-bone contact. It is considered to be the most reliable pathognomic criterion for diagnosing the presence of OA (Jurmain, 1999). Eburnation can be difficult to observe. It was found that shining a bright light obliquely across the surface of the joint increased the visibility of any eburnation present especially when used in conjunction with a hand lens.

The severity is scored on a 4-point scale from 0-3. The extent of eburnation coverage of the total facet surface is scored on a 4-point scale (see Table 5-7). The scores used do not represent exact measurements but represent a ranked order of extent and severity of the OA criteria being assessed due to the difficulty in quantitatively measuring changes due to OA (Robb, 1998).

Eburnation Severity	Score
Absence of eburnation	0
Eburnation barely visible	1
Polished surface visible	2
Eburnation visible as polished surface with grooves	3
Eburnation extent	Score
Eburnation extent No eburnation	Score 0
Eburnation extent No eburnation <1/3 of surface area affected	Score 0 1
Eburnation extent No eburnation <1/3 of surface area affected	Score 0 1 2

Table 5-7 Eburnation scores

(Severity Score after Säger (1969) and Extent Score after Buikstra and Ubelaker (1994:123))

5.7.6.2 Pitting

In dry bone samples, pitting appears as multiple, irregularly shaped holes with sharp edges (Jurmain, 1999). Jurmain (1999) states that, although pitting is a frequently used criterion in the assessment of OA, the actual process is poorly defined and inconsistently applied in osteological research. In a controlled study, Rothschild (1997) demonstrated that there is no correlation between pitting and eburnation or also pitting and osteoarthritis and that porosity should not be considered a diagnostic feature of osteoarthritis. Woods (1995) recommends that the scoring of

pitting should be kept separate during the assessment of OA as its distribution on a joint surface is not concordant with that of eburnation and occurs in areas of least contact between the joint surfaces. He considers that the pores seen in pitting may occur as a result of defects in the articular cartilage or poor nutritional supply to the bone; in this case, pitting may not directly contribute to severe OA. Milgram (1983) identifies the pits as holes seen in the thinnest part of the bone as the subchondral bone is eroded. He describes them histologically as tiny cysts filled with reparative cartilage cells that can vary in size and cluster pattern. Milgram (1983) describes the pits as being part of a pathological process and although present in conjunction with degenerative disease are not necessarily part of the same disease process. Rogers and Waldron (1995) and Waldron (1995) have used the presence of pitting in combination with osteophytes or changes in joint contour as part of their diagnostic criteria for OA. The lack of clarity of the diagnostic suitability of pitting means that, for this study, although pitting will be assessed separately, it will not be considered suitable for differential diagnosis of OA unless it is in combination with osteophytes or joint contour change.

Pitting is defined as small perforations on the surface of the bone where there is erosion of the subchondral layer, exposing the trabeculae below (Waldron, 2009). It is scored on a five-point scale from 0-4. The score was applied to the largest pit visible. The extent of pitting over the total facet surface is scored on a 4-point scale (see Table 5-8). The scores used do not represent exact measurements but represent a ranked order of extent and severity of the OA criteria being assessed due to the difficulty in quantitatively measuring changes due to OA (Robb, 1998).

Pitting Severity	Score
Absence of pitting	0
Small pits <0.5 mm in diameter	1
Medium pits 0.5-1.0 mm in diameter	2
Large pits >1.0 and ≤ 1.5mm in diameter	3
Craters ≥ 1.5 mm in diameter	4
Pitting Extent	Score
No pitting	0
<1/3 of surface area affected	1
1/3-2/3 surface area affected	2
1/3-2/3 surface area affected	3

Table 5-8 Pitting scores

5.7.6.3 Osteophytes

The restabilisation process of the third phase in the degeneration cascade (described in Section 4.3.4.3) is the probable cause of marginal osteophyte formation (Rogers and Waldron, 1995). Marginal osteophytes develop around the margins of joints and are frequently seen close to the attachment of the joint capsule (Jurmain, 1999). Their presence can also be age related (Rogers and Waldron, 1995). Although there is lack of agreement between the use of individual diagnostic criteria for OA, it is recommended that the presence, absence and degree of eburnation, pitting and osteophyte development is used in the scoring of OA in osteological specimens (Bourke, 1967; Rogers et al.1987; Jurmain and Kilgore 1995; Weiss and Jurmain, 2007.

Osteophytic growth is defined as being new bone on or around the joint surface (Jurmain, 1999). The severity was scored following the Säger (1969) system with a 5-point scale. In practice the highest score of 4 was not used as no fusion of facets was observed in the samples studied. The extent of the edge of the facet that was affected by osteophytic growth was scored using a 4-point scale (see Table 5-9). The scores used do not represent exact measurements but represent a ranked order of extent and severity of the OA criteria being assessed due to the difficulty in quantitatively measuring changes due to OA (Robb, 1998). Missing data were scored as unobserved and not included into any calculations of frequency

Osteophyte Severity	Score
Absence of osteophytes	0
One or two small osteophytes around facet	1
Multiple small osteophytes	2
Multiple large osteophytes	3
Fusion of facets	4
Osteophyte extent	Score
No osteophytes	0
<1/3 facet edge affected	1
1/3-2/3 facet edge affected	
	2

Table 5-9 Osteophyte scores

(After Säger (1969)

5.7.6.4 Vertebral body osteophytosis

Vertebral osteophytosis (VOS) results from degeneration of the intervertebral disc (Maat et al., 1995). It manifests as osteophytic growth around the perimeter of the vertebral body. There is argument about the order in which the two degenerative processes occur with Butler et al. (1989) reporting that discs degenerate before facets and Eubanks et al. (2007) reported the opposite, facet OA precedes vertebral osteophytosis. Therefore, the presence of VOS was considered as a potential confounder for this study. Osteophytosis of the vertebral body was scored following Säger (1969).

Table 5-10 Osteophytosis scores (Säger, 1969)

Osteophytosis Severity	Score
Absence of osteophytes	0
Mild arthrosis with up to 50% vertebral margin affected	1
Moderate arthrosis with 50-100% of vertebral margin affected	2
Severe arthrosis with 100% of vertebral margin affected and	3
presence of hypertrophic osteophytes bridging joint space	-
Complete ankylosis	4

5.7.6.5 Prevalence rates

Prevalence rate calculations were performed for each of the scores achieved for eburnation, pitting and osteophytes, as a combined score of their grade and extent, following the method described by Waldron (2007). It was also calculated for vertebral osteophytosis (see section 5.7.6.4) and presence/absence of osteoarthritis (OA).

The true prevalence rate (TPR) was calculated following Roberts and Cox, (2003) and Waldron (2007), using the formula:

TPR = <u>Number of facets affected</u> X 100%

Number of facets

The crude prevalence rate (CPR) was calculated as described by Mays et al. (2004:7) using the formula:

CPR = <u>Number of individuals with condition</u> X100%

Number of individuals sampled

It is useful to record true and crude prevalence rates in order to facilitate comparison with other studies (Waldron, 2007). However, Mays et al. (2004:7) recommend that, when comparing results

within and between samples, it is appropriate to use the crude prevalence rate value as "observations on several bones or teeth from a given individual cannot be considered independent for statistical purposes".

Crude prevalence rates can under-represent or over-represent the true prevalence (Roberts and Cox, 2003:41), because this method assumes that all bones of all skeletons are equally preserved. This is unlikely to be the case in archaeological samples. True prevalence rates identify the number of facets available for observation and provide a clearer picture of prevalence. This study calculated the true and crude prevalence rates for each age group in turn with the *a priori* assumption that the prevalence of OA indicators increased with age to minimise merging of data between age groups (Waldron, 2007). Crude prevalence rates were compared within and between samples by sex. Prevalence rates were compared for inter-sample differences using a Kruskal-Wallace test with Mann-Whitney post hoc to differentiate between pairs of samples.

5.7.7 Extrinsic factors

Lovejoy et al. (1982) have criticised the use of linear measurements and indices alone when using bone dimensions to estimate sex arguing that there should be a greater emphasis on the use of functional correlates and consideration of the relevance of findings to biological adaptation. While sex differences in vertebral shape and size have been noted, there has been no attempt to provide a functional interpretation for these differences. This is addressed in the current study whereby the relationship between facet size and angle are correlated with markers of occupational stress (femoral robusticity, humeral directional asymmetry, eburnation, pitting, osteophyte formation around the perimeter of the facet joint, osteophytosis (osteophyte formation around the perimeter of the vertebral body) and the presence/absence of OA) was assessed. This also enabled comparison of degree of physical activity between samples. The statistical modelling used to control these factors is discussed in section 5.7.8

5.7.7.1 Femoral robusticity

The term robusticity refers to the study and quantification of variation in skeletal size and shape (Stock and Shaw, 2007). Bone strengthens by the addition of new tissue in response to loading placed upon it (Ruff et al., 2006) (See section 4.2.3). The relationship between mechanical loading and functional adaptation of bone has been used to identify changes in behaviour in populations, with an increased robusticity considered to develop as a consequence of strenuous physical activity (Bridges et al., 2000; Steckel and Rose, 2002; Stock and Pfeiffer, 2004) (See section 4.3.2). This study examined the facet joints for changes that may occur as a response to mechanical loading. Robusticity measures of long bones may help to provide a more complete picture of the

lifestyle of an individual in respect of their activity levels and mechanical loading undertaken by the body. The use of long bone external dimensions standardized to bone length (Bräuer, 1988; Wescott, 2002; Ruff et al., 2003) and cross-sectional geometry (Lovejoy et al., 1976; Trinkaus et al., 1994, Larsen, 1997; Stock, 2006) are the most commonly used methods to estimate robusticity.

Cross-sectional geometry of the bone diaphysis by application of beam theory relies on accurate determination of periosteal and endosteal contours of the bone (Lovejoy et al., 1976). This method was not appropriate for this thesis as it is a destructive method that involves sectioning of the bone under study. Non-invasive methods of contour determination involve the use of computed tomography (Jungers and Minns, 1979) or silicone moulds used in conjunction with biplanar radiographs (Stock, 2002). Neither of these two methods was suitable due to the cost and difficulty of imaging samples from a number of locations, thus the use of external long bone dimensions was more appropriate for this study. Stock and Shaw, (2007) reviewed the external methods used to estimate robusticity and identified that, if direct sectioning and CT scanning was not available then the measurement of total sub-periosteal area provided the best externally based method of evaluating robusticity.

To estimate bone strength of the femur, total sub-periosteal area, (TA) was calculated from the antero-posterior and medio-lateral diameters of the femoral midshaft. The mid-shaft was chosen for measurement as this represents the true size of the diaphysis. These measurements can be standardized to body size in order to interpret the measure of robusticity because body size is a primary influence on skeletal mass (Stock and Shaw, 2007). Bräuer (1988) standardized the bone external dimensions against bone length. Trinkaus et al. (1994) proposed standardization of second moments of area by dividing them by the bone length raised by the power of 2.33. (Second moments of area are a measure of the resistance to bending of a loaded section (Lieberman et al., 2004)). Ruff (2008) identified that using bone length raised to the power of 5.33 standardizes second moments to body size and bone length. Bone length is strongly correlated with stature and less strongly correlated with body mass; however body mass has a more significant influence on mechanical loading and long bone dimensions (Ruff et al., 1993). Femoral head diameter can also be used to estimate body mass (Ruff, 2002), Wescott (2006) standardized the external dimensions of the femoral diaphysis to the femoral head diameter.

The anteroposterior (AP) midshaft diameter of the femur was measured at the midpoint of the diaphysis in a sagittal orientation at the highest point of the linea aspera, perpendicular to the bone surface The mediolateral (ML) midshaft diameter was measured in a transverse orientation at the midpoint of the diaphysis, perpendicular to the antero-posterior diameter. Both

measurements were made using a Mitutoyo digital caliper (calibrated to 0.01mm, measured to 0.1mm) and recorded in millimetres. Both femoral diaphyses were measured if present and TA calculated for both if present in order to assess for laterality of mechanical loading.

The standardized total sub-periosteal area (TA) was calculated using the equation:

 $TA = \frac{\left[\pi\left(\frac{AP}{2}\right)\left(\frac{ML}{2}\right)\right]}{Max \text{ length}^3}$ (Ruff et al., 2002; Auerbach and Ruff, 2004a).

5.7.7.2 Directional asymmetry

When examining an East London population from Spitalfields, Waldron (1991) noted that there was a change in laterality of distribution of OA in the facet joint in different regions of the spine, with OA being more common on the left in the cervical region, in the right in the thoracic region and with mostly bilateral distribution in the lumbar region. He theorized that the bilaterality observed in the lumbar spine was due to mobility or forces acting through the spine, but that movement was not the complete answer to this patterning. Merbs (1983), Bridges (1994) and Sofaer Deverenski (2000) also reported similar pattern of distribution. There are a number of suggested causes for this pattern, both anatomical and functional.

Nathan (1962) considered that the position of the aorta on the left of the upper thoracic spine caused inhibition of osteophyte formation on the vertebral body, but the extrapolation of this argument for the formation of degenerative change on the zygapophyseal joints is not clear. Research has shown that although the spine is generally considered to be straight in the coronal plane a lateral curvature is frequently observed (Tallroth et al., 2009). This slight thoracic scoliotic curvature usually occurs in the mid to lower thoracic region, convex to the right (Taylor, 1983; Taylor, 1986; De Smet, 1985; Goldberg et al., 1990). It is considered to be the normally occurring alignment and not related to handedness (Williams, 2000).

Merbs (1983) and Bridges (1994) suggest that asymmetries of OA distribution and severity in the facet joints may be due to the influence of handedness in individuals due to the origin points on the spine of muscles that insert into the upper arm (trapezius and superior and inferior rhomboid muscles). The predominance of right-handedness within the general population (90% across all groups studied (Marchant et al., 1995)) may support this theory, as this would affect the upper thoracic region which is the area of greatest asymmetry (Bridges, 1994), There is a lack of clinical evidence to support this theory. Bridges (1994) tested this hypothesis by comparing the results of OA distribution analysis in North American prehistoric groups and concluded that the asymmetrical patterns of distribution of OA were caused by a wide variety of different activities that involved use of the right arm, rather than a specific activity. In contrast, Wilczak and Kennedy

(1998) consider that side differences in OA distribution represent the effects of specific physical activities rather than age-related changes.

In order to examine the potential confounding effect of hand preference on asymmetry patterns of facet morphology and OA distribution found in the spine, it was necessary to try and identify the dominant hand in each of the skeletons analysed for this research. Identification of hand preference is traditionally based upon the assumption that the dominant hand is used most in habitual activity and this leads to a directional bilateral asymmetry in the upper limbs through mechanical driven bone growth and remodelling as an adaptive response (Steele, 2000; Shaw and Stock, 2009a; Shaw and Stock, 2009b; Shaw, 2011). This assumption has been supported by invivo studies of self-reported handedness and measurement of skeletal asymmetries (Lazenby, 2002; Medland et al., 2004; Shaw, 2011). Several studies have highlighted the fact that there are variable degrees of asymmetry in different skeletal elements from the same individual (Stock and Pfeiffer, 2004; Auerbach and Ruff, 2006; Auerbach and Raxter, 2008). The humerus has been identified as a suitable element in studies of bilateral asymmetry (Stirland, 1993; Steele and Mays, 1995; Steele, 2000; Blackburn and Knüsel, 2006). The lateral asymmetries in length of the humerus and in the humerus + radius have been used to deduce handedness (Steele and Mays, 1995; Lazenby, 2002; Shaw, 2011). Auerbach and Ruff (2006) identified that in the humerus, the average diaphyseal breadths demonstrate the greatest absolute and directional asymmetry. For this reason, diaphyseal breadth was used in this study to assess handedness in each individual. The anteroposterior (AP) midshaft diameter of the humerus was measured at the midpoint of the diaphysis in a sagittal orientation at the maximum diameter; perpendicular to the bone surface. The medio-lateral (ML) midshaft diameter was measured in a transverse orientation at the midpoint of the diaphysis, perpendicular to the antero-posterior diameter at the minimum diameter. Both measurements were made using a Mitutoyo digital caliper (calibrated to 0.01mm, measured to 0.1 mm) and recorded in millimetres.

Directional asymmetry (DA) is a measure of the effects of lateralized behaviour in the human skeleton (Stirland, 1993; Mays, 2002). Differences in the magnitude of DA can be attributed to differential mechanical loading during bone growth (Auerbach and Ruff, 2006; Auerbach and Raxter, 2008; Özener, 2010). Directional asymmetry for the facet joints was calculated based on the facet size as described in section 5.7.4.1 and the asymmetry equation calculated using right and left facet sizes.

Percentage directional asymmetry (%DA) (the difference between the left and right sides) was calculated using the formula: $\% DA = \frac{(Right-Left)/(Right+Left)}{2}$ X100.

(Mays et al., 1999; Auerbach and Raxter, 2008)

112

Positive values indicate a right biased asymmetry and negative values a left biased asymmetry. This provided information about the predominant directional bias and a measure of the magnitude of asymmetry for the humerus and facets.

5.7.7.3 Sexual dimorphism index (SDI)

A sexual dimorphic index (SDI) provides definition of the degree of difference between the sexes and calculates the magnitude of sexual dimorphism between males and females. There are three main assumptions when calculating an SDI (Reno et al., 2003):

- Both sexes are present within the sample (documentary evidence to support this for St Bride's Crypt and morphological examination as described in section 5.7.2 to estimate the sex of individuals from St Brides' Lower and the Anglo-Saxon sample)
- 2) Any individual has an equal prior probability of being male or female
- 3) When 2 individuals are of different sex, the larger is male.

Smith (1999) summarised a number of methods to calculate this index, including ratios and regression formulae. Regression analysis has not been shown to be any more accurate than ratio calculations (Smith, 1999) and therefore a sexual dimorphism index (SDI) for facet size and angle was calculated using the formula:

(Male mean measurement/female mean measurement) x 100 (Smith, 1999)

Results greater than 100% indicate that the male measurement is greater than that for the female, however this does not clearly differentiate between ambiguous individuals. In this study, SDI values were allocated to categories relating to sexual identity as described in Buikstra and Ubelaker (1994) as follows: female (<90), probably female (90-99) ambiguous sex (100), probably male (101-110) male >110). This allowed for strongly sexually dimorphic individuals to be clearly identified.

5.7.8 Data management and statistical analysis

All data was recorded on Excel Spread sheets and exported into SPSS (v 24, IBM, Chicago).

5.7.8.1 Missing data

Missing data due to poor skeletal preservation and outlying data due to the wide range of natural variation limited the choices of statistical analysis for this study. Whilst there are statistical tests that would interrogate and analyse the data in a more efficient manner, the data from this study was not robust enough to undergo these types of analysis.

5.7.8.2 Descriptive statistics

Descriptive statistics using SPSS v 24 (mean, maximum, minimum and standard deviation) of the size of the facet surfaces and the sagittal angle for each vertebra were calculated, to identify any outliers. The normality of frequency distribution for the continuous values was found. The descriptive statistics results are summarised in Appendix B and C. The variables used in statistical analysis are listed in Table 5-11.

5.7.8.3 Bonferroni adjustment

Abelson (1995) postulated that if enough measurements were taken, a number of significant relationships would be found that could be attributed to chance or random processes. Therefore, a Bonferroni correction was applied to most of the sets of analyses to reduce the risk of Type I errors. The Bonferroni correction was calculated by dividing the alpha value (typically 0.05) by the number of statistical tests and the value of the correction applied is identified with each individual set of results. However, a problem associated with using Bonferroni adjustment is that it dramatically lowers the chance of detecting a real difference if one exists (Kim and Bang, 2016). Greenland et al. (2016) recommend designating a primary hypothesis and answering this with adjusted p values and to present adjusted and unadjusted p-values together as secondary information. Pergener (1998) does not recommend using the Bonferroni adjustment because it can be a bad fit to the real significance (expectation) increasing the risk of Type II errors and therefore to balance this, results with Bonferroni adjustment (p<0.0005) and p<0.05 are included in results tables. Significant p values for p<0.05 and Bonferroni adjusted values are highlighted in bold in the appendices.

5.7.8.4 Statistical analysis of the relationship between sex, facet size and angle

Although the data were normally distributed, a number of outlier values were identified and therefore a non-parametric test (Mann-Whitney U) was used with a Bonferroni adjustment (α = 0.0005) to test for differences between the two independent groups (male and female) on a continuous measurement (facet size/angle).

5.7.8.5 Correlation analysis

A correlation is simply defined as the relationship between two or more variables and correlation analysis is used to identify if there is a connection between variables (Field, 2013). For correlation studies in this research, the strength of the association as indicated by the correlation coefficient in terms of: no relationship ($r_s < 0.01$), weak relationship ($r_s = 0.1-0.29$), medium ($r_s = 0.3-0.49$) and strong ($r_s = 0.5-0.1$) was recorded after the method described by Cohen (1998: 79-81). Pallant

(2013) recommends focussing on the strength of the relationship as the significance (p value) can be influenced by the size of the sample and the test of strength of a relationship is independent of significance testing. For this study, both significance value and correlation strength are reported.

5.7.8.6 Age groups v actual age

Facet size and actual age are continuous data, whereas age group is categorical data. In order to check for loss of resolution the relationship between aggregated data (age group) and facet size is compared with the correlational relationship between disaggregated data (actual age at death) and facet size.

5.7.9 Assessment of the roles of predictor variables in assessing sex

This section describes the methods used to test the degree of accuracy with which the sex of an individual can be estimated from facet size and angle using discriminant function analysis. Discriminant function is frequently used to derive equations for sex estimation (Schwartz, 2008). A summary table of predictor variables that summarise statistically significant analyses of all the data in relation to sex differences and facet size and angle was created to clarify results.

5.7.9.1 Discriminant function analysis

Discriminant function analysis was used to classify vertebrae into dichotomous groups based on the male/female variable based on a linear combination of interval predictor variables. In the first instance the variables will be facet size for all four facets, secondly facet angle for all four facets and finally a combination of predictor variables. The advantage of using discriminant function analysis in skeletal measurements is that it reduces the level of subjectivity that can occur when morphology is used to determine sex and is reported to be an effective tool on sex identification (Isçan, 1988).

SPSS was used to select and weight variables to create discriminant functions for each vertebral level using a stepwise process. Variable preservation of the skeletons sampled led to a number of missing values and for the purpose of analysis; the missing values were replaced with mean values calculated by SPSS. Discriminant analysis is highly sensitive to outliers and therefore these were removed from the data and treated as missing values. Cross-validated classification was used to assess the accuracy of the discriminant equations for each vertebra.

The discriminant function used was:

F(n) = C + U1X1 + U2X2 + U3X3 + U4X4 (Field, 2013).

Where F is the discriminant function total, (n) is the discriminant function equation number (1,2 or 3), U1, U2, U3 and U4 are the unstandardized coefficients of X1 (left superior facet), X2 (right superior facet), X3 (left inferior facet) and X4 (right superior facet).

Application of the discriminant function to the mean measurements for each facet for males and females created the group centroids (Zm and Zf), representing males (Zm) and females (Zf). The demarcation point (Z₀) was calculated by taking the weighted means of values at the group centroids for males and females using the following formula (Xavier, 2003):

$$Z_0 = \frac{(Zm*Nf) + (Zf*Nm)}{(Nm+Nf)}$$

The discriminant scores were calculated separately for each individual of known sex from St Bride's crypt by calculating the discriminant function for each of the vertebral levels. The score was compared against the demarcation point. Scores above the demarcation point were deemed to be male and those below were female. The outcome was compared against the known sex of the individual and the percentage accuracy of correct grouping was noted. This was repeated for the St Bride's Lower and Anglo-Saxon populations. Population specificity of the discriminant functions was tested by substituting the data from for St Bride's Lower and Anglo-Saxon into the functions for St Bride's Crypt and comparing the percentage accuracy of correct grouping.

Variable and Section	How Used in Analysis	Type of Data /
Reference		logistic Code
Sex	Categorised into 2 groups	Categorical
5.7.2.4		0=Male
		1= Female
Age	Categorised into 3 age groups	Categorical
5.7.3.2		1 = 21-34
		2 = 35-45
		3 = 46+
Facet Size	Area = π (width x height) ÷ 4	Continuous
5.7.4.1		
Facet Angle	Superior angles for left and right	Continuous
5.7.4.2	Inferior angles for left and right	
Eburnation	Score for extent and severity added together	Continuous
5.7.6.1		
Pitting	Score for extent and severity added together	Continuous
5.7.6.2		
Osteophytes	Score for extent and severity added together	Continuous
5.7.6.3		
Prevalence rate	Calculated as:	Continuous
5.7.6.5	Total prevalence rate =	
	Number of facets affected Number of facets X100%	
	Crude prevalence rate =	
	Number of individuals with condition Number of individuals sampled X100%	
	Calculated separately for combined severity and extent scores for	
	eburnation, pitting and osteophytes, osteophytosis and	
	presence/absence OA	

Table 5-11 Definition of variables

Table 5-11	Definition	of variables	continued
------------	------------	--------------	-----------

Variable and Section Reference	How Used in Analysis	Type of Data / logistic Code
Femoral robusticity 5.7.7.1	Sub periosteal area = $[\pi(AP/2)(ML/2)]/max length^3$	Continuous
Vertebral osteophytosis 5.7.6.4	Score for superior and inferior margins added together	Continuous ordinal
Vertebral asymmetry 5.7.7.2	Difference between right and left sides for facet angle and size Directional asymmetry %DA = (Right-Left)/(Right + Left)/2 X100	Continuous
Humeral directional asymmetry 5.7.7.2	Humeral directional asymmetry %DA = (Right-Left)/(Right + Left)/2 X100	Continuous
Sexual Dimorphism Index 5.7.7.3	(Male mean measurement/female mean measurement) x 100	Continuous
Population sample 5.1.3	Cemetery	Categorical 1 = St Bride's Crypt 2 = St Bride's Lower 3 = Anglo-Saxon
Time Period	Categorised into groups	1 = Anglo-Saxon 2 = St Bride's
5.8 Pilot Study

A pilot study of the methods used in this study was carried out on individuals from a Great Chesterford (N = 20); an Anglo-Saxon cemetery in Essex. The analysis of the Great Chesterford sample took place at the University of Southampton. The use of standard methods across all populations under study ensured that there was consistent recording of the data required to test the hypothesis for this research. The methods used in this study were assessed for intra-observer reliability. The researcher took all repeat measurements. The results are discussed in section 5.8.1.

Limited preservation of the bones led to a relatively small sample being available for the pilot study of analysis of individuals from Great Chesterford (N=20). It was not possible to randomly sample from this cemetery due to the varied preservation of the bones. Limited access to samples other than Great Chesterford meant that this sample only was available to test for intra-rater reliability. Initial data collecting by the author occurred in 2012 and was repeated without reference to the original data in 2014 and 2015.

Age Group	Males (% total number)	Females (% total number)
21-34	5 (56)	3 (27)
35-45	4 (44)	7 (64)
46+	0	1 (9)

Table 5-12 Great Chesterford demographic

5.8.1 Analysis of intra-observer reliability

This section discusses the analysis of intra -observer reliability. Observer error analysis is an important part of any scientific study as it confirms the reliability of the methods used for data collection. For a method to be considered reliable, the data collected must be replicable. The reliability of this study was tested using intra-observer error studies by repeated analysis of the pilot study material (Great Chesterford).

5.8.1.1 Intra-observer error

Intra-observer error analyses were performed using Fleiss' kappa statistic for categorical data and One-Way Repeated Measures of Analysis of Variance (ANOVA) for continuous data.

Fleiss's kappa statistic is a well-known index for assessing the reliability of agreement between raters (Falotico and Quatto, 2014). This method is an adaptation of Cohen's Kappa and is applicable to three or more observations (Sim and Wright, 2005). The goal of the assessment of

Chapter 5 Materials and Methods

intra- and inter-observer error is to assess the rate of agreement and thus this statistical test is appropriate. The following criteria were used to measure levels of agreement between observations: $\kappa = +1.00$ (perfect agreement) and $\kappa = -1.00$ (no agreement), as κ approaches 0 it is more likely that the agreement is a chance event (Pallant, 2013). Peat (2001) considers that a $\kappa =$ 0.5 to be representative of a moderate agreement, $\kappa < 0.7$ a good agreement and $\kappa < 0.8$ a very good agreement. Landis and Koch (1977) further classify the strength of agreement using the following descriptors: poor (<0.00), slight (0.00-0.20), fair (0.21-0.4), moderate (0.41-0.60), substantial (0.61-0.80) and almost perfect (0.81-1.00). The classification system of Landis and Koch (1977) will be used in this study as it has more specific definition between the levels of agreement than the one used by Peat (2001). The results of the intra-observer error for categorical data are reported in Appendix A and summarised in Table 5—13.

The majority of repeated categorical scores demonstrated high levels of agreement. The highest agreement was observed in vertebral levels where the feature was mostly absent. The lowest level of agreement with 3 observations overall was moderate. The reproducibility of the results indicate that the method chosen for this study was appropriate.

Criterion	Criterion Section of Spine (Lan (Lan		Count of All Facets at Level of	Percentage of Facets at Same Level of
		Koch, 1977)	Agreement	Agreement per
			10	area
		Perfect	40	/1%
	Cervical	Almost Perfect	16	29%
		Substantial	0	0%
		Moderate	0	0%
		Perfect	43	45%
Eburnation	Thoracic	Almost Perfect	48	50%
		Substantial	4	4%
		Moderate	1	1%
		Perfect	7	17%
	Lumbar	Almost Perfect	33	83%
	Lambar	Substantial	0	0%
		Moderate	0	0%
		Perfect	6	11%
Pitting	Cervical	Almost Perfect	47	83%
		Substantial	2	4%
		Moderate	1	2%
	Thoracic	Perfect	14	15%
		Almost Perfect	81	84%
		Substantial	1	1%
		Moderate	0	0%
		Perfect	5	13%
	Lunchen	Almost Perfect	35	87%
	Lumbar	Substantial	0	0%
		Moderate	0	0%
		Perfect	39	71%
	Comical	Almost Perfect	16	29%
	Cervical	Substantial	0	0%
		Moderate	0	0%
		Perfect	43	45%
		Almost Perfect	48	50%
Osteophytes	Thoracic	Substantial	4	4%
		Moderate	1	1%
		Perfect	7	17%
		Almost Perfect	33	83%
	Lumbar	Substantial	0	0%
		Moderate	0	0%

Table 5-13 Fleiss' kappa level of agreement for intra-observer error

5.8.1.2 Continuous data

One-way repeated measures analysis of variance (ANOVA) is a statistical method used to test for significant differences in multiple repeated measures between groups (Pallant, 2013). This is an appropriate statistical method for comparison of the repeated measurements of facet size and sagittal angle. The Wilks' Lambda value and probability value p are reported. A significant p value of <0.05 for the differences in repeat measurements indicates that differences between the measurements are unlikely to be due to chance and that there was a change in confidence score across the three repeated measures (Field, 2013).

One-way repeated measures ANOVA was performed to compare repeated measurements of the width, height and sagittal angle of the facets as the data set consisted of continuous variables that were of normal distribution. The data was collected in 2009, 2013 and 2016 from the Great Chesterford sample. The overall results tables can be found in Appendix A and are summarised in Table 5-14.

Spinal Region	Criterion	Count of Facets with Significant Difference Between Repeated Measurement (p<0.05)	Percentage of Facets with Significant Difference in Repeated Measure per Region
Cervical	Facet Width	2	7%
	Facet Height	0	0%
	Facet Angle	1	4%
Thoracic	Facet Width	3	6%
	Facet Height	4	8%
	Facet Angle	6	12%
Cervical	Facet Width	1	5%
	Facet Height	0	0%
	Facet Angle	4	2%

	· · ·	• • • •		1.00		
Table 5-14 Summarv	of count of	facets showing	significant	difference	in repeated i	measures

The results show that for the majority of facet levels, there is no significant difference between all three repeated measurements for facet width, height and angle when p<0.05. Overall 10% of the facet width, 3% of the facet height measurement and 10% of the facet angle measurements showed statistically significant differences in the repeated measures. This could be due to lack of operator experience in determining measurement points in the first data set collection. The overall results indicate that the methods used are appropriate and reproducible.

This chapter presents the results of data analysis to answer the key questions identified in Chapter 1. It will begin by reviewing the demographic data of the samples under study and then will present the results of the statistical analyses. Facet size, facet angle and inter-sample analyses of predictor variables will be presented separately for clarity. The main tables of results can be found in the attached appendices (Appendix B-E) with summaries of the significant or important results being presented in this chapter. The reported results are arbitrarily divided into cervical, thoracic and lumbar regions, based on characteristics and function. Transitional vertebrae between spinal regions (see Section 2.3.7) were excluded from the analyses.

6.1 Sample demography

The total number of individuals used in this study is summarised in Table 6-1. These numbers are not representative of the cemetery population as a whole but are a sample of the individuals that are interred. Only skeletons with sufficient preservation to allow for adequate data analysis were included in this study.

The St Bride's Crypt context is described in detail in section 5.4.1, St Bride's Lower cemetery in section 5.4.2 and the Anglo-Saxon cemetery contexts in sections 5.1, 5.2 and 5.3. Due to the small number of individuals from the Anglo-Saxon cemeteries, the data from the three cemeteries were amalgamated.

As the relationship between age, facet size and angle was explored in this study, there was a risk that imbalances in proportion of age and sex categories would skew the results of comparisons between groups. To check for this, the individuals were placed into approximate age groups (Young Adult, 21-34 years; Middle Adult, 35-45 years and Old Adult, 46+) based upon their estimated age at death (see 5.7.3). The number of males and females in each age group are summarised in Table 6-1 overleaf. Fisher's exact test was used to test for imbalances in age group membership and revealed that there was no significant difference in the number of males and females in each age group ($\chi^2_{(2, n=180)}$ =0.487, p=0.784, phi = 0.052).

Cemetery	Sex	Number of Individuals	Young Adult	Middle Adult	Old Adult
Great	Male	8	3	5	0
Chesterford	Female	12	6	6	0
Buckland	Male	4	3	0	1
	Female	4	3	1	0
St Anne's	Male	17	11	6	0
	Female	8	1	5	2
St Bride's Crypt	Male	26	6	6	14
	Female	41	13	11	17
St Bride's Lower	Male	35	3	17	15
	Female	24	7	6	11

Table 6-1 Summary of cemetery origin and age distribution of skeletons used in this study

6.2 Inter-sample comparison of predictor variables

Prior to presenting the results of analysis of facet size and angle by sample, the results of analysis of inter-sample variation between the predictor variables and prevalence of eburnation, pitting, osteophytes and vertebral osteophytosis are reported.

6.2.1 Inter-sample comparison of femoral robusticity

Femoral robusticity data for males and females were tested for statistically significant differences between the samples.

Kruskal-Wallace tests did not demonstrate any statistically significant difference between left or right femoral robusticity across three samples.

Left femoral robusticity: males – (SBC, n=13: SBL, n=26: AS, n=19). X² (2, n=58) =0.83, p=0.66

Left femoral robusticity: females – (SBC, n=23: SBL, n=15: AS, n=23). X² (2, n=49) =0.08, p=0.96

Right femoral robusticity: males - (SBC, n=12: SBL, n=26: AS, n=23). X² (2, n=61) =0.14, p=0.99

Right femoral robusticity: females – (SBC, n=19: SBL, n=13: AS, n=14). X² (2, n=46) =0.81, p=0.67).

These results demonstrate that there is no statistically significant inter-sample difference in femoral robusticity

6.2.2 Inter-sample comparison of humeral directional asymmetry

Kruskal-Wallis tests did not demonstrate any statistically significant difference between humeral directional asymmetry across the three samples.

Humeral asymmetry: males - (SBC, n=18: SBL, n=27: AS, n=12). X² (2, n=57) =1.354, p=0.51

Humeral asymmetry: females – (SBC, n=31: SBL, n=21: AS, n=5). X² (2, n=57) =0.835, p=0.66).

These results demonstrate that there is no statistically significant inter-sample difference in humeral directional asymmetry between the three samples.

6.2.3 Inter-sample comparison of prevalence rates

6.2.3.1 Crude prevalence rates

Crude prevalence rates were calculated as described in section 5.7.6.5. The results are presented in Table 6-2 and illustrated in Figure 6-1. The crude prevalence rate is a ratio between the number of individuals with a condition and the number of individuals in a sample and presented as a percentage. It measures the number of individuals in a study with the characteristic of interest. This method is limited in scope as it divides the samples into afflicted/not-afflicted categories. Whilst it is useful to be able to visualise the number of individuals in each sample with the particular characteristic it does not differentiate the degree with which it occurs.

Table 6-2 Percentage crude prevalence rates for eburnation, pitting, osteophytes andosteophytosis

Sample	Sex	Crude	Crude	Crude	Crude
		prevalence	prevalence	prevalence	prevalence
		rate	rate pitting	rate	rate
		eburnation		osteophytes	osteophytosis
SBC	Female	88	90	100	39
	Male	96	96	100	54
SBL	Female	63	100	100	54
	Male	71	100	100	60
AS	Female	63	88	63	63
	Male	52	100	97	31



Figure 6-1 Inter-sample crude prevalence rates

6.2.3.2 True prevalence rates

True prevalence rates identify the number of facets available for observation and provide a clearer picture of the characteristic being investigated. Appendix D-1 presents the true prevalence rates as a percentage for eburnation, pitting, osteophytes and vertebral osteophytosis by individual. These values are represented graphically in Figure 6-2,6-3,6-4 and 6-5. The values represent a count of the facets at each vertebral level with the characteristic as a percentage of the facets present. This allows for the management of missing data due to poor preservation.



Figure 6-2 Inter-sample variation in the percentage of facets with eburnation



Figure 6-3 Inter-sample variation in the percentage of facets with pitting





Figure 6-4 Inter-sample variation in percentage of facets with osteophytes







Figure 6-5 Inter-sample variation in the percentage of vertebral bodies with osteophytosis

Kruskal-Wallis tests with Bonferroni adjustment (α =0.0002) show statistically significant

differences in percentage true prevalence for pitting on both males and females across the three samples. Table 6-3 summarises the results of the Kruskal Wallis test. The true prevalence rates for pitting show statistically significant inter-sample differences (p value highlighted). However, it is not possible to differentiate between the samples and therefore a series of Mann-Whitney U tests between pairs of samples for pitting score of all vertebrae were performed to make the differentiation. The results of this analysis can be seen in Table 6-4. After Bonferroni adjustment, (α =0.0002) statistically significant differences in true prevalence of pitting can be seen between the Anglo-Saxon sample and St Bride's Crypt and St Bride's Lower (p value highlighted). Table 6-5 presents the combined median scores for true prevalence of pitting for each sample at all vertebral levels and clearly demonstrates that the true prevalence of pitting is much higher for males and females from the Anglo-Saxon sample.

Table 6-3 Summary of Kruskal-Wallace analysis of inter-sample differences in true prevalence of

Variable	Sample	Sex	n	df	Ν	X ²	р
	SBC		26				
	SBC	Male	35	2	94	6.786	0.034
Eburnation	AS		33				
Loumation	SBC		41				
	SBC	Female	35	2	86	8.576	0.014
	AS		33				
	SBC		26				
	SBC	Male	35	2	94	19.784	0.0001
Ditting	AS		33				
Titting	SBC		44				
	SBC	Female	24	2	86	37.859	0.0001
	AS		21				
	SBC		26				
	SBC	Male	35	2	94	2.566	0.277
Osteonhytes	AS		33				
Osteophytes	SBC		41				
	SBC	Female	24	2	86	10.32	0.006
	AS		21				
	SBC		26				
	SBC	Male	35	2	94	5.795	0.005
VOS	AS		33				
VO3	SBC		44				
	SBC	Female	21	2	86	2.113	0.348
	AS		24				

eburnation, pitting, osteophytes and osteophytosis

Table 6-4 Summary of Mann-Whitney U test for differences in true prevalence between sample

pairs

Predictor Variable True Prevalence	Sex	Sample Comparison	Z	r (Effect size)	N	р
		SBC/SBL	-2.911	-0.373	61	0.004
	Male	SBC/AS	-1.549	-0.202	59	0.121
Eburnation		SBL/AS	-0.095	-0.012	68	0.924
Loumation		SBC/SBL	-1.694	-0.21	65	0.09
	Female	SBC/AS	-1.991	-0.253	62	0.046
		SBL/AS	-2.643	-0.394	45	0.008
		SBC/SBL	-1.327	-0.17	61	0.184
	Male	SBC/AS	-3.855	-0.502	59	0.0001
Ditting		SBL/AS	-3.595	-0.436	68	0.0001
Fitting	Female	SBC/SBL	-1.543	-0.191	65	0.123
		SBC/AS	-5.697	-0.724	62	0.0001
		SBL/AS	-4.937	-0.736	45	0.0001
Ostooshutoo		SBC/SBL	-1.838	-0.235	61	0.066
	Male	SBC/AS	-0.382	-0.05	59	0.703
		SBL/AS	-0.743	-0.09	68	0.458
Osteophytes		SBC/SBL	-0.2603	-0.032	65	0.009
	Female	SBC/AS	-2.67	-0.339	62	0.008
		SBL/AS	-0.182	-0.027	45	0.856
		SBC/SBL	-0.387	-0.05	61	0.699
	Male	SBC/AS	-1.742	-0.227	59	0.081
Vertebral		SBL/AS	-2.351	-0.285	68	0.19
Osteophytosis		SBC/SBL	-1.425	-0.177	65	0.154
	Female	SBC/AS	-1.12	-0.142	62	0.901
		SBL/AS	-0.975	-0.145	45	0.33

Sample	Male True Prevalence Pitting Median	Female True Prevalence Pitting Median
SBC	15.61	17.7
SBL	18.39	23.98
AS	56.84	78.43

Table 6-5 Median scores for true prevalence of male and female pitting scores

6.2.4 Prevalence rates of osteoarthritis.

Osteoarthritis (OA) is diagnosed as described in 5.2.4. Eburnation alone or pitting and osteophytes co-occurring are considered diagnostic of OA. To further clarify inter-sample differences in eburnation, pitting and osteophytes, the crude and true prevalence rates of OA were calculated for each sample (5.7.6.5).

6.2.5 Inter-sample comparison of OA prevalence rates.

Crude prevalence rates were calculated as described in section 5.7.6.5. The results are presented in Table 6-6. In males, the highest crude prevalence rate for OA (88%) was seen in the St Bride's Crypt sample, closely followed by the Anglo-Saxon Males at 83%. In females, the Anglo-Saxon sample have the highest crude prevalence rate for OA at 100%, i.e. all females in this group showed OA on at least one facet level in their spines.

Table 0-0 Clude prevalence rates for Osteoartinitis

Sample	Sex	Crude prevalence rate OA
SBC	Female	56%
	Male	88%
SBL	Female	58%
	Male	66%
٨٢	Female	100%
AS	Male	83%

True prevalence rates (see section 5.7.6.5) identify the number of facets available for observation and provide a clearer picture of prevalence. Appendix D-2 presents the true prevalence rates as a percentage for OA by individual. Kruskal-Wallis test with Bonferroni adjustment (p=0.0002) (Table 6-7) showed a statistically significant difference in percentage true prevalence for OA for females across the three samples (significant values highlighted).



Figure 6-6 Inter-sample variation in true prevalence of OA

Variable	Sample	Sex	n	df	Ν	X ²	р
	SBC		26				
	SBC	Male	35	2	93	13.976	0.001
04	AS		32				
UA	SBC		39		85		<0.0001
	SBC	Female	24	2		24.735	
	AS		22				

Table 6-7 Kruskal-Wallace test for inter-sample variation in true prevalence rates of OA

The results summarised in Table 6-7 indicate that there is a sample difference in true prevalence rates for OA between the three samples under study but does not identify which of the samples is statistically significantly different from one another, therefore a series of Mann-Whitney U tests between pairs of samples were performed allowing for differentiation. The results of this analysis can be seen in Table 6-8. After Bonferroni adjustment (α =0.0001) statistically significant differences in true prevalence of OA can be seen between the female Anglo-Saxon sample and St Bride's Crypt and St Bride's Lower with strong effect size (using Cohen, 1988 criteria of 0.1 = weak effect, 0.3 = medium effect and 0.5 = strong effect) and also between the male Anglo-Saxon sample and St Bride's Lower. Table 6-9 presents the median scores for true prevalence of OA for each sample and clearly demonstrates that the true prevalence of OA is much higher for males and females from the Anglo-Saxon sample.

Predictor Variable True Prevalence	Sex	Sample Comparison	Z	r (Effect size)	Ν	р
04	Male	SBC/SBL	-1.947	0.25 (Weak)	61	0.052
		SBC/AS	-2.043	0.27(Weak)	58	0.041
		SBL/AS	-3.531	0.43 (Medium)	67	<0.0001
UA		SBC/SBL	-0.103	0.01 (None)	63	0.918
	Female	SBC/AS	-4.60	0.95 (Strong)	61	<0.0001
		SBL/AS	-4.089	0.6 (Strong)	46	<0.0001

|--|

Table 6-9 Median scores for male and female true prevalence OA

Sample	Male True Prevalence OA Median	Female True Prevalence OA
		Median
SBC	5.44	2.41
SBL	1.64	1.91
AS	13.23	20.21

In order to further clarify the above results, Mann-Witney U tests were performed to identify differences in true prevalence of OA between males and females from each sample. Bonferroni adjustment was not applied to this analysis due to the small number of comparisons. Table 6-10 presents the results of inter-sex comparison by sample. Males and females from St Bride's Crypt show a statistically significant difference in true prevalence rate of OA between males and females. This observation was not seen in St Bride's Lower and the Anglo-Saxon individuals.

Table 6-10 Mann-Whitney U test for inter-sex	differences in true prevalence of OA.
--	---------------------------------------

Sample	Ν	Ν	Total N	R (Effect size)	Z	р
SBC	Males	26	65	0.27 (Weak	-2.217	0.027
	Females	39		effect)		
SBL	Males	35	59	0.04 (No effect)	-0.301	0.763
	Females	24				
AS	Males	32	54	0.14 (Weak	-1.021	0.307
	Females	22		effect)		

This section investigated inter-sample differences of the predictor variables used in this study. The most significant variations seen were the true prevalence rate of pitting and OA in the Anglo-Saxon sample.

6.3 Facet size

This section presents the results generated by analysis of sexual dimorphism and the relationship between facet size with age, femoral robusticity, humeral directional asymmetry, facet eburnation, pitting, osteophyte scores and vertebral body osteophytosis. Facet size was calculated by multiplying the height and width of each facet (see Section 5.7.4).

6.3.1 Facet Size metric analysis

Each facet was analysed separately in this analysis (i.e. left and right superior and inferior facets for each vertebral level). The number of facets analysed for each sample is summarised in Table 6-11. The Anglo-Saxon sample was poorly preserved and had fewer intact facets available for study.

Table 6-11 summary of number of facets analysed for each sample

Sample	Number of Facets used for Analysis
St Bride's Crypt	5808
St Bride's Lower	4791
Anglo-Saxon	2997

A number of outlying values were observed and are reported for each spinal region and sample in Table 6-12. These values are due to natural variation in facet size. Many statistical techniques are sensitive to outliers, with correlation studies being particularly affected and although the majority of the data was of normal distribution, non-parametric methods were used to reduce the effect of the outlying data.

Spinal Region	Number of Outliers for Facet Size			
op	St Bride's Crypt	St Bride's Lower	Anglo-Saxon	
Cervical	78 (13%)	42 (5%)	21 (4%)	
Thoracic	116 (10%)	82 (6%)	48 (5%)	
Lumbar	31 (6%)	26 (5%)	9 (2%)	

Table 6-12 Count of data outlier	's by spina	I region and	l sample
----------------------------------	-------------	--------------	----------

Descriptive statistics for facet size for each sample using age combined, sex-divided groups are reported in Appendix B-1. This section is focussed on differences in size between males and females and not the effect of age on facet size (reported in Section 6.3.5) and therefore the age groups were amalgamated.

There is a wide range of standard deviation for facet size means across all facets for both sexes. This represents the wide variation in size that can be seen in the same facets from different individuals. The extremes of standard variation results and the facet levels at which they occurred at are summarised in Table 6-14.

	St Bride	's Crypt	St Bride	t Bride's Lower Anglo-Sax		
	Male	Female	Male	Female	Male	Female
Minimum Standard	12.65	14.58	13.35	12.71	14.11	12.41
Deviation	T7RI	T8LtS	C7RI	T8LtS	C4RI	L4LtS
Maximum Standard	72.59	56.61	70.2	79.35	77.23	66.6
Deviation	C2LtS	L4RS	T9LtS	L5LtS	L3RS	L4LtS

Table 6-13 Maximum and minimum standard deviation values for facet size

Key: C=cervical, T=thoracic, L=lumbar, R=right, Lt=left, S=superior, I=inferior

6.3.2 Facet size sexual dimorphism index

This section presents the results of calculation of the sexual dimorphic index (SDI) for each sample. The indices were generated from the average facet size calculated for each individual facet (i.e. left and right superior and inferior) (See section 5.7.4.1). The SDI demonstrates the presence or absence of differences in size between males and females and is a quick method to check for sexual dimorphism. However, it does not indicate the statistical significance of any

differences observed and therefore further analysis was performed to test for this and reported in this section

Table 6-14 summarises the results of SDI analysis presented in Appendix B-1. The results show the count of facets for each region at "SDI >110" (male), "< 90" (female) or "unclassified" (SDI values between 109 and 91). Male indices were greater than female and unclassified individuals in 85% (82/96) of facets from the St Bride's Crypt sample, and the St Bride's Lower (i.e. male facets were larger than female). This decreases to 47% (45/96) from the Anglo-Saxon sample (indicating that there was less difference in facet size between males and females); the left inferior facet of C2 for this sample had an SDI value of 89.24 indicating that the female index value was greater that the male mean and was the only facet to show this. There was a greater number of facets from the Anglo-Saxon sample that fell into the range between classification levels set for males and females indicating that there was much less sexual dimorphism in this sample. The results of statistical analysis to test if this difference is significant are presented in the next section.

Sample	Spinal Region	SDI>110	SDI<90	Unclassified
	Cervical	22	0	6
St Bride's Crypt	Thoracic	41	0	7
	Lumbar	19	0	1
	Cervical	19	0	9
St Bride's Lower	Thoracic	44	0	4
	Lumbar	19	0	1
	Cervical	12	1	15
Anglo-Saxon	Thoracic	26	0	22
	Lumbar	7	0	13

|--|

Figure 6.7 illustrates the comparison in SDI values between the three samples by facet level. It can be seen that for St Bride's Crypt and the Anglo-Saxon samples the SDI values for facet size were evenly distributed throughout the vertebral column with no region having consistently high scores, however for St Bride's Crypt, the higher values (demonstrating bigger male values) were distributed throughout the lower half of the vertebral column.



Figure 6-7 Sexual dimorphism index scores for facet size

6.3.3 Statistical analysis of the relationship between sex and facet size

The previous section has identified that there is a pattern of sexual dimorphism in facet size. Statistical significance of differences in facet size between males and females are reported in this section. Mann-Witney U test with Bonferroni adjustment (α =0.0005) was used to test for differences between the two independent groups (male and female) on a continuous measurement (facet size).The full results for these analyses can be seen in Appendix B-2. The percentage frequency of facets showing significant results are summarised by region in Table 6-15 and are represented graphically in Figure 6-8.

Table 6-15 Percentage frequency of facets with statistically significant differences in facet size between sexes

Spinal Region	Sample	Percentage significant at p<0.0005	Percentage significant at p<0.05
	SBC	18	46
Cervical	SBL	18	50
	AS	0	1
	SBC	54	83
Thoracic	SBL	5	77
	AS	4	21
	SBC	40	85
Lumbar	SBL	20	95
	AS	0	10



Figure 6-8 Percentage frequency of significant differences in facet size by sample and spinal

region

The Bonferroni correction reduced the significant p value to p<0.0005, but despite this, statistically significant differences in facet size between the sexes were identified.

The null hypothesis tested in this section (See Chapter 1) is that there will be no measurable sexual dimorphism of facet joint size. This null hypothesis can be rejected for all three samples, but it should be noted that the degree of difference on facet size between males and females is less pronounced in the Anglo-Saxon samples studied.

6.3.4 Stepwise discriminant function analysis of facet size.

The previous section has identified that there are measurable differences in facet joint size between males and females and therefore this parameter may be of use in estimating the sex of an individual. The degree of difference between facet sizes is variable and the weighted importance of each value is not known. Discriminant function analysis is a statistical procedure that is frequently used to classify an individual into dichotomous groups based on the male/female variable (see Section 5.8.2.1). SPSS v24 was used to select and weight variables to create discriminant functions for each vertebral level, selecting the variables that best discriminated between the two groups. The mean facet size measurements for each facet were entered into the discriminant function equations and the group centroids for males and females (Zm and Zf) were noted. The demarcation point (Z₀) was calculated from the weighted means of values at the group centroids for males and females (see section 5.8.2.1 for further details). Size measurements above the demarcation point are deemed to indicate a male individual and those below the demarcation point are deemed to indicate a female individual. The number of males and females identified for each vertebral level (Nm and Nf) were also recorded. Discriminant function analysis then compared this value to the sex recorded for that individual and the probability of correct allocation into the appropriate group was also recorded as a percentage (calculated by SPSS).

Stepwise discriminant function analysis was performed as described in 5.8.2.1 for each sample, as discriminant function equations are population specific. The results can be seen in Table 6-16. The p value result is from Wilks' Lambda test for goodness of fit for the model, and tests for significant differences between the male and female groups. In this analysis, the p value was set at 0.0006 (with Bonferroni correction). Results for each sample are included in the table. This analysis generated discriminant functions to identify the sex of an individual based on the size of the facets at each vertebral level. In the event of little or no difference in facet size between the sexes, no variables qualified for inclusion in the equation.

145

The total percentage accuracy of correct classification listed in this table will be used to compare accuracy against the use of predictor variables in deriving discriminant function equations in Table 6-64.

Abbreviation Key:

- C = cervical,
- T = thoracic,
- L = lumbar

Numbers represent the vertebral level

Lt = left,

R = right,

S = superior,

I = inferior

E.g. C1LtSA = facet size for C1, left superior facet and C1LS = facet angle for C1 left superior facet

SBC = St Bride's Crypt,

SBL = St Bride's Lower,

AS = Anglo-Saxon,

Zm = male group centroid,

Zf = female group centroid,

Nm = number of males identified,

Nf= number of females identified,

Z₀= demarcation point.

Table 6-16 Stepwise discriminant function analysis for facet size

Vertebral	Sample	р	F(n)	Zm	Zf	Nm	Nf	ZO	% Accuracy		
Level									М	F	Total
C1	SBC	<0.001	F=(-6.74)+(0.03)C1LtISize	0.62	-0.45	14	25	0.09	61	78	71
	SBL	<0.001	F=(- 8.348)+(0.02)C1LtISize +(0.02)C2RSSize	0.75	-0.94	20	17	-0.11	77	85	80
	AS		No qualifying variables								
C2	SBC	<0.001	F=(-5.73)+(0.03)C2LtSSize	0.73	-0.44	9	30	0.15	39	86	67
	SBL	<0.001	F=(-6.45)+(0.03)C2LtSSize	0.62	-0.9	25	16	-0.14	86	80	84
	AS		No qualifying variables								
C3	SBC	0.042	F=(-3.33)+(0.29)C3LtlSize	0.38	-0.23	6	30	0.07	30	97	71
	SBL	0.024	F=(-3.11)+(0.03)C3LtSSize	0.29	-0.39	22	10	-0.05	79	48	65
	AS	0.062	F=(-5.43)+(0.04)C3RISize	0.58	-0.44	10	5	-0.048	67	64	65
C4	SBC	0.006	F=(-3.59)+(0.03)C4LtSSize	0.53	-0.34	6	25	0.095	32	89	66
	SBL	0.025	F=(-3.33)+(0.03)C4LtSSize	0.28	-0.38	24	9	-0.053	83	41	65
	AS		No qualifying variables								
C5	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS		No qualifying variables								
C6	SBC	0.004	F=(-2.99)+ (0.07)C6LtISize- (0.05)C6RISize	0.63	-0.34	9	32	0.146	45	87	72
	SBL		No qualifying variables								
	AS		No qualifying variables								
C7	SBC	0.003	F=(-3.86)+(0.03)C7RISize	0.5	-0.3	8	35	0.1	32	68	67
	SBL	0.008	F=(-3.96)+(0.04)C7RSSize	0.34	-0.42	24	16	-0.042	75	67	71
	AS		No qualifying variables								

Vertebral	Sample	р	F(n)	Zm	Zf	Nm	Nf	Z0	%	Αссι	uracy
Level									М	F	Total
T1	SBC	0.006	F=(-3.93)+(0.04)T1RISize	0.31	-0.48	29	11	-0.08	83	48	69
	SBL		No qualifying variables								
	AS		No qualifying variables								
T2	SBC	0.002	F=(-4.82)+(0.05)T2RISize	0.5	-0.34	13	33	0.08	50	87	72
	SBL	0.001	F=(-4.09)+(0.04)T2LtSSize	0.38	-0.56	23	13	- 0.094	68	59	64
	AS		No qualifying variables								
Т3	SBC	<0.001	F=(-4.92)+(0.05)T3LtISize	0.64	-0.41	12	31	0.12	48	80	67
	SBL	0.005	F=(-3.27)+(0.03)T3LtSSize	0.36	-0.49	26	13	- 0.068	90	62	78
	AS	0.01	F=(-5.07)+(0.05)T3LtISize	0.41	-0.62	21	6	-0.01	96	50	79
T4	SBC	<0.001	F=(-4.59)+(0.02)T4LtISize +(0.03)T4LSSize	0.67	-0.64	15	32	0.01	58	89	76
	SBL	0.003	F=(-0.35)+(0.04)T4LtSSize	0.46	-0.73	24	15	-0.13	80	79	80
	AS	0.001	F=(-7.96)+(0.08)T4RSsize	0.64	-0.97	17	11	- 0.162	74	26	72
T5	SBC	<0.001	F=(-3.32)+(0.038)T5LtSSize	0.65	-0.47	13	34	0.09	52	92	76
	SBL		No qualifying variables								
	AS	0.023	F=(-3.48)+(0.03)T5RISize	-0.36	-0.49	14	10	-0.42	70	71	71
Т6	SBC	<0.001	F=(-4.45)+(0.05)T6LtISize	0.7	-0.51	16	33	0.1	62	87	77
	SBL	<0.001	F=(-5.03)+(0.06)T6RISize	0.49	-0.63	22	14	-0.07	79	67	74
	AS	0.034	F=(-4.15)+(0.04)T6RSSize	0.42	-0.45	13	8	-0.02	72	62	68

Vertebral	Sample	р	F(n)	Zm	Zf	Nm	Nf	Z0	%	% Accuracy	
Level									М	F	Total
Τ7	SBC	<0.001	F=(-5.61)+(0.04)T7LtSSize	0.74	-0.49	12	33	0.13	46	87	70
			+(0.03)T7RSSize								
	SBL	<0.001	F+(-5.97)+(0.08)T7RISize	0.56	-0.85	28	14	-0.15	90	70	82
	AS		No qualifying variables								
Т8	SBC	0.004	F=(-5.09)+(0.02)T8LtISize	0.54	-0.38	12	29	0.08	48	81	67
			+(0.2)T8RSSize								
	SBL	<0.001	F=(-5.5)+(0.07)T8LtSSize	0.53	-0.97	29	12	-0.22	88	60	77
	AS	0.23	F=(-4.62)+(0.02)T8RSSize	0.41	-0.47	11	8	-0.03	65	53	59
Т9	SBC	<0.001	F=(-4.2)+(0.05)T9TISize	0.64	-0.42	14	35	0.11	56	92	78
	SBL	<0.001	F=(-4.16)+(0.05)T9LtSSize	0.46	-0.67	23	15	-0.1	74	75	75
	AS	0.019	F=(-4.02)+(0.04)T9RISize	0.43	-0.49	11	9	-0.03	69	64	67
T10	SBC	<0.001	F=(-3.22)+(0.03)T10tLISize	0.61	-0.39	13	35	0.11	50	90	74
	SBL	<0.001	F=(-3.12)+(0.03)T10RISize	0.45	-0.65	24	12	-0.1	80	60	72
	AS		No qualifying variables								
T11	SBC	<0.001	F=(-4.09)+(0.02)T11RISize	0.66	-0.423	11	32	0.1185	44	82	67
			+(0.03)T11RSSize								
	SBL	0.001	F=(-3.39)+(0.03)T11LtSSize	0.44	-0.6	21	14	-0.08	75	70	73
	AS	0.054	F=(-3.14)+(0.03)T11LtlSize	0.36	-0.46	12	7	-0.05	67	54	61
T12	SBC	0.004	F=(-3.16)+(0.03)T12LtSSize	0.48	-0.31	7	32	0.085	29	87	64
	SBL	0.006	F=(-2.63)+(0.03)T12LtSSize	0.36	-0.479	21	14	-0.06	75	67	71
	AS		No qualifying variables								

Table 6-16 Stepwise discriminant function analysis for facet size continued

Vertebral	Sample	р	F(n)	Zm	Zf	Nm	Nf	Z0	%	% Accuracy	
Level									М	F	Total
L1	SBC	0.005	F=(-3.09)+(0.02)L1RSSize	0.44	-0.33	12	26	0.59	48	74	63
	SBL	0.005	F=(-3.88)+(0.03)L1RISize	0.37	-0.5	21	12	0.25	75	60	69
	AS		No qualifying variables								
L2	SBC	<0.001	F=(-4.56)+(0.03)L2LtISize	0.63	-0.46	14	30	0.09	54	83	71
	SBL	0.004	F=(-3.17)+(0.02)L2LtSSize	0.35	-0.6	23	12	-0.12	79	60	71
	AS		No qualifying variables								
L3	SBC	0.005	F=(-4.17)+(0.02)L3RSSize	0.43	-0.32	13	27	0.06	50	77	66
	SBL	0.003	F=(-4.71)+(0.03)L3LtISize	0.47	-0.55	22	15	-0.04	82	71	77
	AS		No qualifying variables								
L4	SBC	0.003	F=(-3.94)+(0.02)L4LtISize	0.46	-0.32	11	33	0.07	42	87	69
	SBL	0.004	F=(-04.84)+(0.02)L4LtSSize	0.37	-0.66	23	8	-0.15	92	53	78
	AS		No qualifying variables								
L5	SBC	0.002	F=(-4.17)+(0.02)L5LtSSize	0.49	-0.39	12	31	0.05	48	91	73
	SBL	<0.001	F=(-4.45)+(0.02)L5LtSSize	0.51	-0.73	22	12	-0.11	76	63	71
	AS		No qualifying variables								

Table 6—16 Stepwise discriminant function analysis for facet size continued

6.3.5 Relationship between facet size and age.

This section presents the results of analysis of the relationship between facet size with age. There is conflicting evidence in the literature about the influence of age on vertebral morphology (see Section 4.1).

Samples were divided into age group categories as described in section 5.7.3. Kruskal Wallace one-way between–groups analysis of variance with Bonferroni correction (p<0.0005) was used to identify any significant differences between facet size across the three age groups. (See Table 6-1 for sample size). Males and females were analysed separately to reduce the influence on facet size due to sexual dimorphism and also by sample to examine each group individually. The full results can be seen in Appendix B-3. Table 6-17 summarises the results of the Kruskal Wallace analysis of variation of facet size with age group. With p < 0.0005 there was no statistically significant difference in facet size with age across all samples for either males or females. When p<0.05 is considered small significant differences were identified as reported in the table below.

	Sampla	Percentage	significant at	Percentage significant at		
Spinal Region	Sample	p<0.	0005	p<0.05		
		Male	Female	Male	Female	
	SBC	0	0	7	0	
Cervical	SBL	0	0	0	12	
	AS	0	0	0	7	
	SBC	0	0	2	13	
Thoracic	SBL	0	0	2	2	
	AS	0	0	2	2	
	SBC	0	0	0	0	
Lumbar	SBL	0	0	0	20	
	AS	0	0	5	0	

 Table 6-17 Percentage frequency of statistically significant results for facet size with age group

 analysis

These results indicate that age at death is not a significant variable affecting facet size for males and females for all three samples. The null hypothesis tested in this section was that there would be no difference in facet size with age group. This null hypothesis cannot be rejected. The results here suggest that increasing age at death is associated with small difference in facet size when p<0.05.

6.3.6 Age groups v actual age

This section presents the results of testing aggregated and disaggregated data from St Bride's Crypt to test changes in facet size with age. The sex and age of the individuals from St Bride's Crypt is documented and provides a control sample, providing the perfect opportunity to test the hypothesis that there is loss of resolution of the archaeological record when broad age ranges are used to define the age at death of a sample.

In the previous section, the relationship between facet size and age at death using age groups for the St Bride's Crypt sample was explored using Kruskal-Wallis one-way between groups analysis of variance with Bonferroni adjustment (p=0.0005). The results showed that there were no statistically significant relationships between facet size and age group for both males and females. (See Table 6-17).

Spearman's Rho correlational analysis with Bonferroni correction (p<0.0005) was used to test the interrelationship between facet size and age at death of males and females (disaggregated data) from St Bride's Crypt. The strength of correlation was derived as described in the introduction to this chapter. The results can be seen in Appendix B--4.The results are summarised by spinal region in Table 6-18. Both males and females demonstrated facets with negative correlation for size with age (n =13, 14% of facets for males and n = 40, 42% for females). This implies that for certain vertebral levels, facet size decreases with advancing age. Table 6-19 presents the correlation results by spinal region. Strong correlations were seen in the male cervical and lumbar regions and female thoracic region after Bonferroni adjustment was applied. These results show that the ascribing an individual to an age group can "dilute" the relationship between facet size and age such that strong relationships could be overlooked.

Table 6-18 Percentage frequency of correlation classifications between facet size and actual ageat death

Sex	No Correlation (r _s <0.01)	Weak Correlation (r _s =0.1-0.29)	Medium Correlation (r _s =0.3-0.49)	Strong Correlation (r _s =0.5-1.0)	Negative Correlation	P<0.0005
Male	28	40	20	2	13	0
Female	52	39	8	1	40	0

Table 6-19 Percentage frequency of correlation classifications between facet size and actual ageat death by spinal region

Spinal Region	Sex	Negative correlation	No Correlation (r _s <0.01)	Weak Correlation (r _s =0.1-0.29)	Medium Correlation (r _s =0.3-0.49)	Strong Correlation (r _s =0.5-1.0)
Convical	Male	14	28	39	29	4
Cervical	Female	50	50	39	11	0
Thoracic	Male	13	29	50	21	0
moracic	Female	42	55	33	12	2
Lumbar	Male	5	25	65	10	5
	Female	30	55	35	10	0

6.3.7 Relationship between facet size and sample and temporal patterning

The accuracy of discriminant functions used to estimate sex relies on similarity between the individual/group being tested and the sample population on which the function was originally derived. This section tests the null hypothesis was that there would be no difference in facet size between the populations.

The previous results indicate that facet size sexual dimorphism exists in the samples under study and it was observed that there was less difference in facet size between male and females in the Anglo-Saxon group. The next step was to explore inter-sample difference in facet size. Kruskal-Wallace tests were used to identify statistically significant differences in facet size between all three samples for males and females, (p<0.0005 with Bonferroni adjustment). This was followed by a post-hoc Mann-Whitney U test to identify which groups show a statistically significant difference from each other. Kruskal-Wallis tests were performed on all facet levels for the three samples and the percentage frequency of facets showing significant differences are summarised in Table 6-20. (The full table of results can be seen in Appendix B-5).

The results show that with Bonferroni adjustment, there is 1 cervical facet showing a significant difference between samples in facet size for males and 1 cervical and 8 thoracic facets for females. However, if Perneger's (1998) recommendation with regard to p<0.05 is applied, a number of differences between the samples become apparent with 19 male facets and 58 female facets showing significant size difference between samples in facet size.

Table 6-20 Percentage frequency of significant inter-sample differences in facet size using Kruskal-Wallace test

Spinal Region	M	ale	Female			
	p<0.0005	p<0.05	p<0.0005	p<0.05		
Cervical	4	32	4	57		
Thoracic	0	21	17	67		
Lumbar	0	0	0	50		

The next step was to perform a Mann-Witney U test to identify which groups are statistically different from each other. The results are summarised in Table 6--21 (the full table of results is available in Appendix B-6).

		P<().0005	p<0.05		
Spinal Region	Inter-sample Comparison		r of facets	Number of facets		
		Male	Female	Male	Female	
	St Bride's Crypt/St Bride's Lower	0	0	7	4	
Cervical	St Bride's Crypt /Anglo-Saxon	0	39	29	50	
	St Bride's Lower/Anglo-Saxon	39	4	50	75	
	St Bride's Crypt/St Bride's Lower	0	0	0	6	
Thoracic	St Bride's Crypt /Anglo-Saxon	0	23	21	71	
	St Bride's Lower/Anglo-Saxon	23	19	71	65	
	St Bride's Crypt/St Bride's Lower	0	0	0	0	
Lumbar	St Bride's Crypt /Anglo-Saxon	0	5	5	75	
	St Bride's Lower/Anglo-Saxon	5	0	75	80	

Table 6-21 Percentage frequency of significant Mann-Whitney U test results for facet sizebetween sample groups by spinal region

For males, 23 facets show significant difference between St Bride's Lower and Anglo-Saxon samples and for females, 23 facets between St Bride's Crypt and Anglo-Saxon and 10 facets between St Bride's Lower and Anglo-Saxon samples when p<0.0005. There is no difference between St Bride's Crypt and St Bride's Lower for males and females at this significance.

These results indicate that direct comparison in facet size cannot be made between the different samples used in this study, particularly when considering females due to population differences. The null hypothesis that there is no difference in facet size between samples cannot be rejected between St Bride's Crypt and St Bride's Lower but can be rejected between St Bride's Crypt and Anglo-Saxon and St Bride's Lower and Anglo-Saxon.

6.3.8 Effect of sample and temporal variation on the accuracy of sex estimation using DFA equations

To date, there has been little comparative research exploring differences in basal dimensions between samples and how significant any variation might be in terms of the accuracy of discriminant functions used for estimating sex. The discriminant function equations derived for facet size from St Bride's Lower and the Anglo-Saxon samples were used in turn as a test of probability for correct allocation sex of the St Bride's Crypt group as a way of testing the effect of size variation between the three samples identified in the previous section. The equations
were substituted with the St Bride's Crypt data as this is from a documented sample and the actual sex of each individual is known. Table 6-22 summarises the results of the substitutions. Although some St Bride's Lower/St Bride's Crypt percentage accuracies are similar in value, the degree of accuracy is higher in C1, T7, T11, T12 and L1, the same at C2, T8, L5 and reduces for all the other vertebral levels where the degree of accuracy could be calculated (C3, C4, C7, T2, T3, T9, T10, L3, L4). The Anglo-Saxon equations produced lower degrees of accuracy of classification when applied to St Bride's Crypt data across all vertebral levels where substitution was possible. One-way repeated-measures ANOVA was conducted to compare the percentage probability scores when discriminant functions were substituted with the St Bride's Crypt data. There was a statistically significant difference between the results with strong effect size, Wilks' Lambda = 0.136, F (2,5) = 15.81, p = 0.007, multivariate eta squared = 0.864.

Table 6-22 Percentage accuracy of correct classification achieved after discriminant functionsubstitution (vertebral levels for which discriminant functions could not becalculated as a result of missing data or lack of sexual dimorphism are left blank)

Vertebral	St Bride's Crypt with St	St Bride's Lower with St	Anglo Saxon with St
Level	Bride's Crypt data	Bride's Crypt data	Bride's Crypt data
C1	71	73	0
C2	67	67	0
C3	71	67	64
C4	66	61	0
C5	0	0	0
C6	72	0	0
C7	67	51	0
T1	69	0	0
T2	72	63	0
Т3	67	63	51
T4	76	45	55
T5	76	0	55
Т6	77	70	60
T7	68	69	0
Т8	67	67	60
Т9	78	64	57
T10	74	67	0
T11	67	72	6
T12	64	67	0
L1	63	66	0
L2	71	72	0
L3	66	63	0
L4	69	67	0
L5	73	73	0

The aim of this section was to test whether discriminant functions based on facet size could accurately estimate sex in other samples. The results so far have demonstrated that significant variation exists in the dimensions of facet size between the three samples and that these differences can affect the accuracy of discriminant functions when applied to a different sample. The hypothesis that discriminant function equations are not interchangeable between samples cannot be rejected.

Many factors may be responsible for the differences between the groups and the relationship between predictor variables and facet size are analysed and reported in the next section.

6.3.9 Relationship between facet size and predictor variables

This section presents the results of analyses of predictor variables identified in Section 5.7.7. Each variable was analysed separately. Percentage frequencies of strong correlations for each predictor variable by facet level are summarised in table and graph form in this section, whilst the complete tables of results can be found in the attached appendices.

6.3.9.1 Correlation between femoral robusticity and facet size

The femoral robusticity measure was calculated for the right and left femora present using the method described in section 5.7.7.1. The number of individuals and femora used in this correlation study are summarised in Table 6-23. Some individuals had damaged or incomplete femora which meant that both could not be included in the analysis.

Cemeterv	Γ	Male		Female				
,	N (Individuals)	N (Right)	N (Left)	N (Individuals)	N (Right)	N (Left)		
St Bride's Crypt	25	13	12	42	23	19		
St Bride's Lower	35	26	26	24	15	13		
Anglo-Saxon	29	22	15	24	24	18		

Correlation analysis was performed between the results from the femoral robusticity calculations (described in 5.7.7.1) and facet size. The Spearman Ranked Order correlation coefficient (r) was calculated (p<0.0005 after Bonferroni adjustment). This method determines correlation between continuous variables and provides a measure of how the two variables are associated (Pallant 2013).

The results of the correlation analysis were classified according to the method of Cohen (1998) (described in the introduction to this chapter) and presented in Table 6-24. The overall results and two-tailed significance values were also reported in Appendix B-7, column p.

Table 6-24 presents the results of the correlation analysis. The numbers in the table represent the percentage frequency of each facet with the attribute. Both males and females showed negative correlation. Negative correlation was distributed throughout the spinal regions and indicates that as femoral robusticity increases facet size decreases. It is not clear why males should have much higher levels of negative correlation than females in the St Bride's Crypt and Anglo-Saxon populations, whilst the converse is seen In the St Bride's Lower population where females have much higher levels of negative correlation than males for the right side facet joints.

Table 6-24 Percentage frequency of correlation classifications between femoral robusticity and facet size

Sex	No Correlati (rs<0.01		o lation).01)	Weak Correlation (r _s =0.1- 0.29)		Medium Correlation (rs=0.3-0.49)		Strong Correlation (r _s =0.5-1.0)		Negative Correlation		P<0.0005	
		Lt	R	Lt	R	Lt	R	Lt	R	Lt	R	Lt	R
SI	SBC	32	34	51	60	18	30	0	1	65	86	0	0
Male	SBL	38	53	60	46	9	5	0	0	20	60	0	0
	AS	21	25	46	47	28	29	6	3	67	60	0	0
Fomalo	SBC	34	27	43	46	15	25	1	2	38	44	0	0
Female	SBL	32	21	49	63	12	19	8	2	36	89	0	0

Table 6-25 summarises correlation strength by spinal region. In the St Bride's Crypt group, a strong correlation was seen between right femoral robusticity and the right superior facet of T5, the left superior facet of T7, the right superior facets of T8 and T9, the right inferior facet of T12, and the left superior facet of L3 in males and the right superior facet of C5, the right superior facet of T2 and the right inferior facet of T4 in females. A strong correlation was found between left femoral robusticity and the right inferior facet of T7.

No strong correlations were found between femoral robusticity and facet size in males from the St Bride's Lower Group. In females, a strong correlation was found between left femoral robusticity and the left inferior facet of C4, the right inferior facet of T2, the left and right superior and right inferior facets of T3, the right superior facet of T4, the left superior facet of T5, the left superior facet of T12 and all facets of L4 and also between right femoral robusticity and the right inferior facet of T4 and right superior facet of T11.

In the Anglo-Saxon samples, a strong correlation between facet size and left femoral robusticity was found in the male spine at the right inferior facets of C4, T4 and T5, the right superior facet of T6, and the left inferior facet of L2 and L3. In the female spine, a strong correlation was found

between the right inferior facet of C2, the right superior facet of C3, the left superior and right inferior facet of C4, the left and right superior facets of C6, the left and right superior and left inferior facets of C7, the right inferior facet of T4, the left inferior facet of T12. A strong correlation was also found between right femoral robusticity and the right inferior facet of C2, the left and right superior facets of C3, the left superior facet of C4, the left and right inferior facets of C6, the left and right superior and left inferior facet of C7, the right inferior facet of T4 and the left inferior facet of T12.

There was a higher percentage frequency of correlations between facet size and femoral robusticity in the Anglo-Saxon sample than in the other two samples, however no clear pattern of correlation can be seen for individual facets. Overall, femoral robusticity does not demonstrate a linear relationship with facet size for most vertebral levels.

Table 6-25 Percentage frequency of correlation classifications between femoral robusticity andfacet size by spinal region

Spinal Region	Sex	Sample	No Correlation (r₅<0.01)		Weak Correlation (rs=0.1-0.29)		Medium Correlation (r _s =0.3-0.49)		Strong Correlation (r _s =0.5-1.0)	
		$\begin{tabular}{ c c c } \hline & & & & & & & & & & & & & & & & & & $	R	Lt	R	Lt	R	Lt	R	
		SBC	36	50	46	39	18	11	0	0
	Male	SBL	21	64	64	36	14	0	0	0
Convical		AS	36	32	28	36	32	29	4	4
Cervicai		SBC	43	18	39	43	14	36	4	4
	Female	SBL	36	29	50	50	11	21	4	0
		AS	7	7	28	32	25	25	39	36
	Male	SBC	36	25	50	27	15	36	0	10
		SBL	21	42	67	58	10	8	0	0
Thoracic		AS	17	25	50	52	27	27	6	0
THUTACIC		SBC	33	31	42	39	19	27	4	4
	Female	SBL	27	10	54	75	15	21	15	4
		AS	35	42	42	48	13	13	4	4
		SBC	15	25	60	25	25	40	0	5
	Male	SBL	60	65	40	30	0	5	0	0
Lumbar		AS	15	15	60	50	25	35	10	10
Luinbai	F	SBC	25	30	50	65	5	5	0	0
	Female	SBL	35	35	35	55	10	10	20	0
		AS	356	45	40	35	25	20	0	0



Figure 6-9 Percentage frequency of strong correlations between femoral robusticity and facet size

The previous results identify strong correlation between femoral robusticity and individual facets. The next section compares the correlation classifications between superior and inferior facets on the same side of a vertebra to give an overall picture of lateral agreement and allows testing for loss of resolution if the results are amalgamated by side, rather than each facet reported separately. Table 6-26 presents the results achieved when the correlation classification of the left superior/inferior and right superior/inferior facets are compared. The numbers represent the percentage frequency of pairs of facets from the same vertebra with the same classification. This tests for loss of resolution when facet data is amalgamated. The results demonstrate a loss of resolution and indicate that greater accuracy is achieved when the percentage frequency of correlation classification for each facet is reported.

Table 6-26 Percentage frequency of superior and inferior facets with matching correlation

classification

Spinal Region Sex		Sample	No Match		No Correlation (rs<0.01)		Weak Correlation (r _s =0.1-0.29)		Medium Correlation (r _s =0.3-0.49)		Strong Correlation (rs=0.5-1.0)	
			L	R	L	R	L	R	L	R	L	R
		SBC	25	25	7	14	14	11	4	0	0	0
	Male	SBL	25	21	4	18	21	11	0	0	0	0
Corricol		AS	25	28	14	7	4	7	7	7	0	0
Cervical		SBC	28	32	14	4	7	14	0	0	0	0
	Female	SBL	32	28	7	4	11	14	0	4	0	0
		AS	39	36	0	28	7	4	0	7	4	4
N		SBC	35	21	4	17	10	10	0	2	0	0
	Male	SBL	27	25	6	21	17	4	0	0	0	0
Thoracic		AS	38	35	0	4	8	6	4	4	0	0
moracic		SBC	42	38	4	2	4	6	0	4	0	0
	Female	SBL	31	33	2	6	13	8	0	2	4	0
		AS	31	27	10	6	2	13	0	0	6	4
		SBC	35	25	10	15	0	5	0	0	0	0
	Male	SBL	20	30	5	5	25	15	0	0	0	0
Lumbar		AS	30	25	0	0	5	15	10	10	5	0
Lumbar	F	SBC	30	25	0	5	15	5	0	10	0	0
	Female	SBL	40	25	5	0	0	25	0	0	5	0
		AS	40	45	10	0	0	5	0	0	0	10



Figure 6-10 Percentage frequency of matching strong correlations between femoral robusticity and facet size in left and right superior and inferior facets

The correlation classifications between right leg and right facet relative to left facet and right leg and similarly for the left side were compared and the number of facets with matching correlations for ipsilateral and contralateral leg and facets are reported in Table 6-27. The numbers represent the percentage frequency of facets that have the same correlation classification for each comparison. Lt represents the count of facets with the same correlation classification between left facet, left leg and left facet right leg. R represents the count of same correlation classification for the right side.

Table 6-27 Percentage frequency of ipsilateral and contralateral matching correlations femoral
robusticity and facet size

Spinal	St Bride's Crypt			St Bride's Lower				Anglo-Saxon				
Region	Male		Female		Male		Female		Male		Female	
Region	Lt	R	Lt	R	Lt	R	Lt	R	Lt	R	Lt	R
Cervical	57	25	29	18	21	14	21	11	36	29	21	32
Thoracic	17	13	25	18	13	10	25	13	33	27	13	23
Lumbar	20	10	25	25	35	20	15	20	20	30	35	30

These results indicate that for correlation of femoral robusticity and facet size, although there is a degree of symmetry, apart from Anglo-Saxon males, less than 50% of facets show relative differences.

6.3.9.2 Correlation between humeral and facet size directional asymmetry.

Poor preservation restricted the number of humeri available for analysis and the total number used for each population is listed in Table 6-28. Correlation analyses were limited for some facet levels due to lack of suitable data.

Cemeterv	N	lale	Female			
,	N left humeri	N right humeri	N left humeri	N right humeri		
St Bride's Crypt	10	8	19	21		
St Bride's Lower	5	22	13	8		
Anglo-Saxon	8	4	2	3		

Table 6-28 Number of humeri included in correlation

The directional asymmetry between right and left upper limb and right and left superior and inferior facet size was calculated using the method described in 5.7.7.2. The correlation between the two directional asymmetries was then analysed using a 2 x 2 table with Yates' continuity correction used to compensate for over estimation of the Chi-squared value (p<0.001 with Bonferroni correction). The full results are presented in Appendix B-8 and summarised in Table 6-29 and Table 6-30. The phi coefficient was used to measure the degree of association between the two directional asymmetries and classified according to Cohen's (1988) criteria as described in the introduction to this chapter.

Table 6-29 presents the results of the correlation analysis. No correlations showed statistical significance but strong correlations between humeral and facet size asymmetry were noted in the superior and inferior facets of L2 in males and the inferior facets of T9 and the superior facets of L3 in females from St Bride's Crypt; the superior facets of C4 and inferior facets of T2 in females from St Bride's Lower; the superior facets of C4, C7, T10, T11, T12, L2 and L3, inferior facets of C5, T11, L1, and L4 in males and the superior facets of T5, T6, T12, L2, L3, L4 and inferior facets of T2, T6, T7, T9, L3 and L4 in females from the Anglo-Saxon sample.

Table 6-30 summarises the results of correlation analysis by spinal region. The numbers represent the percentage frequency of paired (left and right superior and left and right inferior) facets showing the same correlation classification. It can be seen that the association between humeral and facet size directional asymmetry is non-existent or weak in the majority of vertebral levels for St Bride's Crypt, with the highest count of facets with medium strength associations occurring in the thoracic spine (females having 7 levels both superior and inferior and males 5 facets). Counts of paired facets with strong associations were noted at the superior facet of C4 and the inferior facet of T3 in females from St Bride's Lower, with the highest count of medium strength associations occurred in the thoracic spine (females, 5 and males 3 facets). Strong correlations were also noted in the Anglo-Saxon group (11 male and 12 female) at the superior facets of C4, C7, T10, T11, T12, L2 and L3 and inferior facets of C5, T11, L1 and L4 in males and the superior facets of T5, T6, T12, L2, L3 and L4 and inferior facets of T2, T6, T7, T9, L3, and L4 in females.

Sex	Cemetery	No Correlation (r _s <0.01)	Weak Correlation (r _s =0.1-0.29)	Medium Correlation (r _s =0.3-0.49)	Strong Correlation (r _s =0.5-1.0)	Negative Correlation	P<0.001
Male	SBC	29	17	9	2	21	0
	SBL	18	27	4	0	23	0
	AS	9	15	14	11	21	0
Female	SBC	20	16	11	2	26	0
	SBI	10	27	9	2	17	0

Table 6-29 Percentage frequency of correlation classifications between humeral and facet size directional asymmetry

illustrates the results reported in Table 6-30 and demonstrates the difference in percentage frequency of strong correlations between the samples. Humeral directional asymmetry shows limited use as a predictor of facet size asymmetry in the St Bride's Crypt and St Bride's Lower samples. However there is a greater relationship between these two measures particularly in the male thoracic region and female thoracic and lumbar regions in the Anglo-Saxon sample.

11

8

13

17

0

AS

0

Spinal Region	Sex	Sample	No Correlation (rs<0.01)	Weak Correlation (r _s =0.1-0.29)	Medium Correlation (r₅=0.3-0.49)	Strong Correlation (rs=0.5-1.0)
		SBC	18	18	11	0
	Male	SBL	14	10	0	0
Convical		AS	14	11	11	11
Cervical		SBC	21	25	4	0
	Female	SBL	14	18	14	4
		AS	0	18	0	0
	Male	SBC	19	19	10	0
		SBL	21	21	6	0
Thoracic		AS	8	17	17	8
moracic		SBC	21	10	17	2
	Female	SBL	6	29	10	2
		AS	0	6	15	15
		SBC	20	10	5	10
	Male	SBL	15	30	5	0
Lumbar		AS	5	15	10	20
Lunibal	F	SBC	15	15	15	5
	Female	SBL	15	35	0	0
		AS	0	15	5	25

Table 6-30 Percentage frequency of correlation classification between humeral and facet size

asymmetry by spinal region



Figure 6-11 Percentage frequency of strong correlations between humeral and facet size directional asymmetry

These results indicate that humeral facet directional and facet size asymmetry show more strong correlations in the Anglo-Saxon sample than the St Bride's Samples.

6.3.9.3 Diagnostic criteria for osteoarthritis as predictor variables

The next three sections will consider the effects of eburnation, pitting and osteophytes as predictor variables before assessing correlation of the presence/absence of osteoarthritis as a predictor variable for facet size and angle.

6.3.9.4 Correlation between eburnation score and facet size.

The presence or absence and severity of eburnation seen on each facet were recorded using the method described in 5.7.6.1. The scores for severity and extent were summed to create a new variable called ebtot. The scores represent a ranked order of extent and severity.

The correlation between ebtot and facet size was analysed using Spearman's Ranked Order Correlation with p<0.001 with Bonferroni adjustment. The correlation coefficient was used to measure the degree of association between ebtot and facet size and classified according to Cohen's (1988) criteria as described in 6.2.13.

The full results are presented in Appendix B-9 and summarised below in Table 6-31 and Table 6-32.

Table 6-31 Percentage frequency of correlation classifications between ebtot (total	eburnation
---	------------

Sex	Sample	No Correlation (rs<0.01)	Weak Correlation (r _s =0.1- 0.29)	Medium Correlation (r _s =0.3-0.49)	Strong Correlation (r _s =0.5-1.0)	Negative Correlation	P<0.001	P<0.05
Male	SBC	20	30	10	3	34	0	2
	SBL	19	42	8	2	40	0	4
	AS	28	41	9	10	70	0	8
Female	SBC	30	32	7	0	42	0	0
	SBL	21	18	9	3	34	0	1
	AS	13	45	10	16	48	0	5

score) and facet size

The numbers on the table represent the percentage frequency of facets showing the same correlation classification. Negative correlation was noted in 33 (34%) of male and 40 (42%) of female facets from St Bride's Crypt, 39 (40%) of males and 33 (34%) of female facets from St Bride's Lower and 67 (70%) of male and 46 (48%) of female facets from the Anglo-Saxon sample. This indicates that in these facets, increased ebtot was correlated with smaller facet size. No correlations showed a statistical significance at p<0.0001, however, 14 males and 6 female facets showed significant correlation across all categories at p<0.05. Strong correlations were noted in the male lumbar region for St Bride's Crypt, male thoracic region for St Bride's Lower and thoracic and lumbar for males and all three regions for females from the Anglo-Saxon samples. Degree of OA is known to be related to age. The samples could not be divided into age groups to control for this confounding factor as some groups were too small for analysis.

Table 6-32 summarises the results of the correlation analysis by spinal region. The percentage frequency of facets without eburnation is identified in the "No Eburnation" column and is represented by blank cells in Appendix B-9. For St Bride's Crypt, males and females have similar numbers of facets without eburnation in the cervical region (21% of facets each). This changes in the thoracic region, where males have 42% of facets without eburnation and females have 8%. A similar pattern is seen in the lumbar region where males have 15% and females have 10% of facets without eburnation. The association between eburnation score and facet size is non-existent or weak in the majority of facets with strong correlations between eburnation and facet size seen in the left superior facet of T1, the right superior facet of T3 and the left superior facet of T4 in males, with no strong correlations seen in female spines.

For St Bride's Lower, females have a higher number of facets without eburnation in the cervical region (males 4%, females, 21%). In contrast, in the thoracic region males and females have similar numbers of facets without eburnation (males 40%, females 42%). The lumbar region

pattern reverses that seen in the cervical region with males having 45% and females 25% of facets without eburnation. The association between eburnation score and facet size in non-existent or weak in the majority of facets with strong correlations between eburnation and facet size seen in the right superior facet of C6 and the right superior facets of C3 and C7 and the left superior facet of L5 in females.

For the Anglo-Saxon sample, females have more facets without eburnation in the cervical region than males (14% females, 0% males). In the thoracic and lumbar regions all facets included in this analysis have some eburnation. Strong correlations between eburnation and facet size were seen in the left inferior facet of T1, the right inferior facet of T5, the left inferior facet of T2, the left and right inferior facets of T11, the right and left inferior facets of T12, the left inferior facet of L1 and L2, the left superior facet of L3 in males and in the left and right inferior facets of C2, the left and right superior facets of C3, the right superior facet of C4, the right superior and left and right inferior facets of C5, the right superior and left inferior facets of T11 and the left superior facet of L1 in the female spine.

Figure 6-12 illustrates the results listed in Table 6-32 and clearly identifies that the highest count of strong correlations can be seen in the cervical region of Anglo-Saxon females and thoracic region of Anglo-Saxon males.

The results indicate that ebtot shows the highest count for strong correlation with facet size for the Anglo-Saxon sample and thus has the potential to be a predictor for facet size in this sample. This is not the case for the St Bride's samples.

Table 6-32 Percentage frequency of correlations between ebtot (total eburnation score) and

facet size by spinal region

Spinal Region	Sex	Sample	No Eburnation	No Correlation (rs<0.01)	Weak Correlation (r _s =0.1- 0.29)	Medium Correlation (rs=0.3-0.49)	Strong Correlation (rs=0.5-1.0)
Cervical	Male	SBC	21	21	32	25	0
		SBL	4	14	61	18	4
		AS	0	43	43	18	0
	Female	SBC	21	18	43	18	0
		SBL	21	21	25	29	8
		AS	14	11	21	14	39
Thoracic	Male	SBC	42	19	43	8	6
		SBL	40	23	38	0	0
		AS	100	25	46	0	25
	Female	SBC	8	46	38	8	0
		SBL	42	19	21	0	0
		AS	100	13	56	0	6
Lumbar	Male	SBC	15	25	40	15	0
		SBL	45	15	25	15	0
		AS	100	40	25	20	15
	Female	SBC	75	10	5	10	0
		SBL	65	25	0	5	5
		AS	100	15	50	30	5



Figure 6-12 Percentage frequency of strong correlations between ebtot (total eburnation score) and facet size

6.3.9.5 Correlation between pitting score and facet size

The presence or absence and severity of pitting seen on each facet were recorded using the method described in 5.7.6.2. As with eburnation scores, the pitting scores for severity and extent were added together to create a new variable (pitot). Statistical analysis was performed as for eburnation (see Section 6.3.9.4).

The full results are presented in Appendix B-10 and summarised below in Table 6-33 and Table 6-34.

Sex	Sample	No Correlation (r _s <0.01)	Weak Correlation (r _s =0.1-0.29)	Medium Correlation (rs=0.3-0.49)	Strong Correlation (rs=0.5-1.0)	Negative Correlation	P<0.001
Male	SBC	22	53	23	2	51	0
	SBL	47	39	8	2	36	0
	AS	25	46	13	6	66	0
Female	SBC	38	46	14	1	45	0
	SBL	23	41	13	5	45	0
	AS	40	33	10	21	45	0

Table 6-33 Percentage frequency of correlation classifications between pitot and facet size

Negative correlation was noted in 51% of male and 45% of female facets from St Bride's Crypt; 36% of males and 45% of female facets from St Bride's Lower and 66% of male and 45% of female facets from the Anglo-Saxon sample. This indicates that in these facets, increased pitot score was correlated with smaller facet size. Although no correlations showed a statistical significance, strong correlations were noted in the thoracic and lumbar region in males and the thoracic region for females from St Bride's Crypt, male cervical and thoracic region, female thoracic and lumbar regions for St Bride's Lower and all three regions for males and females from the Anglo-Saxon samples.

Table 6-34 summarises the percentage frequency of facets with each correlation classification by spinal region. The number facets without pitting is identified in the "No Pitting" column and are represented by blank cells in Appendix B-10 The results are illustrated in Figure 6-13

Table 6-34 Percentage frequency of co	rrelation classifications between pitot and facet size by
spinal region	

Spinal Region	Sex	Sample	No pitting	No Correlation (rs<0.01)	Weak Correlation (r _s =0.1- 0.29)	Medium Correlation (rs=0.3-0.49)	Strong Correlation (r _s =0.5-1.0)
Cervical	Male	SBC	0	21	57	21	0
		SBL	0	32	50	14	4
		AS	0	25	50	18	8
	Female	SBC	0	29	61	11	0
		SBL	4	39	39	18	0
		AS	0	21	21	18	43
Thoracic	Male	SBC	0	23	44	31	2
		SBL	0	54	38	0	2
		AS	0	21	52	0	6
	Female	SBC	0	35	44	18	2
		SBL	15	19	46	0	2
		AS	0	38	38	0	10
Lumbar	Male	SBC	0	20	70	5	5
		SBL	0	50	25	20	0
		AS	0	35	25	35	5
	Female	SBC	0	60	30	10	0
		SBL	5	10	30	35	20
		AS	0	20	40	25	15

For St Bride's Crypt, all facet levels have a degree of pitting. The association between pitot and facet size is non-existent or weak in the majority of the facets with some medium strength

associations occurring in all spinal regions. Strong correlations between pitting and facet size were seen in the left inferior facet of T3 and the left superior facet of T2 in males.

For St Bride's Lower, males have pitting in all regions of the spine whereas females have low numbers of facets without pitting in the cervical (4%) and lumbar regions (5%) and a higher number of facets (15%) without pitting in the thoracic region. It can be seen that the association between pitot and facet size is non-existent or weak in the majority of the facets with most medium strength associations occurring in the cervical and lumbar regions for males and females. Strong correlation between pitot and facet size was seen at the right superior facets of C6 and T9 in males and the left inferior facet of T4 in the female spine.

For the Anglo-Saxon sample, all facet levels had a degree of pitting. It can be seen that the association between pitot and facet size is non-existent or weak in the majority of facets with most medium strength associations occurring in the cervical and lumbar regions for males and females. Strong correlations between pitot and facet size were seen in the right inferior facet of C4, the right superior facet of C5 and T2, the left inferior facet of T7 and T8 and the left superior facet of L3 in male spines and the right and left inferior facets of C1, the left and right superior and inferior facets of C2, the right superior facet of C3, the left and right superior facets of C4, the right superior facets of C5, the left superior facet of C6, the right inferior facet of T7, the left inferior facet of T12, the right inferior facet of L1, the left inferior facet of L4 and L5 in the female spine.



Figure 6-13 Percentage frequency of strong correlations between pitot (total pitting score) and facet size

The results indicate that the highest scores for strong correlation classification between pitot and facet size can be seen in the female Anglo-Saxon sample for all spinal regions and in the lumbar region for St Bride's Lower females. These results combined with the low levels of correlation seen in the males indicate that the relationship between pitot and facet size is sample and sex specific.

6.3.9.6 Correlation between osteophyte score and facet size

The method and statistical analysis for correlation between osteophyte score and facet size are the same as used for eburnation (see section 6.3.9.4), creating the new variable ostot. The full results are presented in Appendix B-11 and summarised below in Table 6-35 and Table 6-36.

Sex	Sample	No Correlation (r _s <0.01)	Weak Correlation $(r_s=0.1-0.29)$	Medium Correlation (rs=0.3-0.49)	Strong Correlation (r _s =0.5-1.0)	Negative Correlation	P<0.001
Male	SBC	28	46	11	4	55	0
	SBL	22	46	7	0	50	0
	AS	22	29	10	5	44	0
Female	SBC	41	48	7	1	41	0
	SBL	31	38	10	4	36	0
	AS	21	34	11	8	42	0

Table 6-35 Percentage frequency of correlation classifications between ostot (total osteophyte score) and facet size

Negative correlation was noted in 55% of male and 41% of female facets from St Bride's Crypt, 50% of males and 36% of female facets from St Bride's Lower and 44% of male and 42% of female facets from the Anglo-Saxon sample. This indicates that in these facets increased ostot was correlated with smaller facet size. Although no correlations showed a statistical significance, strong correlations were noted in all regions of the spine in males and the cervical region for females from St Bride's Crypt, female thoracic and lumbar regions for St Bride's Lower and all three regions for males and females from the Anglo-Saxon samples.

Table 6-36 summarises the percentage frequency of correlations by spinal region. The number of facets without osteophytes is identified in the "No Osteophytes" column, and represented by blank cells in Appendix B-11 Figure 6-14 illustrates the results presented in Table 6-36.

For St Bride's Crypt, there are a number of facet levels without osteophytes in the male cervical (14%) and thoracic (13%) regions and female cervical (11%) region. It can be seen that the association between ostot and facet size is non-existent or weak in the majority of the facets with some medium strength associations occurring in all spinal regions with the exception of the male lumbar spine. Strong correlation between ostot and facet size was seen in the left inferior facet of C5, the left superior facet of T5, the right superior facet of T12, and the right superior facet of L4 in males and the right superior facet of C4 in females.

Spinal Region	Sex	Sample	No Osteophytes	No Correlation (r₅<0.01)	Weak Correlation (r _s =0.1- 0.29)	Medium Correlation (rs=0.3-0.49)	Strong Correlation (r _s =0.5-1.0)
Cervical	Male	SBC	14	25	50	7	4
		SBL	7	11	57	25	0
		AS	36	4	36	21	4
	Female	SBC	11	36	46	4	4
		SBL	21	39	32	11	0
		AS	36	11	25	25	7
Thoracic	Male	SBC	13	31	33	19	4
		SBL	17	23	52	0	0
		AS	25	27	23	0	6
	Female	SBC	0	38	54	8	0
		SBL	8	25	46	0	6
		AS	2	19	40	0	10
Lumbar	Male	SBC	0	25	70	0	5
		SBL	0	35	65	0	0
		AS	5	35	35	20	5
	Female	SBC	0	55	35	10	0
		SBL	0	35	25	35	5
		AS	0	40	35	20	5

 Table 6-36 Summary of count of correlation classification between ostot (total osteophyte

score) and facet size by spinal region

For St Bride's Lower, females have more facets without osteophytes in the cervical region (males 7%, females 21%). This pattern is reversed in the thoracic region (males 17%, females 8%) and no facets are without osteophytes in the male and female lumbar regions. It can be seen that the association between ostot and facet size is non-existent or weak in the majority of the facets with medium strength associations occurring in the cervical region for males and the cervical and lumbar regions for females. Strong correlation between ostot and facet size was seen at the left superior facet of T2, the left and right inferior facets of T4 and the left inferior facet of L4 in females.

For the Anglo-Saxon sample, 36% of facets do not have osteophytes in the cervical region for males both males and females. In the thoracic region it is mainly male facets that do not have osteophytes (males 25%, females 2%) and in the lumbar region only 1 male facet (5%) does not have osteophytes. It can be seen that the association between ostot and facet size is non-existent

or weak in the majority of the facets with most medium strength associations occurring in the cervical and lumbar regions for males and females. Strong correlation between ostot and facet size was seen in the left superior facet of C2 and C4, the left and right superior facets of T8 and the right inferior facet of L1 in males and the right superior facet of C3, the left inferior facets of C7 and T1, the left superior facet of T8, the right superior facet of T10, the right superior and left inferior facets of T11 and the left inferior facet of L4 in females.



Figure 6-14 Percentage frequency of strong correlation classifications between ostot (total osteophyte score) and facet size

Overall, the count of strong correlations is low, although the highest score was once again seen in the Anglo-Saxon sample (thoracic region). This indicates that there is a poor relationship between ostot and size.

6.3.9.7 Correlation between vertebral osteophytosis score and facet size

The presence of vertebral osteophytosis was scored as described in 5.7.6.4. The score for the superior and inferior margins of the vertebral body were added together, creating the new variable, cvostot. The statistical analysis details can be seen in Section 6.3.9.4. C1 does not have a vertebral body and is therefore excluded from this analysis.

The full results are presented in Appendix B-12 and summarised below in Table 6-37and Table 6-38.

Table 6-37 Count of correlation classification between cvostot (total score for vertebral

Sex	Sample	No Correlation (r _s <0.01)	Weak Correlation (r _s =0.1-0.29)	Medium Correlation $(r_s=0.3-0.49)$	Strong Correlation (r _s =0.5-1.0)	Negative Correlation	P<0.001
Male	SBC	20	51	19	6	28	0
	SBL	51	36	7	1	35	0
	AS	20	40	13	14	38	0
Female	SBC	45	39	11	1	38	0
	SBL	32	40	14	4	45	0
	AS	11	27	11	17	31	0

osteophytosis) and facet size

Negative correlation was noted in 28% of male and 38% of female facets from St Bride's Crypt, 35% of males and 45% of female facets from St Bride's Lower and 38% of male and 31% of female facets from the Anglo-Saxon sample. This indicates that in these facets, increased cvostot was correlated with smaller facet size. Although no correlations showed a statistical significance, strong correlations were noted in the cervical region in males and the cervical region for females from St Bride's Crypt, the cervical region in males and all three regions in females from St Bride's Lower and all three regions for males and thoracic and lumbar regions in females from the Anglo-Saxon samples. There were no intact female cervical vertebral bodies from the Anglo-Saxon population under study and therefore no conclusions could be drawn about this region.

Table 6-38 summarises the results of the correlation analysis by spinal region. Vertebral bodies that were missing are listed in the "No Vertebral Body" column. This limited the number of analyses that could be performed and is represented by blank cells in Appendix B-12. The shaded area in the appendix table represents the lack of vertebral bodies for C1. No correlation could be drawn for this vertebral level of the spine. Figure 6-15 illustrates the results presented in Table 6-38.

For St Bride's Crypt, it can be seen that the association between cvostot and facet size is nonexistent or weak in the majority of the facets with some medium strength associations occurring in all spinal regions for both sexes. Strong correlation between cvostot and facet size was seen in the left inferior facet of C6, the left inferior facet of T2, the right inferior facet of T3, the left inferior facet of T7, the right and left inferior facets of L4 in males and the right superior facet of T5 in females.

Spinal Region	Sex	Sample	No Vertebral bodies	No Correlation (r₅<0.01)	Weak Correlation (r _s =0.1- 0.29)	Medium Correlation (rs=0.3-0.49)	Strong Correlation (r₅=0.5-1.0)
Cervical	Male	SBC	0	25	57	4	4
		SBL	0	54	25	4	4
		AS	4	11	14	11	25
	Female	SBC	0	29	46	14	0
		SBL	0	32	50	4	4
		AS	21				
Thoracic	Male	SBC	0	15	54	48	6
		SBL	0	42	50	8	0
		AS	0	31	56	13	2
	Female	SBC	0	50	35	13	2
		SBL	0	35	38	23	2
		AS	2	19	38	17	21
Lumbar	Male	SBC	0	25	35	25	10
		SBL	0	70	20	5	0
		AS	0	10	45	15	25
	Female	SBC	0	55	35	5	0
		SBL	5	25	35	5	10
		AS	0	10	40	15	30

Table 6-38 Percentage frequency of correlations between cvostot and facet size by spinal region

For St Bride's Lower, it can be seen that the association between cvostot and facet size is nonexistent or weak in the majority of the facets with medium strength associations occurring in all spinal regions for both sexes. Strong correlation between cvostot and facet size was seen at the right inferior facet of C5 in males and the left superior facet of C5, the right superior facet of T12 and the left and right inferior facets of L5 in females.

For the Anglo-Saxon sample, no female cervical vertebral bodies were sufficiently preserved for data collection. It can be seen that the association between cvostot and facet size is none-existent or weak in the majority of the facets with most medium strength associations occurring in all observable spinal regions for males and females. Strong correlation between cvostot and facet size was seen in the right superior and right inferior facets of C5, all facets of C6, the left superior facet of C7, the left superior facet of T5, the left inferior facet of L2 and L3, the right superior and left inferior facet of L4, the right inferior facet of L5 in males and the right and left inferior facets of T4, the left of T6, the left superior and left inferior facets of T7, the left and right

inferior facets of T9, the left and right superior facets of L4 and the right superior and right inferior facets of L5 in females.



Figure 6-15 Percentage frequency of strong correlations between CVOS (total vertebral body osteophytosis score) and facet size

The highest percentage frequency is seen in the thoracic region of Anglo-Saxon females and the cervical region of Anglo-Saxon males. The limited preservation of the Anglo-Saxon female cervical vertebral bodies limits the conclusions that can be drawn from this analysis.

6.4 Facet Angle

This section presents the results generated by analysis of sexual dimorphism of and the relationship between facet angle with age, femoral robusticity, humeral directional asymmetry, facet eburnation, pitting, osteophyte scores and vertebral body osteophytosis.

6.4.1 Facet angle metric analysis

The number of facets analysed for each sample is summarised in Table 6-11. Descriptive statistics for facet angle for each sample using age combined, sex-divided groups are reported in Appendix C-1. As with facet size, a number of outlying values were observed (see Table 6-39). Non-parametric statistical methods were used to reduce the effect of the outlying data (see 6.3.1)

Spinal region	Facet angle					
	St Bride's Crypt	St Bride's Lower	Anglo-Saxon			
Cervical	2	3	2			
Thoracic	3	3	2			
Lumbar	2	3	2			

Table 6-39 Percentage of data outliers by spinal region and sample

The maximum and minimum standard deviation values and the facet levels they occurred at are summarised in Table 6-40. The range represents the natural variation in angle that can be seen at the same facet levels from different individuals.

	St Bride's Crypt		St Bride	's Lower	Anglo-Saxon		
	Male	Female	Male	Female	Male	Female	
Minimum Standard	3.02	2.98	2.67	2.28	3.4	3.82	
Deviation	LtST6	LtST6	RST4	LtSC1	RST9	LtST4	
Maximum Standard	30.76	29.37	27.57	24.19	33.94	37.11	
Deviation	RIT11	LtST12	LtST12	LtIT12	LtIT2	RIT2	

 Table 6-40 Maximum and minimum standard deviation values for facet angle

Key: C=cervical, T=thoracic, L=lumbar, R=right, Lt=left, S=superior, I=inferior

6.4.2 Sexual dimorphism index for facet angle

A sexual dimorphism index (SDI) was used to ascertain the degree of sexual dimorphism in size in facet angle measurements. As in Section 6.2.2, differentiation between the sexes was considered to have occurred at an index value of 110 or above or 90 and below. Values between these figures do not differentiate between the sexes with any certainty.

Table 6-41 summarises the results of SDI analysis presented in Appendix C-1. The SDI results show that the SDI is unable to clearly differentiate between sexes in 85% of cases for St Bride's Crypt, 76% for St Bride's Lower and 73% for the Anglo-Saxon sample. This indicates that there is a low level of sexual dimorphism in facet angle for the St Bride's samples and less so for the Anglo-Saxons. The SDI showed greatest degree of difference in the lumbar region in the St Bride's Crypt and Anglo-Saxon samples.

Figure 6-16 overleaf illustrates the comparison in SDI values for facet angle across the three samples by facet level.

Sample	Spinal Region	SDI>110	SDI<90	Unclassified
	Cervical	1	0	27
St Bride's Crypt	Thoracic	0	5	43
	Lumbar	0	8	12
	Cervical	0	1	27
St Bride's Lower	Thoracic	0	1	47
	Lumbar	0	1	19
	Cervical	10	1	17
Anglo-Saxon	Thoracic	2	0	46
	Lumbar	1	12	7

Table 6-41 Count of Sexual Dimorphism Index (SDI) results by sample









Figure 6-16 Sexual dimorphism index scores for facet angle

6.4.3 Statistical analysis of the relationship between sex and facet angle

There is a less obvious pattern of sexual dimorphism in facet angle when compared to facet size. Mann-Whitney U with Bonferroni correction ($\alpha = 0.0005$) was used to determine the statistical significance of differences in facet angle between males and females. The full results for these analyses can be seen Appendix C-2. The significant results are summarised by region in Table 6-42 and illustrated in Figure 6-17.

		Percentage	Percentage
Spinal Region	Population	significant at	significant at
		p<0.0005	p<0.05
	SBC	0	7
Cervical	SBL	0	11
	AS	0	40
	SBC	2	2
Thoracic	SBL	0	4
	AS	0	4
	SBC	0	5
Lumbar	SBL	0	0
	AS	5	0

Table 6-42 Percentage frequency of statistically significant results, facet angle with sex

The Bonferroni correction reduced the significant p value to p<0.0005 meaning few statistically significant differences in facet angle between the sexes were identified; 1 for St Bride's Crypt in the thoracic region and 1 for Anglo-Saxon in the lumbar region representing 2% and 5% of facets respectively.

The null hypothesis tested in this section (See Chapter 1) is that there will be no measurable difference in sexual dimorphism in facet joint sagittal orientation. This null hypothesis cannot be rejected for all three samples for most facet levels with degree of orientation being statistically insignificant between males and females.



Figure 6-17 Percentage frequency of significant differences in facet angle with sex by sample and spinal region

6.4.4 Stepwise discriminant function analysis for facet angle

The derivation of stepwise discriminant function in this study depends on clear classification into male or female categories. Facet angle does not show statistical significance in angle differences, however the SDI showed greater variation for the Anglo-Saxon sample. Therefore, stepwise discriminant function analysis was performed as described in 5.8.2.1. (The rationale for using this process is explained in section 6.3.4). The results of this analysis can be seen in Table 6-43. The equations relate to each vertebral level and are derived from the values of all four facets angles for that vertebral level. This creates more equations than would be achieved if each facet was considered individually. The p value result is from the Wilk's Lambda test for goodness of fit for the model and tests for significant differences between the male and female groups. In this analysis, the p value was set at 0.0006 (with Bonferroni correction). Results for each sample are included in the table. This analysis generated discriminant functions to identify the sex of an individual based on the angle of the facets at each vertebral level. In the event of little of no difference in facet angle between the sexes, no variables qualified for inclusion in the equation.

The total percentage accuracy of correct assignment listed in this table will be used to compare accuracy against the use of extrinsic variables in deriving discriminant function equations in Section 6.4.

Vertebral	Sample	р	F(n)	Zm	Zf	Nm	Nf	ZO	% Ac	curac	У
Level									М	F	Total
C1	SBC	0.017	F=(-5.84)+(0.17)C1RI	0.39	-0.29	7	29	0.05	30	70	66
	SBL		No qualifying variables								
	AS	0.038	F=(-3.47)+(0.01)C1LtI	0.36	-0.63	13	5	-0.13	93	56	78
C2	SBC		No qualifying variables								
	SBL	0.024	F=(-5.53)+(0.1)C2RI	-0.29	0.403	26	10	0.06	90	50	74
	AS	0.013	F=(-4.59)+(0.07)C2LtI	0.51	-0.76	9	6	-0.13	75	67	71
С3	SBC		No qualifying variables								
	SBL	0.005	F=(-9.56)+(0.12)C3LtI	-0.37	0.5	21	11	0.07	78	52	67
	AS	0.001	F=(-8.29)+(0.09)C3LtI	0.65	-1.01	12	7	-0.18	86	78	83
C4	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS		No qualifying variables								
C5	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS	0.037	F=(-7.64)+(0.08)C5RS	-0.456	0.557	9	6	0.051	82	67	75
C6	SBC		No qualifying variables								
	SBL	0.052	F=(-20.44)+(0.23)C6LtI	-0.25	0.31	23	11	0.03	74	48	63
	AS		No qualifying variables								
C7	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS		No qualifying variables								

Table 6-43 Stepwise discriminant function analysis of facet angle

Key to Table 6-43: SBC = St Bride's Crypt, SBL = St Bride's Lower, AS = Anglo-Saxon, C = cervical, T = thoracic, L = lumbar, 1,2,3 etc. refers to the vertebral level L = left, R = right, S = superior, I = inferior Zm = male group centroid, Zf = female group centroid, Nm = number of males identified, Nf= number of females identified, Z_0 = demarcation point.

Vertebral	Sample	р	F(n)	Zm	Zf	Nm	Nf	Z0	% Ac	curac	у
Level									м	F	Total
T1	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS		No qualifying variables								
T2	SBC	0.001	F=(-22.54)+(0.22)T2RS	0.52	-0.36	14	31	0.08	54	82	70
	SBL		No qualifying variables								
	AS		No qualifying variables								
Т3	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS		No qualifying variables								
T4	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS	0.025	F=(-22.95)+(0.21)T4RS	0.4	-0.55	12	6	-0.07	80	50	67
Т5	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS		No qualifying variables								
Т6	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS		No qualifying variables								

Vertebral	Sample	р	F(n)	Zm	Zf	Nm	Nf	Z0	% Accuracy		
Level									м	F	Total
Т7	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS		No qualifying variables								
Т8	SBC		No qualifying variables								
	SBL	0.016	F=(-20.85)+(0.22)T8LtI	0.27	0.47	27	6	0.37	85	33	66
	AS		No qualifying variables								
Т9	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS	0.053	F=(-22.29)+(0.21)T9RS	0.4	-0.48	10	9	-0.04	77	82	79
T10	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS		No qualifying variables								
T11	SBC	0.015	F=(-2.37)+(0.04)T11RI	-0.39	0.25	12	29	-0.07	48	74	64
	SBL		No qualifying variables								
	AS		No qualifying variables								
T12	SBC	0.015	F=(-2.24)+(0.03)T12RS	-0.4	0.26	15	26	-0.07	63	72	68
	SBL		No qualifying variables								
	AS		No qualifying variables								

Table 6-43 Stepwise discriminant function analysis of facet angle continued

Vertebral	Sample	р	F(n)	Zm	Zf	Nm	Nf	Z0	% Accuracy		/
Level									М	F	Total
L1	SBC	0.002	F=(0.51)+(-	-0.56	0.4	12	30	-0.08	48	83	69
			0.16)L1LtI+(0.12)L1RS								
	SBL		No qualifying variables								
	AS		No qualifying variables								
L2	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS		No qualifying variables								
L3	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS	0.034	F=(-3.32)+(0.11)L3RS	-0.41	0.48	13	9	0.03	81	64	73
L4	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS	0.002	F=(-4.29)+(0.91)Lt4RI	-0.56	0.84	12	8	0.14	80	80	80
L5	SBC		No qualifying variables								
	SBL		No qualifying variables								
	AS		No qualifying variables								

Table 6-43 Stepwise discriminant function analysis of facet angle continued

6.4.5 Relationship between age and facet angle.

This section presents the results of analysis of the relationship between facet angle and age. Samples were divided into age groups are described in Section 5.7.3. Kruskal Wallace one-way between–groups analysis of variance with Bonferroni correction (p<0.0005) was used to identify any significant differences in facet angle across the three age groups. Males and females were analysed separately to reduce the effect of sexual dimorphism. The full results can be seen in Appendix C-3.

Table 6-44 summarises the results of the Kruskal Wallace analysis of differences in facet angle with age. With the p value set at 0.0005 (with Bonferroni correction), there was no statistically significant difference in facet angle between age groups for either males or females for all three samples. When the results from p<0.05 are considered, in the St Bride's Crypt population, 4% of male cervical, 5% of male lumbar, 7% of female cervical and 5% of female lumbar vertebrae

showed change in facet size with age; for St Bride's lower, 2% of male thoracic, 4% of female cervical and thoracic vertebrae, 20% of lumbar and in the Anglo-Saxon population, 5% of male lumbar and 4% for female cervical vertebrae showed change in size with age.

Spinal Region	Population	Signifi p<0.	cant at 0005	Significant at p<0.05		
		Male	Female	Male	Female	
	SBC	0	0	4	7	
Cervical	SBL	0	0	0	4	
	AS	0	0	0	4	
	SBC	0	0	0	0	
Thoracic	SBL	0	0	2	4	
	AS	0	0	0	0	
	SBC	0	0	5	5	
Lumbar	SBL	0	0	0	20	
	AS	0	0	5	0	

Table 6-44 Percentage frequency of statistically significant results for facet angle with age group

The results show that age at death is not a significant variable in relation to facet angle. The null hypothesis tested in this section was that there would be no difference in facet angle with age group. This null hypothesis cannot be rejected. The results presented suggest that increasing age does not cause a change in sagittal orientation of the facet joints of the spine.

6.4.6 Age groups v actual age

Section 6.3.6 describes the opportunity to test the outcomes of analysing facet size with age group and actual age at death when a sample is from a documented population. This process was repeated for facet angle.

In Section 6.4.5, the relationship between facet angle and age at death using age groups for the St Bride's Crypt population was explored using a Kruskal-Wallis one way between groups analysis of variance with Bonferroni adjustment (p=0.0005). The results showed that there were no statistically significant relationship between facet angle and age group for both males and females. (See Table 6-44).

Spearman's Rho correlational analysis with Bonferroni correction (p<0.0005) was used to test the interrelationship between facet angle and age at death of males and females (disaggregated data) from St Bride's Crypt. The results can be seen in Appendix C-4.

The results are presented in Table 6-45. Both males and females demonstrated facets with negative correlation for angle with age (68% for males and 51% for females). This implies that for certain vertebral levels, lower facets angle are associated with older age at death. Table 6-46 summarises the correlation results by spinal region. No strong correlations were observed and there were no statistically significant relationships between facet angle and age.

Table 6-45 Percentage frequency of correlation classification between facet angle and actualage at death, St Bride's Crypt

Sex	No Correlation	Weak Correlation (r _s =0.1-0.29)	Medium Correlation (r _s =0.3-0.49)	Strong Correlation (r _s =0.5-1.0)	Negative Correlation	P<0.0005
Male	28	53	19	0	68	0
Female	41	51	8	0	51	0

Table 6-46 Percentage frequency of correlation classification between facet angle and actualage at death by spinal region

Spinal Region	Sex	No Correlation (r₅<0.01)	Weak Correlation (rs=0.1-0.29)	Medium Correlation (r _s =0.3-0.49)	Strong Correlation (r _s =0.5-1.0)
Convical	Male	14	72	14	0
Cervical	Female	29	54	18	0
Thorasis	Male	38	43	19	0
moracic	Female	40	54	6	0
Lumbar	Male	25	50	25	0
	Female	60	40	0	0

The results of comparison between aggregated and disaggregated data for facet angle and age at death does not reveal any strong correlation results that would have been missed by using aggregated age groups.

For facet angle the null hypothesis that there is no loss of resolution of the archaeological record when broad age ranges are used to define age at death in a sample cannot be rejected.

6.4.7 Relationship between facet angle and sample and temporal patterning

As for facet size inter-sample comparison (see section 6.3.7), Kruskal-Wallace (p<0.0005 with Bonferroni adjustment) with Mann-Whitney U post hoc analysis was used to test the null hypothesis that there was no difference in facet angle means between samples.

Table 6-47 summarises the results for the Kruskal Wallis test. It highlights the number of facets showing significant difference in facet angle between the groups. The full table of results can be found in Appendix C-5. There are a 21 male and 12 female facets showing significant inter-sample differences after Bonferroni adjustment. When p<0.05, 57 male and 35 female facets show significant differences.

Table 6-47 Percentage frequency of significant results from Kruskal Wallis Test for inter-sample differences in facet angle

Spinal Region	Ma	ale	Female			
opinal negion	p<0.0005	p<0.05	p<0.0005	p<0.05		
Cervical	14	71	4	36		
Thoracic	35	63	23	46		
Lumbar	0	30	0	15		

As with facet size, a post hoc Mann-Whitney U test was performed to differentiate between samples and the results are summarised in Table 6-48. The full table of results can be seen in Appendix C-6.

Significant inter-sample differences in facet angle were observed between all samples in all spinal regions (except lumbar region for St Bride's Crypt and Anglo-Saxon) for males and females when p<0.0005. The null hypothesis that there is no difference in inter-sample facet angle can be rejected.
Table 6-48 Percentage frequency of statistically significant differences in facet angle betweensample groups by spinal region

		P<0	0.0005	p<0.05		
Spinal Region	Inter-sample Comparison	Numbe	r of facets	Number of facets		
		Male	Female	Male	Female	
	St Bride's Crypt/St Bride's Lower	4	0	50	21	
Cervical	St Bride's Crypt /Anglo-Saxon	4	4	43	21	
	St Bride's Lower/Anglo-Saxon	21	4	39	25	
	St Bride's Crypt/St Bride's Lower	15	4	38	24	
Thoracic	St Bride's Crypt /Anglo-Saxon	13	13	46	35	
	St Bride's Lower/Anglo-Saxon	35	23	56	48	
	St Bride's Crypt/St Bride's Lower	0	5	5	0	
Lumbar	St Bride's Crypt /Anglo-Saxon	0	0	35	10	
	St Bride's Lower/Anglo-Saxon	5	0	50	10	

6.4.8 Relationship between facet angle and predictor variables

This section presents the results of analyses of predictor variables identified in Section 5.7.7. Each variable was analysed separately. Strong correlations for each predictor variable are summarised in table and graph form in this section, whilst the complete tables of results can be found in the attached appendices.

6.4.8.1 Correlation between femoral robusticity and facet angle

The methods used to calculate femoral robusticity, the number of femora available for inclusion in the correlation analysis and the method of statistical analysis can be seen in 6.3.9.1. The percentage frequency of individual facets with each correlation classification are summarised in Table 6-49. The full table of results can be seen in Appendix C-7.

Table 6-49 presents the results of the correlation analysis with the numbers representing the percentage frequency of each facet with the same correlation attribute. Both males and females showed negative correlation. Negative correlation was distributed throughout the spinal regions and indicates that, as femoral robusticity increases facet angle decreases. The number of

vertebral levels with negative correlation is similar between males and females for all samples with little inter-sample variation

Table 6-49 Percentage frequency of correlation classifications between femoral robusticity and	d
facet angle	

Sex	Cemetery	No Correlation (r _s <0.01)		Weak Correlation (r _s =0.1-0.29)		Medium Correlation (rs=0.3-0.49)		Strong Correlation (rs=0.5-1.0)		Negative Correlation		P<0.0005	
		L	R	L	R	L	R	L	R	L	R	L	R
Male	SBC	39	27	47	48	24	19	10	6	41	44	0	0
	SBL	41	56	59	53	3	3	3	3	51	42	0	0
	AS	23	21	43	52	22	21	11	7	43	46	0	0
Female	SBC	18	26	54	49	21	22	6	2	55	55	0	0
	SBL	18	20	48	43	18	29	22	1	49	35	0	0
	AS	22	24	38	52	27	33	15	13	55	47	0	0

Table 6-50 Percentage frequency of correlation classifications between femoral robusticity and

Spinal Region	Sex	Cemetery	No Correlation (rs<0.01)		Weak Correlation (r₅=0.1-0.29)		Medium Correlation (r _s =0.3-0.49)		Strong Correlation (rs=0.5-1.0)	
			L	R	L	R	L	R	L	R
Cervical	Male	SBC	18	21	43	27	25	28	14	4
		SBL	43	39	54	57	0	0	4	4
		AS	18	14	61	57	11	21	11	7
	Female	SBC	25	21	57	39	11	36	7	4
		SBL	29	29	36	39	18	29	18	4
		AS	14	11	29	39	21	25	36	25
Thoracic	Male	SBC	10	31	46	50	23	13	10	6
		SBL	42	48	60	29	6	2	4	4
		AS	25	23	42	52	23	21	10	6
	Female	SBC	19	27	52	52	27	19	2	2
		SBL	17	15	69	46	10	25	15	0
		AS	21	33	42	44	33	19	6	6
Lumbar	Male	SBC	15	25	55	45	25	20	1	10
		SBL	35	45	65	45	0	10	0	0
		AS	30	25	20	45	35	20	15	10
	Female	SBC	5	30	55	55	25	15	15	0
		SBL	5	20	15	40	35	40	45	0
		AS	35	20	40	40	20	30	5	10

facet angle by spinal region





Figure 6-18 Percentage frequency of strong correlations between femoral robusticity and facet angle by spinal region

Table 6-50 summarises the results of the percentage frequency of individual facets with each correlation classification between femoral robusticity and facet angle by spinal region. In the St Bride's Crypt sample, a strong correlation was seen between facet angle and left femoral robusticity in the male right superior facet of C1, the left inferior facet of C4, the right superior and left inferior facet of C6, the left and right superior and right inferior facets of T7, the right inferior facet of T10, the right superior facet of T11 and the right inferior facet of L3 and between facet angle and right femoral robusticity in the left inferior facet of C7, the left superior facet of L3. For females, a strong correlation between facet angle and left femoral robusticity was seen at the left and right superior facets of C4, the right superior facet of T12, the left and right superior facet of T12, the left and right superior facet of T12, the left and right femoral robusticity was seen at the left and right superior facets of C4, the right superior facet of T12, the left and right inferior facets of L4 and the left superior facets of L5 and between facet angle and right femoral robusticity at the right superior facet of C4 and the right superior facet of T12.

For the St Bride's Lower sample, a strong correlation between facet angle and left femoral robusticity was seen in males at the right inferior facet of C3, T9 and T10, the left superior facet of T11, and between facet angle and right femoral robusticity at the right inferior facet of C3, T9 and T10. In the female spine, a strong correlation between facet angle and femoral robusticity was seen at the left and right superior and right inferior facets of C4, the right superior facet of C5, the right superior facet of C6, the left inferior facet of T1, the left and right inferior facets of T12, the left and right superior and right inferior facets of L1, the right superior facet of L5 and between facet angle and right femoral robusticity at the left superior facet of L5 and between facet angle and right femoral robusticity at the left superior facet of C5, the left and right inferior facets of L3, the left superior facet of L5 and between facet angle and right femoral robusticity at the left superior facet of C5, the left and right inferior facets of T1, the right superior facet of L5 and between facet angle and right femoral robusticity at the left superior facet of C5, the left and right inferior facets of T1, the right superior facet of C5, the left and right inferior facets of T3, the left superior facet of L5 and between facet angle and right femoral robusticity at the left superior facet of C5, the left superior facets of T1, the right superior facets of T2, T3 and T4, the right superior

facet of T5, the left superior facet of T7, the left superior facet of T9 and the left superior facet of T11.

For the Anglo-Saxon sample, a strong correlation between facet angle and left femoral robusticity in males was seen at the left superior and inferior facets of C2, the right inferior facet of C7, the left and right superior facets of T2, the left and right inferior facets of T6, the right inferior facet of T7, the right superior facet of L1, the left inferior facets of L4 and L5 and between facet angle and right femoral robusticity, the left superior and inferior facets of C2, the left and right superior facets of T2, the right inferior facet of T7, the left superior facet of L1 and the left inferior facet of L4. Anglo-Saxon females showed a higher number of strong correlations between facet angle and femoral robusticity than males. A strong correlation between facet angle and left femoral robusticity was seen at the right superior and inferior facets of C1, the left and right superior and left inferior facets of C2, the left and right inferior facets of C4, the left inferior facet of C5, the left inferior facets of C6 and C7, the right superior and inferior facets of T6, the right inferior facet of T12 and between facet angle and right femoral robusticity at the left and right inferior facets of C2, the left inferior facet of C3, the left and right inferior facets of C4, the left inferior facets of C5 and C7, the right inferior facet of T5, the left superior and right inferior facets of C4, the left inferior facets of C5 and C7, the right inferior facet of T5, the left superior and right inferior facets of T12 and the right superior facets of L1 and L2.

As with the results for correlation between facet size and femoral robusticity, the loss of resolution when amalgamated data is used was also tested for facet angle. The results can be seen in Table 6-51. Again, there are superior and inferior facets from the same vertebra that do not have the same classification for strength of correlation and reporting them as pairs leads to loss of resolution of results. Therefore, it is appropriate to record the count of individual facets rather than amalgamated superior and inferior facets with strong correlation classification for this analysis.

Spinal Region	Sex	Cemetery	No Match		No Correlation (rs<0.01)		Weak Correlation (r _s =0.1-0.29)		Medium Correlation (rs=0.3-0.49)		Strong Correlation (rs=0.5-1.0)	
			Lt	R	Lt	R	Lt	R	Lt	R	Lt	R
Cervical	Male	SBC	32	29	0	0	11	18	7	4	0	0
		SBL	36	36	7	4	7	11	0	0	0	0
		AS	25	25	0	0	21	21	0	0	4	4
	Female	SBC	29	32	0	0	21	11	0	7	0	0
		SBL	43	29	4	0	0	11	0	11	4	0
		AS	32	39	0	0	7	11	0	0	11	0
Thoracic	Male	SBC	33	31	4	8	8	6	2	4	2	0
		SBL	25	29	13	13	13	8	2	4	2	0
		AS	35	38	6	2	8	8	0	2	0	0
	Female	SBC	33	28	2	6	15	13	0	4	0	0
		SBL	28	38	2	4	17	4	2	2	2	2
		AS	35	40	0	2	6	4	8	4	0	0
Lumbar	Male	SBC	20	25	5	10	25	40	0	0	0	0
		SBL	40	5	0	5	10	20	0	0	0	0
		AS	30	25	0	5	15	20	5	0	0	0
	Female	SBC	30	25	0	0	20	20	0	5	0	5
		SBL	30	30	0	0	20	0	0	0	0	5
		AS	35	35	0	0	10	15	5	0	0	0

Table 6-51 Percentage Frequency of superior and inferior facets with the same classification of

correlation

Table 6-52 reports the results for the percentage frequency of matching correlations between left facet and left leg relative with left facet and right leg and right facet and right leg with right facet and left leg as a measure of asymmetry. The numbers represent the percentage frequency of facets that have the same correlation classification for each comparison. The highest number of matching correlation strengths (45%) can be seen in St Bride's Lower males for right facet and right leg in agreement with right facet and left leg with the next highest (40%) also being for St Bride's Lower males for left facet and left leg in agreement with left facet and right leg.



Figure 6-19 Percentage frequency of matching correlations between femoral robusticity and facet angle.

Table 6-52 Percentage frequency of ipsilateral and contralateral matching correlations femoral
robusticity and facet angle

Spinal	S	t Brid	e's Cry	/pt	St Bride's Lower				Anglo-Saxon			
Region	M	ale	Female		Male		Female		Male		Female	
negion	L	R	L	R	L	R	L	R	L	R	L	R
Cervical	21	19	18	17	37	32	21	11	18	18	32	36
Thoracic	19	21	23	25	38	40	10	15	27	31	21	29
Lumbar	5	10	20	20	25	45	10	5	15	20	25	25

6.4.8.2 Correlation between humeral and facet angle asymmetry

As in the previous section data analysis was hampered by poor preservation of humeri. The numbers of humeri by cemetery available for inclusion in this correlation study can be seen in Table 6-28. The directional asymmetry between right and left upper limb and right and left superior and inferior facet angle was calculated using the method described in section 5.7.7.2. The correlation between the two directional asymmetries was then analysed using a 2 x 2 table. Yates' continuity correction was used to compensate for over-estimation of the chi-squared value. The phi coefficient was used to measure the degree of association between the two directional asymmetries and classified according to Cohen's (1998) criteria as outlined in the introduction to this chapter (p<0.001 with Bonferroni adjustment). The full results are presented in Appendix C--8 and summarised in Table 6-53 and Table 6-54.

Sex	Cemetery	No Correlation (r _s <0.01)	Weak Correlation (r _s =0.1-0.29)	Medium Correlation (r _s =0.3-0.49)	Strong Correlation (r _s =0.5-1.0)	Negative Correlation	P<0.001
Male	SBC	9	23	10	5	15	0
	SBL	21	19	8	1	21	0
	AS	5	10	16	18	21	0
Female	SBC	17	30	3	0	21	0
	SBL	13	22	13	3	20	0
	AS	2	5	13	12	15	0

directional asymmetry

Table 6-53 presents the results of the correlation analysis. The numbers represent the percentage frequency of facets with each correlation classification (left and right facets were paired together to measure the degree of directional asymmetry). No correlations showed statistical significance but strong correlations between humeral and facet angle asymmetry were noted at the inferior angles C7 and T2, the superior angles of T9, T12 and L3 in males and no strong correlations noted in females from St Bride's Crypt; the superior angles of C2, T9 and L2 in females from St Bride's Lower and the superior angles of C1, C4, C5, C6, T5, T6, T11, T12, L2, L4 and L5 and the inferior angles of C2, C7, T8 and T11 in males and superior angles of T3, T4, T5, T6, T9, T12, L2 and L4 and the inferior angles of T2, T6, and L3 in females from the Anglo-Saxon sample.

Table 6-54 summarises the results of correlation analysis by spinal region. It can be seen that the association between humeral and facet angle directional asymmetry is non- existent or weak in the majority of vertebral levels for St Bride's Crypt, with most medium strength associations occurring in the thoracic spine with females having 7 thoracic facet levels (including superior and inferior levels) and males 5 thoracic levels at medium strength correlation. A strong correlation was seen on 5 vertebral levels for males and none in females. St Bride's Lower population showed less strong correlations (2 males and 3 female) and most medium strength associations occurred in the thoracic spine (overall 8 male and 12 female levels). The Anglo-Saxon sample showed much higher numbers of strong correlations between humeral and facet angle directional asymmetry; 11 male and 12 female, although correlations were not performed for the female cervical spine due to lack of measurable data. Figure 6-20 is a graphical representation of the data in Table 6-54. The graph clearly identifies that increased count of strong correlations seen in the Anglo-Saxon sample when compared the St Bride's Samples.

Table 6-54 Percentage frequency between humeral and facet angle directional asymmetry by

spinal region

Spinal Region	Sex	Sample	No Correlation (rs<0.01)	Weak Correlation (rs=0.1- 0.29)	Medium Correlation (r _s =0.3-0.49)	Strong Correlation (r _s =0.5-1.0)
Cervical	Male	SBC	11	25	11	3
		SBL	18	25	11	3
		AS	0	7	18	25
	Female	SBC	14	32	3	0
		SBL	18	25	3	3
		AS	No data	No data	No data	No data
Thoracic	Male	SBC	10	19	10	6
		SBL	23	15	10	0
		AS	10	10	15	13
	Female	SBC	17	31	2	0
		SBL	10	10	19	2
		AS	4	6	17	17
Lumbar	Male	SBC	5	30	10	5
		SBL	20	30	0	0
		AS	0	15	15	20
	Female	SBC	20	25	5	0
		SBL	10	20	15	5
		AS	0	10	20	15



Figure 6-20 Percentage frequency of strong correlations between humeral and facet angle asymmetry

These results indicate that humeral directional and facet angle asymmetry show more strong correlations in the Anglo-Saxon sample (particularly in the cervical and thoracic region in males and thoracic and lumbar regions in females) than the St Bride's Samples and indicates the potential for humeral directional asymmetry to be a predictor variable in relation to Anglo-Saxon facet angle asymmetry.

6.4.8.3 Correlation between eburnation score and facet angle

The same method of scoring eburnation and statistical method were used as for eburnation and facet size (See Section 6.3.9.4). The full results are presented in Appendix C-9 and summarised below in Table 6-55 and Table 6-56.

Table 6-55 Percentage frequency of correlation classification between ebtot (total eburnationscore)and facet angle

Sex	Sample	No Correlation (r _s <0.01)	Weak Correlation (r _s =0.1-0.29)	Medium Correlation (rs=0.3-0.49)	Strong Correlation (r _s =0.5-1.0)	Negative Correlation	P<0.001
Male	SBC	18	31	18	3	29	0
	SBL	25	33	7	0	24	0
	AS	21	40	15	14	42	0
Female	SBC	24	38	10	0	34	0
	SBL	19	21	13	1	28	0
	AS	13	30	12	15	48	0

Negative correlations were noted between facet angle and ebtot in 28 (29%) of male and 33 (34%) of female facets from St Bride's Crypt, 23 (24%) of males and 27 (28%) of female facets from St Bride's Lower and 40 (42%) of male and 46 (48%) of female facets from the Anglo-Saxon population. This indicates that in these facets increased ebtot was correlated with smaller facet angle. Although no correlations showed statistical significance, strong correlation was noted in the male cervical and lumbar spine from St Bride's Crypt, the female cervical spine from St Bride's Lower, and the male and female cervical and lumbar from the Anglo-Saxon sample. Table 6-56 summarises the results of the correlation analysis by spinal region. The count of facets without eburnation is identified in the "No Eburnation" column and is represented by blank cells in Appendix C-9. Figure 6-21 illustrates the results listed in Table 6-56. The results indicate that whilst the highest count of strong correlations can be seen in the cervical region for Anglo-Saxon males and females, the low levels and in some cases lack of correlation mean that there is a sample specific relationship between ebtot and facet angle.

Table 6-56 Percentage frequency of correlation classification between ebtot (total eburnationscore) and facet angle by spinal region

Spinal Region	Sex	Sample	No Eburnation	No Correlation (r _s <0.01)	Weak Correlation (r _s =0.1- 0.29)	Medium Correlation (rs=0.3-0.49)	Strong Correlation (r _s =0.5-1.0)
Cervical	Male	SBC	21	18	29	25	7
		SBL	3	25	46	25	0
		AS	0	14	39	29	18
	Female	SBC	21	14	57	7	0
		SBL	18	29	18	36	3
		AS	18	3	36	11	36
Thoracic	Male	SBC	44	21	27	10	0
		SBL	40	25	27	0	0
		AS	0	17	46	0	6
	Female	SBC	8	38	38	17	0
		SBL	35	19	21	0	0
		AS	0	17	46	0	6
Lumbar	Male	SBC	20	5	45	25	5
		SBL	45	25	30	0	0
		AS	0	25	40	30	5
	Female	SBC	75	10	15	0	0
		SBL	60	5	25	10	0
		AS	0	20	30	40	10





6.4.8.4 Correlation between pitting score and facet angle

The method and statistical analysis for correlation between pitting score and facet angle are the same as used for eburnation (see section 6.3.9.4), generating the new variable pitot. The full results are presented in Appendix C-10 and summarised below in Table 6-57 and Table 6-58.

able 6-57 Percentage frequency of correlation classification between pitot (total pitting s	score)
and facet angle	

Sex	Sample	No Correlation (r _s <0.01)	Weak Correlation (r _s =0.1-0.29)	Medium Correlation (rs=0.3-0.49)	Strong Correlation (r _s =0.5-1.0)	Negative Correlation	P<0.001
Male	SBC	38	34	25	3	46	0
	SBL	26	49	11	3	36	0
	AS	29	39	14	9	45	0
Female	SBC	35	52	11	1	53	0
	SBL	16	40	17	6	45	0
	AS	19	31	17	22	55	0

Negative correlations were noted between facet angle and extent of pitting in 44 (46%) of male and 51 (53%) of female facets from St Bride's Crypt, 35 (36%) of males and 43 (45%) of female facets from St Bride's Lower and 43 (45%) of male and 53 (55%) of female facets from the Anglo-Saxon sample. This indicates that in these facets increasing pitot was correlated with smaller facet angle. Although no correlations showed a statistical significance, strong correlations were noted in the male cervical and thoracic and female thoracic spine from St Bride's Crypt, the male thoracic spine and all regions in the female spine from St Bride's Lower, and all regions in the male and female spine from the Anglo-Saxon sample.

Table 6-58 summarises the results of the correlation analysis by spinal region. The number of facets without pitting is identified in the "No Pitting" column and represented by blank cells in Appendix C-10. Figure 6-22 illustrates the results presented in Table 6-58.

Table 6-58 Percentage frequency of correlation classifications between pitot (total pitting score)and facet angle by spinal region

Spinal Region	Sex	Sample	No Pitting	No Correlation (r₅<0.01)	Weak Correlation (r _s =0.1- 0.29)	Medium Correlation (rs=0.3-0.49)	Strong Correlation (r _s =0.5-1.0)
Cervical	Male	SBC	0	32	29	29	11
		SBL	0	14	61	25	0
		AS	0	39	39	21	3
	Female	SBC	0	32	57	11	0
		SBL	3	7	54	29	7
		AS	0	0	14	36	46
Thoracic	Male	SBC	0	57	71	39	3
		SBL	0	23	52	0	4
		AS	0	29	40	0	13
	Female	SBC	0	35	56	6	2
		SBL	15	21	35	0	2
		AS	2	21	44	0	6
Lumbar	Male	SBC	0	55	15	25	0
		SBL	5	50	25	20	0
		AS	0	30	25	30	15
	Female	SBC	0	40	35	35	0
		SBL	5	5	35	40	15
		AS	0	0	40	35	15

For St Bride's Crypt, all males and female facets have an overall measure of pitting in all regions. It can be seen that the association between pitot and facet angle is non-existent or weak in the majority of the facets with medium strength associations occurring in all spinal regions for males and females. Strong correlation between pitot and facet angle was seen in the right superior facet of C4, the left inferior facet of C5, the left inferior facet of C7 and the right superior facet of T11 in males and the left inferior facet of T9 in females.

For St Bride's Lower, females have a slightly higher number of facets without pitting in the cervical region (males 0, females 1), with a similar pattern seen in the thoracic region (males 0, females 7). In the lumbar region, males and females both have 1 facet without pitting. It can be seen that the association between pitot and facet angle is non-existent or weak in the majority of the facets with most medium strength associations occurring in the cervical and lumbar spine in males and females. Strong correlation between pitot and facet angle were seen at the right inferior facet of T4 and the left superior facet of T8 in males and the left inferior facet of C2 and C5, the right inferior facet of T1, the left superior facet of L3, the left inferior facet of L4 and L5 in females.

For the Anglo-Saxon sample, most regions have facets with pitting, the exception being in the thoracic region of the female spine (n=1). It can be seen that the association between pitot and facet angle is non-existent or weak in the majority of the facets with some medium strength associations occurring in the cervical and lumbar regions for males and females. Strong correlation between pitot and facet angle were seen at the left inferior facet of C7 the right superior facet of T1, the left inferior facet of T5 and T6, the right inferior facet of T9 and T11, the right superior facet of T12 and the left superior and right inferior facet of L5 in males and the left superior facet of C2, the left and right inferior facets of C2, the left and right inferior facets of C3, the right superior facet of C4, the left and right superior and right inferior facets of C7, the right superior and left and right inferior facets of C5, the left superior and left and right inferior facet of T6, the right superior facet of T12, the right superior facets of T5, the right superior facet of T6, the right superior facet of T12, the right superior facet of T12, the right superior facet of T12, the right superior facet of T5, the right superior facet of T6, the right superior facet of T12, the right superior and right inferior facet of T6, the right superior facet of T12, the right superior and right inferior facet of T12, the right superior and right inferior facet of T6, the right superior facet of T12, the right superior and right inferior facet of L3 in females.

The results indicate that the highest score for strong correlation classification between pitot and facet angle can be seen in the female Anglo-Saxon sample in the cervical region. The low level of correlation overall and lack of statistical significance implies that there is a poor relationship between pitot and facet angle.



Figure 6-22 Percentage frequency of strong correlations between pitot (total pitting score) and facet angle

6.4.8.5 Correlation between osteophyte score and facet angle

See Section 6.3.9.4 for details of data handling and statistical analysis. The new variable ostot was created. The full results are presented in Appendix C-11 and summarised below in Table 6-59 and Table 6-60.

Sex	Sample	No Correlation (r _s <0.01)	Weak Correlation (r _s =0.1-0.29)	Medium Correlation (r _s =0.3-0.49)	Strong Correlation (r _s =0.5-1.0)	Negative Correlation	P<0.001
Male	SBC	27	44	14	6	52	0
	SBL	25	48	11	1	34	0
	AS	21	29	10	5	26	0
Female	SBC	34	48	13	2	34	0
	SBL	27	36	8	4	40	0
	AS	15	34	6	23	48	0

Table 6-59 Count of correlation classifications between ostot (total osteophyte score) and facet angle

Negative correlations between facet angle and extent of osteophytes were noted in 50 (52%) of male and 33 (34%) of female facets from St Bride's Crypt, 33 (34%) of males and 38 (40%) of female facets from St Bride's Lower and 25 (26%) of male and 46 (48%) of female facets from the Anglo-Saxon sample. This indicates that in these facets increased ostot was correlated with smaller facet angle. Although no correlations showed a statistical significance, a strong correlation was noted in all regions of the male spine and in the cervical and thoracic regions of female spines

from St Bride's Crypt, the male thoracic spine and all regions in the female spine from St Bride's Lower, and all regions in the male and female spine from the Anglo-Saxon sample.

Table 6-60 summarises the results of the correlation analysis by spinal region. The number of facets without osteophytes is identified in the "No Osteophyte" column and are represented by blank cells in Appendix C-11. Figure 6-23 illustrates the results presented in Table 6-60.

Table 6-60 Percentage frequency of correlation between ostot (total osteophyte score) and
facet angle by spinal region

Spinal Region	Sex	Sample	No osteophytes	No Correlation (r₅<0.01)	Weak Correlation (r _s =0.1- 0.29)	Medium Correlation (rs=0.3-0.49)	Strong Correlation (rs=0.5-1.0)
Cervical	Male	SBC	14	25	46	11	3
		SBL	7	25	46	25	0
		AS	36	14	21	18	11
	Female	SBC	11	32	46	7	3
		SBL	21	29	29	18	3
		AS	36	3	11	7	43
Thoracic	Male	SBC	13	33	35	13	6
		SBL	17	27	27 44 0		2
		AS	33	21	29	0	4
	Female	SBC	0	38	52	8	2
		SBL	8	27	35	0	2
		AS	2	23	40	0	15
Lumbar	Male	SBC	0	15	60	15	10
		SBL	0	20	60	20	0
		AS	5	30	40	25	0
	Female	SBC	0	30	40	30	0
		SBL	0	25	50	15	10
		AS	0	10	55	20	15

For St Bride's Crypt, all males and female facets have an overall measure of osteophytes in the lumbar region but not in the cervical and thoracic regions. It can be seen that the association between ostot and facet angle is non-existent or weak in the majority of the facets with medium strength associations occurring in all spinal regions for males and females. Strong correlation between ostot and facet angle was seen in the right superior facet of C4, the right inferior facet of T4, the right inferior facet of T10, the right superior facet of T12, the right superior facet of L1, and the left superior facet of L2 in males and the left inferior facet of C3 and right superior facet of T5 in females.

For St Bride's Lower, females have more facets without osteophytes in the cervical region (males 2, females 6) and in the thoracic region (males 8, females 4). In the lumbar region all facets demonstrated osteophytes. It can be seen that the association ostot and facet angle is non-existent or weak in the majority of the facets with most medium strength associations occurring

in the cervical and lumbar spine in males and females. Strong correlation between ostot and facet angle were seen at the left inferior facet of T10 in males and right inferior facet of C4, T10, L2 and L5 in females.

For the Anglo-Saxon sample, 10 facets did not have osteophytes in the cervical region for both males and females. In the thoracic region, males had a much higher number of facets without osteophytes compared to females (males 16, females 1). In the lumbar region there was only 1 male facet without osteophytes. It can be seen that the association between ostot and facet angle is non-existent or weak in the majority of the facets with some medium strength associations occurring in the cervical and lumbar for males and females. Strong correlation between ostot and facet size were seen at the right inferior facet of C4, the left superior facet of C6, the right inferior facet of C7 and T5 and the left superior facet of C6 in males and the left and right superior facets of C3, the right superior and left and right inferior facet of C6, the left and right superior and right inferior facets of C5, the left superior facet of C6, the left and right superior and right inferior facets of C7, the tight superior facet of C2, the right inferior facet of T6, the left and right superior and left and right inferior facet of T2, the right inferior facet of T6, the left inferior facet of T9, the left superior and right inferior facets of T10, the right superior facet of T11, the right inferior facet of T12 and L2, the right superior facets of L4 and L5 in females.

The highest number of strong correlations can be seen in the cervical and thoracic regions of Anglo-Saxon females. The overall count of strong correlations is low indicating a poor relationship between ostot and facet angle.



Figure 6-23 Percentage frequency of strong correlations between ostot (total osteophyte score) and facet angle

6.4.8.6 Correlation between vertebral osteophytosis score and facet angle

See Section 6.3.9.7 for details of data management and statistical analysis. The full results are presented in Appendix C-12 and summarised below in Table 6-61 and Table 6-62.

Table 6-61 Percentage frequency of correlation classification between cvostot (total vertebralbody osteophytosis score) and facet angle

Sex	Sample	No Correlation (rs<0.01)	Weak Correlation (r _s =0.1-0.29)	Medium Correlation (r _s =0.3-0.49)	Strong Correlation (r _s =0.5-1.0)	Negative Correlation	P<0.001
Male	SBC	47	35	11	2	44	0
	SBL	44	35	13	4	45	0
	AS	24	19	28	17	40	0
Female	SBC	60	33	2	0	38	0
	SBL	33	33	13	3	43	0
	AS	19	22	15	11	24	0

Negative correlations between facet angle and vertebral osteophytosis were noted in 42 (44%) of male and 36 (38%) of female facets from St Bride's Crypt, 42 (45%) of males and 41 (43%) of female facets from St Bride's Lower and 38 (40%) of male and 23 (24%) of female facets from the Anglo-Saxon sample. This indicates that in these facets increased cvostot was correlated with

smaller facet angle. Although no correlations showed a statistical significance, strong correlations were noted in all regions of the male spine and in the cervical and thoracic regions of female spines from St Bride's Crypt, the male thoracic and lumbar spine and lumbar region in the female spine from St Bride's Lower, and all regions in the male and female spine from the Anglo-Saxon sample, apart from the female cervical where no analysis could be performed due to missing vertebral bodies.

Table 6-62 summarises the results of the correlation analysis by spinal region. Figure 6-24 illustrates the results reported in Table 6-62. The full results can be seen in Appendix C-12. See section 6.3.9.7 for details of recording of missing data within this table.

Spinal Region	Sex	Sample	No vertebral bodies	No Correlation (rs<0.01)	Weak Correlation (r _s =0.1- 0.29)	Medium Correlation (rs=0.3-0.49)	Strong Correlation (r _s =0.5-1.0)
Cervical	Male	SBC	0	32	43	7	6
		SBL	0	54	25	11	0
		AS	2	0	21	21	11
	Female	SBC	0	57	29	3	0
		SBL	0	36	43	11	0
		AS	6				
Thoracic	Male	SBC	SBC 0 58 35		35	6	0
		SBL	0	50	42	6	2
		AS	0	33	17	38	15
	Female	SBC	0	63	35	2	0
		SBL	0	40	56	6	6
		AS	1	19	35	23	15
Lumbar	Male	SBC	0	40	25	30	0
		SBL	0	15	35	30	15
		AS	0	35	20	15	30
	Female	SBC	0	60	35	0	0
		SBL	5	5	15	30	15
		AS	0	45	20	15	20

 Table 6-62 Percentage frequency of correlations between cvostot (total score vertebral osteophytosis) and facet angle by spinal region

For St Bride's Crypt, all males and female spinal regions have a sufficient number of intact vertebral bodies for analyses to be performed for all areas. It can be seen that the association between cvostot and facet angle is non-existent or weak in the majority of the facets with

medium strength associations occurring in all spinal regions for males and females. Strong correlation between cvostot and facet angle was seen in the right inferior facet of C5 and right superior facet of C6 in males and no facets in females.

For St Bride's Lower, all males and female spinal regions with the exception of Female L4 have a sufficient number of intact vertebral bodies for analyses to be performed for all areas. It can be seen that the association between cvostot and facet angle is non-existent or weak in the majority of the facets with most medium strength associations occurring in the cervical and lumbar spine in males and females. Strong correlation between cvostot and facet angle were seen at the left inferior facet of T10, the right superior and right inferior facets of L3 and the left inferior facet of L4 in males and the left inferior facet of L2, the right superior and right inferior facets of L3 in females.

For the Anglo-Saxon sample, vertebral bodies were absent at C2 and C4 in males and C2, C3, C4, C5, C6, C7 and T1 in females. It can be seen that the association between cvostot and facet angle is non-existent or weak in the majority of the facets with some medium strength associations occurring in all spinal regions for males and females. However, in this population, much higher numbers of strong correlations were observed despite the amount of missing data. Strong correlation between cvostot and facet angle was seen at the right and left inferior facets of C5, the left superior facet of C7, the left superior facet of T1, the left and right inferior facets of T2, the right superior facet of T4, the left inferior facets of T6 and T9, the lefty and right inferior facets of T2, the right inferior facet of T5, the left superior and right inferior facets of T2, the left superior facet of T5, the left superior and right inferior facets of T2, the left superior facet of T1, the left and right inferior facets of T2, the left superior facet of T5, the left superior and right inferior facets of T2, the left superior facet of T5, the left superior and right inferior facets of T3, the left superior facet of T4, the left inferior facet of T4, the left inferior facets of T6, the left superior facet of T5, the left superior and right inferior facets of T2, the right inferior facet of T5, the left superior facet of T4, the left inferior facet of T6, the left superior facet of T5, the left superior and right inferior facets of T6, the left inferior facet of T3, the left inferior facet of T4, the left inferior facet of T4, the left inferior facet of L1 and the left superior facet of L3 and the left superior and right inferior facets of L4 in females.



Figure 6-24 Percentage frequency of strong correlations between cvostot (total vertebral body osteophytosis score and facet angle)

The highest count of strong correlations can be seen in the Anglo-Saxon sample (thoracic and lumbar regions). The lack intact vertebral bodies in the female Anglo-Saxon sample meant that no conclusions could be drawn from analysis of the cervical region.

6.5 Stepwise discriminant function with extrinsic variables

Stepwise discriminant function equations were calculated for each vertebral level using extrinsic variables, facet size and angle. The functions are presented at the end of this section. Finally, the percentage accuracy of correct classification of sex using functions derived from facet size and facet angle alone was compared with those for functions including predictor variables to measure the effectiveness of considering lifestyle markers in predicting sex.

6.5.1 Derivation of discriminant function equations for all parameters

Facet size, angle and all extrinsic variables (femoral robusticity, humeral directional asymmetry, eburnation, pitting, osteophytes and osteophytosis) were subjected to multivariate discriminant function analysis to create equations using stepwise selection to identify the combination of variables that best discriminate between males and females. The method is described in Section 5.2.9.2 and the results can be seen in Table 6-63.

The p value from the Wilk's Lambda test for goodness of fit for the model tests for significant differences between male and female groups. In this analysis p was set at p<0.0006 with Bonferroni adjustment (significant results are written in bold). Results for all vertebral levels for each sample are included in the table. This analysis generated discriminant functions for each vertebra that identify the sex of an individual based on the size and angle of all four facets for each vertebral level and extrinsic variables relevant to the facets. In the event of little or no difference between sexes, no variables qualified for inclusion in the equation.

The total percentage of correct classification listed in this table was used to compare accuracy in using extrinsic variables rather than facet size or angle alone and the results presented in this section.

Key for Table 6-63:

C = cervical,

T = thoracic,

L = lumbar

Numbers represent the vertebral level

Lt = left,

R = right,

S = superior,

I = inferior

E.g. C1LtSA = facet size for C1, left superior facet and C1LS = facet angle for C1 left superior facet.

ebtot = eburnation severity and extent added together

pitot = pitting severity and extent added together

ostot = osteophyte severity and extent added together

FemrobustL = femoral robusticity of left femur

FemrobustR = femoral robusticity of right femur

Hasym = humeral asymmetry

Vertebral	Sample	Wilks	Discriminant Function Equation F(n)	Zm	Zf	Nm	Nf	Z0	9	6 Correct Classi	fication
Level		Lambda p							М	F	Total
C1	SBC	0.001	F=(-0.384)+(0.029)C1LtIArea+(1.488)C1LtIpitot-(2.11)C1LtS	1.926	-1.051	10	29	0.44	44	91	71
	SBL	0.004	F=(-7.717)-(0.28)C1LtlArea+(1.731)C1tLlebtot+(0.508)C1LtS	-0.83	1.661	21	9	0.42	81	45	65
	AS	0.012	F=(-67.953)-(13.356)C1LtIpitot+(4.416)C1LS+(2.031)C1RIpitot	-14.97	59.879	8	4	22.45	53	50	52
C2	SBC	<0.001	F=(7.009) –(0.36)C2LtSArea+(0.692)C2LtSebtot+(0.032)Hasym	-1.833	1.166	10	20	-0.33	63	71	68
	SBL	0.002	F=(0.029)+(0.02)C2LtSArea-(0.084)C2RI	0.589	-1.474	26	11	-0.44	90	55	76
	AS	0.16	F=(-33.155)+(0.471)C2RI	2.907	-5.814	11	5	-1.45	85	56	73
C3	SBC	0.025	F=(-3.836)+(0.037)C3RSArea	0.809	-0.486	7	26	0.16	35	81	64
	SBL	0.001	F=(-13.74)+(0.228)C3LtI+(1.01)C3LtIpitot-(0.044)C3LSArea- (0.71)C3LtSostot	-0.971	1.943	20	14	0.49	74	67	71
	AS	0.298	F=(-5.103)+(0.061)C3LtI	1.138	-2.275	8	1	-0.57	100	11	68
C4	SBC	0.055	F=(-0.839)+(0.186)FemrobustLt	-0.593	0.461	11	7	-0.07	85	30	50
	SBL	0.001	F=(4.55)-(0.054)C4LtIArea-(0.77)C4LtSostot- (1.17)C4Rlostot)+(1.19)C4RSostot	-0.861	1.844	24	11	0.49	83	52	70
	AS	0.181	F=(-22.062)+(0.283)C4LtS	-2.404	4.808	5	7	1.20	36	78	52

Table 6-63 Stepwise discriminant function analysis with extrinsic variables

Table 6-63 Stepwise discriminant function analysis with extrinsic variables continued

		Wilks							% Co	orrect Classific	ation
Vertebral	Sampla	Lambda	Discriminant Eurotion Equation E(n)	Zm	7f	Nim	NIF	70	NA	E	Total
Levei	Sample	μ		2111	Z1	INIII	INI	20	IVI	Г	Total
C5	SBC	0.007	F=(-4.557)+(0.043)C5LtSArea	0.801	-0.712	12	23	0.04	52	68	61
	SBL	0.029	F=(-0.419)+(1.316)C5Rlostot	-0.419	0.605	24	6	0.09	83	25	57
	AS	0.095	F=(-47.412)+(0.453)C5RS+(0.537)FemrobustR	-4.293	12.878	8	2	4.29	100	29	67
C6	SBC		No variables qualify for this equation								
	SBL		No variables qualify for this equation								
	AS	0.006	F=(136.938)-(4.07)C6LtI+(1.32)C6LtIArea+(0.612)C6RI	24.16	-96.651	9	5	-36.25	75	63	70
C7	SBC	0.016	F=(-4.999)+(0.041)C7LtlArea	1.006	-0.559	10	35	0.22	40	90	70
	SBL	0.008	F=(-1.689)-(0.034)C7LtSArea+(0.572)C7RIpitot+(0.046)C7RSArea	0.579	-1.094	22	13	-0.26	73	54	65
	AS		No variables qualify for this equation								
T1	SBC		No variables qualify for this equation								
	SBL	0.028	F=(-0.455)+(1.138)T1RIpitot	-0.321	0.683	30	8	0.18	86	35	66
	AS	0.197	F=(-55.154)+(0.471)T1	-123	4.243	13	2	-59.38	93	18	60
T2	SBC	0.003	F=(-6.597)+(0.065)T2RIArea	1.141	-0.685	14	31	0.23	54	82	70
	SBL	0.006	F=(2.785)-(0.3)T2LIArea+(1.149)T2RSostot	-0.575	1.078	30	4	0.25	88	18	61
	AS	0.013	F=(-55.973)+(0.55)Hasym+(28.136)T2Ltlebtot	55.256	-55.256	4	3	0.00	36	100	50

Table 6-63 Stepwise discriminant function analysis with predictor variables continued

		Wilks							% Co	rrect Classific	ation
Vertebral Level	Sample	Lambda p	Discriminant Function Equation F(n)	Zm	Zf	Nm	Nf	ZO	М	F	Total
Т3	SBC	<0.001	F=(-56.2)+(-0.19)Hasym-(0.42)T3Ll+(0.19)T3LlArea+(3.08)T3Ltlebtot	8.137	-4.882	11	22	1.63	69	73	72
			+T3LtSebtot+(0.75)T3RI+(13.2)T3RSebtot-(3.424)T3RSpitot								
	SBL		No variables qualify for this equation								
	AS	0.051	F=(-752.247)+(7.211)T3LtI-(4.443)T3RSpitot	13.7	-13.73	10	4	-0.02	100	44	74
T4	SBC	0.001	F=(-15.188)+(1.954)T4Llebtot-(0.027)T4LtSarea+(0.144)T4Rlarea	2.398	-2.877	16	19	-0.24	70	76	73
	SBL	<0.001	F=(-13.3)+(1.28)T4LtI+(8.33)T4LtIpitot-(1.29)T4LtS)- (0.23)T4LtSArea+(25.52)T4tLSostot+(1.64)T4RI-(1.537)T4RSArea	-15.032	40.085	26	8	12.53	93	44	74
	AS	0.074	F=(-82.779)-(0.875)T4RSArea	19.991	-9.995	14	11	5.00	61	69	64
Т5	SBC	0.002	F=(-4.403)+(0.048)T5RSArea-(0.757)T5RSebtot	1.335	-0.935	13	31	0.20	52	84	71
	SBL	0.024	F=(-0.427)+(1.708)T5LStostot	-0.427	0.939	22	5	0.26	79	25	56
	AS	0.018	F=(-117.66)+(10.34)T5LtSostot+(0.876)T5RSArea+(5.549)T5RSostot	26.741	-17.827	8	10	4.46	36	71	50
Т6	SBC	<0.001	F=(-113.77)+(1.23)T6LS-(0.119)T6LtSArea-(0.07)T6RIArea+T6RSebtot	-4.755	3.17	16	10	-0.79	62	77	71
	SBL		No variables qualify for this equation								
	AS	0.024	F=(-192.359)+(1.753)T6LtSArea+(15.984)T6LtSostot	51.69	-17.23	3	12	17.23	18	100	52
Т7	SBC	0.001	F=(12.37)+(0.056)T7LtSArea-(0.159)T7RI	1.461	-0.93	19	25	0.27	76	66	70
	SBL	0.001	F=(11.481)+(-0.421)T7RI+(0.194)T7LtS+(0.114)T7RI	0.908	-1.686	26	11	-0.39	93	55	77
	AS	0.1	F=(-42.38)+(-0.782)T7LtSArea+(3.216)T7LtSpitot+(4.991)T7RSArea	27.262	-41.194	5	9	-6.97	45	75	61

Table 6-63 Stepwise discriminant function analysis with predictor variables continued

		Wilks							% Cc	orrect Classific	ation
Vertebral	Sample	Lambda	Discriminant Eulertion Equation $F(n)$	Zm	7f	Nm	Nf	70	М	E	Total
Level	Jampie	Ρ		2111	21	INITI	INI	20	IVI	I	TOtal
Т8	SBC	0.003	F=(-15.549)+(0.158)FemrobustLt-(0.105)T8tLlArea+(0.221)T8LtS	-1.744	1.047	11	15	-0.35	92	79	83
	SBL	0.009	F=(-5.203)+(0.064)T8LtSArea	0.356	-1.009	32	9	-0.33	97	45	77
	AS		No variables qualify for this equation								
Т9	SBC	0.001	F=(-22.055)+(0.051)Hasym+(0.275T9LtS)-(0.069)T9RSArea	-0.001	1.072	10	25	0.54	56	83	73
	SBL	0.004	F=(-4.687)+(0.057)T9LtSArea	0.461	-0.923	26	11	-0.23	84	55	73
	AS	<0.001	F=(-107.503)-(1.876)T9LtSArea+(10.075)T9LtSebtot+(2.456)T9RSArea	-32.685	32.685	16	4	0.00	89	25	59
T10	SBC	0.002	F=(-11.733)+(0.067)T10tLI+(0.047)T10LtIArea+(0.7)T10LtSostot	1.634	-1.144	18	31	0.25	72	80	77
	SBL	0.011	F=(-4.052)+(0.044)T10RIArea	0.403	-0.806	27	9	-0.20	90	45	72
	AS	0.039	F=(-1.501)+(0.584)T10pitot	1.223	-0.918	9	7	0.15	45	78	55
T11	SBC	0.01	F=(0.038)+(0.041)T11RSArea	0.785	-0.571	8	36	0.11	31	92	68
	SBL	0.015	F=(-4.093)+(0.038)T11tLSArea	0.381	-0.762	25	8	-0.19	89	40	69
	AS	0.001	F=(-226.699)+(19.956)T11LtSostot+(1.546)T11RSArea	18.329	-27.494	10	10	-4.58	63	83	71
T12	SBC	0.008	F=(-2.371)+(-0.177)FemrobustLt+(0.028)T12LtIArea	1.004	-0.73	8	17	0.14	62	90	78
	SBL		No variables qualify for this equation								
	AS	0.007	F=(-75.904)+((0.327)T12LtIArea+(13.189)T12LtIebtot+(1.947)T12RSebtot	51.734	-34.489	3	8	8.62	21	80	45

Table 6-63 Stepwise discriminant function analysis with	predictor variables continued
---	-------------------------------

Vertebral	Sample	Wilks	Discriminant Function Equation F(n)	Zm	Zf	Nm	Nf	ZO	% Correct Classification		ation
Level		Lambda p					М	F	Total		
L1	SBC	0.004	F=(-3.527)+(0.135)FemrobustLt+(0.112)L1RS-(0.653)L1RSpitot	-1.28	0.931	10	18	-0.17	83	90	88
	SBL	0.011	F=(-3.667)+(0.093)t+(0.392)L1RSostot	-0.497	0.994	23	5	0.25	82	25	58
	AS	0.015	F=(-212.902)+(3.251)L1LtS+(0.749)L1RIArea+(23.164)L1VOS	20.255	-30.383	7	5	-5.06	70	63	67
L2	SBC	0.003	F=(-4.152)-(0.177)FemrobusttL+(0.03)L2RIArea	1.141	-0.913	9	17	0.11	69	85	79
	SBL	0.062	F=(-2.594)+(0.015)L2LtSArea	0.3	-0.563	26	6	-0.13	90	31	65
	AS	0.001	F=(-14.976)+(0.429)L2LtI+(1.229)L2Rlosot	-3.074	3.074	11	8	0.00	69	80	73
L3	SBC	0.008	F=(-5.937)-(0.048)Hasym+(0.035)L3RIArea		-0.953	6	25	0.16	33	89	67
	SBL		No variables qualify for this equation								
	AS	0.006	F=(-303.28)-(0.17)L3LtlArea- (6.5)L3Ltlpitot+(23.37)L3Ltlpitot+(0.92)L3LtSArea	29.716	-29.716	3	14	0.00	18	88	52
L4	SBC	0.001	F=(-4.619)-(0.147)FemrobustLt+(0.031)L4LtIArea	1.342	-1.073	10	18	0.13	77	86	82
	SBL	0.002	F=(-56.042)+(1.1)Hasym-(1.366)L4LS+(0.777)L4LtSArea- (16.392)L4LtSostot	35.826	-71.652	16	7	-17.91	89	54	74
	AS		No variables qualify for this equation								
L5	SBC	0.024	F=(-1.1877)-(0.195)FemrobustLt+(0.012)L5LtlArea	0.913	-0.913	9	12	0.00	75	67	70
	SBL	0.053	F=(-6.459)+(0.123)L5LtI	-0.285	0.713	23	4	0.21	82	21	57
	AS		No variables qualify for this equation								

6.5.2 Comparison of percentage correct classification using stepwise discriminant function analysis (DFA) with and without predictor variables

The percentage of correct classification results for stepwise discriminant function analysis using facet size, facet angle alone and predictor variables were compared to see if the use of predictor variables increased the degree of precision in sex identification. The results are shown in Table 6-64. In some circumstances there were no suitable variables for calculation of a discriminant function equation and therefore the percentage accuracy could not be calculated (marked as a blank cell). Discriminant function equations could not be created in 21/72 (29%) of analyses for facet size only, 55/72 (76%) of analyses for facet angle only and 11/72 (15%) of analyses using all predictor variables. The highest percentage correct classification for each sample and vertebral level is highlighted for clarity.

Table 6-64 Comparison of percentage of correct classification of sex identification using stepwise DFA

Vertebral Level	Sample	% Accuracy Facet Size	% Accuracy Facet Angle	% Accuracy All Predictor Variables
C1	SBC	71	66	71
	SBL	80		65
	AS		78	52
C2	SBC	67		68
	SBL	84	74	76
	AS		71	73
С3	SBC	71		64
	SBL	65	67	71
	AS	65	83	68
C4	SBC	66		50
	SBL	65		70
	AS			52
C5	SBC			61
	SBL			57
	AS		75	67
C6	SBC	72		
	SBL		63	
	AS			70
C7	SBC	67		70
	SBL	71		65
	AS			

T1	SBC	69		
	SBL			66
	AS			60
T2	SBC	72	70	70
	SBL	64		61
	AS			50
Т3	SBC	67		72
	SBL	78		
	AS	79		74
T4	SBC	76		73
	SBL	80		74
	AS	72	67	64
T5	SBC	76		71
	SBL			56
	AS	71		50
Т6	SBC	77		71
	SBL	74		
	AS	68		52
T7	SBC	70		70
	SBL	82		77
	AS			61
Т8	SBC	67		83
	SBL	77	66	77
	AS	59		
Т9	SBC	78		73
	SBL	75		73
	AS	67	79	59

Table 6-64 Comparison of percentage of correct classification of sex identification usingstepwise DFA continued

Table 6-64 Comparison of percentage of correct classification of sex identification using

stepwise	DFA	continued
----------	-----	-----------

Vertebral Level	Sample	% Accuracy Facet Size	% Accuracy Facet Angle	% Accuracy All Predictor Variables
T10	SBC	74		77
	SBL	72		72
	AS			55
T11	SBC	67	64	68
	SBL	73		69
	AS	61		71
T12	SBC	64	68	78
	SBL	71		
	AS			45
L1	SBC	63	69	88
	SBL	69		58
	AS			67
L2	SBC	71		79
	SBL	71		65
	AS			73
L3	SBC	66		67
	SBL	77		
	AS		73	52
L4	SBC	69		82
	SBL	78		74
	AS		80	
L5	SBC	73		70
	SBL	71		57
	AS			

Table 6-65 presents a summary of the highest percentages of correct classification highlighted in the previous table. Where there are equally high percentages of correct classification in two

results, they were counted twice (e.g. vertebral level C1 for St Bride's Crypt where both % accuracy for facet size and all predictors has the same value of 71%). Facet angle alone is generally a poor predictor for sex difference for St Brides Crypt and St Bride's Lower, however this is less evident in the Anglo-Saxon sample The stepwise discriminant function analysis using all predictor variables gave the highest total levels of correct classification for St Brides Crypt (14/72) and Anglo-Saxon (11/70, taking missing data into account) whereas facet size gave the highest total levels of correct (18/72). Facet size is an important factor in sexual dimorphism particularly in the thoracic region of all three samples.

Sample	Spinal Region	% Size	% Angle	% All	Missing data
	Cervical	4		4	
SBC	Thoracic	7		6	
	Lumbar	1		4	
	Total	12	0	14	
	Cervical	3	1	3	
SBL	Thoracic	10		4	
	Lumbar	5		1	
	Total	18	1	8	
AS	Cervical		3	3	1
	Thoracic	5	1	6	
	Lumbar		2	2	1
	Total	5	6	11	2

Table 6-65 Summary of highest percentages for correct classification

Chapter 7 Discussion

This study is divided into two parts. The first section used statistical methods to estimate sex in three British samples. Discriminant functions derived from measurements of facet size and angle for the whole spine were used in this process. The second part explored the relationship between extrinsic factors and facet morphology using a comparative approach; the relationship between facet morphology and gendered division of labour was explored in the three samples in response to Armelagos et al. (1982: 318-9) comment:

"Although a methodology for interpreting the functional significance of sexual dimorphism [in postcranial dimensions] could have been developed, it was not. Physical anthropologists instead created statistical techniques for determining sex without concern for the functional factors that lead to these differences"

7.1 Summary of Results – Sexual Dimorphism

The analysis of samples from rural and urban settings described in Chapter 5 has identified a number of interesting findings in relation to the use of facet size and to a lesser degree facet angle in the determination of sex. The main findings for each skeletal sample will be outlined in this section followed by a discussion of the over-arching trends relating to statistical analysis for determination of sex using the facet joints of the spine.

Sexual Dimorphism

Sexual dimorphism can be defined as "systematic difference in form between individuals of different sex in the same species" (Plavcan and van Schaik, 1994). The metric differences between the sexes can be quantified using a ratio. The SDI method used for this study provides results that indicate a feature is similar in males and females when SDI value is close to 100, a value > 110 indicates that male measurements are higher than females conversely, a value < 90 indicates that the female measurement is higher.

Sexual dimorphism has been much studied in humans and nonhuman primates, with males being on average larger than females in general body size and other skeletal parts (Stini 1974; Ross and Ward 1982; Clutton-Brock 1985; Frayer and Wolpoff, 1985; Leutenegger and Cheveraud, 1985; Ruff 1987, 2000; Plavcan and van Schaik 1994; Spradley and Jantz, 2011).

Chapter 7 Discussion

The male larger than female pattern has also been observed in human vertebral measurements (Taylor and Twomey, 1984; MacLaughlin and Oldale, 1992; Marino 1995; Grave et al., 1999; Wescott 2000; Bonivitch et al., 2006; Rühli et al., 2006; Yu et al 2008; Marlow and Pastor 2011; Zheng et al 2012; Hou et al., 2012; Amores et al 2013; Bethard and Seet, 2013; Bastir et al., 2014).

Evaluation of Sexual Dimorphism

This study has demonstrated a difference in the degree of sexual dimorphism of facet size between samples, with St Bride's Crypt being the most sexually dimorphic and the Anglo-Saxon the least for facet size. Facet angle does not show the same degree of variation. Milne (1991) obtained similar results for facet angle in the cervical spine, being unable to identify any significant differences between male and female facet angle. Differences in adult vertebral dimensions reflect differences in bone growth that evolves during early skeletal development, and so changes in bone size during childhood have important biomechanical implications in respect to loading capacity of the skeleton in adulthood (Bonjour et al., 1991; Einhorn 1992). The gendered division of labour can also affect sexual dimorphism because bones respond to external force and remodel according to the loading on them (Ruff, 1987; Steyn and Işcan, 1999).

Historically, there have been a number of studies of sexual dimorphism in vertebrae with varying degrees of accuracy for estimation of sex (from 60-91%) when using multivariable discriminant function equations (Schaffler et al., 1992; MacLauglin and Oldale, 1992; Marino, 1993; Haughen, 1994; Wescott, 2000; Marlow and Pastor, 2011; Swenson, 2013; Bethard and Seet, 2013; Amores et al., 2014; Bastir, 2014). Many of these studies have limitations as a result of data being collected from single vertebral levels or from a spinal region, using unilateral measurements, employing different methodologies to derive discriminant function equations, using only single samples and/or changes due to age not being taken into consideration. None of the studies reviewed had collected data from the whole spine. Most of the studies justified their choice of vertebra by selecting those with easily recognisable morphology (Marino, 1995; Wescott, 2000; Yu et al., 2008; Amores et al., 2014). Other studies were replications of existing studies with no further justification for use of a particular vertebra (Marlow and Pastor, 2011; Bethard and Seet, 2013). Finally, some studies gave no rationale for the choice of vertebra (Hou et al., 2012; Zheng et al., 2012). The majority of published facet angle studies are clinical in nature, rather than archaeological, and used male spines for data

collection (Pal and Routal, 2001; Masharawi et al., 2007). A few studies, however, did identify differences in sagittal angle between sexes when looking at spinal regions rather than individual vertebrae (Tulse and Hermanis, 1993; Boyle et al., 1996; Masharawi et al., 2004; Whitcombe et al., 2007), but Milne (1991) found no sex differences in his analysis of upper cervical spine facet angles. None of these studies considered using sagittal angle as a method to estimate the sex of an individual. Discriminant function equations were not calculated from the data collected in any of these studies.

The current study examined the zygapophyseal joints of the whole spine in order to assess whether all vertebrae were equally dimorphic and to give a more complete picture of the effects of lifestyle on the spine. The zygapophyseal (facet) joints of the spine were selected for measurement in the present study because, although there is existing evidence for sexual dimorphism in vertebrae based on multiple measurements (Marino, 1995; Wescott, 2000; Yu et al., 2008; Marlow and Pastor, 2011; Bethard and Seet, 2013 Hou et al., 2012; Zheng et al., 2012; Amores et al., 2014), the facet joint has not been a point of focus. The facet joints are considered susceptible to degenerative changes caused by activity due to the direct influence of load bearing (Moore and Petty, 2005). This could make them a suitable candidate for investigating the effect of extrinsic influences. It was not possible to make direct comparison between actual facet joint dimensions from this study with other published research due to the variation in methods used to collect data. This results in a wide range of reported osteometric values. Furthermore size sexual dimorphism is also known to depend on the genetic make up and environment of individuals and thus is therefore population specific (Cheveraud, 1988).

St Bride's Crypt

Metric analysis of St Bride's Crypt facet joint size found that the degree of difference between the sexes appeared to be high, with many facets (85%) showing a sexual dimorphism index (SDI) of >110. All spinal regions exhibited a high percentage of facets with SDI > 110 (cervical 79%, thoracic 85%, lumbar, 95%). Only 6/28 cervical, 7/48 thoracic and 1/20 lumbar facets were not classifiable for sex due to a lack of size difference between males and females. Male mean facet size was greater than that of females for most facets.

A clear pattern of facet size sexual dimorphism was confirmed in this sample after statistical analysis using Mann-Whitney U test. The thoracic region demonstrated the highest overall

Chapter 7 Discussion

number of facets showing significant difference in facet size between males and females, followed by the lumbar then cervical region.

Facet angle analysis produced a different pattern than that seen for facet size in this sample. There was much lower sexual dimorphism seen, with only one cervical facet having SDI > 110. However, index values of < 90 were seen in 5 thoracic and 8 lumbar facets. Regionally, 27 cervical, 43 thoracic and 12 lumbar facets could not be classified from their index values indicating a lack of sexual dimorphism for facet angle. The diminished degree of sexual dimorphism in facet angle was confirmed by statistical analysis with only one thoracic facet showing significant difference in facet angle between males and females. There is thus little sexual dimorphism in facet angle indicating that this parameter would be a poor indicator of sex in this sample.

St Bride's Lower

The two St Bride's samples show similarity in the degree of sexual size difference in facet joint size, with both having 85% of facets with an SDI > 110. The distribution of facets with an SDI > 110 (cervical 68%, thoracic, 92%, lumbar, 95% for St Bride's Lower) differed from the St Bride's Crypt sample.

Fewer facet size sex differences showed statistical significance than in the St Bride's Crypt sample, again, the highest number of significant differences was seen in the thoracic region, followed by the cervical and lumbar regions. There are fewer significant facet size differences than seen in the Crypt sample indicating that the degree of sexual dimorphism for facet size is lower in the St Bride's Lower sample. This may be an indication that these individuals had a less gendered division of labour or division of labour but equal degrees of physical activity throughout their lives than the St Bride's Crypt group thereby resulting in reduced sexual dimorphism in facet size.

Facet angle analysis revealed that there was no statistically significant difference in facet angle between the sexes. This lack of sexual dimorphism was confirmed using SDI values. No SDI values were >110 and only one facet from each spinal region demonstrated a mean value of <90. Ninety-five percent of facets (27 cervical, 47 thoracic, and 19 lumbar) did not demonstrate sexual dimorphism, hence making this a poor indicator of sex.
Anglo-Saxon

The previous samples are of urban origin from the same time period but of contrasting socioeconomic status. The Anglo- Saxon sample was included in this study to provide a contrasting rural sample from a different time period. The difference in lifestyle of this group is expected to display a difference in facet size and angle due to the differing degrees of activity and gendered division of labour undertaken by this group.

Calculation of SDI for this sample showed a lower degree of sexual dimorphism for facet size when compared to the two St Bride's samples, with only 47% of facets having an SDI>110 and 1% of facets <90. This meant that over half (52%) of the facets did not show size difference between the sexes. Statistical analysis indicated significant difference in facet size for just two facets in the thoracic region. Differences in facet size between males and females are a poor indicator of sex for this sample.

Facet angle showed a greater degree of sexual dimorphism than the other two samples when SDI was calculated. SDI>110 was observed in 13% of facets (10% of this value being observed in the cervical region) and < 90 in 13% of facets (13% of this value being observed in the lumbar region), however, only one lumbar value showed statistical significance indicating that facet angle is a poor indicator of sex for this sample. The facet angle results from the lumbar region may be due to the deepening of the lumbar lordosis in females during pregnancy. This acts as a counterbalance to the growing weight of the foetus (Whitcombe et al., 2007). Whitcombe et al. (2007) found that the females present with a longer series of dorsally wedged lumbar vertebrae and lumbar facets that are more frontally orientated than males and consider this to be a physiological adaptation increasing resistance to forward displacement of the vertebrae. Masharawi et al. (2004) were unable to confirm this finding, suggesting that female adaptation to pregnancy stems from combination of a smaller kyphotic vertebral body wedging into the thoracic and upper lumbar vertebrae, the relatively greater interspinous spacing and the larger interfacet width in the lumbar spine. This contradiction in results may be the result of using different methods (calliper v digitiser) and/or because the data was collected from different samples. Hay et al. (2015) urge caution with interpretation of Whitcombe et al. (2007) and Marashawi et al. (2004) studies as it is not clear whether the bone modifications described in them equal real differences in lordosis between the sexes.

This study has identified variation in degree of vertebral facet orientation in the lumbar region in males and females with the St Bride's Lower and the Anglo-Saxon sample matching the

Whitcombe et al. (2007) results and the St Bride's Crypt sample matching the Masharawi et al. (2004) result. These conflicting outcomes represent an area for further research. For the majority of facet comparisons, particularly in the upper and mid-thoracics, there was no clear differential between males' and females' mean facet angle and thus facet angle is a poor indicator of sex.

Discriminant Functions

The use of discriminant functions for classification of sex is a much-used practice in bioarchaeology (Schwartz, 2006). The accuracy of any discriminant function (DF) relies on similarity between the individual/group being tested and the sample from which the function is derived. The strength of sexual dimorphism displayed at population level is thus of great importance. Some populations display a greater degree of dimorphism than others. The existing vertebral discriminant functions are population specific and, when used in other ecogeographic or time contexts, may provide misleading results, with accuracies lower than that reported in the reference sample (Kotěrová et al., 2017; Hora and Sládek, 2018). Kotěrová et al. (2017) tested the effect of ignoring population specificity when using DF equations for tibial measurements from Portuguese, European and North American populations on measurements from a Czech sample and found that correct sex assessment was significantly decreased. Earlier studies have reported population differences in vertebral dimensions (Pastor, 2005; Yu et al., 2008; Marlow and Pastor, 2011).

To control for this factor, stepwise discriminant functions were calculated individually for each sample as it is not clear to what degree temporal or socioeconomic factors affect the degree of accuracy. Garvin et al. (2014) recommend that stepwise discriminant function should be utilised when estimating sex if the weighting of the factors under study is unknown. Methods of sex assessment using stepwise discriminant function equations are generally considered useful if the degree of accuracy is at least 80% (Rogers, 1999). In this study, the degree of accuracy achieved for facet size ranged from 64% -77% for St Bride's Crypt, 64-84% for St Bride's Lower and 59-79% for the Anglo-Saxon sample. Similarly, the degree of accuracy for facet angle ranged from 64-70% for St Bride's Crypt, 63-74% for St Bride's Lower and 67-83% for the Anglo-Saxon sample. Missing data or lack of sexual dimorphism in measurements from the samples meant that some equations could not be calculated (e.g. facet size: 1/24 for St Bride's Crypt; 6/24 for St Bride's Lower and 15/24 for Anglo-Saxon; facet angle: 19/24 for St Bride's Crypt, 20/24 for St Bride's Lower and 16/24 for Anglo-Saxon). These results meet

Rogers (1999) level of useful degree of accuracy greater than 80% (e.g. facet size of C1, C2, T4, T7 from St Bride's Lower; facet angle of C3 and L4 from Anglo-Saxon), whilst many of lower degrees of accuracy obtained are comparable to those achieved for vertebral levels using different parameters in other studies.

Marino (1995) achieved a 75-81% accuracy rate for C1 size: this compares favourably with the 71% for St Bride's Crypt and 80% for St Bride's Lower (even though different vertebral size parameters were used. Wescott (2000) achieved a range of accuracy of 81.7% to 83.4% for C2. Marlow and Pastor (2011) achieved degrees of accuracy of 70.91% to 75.89% when replicating Wescott's method applied to an English (Spitalfields) rather than American (Hamann-Todd and Terry) sample. Furthermore, they increased the degree of accuracy to 83.3% by using measurements from the Spitalfields collection supporting the hypothesis that discriminant function equations are population specific. Bethard and Seet (2013) further replicated Wescott (2000) using the William M. Bass and Hamilton County Forensic Centre samples and achieved a degree of accuracy between 82.3 and 86.7%. The current study achieved an accuracy of 67% for St Bride's Crypt and 84% for St Bride's Lower, but again, different vertebral size parameters were used in the studies. The St Bride's Lower results agree favourably with the results from studies listed above.

Amores et al. (2014) and MacLaughlin and Oldale (1992) achieved degrees of accuracy of 80% and 90% respectively for C7 using a combination of vertebral size measurements. These are higher than the results for St Bride's Crypt (67%) and St Bride's Lower (71%). The degree of accuracy for T12 in comparative studies varies from 80.2% (Amores, 2014), 83.3% (Yu et al., 2008) to 87% (MacLaughlin and Oldale, 1992). This study achieved degrees of accuracy of 64% for St Bride's Crypt and 71% for St Bride's Lower. The other studies used multiple parameters rather than just facet size for their equation derivation and this may explain the lower degree of accuracy achieved by this study. Discriminant function equations could not be calculated for C7 in the Anglo-Saxon sample due to a lack of qualifying variables and therefore no comparison for C7 could be drawn for this sample.

In the lumbar region, this study achieved accuracy of between 66% at L3 (St Bride's Crypt) and 78% at L4 (St Bride's Lower). This contrasts with Ostrofsky and Churchill (2015) who achieved 76% at L5 and 85.9% at L3. Once again, St Bride's Lower results compare favourably.

The Anglo-Saxon sample showed a lower degree of statistically significant sex differences in facet size when compared with the other two samples. This is reflected in the lack of

discriminant function equations that could be derived, and hence no results could be generated for the lumbar region for this sample. There are no published equations for discriminant function for facet angle.

Facet size has a significant ability to classify males and females, with a level of accuracy that can be compared relatively favourably with other single bone methods e.g. forearms: 76-86% (Barrier and Abbe, 2008) and humerus: 76-86% (Frutos, 2005). There is, however, the caveat that there must be a sufficient degree of sexual size dimorphism between the sexes in the sample under study. Facet angle is less suitable for separation of males and females due to lack of sexual dimorphism. However, for the vertebral levels where a discriminant equation could be calculated, there was again a significant ability to classify males and females with a reasonably high degree of accuracy. The greatest degree of precision was seen at C3 and L4 in the Anglo-Saxon sample, with percentage accuracy of 83% and 80% respectively.

The value of vertebral discriminant functions in bioarchaeology is limited due to the unknown degree of population specificity. Dimensions of social identity such as age, race or class may actually account for more of the variation in a specific human population than does biological sex (Agarwal, 2012). Furthermore, there has been little inter-population research related to vertebral dimensions and none that considers the remodelling effects of extrinsic factors and their relationship to facet size and angle.

In summary, a reasonable degree of facet size sexual dimorphism can be seen in the St Bride's Crypt and St Bride's Lower samples with less seen in the Anglo-Saxon sample. No vertebra stands out as being more dimorphic than the others, and the cervical region has the least number of statistically significant sex differences in size. There is little sexual dimorphism in facet angle, with the degree of orientation being similar for males and females.

This section has discussed the results relating to sexual dimorphism and the validity of using discriminant functions in the estimation of sex for the samples under investigation. It was noted that there were differing degrees of sexual dimorphism in facet size between the samples. The statistical method of estimating sex criticised by Armelagos et al. (1983:318-9) is not able to offer any valid reason for the differences seen between the samples. The next section summarises the results from exploration of the relationship between extrinsic factors with facet size and angle. Inter-sample differences in these factors may indicate differing levels of physical activity and the gendered division of labour, leading to a greater understanding of populational differences in sexual dimorphism.

7.2 Summary of Results Extrinsic Factors

Whilst sex differences have been studied in a number of post-cranial elements, there has been little attempt to provide a functional interpretation of these differences. This part of the discussion begins with a summary of the results for each extrinsic factor analysed in this study and concludes with a discussion of the overall trends.

7.2.1 Age related changes.

It could be argued that aging is a biological process and not an extrinsic factor, however the process of aging has been shown to alter dimorphic morphology and affect sex determination methods (Walker, 1995) with older individuals tending to become less discriminable between sexes and therefore it is appropriate to include it in this section. The effect of age-related morphological change on facet size and angle was tested in this study. The results show that there is no statistically significant relationship between facet size and angle with age. The lack of significant change with age may be explained by the variation inherent in facet size and angle in a group of individuals of the same age and sex (large standard deviations were recorded for most measurements), thereby masking small changes with increasing age. A further explanation for the lack of change is that longitudinal data was not used to assess true change,

Previous research has identified small but significant changes in spinal dimensions when analysed with respect to age (Erikson, 1976; Jankauskas, 1994; Humphries et al., 1998; Mosekilde and Mosekilde, 1990; Tatarek, 2001). These studies focussed on vertebral body and spinal canal dimensions. Whitmarsh et al. (2012) also found increase in vertebral body width in relation to aging with concurrent decrease in height of the lumbar vertebrae and intervertebral discs. The decrease in height leads to overloading of the facet joints, which increases facet hypertrophy and risk of osteophyte development at the margins of the facets. Facet hypertrophy was not identified in the samples used in this study. Wang and Yang (2009) reported significant differences in sagittal orientation of the lumbar facet joints by age in a clinical sample from a Chinese sample. Their data was collected from CT images rather than from dry bones. The differences between their outcome and the results of the current study may be explained by the different samples and methods used.

The accuracy of age at death estimations, particularly when ageing the skeletons of older individuals, is a key concern in age studies. Jackes (2000) argued that the study of degenerative

or age-related conditions could be limited by the current challenges of accurately estimating the age of death over 50. Age at death estimations assess the physiological age, not the chronological age, of an individual, thereby normally leading to the use of broad age ranges (Campanacho et al., 2012). A conservative approach was used in this study to manage the inaccuracies of precise age estimates. Individuals were assigned to the broad age groups of Young Adult (21-35 years), Middle Adult (36-45 years) and Old Adult (46+ years). To allow for direct comparison between samples, this division of individuals into age groups was also applied to the St Bride's Crypt sample even though the actual age at death for each individual was known.

The St Bride's Crypt data provided a perfect opportunity to test the outcome of using aggregated and disaggregated data to identify any loss of resolution when assessing age related changes in facet size and angle (Section 6.3.6). The results indicate that ascribing an individual to an age group can "dilute" the relationship between facet size and age, with the risk that a small number of strong correlational relationships could be overlooked. The "dilution effect" was not apparent for facet angle. One of the principal effects of using ageing techniques to estimate the age of an individual is to drag the estimated ages of old individuals downwards: this has the effect of reducing age-estimation accuracy in the uppermost age range (Milner and Boldsen, 2012). This can lead to an apparent shortage of individuals aged 46+ and may be a contributory factor to the loss of resolution identified above.

Facet measurements were taken as described in Section 5.2.3. Osteophytes could be seen on facets from some older individuals but were not included in the size measurements (articular surface only), which may account for the differences in size with age seen in other studies.

In summary, no significant changes in facet size and angle with age were identified in the samples used in this study. The large range of variation that occurs within each sample means that the differences in facet size and angle are not statistically significantly different between age groups and therefore it is valid to pool the age groups for comparative purposes.

7.2.2 Sample and Temporal Patterning

There is no single factor that determines the variation in sexual dimorphism between populations, but rather the patterning results from an interaction of many factors (Frayer and Wolpoff, 1985). There is variation in the expression of sexual dimorphism between and within populations (Isçan et al., 1998; Rios-Frutos, 2005 Charisi et al. 2011). Traits that are sexually dimorphic in one sample may be less so in another, with spatial variation within and between populations (Mall et al., 2001). Variation in average male and female morphology depends on the population; this means that it is necessary to re-evaluate the accuracy of sex estimation from a specific trait each time a new population is studied (Garvin et al., 2014). Low levels of dimorphism can be seen in populations that have experienced environmental stress (e.g. nutritional stress, lower socio-economic status) and high levels in populations that are not subject to high stress levels (Rickland and Tobias, 1986).

Previous research has identified the existence of population specific differences in vertebral morphometrics (Marlow and Pastor, 2011; Yu et al., 2008; Pastor, 2005; Bethard and Seet. 2012). The concept of population/sample specific variation in facet size and angle was explored in this study. All three samples investigated in this study are from South–East England, however they are from different time-periods, socioeconomic groups and environments (urban v rural). The aim was to identify inter-sample variation by comparison of mean measurement values for each facet. Meindl et al. (1985) consider that discriminant function techniques that use multiple linear measurements are population specific and environmentally labile. This theory was tested by comparison of the degree of accuracy with which sex could be estimated using the St Bride's Crypt control group data and substituting the discriminant function equations derived from the St Bride's Lower and Anglo-Saxon samples.

The results from this study showed that there are no statistically significant inter-sample differences for facet size after Bonferroni adjustment was applied. The application of the Bonferroni adjustment reduces the risk of Type I errors but can increase the risk of Type 2 errors where truly important differences can be deemed non-significant (Armstrong, 2014). Pergener (1998) advocates discussing the possible interpretations of each result from multiple testing to reach a reasonable conclusion without the help of Bonferroni adjustments. Applying Perneger's ideas to the results shows that there is little difference in facet size for males and females between St Brides' Crypt and St Bride's Lower for all age groups. This is not an unrealistic result as both samples are from the same area of London and same time period but derive from different socioeconomic groups. Greater inter-sample difference in facet size can be seen between the St Bride's Crypt and Lower groups when compared with the Anglo-Saxon sample, with the Anglo-Saxon facets being predominantly larger (particularly in females). Differences in facet angle were less obvious, however the Anglo-Saxon sample had predominantly greater means than the other two samples. Temporal trends in vertebral size may be a measure of bone remodelling as a result of adaptive mechanisms due to differences

in living conditions and physical activity levels (the Anglo-Saxon sample were from a rural setting with increased physical activity related to working on the land). It is not possible to fully understand the factors that lie behind the decrease in facet size and to a degree facet angle from Anglo-Saxon to Georgian times. Almost certainly, the Anglo-Saxon sample would have undertaken greater physical activity in childhood than the higher ranked Georgian sample (Stoodley, 2011), as the children would have likely helped on the land. There is evidence that there was not a rigid gendered division of labour and, although females had domestic duties to fulfil, they also joined the males in the fields to help with seasonal agricultural activities and the equally strenuous lifestyles of males and females would have begun early in life (Hagen, 2010). The concept that the Anglo-Saxon sample undertook greater activity is not entirely supported by the results achieved in this study when femoral robusticity is compared between the three samples (See next section). Higher levels of physical activity that occur in early life could affect facet size and angle due to phenotypic plasticity. Facet size and angle has evidently decreased from Anglo Saxon times, but it is not possible to give a simple or single explanation for this phenomenon. Low levels of physical activity are thought to be a contributory factor in reduced vertebral size (Junno et al., 2009). Differences in activity levels between rural Anglo-Saxon individuals and Urban Georgian individuals may play a role. Rühli and Henneberg (2003) identified that, from a clinical perspective, there is a lack of research into secular trends in vertebral dimensions. This is an area for further research, answers from which may help to explain the differences seen in this study.

To date, there has been little comparative research exploring differences in facet morphometrics between populations and the significance any such variation may have in terms of accuracy of discriminant function. To address this, the discriminant function equations for sex estimation using facet size from St Bride's Lower and the Anglo-Saxon sample were substituted into the St Bride's Crypt functions and the accuracy of classification achieved was tested for statistically significant difference. The results indicated that the DF were not interchangeable between the samples used in this study. Gapert et al. (2009) expressed concerns over the accuracy of discriminant functions when applied to groups from dissimilar temporal contexts and the results of this study support that concern. The results support the argument for generation of sample specific discriminant functions.

7.2.3 Femoral robusticity

Bone adapts to and is shaped by its mechanical environment (Trinkaus et al., 1994; Ruff, 2008). Mechanical loading, as it relates to lifestyle and habitual activity, plays a leading role in explaining robusticity (Stock, 2006; Martin, 2007), thus inter-sample variation in femoral robusticity may provide evidence of differing lifestyle between the samples in this study.

Previous studies of the relationship between extrinsic factors and post-cranial element size have tended to focus on measures of overall size (Oura, 2016). Ruff (1987) is critical of this approach and argues that size alone is a relatively general and imprecise indicator of biological adaptation. He argues that changes in bone geometry or shape may be more informative as to specific environmental adaptations, in particular, adaptation to mechanical forces that are indicative of functional use and thus behavioural differences (Ruff et al., 1984).

The femur and lumbar vertebrae have a similar function in relation to carrying upper body mass (Junno et al., 2009). Ruff (2008) describes the femoral midshaft section as being phenotypically flexible in response to physical activity, with the cross-sectional dimensions having high correlation with the individual's physical activity level. Similarly, Junno et al., (2009) identified that vertebral body dimensions (height and width) are connected to the individual's overall skeletal robusticity and level of physical activity. In the current study, no clear pattern of correlation between facet size, angle and femoral robusticity was observed for individual facets, however there were a higher number of strong correlations seen in Anglo-Saxons (particularly in females) than in the other two samples. This may support the suggestion that this sample had a more physically active lifestyle than the other groups. However, inter-sample comparison of femoral robusticity did not demonstrate any significant difference between left or right femoral robusticity across all three samples for males and females, which suggests that there was no difference in activity level. This is a somewhat unexpected result, which might be due to the lack of complete femora in the Anglo-Saxon population, thereby skewing the data. This is an area for further research using a larger sample size.

7.2.4 Humeral asymmetry

Humeral asymmetry has the potential to be extrinsic factor relating to facet size and angle. Differential mechanical loading, such as hand preference, is thought to result in bilateral asymmetry of paired elements (Jaskulska, 2009). It could be argued that this trait is the result

of the sexual division of labour as differences in mechanical load experienced as a result of activity by an individual may result in differences in observed bone proportions. There is much behavioural, lifestyle and occupational evidence to demonstrate variable use of the upper limbs both within and between populations (Bridges, 1991).

In this study, the relationship between intra-individual humeral and facet directional asymmetry was explored to understand how physical activity can influence facet size and angle. This analysis was hampered by the lack of intact humeri and cervical vertebra morphometric data, particularly in the female Anglo-Saxon sample, and this should be taken into account when considering the results. The greatest number of strong correlations between humeral and facet asymmetry were observed in the Anglo-Saxon sample, particularly in the thoracic and lumbar regions (no results could be calculated for the Anglo-Saxon female cervical region). The lack of statistical significance after Bonferroni adjustment means that these outcomes should be viewed with caution. No inter-sample difference in humeral directional asymmetry was identified for males and females. This is not an entirely unexpected result given that the humerus and upper arm perform a wide variety of activities and experience a wide variation in mechanical loading (Stirland, 1993; Weiss, 2003). Strong muscle groups insert into the humerus (biceps, latissimus dorsi, teres major, pectoralis major (Moore and Dalley, 2006) and differences in engagement of these muscles may impact on humeral robusticity. Lack of complete elements may have affected the outcome of this analysis.

7.2.5 Osteoarthritis

Intensive physical demands and mechanical strains are implicated in the development of osteoarthritis (OA) (Rogers and Waldron, 1995; Goodman and Martin, 2002; Weiss and Jurmain, 2007). OA is associated with activity leading to its use as an indicator of social stratification and inequality (Woo and Sciulli, 2011; Merbs, 1983). The skeletal distribution and severity of osteoarthritis has been used to reconstruct past societal changes such as gendered division of labour and occupational or habitual activity patterns (Lai and Lovell, 1992; Sofaer Deverenski, 2000; Molnar et al., 2011).

This study is particularly interested in bone plasticity and remodelling and the effect this may have on the degree of sexual dimorphism seen in the facets. Changes in the contour of bones (widening and flattening) are a well-reported effect of osteoarthritis (Waldron, 2009), making this a potentially important extrinsic factor when considering sex differences in facet size and angle. Factors considered to be important precipitants for osteoarthritis (OA) include age (Prieto-Alhambra et al., 2014), genetic predisposition (Johnson and Hunter, 2014), sex (Cross et al., 2010), race (Chopra, 2013), obesity (De Angelis and Chen, 2013), trauma (Jurmain et al., 2012), vascular (Conaghan et al., 2005) and neurologic factors (Tait and Bird, 1994) and, potentially most important, movement (Jurmain et al., 2012). OA is a disease of articular cartilage and the morphological changes that occur secondarily to breakdown of the cartilage (Larsen, 1997). Movement is thus the essential prerequisite for the development of OA: joints that do not move do not develop OA (Waldron, 2012). If the samples under investigation had dissimilar levels of movement (physical activity or lifestyle), this should manifest as differing degrees of OA in the spine (and other joints). Given that a number of factors affect the frequency and type of OA, it is not possible to directly equate it with type of physical activity or lifestyle, but Larsen (1997) generalised by commenting that populations that had demanding physical lifestyles had more OA than populations living under less demanding circumstances.

The major diagnostic characteristics of OA are considered to be the development of osteophytes and pitting together or eburnation alone (Bourke, 1967; Rogers et al., 1987; Jurmain and Kilgore, 1995; Weiss and Jurmain, 2007). Correlation studies in this thesis revealed that Anglo-Saxon females had the highest levels of strong correlation with facet size for eburnation (cervical region), pitting (cervical region) and osteophyte formation (cervical and thoracic regions), whilst Anglo-Saxon males had the highest levels of strong correlation with facet size for eburnation in the thoracic and lumbar regions. Given the variation in numbers of strong correlations between diagnostic factors with facet size, angle and inter-sample variation in facet size and angle, differences in prevalence rates for the diagnostic factors and also OA were estimated.

Crude prevalence rate is the prevalence of a disease presented as the percentage of individuals displaying pathological changes associated with the disease within the sample group studied (Klaus, 2014). Crude prevalence rates for the OA diagnostic factors were calculated for the three samples in this study to provide an overall picture of OA prevalence. The crude rates indicated inter-sample differences in rates for eburnation and pitting, but no inter-sample differences for osteophyte formation. This method does not always take into account the preservation and observability of the specific skeletal elements than manifest the pathological changes (Dutour, 2008:133). Roberts and Cox (2003:41) argue that crude prevalence can misrepresent the true prevalence because this method assumes that all bones of all skeletons are equally preserved. This is not the case for the samples used in this study; the Anglo-Saxon

sample had a poorer level of bone preservation than the St Bride's samples, therefore, true prevalence rates of OA were also calculated to facilitate inter-sample comparison. True prevalence rates represent a percentage value of the number of facets observed and the number affected by pathological change due to OA (Waldron, 2007: 62). Crude prevalence rates should be presented alongside true prevalence rates for facet joint OA. The number of facets displaying OA in skeletons within a sample is not a direct indicator of the number of individuals who may have been affected by OA. This is because individuals may display varying degrees of vertebral involvement, from one facet to the entire spine. Differences between true and crude prevalence rates between samples may be the result of the involvement of a greater distribution of OA across the spine in some individuals (i.e. OA may have been found on multiple facets in some individuals, increasing the true prevalence). The two methods of calculating prevalence can perform different roles in the analysis of OA of individuals within a sample. While crude prevalence can help aid in understanding the potential extent that facet OA affected the population, true prevalence can provide a more accurate understanding of the distribution of facet OA throughout the spine of individuals.

The different methods used to diagnose OA make inter-assemblage comparison of crude and true prevalence rates difficult. It is not always clear which factors have been used to diagnose OA nor which denominators have been used in the calculation of prevalence (Waldron, 2012). For this reason, only inter-sample prevalence comparison is discussed in this study.

Statistical analysis of the true prevalence rates of the diagnostic features of OA identified differences between the male and female Anglo-Saxon sample and the St Bride's samples for pitting, but not eburnation nor osteophytes. The true prevalence rates for OA are significantly higher for Anglo-Saxon males and females than in the St Bride's sample. No significant difference in prevalence of OA was observed between the St Bride's Crypt and St Bride's Lower males and females. It is well established that the gendered division of labour could result in differences in the prevalence of OA seen between males and females (Bridges, 1992; Larsen et al., 2008). Bridges (1992) identified that males show higher prevalence of OA than females as they are generally involved in more and heavier labour tasks. The data from St Bride's Crypt support this hypothesis, however the lack of significant difference in prevalence of OA between males and females from St Bride's Lower and Anglo-Saxon samples does not.

The differences in true prevalence of OA between the samples cannot directly be linked to specific physical activities however, the differences do indicate differing degrees of physical

238

activity undertaken between the samples. The only significant inter-sex differences for OA were seen in the St Bride's Crypt sample, indicating that these males and females had differing levels of physical activity whereas the males and females from the other samples may have undertaken the same degree of physical activity. These latter two groups are of lower socioeconomic status and so the distribution of overall activity between the sexes and age groups was probably different from the high socio-economic status St Bride's Crypt group.

Brocher and Willert (1980) report degenerative changes in the cervical region are generally seen in the most flexible part of the cervical spine, in the region of C5-C6. Aufderheide and Rodriguez-Martin (1998) expand this to identify the most commonly affected areas as C6-T1, T2-T5 and L2-L4. Waldron (1991) agrees but uses a broader description of the cervical and upper thoracic regions and also the lower lumbar spine. The cervical spine of males and females from all three samples show the presence of OA, with more found in females than males. Sindermann and Flohr (2007) attribute this to anatomical causes rather than physical workload, describing the female neck as being proportionally longer and less muscular than its male equivalent and therefore more prone to early onset degeneration. Srikanth et al. (2005) undertook a meta-analysis of 40 papers to resolve uncertainty regarding sex differences in prevalence and severity of OA. Eleven of the papers analysed were related to spinal OA. They failed to identify any association between sex, age at onset and prevalence of cervical OA. This supports the finding that physical work rather than anatomy is the precursor to cervical OA.

For the thoracic region, the female mean values of true prevalence between T7 and L5 vary between 1.7 (St Bride's Lower), 4.1 (St Bride's Crypt) to 14.8 (Anglo-Saxon) and male mean true prevalence values between 2.9 (St Bride's Lower), 5.0 (St Bride's Crypt) to 10.8 (Anglo-Saxon). Clearly, there is a lifestyle difference between the St Bride's and Anglo-Saxon samples. Waldron (2012) states that joints that do not move do not develop OA. From this it can be conjectured that St Bride's individuals have a factor that restricts movement of their spines, which in turn inhibits the development of OA. Gibson (2015) examined a number of female skeletons including 18 from St Bride's Lower and identified skeletal markers (flattened spinous processes) consistent with long-term pressure. She was able to calculate waist sizes (derived from coronal rib measurements) and found them to be considerably smaller in the St Bride's Lower group than in modern women. These factors were seen in older individuals and are thought to be a measure of the effect of long-term tight-laced corseting. Corseting was very popular from the 1600's (Bendall, 2014). In the early 19th Century, corseting became very popular and women were expected to wear them, regardless of their socioeconomic status (Bendall, 2014; Steele, 2010). Male corsets were also worn in the 18th and early 19th century (Cole, 2010; Schwartz, 1979). There are few bioanthropological studies measuring the physical changes caused by corseting, however, Moore and Buckberry (2016) reported on a vertebral compression changes associated with the wearing of a corset identified in a 19th century adult male with Pott's disease of the vertebral column. It is thought that the corset was worn as a corrective treatment for the spinal deformity caused by the Pott's disease. It cannot be precisely determined if the corset was for sartorial rather than medical reasons. Moore and Buckberry (2016) note that the types of thoracic deformation seen in this individual have been described in a small number of female skeletons without TB. The effect of wearing corsets may be a factor in differences in true prevalence of OA from the St Bride's samples when compared to the Anglo-Saxon sample. However, this may be an oversimplification and the differences seen may be due to generally lower levels of physical activity.

The results of this section indicate that the facet joints of Anglo-Saxon individuals were significantly more physically stressed than the St Bride's individuals, which supports the assertion that both sexes were subject to broadly similar forms and levels of stress over the life course. This may be associated with the reduction in degree of sexual dimorphism identified for this group.

7.2.6 Vertebral body osteophytosis

Vertebral osteophytes (osteophytosis) are thought to develop as an adaptive response to joint instability as a result of intervertebral disc degeneration and/or the physiological effects of continual loading through the vertebral column (Nathan et al., 1994). Osteophytosis has been widely studied and shown to be associated with sex and age (Nathan et al., 1994; Sofaer Deverenski, 2000; Snodgrass, 2004). However, the frequency of osteophytosis distribution varies between groups of individuals and increases with age to the extent that Watanabe and Terazawa (2006) consider that osteophytosis formation is very specific to each individual. Lindblom (1951) ascribed differences in osteophytosis distribution to differences in lifestyle from the comparison of spines from Swedish prehistoric skeletons with those of modern living individuals (identifying osteophytosis distribution from radiographs). Studies have identified a higher frequency of osteophytes in males compared to females (Van de Merwe et al., 2006; Kacar et al. 2017). These studies reported that males had a significantly higher frequency of osteophytes than females in the lower segments of the thoracic region and lumbar region. Kacar et al., (2017) suggested that this distribution could be explained by the fact that males generally have a higher body mass, are more robust, and participate in more strenuous activity than females, thereby leading to greater pressure on the spine and thus the development of more osteophytes. Males or females engaged in heavy manual work are more likely to develop osteophytosis than sedentary workers (Nathan, 1962; O'Neill, 1999). Occupational and lifestyle factors play an important role in the development of osteophytosis, with sex differences in osteophytosis development being more pronounced in the gendered division of labour where heavy physical labour is structured by sex (Kennedy, 1989). The males and females from the Anglo-Saxon sample had a similar pattern of osteophytosis development supporting the hypothesis that there was much less gendered division of labour in terms of overall activity but not in terms of specific activities than in the St Bride's samples. This sample also demonstrated a higher true prevalence of osteophytosis indicating a more physical lifestyle.

Some clinical research has indicated that OA follows and may even result from osteophytosis (Prescher, 1988; Fujiwara et al., 2001; Benoist, 2003). In this study, not all vertebrae with osteophytosis of the vertebral body also have facet joint OA, particularly in the St Bride's samples. This may be because the individuals died before they had time to develop OA.

There are no published studies that look at the relationship between vertebral osteophytosis and facet size and angle. The strongest correlations between vertebral osteophytosis and facet size were seen in Anglo-Saxon males and females, particularly in the cervical (male), thoracic (female) and lumbar (male and female) spines. A similar pattern was seen for facet angle with the strongest correlations being seen in the cervical (male), thoracic and lumbar (male and female) Anglo-Saxon. Increased scores of vertebral osteophytosis are negatively correlated with facet size and angle in many individuals from all three samples. Smaller facets and angles would lead to greater dissipation of load through the vertebral bodies leading to the development of vertebral osteophytosis. This would act as a compensation mechanism to increase the body area.

7.2.7 DFA using Extrinsic Factor Variables

Whilst many studies have successfully used vertebral measurements to estimate sex (MacLaughlin and Oldale, 1992; Liguoro, 1994; Marino, 1995; Wescott, 2000; Marlow and Pastor, 2011; Hou et al., 2012), there has been no attempt to explain differences in degree of sexual dimorphism between samples. The present study demonstrated sexual dimorphism in St Bride's sample facet size. Some facets from the St Bride's Crypt and St Bride Lower samples

have mean metric measurements that are larger in males than females (in agreement with the studies of Wescott, 2000; Yu et al., 2008; Hou et al., 2012 Zheng et al., 2012). The converse is true for the Anglo-Saxon sample, but also associated with little difference facet size sexual dimorphism. The lack of dimorphism seen in the Anglo-Saxon sample meant that it was not possible to generate discriminant function equations to estimate sex for many vertebral levels.

The vertebral measurements used in this study reveal the biological differences between males and females. The effect of biomechanical differences leading to functional adaptation of the facets was assessed by evaluating the relationship between extrinsic factors that are indicative of activity and facet size and angle. Multiple discriminant functions generated from the seven variables (facet size, facet angle, femoral robusticity, humeral and facet asymmetry, eburnation pitting and osteophyte and osteophytosis score) for all vertebrae from the three samples were used to estimate sex. The percentage accuracy of correct classification when using functions that included extrinsic factors was compared to the outcome of using functions derived from facet size or angle alone. Varying degrees of accuracy were obtained when extrinsic factors were included in functions, with T8 and L1 (St Bride's Crypt) and L4 (St Bride's Lower) achieving posterior probability scores of 80% or over. Rogers (1999) reports that this is the generally accepted standard for determining the utility of a sex assessment method with the caveat that it corresponds between methods where true sex is not known. In comparison, functions generated from facet size alone failed to generate posterior probability scores of 80% or over in the St Bride's Crypt sample, but reached that score for C1, C2, T4 and T7 (St Bride's Lower). Functions derived from facet angle achieved a posterior probability of 83% for C3 (Anglo-Saxon) only. The lack of dimorphism in facet angle meant that few functions could be calculated.

The overall results demonstrate that a higher number of discriminant functions can be generated to estimate sex when extrinsic factors are included, particularly in samples that do not demonstrate high degrees of dimorphism.

7.2.8 General discussion

The use of discriminant functions to estimate sex is well established in the field of bioarchaeology (Moore, 2013). Many equations have been created for different skeletal elements with varying degrees of accuracy of classification (Schwartz, 2006). This study clearly demonstrates that vertebral discriminant functions (DF) from a suitably dimorphic sample can be used for sex classification. However, there are two major limitations to the use of discriminant functions for sex estimation that should be taken into consideration; there must be demonstrable sexual dimorphism in the bones being measured, and the equations are population specific and hence should be developed for each population or region separately (Hora and Sládek, 2018). The lack of sexual dimorphism in facet size between males and females from the Anglo-Saxon sample limited the number of vertebral levels for which functions could be generated. A similar issue arose for facet angle for all three samples. Generation of such discriminant functions was also hampered by poor preservation and missing elements, particularly in the Anglo-Saxon female sample.

The population specificity of discriminant functions is a much-discussed topic (Bidmus and Dayal, 2004; Dabbs, and Moore-Jansen, 2004; Hora and Sládek, 2018), however, few studies attempt to explain the rationale behind this in relation to lifestyle and activity factors. Lower sex classification accuracy is seen in discriminant functions derived from geographically distant samples (Hou et al., 2011). The samples in this study are geographically close but the degree of accuracy achieved when data from St Bride's Lower and Anglo-Saxon samples were substituted into the discriminant functions for St Bride's Crypt were significantly lower for some vertebral levels. There was some reduction in accuracy for St Bride's Lower and a much lower degree of accuracy for the Anglo-Saxon sample, indicating that the derived functions are sample specific. Clearly geographical distance is not the only factor to be considered. Associated environmental factors relating to each population in terms of historical period and socioeconomic status also should be considered.

Previous studies have shown that variation in sexual dimorphism can be used as a sensitive indicator of the quality of life as it has been demonstrated that dimorphism responds to environmental stress (Stini, 1982; Stinson, 1985; Charisi et al., 2011). Major differences in the degree of sexual dimorphism were observed between the St Bride's and the Anglo-Saxon samples. Could this be attributable to socio-economic or lifestyle factors? Sex differences in physical activity would have been a key component affecting dimorphism in the samples under study. Biomechanical force (in the form of physical activity) directly affects bone remodelling (Ruff et al., 2006). Both males and females from the rural Anglo-Saxon samples would have been engaged in hard labour associated with a farming lifestyle (Welch, 2000; Hines, 2003). Documentary evidence exists that there was not a rigid sexual division of labour (Fisher, 1995). Females would have worked in the fields with the males as dictated by the seasons (e.g. harvest) but would also have been associated with spinning and weaving as well as dairy work

243

and domestic chores such as bread making (Fell, 1984). The lifestyles of males and females would have been equally strenuous and started early in life (Welch, 2000).

St Bride's Lower was known to be a resting place for the poor of the parish, prostitutes, lodgers, the incarcerated from Fleet Street prison or workhouse inmates (Miles and Conheeney, 2010). Miles and Conheeney (2010) identify that those that were working were employed as merchants, craftsmen and artisans, however many individuals were extremely poor and relied on charity and workhouses for survival. The remains of affluent families were buried in St Bride's Crypt, in contrast those of an impoverished society from a different social class and lifestyle were interred in St Bride's Lower.

The pattern and degree of sexual dimorphism seen in the Anglo-Saxon samples differ from that seen in the St Bride's samples. Nutrition, reproduction and physical activity would have played an important role as non-genetic factors influencing vertebral morphology. Nutrition and reproduction are difficult parameters to measure in archaeological samples and therefore, in this study, focus was placed on markers of physical activity. Both sexes from the Anglo-Saxon sample demonstrated similar patterns of eburnation, pitting and osteophytes as a result of degenerative changes (OA). This result is similar to that seen during analysis of a sample from Wharram Percy by Sofaer-Deverenski (2000). The pattern of changes supports the assertion that both sexes were subject to broadly similar forms and levels of physical stress in their lifetime.

Bioarchaeological analysis of rural and urban samples enables comparison between the type and intensity of physical activity. The samples used in this study did not demonstrate intersample differences in femoral robusticity (as measured by cross-sectional geometry of the femur) for males or females from rural and urban settings. This does not support the assertion that there are differing levels of physical activity between the sexes in a rural setting. This is a somewhat surprising result. Increasing bone robusticity has long been considered a result of repeated heavy physical activity patterns associated with growing crops (Bridges et al., 2000). However, the relationship between lifestyle and robusticity is not fully understood (Stock and Pfeiffer, 2001; Ruff, 2005). Bone robusticity represents bone strength according to the physical properties of that bone (Stock and Shaw, 2007). The use of external bone measurements to estimate robusticity, whilst still considered an appropriate method to assess mechanical loading (Stock and Shaw. 2007) is being superseded by the use of cross-sectional bone geometry (Stock and Pfeiffer, 2001; Stock and Shaw, 2007; Ruff, 2005). The use of external measurements has been criticised by Ruff (1987) as these are only a partial reflection of the cross-sectional geometric distribution of bone and cannot measure variation in internal contours, thereby reducing the accuracy with which they can provide information about mechanical loading an individual has undergone. Pearson and Lieberman (2004: 76) suggest that mechanical loads during adulthood have little effect on the external dimensions of long bone diaphyses but result in greater cross-sectional areas from smaller medullary cavities. Ruff and Haynes (1983) documented an increase in femoral robusticity with age but the Campanacho et al. (2012) reported the converse. The age effects on femoral robusticity were not controlled for in the present study, and this may have affected the degree of inter-sample difference observed.

Similarly, there were no significant differences in mean humeral directional asymmetry between the samples or between sexes. This indicates that work undertaken by the individuals in this study was generally similar in magnitude of strength (Mays, 2007), but comparison of mean humeral robusticity failing to identify a single individual who had significant difference in humeral robusticity due to a repetitive one-handed task. A further complication of this analysis was the poor state of preservation of female Anglo-Saxon humeri, thus limiting the conclusions that can be drawn.

This is one of the first studies to examine facet joints for all vertebrae and to test their application for sex estimation in three samples from a similar region of England from different time periods and of varied socioeconomic status. This study examined sexual dimorphism in Georgian and Anglo-Saxon samples and generated nine multivariate discriminant functions for sex estimation with overall accuracy rates of 80% or above (4 for the St Bride's Lower sample based on facet size, 2 for the Anglo-Saxon sample based on facet angle and 3 for the St Bride's Crypt sample based on inclusion of extrinsic factors). The results of this study have implications for the use of discriminant functions for determining sex from vertebral measurements. Generation of discriminant functions for less sexually dimorphic samples using extrinsic factors not only increases the number of functions that can be generated improving the opportunity to estimate sex but also provides evidence for the effects of physical activity on the facets as a factor in the degree of dimorphism seen.

Variations in the accuracy of such techniques have been attributed to population specificity alone. The results of the present study suggest that specific mechanical factors may explain some of the lack of dimorphism seen in the results. Thus, variations in the accuracy of sex

discrimination between different samples may reflect at least partly, the degree of sexual dimorphism in activity patterns in those populations. Environmental factors that may influence dimorphism in a particular historic or socioeconomic setting should be considered and used to test population homogeneity when comparing samples from a similar period. Any future attempts to employ these methods in sex determination should take into account these population differences in behaviour.

Chapter 8 Conclusion

The current study identified two main questions in relation to zygapophyseal joints:

1) Do vertebral facet size and sagittal angle demonstrate sexual dimorphism?

2) Does the inclusion of extrinsic factors (relating to physical activity) in discriminant functions increase the degree of accuracy for sex estimation when there are low levels of sexual dimorphism?

The results of the current study have demonstrated that there is sexual dimorphism in the facet joints, particularly in terms of facet size, however, the degree of dimorphism varies for different samples. This indicates a complexity that has not been previously commented on by other researchers in this field, but partly explains the population specificity of discriminant functions used to estimate sex. The inclusion of extrinsic factors (related to physical activity) allows for generation of discriminant functions for vertebral levels where the degree of sexual dimorphism is too small for derivation from facet size only, thereby increasing the opportunity to estimate the sex of an unknown individual from the same population.

Analysis of facet size sexual dimorphism showed clear difference between males and females from the St Bride's samples but this was less apparent in the Anglo-Saxon sample. Calculation of a sexual dimorphic index for facet size demonstrated a pattern of male dominated indices throughout the spine for St Bride's samples and less so in the Anglo-Saxon sample. A much higher number of facets remained unclassified for sex from their index values in the Anglo-Saxon sample. A different picture was seen when sexual dimorphism indices were calculated for facet angle with many facets being unclassified. In the St Bride's Crypt and Anglo-Saxon sample, a high proportion of lumbar facets were classified as "female". The lumbar facet angles may be due to the deepening of the lumbar lordosis in females during pregnancy, which acts to counterbalance the growing foetal weight (Whitcombe et al., 2007). Overall, there were fewer vertebral levels across all three samples with facet angles showing sexual dimorphism leading to difficulty in generation of discriminant functions. Facet angle is a poor determinant of sex.

Inter-sample comparison of facet size and angle showed a significant difference in facet size and angle particularly between the St Bride's sample and the Anglo-Saxon females. How could these differences be explained? An interesting picture emerged in this study when the

Chapter 8 Conclusion

relationship between extrinsic factors and facet size and angle were explored. The use of Bonferroni adjustment to reduce the risk of Type I error (where a true null hypothesis is rejected i.e. creation of a false positive) increased the risk of Type II error (where truly important results are deemed insignificant i.e. creation of a false negative) masking the true number of strong relationships. However, the highest number of strong correlations for each relationship was always seen in the Anglo-Saxon sample in the females for both facet size and angle. These results differentiate this group of females from the others. Given that the strong correlations were between facet morphology and extrinsic factors relating to activity, the implication is that this group undertook a different degree of activity to the other female samples. The results do not indicate a specific activity, only that this group were more physically active. This is a clear indication that facet morphology is moderated by physical activity.

Sexually dimorphic features of the skeleton are of fundamental importance when constructing a biological profile from unidentified human remains. It is not uncommon to encounter broken or poorly preserved skeletons and therefore there is a need to be able to use as many elements as possible for sex determination. Most metric studies of skeletal sexual dimorphism focus on bones that manifest gross size differences whilst avoiding those that do not (Wescott, 2000). However, bioarchaeologists sometimes need to reliably estimate sex from unknown individuals represented by only a few bones. This research aimed to look at the effectiveness with which vertebral facet joints could be used as a predictor of sex and was able to demonstrate that, for a sexually dimorphic sample, a good degree of accuracy of correct classification could be achieved at some vertebral levels. This study also responded to the call from a number of researchers for renewed emphasis on function and shape-related rather than simply size-related characteristics when analysing sex differences in skeletal structure (Van Gerven, 1972; Armelagos et al., 1982; DiBennardo and Taylor, 1982, Ruff, 1987). The present study illustrates one way in which a biomechanical approach, combining behavioural reconstruction with metric analysis, can be used to contribute new insights into this area of research. This study also shows that facet morphology may contain important information regarding past mechanical loading history and thus behavioural patterns.

The study began with a detailed comparison of sexual dimorphism in facet size and angle in three archaeological samples, two from Georgian London of contrasting socioeconomic status and a third from Anglo-Saxon southeast England. Sex differences in facet size were noted in the St Bride's samples with male facet size generally than female. This sex difference was much lower in the Anglo-Saxon sample with females and males having more comparable facet size. Results were then interpreted in light of functional adaptation as a result of physical activity, leading to the following conclusions:

- 1) Sexual dimorphism in facet size and of varying degree is characteristic of the samples under study. These results can be compared to many human populations from England to the US, Europe to China (Wescott, 2008; Amores et al., 2014; Yu et al., 2008).
- 2) There is consistent systematic variation between populations in the degree of sexual dimorphism seen in vertebral facets. This may be related to the gendered division of labour. In this study, the urban samples showed the greatest sexual dimorphism and rural agricultural samples the least. This corresponds to an increase in in the sexual division of labour from rural to urban lifestyle.
- 3) Techniques for the assessment of sex from vertebral measurements should consider systematic population differences. Inclusion of extrinsic factors related to physical activity, in addition to size measures, may aid in the discrimination between the sexes, whilst providing an understanding of the lifestyle of the sample under study. It is recommended that this process be added to studies of sexual dimorphism to provide evidence of different degrees of physical activity and functional adaptation of the element under study.

In conclusion, in order to gain the most comprehensive understanding of the complexities of sex estimation using the facet joints, it is recommended that the most inclusive method possible should be used. This increases understanding of sample variation and the effects that the gendered division of labour can have on the sexual dimorphism of a joint.

8.1 Future directions

The results from the present study have identified a number of potential avenues through for future research. It would be useful to incorporate geometric 3D data from the samples studied. The method used in this study measures the gross area of the facets without taking facet depth or curvature into account. Information about the depth of curvature of the facet would add another useful dimension for analysis. Furthermore, measurement of femoral cross-sectional geometry would elicit a greater degree of accuracy for robusticity measurements than those derived from metric dimensions. This could be particularly true

Chapter 8 Conclusion

when assessing humeral directional asymmetry as it has been suggested that metric dimensions are less asymmetric than geometric ones (Auerbach and Ruff, 2006).

In terms of metric analysis, there is scope for adding more vertebral measurements (e.g. vertebral body height and diameter) to subsequent analyses, which may benefit from the incorporation of more dimensions in line with previously published studies of sex determination from vertebrae. This would identify the optimum data needed to maximise the accuracy of sex classification. The studies previously discussed in Chapter 3 focus on one or two vertebral levels without rationale for choice of level. Application of a multidimensional approach to all vertebrae would help in identifying if certain levels offer a greater degree of classification accuracy.

The lifestyle, occupation and body size of individuals were unknown factors in this study. Each of these factors can have an influence on facet morphology. Whilst lifestyle in terms of physical activity can be assessed by examination of degenerative changes seen on the facets, actual occupation remains unknown. Body size correlates with vertebral body dimensions, with males being larger than females (Gilsantz, 1997). The relationship between body and facet size is not clear from the existing literature and is an area for further investigation. Body size (stature) calculated using Raxter et al.'s (2006) revised Fully method, body weight calculated from femoral head breadth (Auerbach and Ruff, 2004) or other regression equations for incomplete remains (Ruff et al., 2012) could be included as further extrinsic factors and the correlation between them and facet size and angle measured.

Statistically, the number of analyses performed meant that Bonferroni adjustment was applied to reduce the risk of Type 1 errors; this however increased the risk of Type II errors. A true picture of the relationship between facet size/angle and extrinsic factors may not have been seen. This could be addressed by using Bayesian statistical analysis.

The inclusion of extrinsic factors in discriminant functions for the less sexually dimorphic Anglo-Saxon sample allows the generation of discriminant functions for a greater number of vertebral levels than if these factors were ignored. However, when the extrinsic factor functions were used, the percentage degree of accuracy of classification ranged between 47-70% for this sample. Hora and Sládek (2018) demonstrated that increasing the posterior probability in discriminant function analysis increased the percentage accuracy of correct of classification to over 90% for the samples in their study. Application of this method to the Anglo-Saxon sample in this study may increase the degree of certainty of correct classification. This would lead to greater confidence in estimating sex of individuals whose results fell above the increased prior probability. However, the number of individuals that could not be classified would be increased.

The current study identified many interesting findings in relation to the incorporation of functional adaptation and shape in the analysis of sex differences between samples. Further investigation using a wider range of samples will allow for better understanding of the complexities of sex estimation from vertebrae.

Abbreviations used in appendices

C – cervical

T – thoracic

L-lumbar

Lt- left

R- right

S- superior

I-inferior

H – height

W – width

A-angle

Appendix A. Measure of Intra-Observer Agreement

Facet	Wilks' Lambda	Ρ
C1SLW	0.759	0.288
C1SRW	0.989	0.953
C1SLH	0.943	0.769
C1SRH	0.706	0.209
C1ILW	0.976	0.909
C1IRW	0.809	0.384
C1ILH	0.479	0.053
C1IRH	0.887	0.582
C2SLW	0.897	0.552
C2SRW	0.871	0.501
C2SLH	0.954	0.774
C2SRH	0.892	0.563
C2ILW	0.616	0.07
C2IRW	0.802	0.298
C2ILH	0.882	0.502
C2IRH	0.681	0.12
C3SLW	0.967	0.831
C3SRW	0.87	0.466
C3SLH	0.948	0.747
C3SRH	0.788	0.27
C3ILW	0.9	0.561
C3IRW	0.771	0.239
C3ILH	0.666	0.107
C3IRH	0.987	0.93

Facet	Wilks' Lambda	Р
C4SLW	0.852	0.448
C4SRW	0.537	0.045
C4SLH	0.925	0.676
C4SRH	0.979	0.9
C4ILW	0.998	0.992
C4IRW	0.85	0.444
C4ILH	0.809	0.386
C4IRH	0.819	0.367
C5SLW	0.78	0.225
C5LRW	0.959	0.779
C5SLH	0.859	0.403
C5SRH	0.897	0.521
C5ILW	0.82	0.305
C5IRW	0.802	0.266
C5ILH	0.922	0.613
C5IRH	0.928	0.639
C6SLW	0.544	0.035
C6SRW	0.744	0.196
C6SLH	0.852	0.414
C6SRH	0.872	0.472
C6ILW	0.966	0.825
C6IRW	0.895	0.543
C6ILH	0.995	0.973
C6IRH	0.642	0.087

Appendix	A—1 One	-way repeated-	measures ANO	VA for facet	width and	height cont	inued
Appendix	A I Olie	-way repeated-	incasures ANO	VAIOI lacet	widthand	neight cont	mueu

Facet	Wilks' Lambda	Ρ
C7SRW	0.961	0.82
C7SLH	0.956	0.782
C7SRH	0.959	0.812
C7ILW	0.803	0.299
C7IRW	0.779	0.253
C7ISH	0.725	0.17
C7IRH	0.893	0.537
T1SLW	0.993	0.963
T1SRW	0.761	0.194
T1SLH	0.933	0.683
T1SRH	0.997	0.985
T1ILW	0.983	0.905
T1IRW	0.56	0.031
T1ILH	0.853	0.386
T1IRH	0.789	0.24
T2SLW	0.99	0.93
T2SRW	0.83	0.272
T2SLH	0.849	0.319
T2SRH	0.821	0.252
T2ILW	0.982	0.88
T2IRW	0.899	0.473
T2ILH	0.701	0.083
T2IRH	0.852	0.327

Facet	Wilks' Lambda	Р
T3SLW	0.937	0.613
T3SRW	0.805	0.196
T3SLH	0.824	0.234
T3SRH	0.951	0.684
T3ILW	0.884	0.398
T3IRW	0.887	0.406
T3ILH	0.895	0.433
T3IRH	0.831	0.25
T3SLW	0.976	0.822
T3SRW	0.92	0.512
T4SLH	0.973	0.802
T4SRH	0.811	0.187
T4ILW	0.915	0.515
T4IRW	0.983	0.879
T4ILH	0.951	0.687
T4IRH	0.9	0.455
T5SLW	0.973	0.814
T5SRW	0.975	0.826
T5SLH	0.941	0.634
T5SRH	0.941	0.632
T5ILW	0.88	0.409
T5IRW	0.793	0.197
T5ISH	0.893	0.453
T5IRH	0.94	0.649

Facet	Wilks' Lambda	Р
T6SLW	0.631	0.032
T6SRW	0.994	0.956
T6SLH	0.912	0.502
T6SRH	0.949	0.673
T6ISW	0.992	0.946
T6IRW	0.962	0.764
T6ILH	0.612	0.032
T6IRH	0.796	0.202
T7SLW	0.966	0.759
T7SRW	0.921	0.52
T7SLH	0.799	0.166
T7SRH	0.929	0.556
T7ILW	0.995	0.96
T7IRW	0.807	0.2
T7ILH	0.811	0.187
T7IRH	0.824	0.235
T8SLW	0.962	0.736
T8SRW	0.78	0.136
T8SLH	0.808	0.182
T8SRH	0.99	0.925
T8ILW	0.885	0.377
T8ILH	0.834	0.233
T8IRH	0.97	0.782
T8IRH	0.9	0.429

Facet	Wilks' Lambda	Р
T9SLW	0.927	0.527
T9SRW	0.895	0.369
T9SLH	0.84	0.209
T9SRH	0.776	0.102
T9ILW	0.869	0.284
T9IRW	0.847	0.223
T9ILH	0.891	0.353
T9IRH	0.875	0.3
T10SLW	0.77	0.095
T10SRW	0.892	0.356
T10SLH	0.704	0.042
T10SRH	0.82	0.168
T10ILW	0.772	0.098
T10IRW	0.968	0.749
T10ILH	0.915	0.451
T10IRH	0.994	0.949
T11SLW	0.918	0.505
T11SRW	0.864	0.29
T11SLH	0.683	0.048
T11ILW	0.907	0.436
T11IRW	0.967	0.749
T11ILH	0.915	0.47
T11IRH	0.955	0.69

Facet	Wilks' Lambda	Р
T12SLW	0.105	0.045
T12SRW	0.074	0.05
T12SLH	0.806	0.177
T12SRH	0.815	0.176
T12ILW	0.608	0.019
T12IRW	0.982	0.859
T12ILH	0.962	0.733
T12IRH	0.615	0.016
L1SLW	0.469	0.002
L1SRW	0.874	0.341
L1SLH	0.878	0.352
L1SRH	0.961	0.727
L1ILW	0.871	0.308
L1ILH	0.959	0.703
L1IRH	0.86	0.278
L2SLW	0.728	0.079
L2SRW	0.959	0.718
L2SLH	0.76	0.112
L2SRH	0.819	0.202
L2ILW	0.878	0.402
L2IRW	0.971	0.804
L2ILH	0.904	0.492
L2IRH	0.931	0.587

-		
Facet	Wilks' Lambda	Р
L3SLW	0.956	0.681
L3SRW	0.998	0.985
L3SLH	0.842	0.232
L3SRH	0.891	0.373
L3ILW	0.976	0.81
L3IRW	0.905	0.43
L3ILH	0.945	0.62
L3IRH	0.884	0.35
L\$SLW	0.986	0.884
L4SRW	0.657	0.028
L4SLH	0.9	0.408
L4SRH	0.928	0.53
L4ILW	0.924	0.511
L4IRW	0.895	0.388
L4ILH	0.916	0.494
L4IRH	0.963	0.739
L5SLW	0.883	0.393
L5SRW	0.931	0.586
L5SLH	0.869	0.348
L5SRH	0.65	0.039
L5ILW	0.965	0.763
L5IRW	0.649	0.039
L5ILH	0.772	0.144
L5IRH	0.841	0.272

Appendix A-1 One-way repeated measures ANOVA facet angle

-		
Facet	Wilks' Lambda	Р
C1SLA	0.483	0.5
C1SRA	0.566	0.103
C1ILA	0.952	0.843
C1IRA	0.839	0.496
C2SLA	0.725	0.235
C2SRA	0.626	0.122
C2ILA	0.719	0.226
C2IRA	0.986	0.937
C3SLA	0.76	0.291
C3SRA	0.641	0.108
C3ILA	0.98	0.914
C3IRA	0.724	0.863
C4SLA	0.714	0.22
C4SRA	0.936	0.743
C4ILA	0.956	0.834
C4IRA	0.902	0.629
C5SLA	0.963	0.813
C5SRA	0.954	0.773
C5ILA	0.906	0.61
C5IRA	0.527	0.041
Facet	Wilks' Lambda	Р
-------	---------------	-------
C6SLA	0.879	0.524
C6SRA	0.658	0.123
C6ILA	0.607	0.106
C6IRA	0.929	0.744
C7SLA	0.616	0.113
C7SRA	0.828	0.427
C7ISA	0.606	0.105
C7IRA	0.909	0.65
T1SLA	0.881	0.529
T1SRA	0.71	0.18
T1ILA	0.672	0.137
T1IRA	0.846	0.434
T2SLA	0.883	0.473
T2SRA	0.824	0.313
T2ILA	0.826	0.318
T2IRA	0.956	0.763
T3SLA	0.938	0.66
T3SRA	0.669	0.073
T3ILA	0.789	0.214
T3IRA	0.723	0.122
T4SLA	0.87	0.378
T4SRA	0.951	0.702
T4ILA	0.638	0.054
T4IRA	0.644	0.058

Appendix A—2 One-way repeated measures ANOVA facet angle continued

1	1	
Facet	Wilks' Lambda	Р
T5SLA	0.791	0.218
T5SRA	0.677	0.079
T5ILA	0.612	0.052
T5IRA	0.779	0.223
T6SLA	0.89	0.468
T6SRA	0.549	0.02
T6ILA	0.666	0.087
T6IRA	0.717	0.136
T7SLA	0.949	0.695
T7SRA	0.941	0.654
T7ILA	0.578	0.028
T7IRA	0.831	0.3
T8SLA	0.913	0.53
T8SRA	0.547	0.015
T8ILA	0.807	0.223
T8IRA	0.833	0.279
T9SLA	0.949	0.659
T9SRA	0.745	0.095
T9ILA	0.545	0.008
T9IRA	0.752	0.102
T10SLA	0.821	0.206
T10SRA	0.786	0.146
T10ILA	0.734	0.084
T10IRA	0.685	0.049

Appendix A—2 One-way repeated measures ANOVA facet angle continued

Facet	Wilks' Lambda	Р
T11SLA	0.672	0.051
T11SRA	0.571	0.015
T11ILA	0.918	0.527
T11IRA	0.985	0.896
T12SLA	0.83	0.246
T12SRA	0.914	0.509
T12ILA	0.86	0.324
T12IRA	0.994	0.957
L1SLA	0.959	0.748
L1SRA	0.803	0.216
L1ILA	0.69	0.062
L1IRA	0.825	0.237
L2SLA	0.481	0.006
L2SRA	0.832	0.275
L2ILA	0.466	0.015
L2IRA	0.738	0.162
L3SLA	0.854	0.307
L3SRA	0.631	0.032
L3ILA	0.898	0.446
L3IRA	0.897	0.442
L4SLA	0.987	0.907
L4SRA	0.852	0.302
L4ILA	1	1
L4IRA	0.803	0.214

Appendix A—2 One-way repeated measures ANOVA facet angle continued

Appendix A Measure of intra-observer agreement

Facet	Wilks' Lambda	Р
L5SLA	0.834	0.307
L5SRA	0.996	0.974
L5ILA	0.932	0.632
L5IRA	0.523	0.015

Appendix A—2 One-way repeated measures ANOVA facet angle continued

Vertebral Level	N	Measure of Agreement (κ value)	Level of Agreement (Landis and Koch, 1977).
C1LSscore	11	1	Perfect
C1RSscore	11	1	Perfect
C1LSextent	11	1	Perfect
C1RSextent	11	1	Perfect
C1LIscore	10	1	Perfect
C1RIscore	11	1	Perfect
C1Llextent	10	1	Perfect
C1Rlextent	10	1	Perfect
C2LSscore	13	1	Perfect
C2RSscore	13	1	Perfect
C2LSextent	13	0.833	Almost Perfect
C2RSextent	13	1	Perfect
C2LIscore	13	0.833	Almost Perfect
C2RIscore	13	1	Perfect
C2Llextent	13	1	Perfect
C2Rlextent	13	1	Perfect
C3LSscore	13	1	Perfect
C3RSscore	13	0.966	Almost Perfect
C3LSextent	13	1	Perfect
C3RSextent	13	0.947	Almost Perfect
C3Llscore	13	0.978	Almost Perfect
C3RIscore	13	1	Perfect
C3Llextent	13	1	Perfect
C3Rlextent	13	1	Perfect

Appendix A-3 Fleiss' Kappa Measure of Agreement for Eburnation

Appendix A Measure of intra-observer agreement

Appendix A-2 Fleiss' kappa measure of agreement for eburnation continued

Vertebral Level	Ν	Measure of Agreement (к value)	Level of Agreement (Landis and Koch, 1977).
C4LSscore	12	1	Perfect
C4Rsscore	12	1	Perfect
C4Lsextent	12	1	Perfect
C4Rsextent	12	1	Perfect
C4Liscore	12	0.902	Almost Perfect
C4Riscore	12	0.947	Almost Perfect
C4Liextent	12	0.928	Almost Perfect
C4Riextent	11	0.949	Almost Perfect
C5Lsscore	14	0.954	Almost Perfect
C5Rsscore	14	1	Perfect
C5Lsextent	14	0.948	Almost Perfect
C5Rsextent	14	1	Perfect
C5Liscore	14	1	Perfect
C5Riscore	14	1	Perfect
C5Liextent	14	1	Perfect
C5Riextent	14	1	Perfect
C6Lsscore	13	1	Perfect
C6Rsscore	13	1	Perfect
C6Lsextent	13	1	Perfect
C6Rsextent	13	0.813	Almost Perfect
C6Liscore	13	0.928	Almost Perfect
C6Riscore	13	1	Perfect
C6Liextent	13	1	Perfect
C6Riextent	13	0.838	Almost Perfect

Vertebral Level	Ν	Measure of Agreement (κ value)	Level of Agreement (Landis and Koch, 1977).
C7Lsscore	12	1	Perfect
C7Rsscore	11	1	Perfect
C7Lsextent	12	1	Perfect
C7Rsextent	11	1	Perfect
C7Liscore	13	0.894	Almost Perfect
C7Riscore	12	0.959	Almost Perfect
C7Liextent	12	1	Perfect
C7Riextent	13	1	Perfect
T1Lsscore	14	0.96	Almost Perfect
T1Rsscore	14	0.958	Almost Perfect
T1LSExtent	14	0.588	Moderate
T1RSExtent	14	1	Perfect
T1Liscore	14	1	Perfect
T1Riscore	14	1	Perfect
T1LIExtent	14	0.889	Almost Perfect
T1RIExtent	14	1	Perfect
T2Lsscore	16	1	Perfect
T2Rsscore	16	0.963	Almost Perfect
T2LSExtent	16	1	Perfect
T2RSExtent	16	0.952	Almost Perfect
T2Liscore	16	1	Perfect
T2Riscore	16	0.971	Almost Perfect
T2LISExtent	16	1	Perfect
T2RIExtent	16	1	Perfect

Appendix A-3 Fleiss' kappa measure of agreement for eburnation continued

Ar	ppendix	A-3	Fleiss'	kappa	measure	of	agreement	for	eburnation	continued
ጉኮ	penaix	73	110133	Kuppu	measure	01	ugreement		countration	continucu

Vertebral Level	Ν	Measure of Agreement (к value)	Level of Agreement (Landis and Koch, 1977).
T3LSscore	17	1	Perfect
T3RSscore	17	1	Perfect
T3LSExtent	17	1	Perfect
T3RSExtent	17	1	Perfect
T3LIscore	17	0.912	Almost Perfect
T3RIscore	17	0.885	Almost Perfect
T3LIExtent	17	0.912	Almost Perfect
T3RIExtent	17	1	Perfect
T4LSscore	18	0.962	Almost Perfect
T4RSscore	18	1	Perfect
T4LSExtent	18	0.634	Substantial
T4RSExtent	18	0.968	Almost Perfect
T4LIscore	18	1	Perfect
T4RIscore	18	0.976	Almost Perfect
T4LIExtent	18	1	Perfect
T4RIExtent	18	0.961	Almost Perfect
T5LSscore	17	0.73	Substantial
T5RSscore	17	0.95	Almost Perfect
T5LSExtent	17	0.927	Almost Perfect
T5RSExtent	17	0.954	Almost Perfect
T5Llscore	17	1	Perfect
T5RIscore	13	1	Perfect
T5LIExtent	16	1	Perfect

Vertebral Level	Ν	Measure of Agreement (κ value)	Level of Agreement (Landis and Koch, 1977).
T6LSscore	16	1	Perfect
T6RSscore	17	0.952	Almost Perfect
T6LSExtent	17	1	Perfect
T6RSExtent	17	0.844	Almost Perfect
T6LIscore	16	1	Perfect
T6RIscore	16	0.882	Almost Perfect
T6LIExtent	16	1	Perfect
T6RIExtent	16	1	Perfect
T7LSscore	18	1	Perfect
T7RSscore	18	1	Perfect
T7LSExtent	18	1	Perfect
T7RSExtent	18	0.943	Almost Perfect
T7LIscore	17	0.742	Substantial
T7RIscore	18	0.955	Almost Perfect
T7LIExtent	17	1	Perfect
T7RIExtent	18	1	Perfect
T8LSscore	18	0.971	Almost Perfect
T8RSscore	17	0.661	Substantial
T8LSExtent	18	1	Perfect
T8RSExtent	18	1	Perfect
T8Llscore	18	1	Perfect
T8RIscore	18	0.966	Almost Perfect
TRUEvtont	19	0.026	Almost Perfect
T&RIEvtent	18	0.320	Almost Perfect

Appendix A-3 Fleiss' kappa measure of agreement for eburnation continued

Appendix A-3 Fleiss	' kappa measure	of agreement fo	r eburnation continued
---------------------	-----------------	-----------------	------------------------

Vertebral Level	Ν	Measure of Agreement (κ value)	Level of Agreement (Landis and Koch, 1977).
T9LSscore	20	0.949	Almost Perfect
T9RSscore	20	0.954	Almost Perfect
T9LSExtent	20	0.844	Almost Perfect
T9RSExtent	20	1	Perfect
T9LIscore	20	0.846	Almost Perfect
T9RIscore	20	0.913	Almost Perfect
T9LIExtent	19	0.907	Almost Perfect
T9RIExtent	19	1	Perfect
T10LSscore	20	0.966	Almost Perfect
T10RSscore	20	0.928	Almost Perfect
T10LSExtent	20	0.936	Almost Perfect
T10RSExtent	20	0.939	Almost Perfect
T10Llscore	20	0.964	Almost Perfect
T10RIscore	20	1	Perfect
T10LIExtent	20	0.942	Almost Perfect
T10RIExtent	20	0.979	Almost Perfect
T11LSscore	19	0.982	Almost Perfect
T11RSscore	19	0.952	Almost Perfect
T11LSExtent	19	1	Perfect
T11RSExtent	19	1	Perfect
T11Llscore	18	0.98	Almost Perfect
T11Rlscore	19	1	Perfect
T11LIExtent	19	0.918	Almost Perfect
T11RIExtent	19	1	Perfect

Vertebral Level	Ν	Measure of Agreement (κ value)	Level of Agreement (Landis and Koch, 1977).
T12LSscore	19	0.977	Almost Perfect
T12RSscore	19	0.953	Almost Perfect
T12LSExtent	19	0.954	Almost Perfect
T12RSExtent	19	0.975	Almost Perfect
T12LIscore	18	1	Perfect
T12RIscore	18	0.833	Almost Perfect
T12LIExtent	18	0.934	Almost Perfect
T12RIExtent	19	1	Perfect
L1LSscore	18	0.927	Almost Perfect
L1RSscore	18	0.914	Almost Perfect
L1LSextent	18	0.949	Almost Perfect
L1RSextent	18	0.969	Almost Perfect
L1LIscore	19	0.975	Almost Perfect
L1RIscore	19	0.933	Almost Perfect
L1Llextent	19	0.937	Almost Perfect
L1Rlextent	19	1	Perfect
L2LSscore	18	0.955	Almost Perfect
L2RSscore	18	0.98	Almost Perfect
L2LSextent	18	0.961	Almost Perfect
L2RSextent	18	0.981	Almost Perfect
L2LIscore	16	1	Perfect
L2RIscore	16	1	Perfect
L2Llextent	16	1	Perfect
L2Rlextent	16	0.966	Almost Perfect

Appendix A-3 Fleiss' kappa measure of agreement for eburnation continued

Appendix A Measure of intra-observer agreement

Appendix A-3 Fleiss	' kappa measure	of agreement fo	r eburnation	continued
---------------------	-----------------	-----------------	--------------	-----------

Vertebral Level	N	Measure of Agreement (κ value)	Level of Agreement (Landis and Koch, 1977).
L3LSscore	19	0.95	Almost Perfect
L3RSscore	19	0.927	Almost Perfect
L3LSextent	19	0.94	Almost Perfect
L3RSextent	19	0.984	Almost Perfect
L3LIscore	19	0.925	Almost Perfect
L3RIscore	19	1	Perfect
L3Llextent	19	0.964	Almost Perfect
L3Rlextent	19	0.91	Almost Perfect
L4LSscore	19	0.883	Almost Perfect
L4RSscore	19	0.97	Almost Perfect
L4LSextent	19	0.935	Almost Perfect
L4RSextent	19	0.959	Almost Perfect
L4LIscore	19	1	Perfect
L4RIscore	19	1	Perfect
L4Llextent	19	0.928	Almost Perfect
L4Rlextent	18	0.96	Almost Perfect
L5LSscore	17	0.944	Almost Perfect
L5RSscore	17	0.907	Almost Perfect
L5LSextent	17	0.984	Almost Perfect
L5RSextent	17	0.985	Almost Perfect
L5LIscore	17	0.986	Almost Perfect
L5RIscore	7	0.983	Almost Perfect
L5Llextent	17	0.908	Almost Perfect
L5Rlextent	17	0.984	Almost Perfect

Vertebral	N	Measure of Agreement (κ	Level of Agreement (Landis and Koch,
		value)	1977).
C1LSscore	12	0.933	Almost Perfect
C1RSscore	12	0.936	Almost Perfect
C1LSextent	12	1	Perfect
C1RSextent	12	0.894	Almost Perfect
C1Llscore	11	0.93	Almost Perfect
C1RIscore	12	0.966	Almost Perfect
C1Llextent	11	0.529	Moderate
C1Rlextent	12	0.953	Almost Perfect
C2LSscore	13	0.952	Almost Perfect
C2RSscore	13	0.968	Almost Perfect
C2LSextent	13	0.935	Almost Perfect
C2RSextent	13	1	Perfect
C2LIscore	13	0.967	Almost Perfect
C2RIscore	13	0.888	Almost Perfect
C2Llextent	13	1	Perfect
C2Rlextent	13	0.943	Almost Perfect
C3LSscore	13	0.967	Almost Perfect
C3RSscore	13	0.919	Almost Perfect
C3LSextent	13	0.932	Almost Perfect
C3RSextent	13	0.98	Almost Perfect
C3Llscore	13	0.942	Almost Perfect
C3RIscore	13	0.822	Almost Perfect
C3Llextent	13	0.877	Almost Perfect
C3Rlextent	13	0.973	Almost Perfect
C4LSscore	12	0.607	Substantial
C4RSscore	12	0.955	Almost Perfect
C4LSextent	12	0.975	Almost Perfect
C4RSextent	12	0.968	Almost Perfect
C4LIscore	12	0.875	Almost Perfect
C4RIscore	12	1	Perfect
C4Llextent	12	0.909	Almost Perfect
C4Rlextent	12	0.955	Almost Perfect
C5LSscore	14	0.941	Almost Perfect
C5RSscore	13	0.957	Almost Perfect
C5LSextent	14	0.9	Almost Perfect
C5RSextent	14	0.944	Almost Perfect
C5Llscore	14	0.925	Almost Perfect
C5RIscore	14	0.974	Almost Perfect
C5Llextent	14	0.933	Almost Perfect
C5Rlextent	14	0.931	Almost Perfect

Appendix A-3 Fleiss' kappa measure of agreement for pitting

Appendix A-4 Fleiss' kappa measure of agreement for pitting continued

Vortobral		Measure of	Level of Agreement
level	N	Agreement (κ	(Landis and Koch,
		value)	1977).
C6LSscore	13	1	Perfect
C6RSscore	13	0.935	Almost Perfect
C6LSextent	13	0.982	Almost Perfect
C6RSextent	13	1	Perfect
C6LIscore	13	0.91	Almost Perfect
C6RIscore	13	0.858	Almost Perfect
C6Llextent	13	0.748	Substantial
C6Rlextent	13	0.981	Almost Perfect
C7LSscore	13	0.942	Almost Perfect
C7RSscore	11	0.946	Almost Perfect
C7LSextent	12	0.913	Almost Perfect
C7RSextent	11	0.888	Almost Perfect
C7Llscore	12	0.953	Almost Perfect
C7RIscore	12	0.944	Almost Perfect
C7Llextent	12	0.981	Almost Perfect
C7Rlextent	12	0.981	Almost Perfect
T1LSscore	15	1	Perfect
T1RSscore	15	0.964	Almost Perfect
T1LSExtent	15	0.985	Almost Perfect
T1RSExtent	15	0.982	Almost Perfect
T1LIscore	14	1	Perfect
T1RIscore	14	0.923	Almost Perfect
T1LIExtent	14	0.925	Almost Perfect
T1RIExtent	14	0.97	Almost Perfect
T2LSscore	16	1	Perfect
T2RSscore	16	1	Perfect
T2LSExtent	16	0.985	Almost Perfect
T2RSExtent	16	1	Perfect
T2LIscore	16	1	Perfect
T2RIscore	15	0 982	Almost Perfect
T2LISExtent	16	1	Perfect
T2BIExtent	16	0.986	Almost Perfect
	10	1	Perfect
	17	0.973	Almost Perfect
TRISEvtont	17	0.575	Almost Perfect
T2DCEvtont	17	0.97	Almost Porfact
	16	0.972	Almost Perfect
Tables	10	0.992	Almost Perfect
TOULS	1/	0.989	Almost Perfect
T3LIExtent	17	0.991	Almost Perfect
T3RIExtent	17	0.987	Almost Perfect

Vertebral		Measure of	Level of Agreement
Level	N	Agreement (κ	(Landis and Koch,
	18		1977). Almost Perfect
TAPScore	10	0.55	Almost Perfect
TALSEvtont	10	0.884	Almost Perfect
TADSExtent	10	0.964	Almost Perfect
TALLESEN	18	0.934	Almost Perfect
14Liscore	18	0.955	Almost Perfect
14Riscore	18	0.984	Almost Perfect
14LIExtent	18	0.948	Almost Perfect
T4RIExtent	18	0.876	Almost Perfect
T5LSscore	18	1	Perfect
T5RSscore	18	0.93	Almost Perfect
T5LSExtent	18	0.973	Almost Perfect
T5RSExtent	18	0.966	Almost Perfect
T5LIscore	16	0.989	Almost Perfect
T5RIscore	17	0.99	Almost Perfect
T5LIExtent	17	1	Perfect
T5RIExtent	17	1	Perfect
T6LSscore	17	0.987	Almost Perfect
T6RSscore	17	0.968	Almost Perfect
T6LSExtent	17	0.91	Almost Perfect
T6RSExtent	17	0.931	Almost Perfect
T6LIscore	16	0.984	Almost Perfect
T6RIscore	16	0.953	Almost Perfect
T6LIExtent	16	0.974	Almost Perfect
T6RIExtent	16	0.54	Moderate
T7LSscore	18	0.973	Almost Perfect
T7RSscore	18	0.981	Almost Perfect
T7LSExtent	18	0.959	Almost Perfect
T7RSExtent	18	1	Perfect
T7LIscore	17	0.987	Almost Perfect
T7RIscore	18	0.992	Almost Perfect
T7LIExtent	17	0.859	Almost Perfect
T7RIExtent	18	0.979	Almost Perfect
T8LSscore	19	0.983	Almost Perfect
T8RSscore	19	0.991	Almost Perfect
T8I SExtent	19	0.981	Almost Perfect
TRRSEvtent	19	1	Perfect
TSI Iscore	19	 	Almost Perfect
T&RIscore	10	0.914	Almost Perfect
TONISCOLE	10	0.332	Almost Porfact
TODIEXLEIIL	19	0.978	Almost Perfect
ISKIEXtent	19	0.989	Almost Perfect

Appendix A-4 Fleiss' kappa measure of agreement for pitting continued

Appendix A-4 Fleiss' kappa measure of agreement for pitting continued

Vertebral Level	N	Measure of	Level of Agreement
		Agreement (κ	(Landis and Koch,
TOLCOOR	20	value)	1977).
TODSecore	20	0.967	Almost Perfect
T9RSscore	20	0.948	Almost Perfect
19LSExtent	20	0.975	Almost Perfect
19RSExtent	20	0.938	Almost Perfect
T9LIscore	20	0.896	Almost Perfect
T9RIscore	20	0.945	Almost Perfect
T9LIExtent	19	0.989	Almost Perfect
T9RIExtent	20	0.979	Almost Perfect
T10LSscore	19	0.993	Almost Perfect
T10RSscore	20	1	Perfect
T10LSExtent	20	0.951	Almost Perfect
T10RSExtent	20	0.953	Almost Perfect
T10LIscore	20	0.92	Almost Perfect
T10Rlscore	20	0.943	Almost Perfect
T10LIExtent	19	0.969	Almost Perfect
T10RIExtent	20	0.945	Almost Perfect
T11LSscore	19	0.923	Almost Perfect
T11RSscore	19	0.989	Almost Perfect
T11LSExtent	19	0.989	Almost Perfect
T11RSExtent	19	0.979	Almost Perfect
T11LIscore	19	0.923	Almost Perfect
T11Rlscore	19	0.989	Almost Perfect
T11LIExtent	19	0.831	Almost Perfect
T11RIExtent	19	0.908	Almost Perfect
T12LSscore	19	0.93	Almost Perfect
T12RSscore	19	0.919	Almost Perfect
T12LSExtent	19	0.959	Almost Perfect
T12RSExtent	19	0.971	Almost Perfect
T12LIscore	18	0.992	Almost Perfect
T12RIscore	19	0.967	Almost Perfect
T12LIExtent	18	0.939	Almost Perfect
T12RIExtent	19	0.847	Almost Perfect
L1LSscore	18	0.984	Almost Perfect
L1RSscore	18	1	Perfect
111 Sextent	18	0.962	Almost Perfect
L1RSextent	18	0.976	Almost Perfect
	19	0 992	Almost Perfect
11Riscore	19	0.332	Almost Perfect
111 lovtont	10	0.000	Almost Perfect
	19	0.921	Almost Perfect
LIKIextent	19	0.949	Almost Perfect

Vertebral		Measure of	Level of Agreement
Level	Ν	Agreement (κ	(Landis and Koch,
		value)	1977).
L2LSscore	18	0.991	Almost Perfect
L2RSscore	18	1	Perfect
L2LSextent	17	1	Perfect
L2RSextent	17	0.988	Almost Perfect
L2LIscore	16	0.991	Almost Perfect
L2RIscore	16	0.934	Almost Perfect
L2Llextent	16	0.943	Almost Perfect
L2Rlextent	16	1	Perfect
L3LSscore	19	0.985	Almost Perfect
L3RSscore	19	0.991	Almost Perfect
L3LSextent	19	0.956	Almost Perfect
L3RSextent	19	0.988	Almost Perfect
L3LIscore	19	0.888	Almost Perfect
L3RIscore	19	0.919	Almost Perfect
L3Llextent	19	0.987	Almost Perfect
L3Rlextent	19	0.919	Almost Perfect
L4LSscore	19	0.987	Almost Perfect
L4RSscore	19	0.948	Almost Perfect
L4LSextent	19	0.993	Almost Perfect
L4RSextent	19	0.965	Almost Perfect
L4LIscore	19	0.989	Almost Perfect
L4RIscore	19	0.984	Almost Perfect
L4Llextent	19	0.98	Almost Perfect
L4Rlextent	18	0.993	Almost Perfect
L5LSscore	19	0.967	Almost Perfect
L5RSscore	18	0.987	Almost Perfect
L5LSextent	17	0.925	Almost Perfect
L5RSextent	17	1	Perfect
L5LIscore	17	0.988	Almost Perfect
L5RIscore	17	0.986	Almost Perfect
L5Llextent	17	0.974	Almost Perfect
L5Rlextent	17	0.953	Almost Perfect

Appendix A-4 Fleiss' kappa measure of agreement for pitting continued

Appendix A Measure of intra-observer agreement

Appendix A-4 Fleiss' kappa measure of agreement for osteophytes

Vertebral		Measure of	Level of Agreement
Level	N	Agreement (κ	(Landis and Koch,
		value)	1977).
C1LSscore	11	1	Perfect
C1RSscore	11	1	Perfect
C1LSextent	11	1	Perfect
C1RSextent	11	1	Perfect
C1Llscore	10	1	Perfect
C1RIscore	11	1	Perfect
C1Llextent	10	1	Perfect
C1Rlextent	10	1	Perfect
C2LSscore	13	1	Perfect
C2RSscore	13	1	Perfect
C2LSextent	13	0.833	Almost Perfect
C2RSextent	13	1	Perfect
C2LIscore	13	0.833	Almost Perfect
C2RIscore	13	1	Perfect
C2Llextent	13	1	Perfect
C2Rlextent	13	1	Perfect
C3LSscore	13	1	Perfect
C3RSscore	13	0.966	Almost Perfect
C3LSextent	13	1	Perfect
C3RSextent	13	0.947	Almost Perfect
C3Llscore	13	0.978	Almost Perfect
C3RIscore	13	1	Perfect
C3Llextent	13	1	Perfect
C3Rlextent	13	1	Perfect
C4LSscore	12	1	Perfect
C4RSscore	12	1	Perfect
C4LSextent	12	1	Perfect
C4RSextent	12	1	Perfect
C4LIscore	12	0.902	Almost Perfect
C4RIscore	12	0.947	Almost Perfect
CAllextent	12	0.928	Almost Perfect
C4Rlextent	11	0.949	Almost Perfect
C5I Sscore	14	0.954	Almost Perfect
C5RSscore	1/	1	Perfect
(5) Sextent	14	n 948	Almost Perfect
CSRSevtent	1/	1	Derfect
CEllecoro	14	1	Derfect
CERICCORO	14	1	Borfoot
CELleutent	14	1	Periect
CSLiextent	14	1	Perfect
C5Rlextent	14	1	Perfect

		Measure of	Level of Agreement
Vertebral	N	Agreement (κ	(Landis and Koch,
Level		value)	1977).
C6LSscore	13	1	Perfect
C6RSscore	13	1	Perfect
C6LSextent	13	1	Perfect
C6RSextent	13	0.813	Almost Perfect
C6LIscore	13	0.928	Almost Perfect
C6RIscore	13	1	Perfect
C6Llextent	13	1	Perfect
C6Rlextent	13	0.838	Almost Perfect
C7LSscore	12	1	Perfect
C7RSscore	11	1	Perfect
C7LSextent	12	1	Perfect
C7RSextent	11	1	Perfect
C7LIscore	13	0.894	Almost Perfect
C7Rlscore	12	0.959	Almost Perfect
C7Llextent	12	1	Perfect
C7Rlextent	13	1	Perfect
T1LSscore	14	0.96	Almost Perfect
T1RSscore	14	0.958	Almost Perfect
T1LSExtent	14	0.588	Moderate
T1RSExtent	14	1	Perfect
T1LIscore	14	1	Perfect
T1RIscore	14	1	Perfect
T1LIExtent	14	0.889	Almost Perfect
T1RIExtent	14	1	Perfect
T2LSscore	16	1	Perfect
T2RSscore	16	0.963	Almost Perfect
T2LSExtent	16	1	Perfect
T2RSExtent	16	0.952	Almost Perfect
T2LIscore	16	1	Perfect
T2RIscore	16	0.971	Almost Perfect
T2LISExtent	16	1	Perfect
T2RIExtent	16	1	Perfect
T3LSscore	17	1	Perfect
T3RSscore	17	1	Perfect
T3LSExtent	17	1	Perfect
T3RSExtent	17	1	Perfect
T3LIscore	17	0.912	Almost Perfect
T3Rlscore	17	0.885	Almost Perfect
T3LIExtent	17	0.912	Almost Perfect
T3RIExtent	17	1	Perfect

Appendix A-5 Fleiss' kappa measure of agreement for osteophytes continued

Appendix A-5 Fleiss' kappa measure of agreement for osteophytes continued

		Measure of	Level of Agreement
Vertebral	N	Agreement (κ	(Landis and Koch,
Level		value)	1977).
T4LSscore	18	0.962	Almost Perfect
T4RSscore	18	1	Perfect
T4LSExtent	18	0.634	Substantial
T4RSExtent	18	0.968	Almost Perfect
T4LIscore	18	1	Perfect
T4RIscore	18	0.976	Almost Perfect
T4LIExtent	18	1	Perfect
T4RIExtent	18	0.961	Almost Perfect
T5LSscore	17	0.73	Substantial
T5RSscore	17	0.95	Almost Perfect
T5LSExtent	17	0.927	Almost Perfect
T5RSExtent	17	0.954	Almost Perfect
T5LIscore	17	1	Perfect
T5RIscore	13	1	Perfect
T5LIExtent	16	1	Perfect
T5RIExtent	16	1	Perfect
T6LSscore	16	1	Perfect
T6RSscore	17	0.952	Almost Perfect
T6LSExtent	17	1	Perfect
T6RSExtent	17	0.844	Almost Perfect
T6LIscore	16	1	Perfect
T6RIscore	16	0.882	Almost Perfect
T6LIExtent	16	1	Perfect
T6RIExtent	16	1	Perfect
T7LSscore	18	1	Perfect
T7RSscore	18	1	Perfect
T7LSExtent	18	1	Perfect
T7RSExtent	18	0.943	Almost Perfect
T7LIscore	17	0.742	Substantial
T7RIscore	18	0.955	Almost Perfect
T7LIExtent	17	1	Perfect
T7RIExtent	18	1	Perfect
T8LSscore	18	0.971	Almost Perfect
T8RSscore	17	0.661	Substantial
T8LSExtent	18	1	Perfect
T8RSExtent	18	1	Perfect
T8LIscore	18	1	Perfect
T8RIscore	18	0.966	Almost Perfect
T8LIExtent	18	0.926	Almost Perfect
T8RIFxtent	18	0.861	Almost Perfect
1011EAtent	-0	0.001	,

Vertebral		Measure of	Level of Agreement
Level	Ν	Agreement (κ	(Landis and Koch,
		value)	1977).
19LSscore	20	0.949	Almost Perfect
T9RSscore	20	0.954	Almost Perfect
T9LSExtent	20	0.844	Almost Perfect
T9RSExtent	20	1	Perfect
T9Llscore	20	0.846	Almost Perfect
T9RIscore	20	0.913	Almost Perfect
T9LIExtent	19	0.907	Almost Perfect
T9RIExtent	19	1	Perfect
T10LSscore	20	0.966	Almost Perfect
T10RSscore	20	0.928	Almost Perfect
T10LSExtent	20	0.936	Almost Perfect
T10RSExtent	20	0.939	Almost Perfect
T10LIscore	20	0.964	Almost Perfect
T10RIscore	20	1	Perfect
T10LIExtent	20	0.942	Almost Perfect
T10RIExtent	20	0.979	Almost Perfect
T11LSscore	19	0.982	Almost Perfect
T11RSscore	19	0.952	Almost Perfect
T11LSExtent	19	1	Perfect
T11RSExtent	19	1	Perfect
T11LIscore	18	0.98	Almost Perfect
T11RIscore	19	1	Perfect
T11LIExtent	19	0.918	Almost Perfect
T11RIExtent	19	1	Perfect
T12LSscore	19	0.977	Almost Perfect
T12RSscore	19	0.953	Almost Perfect
T12LSExtent	19	0.954	Almost Perfect
T12RSExtent	19	0.975	Almost Perfect
T12LIscore	18	1	Perfect
T12RIscore	18	0.833	Almost Perfect
T12LIExtent	18	0.934	Almost Perfect
T12RIExtent	19	1	Perfect
L1LSscore	18	0.927	Almost Perfect
L1RSscore	18	0.914	Almost Perfect
L1LSextent	18	0.949	Almost Perfect
L1RSextent	18	0.969	Almost Perfect
L1LIscore	19	0.975	Almost Perfect
L1RIscore	19	0.933	Almost Perfect
L1Llextent	19	0.937	Almost Perfect
L1Rlextent	19	1	Perfect
		-	

Appendix A-5 Fleiss' kappa measure of agreement for osteophytes continued

Appendix A-5 Fleiss'	kappa measure of	agreement for	osteophytes (continued
		-0		

Vertebral		Measure of	Level of Agreement
Level	N	Agreement (κ	(Landis and Koch,
	18	0.955	1977). Almost Perfect
	18	0.955	Almost Perfect
1 21 Sextent	18	0.961	Almost Perfect
L2ESextent	10	0.901	Almost Perfect
	10	0.981	Annost Perfect
L2LISCOTE	10	1	Perfect
L2Riscore	10	1	Perfect
L2Llextent	16	1	Perfect
L2Rlextent	16	0.966	Almost Perfect
L3LSscore	19	0.95	Almost Perfect
L3RSscore	19	0.927	Almost Perfect
L3LSextent	19	0.94	Almost Perfect
L3RSextent	19	0.984	Almost Perfect
L3LIscore	19	0.925	Almost Perfect
L3RIscore	19	1	Perfect
L3Llextent	19	0.964	Almost Perfect
L3Rlextent	19	0.91	Almost Perfect
L4LSscore	19	0.883	Almost Perfect
L4RSscore	19	0.97	Almost Perfect
L4LSextent	19	0.935	Almost Perfect
L4RSextent	19	0.959	Almost Perfect
L4LIscore	19	1	Perfect
L4RIscore	19	1	Perfect
L4Llextent	19	0.928	Almost Perfect
L4Rlextent	18	0.96	Almost Perfect
L5LSscore	17	0.944	Almost Perfect
L5RSscore	17	0.907	Almost Perfect
L5LSextent	17	0.984	Almost Perfect
L5RSextent	17	0.985	Almost Perfect
L5LIscore	17	0.986	Almost Perfect
L5Rlscore	7	0.983	Almost Perfect
L5Llextent	17	0.908	Almost Perfect
L5Rlextent	17	0.984	Almost Perfect

Appendix B. Results for Facet Size Analyses

Facet Level	Sample	Male				SDI		
		N	Mean	Std dev.	N	Mean	Std dev.	
C1LS	SBC	23	202.09	44.30	32	180.18	38.22	112.16
	SBL	26	206.79	31.34	20	178.35	36.15	115.95
	AS	17	214.66	46.08	10	206.16	35.58	104.12
C1RS	SBC	23	197.85	40.44	32	173.93	39.95	113.75
	SBL	27	207.96	30.28	20	171.05	23.18	121.58
	AS	19	208.42	33.00	10	205.80	36.33	101.27
C1LI	SBC	23	225.49	34.94	32	192.86	27.21	116.92
	SBL	26	222.74	31.19	20	178.90	27.07	124.51
	AS	15	229.23	28.26	9	199.03	51.18	115.17
C1RI	SBC	23	219.30	40.25	32	193.23	34.41	113.49
	SBL	27	222.37	25.87	20	189.45	27.93	117.38
	AS	17	224.00	33.31	10	219.27	19.16	102.16
C2LS	SBC	23	239.24	45.27	35	199.86	33.28	119.7
	SBL	29	234.41	39.40	20	183.83	20.70	127.52
	AS	14	235.17	20.69	14	199.04	45.25	118.15
C2RSA	SBC	23	220.12	46.59	35	187.70	37.14	117.27
	SBL	29	213.22	31.05	20	176.52	29.59	120.79
	AS	17	239.13	48.82	13	198.59	43.20	120.42
C2LI	SBC	22	140.74	74.59	35	97.33	33.51	114.6
	SBL	29	110.48	31.70	20	91.00	22.31	121.41
	AS	13	112.40	31.04	12	125.95	20.34	89.24
C2RI	SBC	22	119.31	30.56	35	98.55	18.71	121.06
	SBL	29	98.33	22.69	20	90.99	26.44	108.06
	AS	14	114.95	25.19	12	112.95	25.71	101.77
C3LS	SBC	19	128.45	71.27	32	98.34	41.27	130.61
	SBL	28	109.85	36.91	21	90.12	25.75	121.89
	AS	17	111.68	28.33	13	101.77	18.71	109.74

Appendix B Facet Size

Facet Level	Sample	Male Female			SDI			
		N	Mean	Std dev.	N	Mean	Std dev.	
C3RS	SBC	20	115.44	37.32	32	97.73	23.73	118.13
	SBL	28	103.48	22.43	20	94.88	32.23	109.06
	AS	18	114.77	32.57	13	106.10	19.45	108.17
C3LI	SBC	20	130.39	46.97	31	107.03	24.16	121.83
	SBL	27	108.74	32.26	21	101.03	23.68	107.63
	AS	17	129.04	19.73	13	133.85	65.72	96.41
C3RI	SBC	20	115.40	39.22	31	110.57	36.57	104.37
	SBL	28	104.47	24.90	20	103.12	32.39	101.31
	AS	15	131.48	23.81	13	112.71	19.59	116.65
C4LS	SBC	19	119.26	42.87	28	95.87	16.12	124.4
	SBL	29	110.48	37.30	22	92.42	19.89	119.54
	AS	15	117.87	20.49	12	105.33	17.04	111.91
C4RS	SBC	19	124.96	67.67	28	102.94	24.10	121.38
	SBL	29	105.56	28.79	22	98.69	30.70	106.96
	AS	17	115.11	25.61	13	119.67	16.71	96.19
C4LIA	SBC	20	114.13	27.61	29	111.14	35.81	102.69
	SBL	29	108.75	30.86	23	99.54	23.68	109.26
	AS	15	124.37	27.98	12	124.11	28.62	100.21
C4RI	SBC	20	117.72	52.02	29	107.11	32.80	109.91
	SBL	29	106.90	27.71	22	97.27	17.31	109.89
	AS	16	122.37	31.67	12	117.97	14.57	103.72
C5LS	SBC	23	113.58	31.41	34	99.36	33.11	114.31
	SBL	30	103.55	28.23	24	97.04	25.03	106.70
	AS	12	109.75	20.42	14	109.62	21.84	100.12
C5RS	SBC	24	121.68	62.85	34	104.63	31.15	116.29
	SBL	30	104.95	22.07	24	94.30	20.35	111.29
	AS	12	112.44	29.29	14	103.57	19.32	108.56
C55LI	SBC	24	102.58	42.68	34	91.47	19.48	112.14
	SBL	29	105.47	29.95	24	92.12	25.88	114.49
	AS	12	131.18	19.89	13	118.16	26.22	111.02

Facet Level	Sample		Male			Female	1	SDI
		N	Mean	Std dev.	N	Mean	Std dev.	
C5RI	SBC	24	108.90	56.73	34	96.02	18.99	113.41
	SBL	29	104.29	30.55	24	97.31	25.40	107.17
	AS	12	134.44	37.57	14	112.08	23.81	119.94
C6LS	SBC	20	104.96	40.24	37	91.44	18.33	114.79
	SBL	30	98.89	21.61	23	87.31	18.06	113.27
	AS	14	110.39	23.93	10	109.00	18.26	101.28
C6RS	SBC	20	107.16	50.81	37	89.81	21.39	119.32
	SBL	29	101.48	18.70	24	91.91	16.73	110.41
	AS	15	125.51	34.06	11	111.67	23.72	112.39
C6LIA	SBC	20	109.11	26.92	37	92.39	20.36	118.1
	SBL	31	98.40	22.07	23	86.47	25.26	113.80
	AS	14	157.35	61.12	11	130.38	19.50	120.68
C6RI	SBC	20	73.99	24.29	37	71.38	18.71	103.62
	SBL	30	75.04	17.26	23	66.69	21.81	112.52
	AS	14	114.58	36.17	11	94.48	17.09	121.27
C7LS	SBC	23	92.10	23.49	39	92.40	25.44	99.67
	SBL	32	104.39	30.04	24	86.03	23.97	121.34
	AS	13	138.10	71.29	7	110.21	32.41	125.31
C7RSA	SBC	24	99.46	21.77	38	95.46	22.89	99.67
	SBL	32	106.99	26.64	24	87.77	22.08	121.90
	AS	15	130.39	37.06	8	117.71	27.09	110.77
C7LIA	SBC	25	129.75	32.80	39	114.19	26.51	104.18
	SBL	32	125.86	44.15	24	103.75	35.12	121.32
	AS	15	144.45	34.82	7	135.06	17.03	106.95
C7RI	SBC	25	138.09	37.06	39	114.30	30.43	113.62
	SBL	32	125.11	32.25	24	112.03	27.94	111.68
	AS	16	147.01	40.29	8	147.70	27.89	99.53
T1LS	SBC	23	120.36	41.19	39	118.66	31.75	120.82
	SBL	34	127.79	34.83	22	117.40	42.57	108.85
	AS	15	134.18	44.03	12	129.03	34.29	103.99

Appendix B Facet Size

Facet Level	Sample		Male	1		Female	1	SDI
		N	Mean	Std dev.	N	Mean	Std dev.	
T1RS	SBC	24	133.02	38.17	40	128.18	38.65	101.43
	SBL	34	132.73	34.30	23	117.95	29.82	112.53
	AS	17	134.02	33.87	13	128.52	25.08	104.28
T1LI	SBC	24	110.50	27.31	40	100.96	22.75	103.77
	SBL	35	123.53	33.18	23	98.64	28.61	125.23
	AS	19	133.52	50.36	13	124.50	20.63	107.24
T1RI	SBC	24	112.41	31.28	40	104.02	21.75	109.44
	SBL	35	121.49	31.41	23	98.81	24.42	122.95
	AS	17	136.47	49.44	13	122.13	29.46	111.74
T2LS	SBC	26	112.37	21.40	38	101.78	20.06	108.06
	SBL	34	116.43	27.85	22	92.12	22.43	126.38
	AS	20	126.37	53.20	17	106.13	23.21	119.07
T2RS	SBC	26	115.39	29.44	38	103.75	19.65	110.4
	SBL	35	115.00	25.13	22	94.21	25.10	122.06
	AS	22	126.57	55.86	17	120.04	23.21	105.44
T2LI	SBC	26	109.93	16.43	38	100.88	28.61	111.22
	SBL	34	118.29	30.25	22	92.91	24.66	127.32
	AS	23	115.09	40.86	15	96.04	18.94	119.83
T2RI	SBC	26	111.94	24.83	38	94.20	18.04	108.96
	SBL	34	113.10	28.23	22	99.44	21.51	113.74
	AS	22	112.18	21.28	15	110.33	36.27	101.68
T3LS	SBC	25	99.05	15.53	39	92.47	26.00	118.83
	SBL	29	108.09	30.26	21	82.87	29.13	130.44
	AS	20	110.19	27.09	16	94.31	17.32	116.83
T3RS	SBC	25	104.76	22.58	39	90.66	18.61	107.12
	SBL	30	107.56	30.11	21	92.10	23.35	116.79
	AS	23	109.61	21.32	17	108.16	32.26	101.3
T3LI	SBC	25	103.97	20.96	39	84.27	17.14	115.54
	SBL	30	110.46	27.45	21	88.56	29.16	124.72
	AS	22	105.40	18.43	14	83.84	16.35	125.71

Facet Level	Sample		Male			Female	1	SDI
		N	Mean	Std dev.	N	Mean	Std dev.	
T3RI	SBC	25	111.93	38.76	39	94.82	29.61	123.37
	SBL	30	117.76	37.42	21	102.85	39.56	114.50
	AS	22	111.08	28.86	16	111.91	27.64	99.25
T4LS	SBC	26	98.85	24.70	37	78.19	17.56	118.04
	SBL	30	105.74	26.90	19	79.26	20.49	133.42
	AS	23	107.92	21.87	18	90.10	19.61	119.78
T4RS	SBC	26	113.11	44.46	36	93.02	28.46	126.42
	SBL	29	104.67	24.43	19	88.18	29.14	118.69
	AS	23	108.18	20.82	18	100.34	22.99	107.82
T4LI	SBC	26	105.87	34.14	36	79.78	16.87	121.59
	SBL	29	101.18	24.83	18	84.06	24.58	120.36
	AS	22	109.96	18.74	16	90.99	18.72	120.86
T4RI	SBC	23	92.34	16.81	25	81.03	17.76	132.7
	SBL	20	97.43	17.18	12	92.17	17.67	105.70
	AS	16	95.24	14.11	12	99.11	13.61	96.10
T5LS	SBC	25	103.35	35.32	37	73.85	17.05	113.96
	SBL	28	89.89	20.09	20	81.75	23.70	109.95
	AS	21	104.77	20.81	17	94.15	20.59	111.28
T5RS	SBC	25	109.19	23.22	37	88.56	24.92	139.95
	SBL	28	109.76	28.91	20	101.88	42.95	107.74
	AS	22	111.73	29.00	17	106.50	25.55	104.91
T5LI	SBC	25	90.35	16.73	36	76.03	22.20	123.29
	SBL	26	99.54	25.86	20	87.08	25.24	114.32
	AS	19	101.72	26.57	15	88.75	21.41	114.62
T5RI	SBC	25	97.27	27.93	36	79.70	24.29	118.84
	SBL	28	101.70	29.55	20	85.28	23.20	119.25
	AS	20	124.62	36.53	15	100.48	26.65	124.02
T6LS	SBC	26	88.53	20.97	36	74.69	21.21	122.04
	SBL	31	96.11	22.65	21	80.49	23.77	119.41
	AS	18	95.78	26.33	14	83.90	12.41	114.15

Appendix B Facet Size

Facet Level	Sample		Male	1		SDI		
		N	Mean	Std dev.	N	Mean	Std dev.	
T6RS	SBC	26	88.79	22.90	36	80.37	19.59	118.53
	SBL	31	101.04	28.11	21	84.77	26.15	119.19
	AS	18	115.16	30.08	14	90.29	19.00	127.54
T6LI	SBC	26	95.87	19.11	38	73.31	17.82	110.47
	SBL	29	94.17	19.50	21	74.64	18.90	126.16
	AS	17	105.38	30.78	14	89.19	25.87	118.16
T6RI	SBC	26	88.34	14.98	38	73.55	21.57	130.77
	SBL	28	85.59	14.58	21	68.51	16.80	124.94
	AS	17	97.36	20.75	14	85.92	16.79	113.32
T7LS	SBC	26	93.50	18.05	38	74.41	17.56	120.11
	SBL	31	89.02	18.77	20	69.39	14.28	128.29
	AS	16	98.69	19.13	17	85.90	15.59	114.89
T7RS	SBC	26	92.57	22.98	38	74.12	15.57	125.66
	SBL	32	87.08	19.39	19	68.09	13.16	127.89
	AS	16	98.69	19.13	17	85.90	15.59	114.89
T7LI	SBC	25	90.68	15.13	38	74.30	16.77	124.89
	SBL	30	92.53	18.76	20	75.34	14.25	122.81
	AS	14	94.86	24.11	17	95.71	20.55	99.11
T7RI	SBC	25	81.29	12.65	38	70.91	16.55	122.04
	SBL	31	85.12	13.35	20	64.71	13.72	131.55
	AS	14	94.62	21.05	15	97.30	16.64	97.25
T8LS	SBC	25	89.17	21.24	36	76.55	14.58	114.64
	SBL	33	87.37	15.30	20	66.88	12.71	130.64
	AS	17	102.80	21.35	17	93.52	18.13	109.92
T8RS	SBC	25	168.50	34.07	36	145.94	30.25	116.49
	SBL	33	171.15	33.86	20	130.95	24.43	130.70
	AS	18	98.54	24.31	18	101.09	23.15	97.48
T8LI	SBC	26	101.10	26.36	36	81.98	27.26	115.46
	SBL	33	100.90	21.00	18	77.80	16.85	129.69
	AS	18	110.11	26.41	16	101.65	16.07	108.32

Facet Level	Sample		Male			Female	1	SDI
		N	Mean	Std dev.	N	Mean	Std dev.	
T8RI	SBC	26	95.65	28.70	36	77.87	23.76	123.32
	SBL	33	87.19	16.94	19	72.99	14.00	119.45
	AS	18	110.11	26.41	16	101.65	16.07	108.32
T9LS	SBC	26	96.97	28.76	39	84.26	22.10	122.84
	SBL	31	91.89	22.06	20	70.29	16.54	130.74
	AS	18	107.93	26.28	18	100.30	18.86	107.61
T9RS	SBC	26	101.52	32.13	39	81.16	21.71	115.08
	SBL	31	88.82	19.39	20	72.09	14.28	123.21
	AS	18	98.54	24.31	18	101.09	23.15	97.48
T9LI	SBC	25	114.89	30.84	39	95.32	25.87	120.53
	SBL	31	110.61	31.45	20	84.37	20.10	131.11
	AS	17	114.21	21.67	16	101.90	23.66	112.08
T9RI	SBC	25	107.37	29.24	38	83.91	16.06	127.95
	SBL	29	97.58	27.22	20	76.24	15.06	127.98
	AS	16	122.83	32.48	16	96.33	21.95	127.51
T10LS	SBC	25	100.16	17.50	39	87.92	22.81	113.93
	SBL	30	98.30	19.87	20	82.24	16.63	119.54
	AS	19	109.22	19.32	14	108.06	21.97	101.07
T10RS	SBC	25	102.09	21.71	39	83.99	21.23	121.56
	SBL	31	92.54	21.14	20	76.18	18.25	121.47
	AS	18	115.59	18.77	14	106.65	23.93	108.38
T10LI	SBC	26	133.17	38.92	39	96.97	31.50	137.33
	SBL	30	128.37	43.06	20	94.21	20.79	136.26
	AS	19	125.63	24.94	14	109.26	13.50	114.99
T10RI	SBC	26	114.58	26.40	39	89.16	29.91	128.51
	SBL	30	111.57	36.34	20	76.97	20.03	144.95
	AS	19	123.34	20.31	14	127.06	40.96	97.07
T11LS	SBC	26	128.99	31.72	39	106.80	34.30	120.77
	SBL	28	126.91	38.58	20	92.51	22.21	137.19
	AS	15	130.87	31.58	14	112.81	18.00	116.01

Appendix B Facet Size

Facet Level	Sample	Male Female			SDI			
		N	Mean	Std dev.	N	Mean	Std dev.	
T11RS	SBC	26	105.40	30.50	39	84.55	17.06	124.66
	SBL	29	105.06	24.38	21	81.78	20.13	128.45
	AS	19	126.90	26.35	15	127.90	49.44	99.22
T11LI	SBC	25	108.36	29.19	39	88.63	22.42	122.27
	SBL	28	107.76	32.17	21	84.99	28.52	126.79
	AS	18	134.07	43.90	15	108.05	19.53	124.08
T11RI	SBC	25	113.55	32.81	39	84.04	31.72	135.11
	SBL	29	104.13	30.09	21	89.62	36.16	116.19
	AS	16	137.05	39.40	12	110.24	20.19	124.31
T12LS	SBC	24	117.83	40.44	37	92.22	25.89	127.76
	SBL	28	105.33	42.62	21	75.81	21.49	138.93
	AS	14	121.71	42.54	15	114.98	25.63	105.85
T12RS	SBC	24	117.64	42.12	37	92.79	33.47	126.79
	SBL	29	106.18	32.52	21	86.66	32.01	122.53
	AS	14	122.37	35.69	15	102.75	33.80	119.09
T12LI	SBC	25	111.30	30.54	37	94.31	25.80	118.01
	SBL	29	113.00	29.86	21	90.82	21.99	124.42
	AS	14	128.58	42.48	12	115.37	23.11	111.44
T12RI	SBC	25	108.05	23.54	37	98.30	30.75	109.92
	SBL	29	108.94	35.19	21	92.76	18.95	117.44
	AS	13	118.37	32.10	14	109.70	32.89	107.90
L1LS	SBC	25	151.06	40.49	36	130.01	43.59	116.19
	SBL	28	144.23	42.79	20	118.25	33.27	121.96
	AS	16	150.38	54.87	17	152.48	30.03	98.62
L1RS	SBC	25	162.70	38.68	35	129.40	51.59	125.73
	SBL	28	146.83	39.26	20	134.75	55.59	108.97
	AS	15	162.83	61.81	15	161.05	41.82	101.11
L1LI	SBC	25	144.31	32.60	36	121.38	32.29	118.89
	SBL	29	149.42	42.85	20	119.30	27.88	125.25
	AS	15	149.24	39.90	17	138.30	25.71	107.91

Facet Level	Sample		Male			Female	1	SDI
		N	Mean	Std dev.	N	Mean	Std dev.	
L1RI	SBC	25	139.91	33.86	35	125.78	39.38	111.23
	SBL	28	154.48	41.60	20	122.03	26.07	126.59
	AS	16	156.80	40.72	16	145.01	42.54	108.13
L2LS	SBC	26	187.56	46.80	36	150.65	46.48	124.5
	SBL	29	194.95	64.68	20	139.18	26.12	140.07
	AS	18	199.63	40.34	14	184.48	29.27	108.21
L2RS	SBC	26	192.53	43.52	36	160.01	48.60	120.33
	SBL	28	199.07	60.52	18	156.48	38.45	127.22
	AS	18	209.62	52.28	14	188.30	39.64	111.32
L2LI	SBC	26	172.98	37.88	36	136.62	29.67	126.61
	SBL	30	169.38	42.85	19	137.72	25.23	122.99
	AS	17	179.75	40.64	14	162.01	45.09	110.95
L2RI	SBC	26	173.07	32.68	36	142.47	30.94	121.48
	SBL	30	171.70	47.76	19	142.83	32.35	120.21
	AS	17	164.97	44.65	15	169.26	33.14	97.47
L3LS	SBC	26	208.48	53.73	37	175.28	45.71	118.94
	SBL	26	206.86	58.91	20	169.58	42.95	121.98
	AS	17	214.49	63.05	19	202.88	47.66	105.72
L3RS	SBC	26	209.99	49.50	35	175.77	42.51	119.46
	SBL	25	225.62	70.20	18	167.62	43.18	134.60
	AS	15	222.04	77.23	14	221.91	46.93	100.06
L3LI	SBC	26	175.96	45.36	37	147.78	34.05	119.07
	SBL	27	179.27	30.29	21	144.52	33.88	124.05
	AS	18	187.31	43.08	18	155.90	35.02	120.15
L3RI	SBC	26	165.86	41.00	37	150.59	33.79	110.14
	SBL	26	183.41	39.79	21	157.24	37.63	116.64
	AS	16	196.92	44.35	17	161.59	29.66	121.87
L4LS	SBC	26	221.07	43.82	38	189.91	49.95	116.41
	SBL	25	223.99	44.75	15	178.61	38.64	125.41
	AS	19	250.43	55.28	17	228.04	66.60	109.82

Appendix B Facet Size

Appendix B-1 Descriptive statistics and SDI for facet size continue	Appendix B-1	Descriptive	statistics an	d SDI for f	acet size	continued
---	--------------	-------------	---------------	-------------	-----------	-----------

Facet Level	Sample	Male				SDI		
		N	Mean	Std dev.	N	Mean	Std dev.	
L4RS	SBC	26	230.80	54.88	38	202.90	56.61	113.75
	SBL	26	229.59	54.27	15	197.21	69.40	116.42
	AS	20	243.20	47.13	17	223.09	51.63	109.01
L4LI	SBC	26	194.60	45.90	38	160.06	42.95	121.58
	SBL	25	200.33	50.49	16	159.54	64.56	125.57
	AS	19	201.84	30.04	15	200.17	49.06	100.83
L4RI	SBC	26	186.89	48.95	38	169.98	49.49	109.95
	SBL	26	191.36	51.88	16	162.69	55.93	117.62
	AS	18	196.90	37.27	15	196.72	44.58	100.09
L5LS	SBC	25	240.03	60.15	34	194.06	44.87	123.69
	SBL	29	254.41	56.03	19	192.10	41.35	132.43
	AS	17	263.92	48.34	15	250.05	54.81	105.55
L5RS	SBC	25	240.15	48.60	34	204.17	42.21	117.62
	SBL	27	243.36	60.64	19	208.01	79.35	117.00
	AS	16	272.32	48.95	15	242.60	62.01	112.25
L5LI	SBC	25	224.41	58.71	34	186.29	55.61	120.46
	SBL	29	213.87	46.94	19	183.93	59.11	116.28
	AS	16	238.04	55.61	14	207.60	34.62	114.66
L5RI	SBC	25	222.28	50.01	34	188.39	47.43	117.99
	SBL	28	221.22	45.48	19	194.85	56.90	113.53
	AS	17	243.72	53.99	15	221.38	48.27	110.09

Key: C=Cervical, T=Thoracic, L=Lumbar, Number = Vertebral level, R=Right, L=Left, S=Superior, I = Inferior, Std Dev = Standard Deviation, SDI = Sexual dimorphism Index

Facet Level SBC SBL AS Mann-Whitney Mann-Whitney Mannр р р Whitney U U U C1LS 285.50 0.16 128.00 <0.001 81.00 0.84 C1RS 247.00 0.04 79.00 < 0.001 85.50 0.66 163.00 <0.001 72.00 <0.001 39.00 0.09 C1LI 249.00 0.04 111.00 <0.001 72.00 0.51 C1RI C2LS 194.50 < 0.001 70.00 < 0.001 40.00 0.01 215.00 <0.001 C2RS < 0.001 117.00 58.00 0.03 C2LI 171.00 <0.001 173.00 0.02 50.00 0.13 C2RI 231.00 0.01 206.00 0.09 77.00 0.72 C3LS 162.00 0.01 194.00 0.04 94.00 0.49 C3RS 0.07 210.00 100.00 224.50 0.14 0.50 C3LI 210.00 0.05 233.00 0.29 81.00 0.22 C3RI 289.00 0.69 250.00 0.53 50.00 0.03 C4LS 168.00 0.03 198.00 0.02 53.00 0.07 208.00 240.00 96.00 0.54 C4RS 0.21 0.13 240.00 C4LI 0.31 248.00 88.00 0.92 0.12 C4RI 254.00 0.46 236.00 0.11 86.00 0.64 C5LS 263.00 0.04 308.50 0.37 83.00 0.96 320.00 C5RS 0.16 261.00 0.08 73.00 0.57 C55LI 371.00 0.56 258.00 0.11 45.00 0.07 C5RI 391.00 0.79 303.00 0.42 48.00 0.06 C6LS 300.00 0.24 0.02 67.00 0.86 211.00 C6RS 295.00 0.21 240.00 0.05 62.50 0.30 C6LI 223.50 0.01 236.00 0.04 60.00 0.35 C6RI 358.00 0.84 251.00 0.09 49.00 0.13 427.00 0.75 227.00 0.01 35.00 0.41 C7LS C7RS 394.00 0.37 219.00 0.01 51.00 0.56 332.00 C7LI 0.03 221.00 0.01 43.50 0.53 C7RI 269.00 <0.001 297.00 0.15 62.00 0.90

Appendix B-2 Mann-Whitney U test for sex differences in facet size

Appendix B-2 Mann-Whitney	U test for sex o	differences in fa	cet size continued
---------------------------	------------------	-------------------	--------------------

Facet Level	SBC		SBL		AS	
	Mann- Whitney U	р	Mann-Whitney U	р	Mann-Whitney U	р
T1LS	437.00	0.87	268.00	0.08	85.00	0.81
T1RS	454.00	0.72	304.00	0.16	104.00	0.79
T1LI	367.00	0.12	201.00	<0.001	123.00	0.98
T1RI	359.00	0.09	240.00	0.01	92.00	0.44
T2LS	336.00	0.03	167.00	<0.001	139.00	0.34
T2RS	375.00	0.10	200.00	<0.001	180.00	0.84
T2LI	303.00	0.01	145.00	<0.001	106.50	0.05
T2RI	260.00	<0.001	267.00	0.07	162.00	0.93
T3LS	318.00	0.02	122.00	0.00	97.00	0.04
T3RS	297.00	0.01	215.50	0.06	174.50	0.57
T3LI	240.50	<0.001	158.00	<0.001	55.00	<0.001
T3RI	322.00	0.02	228.00	0.10	171.00	0.88
T4LS	226.00	<0.001	95.00	<0.001	117.00	0.02
T4RS	288.00	0.01	172.00	0.03	145.00	0.10
T4LI	199.00	<0.001	144.00	0.01	68.00	<0.001
T4RI	173.00	0.02	106.00	0.59	80.00	0.46
T5LS	136.00	<0.001	196.00	0.08	120.00	0.09
T5RS	241.00	<0.001	200.00	0.09	174.00	0.71
T5LI	229.00	<0.001	178.00	0.07	101.00	0.15
T5RI	246.00	<0.001	182.00	0.04	80.00	0.02
T6LS	261.00	<0.001	190.50	0.01	89.00	0.16
T6RS	346.00	0.08	191.50	0.01	71.00	0.04
T6LI	168.00	<0.001	138.00	<0.001	81.00	0.13
T6RI	221.50	<0.001	123.00	<0.001	78.00	0.10
T7LS	201.00	<0.001	113.00	<0.001	93.00	0.19
T7RS	218.00	<0.001	116.00	<0.001	82.00	0.05
T7LI	219.00	<0.001	147.00	<0.001	109.00	0.69
T7RI	246.50	<0.001	87.00	<0.001	97.50	0.74

SBC Facet Level SBL AS Mann-Whitney Mann-Whitney Mannр р р Whitney U U U 106.00 T8LS 289.00 0.02 92.00 < 0.001 0.18 T8RS 283.00 0.01 109.00 < 0.001 88.00 0.05 T8LI 223.00 <0.001 123.00 < 0.001 118.00 0.25 T8RI 220.00 < 0.001 171.00 0.01 114.00 0.30 T9LS 351.00 0.04 < 0.001 129.00 132.00 0.34 T9RS <0.001 < 0.001 293.00 155.00 153.00 0.78 T9LI 293.00 0.01 126.00 < 0.001 92.00 0.11 T9RI < 0.001 < 0.001 68.00 0.02 237.00 129.00 T10LS 299.00 0.01 162.00 0.01 117.00 0.56 260.00 < 0.001 169.00 90.00 T10RS 0.01 0.17 208.00 < 0.001 152.00 < 0.001 71.00 0.02 T10LI 234.00 < 0.001 118.00 < 0.001 112.00 0.44 T10RI T11LS 292.00 < 0.001 111.00 < 0.001 61.00 0.05 <0.001 288.00 < 0.001 148.00 120.00 0.44 T11RS T11LI 274.00 <0.001 173.00 0.01 85.00 0.07 T11RI 211.00 < 0.001 222.00 0.10 51.00 0.04 T12LS 250.00 < 0.001 166.00 0.01 96.00 0.69 T12RS 294.00 0.03 198.50 0.04 75.00 0.19 T12LI 331.00 0.06 148.00 < 0.001 75.00 0.64 0.08 T12RI 340.00 222.00 0.10 75.00 0.44 335.00 0.09 164.00 0.02 0.64 L1LS 123.00 <0.001 0.92 L1RS 244.00 206.00 0.12 110.00 281.00 0.01 160.00 0.01 112.00 0.56 L1LI L1RI 326.00 0.09 146.00 0.01 99.00 0.27 260.00 < 0.001 < 0.001 94.00 L2LS 113.00 0.22 L2RS 264.00 < 0.001 0.01 91.00 131.00 0.18 < 0.001 0.01 89.00 L2LI 211.00 156.00 0.23 L2RI 254.00 <0.001 181.00 0.03 108.00 0.46

Appendix B-2 Mann-Whitney U test for sex differences in facet size continued

Facet Level	SBC		SBL		AS	
	Mann- Whitney U	р	Mann-Whitney U	р	Mann-Whitney U	р
L3LS	306.00	0.01	154.00	0.02	145.00	0.60
L3RS	266.00	0.01	108.00	<0.001	98.00	0.76
L3LI	308.00	0.02	95.00	<0.001	89.00	0.02
L3RI	340.00	0.05	154.00	0.01	70.00	0.02
L4LS	293.50	0.01	66.00	<0.001	127.00	0.27
L4RS	344.00	0.04	118.00	0.04	131.00	0.23
L4LI	249.00	<0.001	102.00	0.01	136.00	0.82
L4RI	383.00	0.13	123.00	0.03	131.00	0.89
L5LS	224.00	<0.001	99.00	<0.001	101.00	0.32
L5RS	239.00	<0.001	139.00	0.01	96.00	0.34
L5LI	252.00	0.01	158.00	0.01	80.00	0.18
L5RI	244.00	0.01	165.00	0.03	92.00	0.18

Appendix B-2 Mann-Whitney U test for sex differences in facet size continued

Key: C=Cervical, T=Thoracic, L=Lumbar, Number = Vertebral level, R=Right, L=Left, S=Superior, I = Inferior,
		Male	Size	Female Size		
Facet	Sample	Chi-Square	P Value	Chi-Square	P Value	
	SBC	0.32	0.85	0.27	0.88	
C1LS	SBL	0.31	0.86	0.57	0.75	
	AS	1.12	0.57	1.32	0.25	
	SBC	2.67	0.26	0.21	0.90	
C1RS	SBL	1.75	0.42	0.15	0.93	
	AS	2.48	0.29	0.88	0.35	
	SBC	1.25	0.54	13.33	<0.01	
C1LI	SBL	0.02	0.99	6.20	0.05	
	AS	3.75	0.15	0.06	0.81	
	SBC	0.72	0.70	5.96	0.05	
C1RI	SBL	0.48	0.79	5.42	0.07	
	AS	0.79	0.67	1.32	0.25	
	SBC	1.15	0.56	6.03	0.05	
C2LS	SBL	0.02	0.99	2.43	0.30	
	AS	0.39	0.82	4.51	0.11	
	SBC	1.62	0.44	6.03	0.05	
C2RS	SBL	1.16	0.56	4.12	0.13	
	AS	1.05	0.59	2.19	0.33	
	SBC	6.29	0.04	1.03	0.60	
C2LI	SBL	4.76	0.09	1.67	0.43	
	AS	2.60	0.27	2.55	0.28	
	SBC	0.02	0.99	2.61	0.27	
C2RI	SBL	1.66	0.44	4.94	0.08	
	AS	2.79	0.25	8.11	0.02	
	SBC	0.73	0.70	0.45	0.80	
C3LS	SBL	2.53	0.28	2.69	0.26	
	AS	2.68	0.26	3.00	0.22	

Appendix B Facet Size

		Male	Size	Female Size		
Facet	Sample	Chi-Square	P Value	Chi-Square	P Value	
	SBC	1.10	0.58	1.63	0.44	
C3RS	SBL	0.01	0.99	2.69	0.26	
	AS	2.87	0.24	5.40	0.07	
	SBC	6.53	0.04	0.58	0.75	
C3LI	SBL	1.99	0.37	7.63	0.02	
	AS	0.34	0.85	1.39	0.50	
	SBC	1.15	0.56	0.88	0.64	
C3RI	SBL	4.96	0.08	2.14	0.34	
	AS	2.88	0.24	4.17	0.12	
	SBC	1.29	0.52	1.52	0.47	
C4LS	SBL	0.94	0.63	8.02	0.02	
	AS	2.94	0.23	3.72	0.16	
	SBC	0.11	0.95	1.88	0.39	
C4RS	SBL	1.29	0.53	2.81	0.24	
	AS	1.07	0.59	1.16	0.56	
	SBC	2.73	0.26	0.48	0.79	
C4LI	SBL	0.21	0.90	7.67	0.02	
	AS	0.24	0.89	3.90	0.14	
	SBC	0.90	0.64	1.45	0.48	
C4RI	SBL	0.90	0.64	8.14	0.02	
	AS	3.72	0.16	2.05	0.36	
	SBC	1.76	0.41	3.02	0.22	
C5LS	SBL	0.16	0.92	10.82	<0.01	
	AS	1.04	0.59	0.22	0.90	
	SBC	0.40	0.82	2.43	0.30	
C5RS	SBL	1.70	0.43	7.38	0.02	
	AS	5.56	0.06	1.71	0.43	
	SBC	2.86	0.24	0.15	0.93	
C55LI	SBL	1.41	0.49	3.87	0.14	
	AS	3.81	0.15	7.81	0.02	

Facet	Sample	Male Size		Female Size		
		Chi-Square	P Value	Chi-Square	P Value	
	SBC	0.16	0.92	0.59	0.74	
C5RI	SBL	5.13	0.08	8.70	0.01	
	AS	2.57	0.28	5.61	0.06	
	SBC	2.03	0.36	0.04	0.98	
C6LS	SBL	2.44	0.30	3.60	0.17	
	AS	1.99	0.37	0.27	0.87	
	SBC	0.35	0.84	1.25	0.54	
C6RS	SBL	2.57	0.28	3.89	0.14	
	AS	3.44	0.18	1.75	0.42	
	SBC	0.74	0.69	3.64	0.16	
C6LI	SBL	0.87	0.65	3.55	0.17	
	AS	0.43	0.81	2.58	0.28	
	SBC	0.57	0.75	1.11	0.57	
C6RI	SBL	0.89	0.64	1.60	0.45	
	AS	1.07	0.58	2.73	0.26	
	SBC	0.43	0.80	0.14	0.93	
C7LS	SBL	2.20	0.33	1.57	0.46	
	AS	3.12	0.21	0.60	0.44	
	SBC	0.11	0.95	3.15	0.21	
C7RS	SBL	2.92	0.23	2.27	0.32	
	AS	1.95	0.38	1.00	0.32	
	SBC	2.03	0.36	0.82	0.66	
C7LI	SBL	1.48	0.48	0.84	0.66	
	AS	0.44	0.80	0.60	0.44	
	SBC	1.93	0.38	1.14	0.57	
C7RI	SBL	0.82	0.66	1.53	0.46	
	AS	0.12	0.94	1.00	0.32	
	SBC	1.96	0.38	1.37	0.50	
T1LS	SBL	5.01	0.08	0.29	0.87	
	AS	0.08	0.96	2.38	0.30	

Appendix B Facet Size

Facet	Sample	Male	e Size	Female Size	
		Chi-Square	P Value	Chi-Square	P Value
	SBC	5.17	0.08	1.75	0.42
T1RS	SBL	0.54	0.76	4.34	0.11
	AS	2.27	0.32	0.66	0.72
	SBC	1.81	0.41	0.28	0.87
T1LI	SBL	0.84	0.66	0.44	0.8
	AS	0.22	0.90	1.98	0.37
	SBC	5.36	0.07	2.95	0.23
T1RI	SBL	2.85	0.24	0.20	0.91
	AS	4.54	0.10	0.22	0.89
	SBC	0.54	0.76	1.06	0.59
T2LS	SBL	0.24	0.89	1.23	0.54
	AS	0.37	0.83	0.73	0.69
	SBC	2.34	0.31	3.75	0.15
T2RS	SBL	1.39	0.50	0.06	0.97
	AS	3.18	0.20	0.61	0.74
	SBC	0.10	0.95	3.91	0.14
T2LI	SBL	0.59	0.74	0.15	0.93
	AS	0.66	0.72	0.58	0.75
	SBC	1.77	0.41	11.55	<0.01
T2RI	SBL	7.48	0.02	1.53	0.47
	AS	4.54	0.10	0.22	0.89
	SBC	0.12	0.94	5.56	0.06
T3LS	SBL	0.43	0.81	0.45	0.80
	AS	0.61	0.74	2.41	0.30
	SBC	1.32	0.52	8.07	0.02
T3RS	SBL	1.59	0.45	0.76	0.68
	AS	0.40	0.82	0.20	0.91
	SBC	4.39	0.11	6.38	0.04
T3LI	SBL	0.04	0.98	0.06	0.97
	AS	4.85	0.09	1.65	0.44

Facet	Sample	Male Size		Female Size		
		Chi-Square	P Value	Chi-Square	P Value	
	SBC	0.16	0.92	7.06	0.03	
T3RI	SBL	1.89	0.39	2.52	0.28	
	AS	2.92	0.23	0.27	0.87	
	SBC	0.30	0.86	13.31	<0.01	
T4LS	SBL	0.40	0.82	3.17	0.20	
	AS	0.90	0.64	1.28	0.53	
	SBC	5.07	0.08	14.98	<0.01	
T4RS	SBL	0.90	0.64	0.87	0.65	
	AS	0.73	0.69	2.25	0.32	
	SBC	0.92	0.63	1.09	0.58	
T4LI	SBL	1.89	0.39	1.28	0.53	
	AS	1.41	0.49	1.52	0.47	
	SBC	0.11	0.94	6.80	0.03	
T4RI	SBL	1.03	0.60	6.04	0.05	
	AS	2.70	0.26	1.34	0.51	
	SBC	0.67	0.71	4.12	0.13	
T5LS	SBL	0.04	0.98	0.26	0.88	
	AS	3.36	0.19	1.20	0.55	
	SBC	6.00	0.05	8.49	0.01	
T5RS	SBL	0.36	0.84	0.75	0.69	
	AS	0.30	0.86	2.06	0.36	
	SBC	1.08	0.58	0.91	0.63	
T5LI	SBL	1.11	0.58	0.12	0.94	
	AS	5.73	0.06	0.22	0.90	
	SBC	2.48	0.29	3.25	0.20	
T5RI	SBL	0.84	0.66	0.12	0.94	
	AS	2.91	0.23	1.22	0.54	
	SBC	1.35	0.51	0.38	0.83	
T6LS	SBL	0.16	0.92	0.91	0.64	
	AS	1.55	0.46	0.18	0.67	

Appendix B Facet Size

Facet	Sample	Male	e Size	Female Size		
		Chi-Square	P Value	Chi-Square	P Value	
	SBC	7.51	0.02	5.50	0.06	
T6RS	SBL	1.02	0.60	0.11	0.95	
	AS	1.67	0.43	0.02	0.89	
	SBC	4.44	0.11	2.05	0.36	
T6LI	SBL	1.09	0.58	2.96	0.23	
	AS	2.69	0.26	0.18	0.67	
	SBC	0.62	0.73	7.78	0.02	
T6RI	SBL	0.53	0.77	0.27	0.88	
	AS	3.34	0.19	0.98	0.32	
	SBC	4.10	0.13	2.00	0.37	
T7LS	SBL	0.28	0.87	0.84	0.66	
	AS	2.70	0.26	0.10	0.75	
	SBC	1.43	0.49	11.55	<0.01	
T7RS	SBL	0.60	0.74	0.65	0.72	
	AS	5.63	0.06	0.71	0.40	
	SBC	0.16	0.92	2.09	0.35	
T7LI	SBL	1.97	0.37	0.68	0.71	
	AS	6.85	0.03	0.40	0.53	
	SBC	1.01	0.60	5.75	0.06	
T7RI	SBL	0.02	0.99	0.34	0.84	
	AS	3.15	0.21	0.02	0.90	
	SBC	0.08	0.96	0.56	0.76	
T8LS	SBL	0.60	0.74	3.51	0.17	
	AS	1.07	0.59	1.07	0.59	
	SBC	3.54	0.17	0.59	0.75	
T8RS	SBL	2.04	0.36	2.82	0.24	
	AS	0.17	0.92	2.74	0.25	
	SBC	2.92	0.23	0.12	0.94	
T8LI	SBL	0.67	0.72	1.37	0.50	
	AS	1.86	0.39	0.25	0.88	

Facet	Sample	Male Size		Female Size		
		Chi-Square	P Value	Chi-Square	P Value	
	SBC	1.86	0.40	0.72	0.70	
T8RI	SBL	0.23	0.89	0.38	0.83	
	AS	2.77	0.25	0.14	0.93	
	SBC	3.94	0.14	2.44	0.30	
T9LS	SBL	1.09	0.58	0.78	0.68	
	AS	2.11	0.35	1.04	0.60	
	SBC	4.63	0.10	4.46	0.11	
T9RS	SBL	0.71	0.70	0.41	0.82	
	AS	2.15	0.34	2.72	0.26	
	SBC	0.01	1.00	0.68	0.71	
T9LI	SBL	0.76	0.68	0.14	0.93	
	AS	3.71	0.16	0.63	0.43	
	SBC	0.08	0.96	2.55	0.28	
T9RI	SBL	0.34	0.84	2.61	0.27	
	AS	0.81	0.67	0.54	0.46	
	SBC	0.59	0.74	2.34	0.31	
T10LS	SBL	1.60	0.45	0.70	0.71	
	AS	3.77	0.15	0.32	0.57	
	SBC	3.77	0.15	2.00	0.37	
T10RS	SBL	0.90	0.64	1.67	0.43	
	AS	0.24	0.89	0.18	0.67	
	SBC	0.47	0.79	2.07	0.36	
T10LI	SBL	0.35	0.84	2.82	0.24	
	AS	0.98	0.61	0.02	0.89	
	SBC	0.60	0.74	3.52	0.17	
T10RI	SBL	0.14	0.93	2.91	0.23	
	AS	2.14	0.34	0.02	0.89	
	SBC	1.34	0.51	0.36	0.83	
T11LS	SBL	0.16	0.92	0.90	0.64	
	AS	0.05	0.82	0.32	0.57	

Appendix B Facet Size

Appendix B-3 Kruskal-Wallace test	for differences	in facet size wit	h age group	continued
	ior uniciciices	m facet Size wit	ii age gioup	continucu

Facet	Sample	Male Size		Female Size		
		Chi-Square	P Value	Chi-Square	P Value	
	SBC	0.18	0.92	0.18	0.91	
T11RS	SBL	1.55	0.46	2.84	0.24	
	AS	0.01	0.93	0.02	0.90	
	SBC	1.62	0.44	0.37	0.83	
T11LI	SBL	2.87	0.24	4.27	0.12	
	AS	0.02	0.89	0.38	0.54	
	SBC	1.34	0.51	0.08	0.96	
T11RI	SBL	1.66	0.44	6.89	0.03	
	AS	1.18	0.28	2.34	0.13	
	SBC	0.28	0.87	0.52	0.77	
T12LS	SBL	3.16	0.21	2.05	0.36	
	AS	0.60	0.44	3.13	0.08	
	SBC	1.73	0.42	0.43	0.81	
T12RS	SBL	0.39	0.82	3.32	0.19	
	AS	1.67	0.20	0.50	0.48	
	SBC	3.07	0.22	1.79	0.41	
T12LI	SBL	0.62	0.73	0.15	0.93	
	AS	1.67	0.20	5.67	0.02	
	SBC	1.95	0.38	0.03	0.99	
T12RI	SBL	0.71	0.70	4.85	0.09	
	AS	0.73	0.39	0.04	0.84	
	SBC	3.58	0.17	6.07	0.05	
L1LS	SBL	2.64	0.27	3.50	0.17	
	AS	1.24	0.27	0.17	0.92	
	SBC	0.30	0.86	0.99	0.61	
L1RS	SBL	1.52	0.47	5.10	0.08	
	AS	0.13	0.72	0.66	0.72	
	SBC	1.25	0.54	5.60	0.06	
L1LI	SBL	1.14	0.57	2.61	0.27	
	AS	0.89	0.35	0.72	0.70	

Facet	Sample	Male Size		Female Size		
		Chi-Square	P Value	Chi-Square	P Value	
	SBC	0.66	0.72	6.11	0.05	
L1RI	SBL	1.72	0.42	4.18	0.12	
	AS	0.00	0.96	0.16	0.92	
	SBC	2.61	0.27	1.47	0.48	
L2LS	SBL	1.63	0.44	2.26	0.32	
	AS	1.51	0.47	2.55	0.28	
	SBC	2.80	0.25	2.06	0.36	
L2RS	SBL	0.86	0.65	1.14	0.57	
	AS	0.29	0.87	1.10	0.58	
	SBC	0.36	0.83	1.57	0.46	
L2LI	SBL	1.41	0.49	4.01	0.13	
	AS	0.82	0.37	0.42	0.81	
	SBC	1.11	0.57	3.73	0.15	
L2RI	SBL	0.32	0.85	3.50	0.17	
	AS	0.49	0.48	2.66	0.26	
	SBC	2.40	0.30	2.70	0.26	
L3LS	SBL	3.98	0.14	6.33	0.04	
	AS	1.61	0.20	0.55	0.76	
	SBC	1.97	0.37	0.81	0.67	
L3RS	SBL	2.06	0.36	2.59	0.27	
	AS	0.00	1.00	0.54	0.76	
	SBC	2.16	0.34	2.08	0.35	
L3LI	SBL	0.18	0.91	1.63	0.44	
	AS	0.90	0.34	0.46	0.79	
	SBC	3.63	0.16	2.40	0.30	
L3RI	SBL	0.19	0.91	11.24	<0.01	
	AS	0.34	0.56	2.70	0.26	
	SBC	1.13	0.57	0.37	0.83	
L4LS	SBL	0.48	0.79	0.50	0.78	
	AS	1.81	0.41	0.01	0.92	

Appendix B Facet Size

Facet	Sample	Male	e Size	Female Size		
		Chi-Square	P Value	Chi-Square	P Value	
	SBC	0.78	0.68	0.31	0.86	
L4RS	SBL	1.12	0.57	8.72	0.01	
	AS	0.07	0.97	2.28	0.13	
	SBC	5.66	0.06	3.61	0.16	
L4LI	SBL	0.36	0.83	8.48	0.01	
	AS	2.77	0.25	1.50	0.22	
	SBC	0.45	0.80	3.89	0.14	
L4RI	SBL	1.03	0.60	2.01	0.37	
	AS	0.22	0.64	0.00	1.00	
	SBC	0.24	0.89	0.83	0.66	
L5LS	SBL	1.51	0.47	2.60	0.27	
	AS	0.09	0.76	0.02	0.90	
	SBC	1.56	0.46	1.72	0.42	
L5RS	SBL	2.25	0.32	2.48	0.29	
	AS	0.19	0.66	0.24	0.62	
	SBC	0.90	0.64	5.42	0.07	
L5LI	SBL	0.56	0.76	5.00	0.08	
	AS	0.08	0.78	0.04	0.84	
	SBC	0.21	0.90	13.69	<0.01	
L5RI	SBL	1.23	0.54	1.44	0.49	
	AS	1.71	0.19	1.82	0.18	

Appendix B-3 Kruskal-Wallace test for differences in facet size with age group continued

Key: C=Cervical, T=Thoracic, L=Lumbar, Number = Vertebral level, R=Right, L=Left, S=Superior, I = Inferior

Males			Females					
Facet	Correlation Coefficient	р	N	Strength of Relationship	Correlation Coefficient	р	N	Strength of Relationship
C1LS	-0.015	0.946	23	None	-0.065	0.722	32	Weak
C1RS	-0.23	0.29	23	Weak	-0.2	0.272	32	Weak
C1LI	0.242	0.267	23	Weak	0.427	0.015	32	Medium
C1RI	0.392	0.392	23	Medium	0.3	0.096	32	Medium
C2LS	0.202	0.355	23	Weak	0.208	0.23	35	Weak
C2RS	0.399	0.059	23	Medium	0.18	0.302	35	Weak
C2LI	0.637	<0.0001	22	Strong	-0.227	0.191	35	Weak
C2RI	0.12	0.594	22	Weak	-0.195	0.262	35	Weak
C3LS	0.377	0.111	19	Medium	-0.166	0.363	32	Weak
C3RS	0.178	0.454	20	Weak	-0.141	0.442	32	Weak
C3LI	0.366	0.113	20	Medium	-0.041	0.826	31	None
C3RI	-0.097	0.684	20	None	0.1	0.593	31	Weak
C4LS	0.258	0.286	19	Weak	0.186	0.345	28	Weak
C4RS	-0.029	0.906	19	None	0.027	0.891	28	None
C4LI	0.379	0.099	20	Medium	0.013	0.947	29	None
C4RI	0.423	0.063	20	Medium	-0.021	0.913	29	None
C5LS	0.35	0.101	23	Medium	0.122	0.492	34	Weak
C5RS	0.12	0.576	24	Weak	0.051	0.775	34	None
C5LI	0.36	0.084	24	Medium	-0.05	0.779	34	None
C5RI	0.054	0.801	24	None	-0.093	0.599	34	None
C6LS	0.072	0.762	20	None	0.032	0.851	37	None
C6RS	0.004	0.987	20	None	0.179	0.289	37	Weak
C6LI	0.226	0.337	20	Weak	-0.31	0.062	37	Medium
C6RI	-0.002	0.995	20	None	-0.144	0.395	37	Weak
C7LS	0.158	0.472	23	Weak	-0.09	0.585	39	None
C7RS	0.007	0.973	24	None	0.224	0.176	38	Weak
C7LI	-0.105	0.619	24	Weak	-0.027	0.87	39	None
C7RI	0.169	0.421	25	Weak	0.068	0.68	39	None
T1LS	0.185	0.399	23	Weak	0.114	0.488	39	Weak
T1RS	0.286	0.175	24	Weak	0.038	0.816	40	None
T1LI	0.156	0.487	24	Weak	0.02	0.904	40	None
T1RI	0.272	0.199	24	Weak	0.058	0.723	40	None
T2LS	0.215	0.291	26	Weak	-0.073	0.661	38	None
T2RS	0.137	0.504	26	Weak	0.031	0.854	38	None
T2LI	0.065	0.751	26	None	0.105	0.529	38	Weak
T2RI	0.152	0.46	26	Weak	0.378	0.019	38	Medium
T3LS	0.005	0.98	25	None	0.038	0.82	39	None
T3RS	0.166	0.427	25	Weak	0.064	0.697	39	None
T3LI	-0.371	0.068	25	Medium	0.291	0.072	39	Weak
T3RI	0.165	0.43	25	Weak	0.291	0.084	39	Weak

Appendix B-4 Spearman's Rho correlation between facet size and actual age

		Mal	es			Fe	males	
Facet	Correlation Coefficient	р	N	Strength of Relationship	Correlation Coefficient	р	Ν	Strength of Relationship
T4LS	0.026	0.901	26	None	-0.022	0.896	37	None
T4RS	0.46	0.018	26	Medium	0.339	0.043	36	Medium
T4LI	0.022	0.917	26	None	0.015	0.929	36	None
T4RI	0.119	0.588	23	Weak	0.505	<0.0001	25	Strong
T5LS	0.209	0.316	25	Weak	-0.082	0.631	37	None
T5RS	0.424	0.035	25	Medium	0.386	0.018	37	Medium
T5LI	-0.099	0.637	25	None	-0.086	0.619	36	None
T5RI	0.267	0.196	25	Weak	0.026	0.878	36	None
T6LS	0.215	0.291	26	Weak	-0.022	0.899	36	None
T6RS	0.379	0.056	26	Medium	0.27	0.112	36	Weak
T6LI	0.261	0.198	26	Weak	-0.081	0.63	38	None
T6RI	0.014	0.946	26	None	0.08	0.635	38	None
T7LS	0.304	0.131	26	Medium	-0.182	0.274	38	Weak
T7RS	0.351	0.079	26	Medium	-0.063	0.706	38	None
T7LI	0.033	0.874	25	None	-0.259	0.116	38	Weak
T7RI	-0.099	0.637	25	None	-0.243	0.141	38	Weak
T8LS	0.061	0.773	25	None	-0.051	0.766	36	None
T8RS	0.179	0.392	25	Weak	0.011	0.948	36	None
T8LI	0.363	0.068	26	Medium	0.028	0.871	36	None
T8RI	0.257	0.205	26	Weak	-0.115	0.504	36	Weak
T9LS	0.364	0.068	26	Medium	0.218	0.182	39	Weak
T9RS	0.329	0.101	26	Medium	0.196	0.231	39	Weak
T9LI	0.194	0.353	25	Weak	-0.202	0.218	39	Weak
T9RI	0.129	0.54	25	Weak	-0.145	0.386	38	Weak
T10LS	0.093	0.657	25	None	0.192	0.241	39	Weak
T10RS	0.294	0.154	25	Weak	0.025	0.879	39	None
T10LI	0.134	0.514	26	Weak	0.085	0.606	39	None
T10RI	0.182	0.373	26	Weak	0.005	0.977	39	None
T11LS	0.14	0.494	26	Weak	-0.042	0.798	39	None
T11RS	-0.06	0.771	26	None	0.072	0.663	39	None
T11LI	0.197	0.344	25	Weak	-0.147	0.373	39	Weak
T11RI	-0.007	0.974	25	None	-0.097	0.559	39	None
T12LS	0.084	0.698	24	None	-0.179	0.289	37	Weak
T12RS	-0.07	0.744	24	None	-0.055	0.748	37	None
T12LI	0.312	0.128	25	Medium	0.035	0.835	37	None
T12RI	0.108	0.608	25	Weak	-0.073	0.67	37	None

Appendix B-4 Spearman's Rho correlation between facet size and actual age continued

		М	ales			Fe	males	
Facet	Correlation Coefficient	р	Ν	Strength of Relationship	Correlation Coefficient	р	N	Strength of Relationship
L1LS	0.185	0.375	25	Weak	-0.093	0.591	36	None
L1RS	0.204	0.329	25	Weak	-0.026	0.882	35	None
L1LI	0.165	0.43	25	Weak	-0.024	0.89	36	None
L1RI	0.081	0.7	25	None	0.019	0.915	35	None
L2LS	0.183	0.37	26	Weak	0.036	0.834	36	None
L2RS	0.196	0.338	26	Weak	0.012	0.944	36	None
L2LI	0.064	0.755	26	None	0.116	0.501	36	Weak
L2RI	0.168	0.411	26	Weak	0.01	0.952	36	None
L3LS	0.126	0.539	26	Weak	0.079	0.64	37	None
L3RS	0.223	0.274	26	Weak	-0.1	0.567	35	Weak
L3LI	0.058	0.779	26	None	0.108	0.524	37	Weak
L3RI	0.258	0.203	26	Weak	0.224	0.183	37	Weak
L4LS	0.115	0.577	26	Weak	0.071	0.674	38	None
L4RS	0.241	0.235	26	Weak	-0.094	0.576	38	None
L4LI	0.61	<0.0001	26	Strong	0.098	0.557	38	None
L4RI	0.353	0.077	26	Weak	0.191	0.25	38	Weak
L5LS	0.17	0.415	25	Weak	-0.104	0.559	34	Weak
L5RS	-0.036	0.864	25	None	0.203	0.25	34	Weak
L5LI	0.355	0.081	35	Medium	0.323	0.062	34	Medium
L5RI	0.067	0.749	25	None	0.4 0.021		33	Medium

Appendix B-4 Spearman's Rho correlation between facet size and actual age continued

Appendix B-5 Kruskal-Wallace test for inter-sample variation in facet size

	Male		Femal	е
Facet	Chi-Square	р	Chi-Square	р
C1LS	0.764	0.683	4.962	0.084
C1RS	1.188	0.552	6.779	0.034
C1LI	0.086	0.958	6.681	0.035
C1RI	0.52	0.771	8.115	0.017
C2LS	0.1	0.951	5.097	0.078
C2RS	3.982	0.137	5.477	0.065
C2LI	4.399	0.111	15.167	0.001
C2RI	7.766	0.021	6.428	0.04
C3LS	0.719	0.698	2.501	0.286
C3RS	1.823	0.402	3.295	0.193
C3LI	9.008	0.011	2.585	0.275
C3RI	9.315	0.009	3.552	0.169
C4LS	1.68	0.432	3.982	0.137
C4RS	1.605	0.448	9.682	0.008
C4LI	2.616	0.27	6.61	0.037
C4RI	2.317	0.314	8.417	0.015
C5LS	2.192	0.334	3.914	0.141
C5RS	0.627	0.731	1.813	0.404
C55LI	11.209	0.004	9.426	0.009
C5RI	7.435	0.024	5.962	0.051
C6LS	1.438	0.487	8.198	0.017
C6RS	6.002	0.05	8.631	0.013
C6LI	13.554	0.001	18.612	<0.001
C6RI	16.653	<0.001	13.838	0.001
C7LS	8.474	0.014	3.899	0.142
C7RS	8.567	0.014	8.664	0.013
C7LI	3.801	0.15	15.173	0.001
C7RI	3.294	0.193	9.559	0.008
T1LS	2.279	0.32	2.102	0.35
T1RS	0.084	0.959	1.05	0.592
T1LI	2.215	0.33	11.491	0.003
T1RI	2.303	0.316	6.007	0.05
T2LS	0.78	0.677	4.716	0.095
T2RS	0.214	0.898	10.158	0.006
T2LI	1.241	0.538	1.83	0.4
T2RI	0.109	0.947	3.543	0.17

	Male		Female	2
Facet	Chi-Square	р	Chi-Square	р
T3LS	2.928	0.231	5.672	0.059
T3RS	1.218	0.544	2.676	0.262
T3LI	0.746	0.689	0.108	0.947
T3RI	1.128	0.569	3.996	0.136
T4LS	3.584	0.167	6.642	0.036
T4RS	0.532	0.766	2.499	0.287
T4LI	3.056	0.217	4.035	0.133
T4RI	0.52	0.771	9.909	0.007
T5LS	6.419	0.04	11.796	0.003
T5RS	0.024	0.988	5.388	0.068
T5LI	1.897	0.387	7.181	0.028
T5RI	1.703	0.005	9.11	0.011
T6LS	0.919	0.632	5.228	0.073
T6RS	9.382	0.009	3.539	0.17
T6LI	1.206	0.547	4.227	0.121
T6RI	3.86	0.145	8.383	0.015
T7LS	1.208	0.546	9.46	0.009
T7RS	5.084	0.079	14.527	0.001
T7LI	0.228	0.892	13.813	0.001
T7RI	5.274	0.072	24.114	<0.001
T8LS	5.793	0.055	19.573	<0.001
T8RS	14.216	0.001	17.831	<0.001
T8LI	2.603	0.272	11.646	0.003
T8RI	9.492	0.009	19.336	<0.001
T9LS	5.096	0.078	21.522	<0.001
T9RS	2.469	0.291	17.091	<0.001
T9LI	1.144	0.564	7.752	0.021
T9RI	6.478	0.039	11.346	0.003
T10LS	3.855	0.145	11.842	0.003
T10RS	13.508	0.001	13.746	0.001
T10LI	0.213	0.899	6.603	0.037
T10RI	3.466	0.177	16.66	<0.001
T11LS	0.93	0.628	6.991	0.03
T11RS	7.925	0.019	13.84	0.001
T11LI	5.063	0.08	9.623	0.008
T11RI	8.289	0.016	10.024	0.007
T12LS	2.09	0.352	15.909	<0.001
T12RS	1.78	0.411	3.235	0.198
T12LI	1.746	0.418	8.803	0.012
T12RI	0.908	0.635	2.395	0.302

Appendix B-5 Kruskal-Wallace test for inter-sample variation in facet size continued

	Male		Female	5
Facet	Chi-Square	р	Chi-Square	р
L1LS	1.22	0.543	7.148	0.028
L1RS	2.003	0.367	5.545	0.063
L1LI	0.004	0.998	6.702	0.035
L1RI	2.585	0.275	5.212	0.074
L2LS	1.151	0.562	10.325	0.006
L2RS	1.153	0.562	6.59	0.037
L2LI	0.79	0.674	5.657	0.059
L2RI	1.146	0.564	6.085	0.048
L3LS	0.278	0.87	5.262	0.072
L3RS	0.572	0.751	12.88	0.002
L3LI	2.572	0.276	2.724	0.256
L3RI	5.45	0.066	1.915	0.384
L4LS	4.273	0.118	9.272	0.01
L4RS	1.604	0.448	4.133	0.127
L4LI	0.566	0.754	9.7	0.008
L4RI	1.114	0.573	6.724	0.035
L5LS	1.63	0.443	13.818	0.001
L5RS	2.548	0.28	5.445	0.066
L5LI	2.626	0.269	5.093	0.078
L5RI	1.86	0.395	5.047	0.08

Appendix B-5 Kruskal-Wallace test for inter-sample variation in facet size continued

			N	1ale					Fe	male		
Facet	SBC/	/SBL	SBC	C/AS	SBL	_/AS	SBC/	/SBL	SBC	C/AS	SBL	_/AS
	Z	р	Z	р	Z	р	Z	р	Z	р	Z	р
C1LS	-0.501	0.616	-0.917	0.359	-1.772	0.076	-0.997	0.319	-1.772	0.076	-2.024	0.043
C1RS	-0.74	0.459	-1.049	0.294	-2.274	0.023	-0.433	0.665	-2.274	0.023	-2.508	0.012
C1LI	-0.04	0.968	-0.194	0.846	-1.26	0.208	-1.919	0.055	-1.26	0.208	-2.263	0.024
C1RI	-0.516	0.606	-0.616	0.538	-2.303	0.021	-0.922	0.357	-2.303	0.021	-2.772	0.006
C2LS	-0.212	0.832	-0.376	0.707	-0.731	0.465	-1.61	0.107	-0.731	0.465	-2.321	0.02
C2RS	-0.599	0.549	-1.3	0.194	-1.391	0.164	-1.522	0.128	-1.391	0.164	-2.141	0.032
C2LI	-1.978	0.048	-1.502	0.133	-3.644	<0.001	-0.367	0.713	-3.644	<0.001	-3.468	0.001
C2RI	-2.453	0.014	-0.195	0.846	-1.584	0.113	-1.75	0.08	-1.584	0.113	-2.147	0.032
C3LS	-0.824	0.41	-0.333	0.739	-1.239	0.216	-0.473	0.636	-1.239	0.216	-1.572	0.116
C3RS	-1.192	0.233	-0.146	0.884	-1.291	0.197	-0.677	0.498	-1.291	0.197	-1.868	0.062
C3LI	-1.743	0.081	-0.731	0.465	-1.029	0.304	-1.054	0.292	-1.029	0.304	-1.385	0.166
C3RI	-0.836	0.403	-2	0.046	-0.812	0.417	-1.196	0.232	-0.812	0.417	-1.946	0.052
C4LS	-0.432	0.666	-0.642	0.521	-1.741	0.082	-0.528	0.598	-1.741	0.082	-1.802	0.072
C4RS	-1.022	0.307	-0.048	0.962	-2.521	0.012	-1.153	0.249	-2.521	0.012	-2.834	0.005
C4LI	-0.651	0.515	-0.867	0.386	-1.633	0.102	-1.225	0.22	-1.633	0.102	-2.572	0.01
C4RI	-0.468	0.64	-0.987	0.324	-2.063	0.039	-0.742	0.458	-2.063	0.039	-3.099	0.002
C5LS	-1.418	0.156	-0.07	0.945	-1.903	0.057	-0.047	0.962	-1.903	0.057	-1.686	0.092
C5RS	-0.749	0.454	-0.101	0.92	-0.499	0.617	-0.789	0.43	-0.499	0.617	-1.495	0.135
C55LI	-1.179	0.238	-2.953	0.003	-2.977	0.003	-0.111	0.912	-2.977	0.003	-2.618	0.009
C5RI	-0.75	0.453	-2.55	0.011	-2.354	0.019	-0.205	0.837	-2.354	0.019	-2.068	0.039
C6LS	-0.238	0.812	-1.19	0.234	-2.469	0.014	-0.449	0.654	-2.469	0.014	-2.86	0.004
C6RS	-0.57	0.569	-2.167	0.03	-2.784	0.005	-0.768	0.443	-2.784	0.005	-2.452	0.014
C6LI	-1.023	0.307	-2.589	0.01	-3.962	<0.001	-1.254	0.21	-3.962	<0.001	-3.736	<0.001
C6RI	-0.673	0.501	-3.534	<0.001	-3.299	0.001	-1.072	0.284	-3.299	0.001	-3.368	0.001
C7LS	-1.57	0.116	-2.948	0.003	-1.422	0.155	-0.934	0.35	-1.422	0.155	-1.984	0.047
C7RS	-1.176	0.24	-2.916	0.004	-2.376	0.017	-1.387	0.165	-2.376	0.017	-2.655	0.008
C7LI	-0.917	0.359	-1.355	0.175	-2.829	0.005	-2.42	0.016	-2.829	0.005	-3.26	0.001
C7RI	-1.287	0.198	-0.695	0.487	-3.085	0.002	0	1	-3.085	0.002	-2.655	0.008
T1LS	-1.301	0.193	-1.299	0.194	-1.132	0.257	-0.511	0.61	-1.132	0.257	-1.441	0.149
T1RS	-0.237	0.813	-0.212	0.832	-0.455	0.649	-0.685	0.493	-0.455	0.649	-1.037	0.3
T1LI	-1.018	0.308	-1.37	0.171	-3.06	0.002	-0.857	0.392	-3.06	0.002	-3.014	0.003
T1RI	-0.679	0.497	-1.429	0.153	-2.191	0.028	-0.657	0.511	-2.191	0.028	-2.19	0.029
T2LS	-0.776	0.438	-0.753	0.451	-0.909	0.363	-1.718	0.086	-0.909	0.363	-1.892	0.058
T2RS	-0.204	0.838	-0.393	0.694	-2.122	0.034	-1.902	0.057	-2.122	0.034	-2.868	0.004
T2LI	-0.686	0.493	-0.601	0.548	-0.206	0.837	-1.243	0.214	-0.206	0.837	-1.071	0.284
T2RI	-0.313	0.754	-0.021	0.983	-1.733	0.083	-1.15	0.25	-1.733	0.083	-0.844	0.399

Appendix B-6 Mann-Whitney U inter-sample comparison of facet size

Appendix B-6 Mann-Whitney U inter-sample comparison of facet size continued

			М	ale					Fei	male		
Facet	SBC/	/SBL	SBC	/AS	SBL	./AS	SBC/	/SBL	SBC	C/AS	SBL	./AS
	z	р	z	р	Z	р	Z	р	Z	р	Z	р
T3LS	-1.258	0.208	-1.599	0.11	-0.975	0.329	-1.837	0.066	-0.975	0.329	-2.166	0.03
T3RS	-0.541	0.589	-1.104	0.27	-1.612	0.107	-0.101	0.92	-1.612	0.107	-1.257	0.209
T3LI	-0.794	0.427	-0.384	0.701	-0.053	0.958	-0.271	0.786	-0.053	0.958	-0.337	0.736
T3RI	-1.099	0.272	-0.235	0.815	-2.037	0.042	-0.411	0.681	-2.037	0.042	-1.332	0.183
T4LS	-1.281	0.2	-1.923	0.054	-2.468	0.014	-0.19	0.849	-2.468	0.014	-2.044	0.041
T4RS	0	1	-0.841	0.4	-1.315	0.189	-0.602	0.547	-1.315	0.189	-1.378	0.168
T4LI	-0.135	0.893	-1.614	0.107	-2.026	0.043	-0.459	0.646	-2.026	0.043	-1.302	0.193
T4RI	-0.755	0.45	-0.314	0.753	-2.952	0.003	-1.882	0.06	-2.952	0.003	-1.097	0.273
T5LS	-1.318	0.187	-1.422	0.155	-3.399	0.001	-1.07	0.285	-3.399	0.001	-2.194	0.028
T5RS	-0.16	0.873	-0.064	0.949	-2.319	0.02	-0.92	0.358	-2.319	0.02	-1.25	0.211
T5LI	-1.112	0.266	-1.22	0.222	-2.439	0.015	-1.761	0.078	-2.439	0.015	-0.933	0.351
T5RI	-0.499	0.618	-3.106	0.002	-2.977	0.003	-0.992	0.321	-2.977	0.003	-1.967	0.04 9
T6LS	-0.897	0.37	-0.692	0.489	-2.161	0.031	-1.257	0.209	-2.161	0.031	-1.145	0.252
T6RS	-1.682	0.093	-2.96	0.003	-1.837	0.066	-0.645	0.519	-1.837	0.066	-1.28	0.201
T6LI	-0.354	0.723	-0.77	0.441	-2.063	0.039	-0.285	0.776	-2.063	0.039	-1.515	0.13
T6RI	-0.26	0.795	-1.515	0.13	-2.434	0.015	-0.728	0.466	-2.434	0.015	-2.761	0.006
T7LS	-0.881	0.378	-0.271	0.787	-2.368	0.018	-1.194	0.232	-2.368	0.018	-2.929	0.003
T7RS	-0.907	0.364	-1.399	0.162	-3.031	0.002	-1.49	0.136	-3.031	0.002	-3.51	<0.001
T7LI	-0.186	0.852	-0.439	0.661	-3.467	0.001	-0.327	0.744	-3.467	0.001	-3.168	0.002
T7RI	-1.294	0.196	-2.079	0.038	-4.291	<0.001	-1.276	0.202	-4.291	<0.001	-4.444	<0.001
T8LS	-0.086	0.931	-2.012	0.044	-3.192	0.001	-2.308	0.021	-3.192	0.001	-4.011	<0.001
T8RS	-0.432	0.666	-3.472	0.001	-3.113	0.002	-1.847	0.065	-3.113	0.002	-4.011	<0.001
T8LI	-0.58	0.562	-1.48	0.139	-3.133	0.002	-0.073	0.941	-3.133	0.002	-2.967	0.003
T8RI	-1.008	0.314	-2.077	0.038	-3.886	<0.001	-0.159	0.873	-3.886	<0.001	-4.11	<0.001
T9LS	-0.272	0.785	-1.719	0.086	-3.163	0.002	-2.482	0.013	-3.163	0.002	-4.358	<0.001
T9RS	-1.442	0.149	-0.215	0.83	-3.306	0.001	-1.329	0.184	-3.306	0.001	-3.931	<0.001
T9LI	-0.915	0.36	-0.218	0.828	-1.651	0.099	-1.841	0.066	-1.651	0.099	-2.533	0.011
T9RI	-1.379	0.168	-1.443	0.149	-2.35	0.019	-1.799	0.072	-2.35	0.019	-3.133	0.002
T10LS	-0.186	0.852	-1.789	0.074	-2.906	0.004	-0.849	0.396	-2.906	0.004	-3.316	0.001
T10RS	-1.59	0.112	-2.462	0.014	-2.821	0.005	-1.761	0.078	-2.821	0.005	-3.426	0.001
T10LI	-0.476	0.634	-0.161	0.872	-2.462	0.014	-0.048	0.962	-2.462	0.014	-2.174	0.03
T10RI	-0.624	0.532	-1.402	0.161	-3.138	0.002	-1.505	0.132	-3.138	0.002	-4.053	<0.001
T11LS	-0.658	0.511	-0.244	0.808	-1.448	0.148	-1.601	0.109	-1.448	0.148	-2.69	0.007
T11RS	-0.067	0.946	-2.436	0.015	-3.47	0.001	-0.457	0.648	-3.47	0.001	-3.232	0.001
T11LI	-0.232	0.817	-2.142	0.032	-2.824	0.005	-0.736	0.462	-2.824	0.005	-2.694	0.007
T11RI	-0.997	0.319	-2.111	0.035	-3.153	0.002	-0.721	0.471	-3.153	0.002	-2.32	0.02
T12LS	-1.248	0.212	-0.151	0.88	-2.681	0.007	-2.435	0.015	-2.681	0.007	-3.536	<0.001
T12RS	-0.822	0.411	-0.454	0.65	-1.435	0.151	-0.736	0.462	-1.435	0.151	-1.65	0.099
T12LI	-0.477	0.633	-1.171	0.242	-2.514	0.012	-0.752	0.452	-2.514	0.012	-2.877	0.004
T12RI	-0.286	0.775	-0.785	0.433	-1.272	0.203	-0.493	0.622	-1.272	0.203	-1.471	0.141

			М	ale					Fer	nale		
Facet	SBC/	'SBL	SBC	/AS	SBL	_/AS	SBC/	'SBL	SBC	C/AS	SBL	/AS
	Z	р	Z	р	Z	р	Z	р	Z	р	Z	р
L1LS	-1.265	0.206	-0.267	0.789	-1.646	0.1	-0.992	0.321	-1.646	0.1	-2.993	0.003
L1RS	-1.39	0.165	-0.265	0.791	-2.102	0.036	-0.385	0.7	-2.102	0.036	-2.17	0.03
L1LI	-0.026	0.979	-0.07	0.944	-2.26	0.024	-0.222	0.824	-2.26	0.024	-2.42	0.016
L1RI	-1.354	0.176	-1.39	0.165	-2.001	0.045	-0.105	0.916	-2.001	0.045	-2.167	0.03
L2LS	-0.152	0.879	-1.003	0.316	-2.4	0.016	-0.855	0.393	-2.4	0.016	-3.426	0.001
L2RS	-0.26	0.795	-1.05	0.294	-2.31	0.021	-0.239	0.811	-2.31	0.021	-2.362	0.018
L2LI	-0.394	0.693	-0.522	0.602	-2.242	0.025	-0.018	0.986	-2.242	0.025	-2.053	0.04
L2RI	-0.476	0.634	-1.118	0.264	-2.312	0.021	-0.177	0.86	-2.312	0.021	-2.113	0.035
L3LS	-0.439	0.66	-0.099	0.921	-2.027	0.043	-0.334	0.738	-2.027	0.043	-2.046	0.041
L3RS	-0.773	0.44	-0.406	0.685	-3.19	0.001	-0.601	0.548	-3.19	0.001	-3.283	0.001
L3LI	-1.139	0.255	-1.408	0.159	-1.145	0.252	-0.591	0.555	-1.145	0.252	-1.717	0.086
L3RI	-1.72	0.085	-2.02	0.043	-1.298	0.194	-0.607	0.544	-1.298	0.194	-0.92	0.358
L4LS	-0.038	0.97	-1.838	0.066	-2.538	0.011	-0.889	0.374	-2.538	0.011	-2.807	0.005
1485	-0 366	0 714	-1 263	0 207	-1 402	0 161	-1.066	0.286	-1 402	0 161	-1 937	0.053
1411	-0.283	0 777	-0 712	0.476	-2.95	0.003	-0.72	0 472	-2.95	0.003	-2 453	0.014
1481	-0.641	0 5 2 2	-0.955	0.34	-2 187	0.029	-0.89	0 373	-2 187	0.029	-2 328	0.02
1515	-0.616	0.538	-1 268	0 205	-3 493	<0.001	-0.074	0.941	-3 493	<0.001	-3 169	0.002
1585	-0.284	0 777	-1 71	0.087	-2.06/	0.030	-0.427	0.67	-2.06/	0.030	-2 112	0.035
1511	-0.702	0.492	-1.016	0.007	-1 921	0.067	-0.510	0.604	-1 921	0.067	-2.115	0.035
ISRI	-0.303	0.462	-1 089	0.31	-2.07	0.007	-0.352	0.725	-2.07	0.007	-1 931	0.023

Appendix B-6 Mann-Whitney U inter-sample comparison of facet size continued

Facet	Sample				М	ale							Fen	nale			
			Left Femo	oral Robus	ticity		Right Fem	oral Robus	sticity		Left Femo	oral Robus	ticity		Right Fem	oral Robus	sticity
		C Coeffic ient	Р	N	Strength of Correlation	C Coeffic ient	Р	N	Strength of Correlation	C Coeffic ient	Р	N	Strength of Correlation	C Coeffic ient	Р	N	Strength of Correlation
C1LS	SBC	0.00	0.99	11	None	-0.25	0.49	10	Weak	-0.15	0.57	17	Weak	-0.37	0.18	15	Medium
	SBL	0.13	0.61	18	Weak	-0.07	0.79	19	None	-0.12	0.74	11	Weak	-0.43	0.22	10	Medium
	AS	-0.23	0.44	13	Weak	-0.16	0.59	14	Weak	0.16	0.74	7	Weak	0.39	0.29	9	Medium
C1RS	SBC	0.03	0.93	11	None	0.07	0.85	10	None	-0.11	0.67	17	Weak	-0.25	0.36	15	Weak
	SBL	-0.15	0.54	19	Weak	-0.04	0.86	20	None	-0.05	0.89	11	None	-0.22	0.54	10	Weak
	AS	-0.39	0.17	14	Medium	-0.19	0.48	16	Weak	0.25	0.59	7	Weak	0.46	0.22	9	Medium
C1LI	SBC	-0.30	0.37	11	Medium	-0.08	0.82	10	None	-0.09	0.73	17	None	-0.36	0.19	15	Medium
	SBL	0.22	0.39	18	Weak	0.08	0.76	19	None	0.14	0.68	11	Weak	-0.15	0.68	10	Weak
	AS	0.08	0.82	11	None	0.01	0.97	12	None	-0.11	0.84	6	Weak	-0.30	0.47	8	Medium
C1RI	SBC	-0.24	0.48	11	Weak	-0.09	0.81	10	None	0.04	0.87	17	None	-0.39	0.15	15	Medium
	SBL	0.16	0.51	19	Weak	-0.05	0.83	20	None	0.36	0.28	11	Medium	-0.32	0.36	10	Medium
	AS	-0.29	0.36	12	Weak	-0.29	0.31	14	Weak	0.42	0.35	7	Medium	0.43	0.25	9	Medium

					М	ale							Fen	nale			
		L	eft Femora	al Robu	sticity	Rig	ht Femora	l Robus	ticity	Lef	t Femora	l Robu	sticity	Rig	ht Femo	ral Robus	ticity
Facet	Sample	C Coefficie nt	Ρ	N	Strength of Correlation	C Coefficien t	Ρ	N	Strength of Correlation	C Coefficient	Ρ	N	Strength of Correlation	C Coefficie nt	Ρ	N	Strength of Correlation
C2LS	SBC	-0.35	0.27	12	Medium	-0.39	0.23	11	Medium	0.07	0.79	19	None	-0.02	0.93	17	None
	SBL	0.12	0.63	20	Weak	0.03	0.90	22	None	0.12	0.72	11	Weak	-0.10	0.77	10	Weak
	AS	0.01	0.98	10	None	0.10	0.78	11	None	0.13	0.72	10	Weak	0.16	0.63	12	Weak
C2RS	SBC	0.02	0.94	12	None	-0.01	0.98	11	None	0.21	0.40	19	Weak	-0.05	0.84	17	None
	SBL	0.35	0.13	20	Medium	0.18	0.42	22	Weak	0.28	0.40	11	Weak	-0.44	0.20	10	Medium
	AS	-0.18	0.57	12	Weak	-0.19	0.51	14	Weak	-0.09	0.82	9	None	0.00	0.99	11	None
C2LI	SBC	-0.29	0.35	12	Weak	-0.13	0.69	11	Weak	0.19	0.45	19	Weak	0.25	0.34	17	Weak
	SBL	-0.16	0.49	20	Weak	-0.12	0.60	22	Weak	-0.18	0.60	11	Weak	0.39	0.27	10	Medium
	AS	-0.03	0.93	9	None	-0.03	0.94	10	None	0.00	0.99	9	None	-0.05	0.90	10	None
C2RI	SBC	-0.06	0.86	11	None	-0.15	0.65	11	Weak	0.20	0.42	19	Weak	0.25	0.34	17	Weak
	SBL	0.13	0.59	20	Weak	0.03	0.89	22	None	0.01	0.97	11	None	-0.29	0.42	10	Weak
	AS	0.43	0.22	10	Medium	0.39	0.21	12	Medium	0.77	0.01	9	Strong	0.79	0.01	10	Strong
C3LS	SBC	-0.24	0.48	11	Weak	-0.13	0.72	10	Weak	-0.06	0.82	17	None	0.11	0.68	16	Weak
	SBL	-0.18	0.45	20	Weak	-0.16	0.46	22	Weak	0.07	0.84	12	None	0.06	0.86	12	None
	AS	0.06	0.85	11	None	0.21	0.49	13	Weak	0.47	0.17	10	Medium	0.51	0.11	11	Strong

					Male	2							Fen	nale			
		Le	eft Femor	al Robus	sticity	Rigl	nt Femor	al Robus	ticity	Lef	t Femora	Robustici	ty	R	ight Femor	al Robu	sticity
Facet	Sample	C Coefficient	Ρ	N	Strength of Correlation	C Coefficie nt	Ρ	N	Strength of Correlatio n	C Coefficient	Ρ	N	Strength of Correlati on	C Coeffici ent	Ρ	N	Strength of Correlation
C3RS	SBC	0.12	0.71	12	Weak	-0.14	0.67	11	Weak	-0.01	0.97	17	None	0.42	0.11	16	Medium
	SBL	0.04	0.86	20	None	-0.10	0.66	22	Weak	0.14	0.69	11	Weak	-0.10	0.76	11	Weak
	AS	0.02	0.94	12	None	0.10	0.73	14	Weak	0.69	0.03	10	Strong	0.70	0.02	11	Strong
C3LI	SBC	-0.25	0.43	12	Weak	-0.42	0.20	11	Medium	-0.06	0.81	17	None	-0.10	0.71	16	Weak
	SBL	0.38	0.11	19	Medium	0.26	0.25	21	Weak	0.02	0.94	12	None	-0.19	0.56	12	Weak
	AS	0.03	0.92	11	None	0.08	0.79	13	None	0.16	0.65	10	Weak	0.23	0.49	11	Weak
C3RI	SBC	-0.13	0.69	12	Weak	-0.09	0.79	11	None	-0.01	0.97	17	None	0.07	0.81	16	None
	SBL	0.41	0.07	20	Medium	0.17	0.46	22	Weak	-0.08	0.82	11	None	-0.19	0.57	11	Weak
	AS	0.46	0.16	11	Medium	0.40	0.20	12	Medium	0.31	0.38	10	Medium	0.32	0.33	11	Medium
C4LS	SBC	-0.28	0.37	12	Weak	-0.29	0.39	11	Weak	-0.14	0.62	15	Weak	-0.24	0.38	15	Weak
0.120	SBI	0.10	0.67	21	None	-0.02	0.94	23	None	-0.13	0.66	14	Weak	0.03	0.93	11	None
		-0.35	0.32	10	Medium	-0.30	0.34	12	Medium	0.13	0.00	8	Strong	0.55	0.13	9	Strong
CARS	SRC	-0.10	0.75	12	None	-0.07	0.27	11	None	0.01	0.20	15	None	-0.03	0.13	15	None
04115	SBC SBI	0.17	0.75	21	Weak	-0.04	0.05	22	None	-0.05	0.50	1/	None	-0.15	0.51	11	Weak
	AS	0.38	0.23	12	Medium	0.32	0.27	14	Medium	0.21	0.59	9	Weak	0.11	0.77	10	Weak

A	• · · · · · · · · · · · · · ·			C	
Appendix B-7 S	Spearman's rno	correlation betweer	h facet size and "	temoral robusticit	v continued

					Male	2							Fen	nale			
		Le	ft Femor	al Robus	ticity	Righ	nt Femor	al Robi	usticity	Lef	t Femoral	Robustici	ty	R	ight Femora	al Robu	sticity
Facet	Sample	C Coefficient	Ρ	Ν	Strength of Correlation	C Coefficie nt	Ρ	N	Strength of Correlation	C Coefficient	Ρ	Ν	Strength of Correlati on	C Coeffici ent	Ρ	N	Strength of Correlation
C4LI	SBC	-0.33	0.30	12	Medium	-0.28	0.40	11	Weak	-0.18	0.52	15	Weak	0.13	0.63	15	Weak
	SBL	-0.05	0.84	21	None	-0.03	0.88	23	None	0.50	0.06	15	Strong	0.00	0.99	12	None
	AS	0.34	0.34	10	Medium	0.34	0.27	12	Medium	0.16	0.71	8	Weak	0.24	0.54	9	Weak
C4RI	SBC	-0.23	0.46	12	None	-0.03	0.92	11	None	0.71	0.00	15	Strong	0.43	0.11	15	Medium
	SBL	0.11	0.62	21	Weak	-0.01	0.98	23	None	0.37	0.20	14	Medium	0.02	0.94	11	None
	AS	0.61	0.05	11	Strong	0.59	0.03	13	Strong	0.60	0.11	8	Strong	0.26	0.51	9	Weak
C5LS	SBC	0.18	0.56	13	Weak	-0.40	0.20	12	Medium	-0.01	0.98	17	None	-0.07	0.83	14	None
	SBL	-0.05	0.83	21	None	-0.09	0.69	22	None	0.27	0.34	15	Weak	-0.17	0.57	13	Weak
	AS	-0.03	0.94	9	None	-0.15	0.68	10	Weak	-0.32	0.36	10	Medium	-0.10	0.76	12	Weak
C5RS	SBC	-0.17	0.59	13	Weak	-0.13	0.69	12	Weak	-0.02	0.95	17	None	0.42	0.13	14	Medium
	SBL	0.38	0.09	21	Medium	0.16	0.49	22	Weak	0.27	0.33	15	Weak	-0.30	0.33	13	Medium
	AS	-0.28	0.47	9	Weak	-0.34	0.34	10	Medium	0.25	0.49	10	Weak	0.10	0.75	12	Weak
C5LI	SBC	-0.04	0.89	13	None	-0.09	0.77	12	None	-0.09	0.74	17	None	0.32	0.26	14	Medium
	SBL	0.15	0.52	20	Weak	0.02	0.94	21	None	0.27	0.34	15	Weak	-0.12	0.71	13	Weak
	AS	-0.37	0.33	9	Medium	-0.38	0.28	10	Medium	0.77	0.02	9	Strong	0.80	0.00	11	Strong

					M	ale							Fen	nale			
			.eft Femora	l Robust	icity	R	ight Femora	al Robusti	city	l	_eft Femora	l Robust	icity	R	ight Femora	al Robu	sticity
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlation
C5RI	SBC		1.00	13	None	-0.14	0.67	12	Weak	0.17	0.51	17	Weak	0.66	0.01	14	Strong
	SBL		0.29	20	Weak	0.08	0.73	21	None	0.27	0.33	15	Weak	-0.33	0.27	13	Medium
	AS		0.52	9	Weak	-0.18	0.62	10	Weak	0.65	0.04	10	Strong	0.54	0.07	12	Strong
C6LS	SBC		0.65	9	Weak	-0.08	0.82	10	None	-0.26	0.24	22	Weak	-0.14	0.58	19	Weak
	SBL		0.64	23	Weak	0.03	0.91	24	None	0.45	0.11	14	Medium	-0.01	0.96	13	None
	AS		0.31	10	Medium	-0.32	0.31	12	Medium	0.52	0.19	8	Strong	0.49	0.22	8	Medium
C6RS	SBC		0.70	9	Weak	-0.17	0.64	10	Weak	-0.26	0.23	22	Weak	-0.14	0.57	19	Weak
	SBL		0.25	22	Weak	0.13	0.54	23	Weak	0.07	0.80	15	None	-0.28	0.36		Weak
	AS		0.78	11	None	-0.04	0.89	13	None	0.62	0.10	8	Strong	0.22	0.57	9	Weak
C6LI	SBC		0.71	9	Weak	0.03	0.94	10	None	0.48	0.02	22	Medium	0.38	0.10	19	Medium
	SBL		0.59	24	Weak	0.03	0.90	25	None	0.10	0.74	14	None	-0.03	0.92	13	None
	AS		0.44	10	Weak	0.01	0.99	12	None	0.35	0.39	8	Medium	0.28	0.47	9	Weak
C6RI	SBC		0.88	9	None	0.12	0.73	10	Weak	0.43	0.05	22	Medium	0.32	0.18	19	Medium
	SBL		0.85	23	None	-0.01	0.96	24	None	-0.20	0.50	14	Weak	-0.04	0.88	13	None
	AS	-0.18	0.63	10	Weak	-0.09	0.78	12	None	0.39	0.34	8	Medium	0.52	0.15	9	Strong

					М	ale							Fen	nale			
		1	.eft Femo	oral Robu	sticity	R	ight Femo	ral Robu	sticity	l	.eft Femora	l Robust	icity	R	ight Femor	al Robu	sticity
Facet	Sample	C Coeffici ent	Р	N	Strength of Correlation	C Coeffici ent	Р	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlation
C7LS	SBC	-0.08	0.81	11	None	-0.08	0.81	11	None	0.11	0.64	21	Weak	0.37	0.13	18	Medium
	SBL	0.23	0.26	25	Weak	0.13	0.54	25	Weak	0.00	0.99	15	None	-0.29	0.34	13	Weak
	AS	-0.36	0.39	8	Medium	-0.26	0.46	10	Weak	0.58	0.23	6	Strong	0.59	0.22	6	Strong
C7RS	SBC	-0.41	0.21	11	Medium	-0.08	0.82	11	None	-0.01	0.97	20	None	0.28	0.28	17	Weak
	SBL	0.24	0.26	24	Weak	0.28	0.18	25	Weak	-0.11	0.70	15	Weak	-0.22	0.48	13	Weak
	AS	-0.04	0.90	11	None	0.21	0.50	13	Weak	0.50	0.25	7	Strong	0.51	0.24	7	Strong
C7LI	SBC	-0.44	0.15	12	Medium	0.08	0.81	11	None	0.31	0.17	21	Medium	0.25	0.31	18	Weak
	SBL	0.21	0.33	24	Weak	-0.07	0.74	26	None	0.14	0.63	15	Weak	-0.05	0.87	13	None
	AS	-0.05	0.90	10	None	-0.09	0.79	12	None	0.50	0.32	6	Strong	0.51	0.30	6	Strong
C7RI	SBC	-0.27	0.40	12	Weak	0.00	0.99	11	None	0.34	0.14	21	Medium	-0.12	0.65	18	Weak
	SBL	0.06	0.79	24	None	0.06	0.75	26	None	0.06	0.85	15	None	-0.24	0.42	13	Weak
	AS	0.21	0.53	11	Weak	0.07	0.83	13	None	0.32	0.48	7	Medium	0.33	0.48	7	Medium
T1LS	SBC	0.01	0.97	11	None	-0.05	0.90	10	None	-0.14	0.54	22	Weak	0.16	0.52	18	Weak
	SBL	-0.12	0.55	25	Weak	-0.07	0.72	26	None	-0.20	0.52	13	Weak	-0.02	0.94	12	None
	AS	-0.12	0.74	11	Weak	-0.16	0.65	11	Weak	0.46	0.30	7	Medium	0.38	0.27	10	Medium

		www.allast?a.a.la.ast.c.a.a.u	faat state at a state	f	
$\Delta n n e n a i x K_{-} / N$	nearman's rno co	rrelation netweer	i tacet size and '	temoral robusticit\	/ continued
			1 10000 3120 0110		continucu

					M	ale							Fen	nale			
		L	eft Femo	ral Robu	sticity	R	ight Femo	ral Robus	sticity	l	_eft Femora	l Robust	icity	R	ight Femor	al Robu	sticity
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	Ν	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Р	N	Strength of Correlation
T1RS	SBC	0.20	0.52	12	Weak	-0.33	0.33	11	Medium	-0.05	0.83	22	None	0.24	0.33	18	Weak
	SBL	0.02	0.92	25	None	0.09	0.65	26	None	-0.21	0.48	14	Weak	-0.08	0.81	12	None
	AS	-0.04	0.90	12	None	0.08	0.79	13	None	-0.05	0.91	8	None	-0.06	0.85	11	None
T1LI	SBC	0.00	0.99	12	None	0.08	0.82	11	None	0.05	0.83	22	None	0.02	0.94	18	None
	SBL	0.43	0.03	26	Medium	0.07	0.73	26	None	0.21	0.46	14	Weak	-0.03	0.93	12	None
	AS	-0.10	0.75	14	Weak	-0.12	0.66	15	Weak	0.21	0.61	8	Weak	0.04	0.91	11	None
T1RI	SBC	0.11	0.72	12	Weak	-0.37	0.27	11	Medium	0.27	0.22	22	Weak	0.36	0.15	18	Medium
	SBL	0.36	0.07	26	Medium	0.03	0.90	26	None	0.04	0.90	14	None	-0.13	0.69	12	Weak
	AS	-0.37	0.23	12	Medium	-0.39	0.18	13	Medium	-0.08	0.85	8	None	0.00	0.99	11	None
T2LS	SBC	-0.28	0.35	13	Weak	-0.05	0.88	12	None	0.26	0.25	21	Weak	0.49	0.05	17	Medium
	SBL	0.14	0.52	25	Weak	-0.15	0.47	25	Weak	0.20	0.52	13	Weak	-0.13	0.69	12	Weak
	AS	-0.35	0.24	13	Medium	-0.37	0.19	14	Medium	-0.16	0.63	11	Weak	-0.18	0.54	14	Weak
T2RS	SBC	-0.30	0.32	13	Medium	-0.20	0.52	12	Weak	0.30	0.19	21	Medium	0.52	0.03	17	Strong
	SBL	0.14	0.50	26	Weak	-0.08	0.71	26	None	0.30	0.33	13	Medium	-0.30	0.34	12	Medium
	AS	-0.37	0.19	14	Medium	-0.36	0.17	16	Medium	0.09	0.79	11	None	0.07	0.81	14	None

					М	ale							Fen	nale			
		Le	eft Femora	l Robusti	city	R	ight Femoral	Robus	ticity		Left Femora	al Robu	sticity	R	ight Femor	al Robu	sticity
Facet	Sample	C Coefficie nt	Ρ	N	Strength of Correlatio n	C Coefficie nt	Ρ	N	Strength of Correlation	C Coeffici ent	Р	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlation
T2LI	SBC	0.06	0.84	13	None	-0.33	0.29	12	Medium	0.19	0.40	21	Weak	0.11	0.67	17	Weak
	SBL	0.23	0.28	25	Weak	-0.05	0.82	25	None	0.42	0.16	13	Medium	-0.32	0.31	12	Medium
	AS	-0.29	0.29	15	Weak	-0.16	0.54	17	Weak	0.02	0.96	10	None	-0.17	0.57	13	Weak
T2RI	SBC	-0.12	0.69	13	Weak	-0.02	0.94	12	None	-0.16	0.48	21	Weak	0.16	0.53	17	Weak
	SBL	0.14	0.50	25	Weak	0.05	0.82	25	None	0.66	0.01	13	Strong	-0.17	0.60	12	Weak
	AS	-0.50	0.06	15	Strong	-0.22	0.41	16	Weak	0.04	0.91	10	None	-0.05	0.87	13	None
T3LS	SBC	-0.09	0.77	13	None	-0.08	0.80	12	None	0.03	0.88	21	None	0.04	0.88	17	None
	SBL	0.04	0.85	21	None	-0.14	0.55	21	Weak	0.53	0.07	12	Strong	-0.21	0.52	12	Weak
	AS	-0.20	0.47	15	Weak	-0.20	0.48	15	Weak	-0.01	0.97	12	None	0.15	0.60	14	Weak
T3RS	SBC	-0.18	0.56	13	Weak	-0.29	0.35	12	Weak	0.01	0.98	21	None	0.43	0.09	17	Medium
	SBL	0.03	0.89	22	None	-0.10	0.65	22	Weak	0.61	0.03	12	Strong	-0.24	0.45	12	Weak
	AS	-0.21	0.44	16	Weak	0.03	0.92	18	None	0.18	0.57	12	Weak	0.14	0.62	15	Weak
T3LI	SBC	0.30	0.32	13	Medium	-0.18	0.57	12	Weak	-0.05	0.84	21	None	-0.05	0.86	17	None
-	SBL	0.10	0.64	22	Weak	-0.19	0.40	22	Weak	0.48	0.12	12	Medium	-0.30	0.35	12	Medium
	AS	0.13	0.64	15	Weak	0.18	0.49	17	Weak	0.02	0.95	11	None	0.17	0.57	13	Weak

					M	ale							Fen	nale			
			Left Femora	l Robustic	city	R	light Femor	al Robi	usticity		Left Femora	l Robu	sticity	F	ight Femor	al Robu	sticity
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlation
T3RI	SBC	-0.14	0.64	13	Weak	-0.22	0.50	12	Weak	-0.11	0.65	21	Weak	-0.15	0.56	17	Weak
	SBL	0.05	0.83	22	None	-0.10	0.65	22	Weak	0.71	0.01	12	Strong	-0.21	0.51	12	Weak
	AS	-0.40	0.14	15	Medium	-0.49	0.05	17	Medium	0.09	0.78	12	None	0.16	0.58	15	Weak
T4LS	SBC	0.14	0.64	13	Weak	-0.16	0.62	12	Weak	0.37	0.11	20	Medium	0.39	0.13	16	Medium
	SBL	0.11	0.63	23	Weak	-0.13	0.57	22	Weak	0.26	0.46	10	Weak	0.21	0.56	10	Weak
	AS	-0.12	0.66	15	Weak	-0.05	0.85	18	None	-0.27	0.35	14	Weak	-0.39	0.13	16	Medium
T4RS	SBC	-0.39	0.19	13	Medium	-0.21	0.51	12	Weak	0.01	0.97	19	None	-0.05	0.87	15	None
	SBL	-0.05	0.82	22	None	-0.12	0.59	21	Weak	0.70	0.02	10	Strong	-0.23	0.52	10	Weak
	AS	-0.23	0.40	15	Weak	-0.20	0.44	18	Weak	-0.14	0.63	14	None	-0.26	0.32	16	Weak
T4LI	SBC	-0.08	0.79	13	None	0.05	0.87	12	None	0.02	0.94	19	None	0.33	0.21	16	Medium
	SBL	0.18	0.41	22	Weak	-0.01	0.96	21	None	0.48	0.19	9	Medium	-0.14	0.72	9	Weak
	AS	0.07	0.82	14	None	0.13	0.61	17	Weak	-0.14	0.64	13	None	0.10	0.72	15	Weak
T4RI	SBC	0.23	0.50	11	Weak	0.06	0.87	10	None	-0.48	0.09	13	Medium	-0.83	0.01	9	Strong
	SBL	0.14	0.61	15	Weak	-0.09	0.77	14	None	0.07	0.90	6	None	-0.53	0.28	6	Strong
	AS	-0.32	0.33	11	Medium	-0.44	0.14	13	Medium	-0.71	0.05	8	Strong	-0.62	0.06	10	Strong

					Μ	lale							Fer	nale			
			Left Femora	al Robus	ticity	R	ight Femora	al Robus [.]	ticity		Left Femora	l Robus	sticity	R	ight Femor	al Robu	sticity
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlation
T5LS	SBC	-0.08	0.80	12	None	-0.07	0.82	12	None	0.17	0.46	20	Weak	0.45	0.07	17	Medium
	SBL	-0.09	0.69	21	None	-0.26	0.26	20	Weak	0.62	0.04	11	Strong	-0.18	0.61	10	Weak
	AS	-0.31	0.30	13	Medium	-0.16	0.55	16	Weak	0.19	0.55	12	Weak	0.07	0.82	14	None
T5RS	SBC	0.01	0.98	12	None	-0.51	0.09	12	Strong	-0.18	0.45	20	Weak	-0.43	0.08	17	Medium
	SBL	0.14	0.54	21	Weak	-0.11	0.65	20	Weak	0.29	0.39	11	Weak	-0.23	0.52	10	Weak
	AS	-0.27	0.35	14	Weak	-0.30	0.25	17	Medium	0.03	0.91	12	None	0.02	0.95	14	None
T5LI	SBC	0.01	0.98	12	None	0.08	0.80	12	None	0.18	0.45	20	Weak	0.06	0.81	17	None
	SBL	0.47	0.04	20	Medium	0.15	0.57	18	Weak	0.23	0.48	12	Weak	-0.03	0.93	10	None
	AS	-0.47	0.13	12	Medium	-0.15	0.62	14	Weak	0.32	0.33	11	Medium	0.30	0.32	13	Medium
T5RI	SBC	-0.17	0.60	12	Weak	-0.14	0.67	12	Weak	0.11	0.65	20	Weak	-0.27	0.29	17	Weak
	SBL	0.21	0.37	21	Weak	-0.06	0.81	20	None	0.04	0.89	12	None	-0.22	0.53	10	Weak
	AS	-0.65	0.02	13	Strong	-0.48	0.07	15	Medium	0.07	0.85	11	None	-0.11	0.72	13	Weak
T6LS	SBC	0.39	0.19	13	Medium	-0.03	0.93	12	None	-0.07	0.76	20	None	0.05	0.84	17	None
	SBL	0.21	0.33	24	Weak	-0.08	0.72	23	None	0.32	0.31	12	Medium	-0.36	0.28	11	Medium
	AS	-0.42	0.20	11	Medium	0.16	0.59	14	Weak	0.17	0.62	11	Weak	-0.09	0.76	13	None

					Ma	ale							Fem	nale			
		L	eft Femoral	l Robustio	city	Ri	ight Femora	l Robus	ticity	L	eft Femora	l Robusti	city	R	ight Femora	al Robu	sticity
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlation	C Coefficie nt	Ρ	N	Strength of Correlation	C Coeffici ent	Р	N	Strength of Correlation
T6RS	SBC	-0.07	0.81	13	None	-0.46	0.13	12	Medium	-0.25	0.30	20	Weak	-0.26	0.31	17	Weak
	SBL	0.28	0.19	24	Weak	0.00	0.99	23	None	-0.05	0.89	12	None	-0.20	0.56	11	Weak
	AS	-0.64	0.03	11	Strong	-0.29	0.31	14	Weak	0.14	0.68	11	Weak	0.05	0.87	13	None
T6LI	SBC	-0.23	0.45	13	Weak	-0.46	0.13	12	Medium	0.33	0.15	21	Medium	-0.05	0.86	17	None
	SBL	0.06	0.77	23	None	-0.18	0.42	22	Weak	0.06	0.86	12	None	-0.19	0.57	11	Weak
	AS	0.22	0.57	9	Weak	0.26	0.41	12	Weak	0.14	0.68	11	Weak	-0.19	0.53	13	Weak
T6RI	SBC	-0.14	0.65	13	Weak	-0.31	0.33	12	Medium	0.16	0.49	21	Weak	0.24	0.35	17	Weak
	SBL	-0.03	0.88	22	None	-0.33	0.15	21	Medium	-0.10	0.75	12	Weak	-0.23	0.50	11	Weak
	AS	0.09	0.80	10	None	0.01	0.98	12	None	0.22	0.51	11	Weak	-0.28	0.35	13	Weak
T7LS	SBC	-0.29	0.33	13	Weak	-0.57	0.05	12	Strong	0.29	0.20	21	Weak	0.42	0.10	17	Medium
	SBL	0.20	0.36	23	Weak	-0.07	0.76	23	None	0.10	0.75	12	Weak	-0.23	0.50	11	Weak
	AS	-0.11	0.78	9	Weak	-0.24	0.47	11	Weak	0.06	0.85	14	None	0.00	1.00	16	None
T7RS	SBC	-0.06	0.85	13	None	-0.33	0.29	12	Medium	0.41	0.06	21	Medium	0.46	0.06	17	Medium
	SBL	0.06	0.80	24	None	-0.18	0.40	24	Weak	-0.18	0.57	12	Weak	-0.19	0.59	10	Weak
	AS	0.00	0.99	10	None	0.03	0.93	12	None	0.10	0.72	14	Weak	-0.05	0.87	16	None

					М	ale							Fer	nale			
			Left Femora	ıl Robusti	city	R	ight Femora	al Robustio	city		Left Femora	l Robust	icity	R	light Femor	al Robusti	city
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlatio n
T7LI	SBC	0.27	0.39	12	Weak	-0.45	0.16	11	Medium	0.67	0.00	21	Strong	0.32	0.21	17	Medium
	SBL	0.21	0.34	23	Weak	-0.08	0.73	22	None	-0.17	0.59	12	Weak	-0.22	0.51	11	Weak
	AS	-0.27	0.49	9	Weak	-0.41	0.21	11	Medium	0.30	0.29	14	Medium	0.18	0.50	16	Weak
T7RI	SBC	0.08	0.81	12	None	0.10	0.78	11	None	0.61	0.00	21	Strong	0.43	0.08	17	Medium
	SBL	-0.05	0.83	23	None	-0.40	0.06	23	Medium	-0.08	0.82	12	None	-0.20	0.55	11	Weak
	AS	-0.21	0.60	9	Weak	-0.22	0.52	11	Weak	0.18	0.57	12	Weak	0.02	0.94	14	None
T8LS	SBC	0.02	0.94	12	None	-0.38	0.25	11	Medium	0.26	0.29	19	Weak	-0.05	0.85	16	None
	SBL	0.33	0.10	25	Medium	0.15	0.49	24	Weak	-0.32	0.31	12	Medium	-0.29	0.37	12	Weak
	AS	0.45	0.19	10	Medium	0.31	0.33	12	Medium	-0.13	0.68	12	Weak	-0.18	0.53	14	Weak
T8RS	SBC	-0.22	0.49	12	Weak	-0.55	0.08	11	Strong	0.20	0.41	19	Weak	-0.08	0.78	16	None
	SBL	0.00	0.98	25	None	-0.20	0.35	24	Weak	-0.12	0.72	12	Weak	-0.32	0.32	12	Medium
	AS	0.32	0.34	11	Medium	0.23	0.45	13	Weak	-0.01	0.97	12	None	-0.16	0.58	14	Weak
T8LI	SBC	-0.20	0.50	13	Weak	-0.34	0.28	12	Medium	-0.09	0.71	19	None	-0.15	0.57	16	Weak
-	SBL	0.03	0.90	25	None	-0.19	0.36	24	Weak	-0.42	0.23	10	Medium	-0.10	0.79	10	Weak
	AS	0.18	0.59	11	Weak	0.04	0.90	13	None	0.26	0.41	12	Weak	0.08	0.78	14	None

Appondix B 7 Spoormon's rbs	corrolation botwoon	facat ciza and fa	moral robucticity	continued
ADDEIIUIX D-7 SDEal IIIdii S III	Correlation between	idlet size and re	illoral lobusticity	continueu

					M	ale							Fen	nale			
			_eft Femora	l Robusti	city	R	light Femora	al Robusti	city		Left Femora	l Robust	icity	R	ight Femora	al Robusti	city
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlatio n
T8RI	SBC	-0.34	0.26	13	Medium	-0.09	0.79	12	None	0.32	0.19	19	Medium	-0.10	0.71	16	Weak
	SBL	0.00	0.98	25	None	-0.20	0.35	24	Weak	-0.12	0.72	12	Weak	-0.32	0.32	12	Medium
	AS	0.20	0.55	11	Weak	0.15	0.64	13	Weak	0.11	0.74	12	Weak	-0.06	0.85	14	None
T9LS	SBC	-0.18	0.56	13	Weak	-0.46	0.13	12	Medium	-0.27	0.23	21	Weak	-0.39	0.11	18	Medium
	SBL	0.15	0.48	25	Weak	-0.12	0.57	24	Weak	-0.07	0.83	11	None	-0.26	0.43	11	Weak
	AS	0.33	0.27	13	Medium	0.31	0.24	16	Medium	-0.10	0.73	14	Weak	-0.27	0.33	15	Weak
T9RS	SBC	-0.06	0.84	13	None	-0.51	0.09	12	Strong	0.06	0.81	21	None	-0.32	0.20	18	Medium
	SBL	0.16	0.45	24	Weak	-0.04	0.85	24	None	-0.03	0.94	12	None	-0.12	0.71	12	Weak
	AS	0.21	0.50	13	Weak	0.20	0.45	16	Weak	-0.03	0.93	14	None	-0.21	0.44	15	Weak
T9LI	SBC	0.20	0.53	12	Weak	-0.18	0.59	11	Weak	0.18	0.44	21	Weak	-0.10	0.70	18	Weak
	SBL	0.15	0.47	24	Weak	-0.12	0.57	24	Weak	-0.14	0.68	12	Weak	-0.12	0.72	12	Weak
	AS	-0.01	0.97	12	None	0.01	0.96	15	None	0.12	0.69	14	Weak	0.04	0.87	15	None
T9RI	SBC	0.40	0.20	12	Medium	-0.36	0.28	11	Medium	0.26	0.25	21	Weak	0.18	0.47	18	Weak
	SBL	0.22	0.31	22	Weak	-0.12	0.61	22	Weak	0.05	0.89	12	None	-0.42	0.18	12	Medium
	AS	0.27	0.43	11	Weak	0.20	0.49	14	Weak	0.09	0.75	14	None	-0.02	0.94	15	None

					Mi	ale			Female								
		L	_eft Femora	l Robustio	city	R	ight Femora	city		Left Femora	l Robusti	icity	Right Femoral Robusticity				
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlatio n
T10LS	SBC	-0.08	0.81	12	None	-0.41	0.21	11	Medium	-0.07	0.75	21	None	-0.29	0.25	18	Weak
	SBL	0.00	0.98	23	None	-0.21	0.35	22	Weak	-0.01	0.97	12	None	0.09	0.78	12	None
	AS	-0.11	0.74	12	Weak	-0.06	0.82	16	None	0.30	0.32	13	Medium	0.14	0.62	14	Weak
T10RS	SBC	-0.46	0.13	12	Medium	-0.17	0.61	11	Weak	-0.09	0.71	21	None	-0.08	0.74	18	None
	SBL	0.13	0.55	23	Weak	0.06	0.79	23	None	0.08	0.80	12	None	-0.21	0.51	12	Weak
	AS	0.04	0.90	12	None	0.09	0.75	15	None	0.22	0.48	13	Weak	-0.07	0.80	14	None
T10LI	SBC	-0.04	0.89	13	None	-0.38	0.23	12	Medium	-0.03	0.89	21	None	-0.10	0.69	18	None
	SBL	-0.04	0.85	23	None	-0.18	0.40	23	Weak	0.10	0.75	12	Weak	-0.20	0.54	12	Weak
	AS	-0.12	0.71	12	Weak	-0.04	0.88	16	None	0.37	0.21	13	Medium	0.01	0.98	14	None
T10RI	SBC	0.08	0.79	13	None	-0.29	0.36	12	Weak	0.16	0.50	21	Weak	0.03	0.90	18	None
	SBL	-0.04	0.86	23	None	-0.17	0.45	23	Weak	0.28	0.37	12	Weak	-0.23	0.47	12	Weak
	AS	0.28	0.38	12	Weak	0.32	0.22	16	Medium	0.22	0.47	13	Weak	-0.03	0.91	14	None
T11LS	SBC	0.20	0.50	13	Weak	-0.45	0.14	12	Medium	0.16	0.50	21	Weak	0.00	0.99	18	None
	SBL	-0.07	0.76	23	None	-0.20	0.36	22	Weak	0.00	1.00	12	None	-0.21	0.51	12	Weak
	AS	0.19	0.57	11	Weak	0.20	0.51	13	Weak	0.09	0.77	13	None	-0.15	0.62	14	Weak

	Sample				Ma	ale			Female								
		l	_eft Femora	l Robusti	city	R	ight Femora	al Robusti	city		Left Femora	l Robusti	city	Right Femoral Robusticity			
Facet		C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlatio n
T11RS	SBC	0.19	0.53	13	Weak	-0.36	0.25	12	Medium	-0.07	0.76	21	None	-0.19	0.45	18	Weak
	SBL	0.23	0.29	23	Weak	0.10	0.66	22	None	0.29	0.34	13	Weak	-0.53	0.08	12	Strong
	AS	-0.15	0.63	12	Weak	-0.10	0.73	15	Weak	0.20	0.52	13	Weak	-0.06	0.84	14	None
T11LI	SBC	-0.16	0.60	13	Weak	-0.32	0.30	12	Medium	0.05	0.83	21	None	0.02	0.95	18	None
	SBL	0.26	0.24	23	Weak	0.20	0.39	21	Weak	0.22	0.47	13	Weak	-0.38	0.22	12	Medium
	AS	0.06	0.86	12	None	0.04	0.90	15	None	0.13	0.68	13	Weak	-0.30	0.30	14	Medium
T11RI	SBC	0.03	0.93	13	None	-0.25	0.43	12	Weak	0.04	0.86	21	None	-0.09	0.71	18	None
	SBL	-0.07	0.74	23	None	-0.17	0.44	22	Weak	-0.01	0.98	13	None	-0.30	0.34	12	Medium
	AS	-0.40	0.25	10	Medium	-0.38	0.20	13	Medium	-0.35	0.30	11	Medium	-0.44	0.15	12	Medium
T12LS	SBC	-0.05	0.88	13	None	-0.18	0.58	12	Weak	0.28	0.25	19	Weak	0.28	0.28	17	Weak
	SBL	0.30	0.18	21	Medium	0.46	0.04	20	Medium	0.55	0.05	13	Strong	-0.32	0.31	12	Medium
	AS	0.01	0.97	10	None	-0.02	0.94	12	None	-0.21	0.48	13	Weak	-0.31	0.28	14	Medium
T12RS	SBC	-0.17	0.58	13	Weak	-0.26	0.42	12	Weak	0.32	0.18	19	Medium	0.22	0.40	17	Weak
	SBL	0.01	0.97	22	None	0.00	0.99	21	None	-0.07	0.82	13	None	-0.32	0.31	12	Medium
	AS	-0.21	0.57	10	Weak	-0.29	0.37	12	Weak	-0.16	0.60	13	Weak	-0.22	0.45	14	Weak

					м	ale			Female								
	Sample	L	_eft Femora	l Robusti	city	R	light Femora	city		Left Femora	ıl Robusti	icity	Right Femoral Robusticity				
Facet		C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlatio n
T12LI	SBC	-0.10	0.74	13	Weak	-0.45	0.14	12	Medium	0.31	0.19	19	Medium	0.23	0.37	17	Weak
	SBL	0.18	0.43	22	Weak	0.00	1.00	21	None	-0.21	0.48	13	Weak	-0.26	0.41	12	Weak
	AS	-0.24	0.51	10	Weak	-0.27	0.39	12	Weak	-0.63	0.04	11	Strong	-0.62	0.03	12	Strong
T12RI	SBC	-0.24	0.43	13	Weak	-0.56	0.06	12	Strong	0.35	0.14	19	Medium	0.20	0.45	17	Weak
	SBL	0.16	0.49	22	Weak	-0.05	0.82	21	None	0.13	0.66	13	Weak	-0.28	0.38	12	Weak
	AS	-0.31	0.39	10	Medium	-0.32	0.34	11	Medium	0.21	0.50	13	Weak	0.10	0.73	14	Weak
L1LS	SBC	-0.31	0.33	12	Medium	-0.32	0.31	12	Medium	0.19	0.41	20	Weak	0.12	0.65	17	Weak
	SBL	0.11	0.64	22	Weak	0.02	0.94	22	None	-0.42	0.17	12	Weak	-0.43	0.17	12	Medium
	AS	-0.46	0.18	10	Medium	-0.54	0.05	13	Strong	0.25	0.44	12	Weak	0.39	0.16	14	Medium
L1RS	SBC	-0.26	0.42	12	Weak	0.02	0.95	12	None	0.00	0.99	19	None	0.13	0.63	17	Weak
	SBL	0.06	0.80	22	None	-0.09	0.68	22	None	-0.20	0.53	12	Weak	-0.22	0.50	12	Weak
	AS	-0.42	0.26	9	Medium	-0.31	0.33	12	Medium	-0.06	0.84	12	None	0.10	0.74	13	Weak
L1LI	SBC	-0.18	0.57	12	Weak	-0.23	0.48	12	Weak	0.38	0.10	20	Medium	0.46	0.06	17	Medium
	SBL	0.19	0.38	23	Weak	0.15	0.48	23	Weak	-0.22	0.50	12	Weak	-0.28	0.38	12	Weak
	AS	-0.51	0.16	9	Strong	-0.44	0.15	12	Medium	0.01	0.97	13	None	-0.04	0.89	15	None

					M	ale			Female										
	Sample		Left Femora	al Robust	icity		Right Femo	oral Robustic	ity		Left Femor	al Robustio	city	Right Femoral Robusticity					
Facet		C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlati on	C Coeffici ent	Ρ	N	Strength of Correlati on	C Coefficie nt	Ρ	N	Strength of Correlatio n		
L1RI	SBC	-0.17	0.59	12	Weak	-0.27	0.40	12	Weak	0.33	0.16	19	Medium	0.25	0.32	17	Weak		
	SBL	0.01	0.96	22	None	-0.01	0.98	22	None	0.11	0.74	12	Weak	-0.20	0.54	12	Weak		
	AS	-0.17	0.64	10	Weak	-0.11	0.72	13	Weak	0.16	0.63	12	Weak	0.09	0.77	14	None		
L2LS	SBC	-0.33	0.28	13	Medium	0.04	0.91	12	None	-0.08	0.75	20	None	0.12	0.63	18	Weak		
	SBL	0.02	0.93	22	None	0.04	0.88	22	None	0.09	0.79	12	None	0.00	0.99	12	None		
	AS	-0.54	0.07	12	Strong	-0.49	0.07	15	Medium	-0.32	0.34	11	Medium	-0.35	0.26	12	Medium		
L2RS	SBC	-0.11	0.72	13	Weak	-0.45	0.14	12	Medium	0.12	0.60	20	Weak	0.17	0.51	18	Weak		
	SBL	0.04	0.87	21	None	0.02	0.94	21	None	0.36	0.25	12	Medium	0.04	0.91	11	None		
	AS	-0.20	0.53	12	Weak	-0.23	0.40	15	Weak	0.07	0.83	11	None	0.07	0.84	12	None		
L2LI	SBC	0.21	0.48	13	Weak	-0.46	0.13	12	Medium	-0.12	0.60	20	Weak	-0.18	0.49	18	Weak		
	SBL	0.15	0.51	23	Weak	0.13	0.55	23	Weak	-0.04	0.91	11	None	-0.36	0.27	11	Medium		
	AS	-0.26	0.46	10	Weak	-0.34	0.25	13	Medium	0.10	0.78	11	None	0.00	1.00	12	None		
L2RI	SBC	-0.15	0.63	13	Weak	-0.48	0.12	12	Medium	0.02	0.92	20	None	0.07	0.79	18	None		
	SBL	0.11	0.61	23	Weak	0.10	0.65	23	None	-0.04	0.89	12	None	0.00	1.00	12	None		
	AS	-0.32	0.37	10	Medium	-0.37	0.22	13	Medium	-0.13	0.70	12	Weak	-0.25	0.40	13	Weak		
			Male									Female							
-------	-------	------------------	--------	---------	----------------------------	------------------	----------	---------	----------------------------	----------------------	---------	----------	----------------------------	----------------------	-----------	----------	----------------------------	--	--
	Sampl	Left	Femora	ıl Robu	ısticity	Righ	nt Femor	al Robu	sticity	Le	ft Femo	ral Robi	usticity	Rigl	ht Femora	ıl Robus	ticity		
Facet	e	C Coefficient	Ρ	N	Strength of Correlation	C Coefficient	Ρ	Ν	Strength of Correlation	C Coefficien t	Ρ	Ν	Strength of Correlation	C Coefficien t	Ρ	N	Strength of Correlation		
L3LS	SBC	-0.10	0.74	13	Weak	-0.60	0.04	12	Strong	0.35	0.12	21	Medium	0.11	0.67	18	Weak		
	SBL	0.03	0.90	20	None	0.09	0.70	20	None	-0.02	0.94	13	None	-0.11	0.75	11	Weak		
	AS	-0.49	0.11	12	Medium	-0.55	0.04	14	Strong	0.20	0.50	14	Weak	-0.05	0.86	15	None		
L3RS	SBC	-0.13	0.66	13	Weak	-0.47	0.12	12	Medium	0.28	0.25	19	Weak	0.29	0.27	17	Weak		
	SBL	0.03	0.91	20	None	0.11	0.66	19	Weak	0.16	0.61	12	Weak	0.02	0.95	11	None		
	AS	-0.27	0.48	9	Weak	-0.33	0.29	12	Medium	-0.03	0.94	11	None	-0.06	0.86	12	None		
L3LI	SBC	0.19	0.54	13	Weak	-0.35	0.27	12	Medium	-0.13	0.59	21	Weak	-0.05	0.85	18	None		
	SBL	-0.01	0.96	21	None	-0.03	0.89	21	None	-0.07	0.82	13	None	-0.18	0.58	12	Weak		
	AS	-0.09	0.78	12	None	0.04	0.88	15	None	0.02	0.95	14	None	0.09	0.74	15	None		
L3RI	SBC	-0.09	0.76	13	None	-0.35	0.26	12	Medium	-0.07	0.76	21	None	-0.28	0.25	18	Weak		
	SBL	0.06	0.82	20	None	-0.08	0.74	21	None	0.07	0.82	13	None	-0.12	0.70	12	Weak		
	AS	-0.29	0.37	12	Weak	-0.08	0.78	13	None	-0.20	0.50	13	Weak	-0.26	0.36	14	Weak		
L4LS	SBC	0.37	0.22	13	Medium	-0.31	0.32	12	Medium	-0.30	0.19	21	Medium	-0.26	0.30	18	Weak		
	SBL	-0.13	0.58	19	Weak	-0.10	0.71	18	None	0.77	0.01	10	Strong	-0.01	0.97	11	None		
	AS	-0.29	0.36	12	Weak	-0.17	0.53	15	Weak	-0.07	0.81	14	None	0.01	0.97	15	None		

Appendix B-7 Spearman's rho correlation between facet size and femoral robusticity continued

A manager of the D T C	n a a musa a la mba a a s		. faaat dina amd	fama and nakes at at	
$\Delta n n e n a i x K - / N$	nearman s rno cor	relation netweel	n tacet size and	temoral robusticity	/ CONTINUED
Appendix D / 3	pearman 5 mo coi		i lucct Size ullu	icilioral robasticit	Continucu

					Μ	ale				Female							
			Left Femor	ral Robust	icity	Right Femoral Robusticity					Left Femor	al Robustic	city	F	Right Femor	al Robustic	city
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlati on	C Coeffici ent	Ρ	N	Strength of Correlati on	C Coefficie nt	Ρ	N	Strength of Correlatio n
L4RS	SBC	0.14	0.65	13	Weak	-0.13	0.70	12	Weak	0.23	0.31	21	Weak	0.04	0.88	18	None
	SBL	0.01	0.95	20	None	-0.02	0.94	19	None	0.65	0.04	10	Strong	-0.14	0.69	10	Weak
	AS	-0.31	0.30	13	Medium	-0.13	0.64	16	Weak	-0.36	0.20	14	Medium	-0.35	0.20	15	Medium
L4LI	SBC	-0.35	0.24	13	Medium	-0.37	0.23	12	Medium	0.26	0.26	21	Weak	0.08	0.76	18	None
	SBL	-0.06	0.82	19	None	0.04	0.88	18	None	0.64	0.05	10	Strong	-0.23	0.50	11	Weak
	AS	-0.24	0.45	12	Weak	-0.12	0.67	15	Weak	-0.30	0.34	12	Medium	-0.29	0.34	13	Weak
L4RI	SBC	-0.08	0.81	13	None	-0.22	0.49	12	Weak	0.40	0.07	21	Medium	0.00	0.99	18	None
	SBL	-0.12	0.60	20	Weak	-0.02	0.94	19	None	0.76	0.01	10	Strong	-0.15	0.65	11	Weak
	AS	0.23	0.46	12	Weak	0.32	0.27	14	Medium	0.13	0.70	12	Weak	0.08	0.80	13	None
L5LS	SBC	0.39	0.22	12	Medium	-0.06	0.85	11	None	0.18	0.47	18	Weak	-0.12	0.67	16	Weak
	SBL	0.14	0.55	22	Weak	0.10	0.65	22	Weak	0.31	0.33	12	Medium	0.15	0.66	11	Weak
	AS	0.10	0.78	10	None	0.14	0.64	13	Weak	-0.20	0.50	13	Weak	-0.25	0.38	14	Weak
L5RS	SBC	-0.16	0.62	12	Weak	-0.01	0.97	11	None	-0.09	0.72	18	None	-0.28	0.29	16	Weak
	SBL	0.02	0.93	20	None	0.11	0.64	20	Weak	-0.11	0.73	12	Weak	0.14	0.69	11	Weak
	AS	0.10	0.77	10	Weak	0.17	0.58	13	Weak	-0.31	0.30	13	Medium	-0.25	0.39	14	Weak

			Male								Female							
Facet	Sample																	
L5LI	SBC	-0.02	0.96	12	None	-0.12	0.72	11	Weak	0.15	0.56	18	Weak	-0.20	0.47	16	Weak	
	SBL	0.12	0.61	22	Weak	-0.39	0.08	22	Medium	0.06	0.86	12	None	-0.05	0.88	11	None	
	AS	-0.03	0.94	9	None	0.04	0.90	12	None	-0.12	0.72	12	Weak	-0.22	0.47	13	Weak	
L5RI	SBC	0.11	0.73	12	Weak	0.03	0.94	11	None	0.15	0.54	18	Weak	-0.02	0.95	16	None	
	SBL	0.05	0.83	21	None	-0.24	0.29	21	Weak	0.19	0.55	12	Weak	-0.05	0.88	11	None	
	AS	0.20	0.59	10	Weak	0.28	0.36	13	Weak	-0.37	0.21	13	Medium	-0.37	0.19	14	Medium	

Appendix B-7 Spearman's rho correlation between facet size and femoral robusticity continued

Appendix B-8 Correlation between humeral and facet size directional asymmetry

			Ma	ale			Fema	le	
Facet	Sample	Continuity Coefficient	Р	Phi	Phi correlation Strength	Continuity Coefficient	Р	Phi	Phi correlation Strength
	SBC	0.00	1.00	0.13	Weak	0.23	0.63	0.17	Weak
C1S	SBL	0.44	0.51	0.26	Weak	0.00	1.00	-0.12	Weak
	AS	0.32	0.57	0.47	Medium	No data			
	SBC	0.00	1.00	-0.16	Weak	0.02	0.90	-0.10	Weak
C1I	SBL	0.00	1.00	-0.09	None	0.00	1.00	-0.06	None
	AS	0.00	1.00	0.15	Weak				
	SBC	0.56	0.46	-0.32	Medium	1.26	0.26	-0.28	Weak
C2S	SBL	0.20	0.65	-0.23	Weak	1.13	0.29	0.38	Medium
	AS	0.10	0.74	-0.42	Medium	0.00	1.00	-0.10	Weak
	SBC	0.00	1.00	-0.05	None	0.00	0.97	0.08	None
C2I	SBL	0.00	1.00	0.07	None	0.00	0.98	0.13	Weak
	AS	0.00	1.00	0.00	None	0.00	1.00	0.10	Weak
	SBC	0.29	0.59	0.29	Weak	0.00	1.00	0.04	None
C3S	SBL	0.11	0.74	0.18	Weak	0.35	0.55	0.26	Weak
	AS	0.00	1.00	-0.15	Weak	0.00	1.00	0.10	Weak
	SBC	0.35	0.55	0.32	Medium	1.67	0.20	-0.34	Medium
C3I	SBL	0.00	0.98	-0.12	Weak	0.00	1.00	0.03	None
	AS	0.00	1.00	-0.15	Weak	0.00	1.00	0.10	Weak
	SBC	0.00	1.00	-0.06	None	0.00	1.00	0.00	None
C4S	SBL	0.00	0.98	-0.12	Weak	3.58	0.06	0.54	Strong
	AS	1.18	0.28	-0.73	Strong				
	SBC	0.00	1.00	0.07	None	0.02	0.88	-0.11	Weak
CC4I	SBL	0.47	0.49	-0.25	Weak	1.88	0.17	0.41	Medium
	AS	0.00	1.00	-0.09	None				
	SBC	0.23	0.64	-0.22	Weak	0.02	0.89	-0.10	Weak
C5S	SBL	0.04	0.83	-0.15	Weak	2.31	0.13	0.43	Medium
	AS	0.00	1.00	0.00	None	0.00	1.00	0.10	Weak

Appendix B-8 Correlation between humeral and facet size directional asymmetry continued

			N	1ale			Fer	nale	
Facet	Sample	Continuity Coefficient	Р	Phi	Phi correlation Strength	Continuity Coefficient	Р	Phi	Phi correlation Strength
	SBC	1.83	0.18	-0.43	Medium	0.00	1.00	0.05	None
C5I	SBL	0.01	0.91	0.13	Weak	0.08	0.78	0.16	Weak
	AS	0.09	0.76	0.50	Strong				
	SBC	0.00	1.00	-0.07	None	0.03	0.87	-0.11	Weak
C6S	SBL	0.00	1.00	-0.07	None	0.00	1.00	0.01	None
	AS	0.11	0.74	-0.42	Medium				
	SBC					0.05	0.82	0.24	weak
C6I	SBL	0.02	0.88	-0.19	Weak	0.04	0.83	0.28	Weak
	AS								
	SBC	0.00	1.00	0.00	None	0.00	1.00	0.03	None
C7S	SBL	0.00	1.00	0.06	None	0.95	0.33	0.31	Medium
	AS	0.37	0.55	0.55	Strong				
	SBC	0.47	0.49	-0.29	Weak	0.00	1.00	-0.01	None
C7!	SBL	0.53	0.47	-0.25	Weak	0.00	1.00	0.01	None
	AS	0.00	1.00	-0.09	None				
	SBC	0.00	1.00	0.00	None	0.12	0.73	-0.14	Weak
T1S	SBL	0.00	1.00	0.06	None	0.00	1.00	-0.08	None
	AS	0.18	0.67	0.45	Medium				
	SBC	0.00	1.00	-0.13	Weak	0.52	0.47	-0.13	Weak
T1I	SBL	0.81	0.37	-0.27	Weak	0.01	0.93	0.12	Weak
	AS	0.00	1.00	0.16	Weak				
	SBC	0.00	1.00	-0.08	None	5.17	0.23	0.48	Medium
T2S	SBL	0.00	1.00	-0.04	None	0.67	0.41	-0.30	Weak
	AS	0.03	0.86	0.33	Medium				
	SBC	0.00	1.00	0.00	None	1.76	0.19	0.31	Medium
T2I	SBL	2.40	0.12	0.40	Medium	4.02	0.05	-0.57	Strong
	AS	0.00	1.00	-0.07	None				
	SBC	0.38	0.54	-0.27	Weak	5.17	0.02	0.48	Medium
T3S	SBL	0.71	0.40	0.29	Weak	0.03	0.87	0.16	Weak
	AS	0.00	1.00	-0.07	None				

Appendix B Facet Size

	-	
Appendix B-8 Correlation between humeral and facet size directional a	svmmetrv	<i>i</i> continued
Appendix B o conclution between numeral and facet size an ethorial a	<i>y</i>	continueu

			1	Male			Fe	emale	
Facet	Sample	Continuity Coefficient	Р	Phi	Phi correlation Strength	Continuity Coefficient	Ρ	Phi	Phi correlation Strength
	SBC	1.42	0.23	0.41	Medium	3.03	0.08	0.39	Medium
тзі	SBL	0.21	0.65	-0.21	Weak	0.00	0.96	0.13	Weak
	AS	2.34	0.63	0.41	Medium				
	SBC	0.00	0.96	0.13	Weak	3.41	0.07	0.42	Medium
T4S	SBL	0.00	1.00	-0.12	Weak	0.33	0.57	0.27	Weak
	AS	0.00	1.00	0.13	Weak	0.00	1.00	-0.17	Weak
	SBC	0.00	1.00	-0.06	None	1.61	0.20	-0.32	Medium
T4I	SBL	0.86	0.36	-0.30	Medium	0.00	1.00	-0.12	Weak
	AS	0.00	1.00	0.00	None	0.00	1.00	-0.41	Medium
	SBC	0.00	1.00	-0.06	None	0.00	1.00	-0.07	None
T5S	SBL	0.46	0.50	0.27	Weak	0.09	0.77	0.21	Weak
	AS	0.05	0.83	-0.25	Weak	0.00	1.00	-0.58	Strong
	SBC	0.23	0.64	-0.22	Weak	0.02	0.89	0.11	Weak
T5I	SBL	0.42	0.52	-0.25	Weak	0.04	0.84	-0.17	Weak
	AS	0.33	0.57	-0.39	Medium	0.00	1.00	-0.33	Medium
	SBC	0.14	0.71	0.20	Weak	0.00	1.00	-0.05	None
T6S	SBL	0.50	0.95	-0.13	Weak	0.00	1.00	-0.11	Weak
	AS	0.00	1.00	0.35	Medium	0.44	0.51	1.00	Strong
	SBC	0.00	1.00	0.08	None	0.00	1.00	-0.07	None
T6I	SBL	0.00	1.00	-0.12	Weak	1.43	0.23	0.40	Medium
	AS	0.00	1.00	-0.07	None	0.00	1.00	0.58	Strong
	SBC	0.00	1.00	0.03	None	0.00	1.00	-0.07	None
T7S	SBL	0.00	1.00	0.10	None	0.14	0.71	0.20	Weak
	AS	0.14	0.71	-0.35	Medium	0.00	1.00	0.41	Medium
	SBC	0.50	0.48	0.31	Medium	3.76	0.05	0.43	Medium
T7I	SBL	0.00	1.00	0.09	None	0.00	1.00	-0.19	Weak
	AS	0.00	1.00	0.10	Weak	0.31	0.58	-0.67	Strong
	SBC								
T8S	SBL								
	AS	0.05	0.83	-0.41	Medium				

Appendix B-8 Correlation between humeral and facet size directional asymmetry continued

Facot Sample			N	1ale		Female					
Facet	Sample	Continuity Coefficient	Р	Phi	Phi correlation Strength	Continuity Coefficient	Р	Phi	Phi correlation Strength		
	SBC	0.68	0.41	0.33	Medium	0.00	1.00	0.01	None		
Т8І	SBL	0.00	1.00	0.06	None	0.03	0.86	-0.18	Weak		
	AS	0.42	0.52	0.41	Medium						
	SBC	0.00	1.00	0.00	None	0.01	0.94	0.09	None		
T9S	SBL	0.15	0.70	0.18	Weak	0.59	0.44	0.31	Medium		
	AS	0.00	1.00	0.16	Weak	0.00	1.00	-0.41	Medium		
	SBC	1.81	0.18	-0.45	Medium	0.00	1.00	0.59	Strong		
Т9!	SBL	0.00	1.00	-0.08	None	0.00	1.00	-0.03	None		
	AS	0.00	1.00	0.07	None	0.05	0.82	-0.61	Strong		
	SBC	0.11	0.74	0.20	Weak	0.01	0.94	-0.08	None		
T10S	SBL	0.93	0.34	0.30	Medium	0.81	0.37	0.33	Medium		
	AS	0.97	0.32	-0.60	Strong	0.00	1.00	-0.41	Medium		
	SBC	0.00	1.00	-0.03	None	0.01	0.94	-0.09	None		
T10I	SBL	0.00	1.00	0.00	None	0.00	1.00	0.04	None		
	AS	0.00	1.00	-0.26	Weak	0.00	1.00	-0.41	Medium		
	SBC	0.70	0.40	0.32	Medium	1.31	0.25	-0.29	Weak		
T11S	SBL	0.00	1.00	-0.01	None	0.00	1.00	0.11	Weak		
	AS	0.97	0.32	0.60	Strong	0.00	1.00	-0.41	Medium		
	SBC	0.03	0.88	-0.17	Weak	0.00	1.00	0.00	None		
T11I	SBL	0.00	1.00	0.02	None	0.00	0.98	0.12	Weak		
	AS	1.41	0.24	0.63	Strong	0.00	1.00	0.17	Weak		
	SBC	0.47	0.49	-0.29	Weak	0.00	1.00	-0.04	None		
T12S	SBL	0.00	1.00	0.12	Weak	1.82	0.18	0.44	Medium		
	AS	0.89	0.35	0.60	Strong	0.31	0.58	0.67	Strong		
	SBC	0.23	0.64	-0.22	Weak	0.07	0.80	-0.12	Weak		
T12I	SBL	0.00	1.00	0.04	None	1.05	0.31	0.35	Medium		
	AS	0.00	1.00	-0.26	Weak	0.00	1.00	0.17	Weak		
	SBC	0.00	1.00	0.03	None	0.00	1.00	-0.03	None		
L1S	SBL	0.21	0.65	-0.21	Weak	0.00	1.00	0.03	None		
	AS	0.00	1.00	0.00	None	0.00	1.00	0.41	Medium		

Appendix B Facet Size

Annow div D. O. Convolation hat was an human and facet size divestional		
Appendix B-X Correlation between numeral and tacet size directional	asymmetry	<i>y</i> continued
		CONTINUCA

			Ν	1ale			Fe	emale	
Facet	Sample	Continuity Coefficient	Ρ	Phi	Phi correlation Strength	Continuity Coefficient	Ρ	Phi	Phi correlation Strength
	SBC	0.14	0.71	0.20	Weak	0.01	0.93	-0.09	None
L1I	SBL	0.00	1.00	0.09	None	0.00	0.96	0.13	Weak
	AS	1.18	0.28	-0.73	Strong	0.00	1.00	0.17	Weak
	SBC	3.85	0.05	0.58	Strong	1.61	0.20	0.32	Medium
L2S	SBL	0.03	0.85	0.15	Weak	0.00	1.00	-0.03	None
	AS	1.60	0.21	0.75	Strong	1.70	0.19	-1.00	Strong
	SBC	0.00	1.00	0.67	Strong	0.02	0.90	-0.10	Weak
L2I	SBL	0.00	1.00	0.00	None	0.23	0.64	-0.22	Weak
	AS	0.08	0.79	0.49	Medium	0.00	1.00	-0.17	Weak
	SBC	0.81	0.37	0.33	Medium	0.00	1.00	0.73	Strong
L3S	SBL	0.54	0.46	0.29	Weak	0.00	0.96	-0.13	Weak
	AS	2.96	0.09	-1.00	Strong	0.31	0.58	0.67	Strong
	SBC	0.35	0.55	0.26	Weak	0.22	0.64	-0.16	Weak
L3I	SBL	0.31	0.58	0.24	Weak	0.09	0.76	-0.22	Weak
	AS	0.32	0.57	-0.47	Medium	1.70	0.19	1.00	Strong
	SBC	0.00	1.00	0.06	None	1.95	0.16	-0.34	Medium
L4S	SBL	0.00	1.00	-0.05	None	0.00	1.00	-0.15	Weak
	AS	0.00	1.00	0.15	Weak	0.31	0.58	-0.67	Strong
	SBC	0.00	1.00	-0.06	None	0.21	0.64	-0.16	Weak
L4I	SBL	0.13	0.71	0.22	Weak	0.11	0.74	0.24	Weak
	AS	0.89	0.35	0.60	Strong	0.31	0.58	0.67	Strong
	SBC	0.00	1.00	0.07	None	0.00	1.00	0.01	None
L5S	SBL	1.06	0.30	-0.34	Medium	0.04	0.84	0.17	Weak
	AS	0.00	1.00	0.26	Weak				
	SBC	1.83	0.18	0.43	Medium	3.40	0.07	0.40	Medium
L5I	SBL	0.22	0.64	0.19	Weak	0.00	1.00	0.04	None
	AS	0.43	0.84	-0.24	Weak	0.00	1.00	-0.17	Weak

Facet	Sample	Male						Female				
		Correlation Coefficient	р	N	Strength of Correlation	Correlation Coefficient	р	N	Strength of Correlation			
C1LS	SBC	0.04	0.86	19	None			21				
	SBL	-0.26	0.29	19	Weak			15				
	AS	-0.03	0.94	11	None			6				
C1RS	SBC	-0.01	0.99	18	None	-0.10	0.66	22	Weak			
	SBL	-0.02	0.93	20	None			15				
	AS	0.01	0.97	14	None			6				
C1LI	SBC	-0.22	0.38	19	Weak	-0.36	0.11	21	Medium			
	SBL	-0.31	0.20	19	Medium	0.00	1.00	15	None			
	AS	0.23	0.52	10	Weak			5				
C1RI	SBC			19				22				
	SBL	-0.37	0.11	20	Medium	-0.31	0.26	15	Medium			
	AS	-0.23	0.44	13	Weak			6				
C2LS	SBC	-0.35	0.13	20	Medium	0.16	0.46	23	Weak			
	SBL	-0.29	0.21	21	Weak	0.03	0.91	14	None			
	AS	-0.23	0.49	11	Weak	-0.39	0.26	10	Medium			
C2RS	SBC	-0.30	0.19	21	Medium	-0.14	0.53	24	Weak			
	SBL	-0.16	0.48	21	Weak	-0.45	0.11	14	Medium			
	AS	-0.38	0.20	13	Medium	-0.18	0.63	10	Weak			
C2LI	SBC	0.16	0.52	18	Weak	0.21	0.34	23	Weak			
	SBL	-0.14	0.56	20	Weak	-0.08	0.80	13	None			
	AS	0.16	0.68	9	Weak	0.58	0.13	8	Strong			
C2RI	SBC			17		0.26	0.24	23	Weak			
	SBL	0.19	0.42	21	Weak	-0.31	0.28	14	Medium			
	AS	0.32	0.40	9	Medium	0.58	0.13	8	Strong			
C3LS	SBC	0.36	0.23	13	Medium	0.20	0.37	23	Weak			
	SBL	0.28	0.20	22	Weak	0.25	0.37	15	Weak			
	AS	-0.25	0.37	15	Weak	0.60	0.09	9	Strong			

Facet	Sample			Male		Female				
		Correlation Coefficient	р	Ν	Strength of Correlation	Correlatio n Coefficient	р	Ν	Strength of Correlatio n	
C3RS	SBC	-0.08	0.80	13	None	0.00	1.00	23	None	
	SBL	-0.26	0.25	22	Weak	-0.56	0.04	14	Strong	
	AS	-0.29	0.30	15	Weak	0.53	0.15	9	Strong	
C3LI	SBC	-0.37	0.22	13	Medium	-0.33	0.14	21	Medium	
	SBL	-0.18	0.44	21	Weak	0.25	0.37	15	Weak	
	AS	-0.09	0.75	15	None	0.17	0.67	9	Weak	
C3RI	SBC			13		0.29	0.19	22	Weak	
	SBL	-0.22	0.32	22	Weak	-0.03	0.91	14	None	
	AS	0.14	0.65	13	Weak	-0.55	0.13	9	Strong	
C4LS	SBC	-0.29	0.31	14	Weak	-0.13	0.57	21	Weak	
	SBL			24				17		
	AS	-0.42	0.18	12	Medium	0.00	1.00	6	None	
C4RS	SBC	0.22	0.50	12	Weak	-0.34	0.15	20	Medium	
	SBL	-0.16	0.45	25	Weak	-0.31	0.25	16	Medium	
	AS	-0.18	0.57	13	Weak	0.62	0.19	6	Strong	
C4LI	SBC	0.30	0.32	13	Medium	0.28	0.22	21	Weak	
	SBL	-0.31	0.13	25	Medium	0.20	0.43	17	Weak	
	AS	-0.09	0.78	12	None	0.21	0.69	6	Weak	
C4RI	SBC	0.39	0.21	12	Medium	0.21	0.36	21	Weak	
	SBL	-0.23	0.28	25	Weak	-0.31	0.25	16	Medium	
	AS	0.08	0.82	12	None	0.39	0.44	6	Medium	

Facet	Sample			Male	-			Female	
		Correlatio n Coefficien t	р	Ν	Strength of Correlat ion	Correlatio n Coefficien t	р	N	Strength of Correlation
C5LS	SBC	0.21	0.37	20	Weak	-0.04	0.85	26	None
	SBL	-0.16	0.45	24	Weak	0.07	0.76	20	None
	AS	-0.05	0.90	8	None	-0.08	0.85	8	None
C5RS	SBC	0.24	0.31	20	Weak	-0.28	0.17	26	Weak
	SBL	-0.44	0.03	25	Medium	-0.26	0.27	20	Weak
	AS	-0.19	0.65	8	Weak	0.52	0.19	8	Strong
C5LI	SBC			20		-0.07	0.75	26	None
	SBL	-0.19	0.38	24	Weak			20	
	AS	-0.06	0.90	8	None	0.74	0.04	8	Strong
C5RI	SBC	-0.47	0.04	20	Medium			26	
	SBL	0.21	0.32	24	Weak	-0.38	0.10	20	Medium
	AS	0.18	0.67	8	Weak	0.74	0.04	8	Strong
C6LS	SBC	-0.17	0.50	19	Weak	-0.14	0.51	24	Weak
	SBL	0.08	0.72	25	None	-0.16	0.53	18	Weak
	AS	-0.08	0.84	9	None	0.36	0.43	7	Medium
C6RS	SBC	-0.01	0.96	19	None	-0.33	0.11	25	Medium
	SBL	0.53	0.01	24	Strong	-0.30	0.22	18	Medium
	AS	0.01	0.98	11	None	0.11	0.80	8	Weak
C6LI	SBC	0.05	0.85	19	None			24	
	SBL	0.17	0.40	26	Weak	0.07	0.78	18	None
	AS	-0.08	0.85	9	None	-0.45	0.26	8	Medium
C6RI	SBC			19				25	
	SBL	-0.32	0.13	24	Medium			18	
	AS	0.03	0.94	10	None	0.25	0.56	8	Weak
C7LS	SBC	0.19	0.44	19	Weak	0.09	0.64	28	None
	SBL	0.01	0.95	26	None	-0.13	0.57	21	Weak
	AS	-0.03	0.93	10	None	0.03	0.95	6	None
C7RS	SBC	0.03	0.90	19	None			27	
	SBL	0.09	0.65	27	None	-0.51	0.02	21	Strong
	AS	-0.19	0.51	14	Weak	0.73	0.06	7	Strong

Facet	Sample			Male				Female	
		Correlati on Coefficie nt	р	N	Strength of Correlati on	Correlati on Coefficie nt	р	N	Strength of Correlati on
C7LI	SBC	0.13	0.60	19	Weak	0.05	0.82	28	None
	SBL	-0.23	0.25	28	Weak	-0.22	0.33	21	Weak
	AS	-0.40	0.17	13	Medium	0.66	0.16	6	Strong
C7RI	SBC			19		-0.38	0.05	28	Medium
	SBL	0.20	0.30	28	Weak			21	
	AS	-0.15	0.61	14	Weak	0.20	0.66	7	Weak
T1LS	SBC	0.52	0.03	18	Strong	-0.04	0.84	29	None
	SBL	-0.23	0.23	30	Weak	-0.44	0.06	19	Medium
	AS	-0.31	0.30	13	Medium	-0.11	0.78	9	Weak
T1RS	SBC	0.02	0.93	19	None	-0.07	0.71	30	None
	SBL	0.07	0.73	29	None	-0.33	0.14	21	Medium
	AS	-0.02	0.94	16	None	0.22	0.55	10	Weak
T1LI	SBC			20		-0.13	0.50	31	Weak
	SBL	0.08	0.66	31	None	-0.26	0.26	21	Weak
	AS	-0.12	0.64	17	Weak	-0.14	0.71	10	Weak
T1RI	SBC			20		-0.10	0.59	31	Weak
	SBL			31		-0.15	0.52	21	Weak
	AS	-0.16	0.57	15	Weak	0.28	0.44	10	Weak
T2LS	SBC			23		-0.04	0.83	28	None
	SBL	0.24	0.18	32	Weak			19	
	AS	-0.04	0.86	19	None	-0.20	0.50	14	Weak
T2RS	SBC	-0.21	0.33	23	Weak	-0.03	0.89	29	None
	SBL	-0.15	0.42	33	Weak			20	
	AS	-0.13	0.61	19	Weak	0.00	0.99	14	None
T2LI	SBC	-0.24	0.29	21	Weak	0.32	0.10	29	Medium
	SBL			33				19	
	AS	-0.50	0.03	20	Strong	0.16	0.58	14	Weak
T2RI	SBC	0.13	0.56	23	Weak	0.25	0.19	29	Weak
	SBL	0.03	0.86	31	None	0.34	0.15	20	Medium
	AS	-0.40	0.11	17	Medium	0.26	0.38	14	Weak

Facet Sample Male Female Correlati Correlati Ν Strength Ν Strength of р р Correlation on of on Coefficie Correlatio Coefficie nt n nt T3LS SBC -0.24 0.30 21 Weak 0.15 0.46 26 Weak SBL 0.12 0.59 22 Weak 0.05 0.83 21 None AS 0.07 0.78 19 None -0.21 0.44 16 Weak T3RS 27 SBC -0.59 0.00 22 -0.08 0.68 Strong None SBL 0.26 0.20 25 Weak 0.30 0.19 21 Medium AS 0.14 0.54 22 Weak 0.15 0.56 17 Weak T3LI SBC 22 0.21 0.29 27 Weak SBL 25 0.38 0.10 20 Medium 0.61 18 0.24 0.41 14 Weak AS 0.13 Weak T3RI SBC -0.22 0.33 22 Weak 0.12 0.55 27 Weak SBL -0.13 0.55 25 Weak 0.05 0.84 20 None AS -0.08 0.76 18 -0.23 0.38 16 None Weak T4LS SBC -0.53 0.02 19 Strong -0.04 0.87 25 None SBL 0.16 0.44 26 Weak 0.39 0.10 19 Medium AS 0.16 0.51 18 Weak -0.32 0.22 17 Medium T4RS SBC -0.43 0.07 19 Medium -0.07 0.74 24 None SBL -0.23 0.26 25 Weak -0.20 0.41 19 Weak AS -0.38 0.10 20 Medium 0.10 0.70 17 Weak T4LI SBC 0.39 0.10 19 Medium 0.10 0.63 24 Weak SBL 25 16 AS -0.09 0.73 17 None -0.12 0.68 15 Weak T4RI SBC 17 0.17 0.51 17 Weak SBL 17 0.48 0.11 12 Medium AS -0.38 0.20 13 Medium -0.18 0.58 12 Weak T5LS SBC -0.17 0.48 19 Weak -0.28 25 Weak 0.18 SBL 0.05 0.82 22 None 0.30 0.21 19 Medium AS -0.20 0.10 18 Weak 0.27 0.31 16 Weak

Facet	Sample		М	ale		Female				
		Correlatio n Coefficien t	р	N	Strength of Correlation	Correlatio n Coefficien t	р	N	Strength of Correlati on	
T5RS	SBC	0.02	0.93	19	None	0.13	0.54	25	Weak	
	SBL	0.08	0.72	21	None	-0.24	0.32	19	Weak	
	AS	-0.45	0.05	20	Medium	0.30	0.26	16	Medium	
T5LI	SBC	-0.05	0.85	17	None	-0.11	0.63	22	Weak	
	SBL			20				19		
	AS	-0.41	0.12	16	Medium	-0.01	0.98	15	None	
T5RI	SBC	-0.21	0.40	18	Weak	-0.17	0.44	22	Weak	
	SBL			23		-0.30	0.22	18	Medium	
	AS	-0.62	0.01	17	Strong	-0.25	0.37	15	Weak	
T6LS	SBC	-0.36	0.16	17	Medium	-0.09	0.67	26	None	
	SBL	-0.37	0.07	24	Medium	0.08	0.74	19	None	
	AS	0.09	0.75	15	None	0.08	0.80	14	None	
T6RS	SBC	0.02	0.95	17	None	-0.03	0.89	26	None	
	SBL	0.07	0.73	24	None	-0.15	0.53	20	Weak	
	AS	-0.20	0.45	16	Weak	-0.17	0.55	14	Weak	
T6LI	SBC			19		0.08	0.67	28	None	
	SBL			23				20		
	AS	0.26	0.39	13	Weak	-0.38	0.18	14	Medium	
T6RI	SBC			19				28		
	SBL			22		0.06	0.80	20	None	
	AS	-0.37	0.20	14	Medium	-0.56	0.04	14	Strong	
T7LS	SBC			20		0.06	0.76	27	None	
	SBL	-0.01	0.98	23	None	0.27	0.27	18	Weak	
	AS	-0.15	0.62	14	Weak	-0.11	0.70	16	Weak	
T7RS	SBC			20		-0.24	0.24	26	Weak	
	SBL	-0.29	0.17	24	Weak	-0.05	0.85	17	None	
	AS	-0.29	0.29	15	Weak	0.38	0.15	16	Medium	
T7LI	SBC			19				26		
	SBL			19				18		
	AS	-0.65	0.02	12	Strong	0.46	0.08	16	Medium	

Facet Sample Male Female Correlatio Ν Strength of Correlatio Ν Strength р р Correlation of n n Coefficien Coefficien Correlati t t on T7RI SBC 19 26 SBL 0.64 18 22 0.12 Weak 0.28 AS -0.31 0.31 13 Medium 0.34 14 Weak SBC 0.80 21 0.28 T8LS 0.06 None 0.18 24 Weak SBL 25 0.41 0.10 17 Medium W<u>eak</u> AS 0.23 0.40 15 Weak -0.15 0.57 16 T8RS -0.34 0.11 SBC 21 24 Medium SBL 0.11 0.59 25 Weak 0.46 0.07 17 Medium AS 0.11 0.70 15 Weak -0.30 0.26 16 Medium T8LI SBC -0.04 0.87 21 -0.05 0.80 24 None None 25 SBL -0.16 0.44 Weak 15 0.05 0.85 Weak AS 16 None -0.25 0.35 16 T8RI SBC 22 -0.17 24 0.44 Weak SBL 0.02 0.94 25 None 16 AS 0.14 0.62 16 Weak 0.31 0.26 15 Medium T9LS SBC 0.00 1.00 21 None -0.02 0.92 29 None SBL -0.23 0.26 26 Weak 0.04 0.88 16 None AS -0.23 0.36 18 Weak 0.01 0.96 17 None T9RS SBC -0.38 0.09 21 Medium 0.35 0.06 29 Medium 0.25 SBL -0.14 0.49 26 Weak 0.31 16 Medium -0.20 0.44 -0.01 0.98 AS 18 Weak 17 None T9LI SBC 19 -0.16 0.40 29 Weak SBL 0.50 25 16 -0.14 Weak 0.50 AS -0.18 17 Weak 0.54 15 -0.17 Weak T9RI SBC 19 -0.05 0.81 28 None SBL 23 -0.03 0.92 16 None -0.40 0.23 0.40 AS 0.12 16 Medium 15 Weak T10LS SBC -0.14 0.55 20 Weak -0.08 0.69 28 None SBL -0.04 0.85 26 None 14

AS

-0.28

0.25

18

Weak

-0.52

0.07

Appendix B-9 Correlation between eburnation and facet size continued

13

Strong

Facet	Sample		M	ale		Female				
		Correlatio n Coefficien t	р	N	Strength of Correlation	Correlatio n Coefficien t	р	N	Strength of Correlati on	
T10RS	SBC			20		-0.25	0.21	27	Weak	
	SBL	0.03	0.90	27	None	0.14	0.64	14	Weak	
	AS	-0.18	0.49	17	Weak	0.22	0.48	13	Weak	
T10LI	SBC			20		0.06	0.76	28	None	
	SBL			27		0.30	0.29	14	Medium	
	AS	-0.17	0.50	18	Weak	0.32	0.30	13	Medium	
T10RI	SBC			20		-0.03	0.90	27	None	
	SBL			27		-0.20	0.47	16	Weak	
	AS	-0.32	0.20	18	Medium	0.33	0.27	13	Medium	
T11LS	SBC			20		0.08	0.70	28	None	
	SBL	-0.09	0.69	24	None			14		
	AS	-0.28	0.33	14	Weak	0.52	0.07	13	Strong	
T11RS	SBC	0.29	0.21	21	Weak	0.09	0.65	28	None	
	SBL	0.24	0.27	24	Weak	-0.06	0.83	15	None	
	AS	0.11	0.67	17	Weak	0.56	0.04	14	Strong	
T11LI	SBC	-0.26	0.27	20	Weak	0.00	1.00	27	None	
	SBL			23		0.03	0.91	14	None	
	AS	-0.55	0.02	17	Strong	-0.08	0.80	14	None	
T11RI	SBC			20		-0.08	0.71	27	None	
	SBL			25				15		
	AS	-0.59	0.02	15	Strong	-0.46	0.16	11	Medium	
T12LS	SBC	0.12	0.64	18	Weak	0.17	0.40	26	Weak	
	SBL	0.29	0.18	23	Weak	-0.17	0.56	14	Weak	
	AS	-0.42	0.14	14	Medium	-0.25	0.41	13	Weak	
T12RS	SBC	0.02	0.93	18	None	-0.07	0.75	26	None	
	SBL			23				14		
	AS	-0.40	0.16	14	Medium	0.16	0.61	13	Weak	
T12LI	SBC	0.08	0.74	19	None	0.30	0.13	27	Medium	
	SBL	0.23	0.30	23	Weak			14		
	AS	-0.83	0.00	12	Strong	-0.46	0.18	10	Medium	

Facet	Sample		Ma	ale		Female				
		Correlatio n Coefficien t	р	N	Strength of Correlation	Correlatio n Coefficien t	р	N	Strength of Correlati on	
T12RI	SBC	-0.12	0.64	18	Weak			27		
	SBL	0.19	0.38	23	Weak			14		
	AS	-0.57	0.04	13	Strong	0.18	0.59	12	Weak	
L1LS	SBC	-0.24	0.32	19	Weak			25		
	SBL			23		-0.31	0.31	13	Medium	
	AS	-0.35	0.22	14	Medium	0.19	0.51	14	Weak	
L1RS	SBC	0.10	0.70	19	Weak			23		
	SBL	-0.32	0.13	24	Medium			12		
	AS	0.02	0.95	13	None	0.12	0.69	13	Weak	
L1LI	SBC	0.00	1.00	19	None			24		
	SBL	-0.17	0.42	25	Weak			13		
	AS	-0.50	0.08	13	Strong	-0.22	0.41	16	Weak	
L1RI	SBC	0.00	0.99	19	None			23		
	SBL	0.03	0.89	25	None			12		
	AS	-0.37	0.19	14	Medium	-0.34	0.21	15	Medium	
L2LS	SBC	0.11	0.63	21	Weak	0.31	0.13	25	Medium	
	SBL			22				13		
	AS	0.06	0.83	17	None	-0.76	0.00	12	Strong	
L2RS	SBC			21				24		
	SBL			23				11		
	AS	-0.20	0.46	16	Weak	-0.32	0.31	12	Medium	
L2LI	SBC			21				24		
	SBL			24				12		
	AS	-0.61	0.02	14	Strong	-0.17	0.59	13	Weak	
L2RI	SBC			20				24		
	SBL	-0.11	0.62	24	Weak			12		
	AS	0.01	0.97	14	None	-0.48	0.10	13	Medium	
L3LS	SBC	-0.20	0.40	20	Weak			28		
	SBL	-0.06	0.80	20	None			13		
	AS	-0.63	0.01	17	Strong	-0.27	0.27	18	Weak	

Facet	Sample		М	ale		Female			
		Correlatio n Coefficien t	р	N	Strength of Correlation	Correlatio n Coefficien t	р	N	Strength of Correlati on
L3RS	SBC	0.00	1.00	21	None			25	
	SBL	-0.07	0.78	18	None			13	
	AS	-0.45	0.10	15	Medium	0.18	0.55	13	Weak
L3LI	SBC	0.21	0.35	21	Weak	0.10	0.62	29	None
	SBL	-0.14	0.56	20	Weak			15	
	AS	-0.17	0.50	18	Weak	0.00	1.00	16	None
L3RI	SBC	-0.35	0.12	21	Medium			28	
	SBL			20		-0.06	0.83	15	None
	AS	-0.06	0.83	15	None	-0.10	0.72	15	Weak
L4LS	SBC			20				25	
	SBL			19				10	
	AS	-0.06	0.82	17	None	-0.06	0.81	16	None
L4RS	SBC	0.08	0.73	20	None	0.09	0.69	25	None
	SBL			20		0.00	1.00	9	None
	AS	-0.12	0.65	18	Weak	-0.35	0.18	16	Medium
L4LI	SBC	-0.30	0.20	20	Medium			25	
	SBL			18		0.06	0.87	10	None
	AS	0.14	0.61	16	Weak	-0.47	0.10	13	Medium
L4RI	SBC	0.30	0.20	20	Medium	-0.31	0.13	26	Medium
	SBL			20				10	
	AS	0.39	0.15	15	Medium	-0.01	0.99	13	None
L5LS	SBC	-0.13	0.59	19	Weak	0.10	0.68	20	Weak
	SBL	-0.15	0.47	24	Weak	-0.51	0.08	13	Strong
	AS	0.04	0.89	14	None	-0.45	0.12	13	Medium
L5RS	SBC	-0.17	0.48	19	Weak			20	
	SBL	0.36	0.10	22	Medium	-0.08	0.80	13	None
	AS	0.04	0.90	13	None	-0.20	0.51	13	Weak
L5LI	SBC	0.10	0.68	19	Weak			18	
	SBL	-0.35	0.10	24	Medium	-0.08	0.80	13	None
	AS	0.10	0.76	11	Weak	-0.23	0.88	13	Weak

Facet Sample Male Female Correlatio Ν Strength of Correlatio Ν Strength р р Correlation of n n Coefficien Coefficien Correlati t t on L5RI -0.09 SBC 0.73 19 None 19 SBL 0.23 0.30 23 Weak 13 AS 0.06 0.84 14 None -0.15 0.64 13 Weak

			Ma	ale			Ferr	nale	
Facet	Sample	Correlati on			Strength of	Correlati on			Strength of
		Coefficie nt	р	N	Correlati on	Coefficie nt	р	N	Correlati on
	SBC	-0.15	0.57	18	Weak	-0.46	0.03	23	Medium
C1LS	SBL	-0.03	0.92	19	None	0.31	0.28	14	Medium
	AS	0.15	0.65	12	Weak	0	1	7	None
	SBC	0.26	0.31	18	Weak	-0.44	0.04	23	Medium
C1RS	SBL	-0.17	0.47	20	Weak			14	
	AS	0.26	0.37	14	Weak	0.36	0.43	7	Medium
	SBC	-0.34	0.17	18	Medium	-0.28	0.2	23	Weak
C1LI	SBL	-0.23	0.35	19	Weak	0.03	0.91	14	None
	AS	0.01	0.97	10	None	-0.54	0.27	6	Strong
	SBC	-0.06	0.8	18	None	-0.13	0.57	23	Weak
C1RI	SBL	-0.18	0.45	20	Weak	-0.1	0.73	14	Weak
	AS	-0.25	0.42	13	Weak	0.61	0.14	7	Strong
	SBC	0.12	0.62	20	Weak	-0.01	0.97	25	None
C2LS	SBL	-0.28	0.24	20	Weak	0	1	13	None
	AS	0.01	0.99	11	None	0.66	0.03	11	Strong
	SBC	0.38	0.1	20	Medium	-0.21	0.3	25	Weak
C2RS	SBL	0.01	0.98	21	None	-0.04	0.89	13	None
	AS	-0.11	0.72	13	Weak	0.55	0.1	10	Strong
	SBC	-0.4	0.1	18	Medium	-0.01	0.96	25	None
C2LI	SBL	0.3	0.21	19	Medium	0.1	0.76	13	None
	AS	0.11	0.79	9	Weak	0.4	0.28	9	Medium
	SBC	-0.15	0.56	17	Weak	0.03	0.9	25	None
C2RI	SBL	0.26	0.26	21	Weak	-0.39	0.19	13	Medium
	AS	0.03	0.95	9	None	0.69	0.04	9	Strong
	SBC	0.11	0.72	13	Weak	-0.26	0.25	22	Weak
C3LS	SBL	-0.06	0.81	21	None	0.04	0.89	17	None
	AS	-0.07	0.82	15	None	0.09	0.84	8	None
	SBC	-0.37	0.22	13	Medium	0.18	0.42	22	Weak
C3RS	SBL	0	0.99	21	None	-0.27	0.32	16	Weak
	AS	-0.21	0.46	15	Weak	0.51	0.19	8	Strong
	SBC	-0.32	0.29	13	Medium	0.13	0.59	20	Weak
C3LI	SBL	0.09	0.69	20	None	-0.01	0.96	17	None
	AS	0.2	0.48	15	Weak	0.37	0.37	8	Medium
	SBC	-0.09	0.77	13	None	-0.12	0.61	20	Weak
C3RI	SBL	0.15	0.52	21	Weak	-0.12	0.67	16	Weak
	AS	0.39	0.19	13	Medium	-0.03	0.95	8	None

			М	ale			Fen	nale	
Facet	Sample	Correlation Coefficient	р	N	Strength of Correlation	Correlation Coefficient	р	N	Strength of Correlation
	SBC	-0.08	0.79	13	None	0.22	0.36	20	Weak
C4LS	SBL	-0.04	0.85	23	None	0.02	0.94	16	None
	AS	-0.18	0.58	12	Weak	0.52	0.3	6	Strong
	SBC	-0.02	0.96	11	None	0.2	0.38	21	Weak
C4RS	SBL	-0.08	0.72	24	None	0.26	0.36	15	Weak
	AS	0.39	0.19	13	Medium	0.62	0.19	6	Strong
	SBC	-0.15	0.63	13	Weak	0.18	0.42	23	Weak
C4LI	SBL	-0.44	0.03	24	Medium	0.04	0.89	16	None
	AS	0.26	0.42	12	Weak	0.35	0.49	6	Medium
	SBC	-0.03	0.92	12	None	0.27	0.21	23	Weak
C4RI	SBL	-0.02	0.93	24	None	-0.31	0.26	15	Medium
	AS	0.52	0.08	12	Strong	0.1	0.85	6	Weak
	SBC	0.24	0.3	20	Weak	-0.26	0.18	27	Weak
C5LS	SBL	-0.26	0.22	24	Weak	-0.09	0.72	20	None
	AS	-0.24	0.56	8	Weak	0.26	0.53	8	Weak
	SBC	-0.22	0.35	20	Weak	-0.1	0.64	27	None
C5RS	SBL	0.11	0.59	25	Weak	0.03	0.91	17	None
	AS	-0.56	0.15	8	Strong	0.85	0.01	8	Strong
	SBC	-0.3	0.2	20	Medium	0.38	0.05	27	Medium
C5LI	SBL	-0.13	0.54	24	Weak	0.1	0.68	20	Weak
	AS	-0.29	0.48	8	Weak	0.58	0.13	8	Strong
	SBC	-0.2	0.4	20	Weak	0.09	0.66	27	None
C5RI	SBL	0.22	0.29	24	Weak	0.06	0.8	20	None
	AS	-0.41	0.31	8	Medium	0.58	0.13	8	Strong
	SBC	-0.13	0.6	18	Weak	-0.15	0.49	25	Weak
C6LS	SBL	0.12	0.58	23	Weak	0.11	0.66	19	Weak
	AS	-0.07	0.86	9	None	0.51	0.24	7	Strong
	SBC	0.05	0.84	18	None	0	1	26	None
C6RS	SBL	0.52	0.01	22	Strong	0.12	0.63	19	Weak
	AS	-0.38	0.24	11	Medium	0.29	0.49	8	Weak
	SBC	0.15	0.56	18	Weak	0.13	0.53	25	Weak
C6LI	SBL	-0.27	0.19	26	Weak	-0.25	0.31	19	Weak
	AS	-0.3	0.44	9	Weak	-0.22	0.61	8	Weak
	SBC	0.26	0.3	18	Weak	0.19	0.35	25	Weak
C6RI	SBL	-0.16	0.47	22	Weak	-0.28	0.24	19	Weak
	AS	-0.36	0.31	10	Medium	0	1	8	None
	SBC	0.14	0.59	18	Weak	-0.2	0.32	28	Weak
C7LS	SBL	-0.31	0.12	26	Medium	-0.21	0.37	20	Weak
	AS	-0.3	0.38	11	Weak	-0.06	0.91	6	None

			M	ale			Fen	nale	
Facet	Sample	Correlati on Coefficie nt	р	N	Strength of Correlati on	Correlati on Coefficie nt	р	N	Strength of Correlati on
	SBC	-0.15	0.56	18	Weak	0.22	0.26	28	Weak
C7RS	SBL	0.45	0.03	25	Medium	-0.41	0.06	22	Medium
	AS	-0.07	0.83	12	None	0.22	0.63	7	Weak
	SBC	-0.28	0.26	18	Weak	-0.03	0.89	29	None
C7LI	SBL	-0.1	0.62	27	Weak	-0.35	0.12	22	Medium
	AS	0.02	0.95	12	None	0.21	0.69	6	Weak
	SBC	0.24	0.34	18	Weak	0.08	0.68	29	None
C7RI	SBL	-0.02	0.91	27	None	0.22	0.32	22	Weak
	AS	0.17	0.58	13	Weak	-0.05	0.92	7	None
	SBC	-0.09	0.76	16	None	-0.44	0.01	31	Medium
T1LS	SBL	0.15	0.45	29	Weak	-0.3	0.2	20	Weak
	AS	-0.27	0.4	12	Weak	0.36	0.31	10	Medium
	SBC	0.04	0.88	18	None	0.04	0.84	32	None
T1RS	SBL	0.05	0.79	29	None	-0.3	0.18	21	Medium
	AS	-0.25	0.38	14	Weak	0.31	0.32	12	Medium
	SBC	-0.31	0.2	19	Medium	0.31	0.08	32	Medium
T1LI	SBL	-0.06	0.75	30	None	0.02	0.92	22	None
	AS	0.15	0.28	14	Weak	0.02	0.94	12	None
	SBC	-0.29	0.22	19	Weak	0.07	0.71	32	None
T1RI	SBL	0.24	0.21	30	Weak	-0.1	0.66	21	Weak
	AS	-0.16	0.58	14	Weak	0.29	0.37	12	Weak
	SBC	0.04	0.83	29	None	-0.44	0.07	18	Medium
T2LS	SBL	0.04	0.83	29	None	-0.44	0.07	18	Medium
	AS	-0.28	0.27	17	Weak	0.05	0.85	16	None
	SBC	-0.1	0.58	34	None	0.41	0.08	19	Medium
T2RS	SBL	-0.1	0.58	34	None	0.41	0.08	19	Medium
	AS	-0.53	0.02	19	Strong	-0.05	0.85	16	None
	SBC	0.26	0.25	21	Weak	0.4	0.02	31	Medium
T2LI	SBL	-0.03	0.85	33	None	0.11	0.65	19	Weak
	AS	-0.23	0.32	20	Weak	-0.04	0.9	14	None
	SBC	-0.16	0.5	21	Weak	-0.15	0.44	30	Weak
T2RI	SBL	0.02	0.92	32	None	-0.15	0.52	20	Weak
	AS	-0.08	0.75	19	None	0.17	0.55	14	Weak
	SBC	-0.06	0.82	20	None	0.44	0.02	28	Medium
T3LS	SBL	0.07	0.73	24	None	0.06	0.8	20	None
	AS	0.03	0.92	18	None	-0.2	0.48	15	Weak

			М	ale		Female			
		Correlati			Strength	Correlati			Strength
Facet	Sample	on	р	N	of	on	р	N	of
		Coefficie nt			on	Coefficie nt			on
	SBC	0.19	0.41	20	Weak	-0.06	0.77	29	None
T3RS	SBL	0.15	0.46	26	Weak	0.28	0.24	20	Weak
	AS	0.11	0.63	21	Weak	0.1	0.72	16	None
	SBC	0.59	0.01	20	Strong	0.35	0.06	29	Medium
T3LI	SBL	-0.14	0.5	24	Weak	0.1	0.67	21	Weak
	AS	-0.18	0.46	20	Weak	0.01	0.98	13	None
	SBC	-0.38	0.1	20	Medium	0.03	0.86	29	None
T3RI	SBL	-0.18	0.38	25	Weak	0.17	0.47	21	Weak
	AS	-0.13	0.59	20	Weak	-0.03	0.92	15	None
	SBC	-0.31	0.25	16	Medium	0.04	0.82	28	None
T4LS	SBL	-0.01	0.95	26	None	0.05	0.85	18	None
	AS	-0.39	0.1	19	Medium	-0.17	0.53	17	Weak
	SBC	0	1	16	None	-0.05	0.82	27	None
T4RS	SBL	0.1	0.65	24	None	0.13	0.61	18	Weak
	AS	0.02	0.93	18	None	0.04	0.89	17	None
	SBC	-0.2	0.45	16	Weak	0.13	0.51	27	Weak
T4LI	SBL	0.03	0.88	24	None	0.49	0.05	16	Strong
	AS	0.1	0.71	17	None	0.17	0.55	15	Weak
	SBC	-0.34	0.24	14	Medium	-0.46	0.05	19	Medium
T4RI	SBL	0.5	0.05	16	Strong	0.45	0.14	12	Medium
	AS	-0.32	0.24	15	Medium	-0.56	0.07	11	Strong
	SBC	-0.23	0.39	16	Weak	-0.11	0.58	28	Weak
T5LS	SBL	0.29	0.19	22	Weak	-0.05	0.83	19	None
	AS	-0.12	0.61	19	Weak	-0.07	0.81	16	None
	SBC	-0.35	0.18	16	Medium	-0.22	0.29	25	Weak
T5RS	SBL	0.4	0.07	21	Medium	-0.12	0.63	19	Weak
	AS	0.19	0.41	19	Weak	-0.13	0.63	17	Weak
	SBC	0.46	0.08	15	Medium	-0.23	0.26	24	Weak
T5LI	SBL	-0.04	0.86	20	None	0.3	0.19	20	Medium
	AS	0.21	0.46	15	Weak	0.52	0.07	13	Strong
	SBC	-0.25	0.35	16	Weak	-0.26	0.22	24	Weak
T5RI	SBL	0.01	0.98	23	None	-0.07	0.78	20	None
	AS	-0.47	0.07	16	Medium	-0.15	0.63	13	Weak
	SBC	-0.36	0.14	18	Medium	0.13	0.53	26	Weak
T6LS	SBL	-0.1	0.63	25	Weak	-0.18	0.47	19	Weak
	AS	-0.14	0.62	15	Weak	-0.2	0.54	12	Weak
	SBC	-0.32	0.2	18	Medium	-0.19	0.36	26	Weak
T6RS	SBL	0.09	0.67	25	None	-0.14	0.57	19	Weak
	AS	-0.06	0.83	16	None	-0.33	0.29	12	Medium

			M	ale			Fen	nale	
Facat	Sampla	Correlati			Strength	Correlati			Strength
Facet	Sample	on Coefficie	р	N	of Correlati	on Coefficie	р	N	of Correlati
	SBC	-0.11	0.66	19	Weak	-0.28	0.16	27	Weak
T6LI	SBL	0.05	0.84	22	None	0.18	0.44	20	Weak
	AS	0.37	0.18	15	Medium	-0.04	0.89	12	None
	SBC	-0.08	0.74	19	None	-0.34	0.08	28	Medium
T6RI	SBL	-0.11	0.61	23	Weak	-0.2	0.43	17	Weak
	AS	0	1	15	None	-0.76	0	12	Strong
	SBC	0.19	0.43	19	Weak	0.28	0.14	29	Weak
T7LS	SBL	0.04	0.85	25	None	0.25	0.33	18	Weak
	AS	-0.46	0.1	14	Medium	0.06	0.83	16	None
	SBC	0.36	0.13	19	Medium	0.04	0.84	27	None
T7RS	SBL	-0.12	0.59	24	Weak	-0.17	0.52	16	Weak
	AS	-0.32	0.25	15	Medium	-0.02	0.93	16	None
	SBC	0.22	0.38	18	Weak	0.06	0.78	28	None
T7LI	SBL	-0.04	0.87	22	None	-0.2	0.43	17	Weak
	AS	-0.54	0.07	12	Strong	0.5	0.05	16	Strong
	SBC	0.17	0.49	18	Weak	-0.13	0.53	28	Weak
T7RI	SBL	-0.11	0.61	23	Weak	-0.2	0.43	17	Weak
	AS	-0.48	0.1	13	Medium	0.49	0.07	14	Medium
	SBC	0.47	0.04	19	Medium	0.3	0.12	28	Medium
T8LS	SBL	0.04	0.85	25	None	0.25	0.33	18	Weak
	AS	-0.27	0.35	14	Weak	0.09	0.75	16	None
	SBC	0.41	0.08	19	Medium	-0.05	0.79	28	None
T8RS	SBL	-0.16	0.45	25	Weak	0.16	0.52	18	Weak
	AS	-0.29	0.3	15	Weak	-0.13	0.63	16	Weak
	SBC	0.28	0.25	19	Weak	-0.28	0.15	28	Weak
T8LI	SBL	-0.12	0.56	25	Weak			16	
	AS	-0.57	0.02	16	Strong	-0.05	0.85	16	None
	SBC	0.04	0.87	20	None	-0.02	0.92	28	None
T8RI	SBL	0.15	0.49	25	Weak	0.36	0.16	17	Medium
	AS	-0.3	0.26	16	Weak	-0.21	0.46	15	Weak
	SBC	-0.34	0.16	19	Medium	0	0.99	30	None
T9LS	SBL	0.29	0.16	26	Weak	-0.09	0.74	17	None
	AS	0.28	0.27	17	Weak	0.12	0.64	18	Weak
	SBC	-0.23	0.35	19	Weak	-0.2	0.28	30	Weak
T9RS	SBL	0.55	0	26	Strong			17	
	AS	0.06	0.82	16	None	0.13	0.62	18	Weak

			M	ale			Fen	nale	
Facet	Sample	Correlati			Strength	Correlati			Strength
Tacci	Sample	on Coefficie nt	р	N	of Correlati on	on Coefficie nt	р	N	of Correlati on
	SBC	-0.17	0.51	18	Weak	-0.23	0.24	29	Weak
T9LI	SBL	-0.08	0.72	25	None			17	
	AS	0.28	0.32	15	Weak	0.15	0.57	16	Weak
	SBC	-0.03	0.89	18	None	-0.17	0.4	28	Weak
T9RI	SBL	-0.22	0.32	22	Weak			17	
	AS	-0.08	0.79	14	None	0.28	0.29	16	Weak
	SBC	0.03	0.91	18	None	0.01	0.97	28	None
T10LS	SBL	0.1	0.65	25	None	0.36	0.17	16	Medium
	AS	-0.34	0.22	15	Medium	-0.05	0.88	14	None
	SBC	0.15	0.57	18	Weak	0.11	0.56	29	Weak
T10RS	SBL	0.26	0.21	26	Weak			16	
	AS	0.08	0.76	16	None	-0.32	0.27	14	Medium
	SBC	0.34	0.17	18	Medium	0.17	0.37	29	Weak
T10LI	SBL	0.07	0.74	26	None			16	
	AS	-0.12	0.68	14	Weak	-0.44	0.11	14	Medium
	SBC	-0.05	0.84	18	None	0.16	0.4	29	Weak
T10RI	SBL	-0.28	0.17	26	Weak			15	
	AS	-0.05	0.86	15	None	0	0.99	14	None
	SBC	0.1	0.69	19	None	0.01	0.96	28	None
T11LS	SBL	-0.04	0.85	23	None	0.15	0.13	16	Weak
	AS	-0.19	0.55	13	Weak	0.01	0.99	14	None
	SBC	-0.12	0.61	19	Weak	-0.06	0.78	29	None
T11RS	SBL	-0.01	0.96	24	None	-0.29	0.28	16	Weak
	AS	0.22	0.4	17	Weak	0.2	0.48	15	Weak
	SBC	0.37	0.13	18	Medium	0.2	0.32	28	Weak
T11LI	SBL	0	0.99	23	None	0.12	0.66	15	Weak
	AS	0.24	0.37	16	Weak	-0.16	0.57	15	Weak
	SBC	0.44	0.07	18	Medium	0.33	0.08	28	Medium
T11RI	SBL	0.01	0.98	24	None	0.42	0.11	16	Medium
	AS	-0.17	0.6	12	Weak	-0.29	0.36	12	Weak
	SBC	0.25	0.33	17	Weak	-0.1	0.64	25	None
T12LS	SBL	-0.06	0.8	22	None	-0.39	0.14	16	Medium
	AS	-0.15	0.53	12	Weak	-0.31	0.31	13	Medium
	SBC	0.21	0.42	17	Weak	-0.05	0.83	25	None
T12RS	SBL	-0.04	0.86	22	None	0.2	0.47	16	Weak
	AS	-0.2	0.51	13	Weak	-0.24	0.44	13	Weak
	SBC	0.25	0.32	18	Weak	0.2	0.32	26	Weak
T12LI	SBL	0.16	0.49	22	Weak	0.23	16	-0.37	Weak
	AS	-0.48	0.1	13	Medium	-0.58	0.05	12	Strong

			M	ale			Fen	nale	
Eacot	Samplo	Correlati			Strength	Correlati			Strength
Facet	Sample	on Coefficie	р	N	of Correlati	on Coefficie	р	N	of Correlati
		nt			on	nt			on
	SBC	-0.14	0.6	17	Weak	0.23	0.25	26	Weak
T12RI	SBL	0.32	0.15	22	Medium	-0.04	0.88	16	None
	AS	-0.38	0.29	10	Medium	-0.02	0.96	12	None
	SBC	0.02	0.95	19	None	-0.2	0.33	25	Weak
L1LS	SBL	0.4	0.07	22	Medium	-0.31	0.31	13	Medium
	AS	-0.23	0.41	15	Weak	-0.07	0.81	15	None
	SBC	-0.23	0.35	19	Weak	0.03	0.9	23	None
L1RS	SBL	-0.13	0.56	24	Weak	0.05	0.88	12	None
	AS	-0.01	0.97	13	None	0.12	0.7	14	Weak
	SBC	0.34	0.16	19	Medium	0.03	0.87	24	None
L1LI	SBL	0.09	0.69	25	None	0.08	0.8	13	None
	AS	-0.4	0.16	13	Medium	-0.23	0.4	16	Weak
	SBC	-0.28	0.25	19	Weak	0.02	0.94	24	None
L1RI	SBL	-0.09	0.68	24	None	-0.4	0.2	12	Medium
	AS	-0.42	0.12	15	Medium	-0.55	0.03	15	Strong
	SBC	0.2	0.38	21	Weak	0.04	0.84	26	None
L2LS	SBL	0.36	0.1	22	Medium	0.31	0.3	13	Medium
	AS	-0.16	0.53	18	Weak	-0.4	0.17	13	Medium
	SBC	0.22	0.35	21	Weak	-0.04	0.83	25	None
L2RS	SBL	0.09	0.7	23	None	0.1	0.77	11	Weak
	AS	-0.45	0.08	16	Medium	0.18	0.55	13	Weak
	SBC	0.16	0.48	21	Weak	-0.13	0.54	25	Weak
L2LI	SBL	0.44	0.03	24	Medium	0.56	0.06	12	Strong
	AS	-0.31	0.28	14	Medium	0.05	0.87	13	None
	SBC	0	0.99	20	None	-0.13	0.54	25	Weak
L2RI	SBL	-0.04	0.86	24	None	0.38	0.22	12	Medium
	AS	-0.44	0.12	14	Medium	-0.11	0.71	14	Weak
	SBC	0.11	0.64	20	Weak	0.07	0.72	27	None
L3LS	SBL	0.2	0.4	20	Weak	-0.4	0.18	13	Medium
	AS	-0.51	0.04	17	Strong	-0.19	0.48	16	Weak
	SBC	-0.13	0.57	21	Weak	0	0.99	25	None
L3RS	SBL	-0.05	0.86	18	None	-0.29	0.35	13	Weak
	AS	-0.07	0.8	15	None	-0.13	0.68	12	Weak
	SBC	0.23	0.32	20	Weak	0.28	0.14	29	Weak
L3LI	SBL	-0.14	0.56	20	Weak	0.3	0.27	15	Medium
	AS	-0.03	0.92	18	None	-0.28	0.32	15	Medium

			М	ale			Female			
Facet	Sample	Correlati on Coefficie nt	р	N	Strength of Correlati on	Correlati on Coefficie nt	р	N	Strength of Correlati on	
	SBC	-0.07	0.77	20	None	0.24	0.22	28	Weak	
L3RI	SBL			20		-0.26	0.37	14	Weak	
	AS	-0.03	0.92	15	None	-0.22	0.46	14	Medium	
	SBC	-0.12	0.61	20	Weak	0.06	0.79	25	None	
L4LS	SBL	-0.07	0.78	19	None			10		
	AS	-0.06	0.83	17	None	-0.25	0.35	16	Medium	
	SBC	-0.11	0.65	20	Weak	0.06	0.77	25	None	
L4RS	SBL	0.35	0.13	20	Medium	0.14	0.73	9	Weak	
	AS	-0.06	0.83	17	None	-0.25	0.35	16	Medium	
	SBC	-0.05	0.84	20	None	-0.25	0.18	30	Weak	
L4LI	SBL	0.07	0.78	18	None	0.61	0.06	10	Strong	
	AS	0.23	0.39	16	Weak	-0.67	0.01	13	Strong	
	SBC	0.57	0.01	19	Strong	-0.32	0.11	26	Medium	
L4RI	SBL	0.03	0.89	20	None	-0.52	0.12	10	Strong	
	AS	0.38	0.16	15	Medium	0.07	0.83	13	None	
	SBC	0.23	0.36	19	Weak	0.37	0.1	21	Medium	
L5LS	SBL	0.02	0.93	24	None	-0.54	0.06	13	Strong	
	AS	0.04	0.9	14	None	-0.15	0.62	13	Weak	
	SBC	0.12	0.63	19	Weak	0.09	0.69	21	None	
L5RS	SBL	-0.19	0.42	21	Weak	-0.17	0.58	13	Weak	
	AS	-0.15	0.64	13	Weak	0.1	0.75	13	None	
	SBC	0.16	0.53	19	Weak	-0.02	0.93	21	None	
L5LI	SBL	-0.04	0.84	24	None	-0.17	0.59	13	Weak	
	AS	0.22	0.49	12	Weak	0.62	0.03	12	Strong	
	SBC	0.14	0.57	19	Weak	0.06	0.81	21	None	
L5RI	SBL	0.14	0.54	22	Weak	-0.39	0.19	13	Medium	
	AS	0.32	0.27	14	Medium	-0.19	0.55	13	Weak	

Appendix B-11 Correlation between osteophyte score and facet size

			Ma	ale		Female			
Facet	Sample	Correlati on Coefficie nt	р	N	Strength of Correlati on	Correlati on Coefficie nt	р	N	Strength of Correlati on
	SBC			18		0	1	23	None
C1LS	SBL	-0.03	0.92	19	None	0.31	0.28	14	Medium
	AS			12				6	
	SBC			18		-0.03	0.88	23	None
C1RS	SBL	-0.17	0.47	20	Weak			14	
	AS			14				6	
	SBC	0.12	0.64	18	Weak			23	
C1LI	SBL	-0.23	0.35	19	Weak	0.03	0.91	14	None
	AS			10				5	
	SBC			18				23	
C1RI	SBL	-0.18	0.45	20	Weak	-0.1	0.73	14	Weak
	AS	-0.46	0.11	13	Medium			6	
	SBC	0.26	0.27	20	Weak			25	
C2LS	SBL	-0.28	0.24	20	Weak	0	1	13	None
	AS	-0.5	0.12	11	Strong			10	
	SBC			20		-0.26	0.22	25	Weak
C2RS	SBL	0.01	0.98	21	None	-0.04	0.89	13	None
	AS			13				9	
	SBC	0.1	0.69	18	Weak	-0.03	0.89	25	None
C2LI	SBL	0.3	0.21	19	Medium	0.1	0.76	13	None
	AS			9				8	
	SBC	-0.2	0.43	17	Weak	0.01	0.95	25	None
C2RI	SBL	0.26	0.26	21	Weak	-0.39	0.19	13	Medium
	AS			9		0.25	0.56	8	Weak
	SBC	0	1	13	None	0.15	0.52	22	Weak
C3LS	SBL	-0.06	0.81	21	None	0.04	0.89	17	None
	AS			15		-0.25	0.55	8	Weak
	SBC	-0.48	0.1	13	Medium	0.32	0.15	22	Medium
C3RS	SBL	0	0.99	21	None	-0.27	0.32	16	Weak
	AS	-0.19	0.51	15	Weak	0.74	0.04	8	Strong
	SBC	-0.39	0.19	13	Medium	0.05	0.85	20	None
C3LI	SBL	0.09	0.69	20	None	-0.01	0.96	17	None
	AS	-0.31	0.26	15	Medium	-0.06	0.88	8	None
	SBC	0.1	0.76	13	Weak	0.08	0.75	20	None
C3RI	SBL	0.15	0.52	21	Weak	-0.12	0.67	16	Weak
	AS			13		0.33	0.42	8	Medium

Male Female Facet Sample Correlation Strength of Correlation Strength of р Ν р Ν Coefficient Correlation Coefficient Correlation SBC -0.24 0.44 13 Weak 0.13 0.59 21 Weak C4LS SBL -0.04 0.85 23 None 0.02 0.94 16 None AS -0.15 0.64 12 Weak 6 0.52 21 SBC -0.08 0.02 Strong 0.81 11 None C4RS -0.08 0.36 15 SBL 0.72 24 None 0.26 Weak AS 0.46 0.12 13 Medium 0.49 0.33 6 Medium SBC 0.19 0.53 13 Weak -0.06 0.78 23 None C4LI SBL -0.44 0.03 24 Medium 0.04 0.89 16 None AS 0.22 0.5 12 Weak 0.28 0.59 6 Weak SBC -0.01 0.97 12 None -0.18 0.42 23 Weak 0.26 C4RI -0.02 0.93 24 15 SBL -0.31 Medium None AS 0.13 0.69 12 Weak 0.37 0.47 6 Medium SBC -0.16 0.51 20 0.1 0.63 27 Weak Weak C5LS -0.26 24 Weak 20 SBL 0.22 -0.09 0.72 None 0.48 0.31 AS 0.23 8 Medium -0.42 8 Medium SBC -0.06 0.8 20 0.47 27 None 0.15 Weak C5RS SBL 0.11 0.59 25 Weak 0.03 0.91 17 None 0.31 AS 0.41 8 Medium 0.25 0.55 8 Weak SBC -0.51 0.02 20 Strong 0.18 0.37 27 Weak -0.13 0.54 24 0.1 0.68 20 C5LI SBL Weak Weak -0.41 8 Medium AS 0.31 8 -0.25 20 Weak SBC 0.3 0.28 0.16 27 Weak C5RI SBL 0.22 0.29 24 Weak 0.06 0.8 20 None -0.25 AS 0.56 8 Weak -0.04 0.93 8 None SBC 0.03 0.92 18 None 0.25 0.24 25 Weak C6LS SBL 0.12 0.58 23 Weak 0.11 0.66 19 Weak 9 7 AS 0.14 0.73 Weak 0.27 SBC 0.27 18 Weak -0.18 0.37 26 Weak SBL 0.52 0.01 22 0.63 19 C6RS 0.12 Weak Strong 0.05 0.91 AS 0.1 0.77 11 Weak 8 None SBC 0.04 0.88 18 None 0.11 0.61 25 Weak C6LI SBL -0.27 0.19 26 -0.25 0.31 19 Weak Weak AS 0.27 0.48 9 Weak -0.25 0.56 8 Weak SBC -0.25 0.32 18 0.64 26 Weak -0.1 Weak C6RI SBL -0.16 0.47 22 Weak -0.28 0.24 19 Weak AS 10 -0.25 0.56 8 Weak SBC -0.17 0.5 18 Weak 0.08 0.66 29 None SBL -0.31 26 0.37 20 C7LS 0.12 Medium -0.21 Weak

AS

-0.1

0.77

11

Weak

Appendix B-11 Correlation between osteophyte score and facet size continued

Medium

0.41

-0.41

6

Appendix B-11 Correlation) between osteophyte score	e and facet size continued
Appendix D II conclution	between obteophyte score	

Male Female				nale					
Facet	Sample	Correlati on Coefficie nt	р	N	Strength of Correlati on	Correlati on Coefficie nt	р	N	Strength of Correlati on
	SBC	-0.25	0.32	18	Weak	0.02	0.94	28	None
C7RS	SBL	0.45	0.03	25	Medium	-0.41	0.06	22	Medium
	AS			13		0.2	0.66	7	Weak
	SBC	-0.06	0.81	18	None	0	0.99	29	None
C7LI	SBL	-0.1	0.62	27	Weak	-0.35	0.12	22	Medium
	AS	0.08	0.79	13	None	0.66	0.16	6	Strong
	SBC	0.19	0.46	18	Weak	-0.23	0.24	29	Weak
C7RI	SBL	-0.02	0.91	27	None	0.22	0.32	22	Weak
	AS	-0.15	0.62	13	Weak	0.41	0.36	7	Medium
	SBC	-0.09	0.74	18	None	-0.02	0.93	29	None
T1LS	SBL	0.15	0.45	29	Weak	-0.3	0.2	20	Weak
	AS	0	1	13	None	0.44	0.24	9	Medium
	SBC	-0.31	0.2	19	Medium	-0.24	0.2	31	Weak
T1RS	SBL	0.05	0.79	29	None	-0.3	0.18	21	Medium
	AS	0.09	0.75	16	None	0.36	0.31	10	Medium
	SBC			20		0.06	0.75	31	None
T1LI	SBL	-0.06	0.75	30	None	0.02	0.92	22	None
	AS			18		-0.7	0.02	10	Strong
	SBC			20		0.13	0.5	31	Weak
T1RI	SBL	0.24	0.21	30	Weak	-0.1	0.66	21	Weak
	AS	-0.2	0.47	16	Weak	0.29	0.42	10	Weak
	SBC	0.04	0.83	29	None	-0.44	0.07	18	Medium
T2LS	SBL	0.04	0.83	29	None	-0.44	0.07	18	Medium
	AS	-0.17	0.48	19	Weak	-0.02	0.94	14	None
	SBC	-0.1	0.58	34	None	0.41	0.08	19	Medium
T2RS	SBL	-0.1	0.58	34	None	0.41	0.08	19	Medium
	AS	-0.3	0.2	21	Medium	0.11	0.72	14	Weak
	SBC	-0.35	0.1	23	Medium	0.34	0.08	29	Medium
T2LI	SBL	-0.03	0.85	33	None	0.11	0.65	19	Weak
	AS	-0.09	0.7	22	None	-0.1	0.73	14	Weak
	SBC	-0.23	0.29	23	Weak	0	0.99	29	None
T2RI	SBL	0.02	0.92	32	None	-0.15	0.52	20	Weak
	AS	-0.37	0.1	21	Medium	0.09	0.75	14	None
	SBC	-0.04	0.87	21	None	0.24	0.23	27	Weak
T3LS	SBL	0.07	0.73	24	None	0.06	0.8	20	None
	AS	-0.35	0.15	19	Medium	-0.42	0.1	16	Medium

Appendix B-11 Correlation between osteophyte score and facet size continued

			M	ale		Female			
Frank	Camala	Correlati			Strength	Correlati			Strength
Facet	Sample	on Coefficie	р	N	of Correlati	on Coefficie	р	N	of Correlati
		nt			on	nt			on
	SBC	-0.29	0.2	22	Weak	0.05	0.8	27	None
T3RS	SBL	0.15	0.46	26	Weak	0.28	0.24	20	Weak
	AS			22		0.18	0.48	17	Weak
	SBC	-0.39	0.07	22	Medium	0.31	0.12	27	Medium
T3LI	SBL	-0.14	0.5	24	Weak	0.1	0.67	21	Weak
	AS	0.3	0.21	19	Medium	0.12	0.69	14	Weak
	SBC	-0.33	0.13	22	Medium	0.32	0.12	26	Medium
T3RI	SBL	-0.18	0.38	25	Weak	0.17	0.47	21	Weak
	AS	-0.04	0.86	19	None	-0.14	0.63	15	Weak
	SBC	-0.14	0.56	19	Weak	0.12	0.57	25	Weak
T4LS	SBL	-0.01	0.95	26	None	0.05	0.85	18	None
	AS	-0.5	0.03	20	Strong	-0.34	0.19	16	Medium
	SBC	-0.12	0.63	19	Weak	0.13	0.56	24	Weak
T4RS	SBL	0.1	0.65	24	None	0.13	0.61	18	Weak
	AS	0.02	0.93	20	None	0.19	0.48	16	Weak
	SBC	-0.41	0.08	19	Medium	0.35	0.09	24	Medium
T4LI	SBL	0.03	0.88	24	None	0.49	0.05	16	Strong
	AS								
	SBC	-0.16	0.54	17	Weak	0.01	0.98	19	None
T4RI	SBL	0.5	0.05	16	Strong	0.45	0.14	12	Medium
	AS	-0.08	0.8	13	None	0.24	0.46	12	Weak
	SBC	-0.18	0.47	19	Weak	0	0.98	25	None
T5LS	SBL	0.29	0.19	22	Weak	-0.05	0.83	19	None
	AS	0.04	0.88	19	None	0.23	0.39	16	Weak
	SBC	-0.05	0.85	19	None	-0.12	0.57	25	Weak
T5RS	SBL	0.4	0.07	21	Medium	-0.12	0.63	19	Weak
	AS	-0.31	0.18	20	Medium	0.41	0.11	16	Medium
	SBC	-0.32	0.21	17	Medium	-0.11	0.63	22	Weak
T5LI	SBL	-0.04	0.86	20	None	0.3	0.19	20	Medium
	AS			14		-0.25	0.37	15	Weak
	SBC	0.07	0.78	18	None	0.1	0.66	22	Weak
T5RI	SBL	0.01	0.98	23	None	-0.07	0.78	20	None
	AS	0.29	0.3	15	Weak	-0.2	0.49	15	Weak
	SBC	-0.69	0	17	Strong	-0.02	0.93	26	None
T6LS	SBL	-0.1	0.63	25	Weak	-0.18	0.47	19	Weak
	AS	0.2	0.46	16	Weak	0.17	0.56	14	Weak
	SBC	-0.45	0.07	17	Medium	0.1	0.62	26	Weak
T6RS	SBL	0.09	0.67	25	None	-0.14	0.57	19	Weak
	AS	0.3	0.26	16	Medium	-0.06	0.84	14	None

Appendix B-11 Correlation betwee	en osteophyte score	and facet size continued

			M	ale		Female				
Facet	Sample	Correlati on Coefficie nt	р	N	Strength of Correlati on	Correlati on Coefficie nt	р	Ν	Strength of Correlati on	
	SBC			19		-0.2	0.31	27	Weak	
T6LI	SBL	0.05	0.84	22	None	0.18	0.44	20	Weak	
	AS	0.38	0.18	14	Medium	0.32	0.26	14	Medium	
	SBC	0.03	0.92	19	None	-0.19	0.34	28	Weak	
T6RI	SBL	-0.11	0.61	23	Weak	-0.2	0.43	17	Weak	
	AS			14		-0.05	0.86	14	None	
	SBC	-0.17	0.46	20	Weak	0.1	0.6	28	Weak	
T7LS	SBL	0.04	0.85	25	None	0.25	0.33	18	Weak	
	AS	-0.05	0.87	14	None	0.04	0.9	15	None	
	SBC	0.15	0.52	20	Weak	-0.11	0.59	26	Weak	
T7RS	SBL	-0.12	0.59	24	Weak	-0.17	0.52	16	Weak	
	AS	-0.01	0.97	15	None	-0.38	0.17	15	Medium	
	SBC			18		-0.18	0.37	27	Weak	
T7LI	SBL	-0.04	0.87	22	None	-0.2	0.43	17	Weak	
	AS	-0.15	0.62	13	Weak	0.43	0.11	15	Medium	
	SBC			19		0.08	0.71	27	None	
T7RI	SBL	-0.11	0.61	23	Weak	-0.2	0.43	17	Weak	
	AS			13		-0.39	0.19	13	Medium	
	SBC	0	1	21	None	-0.13	0.54	24	Weak	
T8LS	SBL	0.04	0.85	25	None	0.25	0.33	18	Weak	
	AS	0.5	0.07	14	Strong	0.57	0.03	15	Strong	
	SBC	-0.02	0.94	21	None	0.19	0.37	24	Weak	
T8RS	SBL	-0.16	0.45	25	Weak	0.16	0.52	18	Weak	
	AS	0.54	0.05	14	Strong	-0.03	0.92	15	None	
	SBC	-0.33	0.14	21	Medium	-0.09	0.68	24	None	
T8LI	SBL	-0.12	0.56	25	Weak			16		
	AS			15		-0.25	0.37	15	Weak	
	SBC	-0.22	0.32	22	Weak	0	0.99	24	None	
T8RI	SBL	0.15	0.49	25	Weak	0.36	0.16	17	Medium	
	AS	-0.31	0.26	15	Medium	-0.07	0.8	14	None	
	SBC	-0.08	0.73	21	None	0.11	0.59	29	Weak	
T9LS	SBL	0.29	0.16	26	Weak	-0.09	0.74	17	None	
	AS	-0.22	0.39	18	Weak	0.36	0.18	16	Medium	
	SBC	0.08	0.73	21	None	0.19	0.32	29	Weak	
T9RS	SBL	0.55	0	26	Strong			17		
	AS	0.01	0.98	18	None	-0.11	0.68	16	Weak	

Appendix B-11 Correlation between osteophyte score and facet size continued

			M	ale		Female			
Facet	Sample	Correlati on Coefficie nt	р	N	Strength of Correlati on	Correlati on Coefficie nt	р	N	Strength of Correlati on
	SBC			19		-0.03	0.9	29	None
T9LI	SBL	-0.08	0.72	25	None			17	
	AS	-0.37	0.14	17	Medium	0.18	0.54	14	Weak
	SBC	-0.09	0.73	19	None	-0.14	0.5	27	Weak
T9RI	SBL	-0.22	0.32	22	Weak			17	
	AS			16		-0.16	0.57	15	Weak
	SBC	-0.07	0.76	20	None	0.03	0.9	28	None
T10LS	SBL	0.1	0.65	25	None	0.36	0.17	16	Medium
	AS	0.22	0.38	18	Weak	-0.43	0.15	13	Medium
	SBC	0.27	0.25	20	Weak	0.01	0.97	28	None
T10RS	SBL	0.26	0.21	26	Weak			16	
	AS	0.45	0.07	17	Medium	-0.07	0.82	13	None
	SBC	-0.22	0.37	19	Weak	0.1	0.63	27	Weak
T10LI	SBL	0.07	0.74	26	None			16	
	AS	0.47	0.06	17	Medium	0.27	0.4	12	Weak
	SBC	-0.12	0.61	19	Weak	-0.02	0.93	27	None
T10RI	SBL	-0.28	0.17	26	Weak			15	
	AS	0.41	0.1	17	Medium	0.62	0.03	12	Strong
	SBC	0.19	0.42	20	Weak	0.03	0.87	29	None
T11LS	SBL	-0.04	0.85	23	None	0.15	0.13	16	Weak
	AS	0.06	0.83	14	None	-0.28	0.36	13	Medium
	SBC	-0.06	0.78	21	None	0.08	0.67	29	None
T11RS	SBL	-0.01	0.96	24	None	-0.29	0.28	16	Weak
	AS	-0.17	0.51	17	Weak	0.51	0.07	14	Strong
	SBC	0.29	0.22	20	Weak	-0.04	0.84	28	None
T11LI	SBL	0	0.99	23	None	0.12	0.66	15	Weak
	AS	-0.02	0.95	17	None	0.07	0.81	14	None
	SBC	0.14	0.57	20	Weak	0.13	0.51	28	Weak
T11RI	SBL	0.01	0.98	24	None	0.42	0.11	16	Medium
	AS			15		0.5	0.12	11	Strong
	SBC	0.03	0.89	18	None	-0.16	0.44	26	Weak
T12LS	SBL	-0.06	0.8	22	None	-0.39	0.14	16	Medium
	AS	-0.14	0.63	14	Weak	0.15	0.63	13	Weak
	SBC	-0.57	0.01	18	Strong	0.18	0.39	26	Weak
T12RS	SBL	-0.04	0.86	22	None	0.2	0.47	16	Weak
	AS	-0.28	0.34	14	Weak	-0.14	0.65	13	Weak

Appendix B-11 Correlation between	osteophyte score and	I facet size continued

			Ma	ale					
Facet	Sample	Correlati on Coefficie nt	р	N	Strength of Correlati on	Correlati on Coefficie nt	р	N	Strength of Correlati on
	SBC	0.02	0.92	19	None	0.14	0.47	27	Weak
T12LI	SBL	0.16	0.49	22	Weak	0.23	16	-0.37	Weak
	AS	-0.18	0.57	12	Weak	-0.37	0.29	10	Medium
	SBC	-0.18	0.48	18	Weak	0	1	27	None
T12RI	SBL	0.32	0.15	22	Medium	-0.04	0.88	16	None
	AS			12		-0.44	0.15	12	Medium
	SBC	-0.13	0.61	19	Weak	0.09	0.68	25	None
L1LS	SBL	0.4	0.07	22	Medium	-0.31	0.31	13	Medium
	AS	-0.08	0.8	14	None	0.18	0.52	15	Weak
	SBC	-0.22	0.39	18	Weak	-0.04	0.85	23	None
L1RS	SBL	-0.13	0.56	24	Weak	0.05	0.88	12	None
	AS	0.06	0.84	13	None	0.07	0.82	14	None
	SBC	0.12	0.64	19	Weak	-0.08	0.73	24	None
L1LI	SBL	0.09	0.69	25	None	0.08	0.8	13	None
	AS			13		-0.1	0.7	16	Weak
	SBC	-0.14	0.57	19	Weak	0.09	0.68	24	None
L1RI	SBL	-0.09	0.68	24	None	-0.4	0.2	12	Medium
	AS	-0.53	0.05	14	Strong	-0.05	0.86	15	None
	SBC	0.17	0.45	21	Weak	0.32	0.11	26	Medium
L2LS	SBL	0.36	0.1	22	Medium	0.31	0.3	13	Medium
	AS	0	0.99	17	None	0.24	0.43	13	Weak
	SBC	-0.17	0.46	21	Weak	-0.01	0.96	25	None
L2RS	SBL	0.09	0.7	23	None	0.1	0.77	11	Weak
	AS	-0.36	0.18	16	Medium	0.05	0.88	13	None
	SBC	0.05	0.82	21	None	0.2	0.34	25	Weak
L2LI	SBL	0.44	0.03	24	Medium	0.56	0.06	12	Strong
	AS	-0.42	0.13	14	Medium	-0.23	0.45	13	Weak
	SBC	-0.27	0.24	20	Weak	-0.1	0.63	25	Weak
L2RI	SBL	-0.04	0.86	24	None	0.38	0.22	12	Medium
	AS	0.1	0.72	15	Weak	-0.23	0.44	14	Weak
	SBC	0.16	0.49	20	Weak	-0.06	0.77	28	None
L3LS	SBL	0.2	0.4	20	Weak	-0.4	0.18	13	Medium
	AS	-0.08	0.75	17	None	0.13	0.63	17	Weak
	SBC	0.27	0.26	20	Weak	-0.08	0.71	25	None
L3RS	SBL	-0.05	0.86	18	None	-0.29	0.35	13	Weak
	AS	-0.04	0.9	15	None	0.02	0.96	12	None
	SBC	0.05	0.85	21	None	-0.24	0.22	29	Weak
L3LI	SBL	-0.14	0.56	20	Weak	0.3	0.27	15	Medium
	AS	0.32	0.2	18	Medium	-0.31	0.26	15	Medium

Facet	Sample	Male				Female			
		Correlati on Coefficie nt	р	N	Strength of Correlati on	Correlati on Coefficie nt	р	N	Strength of Correlati on
L3RI	SBC	-0.1	0.68	21	Weak	0.08	0.7	27	None
	SBL	-0.22	0.39	17	Weak	-0.08	0.78	15	None
	AS	-0.22	0.44	15	Weak	0.3	0.31	13	Medium
L4LS	SBC	0.17	0.48	20	Weak	0.19	0.36	25	Weak
	SBL	-0.07	0.78	19	None			10	
	AS	-0.22	0.39	17	Weak	-0.08	0.78	15	None
L4RS	SBC	-0.17	0.48	20	Weak	-0.23	0.28	25	Weak
	SBL	0.35	0.13	20	Medium	0.14	0.73	9	Weak
	AS	-0.22	0.39	17	Weak	-0.08	0.78	15	None
L4LI	SBC	0.27	0.26	20	Weak	-0.07	0.73	25	None
	SBL	0.07	0.78	18	None	0.61	0.06	10	Strong
	AS	-0.02	0.95	16	None	-0.64	0.02	12	Strong
L4RI	SBC	0.59	0.01	20	Strong	-0.28	0.16	26	Weak
	SBL	0.03	0.89	20	None	-0.52	0.12	10	Strong
	AS	0.25	0.37	15	Weak	0.17	0.59	12	Weak
L5LS	SBC	0.01	0.96	19	None	-0.25	0.27	21	Weak
	SBL	0.02	0.93	24	None	-0.54	0.06	13	Strong
	AS	0.27	0.35	14	Weak	-0.47	0.12	12	Medium
L5RS	SBC	0.04	0.88	19	None	0.06	0.8	21	None
	SBL	-0.19	0.42	21	Weak	-0.17	0.58	13	Weak
	AS	0.41	0.17	13	Medium	0.04	0.9	12	None
L5LI	SBC	-0.22	0.36	19	Weak	0.31	0.17	21	Medium
	SBL	-0.04	0.84	24	None	-0.17	0.59	13	Weak
	AS	0.1	0.76	12	Weak	-0.01	0.98	11	None
L5RI	SBC	0.08	0.75	19	None	-0.06	0.81	21	None
	SBL	0.14	0.54	22	Weak	-0.39	0.19	13	Medium
	AS	-0.11	0.72	14	Weak	-0.35	0.26	12	Medium

Appendix B-11 Correlation between osteophyte score and facet size continued

Male Female Facet Sample Correlation Strength of Correlation Strength of р Ν р Ν Coefficient Correlation Coefficient Correlation SBC C1LS SBL AS SBC C1RS SBL AS SBC C1LI SBL AS SBC C1RI SBL AS -0.138 SBC 0.529 23 -0.028 0.873 None Weak 35 29 C2LS SBL -0.079 0.684 -0.189 0.426 20 Weak None 11 6 AS 23 SBC 0.134 0.541 Weak 0.041 0.817 35 None -0.044 C2RS SBL -0.019 0.929 29 None 0.853 20 None 10 AS 6 22 0.391 SBC -0.103 0.65 Weak -0.15 35 Weak C2LI 0.171 10.75 29 0.054 0.821 20 SBL Weak None 11 6 AS SBC -0.179 0.425 22 Weak -0.207 0.233 35 Weak C2RI SBL 0.016 0.0932 29 None 0.288 0.218 20 Weak 10 5 AS SBC -0.077 0.754 19 Weak 0.144 0.43 Weak 32 C3LS SBL 0.247 0.205 28 Weak -0.098 0.671 21 None 0.058 11 5 AS 0.865 none 0.424 20 SBC 0.189 Weak 0.464 0.007 32 Medium C3RS SBL 0.049 0.803 28 None -0.038 0.874 20 None -0.338 0.309 11 5 AS Medium SBC 0.001 0.996 20 None 0.031 0.87 31 None C3LI SBL 0.055 0.783 27 -0.057 0.807 21 None None 0.126 0.711 11 Weak 5 AS SBC -0.186 0.43 20 31 Weak 0.176 0.345 Weak C3RI SBL 0.024 0.903 28 None -0.171 0.47 20 Weak -0.229 0.525 10 AS Weak 5

Appendix B-12 Correlation between vertebral osteophytosis and facet size
			M	ale			Fema	ale	
Facet	Sample	Correlation Coefficient	р	Ν	Strength of Correlation	Correlation Coefficient	р	Ν	Strength of Correlation
	SBC	-0.043	0.86	19	None	0.169	0.39	28	Weak
C4LS	SBL	0.429	0.02	29	Medium	-0.008	0.971	22	None
	AS			10				Female N Strength of Correlation 39 28 Weak 71 22 None 39 28 Weak 71 22 None 39 28 Weak 71 22 None 39 28 Weak 39 28 Weak 39 28 Weak 39 28 Weak 71 22 None 42 3 None 70 29 Medium 54 22 Weak 70 34 Weak 54 24 Strong 8 17 34 Weak 56 24 Weak 56 24 Weak 56 24 Weak 56 34 None 57 24 Weak 58 24 Weak 58 24 Weak 58 24 Weak 58 </td	
	SBC	-0.207	0.396	19	Medium	0.287	0.139	28	Weak
C4RS	SBL	0.095	0.626	29	None	-0.063	0.781	22	None
	AS			11				7	
	SBC	0.096	0.688	20	None	0.375	0.045	29	Medium
C4LI	SBL	0.072	0.712	29	None	-0.023	0.919	23	None
	AS			11				7	
	SBC	-0.155	0.479	23	Weak	0.458	0.012	29	Medium
C4RI	SBL	0.163	0.397	29	Weak	0.138	0.54	22	Weak
	AS			11				7	
	SBC	-0.155	0.469	24	Weak	0.158	0.372	34	Weak
C5LS	SBL	0.046	0.808	30	None	0.564	0.004	24	Strong
	AS	0.409	0.24	10	Medium			8	
	SBC	-0.09	0.677	24	None	0.241	0.17	34	Weak
C5RS	SBL	0.278	0.137	30	Weak	0.242	0.256	24	Weak
	AS	0.724	0.009	10	Strong			8	
	SBC	-0.204	0.338	24	Weak	-0.019	0.916	34	None
C5LI	SBL	0.025	0.898	29	None	-0.136	0.527	24	Weak
	AS	0.088	0.808	10	None			8	
	SBC	0.046	0.808	30	None	0.106	0.549	34	Weak
C5RI	SBL	0.536	0.003	29	Strong	0.117	0.585	24	Weak
	AS	0.74	0.014	10	Strong			8	
	SBC	0.278	0.137	30	Weak	0.153	0.373	34	Weak
C6LS	SBL	0.077	0.685	30	None	-0.128	0.562	23	weak
	AS	0.717	0.013	11	Strong			6	
	SBC	0.025	0.898	29	None	0.472	0.004	36	Medium
C6RS	SBL	0.07	0.717	29	None	0.248	0.243	24	Weak
	AS	0.53	0.093	11	Strong			7	
	SBC	0.536	0.003	29	Strong	-0.109	0.528	36	Weak
C6LI	SBL	-0.068	0.715	31	None	0.156	0.476	23	Weak
	AS	0.735	0.01	11	Strong			7	
	SBC	-0.222	0.348	20	Weak	0.014	0.937	36	None
C6RI	SBL	-0.037	0.846	30	None	0.201	0.359	23	Weak
	AS	0.646	0.032	11	Strong			7	

			M	ale			Fema	ale	
Facet	Sample	Correlation Coefficient	р	N	Strength of Correlation	Correlation Coefficient	р	Ν	Strength of Correlation
	SBC	-0.245	0.297	20	Weak	0.056	0.744	37	None
C7LS	SBL	-0.008	0.967	32	None	-0.146	0.508	23	Weak
	AS	0.906	0.001	11	Strong			5	
	SBC	-0.221	0.35	20	Weak	0.231	0.176	36	Weak
C7RS	SBL	-0.295	0.101	32	Weak	-0.146	0.505	23	Weak
	AS	0.124	0.716	11	Weak			5	
	SBC	0.13	0.544	24	Weak	0.062	0.715	37	None
C7LI	SBL	0.129	0.481	32	Weak	-0.077	0.727	23	None
	AS	0.057	0.86	12	None			5	
	SBC	0.251	0.236	24	Weak	0.116	0.496	37	Weak
C7RI	SBL	0.254	0.161	32	Weak	-0.136	0.536	23	Weak
	AS	-0.333	0.29	12	Medium			5	
	SBC	0.067	0.77	20	None	0.01	0.951	37	None
T1LS	SBL	0.365	0.103	21	Medium	0.365	0.103	21	Medium
	AS	0.228	0.587	8	Weak			6	
	SBC	-0.11	0.636	21	Weak	0.094	0.576	38	None
T1RS	SBL	0.215	0.337	22	Weak	0.215	0.337	22	Weak
	AS	-0.218	0.572	9	Weak	Weak Use Children Chi		7	
	SBC	0.096	0.68	21	None	0.351	0.03	38	Medium
T1LI	SBL	0.235	0.293	22	Weak	0.235	0.293	22	Weak
	AS	0.245	0.526	9	Weak			7	
	SBC	-0.155	0.502	21	Weak	0.209	0.209	38	Weak
T1RI	SBL	-0.012	0.957	22	None	-0.012	0.957	22	None
	AS	-0.116	0.766	9	Weak			7	
	SBC	-0.018	0.937	21	None	0.001	0.998	22	None
T2LS	SBL	-0.018	0.937	21	None	0.001	0.998	22	None
	AS	0.075	0.838	10	None	-0.306	0.933	8	Medium
	SBC	-0.072	0.75	22	None	-0.072	0.75	22	None
T2RS	SBL	-0.072	0.75	22	None	-0.072	0.75	22	None
	AS	-0.141	0.698	10	Weak	0.455	0.257	8	Medium
	SBC	-0.19	0.385	23	Weak	0.086	0.614	37	None
T2LI	SBL	-0.133	0.557	22	Weak	-0.133	0.557	22	Weak
	AS	-0.103	0.778	10	Weak	-0.307	0.459	8	Medium
	SBC	-0.195	0.373	23	Weak	0.249	0.137	37	Weak
T2RI	SBL	0.461	0.031	22	Medium	0.461	0.031	22	Medium
	AS	-0.108	0.767	10	Weak	0.746	0.034	8	Strong

			M	ale			Fema	ale	
Facet	Sample	Correlation Coefficient	р	N	Strength of Correlation	Correlation Coefficient	р	N	Strength of Correlation
	SBC	-0.021	0.928	22	None	-0.021	0.899	38	None
T3LS	SBL	-0.013	0.957	20	None	-0.013	0.957	20	None
	AS	-0.108	0.737	12	Weak	0.033	0.938	8	None
	SBC	0.234	0.295	22	Weak	0.225	0.174	38	Weak
T3RS	SBL	0.35	0.13	20	Medium	0.35	0.13	20	Medium
	AS	-0.179	0.559	13	Weak	0.292	0.445	9	Weak
	SBC	-0.241	0.279	22	Weak	0.024	0.888	38	None
T3LI	SBL	-0.194	0.412	20	Weak	-0.194	0.412	20	Weak
	AS	-0.287	0.343	13	Weak	0.521	0.186	8	Strong
	SBC	0.576	0.005	22	Strong	0.311	0.057	38	Medium
T3RI	SBL	0.048	0.844	19	None	0.239	0.309	20	Weak
	AS	-0.16	0.603	13	Weak	0.19	0.625	9	Weak
	SBC	0.027	0.902	24	None	0.008	0.964	37	None
T4LS	SBL	-0.157	0.52	19	Weak	-0.157	0.52	19	Weak
	AS	-0.314	0.411	9	Medium	-0.26	0.5	9	Weak
	SBC	0.273	0.197	24	Weak	0.216	0.207	36	Weak
T4RS	SBL	0.008	0.973	19	None	0.008	0.973	19	None
	AS	-0.145	0.71	9	Weak	0.261	0.498	9	Weak
	SBC	0.371	0.074	24	Medium	-0.248	0.145	36	Weak
T4LI	SBL	-0.349	0.155	18	Medium	-0.349	0.155	18	Medium
	AS	-0.115	0.769	9	Weak	-0.567	0.143	8	Strong
	SBC	0.403	0.07	21	Medium	0.375	0.065	25	Medium
T4RI	SBL	0.14	0.638	12	Weak			12	
	AS	0.319	0.538	6	Medium	0.776	0.123	5	Strong
	SBC	0.245	0.259	23	Weak	-0.33	0.053	35	Medium
T5LS	SBL	-0.07	0.775	19	None	-0.07	0.775	19	None
	AS	-0.51	0.197	8	Strong	-0.125	0.767	8	Weak
	SBC	0.417	0.048	23	Medium	0.602	0.001	35	Strong
T5RS	SBL	0.257	0.287	19	Weak	0.257	0.287	19	Weak
	AS	0.201	0.633	8	Weak	0.198	0.638	8	Weak
	SBC	0.257	237	23	Weak	-0.016	0.929	35	None
T5LI	SBL	-0.023	0.925	19	None	-0.023	0.925	19	None
	AS	0.269	0.559	7	Weak	-0.28	0.502	8	Weak
	SBC	0.25	0.25	23	Weak	0.253	0.142	35	Weak
T5RI	SBL	0.175	0.473	19	Weak	0.175	0.473	19	Weak
	AS	-0.17	0.715	7	Weak	-0.248	0.0240.888-0.1940.4120.5210.1860.3110.0570.2390.3090.190.6250.0080.96440.1570.520.260.9730.260.2070.260.9730.2610.4980.2610.4980.2610.4980.2610.4980.2610.4980.2610.4980.2610.4980.2610.4980.2610.4980.2610.1430.2610.1430.2610.1430.3750.0650.3750.1430.3750.0650.7760.1230.7760.7750.1250.7670.6020.0010.12570.2870.1980.6380.1980.5020.2530.4730.1750.4730.1750.473	8	Weak

			M	ale			Fema	ale	
Facet	Sample	Correlation Coefficient	р	N	Strength of Correlation	Correlation Coefficient	р	N	Strength of Correlation
	SBC	0.436	0.033	24	Medium	-0.004	0.981	35	None
T6LS	SBL	0.031	0.893	21	None	0.031	0.893	21	None
	AS	0.258	0.471	10	Weak	-0.256	0.541	8	Weak
	SBC	0.118	0.582	24	Weak	0.084	0.631	35	None
T6RS	SBL	0.013	0.955	21	None	0.013	0.955	21	None
	AS	-0.134	0.713	10	Weak	0.103	0.809	8	Weak
	SBC	0.469	0.021	24	Medium	0.082	0.629	37	None
T6LI	SBL	-0.414	0.062	21	Medium	-0.141	0.062	21	Weak
	AS	-0.046	0.906	9	None	-0.624	0.098	8	Strong
	SBC	0.381	0.066	24	Medium	0.164	0.332	37	Weak
T6RI	SBL	0.234	0.212	30	Weak	0.007	0.977	20	None
	AS	0.095	0.808	9	None	-0.343	0.406	8	Medium
	SBC	0.512	0.008	26	Strong	0.019	0.909	37	None
T7LS	SBL	0.06	0.75	31	None	-0.144	0.543	20	Weak
	AS	0.069	0.871	8	None	-0.512	0.158	9	Strong
	SBC	0.14	0.494	26	Weak	0.042	0.803	37	None
T7RS	SBL	0.259	0.167	30	Weak	-0.383	0.095	20	Medium
	AS	0.303	0.466	8	Medium	-0.259	0.501	0.501 9	None
	SBC	0.233	0.263	25	Weak	0.077	0.653	37	None
T7LI	SBL	0.259	0.175	29	Weak	-0.003	0.989	200	None
	AS	0.296	0.477	8	Weak	0.78	0.013	9	Strong
	SBC	-0.145	0.489	25	Weak	0.041	0.811	37	None
T7RI	SBL	0.234	0.212	30	Weak	0.007	0.977	20	None
	AS	0.188	0.655	8	Weak	-0.154	0.715	8	Weak
	SBC	0.332	0.113	24	Medium	-0.017	0.921	37	None
T8LS	SBL	0.06	0.75	31	None	-0.144	0.543	20	Weak
	AS	0.274	0.511	8	Weak	0.047	0.904	9	None
	SBC	0.037	0.863	24	None	-0.015	0.933	35	None
T8RS	SBL	0.135	0.468	31	Weak	-0.217	0.357	20	Weak
	AS	0.096	0.822	8	None	-0.13	0.738	9	Weak
	SBC	0.49	0.013	25	Medium	0.03	0.865	35	None
T8LI	SBL	0.2	0.28	31	Weak	-0.289	0.245	18	Weak
	AS	-0.163	0.699	8	Weak	-0.345	0.363	9	Medium
	SBC	0.433	0.031	25	Medium	0.011	0.949	35	None
T8RI	SBL	0.11	0.555	31	Weak	-0.009	0.972	19	None
	AS	-0.034	0.936	8	None	0.216	0.576	9	Weak

			M	ale		Female			
Facet	Sample	Correlation Coefficient	р	N	Strength of Correlation	Correlation Coefficient	р	N	Strength of Correlation
	SBC	0.358	0.073	26	Medium	0.34	0.037	38	Medium
T9LS	SBL	0.104	0.583	30	Weak	-0.025	0.916	20	None
	AS	0.434	0.21	10	Medium	-0.448	0.226	9	Medium
	SBC	0.24	0.237	26	Weak	0.285	0.083	38	Weak
T9RS	SBL	0.093	0.627	30	None	0.005	0.985	20	None
	AS	0.082	0.822	10	None	0.09	0.818	9	None
	SBC	0.148	0.481	25	Weak	0.066	0.695	38	None
T9LI	SBL	-0.099	0.604	30	None	-0.335	0.149	20	Medium
	AS	-0.145	0.689	10	Weak	-0.649	0.059	9	Strong
	SBC	0.17	0.416	25	Weak	0.071	0.675	37	None
T9RI	SBL	0.005	0.98	28	None	-0.058	0.808	20	None
	AS	-0.33	0.352	10	None	-0.537	0.136	9	Strong
	SBC	0.082	0.698	25	None	0.153	0.359	38	Weak
T10LS	SBL	0.054	0.781	29	None	-0.117	0.622	20	Weak
	AS	0.004	0.991	10	None	-0.488	0.182	9	Medium
	SBC	0.274	0.185	25	Weak	0.08	0.634	38	None
T10RS	SBL	0.039	0.837	30	None	-0.004	0.987	20	None
	AS	0.021	0.953	10	None	one -0.011 0.977	9	None	
	SBC	0.155	0.449	26	Weak	0.165	0.322	38	Weak
T10LI	SBL	-0.118	0.541	29	Weak	-0.206	0.384	20	Weak
	AS	0.094	0.796	10	None	-0.042	0.915	9	None
	SBC	0.352	0.077	26	Medium	0.176	0.29	38	Weak
T10RI	SBL	-0.155	0.423	29	Weak	-0.124	0.601	20	Weak
	AS	0.004	0.991	10	None	0.227	0.556	9	Weak
	SBC	0.169	0.408	26	Weak	-0.022	0.894	38	None
T11LS	SBL	-0.125	0.534	27	Weak	0.057	0.81	20	None
	AS	-0.225	0.592	8	Weak	0.158	0.685	9	Weak
	SBC	-0.116	0.573	26	Weak	-0.022	0.896	38	None
T11RS	SBL	0.01	0.959	28	None	0.23	0.315	21	Weak
	AS	-0.448	0.227	9	Medium	-0.093	0.812	9	None
	SBC	0.012	0.956	25	None	-0.093	0.578	38	None
T11LI	SBL	-0.053	0.793	27	None	0.408	0.067	21	Medium
	AS	-0.113	0.773	9	Weak	-0.012	0.975	9	None
	SBC	0.156	0.457	25	Weak	-0.132	0.43	38	Weak
T11RI	SBL	0.152	0.44	28	Weak	0.499	0.021	21	Medium
	AS	-0.059	0.881	9	None	0.308	0.457	8	Medium

			M	ale			Fem	ale	
Facet	Sample	Correlation Coefficient	р	Ν	Strength of Correlation	Correlation Coefficient	р	N	Strength of Correlation
	SBC	-0.105	0.624	24	Weak	-0.227	0.19	35	Weak
T12LS	SBL	-0.16	0.417	28	None	0.337	0.136	21	Medium
	AS	-0.031	0.942	8	None	0.194	0.617	9	Weak
	SBC	0.142	0.509	24	Weak	-0.073	0.678	35	None
T12RS	SBL	0.067	0.728	29	None	0.51	0.018	21	Strong
	AS	0.169	0.665	9	Weak	0.552	0.123	9	Strong
	SBC	0.039	0.852	25	None	-0.242	0.162	35	Weak
T12LI	SBL	0.21	0.275	29	Weak	0.435	0.049	21	Medium
	AS	0.04	0.919	9	None	0.213	0.613	8	Weak
	SBC	0.238	0.252	25	Weak	-0.356	0.036	35	Medium
T12RI	SBL	0.105	0.586	29	Weak	0.347	0.123	21	Medium
	AS	-0.266	0.489	9	Weak	0.197	0.612	9	Weak
	SBC	0.46	0.021	25	Medium	-0.105	0.55	35	Weak
L1LS	SBL	0.165	0.4	28	Weak	0.478	0.033	20	Medium
	AS	-0.361	0.306	10	Medium	0.001	0.998	8	None
	SBC	0.438	0.029	25	Medium	-0.196	0.267	34	Weak
L1RS	SBL	-0.14	0.478	28	Weak	0.248	0.292	20	Weak
	AS	-0.231	0.55	9	Weak	0.149	0.725	8	Weak
	SBC	0.302	0.142	25	Medium	-0.085	0.628	35	None
L1LI	SBL	-0.036	0.852	29	None	0.385	0.093	20	Medium
	AS	-0.211	0.559	10	Weak	-0.485	0.186	9	Medium
	SBC	0.073	0.728	25	None	-0.063	0.722	34	None
L1RI	SBL	0.056	0.776	28	None	0.147	0.535	20	Weak
	AS	-0.211	0.559	10	Weak	-0.122	0.754	9	Weak
	SBC	0.303	0.132	26	Medium	0.092	0.6	35	None
L2LS	SBL	0.058	0.764	29	None	0.244	0.3	20	Weak
	AS	-0.156	0.667	10	Weak	-0.027	0.949	8	None
	SBC	0.267	0.188	26	Weak	0.076	0.664	35	None
L2RS	SBL	0.066	0.74	28	None	-0.011	0.964	18	None
	AS	0.388	0.268	10	Medium	-0.255	0.542	8	Weak
	SBC	0.232	0.252	26	Weak	0.142	0.416	35	Weak
L2LI	SBL	-0.092	0.627	30	None	0.146	0.551	19	Weak
	AS	0.507	0.135	10	Strong	-0.34	0.409	8	Medium
	SBC	0.218	0.286	26	Weak	0.009	0.957	35	None
L2RI	SBL	-0.071	0.71	30	None	0.103	0.676	19	Weak
	AS	0.162	0.677	9	Weak	-0.238	0.57	8	Weak

			Ma	ale			Fema	ale	
Facet	Sample	Correlation Coefficient	р	N	Strength of Correlation	Correlation Coefficient	р	N	Strength of Correlation
	SBC	-0.086	0.698	23	None	-0.135	0.455	33	Weak
L3LS	SBL	0.183	0.141	22	Weak	-0.036	0.898	15	None
	AS	0.153	0.693	9	Weak	0.227	0.529	10	Weak
	SBC	0.033	0.88	23	None	-0.413	0.019	32	Medium
L3RS	SBL	-0.038	0.87	21	None	0.126	0.669	14	Weak
	AS	0.035	0.935	8	None	0.244	0.561	8	Weak
	SBC	0.155	0.602	23	Weak	-0.216	0.228	33	Weak
L3LI	SBL	-0.087	0.693	23	None	-0.065	0.811	16	None
	AS	0.6	0.155	7	Strong	0.1	0.784	10	Weak
	SBC	0.267	0.219	23	Weak	-0.022	0.901	33	None
L3RI	SBL	0.27	0.517	8	Weak	0.539	0.108	10	Strong
	AS	-0.114	0.488	8	Weak	-0.027	0.941	10	None
	SBC	0.079	0.701	26	None	-0.06	0.723	37	None
L4LS	SBL	-0.365	0.42	7	Medium			4	
	AS	0.27	0.517	8	Weak	0.539	0.108	10	Strong
	SBC	-0.021	0.918	26	None	-0.042	0.805	37	None
L4RS	SBL	-0.0386	0.393	7	None			3	
	AS	0.27	0.517	8	Weak	0.539	0.108	10	Strong
	SBC	0.347	0.083	26	Medium	-0.269	0.108	37	Weak
L4LI	SBL	0.083	0.859	7	None			4	
	AS	0.587	0.045	12	Strong	0.567	0.111	9	Strong
	SBC	0.577	0.002	26	Strong	-0.152	0.37	37	Weak
L4RI	SBL	0.206	0.658	7	Weak			4	
	AS	-0.079	0.818	11	None	0.29	0.449	9	Weak
	SBC	0.146	0.488	25	Weak	-0.042	0.815	33	None
L5LS	SBL	0.047	0.808	29	None	0.016	19	0.11	None
	AS	0.392	0.234	11	Medium	0.614	0.106	8	Strong
	SBC	0.165	0.43	25	Weak	0.146	0.416	33	Weak
L5RS	SBL	0.218	0.274	27	Weak	0.11	0.653	19	Weak
	AS	0.313	0.412	9	Medium	0.582	0.13	8	Strong
	SBC	0.708	0.001	25	Strong	-0.098	0.588	33	None
L5LI	SBL	0.061	0.751	29	None	0.697	0.001	19	Strong
	AS	0.116	0.766	9	Weak	0.492	0.216	8	Medium
	SBC	0.403	0.046	25	Medium	0.062	0.737	32	None
L5RI	SBL	-0.022	0.911	28	None	0.739	0.001	19	Strong
	AS	0.588	0.096	9	Strong	0.716	0.03001.10.52910010.66914410.561810.5613.310.78410010.78410010.78410010.9013.310.9013.310.9013.310.9013.310.90110010.90110010.9013.710.10810010.1083.710.1083.710.1083.710.1083.710.1083.710.1083.710.111910.1123.310.1133.810.13810.13810.216830.7373.210.1341.910.1353.310.1341.910.1353.310.1353.310.1363.310.1373.210.1341.910.1441.910.1543.310.1543.310.1543.310.1543.310.1543.310.1543.310.1543.310.1543.31	Strong	

Appendix C. Results for Facet Angle

Facet Area Sample SDI Male Female Ν Mean Std dev. Ν Mean Std dev. C1LS SBC 23 29.48 4.52 32 29.53 3.82 99.82 SBL 26 26.19 1.83 20 26.20 2.28 99.97 AS 16 26.69 10.11 9 28.56 10.98 93.46 C1RS SBC 23 30.17 32 30.19 99.95 5.97 6.19 SBL 26 26.15 1.74 20 26.55 2.74 98.51 AS 16 27.63 8.79 9 25.56 9.04 108.10 C1LI SBC 33.50 106.04 23 35.52 7.43 32 5.69 SBL 26 34.54 6.03 20 33.45 6.39 103.25 9 AS 14 38.86 9.84 27.89 10.45 139.33 C1RI SBC 23 37.48 32 33.44 112.08 7.11 5.09 SBL 35.08 20 33.75 103.93 26 5.26 5.86 AS 15 38.40 11.02 10 29.60 8.19 129.73 C2LS SBC 23 49.00 9.84 35 45.94 11.30 106.65 SBL 29 43.76 8.22 20 43.10 6.03 101.53 41.27 AS 12 54.92 11.92 11 13.96 133.06 C2RS SBC 23 48.96 11.92 35 45.06 9.50 108.65 107.76 SBL 29 43.48 5.31 20 40.35 4.65 AS 12 49.92 11.29 11 41.82 9.98 119.37 C2LI SBC 67.80 22 64.14 17.46 35 17.64 94.6 SBL 29 50.90 20 56.15 90.64 8.21 13.04 AS 12 77.50 15.20 10 55.70 14.47 139.14 C2RI SBC 23 64.43 35 90.39 15.77 71.29 19.54 SBL 29 51.17 7.19 20 57.80 12.60 88.53 AS 13 74.46 13.68 10 62.90 16.41 118.38 C3LS SBC 19 89.68 82.50 108.71 10.24 32 14.73 28 12.05 SBL 78.32 13.35 21 81.24 96.41 AS 16 72.44 13.62 10 63.60 11.53 113.90

Facet Area	Sample		Male			Female	I	SDI
		N	Mean	Std dev.	N	Mean	Std dev.	
C3RS	SBC	20	84.60	11.50	32	80.25	16.14	105.42
	SBL	28	73.64	15.30	20	79.30	12.39	92.87
	AS	16	74.00	17.75	10	56.40	14.77	131.21
C3LI	SBC	20	84.30	6.81	31	83.45	7.19	101.02
	SBL	27	76.44	10.33	21	82.43	6.95	92.74
	AS	16	96.31	12.99	10	82.00	14.86	117.45
C3RI	SBC	20	90.60	9.02	31	87.74	7.63	103.26
	SBL	28	78.54	8.29	20	84.80	4.87	92.61
	AS	14	105.07	12.63	10	84.00	10.93	125.09
C4LS	SBC	19	95.95	8.99	28	90.43	10.24	106.1
	SBL	29	91.45	9.59	22	92.73	8.11	98.62
	AS	14	85.71	10.37	10	88.70	7.06	96.63
C4RS	SBC	20	91.90	8.10	28	89.75	9.72	102.4
	SBL	29	90.45	9.18	22	90.41	10.77	100.04
	AS	14	89.86	12.35	10	84.60	18.43	106.21
C4LI	SBC	20	84.85	8.90	29	85.93	8.34	98.74
	SBL	29	82.79	10.68	23	83.70	5.12	98.92
	AS	13	98.23	12.59	10	91.70	11.74	107.12
C4RI	SBC	20	89.95	9.82	29	91.48	7.69	98.32
	SBL	29	85.97	5.47	22	86.95	6.43	98.86
	AS	13	96.23	18.65	10	100.90	9.78	95.37
C5LS	SBC	23	98.70	9.16	34	97.26	6.36	101.47
	SBL	30	93.33	14.24	24	95.08	6.09	98.16
	AS	11	91.64	14.20	10	97.20	11.69	94.28
C5RS	SBC	24	97.00	8.13	34	93.91	10.43	103.29
	SBL	30	94.67	6.97	24	94.21	7.30	100.49
	AS	11	86.18	11.65	10	99.50	12.28	86.61
C55LI	SBC	24	88.67	9.78	34	88.91	4.30	99.72
	SBL	29	85.90	8.80	24	87.13	5.28	98.59
	AS	11	95.91	10.36	10	99.50	11.28	96.39

Facet Area Female SDI Sample Male Std dev. Mean Std dev. Ν Mean Ν C5RI SBC 24 93.00 10.55 34 91.56 101.57 6.76 SBL 29 87.83 6.29 90.38 6.32 97.18 24 AS 11 104.18 13.38 10 100.00 10.09 104.18 C6LS SBC 21 37 98.32 9.76 100.33 6.87 SBL 30 98.07 23 99.30 98.75 5.31 6.60 AS 91.83 9.81 9 93.67 10.74 98.04 12 C6RS SBC 21 100.05 9.16 37 96.38 8.15 103.81 29 99.02 SBL 97.86 4.74 24 98.83 5.88 AS 12 88.00 15.86 9 96.11 18.58 91.56 C6LI SBC 37 91.68 6.47 100.15 21 91.81 7.12 SBL 31 88.29 4.58 23 90.96 4.03 97.07 9 97.00 AS 12 91.33 7.36 11.94 94.16 C6RI SBC 94.05 7.05 37 92.97 21 8.06 101.16 SBL 89.26 90.79 31 5.27 24 4.57 98.31 AS 12 94.17 9.43 9 98.78 7.66 95.33 C7LS SBC 25 98.64 39 100.71 7.57 97.95 6.65 SBL 95.84 24 96.04 99.79 32 5.32 4.80 AS 11 91.18 11.01 6 88.50 9.05 103.03 SBC 94.40 7.96 94.50 6.93 99.89 C7RS 25 38 SBL 32 5.90 24 92.58 5.79 100.25 92.81 AS 11 90.73 12.95 6 94.50 9.31 96.01 C7LI SBC 25 93.68 7.12 39 91.28 5.46 102.63 SBL 32 89.47 5.02 24 89.17 5.27 100.34 AS 12 96.00 12.50 6 96.17 12.91 99.83 C7RI SBC 25 94.40 6.51 39 94.23 6.10 100.18 92.79 97.67 SBL 32 90.63 5.24 24 5.82 98.83 6 9.03 103.67 AS 12 11.63 95.33 40 T1LS SBC 21 100.90 6.80 99.40 6.90 101.51 SBL 34 99.59 6.08 23 100.61 5.26 98.99 AS 15 95.00 10.74 12 99.17 95.80 11.15

Facet Area	Sample		Male	1		Female	1	SDI
		N	Mean	Std dev.	N	Mean	Std dev.	
T1RS	SBC	24	97.67	9.02	40	96.90	6.10	100.79
	SBL	34	100.12	5.96	23	98.17	7.15	101.98
	AS	14	99.50	9.57	12	98.17	6.75	101.36
T1LI	SBC	24	95.79	8.58	40	93.20	15.37	102.78
	SBL	35	94.11	5.16	23	92.04	4.24	102.25
	AS	14	108.36	6.52	12	108.42	11.03	99.95
T1RI	SBC	24	102.83	5.96	40	103.85	6.34	99.02
	SBL	35	94.94	7.05	23	97.39	6.74	97.49
	AS	14	110.79	4.46	12	107.25	10.80	103.30
T2LS	SBC	26	105.69	4.61	38	103.16	5.03	102.46
	SBL	34	103.59	5.07	22	105.23	4.07	98.44
	AS	13	102.69	5.34	14	103.86	9.78	98.88
T2RS	SBC	26	104.96	4.65	38	101.00	4.48	103.92
	SBL	35	102.37	5.69	22	103.41	3.46	99.00
	AS	13	106.23	5.99	14	108.79	5.38	97.65
T2LI	SBC	26	99.69	8.72	38	98.42	8.40	101.29
	SBL	34	95.44	4.61	22	93.23	4.08	102.37
	AS	13	111.69	7.66	14	109.86	8.93	101.67
T2RI	SBC	26	105.58	6.59	38	104.34	5.68	101.18
	SBL	34	98.94	5.32	22	98.27	6.57	100.68
	AS	13	109.92	8.46	14	98.07	37.11	112.08
T3LS	SBC	25	106.48	4.39	39	105.08	4.74	101.34
	SBL	28	105.07	3.97	21	104.62	3.09	100.43
	AS	15	105.80	6.21	14	106.50	5.65	99.34
T3RS	SBC	25	104.40	3.54	39	104.79	3.84	99.62
	SBL	29	105.52	3.79	21	105.00	3.08	100.49
	AS	16	110.25	5.89	14	108.00	5.28	102.08
T3LI	SBC	25	97.40	7.33	39	96.62	5.61	100.61
	SBL	29	94.28	4.08	21	94.76	4.65	99.49
	AS	15	111.80	8.49	13	111.62	10.02	100.17

Facet Area Female SDI Sample Male Std dev. Std dev. Ν Mean Ν Mean T3RI SBC 25 104.60 39 105.38 99.26 6.31 5.71 SBL 29 99.59 5.96 99.19 6.52 100.40 21 AS 16 109.06 7.02 14 109.86 5.05 99.28 T4LS SBC 26 37 100.64 105.42 4.25 104.76 3.93 SBL 30 19 99.25 103.63 3.93 104.42 3.29 AS 15 107.07 14 106.36 100.67 5.35 3.82 T4RS SBC 26 104.85 3.89 36 105.25 4.30 99.62 SBL 29 103.76 2.67 19 104.68 3.90 99.12 AS 15 5.41 14 106.00 3.98 103.96 110.20 T4LI SBC 100.42 7.29 103.77 26 36 96.78 6.80 SBL 29 93.34 3.70 18 94.50 4.57 98.78 AS 16 109.00 6.54 13 106.54 4.91 102.31 T4RI SBC 106.85 104.94 26 5.24 36 5.84 101.81 SBL 30 99.78 97.50 4.64 18 7.51 97.72 AS 16 110.25 6.12 108.15 5.61 101.94 13 SBC 100.61 T5LS 25 105.64 3.65 37 105.00 3.18 SBL 20 99.75 28 104.29 3.32 104.55 3.39 AS 16 104.94 5.37 10 105.30 3.95 99.66 SBC 4.98 100.81 T5RS 25 104.68 3.30 37 103.84 SBL 28 4.04 20 104.25 99.69 103.93 3.63 AS 16 6.60 10 5.19 103.54 109.75 106.00 T5LI SBC 25 97.84 7.56 36 97.42 6.17 100.43 SBL 26 94.88 5.02 20 94.45 4.89 100.46 AS 14 108.79 5.48 10 107.60 8.25 101.10 T5RI SBC 25 105.40 4.72 36 105.67 5.76 99.75 20 SBL 28 99.39 4.93 98.05 5.13 101.37 100.43 AS 14 109.07 7.38 10 108.60 7.17 T6LS SBC 26 104.23 3.02 35 104.29 2.98 99.95 SBL 31 21 105.71 3.70 98.56 104.19 4.36 AS 5.85 10 105.50 99.26 14 104.71 5.25

Facet Area	Sample		Male	1		Female	1	SDI
		N	Mean	Std dev.	N	Mean	Std dev.	
T6RS	SBC	26	104.73	4.79	35	102.74	5.53	101.93
	SBL	31	103.71	3.47	21	103.71	3.61	100.00
	AS	14	107.64	5.58	10	107.00	7.30	100.60
T6LI	SBC	26	99.12	7.58	37	95.97	5.89	103.27
	SBL	29	94.62	5.56	21	94.29	3.90	100.36
	AS	12	104.42	6.24	10	105.10	8.41	99.35
T6RI	SBC	26	104.19	6.44	37	102.65	6.19	101.5
	SBL	29	98.10	4.19	21	98.81	5.27	99.29
	AS	12	107.67	6.88	10	106.00	6.58	101.57
T7LS	SBC	26	104.96	3.62	38	105.08	4.60	99.89
	SBL	31	104.03	4.00	20	104.75	3.55	99.31
	AS	13	104.15	5.51	13	104.23	6.93	99.93
T7RS	SBC	26	102.92	4.08	38	104.05	4.58	98.91
	SBL	32	103.44	3.93	19	103.37	2.89	100.07
	AS	13	104.38	6.89	13	108.08	5.85	96.58
T7LI	SBC	25	96.32	5.48	38	96.68	5.36	99.62
	SBL	30	94.87	3.55	20	94.40	4.76	100.49
	AS	11	106.27	9.59	12	109.00	6.37	97.50
T7RI	SBC	25	102.56	5.81	38	102.76	6.86	99.8
	SBL	31	98.19	4.97	20	99.35	5.02	98.84
	AS	11	110.73	8.66	13	105.85	5.16	104.61
T8LS	SBC	25	104.36	4.32	36	105.14	4.06	99.26
	SBL	32	103.44	4.13	20	103.95	5.42	99.51
	AS	13	104.54	4.96	14	103.57	6.01	100.93
T8RS	SBC	25	104.08	4.17	36	103.03	5.17	101.02
	SBL	32	104.09	3.63	20	101.80	2.88	102.25
	AS	13	107.46	5.08	14	107.79	5.49	99.70
T8LI	SBC	26	95.35	7.03	36	94.92	6.22	100.45
	SBL	32	96.59	4.88	18	93.22	3.95	103.62
	AS	13	103.38	7.21	14	101.93	7.27	101.43

Facet Area Female SDI Sample Male Std dev. Std dev. Ν Mean Ν Mean T8RI SBC 102.62 4.96 36 102.22 100.38 26 5.22 SBL 32 96.81 19 97.37 4.40 99.43 5.81 AS 13 104.46 6.96 14 105.93 6.66 98.62 T9LS SBC 26 4.84 39 100.52 103.62 103.08 5.76 SBL 4.40 20 102.95 99.89 31 102.84 4.05 AS 14 102.29 4.41 13 100.69 101.58 6.10 T9RS SBC 26 101.92 4.04 39 102.44 5.00 99.5 SBL 3.64 20 101.15 3.08 101.99 31 103.16 AS 13 108.38 3.40 104.23 5.34 103.99 13 T9LI SBC 25 39 99.39 95.52 6.21 96.10 6.38 SBL 31 95.58 4.43 20 93.80 4.02 101.90 AS 14 100.29 7.35 12 103.00 6.19 97.36 T9RI SBC 98.92 25 102.12 5.22 39 103.23 5.93 SBL 20 96.55 101.88 30 98.37 5.24 5.56 AS 14 101.29 6.34 12 104.08 5.45 97.31 T10LS SBC 39 98.18 25 101.40 6.50 103.28 4.65 SBL 30 20 99.58 103.37 3.91 103.80 3.04 AS 15 104.13 4.47 12 102.08 5.43 102.01 T10RS SBC 4.46 25 103.00 5.59 39 101.33 101.64 SBL 3.93 20 102.10 3.77 101.73 31 103.87 AS 14 4.12 12 107.42 4.50 99.68 107.07 T10LI SBC 26 93.54 11.69 39 94.92 8.92 98.54 SBL 30 92.57 8.20 20 95.45 4.91 96.98 AS 15 104.00 8.25 12 104.67 5.55 99.36 T10RI SBC 26 97.04 13.49 39 99.41 8.15 97.61 20 97.79 SBL 30 96.47 6.20 98.65 6.91 98.11 AS 15 104.73 7.45 12 106.75 9.39 T11LS SBC 26 101.00 12.21 39 103.51 6.42 97.57 SBL 29 5.49 21 104.71 3.66 99.09 103.76 AS 5.57 12 104.75 99.98 15 104.73 3.96

Facet Area	Sample		Male	1		Female	1	SDI
		N	Mean	Std dev.	N	Mean	Std dev.	
T11RS	SBC	26	96.08	14.32	39	100.41	7.33	95.68
	SBL	29	102.97	5.74	21	102.71	3.68	100.24
	AS	15	105.67	8.48	12	108.83	6.18	97.09
T11LI	SBC	25	62.12	23.48	39	72.49	23.94	85.7
	SBL	28	76.39	24.45	21	86.14	18.78	88.68
	AS	14	80.57	25.53	12	78.67	21.77	102.42
T11RI	SBC	25	57.32	30.76	39	75.77	27.70	75.65
	SBL	29	76.79	25.25	21	82.38	21.76	93.22
	AS	14	83.14	25.92	12	85.17	27.03	97.62
T12LS	SBC	24	63.83	29.15	36	77.72	29.37	82.13
	SBL	28	82.71	27.57	21	90.19	21.94	91.71
	AS	13	81.92	27.12	12	80.92	23.96	101.24
T12RS	SBC	24	53.50	29.47	36	72.67	28.83	73.62
	SBL	28	81.50	26.80	21	85.67	24.19	95.14
	AS	13	82.15	24.51	12	81.75	30.63	100.49
T12LI	SBC	25	30.88	7.17	36	36.53	13.62	84.54
	SBL	29	38.90	16.41	21	41.24	20.31	94.32
	AS	13	45.46	33.94	11	44.27	31.04	102.69
T12RI	SBC	25	31.88	8.01	36	33.61	15.22	94.85
	SBL	29	37.55	18.42	21	37.71	14.28	99.57
	AS	12	40.25	32.85	12	29.33	10.42	137.22
L1LS	SBC	25	31.08	10.23	36	37.53	17.57	82.82
	SBL	29	37.62	16.80	20	37.80	19.79	99.53
	AS	16	35.63	29.41	15	31.40	12.03	113.46
L1RS	SBC	25	27.92	7.34	36	33.03	10.42	84.53
	SBL	28	35.96	16.88	20	32.75	11.58	109.81
	AS	14	23.64	6.63	15	31.13	15.02	75.94
L1LI	SBC	25	28.76	6.77	36	26.83	6.38	107.18
	SBL	29	29.86	6.27	20	31.65	10.02	94.35
	AS	15	24.07	9.54	16	29.56	8.52	81.41

Facet Area Sample Female SDI Male Std dev. Std dev. Ν Mean Ν Mean L1RI SBC 25 28.04 35 28.54 98.24 6.13 7.17 SBL 28 28.86 20 31.40 91.90 4.32 8.16 AS 16 25.88 13.81 15 30.33 9.24 85.30 L2LS SBC 26 9.83 37 30.73 9.81 99.25 30.50 SBL 30 20 28.70 4.01 101.97 29.27 7.30 AS 17 8.45 13 25.77 11.66 89.71 23.12 L2RS SBC 26 30.12 10.01 37 28.92 8.25 104.14 6.07 SBL 28 27.54 6.72 19 28.63 96.17 AS 17 23.88 12.01 13 29.31 13.14 81.49 L2LI SBC 26 29.50 8.42 37 30.68 8.02 96.17 SBL 30 32.07 7.03 19 34.05 8.32 94.17 28.25 34.09 AS 16 6.12 11 9.62 82.87 L2RI SBC 30.46 31.84 95.68 26 7.57 37 8.57 SBL 30 19 32.00 102.19 32.70 7.12 7.33 AS 16 26.94 8.61 12 29.75 8.17 90.55 L3LS SBC 26 9.33 36 8.39 102.06 33.42 32.75 SBL 27 20 102.44 34.11 7.13 33.30 9.26 AS 16 24.38 7.85 16 29.50 8.38 82.63 L3RS SBC 26 30.89 8.51 104.34 32.23 7.37 36 SBL 27 9.38 20 32.50 7.42 102.56 33.33 AS 16 26.06 9.43 16 31.75 7.03 82.09 L3LI SBC 26 38.73 8.66 36 39.58 10.96 97.85 27 SBL 40.00 8.20 21 41.19 12.16 97.11 AS 17 36.82 9.72 16 43.19 9.66 85.26 L3RI SBC 26 37.85 9.22 36 42.33 12.65 89.4 25 9.90 103.24 SBL 41.64 21 40.33 10.49 34.75 104.62 AS 16 8.25 14 33.21 7.46 104.37 L4LS SBC 26 41.31 10.49 38 39.58 11.93 SBL 24 42.92 8.38 16 44.38 5.75 96.71 AS 10.98 14 34.86 9.02 95.93 16 33.44

Facet Area	Sample	Male Female				SDI		
		N	Mean	Std dev.	N	Mean	Std dev.	
L4RS	SBC	26	38.62	8.41	38	36.76	9.68	105.04
	SBL	25	37.84	7.36	15	40.27	8.15	93.97
	AS	16	33.69	7.98	14	34.86	5.48	96.64
L4LI	SBC	26	46.58	9.77	38	46.97	11.73	99.16
	SBL	24	50.63	12.26	16	52.63	9.99	96.20
	AS	16	46.88	10.22	13	53.77	7.88	87.18
L4RI	SBC	26	49.88	10.52	38	48.63	11.99	102.58
	SBL	25	53.56	14.83	16	53.56	9.55	100.00
	AS	15	41.07	13.47	12	56.83	6.29	72.26
L5LS	SBC	24	51.83	8.33	34	48.71	11.29	106.42
	SBL	29	52.72	12.30	19	52.11	11.14	101.19
	AS	13	47.77	19.97	13	45.23	13.40	105.61
L5RS	SBC	24	47.88	9.39	34	44.35	11.67	107.94
	SBL	27	49.74	10.85	19	50.00	8.14	99.48
	AS	13	43.08	11.91	13	50.08	9.60	86.02
L5LI	SBC	24	51.54	9.50	34	50.35	9.74	102.36
	SBL	29	51.28	12.36	19	52.84	11.68	97.04
	AS	12	50.42	9.89	12	49.50	9.67	101.85
L5RI	SBC	24	53.17	11.04	34	51.35	8.96	103.53
	SBL	28	52.46	10.02	19	55.89	9.52	93.86
	AS	13	44.69	9.29	12	45.92	5.28	97.33

Appendix C-1 Descriptive statistics and SDI for facet angle continued

Key: C=Cervical, T=Thoracic, L=Lumbar, Number = Vertebral level, R=Right, L=Left, S=Superior, I = Inferior, Std Dev = Standard Deviation, SDI = Sexual dimorphism Index,

SBC SBL Facet Level AS Mann-Mann-Mannр р р Whitney U Whitney U Whitney U C1LS 361.00 0.90 248.50 0.79 61.50 0.55 C1RS 356.50 0.84 251.00 0.84 68.50 0.84 C1LI 314.50 0.36 230.50 0.51 27.00 0.02 C1RI 258.00 0.06 216.50 0.33 40.00 0.05 C2LS 322.00 0.20 287.00 0.95 30.00 0.03 321.00 C2RS 0.19 198.00 0.06 33.50 0.04 C2LI 335.00 0.41 217.00 0.14 18.00 0.01 C2RI 307.50 0.13 206.00 0.09 33.00 0.05 C3LS 227.00 0.13 258.50 0.47 51.50 0.13 C3RS 264.50 0.30 216.00 0.18 34.00 0.02 C3LI 296.50 0.79 167.00 0.02 36.00 0.02 280.00 0.56 150.00 0.01 <0.0001 C3RI 13.50 C4LS 173.50 0.04 296.50 0.67 57.50 0.46 C4RS 229.00 0.28 283.50 0.50 54.50 0.36 C4LI 287.50 0.96 320.00 0.80 33.50 0.05 C4RI 272.00 0.71 280.50 0.46 64.50 0.98 C5LS 342.50 0.43 346.50 0.81 44.00 0.44 C5RS 306.50 0.11 349.50 0.85 0.02 21.50 C55LI 401.00 0.91 338.00 0.86 41.50 0.34 C5RIA 405.00 0.96 304.50 0.44 47.00 0.57 C6LS 349.50 0.53 312.00 0.55 45.50 0.55 C6RS 235.50 0.01 324.50 0.67 39.50 0.30 C6LI 388.00 0.99 242.50 0.04 36.00 0.20 C6RI 365.50 0.71 312.00 0.30 35.00 0.18 C7LS 454.50 0.65 349.00 0.56 28.00 0.61 C7RS 0.86 382.00 0.97 0.76 462.50 30.00 404.50 0.25 357.50 0.66 0.96 C7LI 35.50 C7RI 482.00 0.94 298.50 0.15 30.00 0.57

Appendix C-2 Results of Mann-Whitney u test for sex differences in facet angle

Facet Level	SBC		SBL		AS	
	Mann- Whitney U	р	Mann- Whitney U	р	Mann- Whitney U	р
T1LS	380.00	0.54	347.00	0.47	68.50	0.29
T1RS	446.00	0.64	344.50	0.45	66.50	0.37
T1LI	447.00	0.65	314.00	0.16	77.50	0.74
T1RI	466.00	0.85	332.50	0.26	66.50	0.37
T2LS	321.50	0.02	302.50	0.23	77.00	0.50
T2RS	261.50	<0.000	368.00	0.78	66.50	0.23
T2LI	437.50	0.44	275.00	0.09	80.00	0.59
T2RI	440.50	0.46	342.50	0.60	90.00	0.96
T3LS	424.50	0.38	280.00	0.77	103.50	0.95
T3RS	468.00	0.79	288.50	0.75	92.00	0.40
T3LI	435.00	0.47	297.00	0.88	94.50	0.89
T3RI	448.50	0.59	302.50	0.97	98.00	0.56
T4LS	442.00	0.58	229.50	0.25	105.00	1.00
T4RS	414.00	0.44	244.50	0.51	53.50	0.02
T4LI	313.50	0.03	233.50	0.54	78.50	0.26
T4RI	356.00	0.11	208.00	0.18	85.50	0.42
T5LS	423.00	0.57	251.50	0.54	72.50	0.69
T5RS	416.50	0.51	277.00	0.95	54.50	0.18
T5LI	445.00	0.94	247.00	0.77	60.00	0.56
T5RI	441.50	0.90	237.00	0.37	68.50	0.93
T6LS	440.00	0.82	254.00	0.18	67.50	0.88
T6RS	361.00	0.17	324.50	0.99	66.00	0.81
T6LI	360.50	0.09	303.00	0.98	58.00	0.89
T6RI	433.00	0.50	289.50	0.77	53.50	0.67
T7LS	472.00	0.76	260.00	0.33	83.00	0.94
T7RS	394.50	0.17	297.50	0.90	58.50	0.18
T7LI	466.50	0.90	272.00	0.58	55.50	0.52
T7RI	463.00	0.87	269.00	0.43	47.00	0.15

Appendix C-2 Results of Mann-Whitney u test for sex differences in facet angle continued

Facet Level	SBC		SBL		AS	
	Mann- Whitney U	р	Mann- Whitney U	р	Mann- Whitney U	р
T8LS	400.50	0.47	256.00	0.22	76.00	0.47
T8RS	395.50	0.42	200.00	0.02	87.00	0.85
T8LI	465.50	0.97	167.00	0.01	82.50	0.68
T8RI	453.00	0.83	284.00	0.70	88.00	0.88
T9LS	487.50	0.79	303.50	0.90	73.50	0.39
T9RS	471.00	0.63	211.50	0.05	34.00	0.01
T9LI	477.00	0.88	233.00	0.14	68.50	0.42
T9RI	441.00	0.52	242.00	0.25	60.00	0.22
T10LS	411.50	0.29	285.00	0.76	70.00	0.33
T10RS	370.00	0.10	223.50	0.09	76.00	0.68
T10LI	506.50	0.99	229.00	0.16	84.00	0.77
T10RI	480.50	0.72	241.50	0.24	78.50	0.57
T11LS	470.00	0.62	268.50	0.48	82.00	0.69
T11RS	422.50	0.26	284.50	0.69	65.50	0.23
T11LI	373.00	0.11	218.50	0.13	69.50	0.45
T11RI	326.50	0.03	262.00	0.40	76.50	0.70
T12LS	313.00	0.07	249.00	0.36	64.00	0.44
T12RS	294.50	0.04	260.50	0.50	66.00	0.51
T12LI	319.00	0.05	301.50	0.95	65.00	0.71
T12RI	414.00	0.60	276.00	0.57	71.50	0.98
L1LS	351.00	0.15	259.00	0.53	101.50	0.46
L1RS	297.00	0.02	246.00	0.48	67.00	0.10
L1LI	389.50	0.37	289.50	0.99	83.00	0.14
L1RI	413.50	0.72	202.50	0.10	78.00	0.10
L2LS	471.00	0.89	280.00	0.69	98.50	0.61
L2RS	464.50	0.82	228.00	0.41	81.00	0.22
L2LI	443.00	0.59	235.50	0.31	53.00	0.08
L2RI	449.50	0.66	259.00	0.59	81.50	0.50

Appendix C-2 Results of Mann-Whitney u test for sex differences in facet angle continued

Facet Level	SBC		SBL		AS	
	Mann- Whitney U	р	Mann- Whitney U	р	Mann- Whitney U	р
L3LS	449.50	0.79	252.00	0.70	78.50	0.06
L3RS	395.00	0.30	260.00	0.83	82.50	0.09
L3LI	458.50	0.89	271.00	0.79	86.50	0.07
L3RI	364.50	0.14	244.00	0.68	106.00	0.80
L4LS	452.50	0.57	172.50	0.59	106.50	0.82
L4RS	433.00	0.40	150.00	0.29	108.50	0.88
L4LI	476.00	0.81	169.50	0.53	61.00	0.06
L4RI	468.00	0.72	199.00	0.98	22.50	0.00
L5LS	337.00	0.26	272.00	0.94	73.50	0.57
L5RS	343.50	0.31	233.00	0.60	53.50	0.11
L5LI	397.50	0.87	259.50	0.74	69.00	0.86
L5RI	382.00	0.68	207.50	0.20	66.50	0.53

Appendix C-2 Results of Mann-Whitney u test for sex differences in facet angle continued

Key: C=Cervical, T=Thoracic, L=Lumbar, Number = Vertebral level, R=Right, L=Left, S=Superior, I = Inferior.

Facet	Sample	Male	Size	Female Size	
		Chi-Square	P Value	Chi-Square	P Value
C1LS	SBC	1.60	0.45	0.76	0.68
	SBL	3.57	0.17	0.12	0.94
	AS	2.03	0.36	0.38	0.54
C1RS	SBC	2.63	0.27	0.34	0.84
	SBL	1.23	0.54	3.85	0.15
	AS	1.13	0.57	0.24	0.62
C1L!	SBC	2.57	0.28	0.65	0.72
	SBL	3.41	0.18	0.30	0.86
	AS	2.12	0.35	0.06	0.81
C1RI	SBC	2.65	0.27	0.06	0.97
	SBL	2.76	0.25	1.43	0.49
	AS	2.16	0.34	0.72	0.40
C2LS	SBC	1.14	0.57	1.68	0.43
	SBL	0.28	0.87	2.62	0.27
	AS	2.24	0.33	2.19	0.33
C2RS	SBC	2.69	0.26	1.31	0.52
	SBL	0.01	0.99	1.38	0.50
	AS	1.48	0.48	1.68	0.43
C2LI	SBC	3.39	0.18	1.39	0.50
	SBL	1.61	0.45	1.77	0.41
	AS	4.73	0.09	3.96	0.14
C2RI	SBC	4.31	0.12	1.02	0.60
	SBL	0.73	0.69	3.50	0.17
	AS	0.02	0.99	3.99	0.14
C3LS	SBC	0.53	0.77	4.41	0.11
	SBL	3.40	0.18	3.99	0.14
	AS	0.97	0.62	0.10	0.95

Appendix C-3 Kruskal-Wallace test for differences in facet angle with age group

Annondix C 2 Kruckal Wallaco tost fo	r difforoncos in facol	t angle with age group	a continued
Appendix C-5 Kluskal-Wallace lest lu	i uniterences in lace	t angle with age giou	p continueu

Facet	Sample	Male Size		Female Size	
		Chi-Square	P Value	Chi-Square	P Value
C3RS	SBC	0.43	0.81	2.86	0.24
	SBL	3.31	0.19	6.19	0.05
	AS	1.28	0.53	0.98	0.61
C3LI	SBC	2.15	0.34	5.79	0.06
	SBL	0.29	0.87	1.62	0.44
	AS	0.99	0.61	2.68	0.26
C3RI	SBC	5.01	0.08	1.19	0.55
	SBL	1.35	0.51	5.70	0.06
	AS	2.33	0.31	0.93	0.63
C4LS	SBC	0.97	0.62	0.94	0.63
	SBL	0.81	0.67	0.89	0.64
	AS	0.80	0.67	0.44	0.80
C4RS	SBC	0.34	0.84	5.26	0.07
	SBL	1.17	0.56	2.13	0.34
	AS	1.60	0.45	0.61	0.74
C4LI	SBC	6.32	0.04	5.46	0.07
	SBL	3.41	0.18	1.36	0.51
	AS	2.37	0.31	0.90	0.64
C4RI	SBC	4.28	0.12	8.82	0.01
	SBL	0.46	0.80	1.09	0.58
	AS	0.30	0.86	3.95	0.14
C5LS	SBC	3.33	0.19	2.37	0.31
	SBL	4.67	0.10	1.32	0.52
	AS	2.37	0.31	0.54	0.76
C5RS	SBC	2.08	0.35	3.23	0.20
	SBL	5.84	0.05	1.05	0.59
	AS	1.24	0.54	1.35	0.51
C55LI	SBC	4.12	0.13	6.16	0.05
	SBL	1.20	0.55	1.35	0.51
	AS	1.09	0.58	6.47	0.04

Facet	Sample	Male	Size	Female Size	
		Chi-Square	P Value	Chi-Square	P Value
C5RI	SBC	5.97	0.05	2.67	0.26
	SBL	3.94	0.14	1.84	0.40
	AS	2.69	0.26	2.29	0.32
C6LS	SBC	2.15	0.34	2.36	0.31
	SBL	2.06	0.36	0.31	0.86
	AS	0.02	0.99	1.50	0.47
C6RS	SBC	1.66	0.44	3.28	0.19
	SBL	1.44	0.49	0.55	0.76
	AS	1.97	0.37	4.80	0.09
C6LI	SBC	1.64	0.44	1.19	0.55
	SBL	4.41	0.11	0.48	0.79
	AS	0.77	0.68	0.76	0.69
C6RI	SBC	4.04	0.13	3.59	0.17
	SBL	2.27	0.32	1.19	0.55
	AS	1.27	0.53	0.38	0.83
C7LS	SBC	2.69	0.26	0.58	0.75
	SBL	0.50	0.78	0.51	0.77
	AS	0.68	0.71	0.21	0.64
C7RS	SBC	3.10	0.21	0.35	0.84
	SBL	0.26	0.88	0.74	0.69
	AS	1.09	0.58	0.00	1.00
C7LI	SBC	2.04	0.36	4.09	0.13
	SBL	0.14	0.93	0.22	0.90
	AS	2.24	0.33	3.43	0.06
C7RI	SBC	0.46	0.80	8.45	0.01
	SBL	4.66	0.10	6.41	0.04
	AS	2.10	0.35	0.21	0.64
T1LS	SBC	0.45	0.80	0.29	0.86
	SBL	4.67	0.10	0.36	0.83
	AS	3.63	0.16	0.03	0.86

Appendix C-3 Kruskal-Wallace test for differences in facet angle with age group continued

Annondix C 2 Kruckal Wallaco tost fo	r difforoncos in facol	t angle with age group	a continued
Appendix C-5 Kluskal-Wallace lest lu	i uniterences in lace	t angle with age giou	p continueu

Facet	Sample	Male	e Size	Fema	e Size
		Chi-Square	P Value	Chi-Square	P Value
T1RS	SBC	0.72	0.70	0.45	0.80
	SBL	3.91	0.14	5.55	0.06
	AS	2.76	0.25	2.18	0.34
T1LI	SBC	0.82	0.66	1.60	0.45
	SBL	4.29	0.12	0.33	0.85
	AS	0.29	0.87	1.71	0.43
T1RI	SBC	2.22	0.33	1.63	0.44
	SBL	0.97	0.61	0.24	0.89
	AS	2.26	0.32	0.17	0.92
T2LS	SBC	0.42	0.81	6.06	0.05
	SBL	4.41	0.11	1.61	0.45
	AS	3.39	0.18	0.64	0.73
T2RS	SBC	0.42	0.81	6.06	0.05
	SBL	3.34	0.19	2.10	0.35
	AS	4.11	0.13	3.63	0.16
T2LI	SBC	0.73	0.69	2.33	0.31
	SBL	3.64	0.16	1.20	0.55
	AS	0.08	0.96	1.92	0.38
T2RI	SBC	1.02	0.60	0.52	0.77
	SBL	2.16	0.34	1.38	0.50
	AS	2.75	0.25	0.14	0.93
T3LS	SBC	1.93	0.38	0.61	0.74
	SBL	2.70	0.26	0.24	0.89
	AS	2.13	0.34	0.61	0.74
T3RS	SBC	2.80	0.25	0.32	0.85
	SBL	0.24	0.89	1.77	0.41
	AS	0.90	0.64	1.83	0.40
T3LI	SBC	1.62	0.45	1.58	0.45
	SBL	1.42	0.49	1.49	0.47
	AS	1.77	0.41	0.03	0.99

Facet	Sample	Male	Size	Female Size	
		Chi-Square	P Value	Chi-Square	P Value
T3RI	SBC	0.66	0.72	3.96	0.14
	SBL	0.55	0.76	0.23	0.89
	AS	0.12	0.94	1.57	0.46
T4LS	SBC	0.43	0.80	2.97	0.23
	SBL	1.39	0.50	2.10	0.35
	AS	1.28	0.53	1.31	0.52
T4RS	SBC	2.59	0.27	0.45	0.80
	SBL	1.08	0.58	0.69	0.71
	AS	0.97	0.62	1.87	0.39
T4LI	SBC	1.84	0.40	2.40	0.30
	SBL	0.09	0.95	4.51	0.11
	AS	0.30	0.86	0.31	0.86
T4RI	SBC	1.83	0.40	0.12	0.94
	SBL	2.72	0.26	0.09	0.96
	AS	3.75	0.15	1.52	0.47
T5LS	SBC	5.68	0.06	0.03	0.98
	SBL	4.14	0.13	1.48	0.48
	AS	1.14	0.57	1.17	0.56
T5RS	SBC	0.20	0.90	0.30	0.86
	SBL	0.14	0.93	2.26	0.32
	AS	2.88	0.24	0.59	0.75
T5LI	SBC	4.92	0.09	4.14	0.13
	SBL	1.17	0.56	1.47	0.48
	AS	2.02	0.36	0.20	0.91
T5RI	SBC	2.77	0.25	0.64	0.72
	SBL	1.47	0.48	0.20	0.91
	AS	2.33	0.31	4.08	0.13
T6LS	SBC	4.92	0.09	0.55	0.76
	SBL	1.19	0.55	1.39	0.50
	AS	1.08	0.58	0.05	0.83

Appendix C-3 Kruskal-Wallace test for differences in facet angle with age group continued

Annondix C 2 Kruckal Wallaco tost fo	r difforoncos in facol	t angle with age group	a continued
Appendix C-5 Kluskal-Wallace lest lu	i uniterences in lace	t angle with age giou	p continueu

Facet	Sample	Male Size		Female Size	
		Chi-Square	P Value	Chi-Square	P Value
T6RS	SBC	0.06	0.97	3.65	0.16
	SBL	1.97	0.37	1.21	0.55
	AS	1.92	0.38	1.14	0.29
T6LI	SBC	5.38	0.07	2.85	0.24
	SBL	3.71	0.16	5.07	0.08
	AS	1.48	0.48	0.10	0.75
TGRI	SBC	2.37	0.31	1.14	0.57
	SBL	0.20	0.90	1.63	0.44
	AS	0.11	0.95	1.15	0.28
T7LS	SBC	0.86	0.65	0.32	0.85
	SBL	1.70	0.43	6.64	0.04
	AS	0.66	0.72	0.00	1.00
T7RS	SBC	1.98	0.37	1.94	0.38
	SBL	2.08	0.35	1.60	0.45
	AS	0.66	0.72	0.14	0.71
T7LI	SBC	3.86	0.14	2.01	0.37
	SBL	3.81	0.15	0.26	0.88
	AS	1.20	0.55	0.55	0.46
T7RI	SBC	3.44	0.18	2.13	0.34
	SBL	0.77	0.68	0.37	0.83
	AS	2.12	0.35	0.05	0.82
T8LS	SBC	1.06	0.59	0.51	0.77
	SBL	3.35	0.19	0.14	0.93
	AS	1.31	0.52	4.00	0.14
T8RS	SBC	0.04	0.98	0.07	0.97
	SBL	2.87	0.24	0.51	0.78
	AS	1.59	0.45	3.38	0.18
T8LI	SBC	3.37	0.18	0.87	0.65
	SBL	2.23	0.33	0.87	0.65
	AS	2.21	0.33	2.75	0.25

Facet	Sample	Male Size		Female Size			
		Chi-Square	P Value	Chi-Square	P Value		
T8RI	SBC	0.83	0.66	2.82	0.24		
	SBL	0.40	0.82	0.68	0.71		
	AS	0.69	0.71	3.24	0.20		
T9LS	SBC	3.11	0.21	1.07	0.59		
	SBL	1.80	0.41	1.03	0.60		
	AS	0.53	0.77	0.09	0.96		
T9RS	SBC	1.59	0.45	0.66	0.72		
	SBL	5.41	0.07	4.64	0.10		
	AS	0.10	0.95	1.55	0.46		
T9LI	SBC	0.51	0.77	2.98	0.22		
	SBL	0.42	0.81	0.86	0.65		
	AS	2.57	0.28	0.00	1.00		
T9RI	SBC	0.95	0.62	1.63	0.44		
	SBL	9.00	0.01	6.99	0.03		
	AS	3.15	0.21	1.24	0.27		
T10LS	SBC	1.76	0.42	0.71	0.70		
	SBL	0.93	0.63	1.88	0.39		
	AS	0.35	0.84	0.03	0.86		
T10RS	SBC	0.33	0.85	3.78	0.15		
	SBL	1.85	0.40	0.59	0.74		
	AS	0.46	0.80	0.01	0.93		
T10LI	SBC	1.15	0.56	0.65	0.72		
	SBL	0.18	0.91	0.28	0.87		
	AS	2.56	0.28	0.12	0.73		
T10RI	SBC	1.85	0.40	0.17	0.92		
	SBL	2.50	0.29	1.41	0.50		
	AS	1.91	0.38	0.46	0.50		
T11LS	SBC	2.19	0.33	2.27	0.32		
	SBL	2.51	0.29	0.25	0.88		
	AS	0.69	0.41	0.12	0.73		

Appendix C-3 Kruskal-Wallace test for differences in facet angle with age group continued

Annondix C 2 Kruckal Wallaco tost fo	r difforoncos in faco	t angle with age grou	n continued
Appendix C-5 Kluskal-Wallace lest to	i uniterences in face	it allgle with age giou	p continueu

Facet	cet Sample Male Size		e Size	Female Size			
		Chi-Square	P Value	Chi-Square	P Value		
T11RS	SBC	1.62	0.45	0.14	0.93		
	SBL	2.67	0.26	1.07	0.59		
	AS	0.06	0.81	0.03	0.86		
T11LI	SBC	0.80	0.67	5.45	0.07		
	SBL	0.66	0.72	0.27	0.87		
	AS	0.10	0.75	1.05	0.30		
T11RI	SBC	0.37	0.83	2.35	0.31		
	SBL	0.48	0.79	0.71	0.70		
	AS	0.27	0.61	0.12	0.73		
T12LS	SBC	2.16	0.34	0.33	0.85		
	SBL	0.39	0.82	0.39	0.82		
	AS	2.48	0.12	2.93	0.09		
T12RS	SBC	0.97	0.62	3.37	0.19		
	SBL	0.24	0.89	0.41	0.82		
	AS	0.02	0.89	1.05	0.30		
T12LI	SBC	0.34	0.84	1.22	0.54		
	SBL	0.10	0.95	0.39	0.82		
	AS	0.08	0.78	0.04	0.85		
T12RI	SBC	1.40	0.50	0.31	0.86		
	SBL	0.57	0.75	0.85	0.65		
	AS	0.01	0.94	3.86	0.05		
L1LS	SBC	2.82	0.24	1.77	0.41		
	SBL	1.35	0.51	1.61	0.45		
	AS	0.10	0.75	0.78	0.68		
L1RS	SBC	0.26	0.88	1.68	0.43		
	SBL	0.65	0.72	0.23	0.89		
	AS	0.15	0.70	1.65	0.44		
L1LI	SBC	1.04	0.60	6.35	0.04		
	SBL	0.03	0.98	3.78	0.15		
	AS	5.06	0.02	5.29	0.07		

Facet	Sample	Male	Size	Female Size			
		Chi-Square	P Value	Chi-Square	P Value		
L1RI	SBC	2.59	0.27	13.84	<0.01		
	SBL	0.59	0.74	2.59	0.27		
	AS	1.02	0.31	0.49	0.78		
L2LS	SBC	3.70	0.16	1.25	0.53		
	SBL	0.41	0.81	0.86	0.65		
	AS	2.16	0.14	0.62	0.73		
L2RS	SBC	6.65	0.04	2.52	0.28		
	SBL	0.22	0.90	0.73	0.69		
	AS	3.30	0.07	1.09	0.58		
L2LI	SBC	3.67	0.16	2.28	0.32		
	SBL	1.64	0.44	1.93	0.38		
	AS	1.57	0.21	1.78	0.41		
L2RI	SBC	1.74	0.42	1.28	0.53		
	SBL	1.01	0.60	1.84	0.40		
	AS	0.96	0.33	2.21	0.33		
L3LS	SBC	2.55	0.28	1.01	0.60		
	SBL	0.16	0.92	0.44	0.80		
	AS	0.84	0.36	1.50	0.47		
L3RS	SBC	1.03	0.60	0.42	0.81		
	SBL	1.57	0.46	1.05	0.59		
	AS	0.05	0.83	0.96	0.62		
L3LI	SBC	1.42	0.49	5.09	0.08		
	SBL	1.48	0.48	0.60	0.74		
	AS	3.65	0.06	0.30	0.86		
L3RI	SBC	2.64	0.27	1.26	0.53		
	SBL	3.39	0.18	0.82	0.66		
	AS	3.45	0.06	0.01	0.94		
L4LS	SBC	0.15	0.93	0.38	0.83		
	SBL	0.28	0.87	1.84	0.40		
	AS	0.76	0.38	0.01	0.94		

Appendix C-3 Kruskal-Wallace test for differences in facet angle with age group continued

Facet	Sample	Male	e Size	Female Size			
		Chi-Square	P Value	Chi-Square	P Value		
L4RS	SBC	0.83	0.68	2.33	0.31		
	SBL	1.80	0.41	0.56	0.76		
	AS	0.05	0.83	0.01	0.94		
L4LI	SBC	2.34	0.31	4.06	0.13		
	SBL	0.13	0.94	0.39	0.82		
	AS	1.57	0.21	0.60	0.44		
L4RI	SBC	0.72	0.70	4.25	0.12		
	SBL	0.00	1.00	0.31	0.86		
	AS	0.50	0.48	0.01	0.93		
L5LS	SBC	0.49	0.78	1.47	0.48		
	SBL	0.26	0.88	0.74	0.69		
	AS	0.86	0.35	0.03	0.87		
L5RS	SBC	2.40	0.30	4.27	0.12		
	SBL	1.08	0.58	1.31	0.52		
	AS	0.62	0.43	0.15	0.70		
L5LI	SBC	0.95	0.62	1.43	0.49		
	SBL	2.17	0.34	0.60	0.74		
	AS	1.91	0.17	0.03	0.86		
L5RI	SBC	1.47	0.48	2.05	0.36		
	SBL	2.48	0.29	1.35	0.51		
	AS	1.33	0.25	2.36	0.12		

Appendix C-3 Kruskal-Wallace test for differences in facet angle with age group continued

Key: C=Cervical, T=Thoracic, L=Lumbar, Number = Vertebral level, R=Right, L=Left, S=Superior, I = Inferior,

Appendix C-4 Correlation between facet angle and actual age

		Males				Females		
Facet	C-Coefficient	Р	Ν	Strength of Correlation	C-Coefficient	Ρ	N	Strength of Correlation
C1LS	-0.34	0.11	23	Medium	-0.04	0.84	32	None
C1RS	-0.26	0.23	23	Weak	-0.11	0.56	32	Weak
C1LI	-0.13	0.57	23	Weak	0.07	0.72	32	None
C1RI	-0.02	0.92	23	None	0.08	0.68	32	Weak
C2LS	-0.18	0.42	23	Weak	0.18	0.31	35	Weak
C2RS	-0.28	0.19	23	Weak	0.11	0.52	35	Weak
C2LI	-0.25	0.26	22	Weak	-0.09	0.62	35	None
C2RI	-0.29	0.18	23	Weak	-0.09	0.62	35	None
C3LS	0.00	1.00	19	None	0.32	0.07	32	Medium
C3RS	0.15	0.52	20	Weak	0.00	0.99	32	None
C3LI	-0.23	0.34	20	Weak	-0.15	0.41	31	Weak
C3RI	-0.42	0.06	20	Medium	-0.14	0.45	31	Weak
C4LS	-0.14	0.57	19	Weak	0.06	0.75	28	None
C4RS	0.27	0.25	20	Weak	-0.26	0.18	28	Weak
C4LI	-0.39	0.09	20	Medium	-0.29	0.12	29	Weak
C4RI	-0.28	0.23	20	Weak	-0.39	0.04	29	Medium
C5LS	-0.25	0.25	23	Weak	-0.12	0.49	34	Weak
C5RS	-0.20	0.34	24	Weak	-0.40	0.02	34	Medium
C5LI	-0.22	0.30	24	Weak	-0.39	0.02	34	Medium
C5RI	-0.41	0.05	24	Medium	-0.25	0.15	34	Weak
C6LS	-0.08	0.73	21	None	-0.17	0.31	37	Weak
C6RS	-0.22	0.34	21	Weak	-0.17	0.30	37	Weak
C6LI	-0.18	0.44	21	Weak	-0.16	0.33	37	Weak
C6RI	-0.35	0.12	21	Weak	-0.19	0.27	37	Weak
C7LS	-0.19	0.36	25	Weak	-0.02	0.92	39	None
C7RS	-0.13	0.53	25	Weak	-0.08	0.65	38	None
C7LI	0.16	0.45	25	Weak	-0.13	0.43	39	Weak
C7RI	0.03	0.90	25	None	-0.32	0.04	39	Medium
T1LS	-0.16	0.49	21	Weak	0.07	0.66	40	None
T1RS	0.14	0.53	24	Weak	-0.08	0.64	40	None
T1LI	-0.26	0.22	24	Weak	-0.13	0.44	40	Weak
T1RI	-0.02	0.94	24	None	-0.02	0.93	40	None
T2LS	0.16	0.45	26	Weak	0.19	0.24	38	Weak
T2RS	-0.02	0.91	26	None	0.05	0.76	38	None
T2LI	-0.25	0.21	26	Weak	-0.24	0.15	38	Weak
T2RI	-0.08	0.70	26	None	0.06	0.71	38	None
T3LS	0.32	0.12	25	Medium	0.15	0.37	39	Weak
T3RS	0.33	0.11	25	Medium	0.01	0.96	39	None
T3LI	-0.26	0.22	25	Weak	-0.21	0.21	39	Weak
T3RI	0.00	0.99	25	None	-0.34	0.03	39	Medium

Appendix C-4 Correlation between facet angle and actual age continued

		Males				Females		
Facet	C-Coefficient	Р	Ν	Strength of Correlation	C-Coefficient	Р	Ν	Strength of Correlation
T4LS	0.00	0.99	26	None	-0.11	0.50	37	Weak
T4RS	-0.27	0.19	26	Weak	0.03	0.88	36	None
T4LI	-0.27	0.18	26	Weak	-0.26	0.13	36	Weak
T4RI	-0.19	0.36	26	Weak	-0.01	0.98	36	None
T5LS	-0.19	0.36	25	Weak	0.07	0.66	37	None
T5RS	-0.03	0.88	25	None	-0.07	0.69	37	None
T5LI	-0.27	0.19	25	Weak	-0.34	0.04	36	Medium
T5RI	-0.20	0.33	25	Weak	-0.14	0.41	36	Weak
T6LS	0.39	0.05	26	Medium	0.00	0.99	35	None
T6RS	-0.08	0.71	26	None	0.28	0.11	35	Weak
T6LI	-0.27	0.18	26	Weak	-0.24	0.15	37	Weak
T6RI	-0.04	0.85	26	None	0.20	0.23	37	Weak
T7LS	0.06	0.78	26	Weak	0.04	0.81	38	None
T7RS	-0.21	0.30	26	Weak	0.23	0.17	38	Weak
T7LI	-0.46	0.02	25	Medium	-0.14	0.41	38	Weak
T7RI	-0.08	0.72	25	None	-0.17	0.31	38	Weak
T8LS	0.32	0.12	25	Medium	0.14	0.42	36	Weak
T8RS	-0.05	0.81	25	None	0.05	0.79	36	None
T8LI	-0.42	0.03	26	Medium	-0.12	0.49	36	Weak
T8RI	-0.13	0.51	26	Weak	-0.17	0.32	36	Weak
T9LS	0.11	0.60	26	Weak	0.16	0.33	39	Weak
T9RS	0.05	0.79	26	None	0.10	0.56	39	Weak
T9LI	-0.26	0.20	25	Weak	-0.26	0.11	39	Weak
T9RI	-0.12	0.57	25	Weak	-0.21	0.19	39	Weak
T10LS	0.33	0.11	25	Medium	0.09	0.57	39	None
T10RS	-0.18	0.38	25	Weak	-0.02	0.90	39	None
T10LI	-0.40	0.04	26	Medium	-0.16	0.34	39	Weak
T10RI	-0.38	0.05	26	Medium	0.02	0.92	39	None
T11LS	-0.01	0.98	26	None	-0.10	0.55	39	Weak
T11RS	-0.07	0.73	26	None	0.00	0.99	39	None
T11LI	-0.09	0.67	25	None	0.04	0.80	39	None
T11RI	-0.03	0.88	25	None	0.11	0.50	39	Weak
T12LS	0.02	0.92	24	None	0.18	0.29	36	Weak
T12RS	-0.03	0.89	24	None	0.30	0.07	36	Medium
T12LI	-0.01	0.95	25	None	0.20	0.24	36	Weak
T12RI	-0.20	0.33	25	Weak	-0.09	0.61	36	None
L1LS	0.18	0.40	25	Weak	0.05	0.76	36	None
L1RS	-0.04	0.84	25	None	0.05	0.77	36	None
L1LI	0.16	0.44	25	Weak	0.13	0.44	36	Weak
L1RI	0.38	0.06	25	Medium	-0.03	0.87	35	None

Facat	Males			Females				
Facet	C-Coefficient	Р	Ν	Strength of Correlation	C-Coefficient	Р	Ν	Strength of Correlation
L2LS	0.36	0.07	26	Medium	-0.09	0.58	37	None
L2RS	0.45	0.02	26	Medium	-0.15	0.37	37	Weak
L2LI	0.43	0.03	26	Medium	0.08	0.63	37	None
L2RI	0.23	0.25	26	Weak	0.05	0.79	37	None
L3LS	0.39	0.05	26	Medium	0.07	0.70	36	None
L3RS	0.21	0.31	26	Weak	-0.01	0.96	36	None
L3LI	-0.18	0.38	26	Weak	-0.11	0.51	36	Weak
L3RI	0.04	0.83	26	None	0.05	0.78	36	None
L4LS	0.15	0.48	26	Weak	-0.01	0.97	38	None
L4RS	-0.05	0.82	26	None	0.07	0.66	38	None
L4LI	-0.13	0.53	26	Weak	0.16	0.35	38	Weak
L4RI	-0.06	0.76	26	None	0.16	0.33	38	Weak
L5LS	0.11	0.60	24	Weak	0.13	0.47	34	Weak
L5RS	-0.15	0.47	24	Weak	0.02	0.92	34	None
L5LI	-0.13	0.55	24	Weak	-0.13	0.55	24	Weak
L5RI	0.02	0.92	24	None	-0.18	0.31	34	Weak

Appendix C-4 Correlation between facet angle and actual age continued

Appendix C-5 Kruskal-Wallace test for inter-sample facet angle difference

French	Male		Female			
Facet	Chi-Square	р	Chi-Square	р		
C1LS	7.247	0.027	10.169	0.006		
C1RS	9.701	0.008	7.65	0.022		
C1LI	1.748	0.417	3.761	0.153		
C1RI	1.056	0.59	2.311	0.315		
C2LS	8.278	0.016	1.376	0.503		
C2RS	4.049	0.132	3.473	0.176		
C2LI	21.119	<0.0001	7.232	0.027		
C2RI	21.025	<0.0001	7.54	0.023		
C3LS	14.801	0.001	12.455	0.002		
C3RS	7.349	0.025	14.433	0.001		
C3LI	21.926	<0.0001	0.158	0.924		
C3RI	39.371	<0.0001	4.816	0.09		
C4LS	9.879	0.007	2.648	0.266		
C4RS	0.471	0.79	3.114	0.211		
C4LI	15.146	0.001	6.027	0.049		
C4RI	14.689	0.001	17.88	<0.0001		
C5LS	3.116	0.211	1.501	0.472		
C5RS	9.439	0.009	3.714	0.156		
C5LI	6.129	0.047	8.348	0.015		
C5RI	13.473	0.001	12.833	0.002		
C6LS	6.094	0.048	2.223	0.329		
C6RS	8.027	0.018	0.977	0.613		
C6LI	6.33	0.042	3.057	0.217		
C6RI	6.77	0.034	7.852	0.02		
C7LS	6.233	0.044	6.525	0.038		
C7RS	1.394	0.498	1.326	0.515		
C7LI	5.102	0.078	2.647	0.266		
C7RI	7.39	0.025	1.249	0.535		
T1LS	2.813	0.245	0.671	0.715		
T1RS	1.39	0.499	0.519	0.771		
T1LI	24.764	<0.0001	19.042	<0.0001		
T1RI	37.397	<0.0001	11.415	0.003		
T2LS	3.243	0.198	3.263	0.196		
T2RS	5.176	0.075	18.084	<0.0001		
T2LI	24.726	<0.0001	23.962	<0.0001		
T2RI	23.168	<0.0001	22.568	<0.0001		
T3LS	1.041	0.594	1.483	0.477		
T3RS	10.099	0.006	4.997	0.082		
T3LI	31.009	<0.0001	25.215	<0.0001		
T3RI	17.479	<0.0001	18.719	<0.0001		
Facet	Male	2	Fema	lle		
------------------	------------	---------	------------	---------		
T · · · ·	Chi-Square	p	Chi-Square	p		
T4LS	4.595	0.101	0.91	0.635		
T4RS	15.395	<0.0001	2.278	0.32		
T4LI	41.31	<0.0001	20.7	<0.0001		
T4RI	40.067	<0.0001	8.849	0.012		
T5LS	1.461	0.482	0.043	0.979		
T5RS	11.721	0.003	1.059	0.589		
T5LI	21.453	<0.0001	17.822	<0.0001		
T5RI	22.444	<0.0001	20.669	<0.0001		
T6LS	0.188	0.91	2.255	0.324		
T6RS	5.629	0.06	2.483	0.289		
T6LI	16.336	<0.0001	12.954	0.002		
T6RI	21.216	<0.0001	9.702	0.008		
T7LS	1.242	0.537	0.545	0.761		
T7RS	1.016	0.602	4.865	0.088		
T7LI	13.258	0.001	24.941	<0.0001		
T7RI	24.259	<0.0001	7.519	0.023		
T8LS	0.842	0.656	2.299	0.317		
T8RS	4.921	0.085	9.171	0.01		
T8LI	11.581	0.003	10.97	0.004		
T8RI	18.074	<0.0001	15.224	<0.0001		
T9LS	0.654	0.721	3.459	0.177		
T9RS	18.602	<0.0001	3.266	0.195		
T9LI	4.879	0.087	12.186	0.002		
T9RI	7.69	0.021	14.87	0.001		
T10LS	2.974	0.226	2.281	0.32		
T10RS	6.503	0.039	11.534	0.003		
T10LI	16.184	<0.0001	13.854	0.001		
T10RI	14.896	0.001	4.75	0.093		
T11LS	0.399	0.819	0.308	0.857		
T11RS	9.778	0.008	10.763	0.005		
T11LI	7.044	0.03	5.354	0.069		
T11RI	7.123	0.028	0.83	0.66		
T12LS	6.883	0.032	1.583	0.453		
T12RS	10.465	0.005	4.089	0.129		
T12LI	3,435	0.18	0.155	0.926		
T12RI	1.264	0.532	3.742	0.154		
1115	4,474	0,109	1.68	0 432		
1185	7 797	0.02	1 756	0.416		
1	3 685	0.158	3 286	0 193		
110	5.005	0.150	2.051	0.195		
LTKI	2.331	0.05	2.051	0.359		

Appendix C-5 Kruskal-Wallace test for inter-sample facet angle difference continued

Eacot	Male		Femal	e
Facet	Chi-Square	р	Chi-Square	р
L2LS	7.768	0.021	3.974	0.137
L2RS	7.966	0.019	0.482	0.786
L2LI	4.024	0.134	3.25	0.197
L2RI	3.73	0.155	0.538	0.764
L3LS	14.974	0.001	1.585	0.453
L3RS	5.752	0.056	1.052	0.591
L3LI	2.27	0.321	1.105	0.576
L3RI	5.339	0.069	5.569	0.062
L4LS	7.029	0.03	5.39	0.068
L4RS	2.66	0.265	2.734	0.255
L4LI	1.201	0.549	6.24	0.044
L4RI	6.994	0.03	8.074	0.018
L5LS	1.046	0.593	2.14	0.343
L5RS	1.981	0.371	5.461	0.065
L5LI	0.061	0.97	0.47	0.791
L5RI	5.782	0.056	8.038	0.018

Appendix C-5 Kruskal-Wallace test for inter-sample facet angle difference continued

	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$												
Facet	SBC	/SBL	SBO	C/AS	SB	L/AS	SBC/	SBL	SBO	C/AS	SB	_/AS	
	Z	р	Z	р	Z	р	Z	р	Z	р	Z	р	
C1LS	-2.758	0.006	-1.634	0.102	-0.984	0.325	-3.394	0.001	-0.538	0.591	-1.307	0.191	
C1RS	-3.42	0.001	-1.117	0.264	-0.824	0.41	-2.878	0.004	-1.011	0.312	-0.311	0.755	
C1LI	-0.251	0.802	-0.99	0.322	-1.324	0.186	-0.208	0.836	-1.912	0.056	-1.611	0.107	
C1RI	-1.075	0.282	-0.285	0.776	-0.448	0.654	-0.227	0.821	-1.437	0.151	-1.327	0.185	
C2LS	-1.818	0.069	-1.55	0.121	-2.584	0.01	-0.368	0.713	-1.134	0.257	-0.875	0.382	
C2RS	-1.829	0.067	-0.296	0.767	-1.424	0.154	-1.815	0.07	-1.033	0.301	-0.042	0.967	
C2LI	-2.713	0.007	-2.073	0.038	-4.417	<0.001	-2.338	0.019	-1.926	0.054	-0.132	0.895	
C2RI	-3.008	0.003	-1.897	0.058	-4.254	<0.001	-2.698	0.007	-1.285	0.199	-0.683	0.495	
C3LS	-2.984	0.003	-3.432	0.001	-1.502	0.133	-0.665	0.506	-3.312	0.001	-3.088	0.002	
C3RS	-2.648	0.008	-1.912	0.056	-0.098	0.922	-0.283	0.778	-3.534	<0.001	-3.461	0.001	
C3LI	-2.703	0.007	-3.011	0.003	-4.137	<0.001	-0.253	0.801	-0.183	0.855	-0.403	0.687	
C3RI	-4.515	<0.001	-3.578	<0.001	-5.141	<0.001	-2.209	0.027	-0.974	0.33	-0.464	0.643	
C4LS	-1.778	0.075	-3.029	0.002	-1.884	0.06	-0.696	0.487	-0.997	0.319	-1.737	0.082	
C4RS	-0.765	0.444	-0.298	0.766	-0.039	0.969	-0.883	0.377	-1.198	0.231	-1.69	0.091	
C4LI	-0.877	0.38	-3.322	0.001	-3.572	<0.001	-1.447	0.148	-1.47	0.142	-2.321	0.02	
C4RI	-2.46	0.014	-2.214	0.027	-3.328	0.001	-2.634	0.008	-2.771	0.006	-3.714	<0.001	
C5LS	-1.683	0.092	-1.183	0.237	-0.251	0.802	-1.203	0.229	-0.323	0.747	-0.701	0.483	
C5RS	-1.204	0.229	-2.797	0.005	-2.521	0.012	-0.325	0.745	-1.743	0.081	-1.819	0.069	
C5LI	-0.743	0.457	-1.761	0.078	-2.504	0.012	-0.972	0.331	-2.439	0.015	-2.69	0.007	
C5RI	-1.809	0.07	-2.356	0.018	-3.497	<0.001	-1.31	0.19	-3.261	0.001	-3.046	0.002	
C6LS	-1.344	0.179	-2.362	0.018	-1.579	0.114	-0.252	0.801	-1.431	0.152	-1.303	0.193	
C6RS	-2.286	0.022	-2.119	0.034	-1.622	0.105	-0.957	0.339	-0.514	0.607	-0.101	0.919	
C6LI	-2.27	0.023	-0.038	0.97	-1.782	0.075	-1.153	0.249	-1.309	0.191	-1.348	0.178	
C6RI	-2.58	0.01	-0.075	0.94	-1.483	0.138	-1.575	0.115	-1.89	0.059	-2.641	0.008	

Appendix C-6 Mann- Whitney test for inter-sample variation in facet angle

			N	lale					Fer	nale		
Facet	SBC	C/SBL	SB	C/AS	SBI	L/AS	SBC	C/SBL	SBC	C/AS	SB	L/AS
	Z	р	Z	р	Z	р	Z	р	Z	р	Z	р
C7LS	-1.797	0.072	-2.207	0.027	-1.215	0.224	-1.099	0.272	-2.394	0.017	-1.951	0.051
C7RS	-1.114	0.265	-0.741	0.459	-0.434	0.664	-1.173	0.241	-0.325	0.745	-0.156	0.876
C7LI	-2.142	0.032	-0.261	0.794	-1.44	0.15	-0.784	0.433	-1.325	0.185	-1.46	0.144
C7RI	-2.315	0.021	-0.862	0.388	-2.038	0.042	-0.943	0.345	-0.452	0.651	-0.831	0.406
T1LS	-0.643	0.52	-1.607	0.108	-1.274	0.203	-0.501	0.616	-0.711	0.477	-0.498	0.618
T1RS	-1.038	0.299	-0.925	0.355	-0.331	0.741	-0.688	0.491	-0.417	0.677	-0.052	0.958
T1LI	-0.845	0.398	-3.878	<0.001	-4.905	<0.001	-1.605	0.108	-3.734	<0.001	-3.917	<0.001
T1RI	-3.941	<0.001	-3.717	<0.001	-5.229	<0.001	-3.09	0.002	-0.772	0.44	-2.523	0.012
T2LS	-1.709	0.087	-1.241	0.215	-0.155	0.876	-1.81	0.07	-0.987	0.324	-0.411	0.681
T2RS	-1.548	0.122	-0.96	0.337	-2.037	0.042	-2.048	0.041	-3.835	<0.001	-3.089	0.002
T2LI	-1.933	0.053	-3.476	0.001	-4.836	<0.001	-2.094	0.036	-3.647	<0.001	-4.758	<0.001
T2RI	-3.947	<0.001	-1.764	0.078	-3.733	<0.001	-3.213	0.001	-3.249	0.001	-3.866	<0.001
T3LS	-0.995	0.32	-0.28	0.779	-0.59	0.555	-0.459	0.646	-0.912	0.362	-1.23	0.219
T3RS	-0.768	0.443	-3.031	0.002	-2.548	0.011	-0.164	0.87	-2.111	0.035	-1.942	0.052
T3LI	-2.112	0.035	-4.307	<0.001	-5.132	<0.001	-1.176	0.24	-4.578	<0.001	-4.449	<0.001
T3RI	-2.699	0.007	-1.85	0.064	-3.908	<0.001	-3.212	0.001	-2.426	0.015	-3.641	<0.001
T4LS	-1.601	0.109	-0.802	0.422	-1.867	0.062	-0.191	0.848	-0.889	0.374	-0.811	0.418
T4RS	-0.876	0.381	-3.129	0.002	-3.77	<0.001	-1.167	0.243	-0.559	0.576	-1.408	0.159
T4LI	-4.32	<0.001	-3.757	<0.001	-5.516	<0.001	-1.139	0.255	-3.875	<0.001	-4.319	<0.001
T4RI	-5.251	<0.001	-1.573	0.116	-5.19	<0.001	-1.954	0.051	-1.649	0.099	-2.844	0.004
T5LS	-1.221	0.222	-0.738	0.461	-0.061	0.951	-0.22	0.826	-0.026	0.979	-0.089	0.929
T5RS	-0.619	0.536	-2.785	0.005	-3.246	0.001	-0.21	0.833	-1.073	0.283	-0.687	0.492
T5LI	-1.021	0.307	-3.855	<0.001	-4.331	<0.001	-1.829	0.067	-3.43	0.001	-3.844	<0.001
T5RI	-3.788	<0.001	-1.575	0.115	-3.935	<0.001	-4.138	<0.001	-1.042	0.297	-3.424	0.001

			М	ale					Fer	nale		
Facet	SBC	/SBL	SBC	C/AS	SBI	L/AS	SBC	/SBL	SBC	C/AS	SB	_/AS
	Z	р	Z	р	Z	р	Z	р	Z	р	Z	р
T6LS	-0.379	0.704	-0.357	0.721	-0.086	0.931	-1.566	0.117	-0.041	0.967	-0.679	0.497
T6RS	-1.077	0.281	-1.522	0.128	-2.304	0.021	-0.34	0.734	-1.533	0.125	-1.273	0.203
T6LI	-2.373	0.018	-2.235	0.025	-3.791	<0.001	-1.017	0.309	-3.01	0.003	-3.524	<0.001
T6RI	-3.729	<0.001	-1.102	0.27	-3.865	<0.001	-2.459	0.014	-1.239	0.215	-2.74	0.006
T7LS	-1.088	0.277	-0.24	0.811	-0.621	0.534	-0.443	0.657	-0.639	0.523	-0.469	0.639
T7RS	-0.622	0.534	-0.927	0.354	-0.593	0.553	-0.903	0.366	-1.724	0.085	-2.115	0.034
T7LI	-0.865	0.387	-3.166	0.002	-3.383	0.001	-1.546	0.122	-4.498	<0.001	-4.351	<0.001
T7RI	-2.718	0.007	-3.308	0.001	-4.429	<0.001	-2.018	0.044	-1.093	0.274	-2.669	0.008
T8LS	-0.635	0.526	-0.264	0.792	-0.859	0.391	-0.086	0.932	-1.534	0.125	-1.133	0.257
T8RS	-0.259	0.796	-1.995	0.046	-2.041	0.041	-0.699	0.484	-2.365	0.018	-3.15	0.002
T8LI	-1.247	0.212	-3.049	0.002	-2.874	0.004	-0.786	0.432	-2.784	0.005	-3.185	0.001
T8RI	-3.691	<0.001	-1.181	0.238	-3.164	0.002	-3.224	0.001	-1.286	0.199	-3.426	0.001
T9LS	-0.523	0.601	-0.185	0.853	-0.814	0.415	-0.105	0.917	-1.85	0.064	-1.465	0.143
T9RS	-1.098	0.272	-3.979	<0.001	-3.711	<0.001	-1.387	0.165	-0.77	0.442	-1.649	0.099
T9LI	-0.066	0.947	-1.878	0.06	-2.12	0.034	-1.086	0.278	-2.76	0.006	-3.587	<0.001
T9RI	-2.772	0.006	-0.793	0.428	-1.365	0.172	-3.588	<0.001	-0.153	0.879	-2.986	0.003
T10LS	-1.385	0.166	-1.523	0.128	-0.448	0.654	-1.143	0.253	-0.765	0.445	-1.313	0.189
T10RS	-0.72	0.471	-2.333	0.02	-2.17	0.03	-0.727	0.467	-3.299	0.001	-2.695	0.007
T10LI	-0.835	0.404	-3.081	0.002	-3.972	<0.001	-0.104	0.917	-3.378	0.001	-3.577	<0.001
T10RI	-1.68	0.093	-2.496	0.013	-3.764	<0.001	-0.842	0.4	-1.842	0.065	-1.967	0.049
T11LS	-0.575	0.565	-0.435	0.664	-0.236	0.813	-0.521	0.602	-0.259	0.796	-0.279	0.78
T11RS	-2.502	0.012	-2.716	0.007	-0.881	0.378	-0.988	0.323	-3.124	0.002	-2.597	0.009
T11LI	-2.183	0.029	-2.182	0.029	-0.907	0.364	-2.295	0.022	-0.68	0.497	-1.112	0.266
T11RI	-2.152	0.031	-2.212	0.027	-1.024	0.306	-0.628	0.53	-0.773	0.439	-0.417	0.677

Appendix C-6 Mann- Whitney test for inter-sample variation in facet angle continued

Appendix C-6 Mann- Whitney test for inter-sample variation in facet angle continued

				Male			Female SBC/SBL SBC/AS SBL/AS Z p Z p Z n -1.15 0.25 -0.163 0.87 -0.973 0.3 -1.938 0.053 -1.157 0.247 -0.159 0.8 -0.373 0.709 -0.04 0.968 -0.275 0.7 -1.628 0.103 -0.668 0.504 -1.597 0.6 -0.343 0.732 -1.212 0.226 -1.052 0.2 -0.471 0.637 -1.277 0.202 -0.913 0.3 -1.774 0.076 -0.601 0.548 -1.039 0.2 -1.449 0.147 -0.189 0.85 -0.843 0.3 -1.209 0.227 -1.564 0.118 -1.725 0.0 -0.174 0.862 -0.501 0.616 -0.876 0.3 -1.6 0.11 -1.226 0.22 -0.115 0.5								
Facet	SBC/	/SBL	SBC	/AS		SBL/AS	SBC/	SBL	SBC	/AS	SBL,	/AS			
	Z	р	Z	р	Z	р	Z	р	Z	р	Z	р			
T12LS	-2.471	0.013	-1.768	0.077	-0.575	0.565	-1.15	0.25	-0.163	0.87	-0.973	0.331			
T12RS	-2.911	0.004	-2.515	0.012	-0.182	0.855	-1.938	0.053	-1.157	0.247	-0.159	0.874			
T12LI	-2.072	0.038	-0.463	0.643	-0.286	0.775	-0.373	0.709	-0.04	0.968	-0.275	0.783			
T12RI	-0.583	0.56	-0.731	0.465	-1.048	0.295	-1.628	0.103	-0.668	0.504	-1.597	0.11			
L1LS	-1.608	0.108	-0.724	0.469	-1.817	0.069	-0.343	0.732	-1.212	0.226	-1.052	0.293			
L1RS	-2.322	0.02	-0.823	0.41	-2.245	0.025	-0.471	0.637	-1.277	0.202	-0.913	0.361			
L1LI	-0.678	0.497	-1.373	0.17	-1.861	0.063	-1.774	0.076	-0.601	0.548	-1.039	0.299			
L1RI	-0.431	0.666	-1.689	0.091	-2.548	0.011	-1.449	0.147	-0.189	0.85	-0.843	0.399			
L2LS	-0.239	0.811	-2.49	0.013	-2.497	0.013	-1.209	0.227	-1.564	0.118	-1.725	0.085			
L2RS	-1.295	0.195	-2.465	0.014	-2.22	0.026	-0.174	0.862	-0.501	0.616	-0.876	0.381			
L2LI	-1.564	0.118	-0.364	0.716	-1.779	0.075	-1.6	0.11	-1.226	0.22	-0.115	0.908			
L2RI	-0.924	0.356	-1.129	0.259	-1.896	0.058	-0.104	0.917	-0.642	0.521	-0.716	0.474			
L3LS	-0.357	0.721	-3.313	0.001	-3.603	<0.001	-0.086	0.932	-1.171	0.241	-1.086	0.278			
L3RS	-0.312	0.755	-2.246	0.025	-2.029	0.042	-0.865	0.387	-0.787	0.431	-0.234	0.815			
L3LI	-0.615	0.538	-1.158	0.247	-1.366	0.172	-0.605	0.545	-1.015	0.31	-0.402	0.688			
L3RI	-1.162	0.245	-1.233	0.218	-2.33	0.02	-0.332	0.74	-2.313	0.021	-1.864	0.062			
L4LS	-0.584	0.559	-1.973	0.049	-2.65	0.008	-1.576	0.115	-1.039	0.299	-2.426	0.015			
L4RS	-0.416	0.677	-1.583	0.113	-1.234	0.217	-1.19	0.234	-0.5	0.617	-1.843	0.065			
L4LI	-1.051	0.293	-0.117	0.907	-0.76	0.447	-1.736	0.083	-2.206	0.027	-0.395	0.693			
L4RI	-0.981	0.326	-2.102	0.036	-2.407	0.016	-1.737	0.082	-2.532	0.011	-1.339	0.181			
L5LS	-0.331	0.74	-0.733	0.463	-1.008	0.314	-1.134	0.257	-0.778	0.437	-1.198	0.231			
L5RS	-0.397	0.691	-0.877	0.381	-1.503	0.133	-1.831	0.067	-1.866	0.062	-0.794	0.427			
L5LI	-0.116	0.907	-0.202	0.84	-0.215	0.83	-0.576	0.565	-0.066	0.947	-0.669	0.503			
L5RI	-0.156	0.876	-2.296	0.022	-2.076	0.038	-1.469	0.142	-1.945	0.052	-2.739	0.006			

					M	ale							Fen	nale			
			Left Fem	oral Robu	sticity		Right Ferr	noral Robu	sticity		Left Fem	oral Robu	sticity		Right Fem	oral Robu	sticity
Facet	Sample	C Coeffi cient	Ρ	N	Strength of Correlation	C Coeffi cient	Р	N	Strength of Correlation	C Coeffi cient	Ρ	N	Strength of Correlation	C Coeffi cient	Ρ	N	Strength of Correlation
	SBC	0.37	0.27	11	Medium	0.28	0.43	10	Weak	0.29	0.25	17	Weak	0.37	0.18	15	Medium
C1LS	SBL	-0.29	0.25	18	Weak	-0.23	0.35	19	Weak	-0.03	0.92	11	None	-0.48	0.16	10	Medium
	AS	0.10	0.75	12	Weak	0.30	0.31	13	Medium	0.00	1.00	6	None	-0.13	0.76	8	Weak
	SBC	0.53	0.09	11	Strong	0.44	0.20	10	Medium	0.43	0.08	17	Medium	0.14	0.62	15	Weak
C1RS	SBL	-0.17	0.51	18	Weak	-0.14	0.57	19	Weak	0.09	0.78	11	None	-0.34	0.34	10	Medium
	AS	-0.12	0.71	12	Weak	0.15	0.61	13	Weak	0.71	0.12	6	Strong	0.34	0.41	8	Medium
	SBC	0.08	0.81	11	None	0.28	0.43	10	Weak	-0.20	0.43	17	Weak	0.38	0.16	15	Medium
C1LI	SBL	-0.08	0.76	18	None	-0.13	0.60	19	Weak	0.28	0.41	11	Weak	-0.46	0.19	10	Medium
	AS	0.02	0.96	10	None	-0.12	0.74	11	Weak	0.27	0.61	6	Weak	0.16	0.70	8	Weak
	SBC	0.06	0.86	11	None	0.29	0.42	10	Weak	-0.23	0.38	17	Weak	0.41	0.13	15	Medium
C1RI	SBL	-0.08	0.76	18	None	0.00	1.00	19	None	0.21	0.54	11	Weak	-0.38	0.27	10	Medium
	AS	0.11	0.75	11	Weak	0.04	0.89	12	None	0.51	0.25	7	Strong	0.18	0.64	9	Weak
	SBC	0.11	0.74	12	Weak	0.12	0.73	11	Weak	-0.28	0.25	19	Weak	-0.29	0.27	17	Weak
C2LS	SBL	-0.12	0.61	20	Weak	-0.11	0.61	22	Weak	0.00	1.00	11	None	-0.32	0.37	10	Medium
	AS	0.65	0.08	8	Strong	0.71	0.03	9	Strong	0.57	0.18	7	Strong	0.39	0.31	9	Medium

					M	ale							Fe	male			
		Le	eft Femor	al Robusti	icity	R	ight Fem	oral Robu	sticity	L	eft Femo	oral Robus	sticity	F	Right Fem	oral Robi	usticity
Facet	Sample	C Coeffi cient	Ρ	N	Strength of Correlat ion	C Coeffi cient	Ρ	N	Strength of Correlatio n	C Coeffi cient	Ρ	N	Strength of Correlatio n	C Coeffi cient	Ρ	Ν	Strength of Correlation
	SBC	0.13	0.68	12	Weak	0.20	0.56	11	Weak	-0.16	0.50	19	Weak	-0.23	0.38	17	Weak
C2RS	SBL	-0.22	0.34	20	Weak	-0.24	0.29	22	Weak	0.08	0.83	11	None	-0.25	0.49	10	Weak
	AS	0.22	0.60	8	Weak	0.37	0.32	9	Medium	0.62	0.13	7	Strong	0.42	0.26	9	Medium
	SBC	0.00	0.99	12	None	-0.08	0.80	11	None	-0.08	0.74	19	None	-0.19	0.47	17	Weak
C2LI	SBL	0.04	0.88	20	None	-0.08	0.73	22	None	0.03	0.93	11	None	-0.13	0.71	10	Weak
	AS	-0.67	0.07	8	Strong	-0.65	0.06	9	Strong	0.52	0.23	7	Strong	0.59	0.12	8	Strong
	SBC	0.25	0.43	12	Weak	-0.06	0.86	11	None	0.25	0.30	19	Weak	-0.25	0.34	17	Weak
C2RI	SBL	0.14	0.55	20	Weak	0.02	0.92	22	None	0.15	0.67	11	Weak	-0.10	0.78	10	Weak
	AS	0.24	0.57	8	Weak	0.20	0.59	10	Weak	0.46	0.30	7	Medium	0.55	0.16	8	Strong
	SBC	0.24	0.49	11	Weak	0.26	0.47	10	Weak	-0.31	0.22	17	Medium	-0.39	0.14	16	Medium
C3LS	SBL	0.02	0.93	20	None	-0.08	0.72	22	None	0.18	0.58	12	Weak	0.03	0.91	12	None
	AS	0.00	0.99	10	None	-0.13	0.71	11	Weak	0.08	0.85	8	None	-0.22	0.57	9	Weak
	SBC	0.08	0.81	12	None	0.33	0.32	11	Medium	-0.28	0.28	17	Weak	-0.21	0.43	16	Weak
C3RS	SBL	0.06	0.81	20	None	-0.18	0.41	22	Weak	0.31	0.35	11	Medium	0.14	0.69	11	Weak
	AS	-0.23	0.52	10	Weak	-0.29	0.38	11	Weak	0.25	0.56	8	Weak	-0.13	0.74	9	Weak

					N	lale							Fen	nale			
Facet	Sample	Let	ft Femora	l Robust	ticity	Rigl	ht Femor	al Robust	icity	Le	eft Femor	al Robust	ticity	F	Right Fem	ioral Rob	usticity
		C Coeffi	Р	N	Strength of	C Coeffici	Р	Ν	Strength of	C Coeffici	Р	Ν	Strength of	C Coeffi	Р	Ν	Strength of Correlation
	SBC	-0.18	0.58	12	Weak	0.13	0.70	11	Weak	0.04	0.87	17	None	-0.02	0.95	16	None
C3LI	SBL	-0.09	0.72	19	None	0.29	0.20	21	Weak	0.32	0.31	12	Medium	0.35	0.26	12	Medium
	AS	0.21	0.57	10	Weak	0.22	0.52	11	Weak	-0.47	0.24	8	Medium	-0.50	0.17	9	Strong
	SBC	0.38	0.22	12	Medium	0.41	0.21	11	Medium	0.19	0.46	17	Weak	0.32	0.23	16	Medium
C3RI	SBL	-0.52	0.02	20	Strong	-0.53	0.01	22	Strong	0.16	0.63	11	Weak	-0.13	0.70	11	Weak
	AS	-0.15	0.68	10	Weak	-0.11	0.77	10	Weak	-0.10	0.81	8	Weak	-0.22	0.56	9	Weak
	SBC	0.28	0.37	12	Weak	-0.10	0.78	11	Weak	0.25	0.37	15	Weak	0.31	0.25	15	Medium
C4LS	SBL	-0.17	0.46	21	Weak	-0.26	0.24	23	Weak	-0.68	0.01	14	Strong	-0.02	0.94	11	None
	AS	0.19	0.62	9	Weak	0.24	0.47	11	Weak	-0.03	0.94	7	None	-0.06	0.88	8	None
	SBC	-0.30	0.35	12	Medium	-0.01	0.97	11	None	0.50	0.06	15	Strong	0.53	0.04	15	Strong
C4RS	SBL	-0.24	0.30	21	Weak	-0.24	0.26	23	Weak	-0.96	0.00	14	Strong	0.07	0.84	11	None
	AS	0.08	0.84	9	None	0.14	0.67	11	Weak	-0.20	0.67	7	Weak	-0.18	0.67	8	Weak
	SBC	0.55	0.06	12	Strong	0.30	0.38	11	Medium	0.56	0.03	15	Strong	0.47	0.08	15	Medium
C4LI	SBL	0.06	0.80	21	None	0.07	0.75	23	None	-0.45	0.09	15	Medium	-0.13	0.70	12	Weak
	AS	0.43	0.29	8	Medium	0.47	0.17	10	Medium	0.69	0.09	7	Strong	0.63	0.09	8	Strong

					N	lale							Fe	male			
		Let	ft Femora	l Robust	icity	Righ	t Femoral	Robust	icity	Le	ft Femora	l Robust	icity		Right Fem	ioral Rol	ousticity
Facet	Sample	C Coeffi cient	Ρ	N	Strength of Correlati on	C Coeffici ent	Ρ	N	Strength of Correlati on	C Coeffici ent	Ρ	N	Strength of Correlatio n	C Coeffi cient	Ρ	N	Strength of Correlation
	SBC	0.36	0.24	12	Medium	0.17	0.62	11	Weak	0.48	0.07	15	Medium	0.49	0.06	15	Medium
C4RI	SBL	-0.12	0.62	21	Weak	-0.06	0.77	23	None	-0.64	0.01	14	Strong	0.30	0.37	11	Medium
	AS	0.12	0.78	8	Weak	0.14	0.70	10	Weak	0.62	0.14	7	Strong	0.68	0.06	8	Strong
	SBC	0.21	0.49	13	Weak	0.04	0.90	12	None	0.16	0.54	17	Weak	0.14	0.63	14	Weak
C5LS	SBL	0.21	0.37	21	Weak	0.21	0.36	22	Weak	-0.46	0.08	15	Medium	0.55	0.05	13	Strong
	AS	0.24	0.57	8	Weak	0.32	0.41	9	Medium	-0.30	0.51	7	Medium	-0.29	0.45	9	Weak
	SBC	-0.20	0.52	13	Weak	-0.31	0.32	12	Medium	0.26	0.32	17	Weak	0.33	0.25	14	Medium
C5RS	SBL	0.18	0.44	21	Weak	0.15	0.51	22	Weak	-0.65	0.01	15	Strong	0.47	0.10	13	Medium
	AS	-0.04	0.92	8	None	-0.11	0.78	9	Weak	-0.35	0.45	7	Medium	-0.32	0.40	9	Medium
	SBC	0.26	0.38	13	Weak	0.14	0.66	12	Weak	0.27	0.29	17	Weak	-0.07	0.82	14	None
C5LI	SBL	0.04	0.85	20	None	0.11	0.64	21	Weak	-0.29	0.30	15	Weak	0.23	0.44	13	Weak
	AS	0.29	0.48	8	Weak	0.01	0.99	9	None	-0.77	0.04	7	Strong	-0.70	0.03	9	Strong
	SBC	0.13	0.68	13	Weak	0.32	0.31	12	Medium	0.22	0.40	17	Weak	0.22	0.45	14	Weak
C5RI	SBL	0.11	0.65	20	Weak	0.27	0.24	21	Weak	-0.26	0.35	15	Weak	0.03	0.92	13	None
	AS	0.09	0.83	8	None	0.18	0.65	9	Weak	-0.47	0.29	7	Medium	-0.39	0.30	9	Medium

					N	1ale							Fe	male			
		Let	ft Femora	l Robust	ticity	Righ	t Femoral	l Robust	icity	Le	ft Femora	l Robust	icity		Right Fem	oral Rol	ousticity
Facet	Sample	C Coeffi cient	Ρ	Ν	Strength of Correlat ion	C Coefficie nt	Ρ	N	Strength of Correlat ion	C Coeffici ent	Ρ	Ν	Strength of Correlatio n	C.coef ficient	Ρ	N	Strength of Correlation
	SBC	0.64	0.05	10	Strong	-0.28	0.44	10	Weak	0.08	0.71	22	None	0.15	0.53	19	Weak
C6LS	SBL	-0.04	0.85	23	None	0.07	0.74	24	None	0.08	0.78	14	None	0.02	0.95	13	None
	AS	0.34	0.37	9	Medium	0.23	0.52	10	Weak	-0.09	0.84	7	None	-0.05	0.91	8	None
	SBC	0.28	0.43	10	Weak	-0.19	0.59	10	Weak	0.07	0.76	22	None	0.03	0.91	19	None
C6RS	SBL	-0.17	0.44	22	Weak	-0.07	0.74	23	None	0.59	0.02	15	Strong	0.11	0.73	13	Weak
	AS	0.29	0.45	9	Weak	0.38	0.29	10	Medium	-0.47	0.28	7	Medium	-0.47	0.25	8	Medium
	SBC	0.52	0.12	10	Strong	0.06	0.88	10	None	-0.26	0.24	22	Weak	-0.20	0.41	19	Weak
C6LI	SBL	0.20	0.34	24	Weak	0.12	0.56	25	Weak	-0.02	0.94	14	None	0.09	0.77	13	None
	AS	-0.14	0.72	9	Weak	-0.01	0.99	10	None	0.50	0.26	7	Strong	0.36	0.38	8	Medium
	SBC	0.45	0.19	10	Medium	0.20	0.58		Weak	0.17	0.44	22	Weak	0.09	0.71	19	None
C6RI	SBL	0.13	0.54	24	Weak	0.11	0.60	25	Weak	0.10	0.73	15	None	0.15	0.62	13	Weak
	AS	0.16	0.69	9	Weak	0.27	0.45	10	Weak	0.19	0.69	7	Weak	-0.01	0.99	8	None
	SBC	0.33	0.30	12	Medium	0.08	0.82	11	None	-0.09	0.68	21	None	0.13	0.60	18	Weak
C7LS	SBL	-0.12	0.56	25	Weak	-0.15	0.47	25	Weak	-0.38	0.16	15	Medium	0.05	0.86	13	None
	AS	-0.34	0.40	8	Medium	-0.22	0.57	9	Weak	-0.12	0.83	6	Weak	-0.11	0.84	6	Weak

					Male								Fe	emale			
		Left	Femora	l Robust	ticity	Righ	nt Femo	ral Ro	busticity	Le	ft Femo	ral Ro	busticity	Rig	ht Femo	ral Rob	usticity
Facet	Sample	C Coefficient	Р	N	Strength of Correlation	C Coeffici ent	Р	N	Strength of Correlation	C Coeffi cient	Р	N	Strength of Correlation	C Coeffici ent	Р	N	Strength of Correlation
	SBC	0.49	0.11	12	Medium	0.44	0.17	11	Medium	-0.13	0.59	20	Weak	-0.08	0.76	17	None
C7RS	SBL	0.02	0.93	24	None	-0.05	0.82	25	None	-0.14	0.61	15	Weak	-0.05	0.87	13	None
	AS	0.25	0.52	9	Weak	0.35	0.32	10	Medium	0.13	0.81	6	Weak	0.14	0.79	6	Weak
	SBC	-0.01	0.99	12	None	-0.67	0.03	11	Strong	0.08	0.73	21	None	0.03	0.90	18	None
C7LI	SBL	-0.02	0.94	24	None	0.01	0.95	26	None	0.16	0.58	15	Weak	0.17	0.57	13	Weak
	AS	-0.23	0.55	9	Weak	0.00	1.00	10	None	-0.86	0.03	6	Strong	-0.87	0.02	6	Strong
	SBC	0.15	0.63	12	Weak	-0.38	0.25	11	Medium	-0.09	0.71	21	None	0.32	0.20	18	Medium
C7RI	SBL	0.06	0.77	24	None	0.00	0.99	26	None	0.15	0.60	15	Weak	0.13	0.67	13	Weak
	AS	-0.52	0.16	9	Strong	-0.25	0.49	10	Weak	0.14	0.79	6	Weak	0.14	0.79	6	Weak
	SBC	0.25	0.46	11	Weak	-0.08	0.83	10	None	0.23	0.28	23	Weak	0.08	0.75	19	None
T1LS	SBL	-0.12	0.57	25	Weak	0.03	0.89	26	None	0.20	0.49	14	Weak	0.28	0.39	12	Weak
	AS	0.08	0.82	11	None	0.26	0.39	13	Weak	-0.28	0.40	11	Weak	-0.24	0.48	11	Weak
	SBC	-0.07	0.82	12	None	0.14	0.68	11	Weak	0.12	0.60	22	Weak	0.31	0.21	18	Medium
T1RS	SBL	0.00	0.99	25	None	0.11	0.58	26	Weak	0.20	0.49	14	Weak	-0.30	0.35	12	Medium
	AS	-0.05	0.88	11	None	0.14	0.67	12	Weak	0.41	0.31	8	Medium	0.38	0.28	10	Medium

					Ma	le							Fe	male			
- ·		Le	eft Femor	al Robus	sticity	F	Right Fem	oral Rob	usticity	Le	ft Femo	ral Ro	busticity	Ri	ight Femo	ral Rob	usticity
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffi cient	Р	N	Strength of Correlation	C Coeffi cient	Ρ	N	Strength of Correlation	C Coeffi cient	Ρ	N	Strength of Correlation
	SBC	-0.27	0.39	12	Weak	-0.08	0.82	11	None	0.23	0.31	22	Weak	0.12	0.64	18	Weak
T1LI	SBL	-0.05	0.79	26	None	0.06	0.79	26	None	0.67	0.01	14	Strong	0.61	0.04	12	Strong
	AS	0.41	0.21	11	Medium	0.39	0.22	12	Medium	-0.17	0.68	8	Weak	-0.19	0.60	10	Weak
	SBC	-0.04	0.89	12	None	0.04	0.92	11	None	0.10	0.66	22	None	-0.05	0.84	18	None
T1RI	SBL	0.10	0.62	26	Weak	0.10	0.61	26	Weak	0.23	0.42	14	Weak	0.64	0.02	12	Strong
	AS	-0.25	0.46	11	Weak	-0.28	0.38	12	Weak	0.25	0.54	8	Weak	0.28	0.44	10	Weak
	SBC	0.32	0.29	13	Medium	0.17	0.59	12	Weak	0.20	0.39	21	Weak	0.19	0.47	17	Weak
T2LS	SBL	-0.07	0.73	25	None	0.00	0.99	25	None	0.34	0.25	13	Medium	0.37	0.24	12	Medium
	AS	0.50	0.14	10	Strong	0.62	0.06	10	Strong	0.10	0.77	11	None	0.04	0.91	13	None
	SBC	0.03	0.93	13	None	0.29	0.37	12	Weak	-0.16	0.50	21	Weak	0.14	0.58	17	Weak
T2RS	SBL	-0.08	0.69	26	None	0.04	0.86	26	None	-0.11	0.72	13	Weak	0.66	0.02	12	Strong
	AS	0.53	0.12	10	Strong	0.51	0.14	10	Strong	-0.31	0.36	11	Medium	-0.38	0.20	13	Medium
	SBC	0.03	0.91	13	None	-0.03	0.94	12	None	0.38	0.09	21	Medium	0.17	0.52	17	Weak
T2LI	SBL	0.02	0.93	25	None	0.07	0.74	25	None	-0.23	0.45	13	Weak	0.63	0.03	12	Strong
	AS	0.09	0.80	10	None	0.07	0.84	10	None	-0.44	0.17	11	Medium	-0.46	0.12	13	None

Appendix C-7 Correlation between femoral robusticity and facet angle continued	
--	--

					Mal	е							Fer	male			
- ·		Le	eft Femor	al Robu	sticity	Rig	ght Femo	oral Ro	busticity	Le	eft Femo	ral Rob	ousticity	Righ	t Femor	al Robi	usticity
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffic ient	Р	N	Strength of Correlation	C Coeffic ient	Ρ	N	Strength of Correlation	C Coefficie nt	Ρ	N	Strength of Correlation
	SBC	-0.29	0.34	13	Weak	0.34	0.28	12	Medium	0.20	0.38	21	Weak	-0.03	0.91	17	None
T2RI	SBL	0.18	0.40	25	Weak	0.20	0.34	25	Weak	0.19	0.54	13	Weak	0.33	0.30	12	Medium
	AS	0.05	0.89	10	None	0.11	0.76	10	Weak	-0.07	0.84	11	None	-0.04	0.88	13	None
	SBC	-0.37	0.21	13	Medium	-0.02	0.95	12	None	0.05	0.82	21	None	0.22	0.39	17	Weak
T3LS	SBL	0.17	0.47	21	Weak	0.23	0.34	20	Weak	-0.26	0.41	12	Weak	0.44	0.15	12	Medium
	AS	-0.25	0.46	11	Weak	-0.40	0.23	11	Medium	-0.43	0.19	11	Medium	-0.34	0.25	13	Medium
	SBC	-0.32	0.29	13	Medium	0.12	0.70	12	Weak	0.19	0.41	21	Weak	0.19	0.48	17	Weak
T3RS	SBL	0.05	0.82	22	None	0.10	0.67	21	None	0.28	0.39	12	Weak	0.51	0.09	12	Strong
	AS	-0.09	0.78	11	None	-0.08	0.80	12	None	-0.34	0.30	11	Medium	-0.41	0.16	13	Medium
	SBC	-0.28	0.35	13	Weak	-0.03	0.93	12	None	0.33	0.14	21	Medium	-0.05	0.85	17	None
T3LI	SBL	0.13	0.56	22	Weak	-0.05	0.82	21	None	-0.40	0.20	12	Medium	0.84	0.00	12	Strong
	AS	-0.12	0.74	10	Weak	-0.16	0.63	11	Weak	-0.31	0.36	11	Medium	-0.16	0.61	12	Weak
	SBC	-0.13	0.68	13	Weak	-0.18	0.58	12	Weak	0.42	0.06	21	Medium	0.15	0.57	17	Weak
T3RI	SBL	0.03	0.90	22	None	0.03	0.88	21	None	-0.42	0.18	12	Medium	0.42	0.17	12	Medium
	AS	0.01	0.98	11	None	0.00	0.99	12	None	0.19	0.58	11	Weak	0.22	0.48	13	Weak

					Male	e							Fer	nale			
		Le	eft Femor	al Robu	sticity	Rig	ght Femo	oral Ro	busticity	Le	ft Femo	ral Rob	usticity	Righ	nt Femor	al Robu	usticity
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffic ient	Ρ	N	Strength of Correlation	C Coeffic ient	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlation
	SBC	-0.20	0.52	13	Weak	-0.02	0.95	12	None	-0.11	0.66	20	Weak	-0.28	0.30	16	Weak
T4LS	SBL	-0.02	0.92	23	None	-0.01	0.98	22	None	-0.65	0.04	10	Strong	0.17	0.63	10	Weak
	AS	0.07	0.83	12	None	0.08	0.82	12	None	-0.13	0.68	12	Weak	-0.09	0.77	13	None
	SBC	0.02	0.95	13	None	0.16	0.63	12	Weak	-0.22	0.36	19	Weak	-0.37	0.18	15	Medium
T4RS	SBL	-0.22	0.33	22	Weak	-0.24	0.30	21	Weak	0.11	0.76	10	Weak	0.21	0.56	10	Weak
	AS	0.21	0.52	12	Weak	0.20	0.53	12	Weak	0.16	0.61	12	Weak	0.19	0.53	13	Weak
	SBC	0.04	0.91	13	None	0.03	0.92	12	None	0.28	0.24	19	Weak	0.13	0.64	16	Weak
T4LI	SBL	-0.11	0.63	22	Weak	-0.07	0.77	21	None	-0.19	0.63	9	Weak	0.60	0.09	9	Strong
	AS	0.15	0.64	13	Weak	-0.03	0.92	13	None	-0.33	0.33	11	Medium	-0.18	0.57	12	Weak
	SBC	0.26	0.39	13	Weak	0.29	0.35	12	Weak	-0.10	0.70	19	Weak	-0.30	0.26	16	Medium
T4RI	SBL	-0.07	0.76	23	None	-0.23	0.30	22	Weak	-0.13	0.74	9	Weak	0.51	0.17	9	Strong
	AS	0.14	0.64	13	Weak	0.17	0.58	13	Weak	-0.02	0.96	11	None	-0.09	0.79	12	None
	SBC	-0.09	0.78	12	None	0.18	0.57	12	Weak	-0.20	0.40	20	Weak	-0.01	0.98	17	None
T5LS	SBL	-0.03	0.88	21	None	-0.04	0.87	20	None	-0.28	0.40	11	Weak	0.29	0.42	10	Weak
	AS	-0.15	0.62	13	Weak	-0.22	0.48	13	Weak	-0.05	0.91	8	None	0.22	0.57	9	Weak

					Male								Fer	nale			
			Left Femo	ral Robustic	ity	Ri	ght Fem	oral Rol	ousticity	Le	ft Femor	ral Robus	sticity	Rigl	ht Femo	ral Ro	busticity
Facet	Sample	C Coeffi cient	Ρ	Ν	Strength of Correlatio n	C Coeffi cient	Ρ	N	Strength of Correlation	C Coeffic ient	Ρ	Ν	Strength of Correlatio n	C Coeffici ent	Ρ	N	Strength of Correlation
	SBC	-0.18	0.57	12	Weak	0.06	0.85	12	None	-0.10	0.67	20	Weak	-0.24	0.35	17	Weak
T5RS	SBL	-0.11	0.65	21	Weak	-0.20	0.40	20	Weak	-0.04	0.91	11	None	0.51	0.13	10	Strong
	AS	-0.08	0.81	13	None	-0.15	0.62	13	Weak	-0.48	0.22	8	Medium	-0.24	0.54	9	Weak
	SBC	0.27	0.40	12	Weak	0.11	0.74	12	Weak	0.38	0.10	20	Medium	0.24	0.34	17	Weak
T5LI	SBL	-0.20	0.40	20	Weak	-0.07	0.79	18	None	-0.23	0.48	12	Weak	0.43	0.21	10	Medium
	AS	0.36	0.28	11	Medium	0.33	0.32	11	Medium	-0.27	0.52	8	Weak	0.15	0.71	9	Weak
	SBC	0.21	0.52	12	Weak	-0.17	0.59	12	Weak	0.41	0.07	20	Medium	0.14	0.58	17	Weak
T5RI	SBL	0.12	0.61	21	Weak	0.24	0.31	20	Weak	-0.15	0.65	12	Weak	0.37	0.29	10	Medium
	AS	0.23	0.50	11	Weak	0.20	0.55	11	Weak	0.48	0.23	8	Medium	0.59	0.09	9	Strong
	SBC	-0.13	0.67	13	Weak	-0.32	0.32	12	Medium	0.17	0.48	19	Weak	0.24	0.36	16	Weak
T6LS	SBL	0.05	0.82	24	None	0.10	0.67	23	None	0.01	0.99	12	None	0.23	0.50	11	Weak
	AS	0.31	0.41	9	Medium	0.11	0.75	11	Weak	-0.22	0.58	9	Weak	-0.04	0.90	10	None
	SBC	-0.14	0.66	13	Weak	-0.05	0.87	12	None	-0.18	0.47	19	Weak	-0.24	0.36	16	Weak
T6RS	SBL	-0.24	0.25	24	Weak	-0.20	0.37	23	Weak	0.14	0.65	12	Weak	0.20	0.56	11	Weak
	AS	0.17	0.66	9	Weak	0.11	0.74	11	Weak	-0.63	0.07	9	Strong	-0.43	0.21	10	Medium

					Ma	le							Fema	le			
_ .		Le	eft Femora	al Robus	sticity	Rig	ht Femor	al Rob	usticity	Le	ft Femora	l Robu	isticity	Righ	t Femor	al Rob	usticity
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	Ν	Strength of Correlation	C Coeffici ent	Ρ	N	Strength of Correlation
	SBC	0.48	0.10	13	Medium	-0.12	0.71	12	Weak	0.07	0.78	20	None	0.01	0.98	16	None
T6LI	SBL	-0.19	0.38	23	Weak	-0.16	0.48	22	Weak	0.57	0.05	12	Strong	0.39	0.24	11	Medium
	AS	0.61	0.14	7	Strong	0.44	0.23	9	Medium	-0.47	0.20	9	Medium	-0.21	0.56	10	Weak
	SBC	-0.29	0.34	13	Weak	0.06	0.85	12	None	-0.09	0.70	20	None	-0.12	0.65	16	Weak
T6RI	SBL	-0.02	0.92	23	None	0.11	0.63	22	Weak	0.14	0.66	12	Weak	0.19	0.58	11	Weak
	AS	0.59	0.16	7	Strong	0.06	0.89	9	None	-0.61	0.08	9	Strong	-0.35	0.32	10	Medium
	SBC	0.58	0.04	13	Strong	0.19	0.56	12	Weak	-0.16	0.50	21	Weak	-0.04	0.87	17	None
T7LS	SBL	0.08	0.73	23	None	0.19	0.39	23	Weak	0.07	0.83	12	None	0.53	0.09	11	Strong
	AS	0.39	0.29	9	Medium	0.23	0.52	10	Weak	-0.08	0.80	12	None	0.02	0.95	13	None
	SBC	0.60	0.03	13	Strong	0.40	0.20	12	Medium	0.07	0.77	21	None	-0.13	0.61	17	Weak
T7RS	SBL	-0.04	0.84	24	None	-0.02	0.94	24	None	-0.34	0.29	12	Medium	0.20	0.58	10	Weak
	AS	0.21	0.58	9	Weak	0.16	0.66	10	Weak	-0.10	0.75	12	Weak	0.01	0.97	13	None
	SBC	0.46	0.13	12	Medium	0.05	0.89	11	None	0.09	0.70	21	None	-0.12	0.66	17	Weak
T7LI	SBL	-0.31	0.15	23	Medium	-0.16	0.48	22	Weak	-0.26	0.42	12	Weak	-0.32	0.33	11	Medium
	AS	0.45	0.31	7	Medium	0.42	0.30	8	Medium	-0.17	0.63	11	Weak	-0.04	0.91	12	None

					Ma	le							Fema	le			
- ·		Le	eft Femora	al Robus	sticity	Rig	ht Femo	oral Rob	usticity	Le	ft Femo	ral Robu	isticity	Rig	ght Femo	oral Rob	ousticity
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlation	C Coeffici ent	Ρ	Ν	Strength of Correlation	C Coeffici ent	Ρ	Ν	Strength of Correlation	C Coeffi cient	Ρ	Ν	Strength of Correlation
	SBC	0.58	0.05	12	Strong	0.28	0.40	11	Weak	0.37	0.10	21	Medium	-0.10	0.69	17	Weak
T7RI	SBL	-0.33	0.12	23	Medium	-0.25	0.25	23	Weak	0.03	0.94	12	None	0.19	0.57	11	Weak
	AS	0.67	0.10	7	Strong	0.72	0.04	8	Strong	0.12	0.70	12	Weak	0.14	0.66	13	Weak
	SBC	0.16	0.61	12	Weak	-0.12	0.73	11	Weak	-0.32	0.19	19	Medium	0.07	0.79	16	None
T8LS	SBL	0.07	0.75	24	None	0.06	0.78	23	None	0.09	0.77	12	None	0.35	0.27	12	Medium
	AS	-0.17	0.65	9	Weak	-0.05	0.88	11	None	-0.33	0.30	12	Medium	-0.05	0.87	13	None
	SBC	-0.11	0.74	12	Weak	0.18	0.60	11	Weak	-0.13	0.61	19	Weak	-0.22	0.42	16	Weak
T8RS	SBL	-0.12	0.57	24	Weak	-0.18	0.42	23	Weak	-0.72	0.01	12	Strong	-0.05	0.88	12	None
	AS	-0.09	0.82	9	None	-0.27	0.43	11	Weak	0.32	0.31	12	Medium	0.40	0.17	13	Medium
	SBC	0.19	0.53	13	Weak	-0.10	0.75	12	Weak	0.21	0.40	19	Weak	0.15	0.57	16	Weak
T8LI	SBL	0.03	0.91	24	None	-0.07	0.76	23	None	0.73	0.02	10	Strong	-0.01	0.97	10	None
	AS	0.16	0.67	9	Weak	0.13	0.70	11	Weak	-0.24	0.45	12	Weak	-0.13	0.67	13	Weak
	SBC	0.28	0.36	13	Weak	-0.41	0.18	12	Medium	-0.07	0.78	19	None	0.33	0.21	16	Medium
T8RI	SBL	-0.03	0.88	24	None	-0.04	0.84	23	None	-0.10	0.77	11	Weak	-0.25	0.46	11	Weak
	AS	0.06	0.89	9	None	-0.10	0.77	11	Weak	-0.04	0.89	12	None	0.02	0.94	13	None

					Ma	le							Fema	le			
- ·	Comme	Le	eft Femora	al Robus	sticity	Rig	ht Femo	oral Rob	usticity	L	eft Femc	oral Robu	isticity	Ri	ght Femo	oral Rob	ousticity
Facet	e	C Coeffici ent	Ρ	Z	Strength of Correlation	C Coeffici ent	Ρ	Ν	Strength of Correlation	C Coeffic ient	Ρ	Ν	Strength of Correlation	C Coeffi cient	Ρ	N	Strength of Correlation
	SBC	0.02	0.94	13	None	0.61	0.04	12	Strong	-0.12	0.61	21	Weak	0.09	0.73	18	None
T9LS	SBL	0.00	1.00	24	None	0.02	0.92	24	None	0.24	0.45	12	Weak	0.50	0.09	12	Strong
	AS	0.42	0.22	10	Medium	0.18	0.58	12	Weak	-0.13	0.69	11	Weak	-0.03	0.92	12	None
	SBC	0.13	0.67	13	Weak	-0.26	0.42	12	Weak	-0.28	0.22	21	Weak	-0.05	0.84	18	None
T9RS	SBL	0.20	0.34	24	Weak	0.15	0.50	24	Weak	-0.13	0.69	12	Weak	0.08	0.80	12	None
	AS	0.38	0.31	9	Medium	0.26	0.45	11	Weak	-0.34	0.30	11	Medium	-0.07	0.84	12	None
	SBC	0.46	0.13	12	Medium	-0.47	0.15	11	Medium	0.07	0.76	21	None	0.30	0.23	18	Medium
T9LI	SBL	-0.17	0.43	24	Weak	-0.23	0.27	24	Weak	-0.09	0.77	12	None	-0.14	0.66	12	Weak
	AS	-0.15	0.69	10	Weak	-0.24	0.46	12	Weak	-0.05	0.89	11	None	0.07	0.82	12	None
	SBC	0.06	0.85	12	None	0.14	0.69	11	Weak	0.16	0.50	21	Weak	0.29	0.25	18	Weak
T9RI	SBL	0.53	0.01	23	Strong	0.54	0.01	23	Strong	-0.31	0.33	12	Medium	-0.43	0.16	12	None
	AS	-0.39	0.27	10	Medium	-0.39	0.21	12	Medium	-0.19	0.57	11	Weak	-0.18	0.58	12	Weak
	SBC	-0.22	0.49	12	Weak	0.50	0.12	11	Strong	-0.11	0.63	21	Weak	0.05	0.85	18	None
T10LS	SBL	0.02	0.92	23	None	0.06	0.79	22	None	0.14	0.67	12	Weak	0.28	0.37	12	Weak
	AS	0.34	0.33	10	Medium	0.33	0.28	13	Medium	0.13	0.70	11	Weak	0.13	0.68	12	Weak

					М	ale							Fem	ale			
Facet	Sampl	L	eft Femor	al Robu	sticity	Rig	ght Femo	oral Robu	usticity	L	eft Fem	oral Rob	ousticity	Ri	ght Femo	oral Rol	ousticity
Tacet	e	C Coeffici ent	Р	N	Strength of Correlation	C Coefficie nt	Р	N	Strength of Correlation	C Coeffi cient	Р	N	Strength of Correlation	C Coeffi cient	Р	N	Strength of Correlation
	SPC	0.06	0.85	12	None	-0.01	0.99	11	None	-0.20	0.38	21	Weak	-0.22	0.38	18	Weak
T10RS	CDI	0.11	0.63	23	Weak	0.09	0.69	23	None	0.24	0.45	12	Weak	0.18	0.58	12	Weak
	۸¢	0.04	0.92	10	None	0.26	0.42	12	Weak	0.15	0.66	11	Weak	0.22	0.50	12	Weak
	SPC	-0.49	0.09	13	Medium	-0.08	0.81	12	None	-0.09	0.70	21	None	0.08	0.77	18	None
T10LI	CDI	0.12	0.58	23	Weak	0.10	0.64	23	Weak	0.06	0.86	12	None	0.13	0.68	12	Weak
	Δ <u>ς</u>	0.13	0.71	10	Weak	0.01	0.97	13	None	-0.02	0.95	11	None	0.14	0.67	12	Weak
	SBC	-0.51	0.08	13	Strong	0.20	0.53	12	Weak	-0.30	0.19	21	Medium	-0.25	0.32	18	Weak
T10RI	CDI	0.50	0.02	23	Strong	0.59	0.00	23	Strong	-0.21	0.52	12	Weak	-0.22	0.49	12	Weak
	٨٩	-0.44	0.20	10	Medium	-0.39	0.19	13	Medium	-0.30	0.38	11	Medium	-0.13	0.68	12	Weak
	SBC	-0.42	0.15	13	Medium	-0.27	0.40	12	Weak	-0.40	0.08	21	Medium	-0.13	0.60	18	Weak
T11LS	CDI	0.20	0.35	23	Weak	0.14	0.53	22	Weak	0.02	0.96	13	None	0.63	0.03	12	Strong
	٨٩	-0.19	0.59	10	Weak	-0.17	0.61	12	Weak	0.10	0.77	11	None	0.14	0.66	12	Weak
	SBC	-0.51	0.07	13	Strong	-0.17	0.59	12	Weak	-0.40	0.07	21	Medium	-0.31	0.22	18	Medium
T11RS	SBI	0.30	0.17	23	Medium	0.32	0.15	22	Medium	0.11	0.73	13	Weak	0.37	0.24	12	Medium
	A.C.	-0.04	0.91	10	None	0.00	1.00	12	None	-0.23	0.49	11	Weak	0.04	0.90	12	None

						Male							Fem	ale			
		Le	ft Femor	ral Robus	ticity	Righ	nt Femor	al Robus	ticity	L	eft Femo	oral Rob	usticity	Rig	sht Femor	al Robu	sticity
Facet	Sample	C Coeffi cient	Ρ	N	Strength of Correlat ion	C Coefficient	Ρ	N	Strength of Correlation	C Coeffi cient	Ρ	N	Strength of Correlation	C Coeffi cient	Ρ	N	Strength of Correlati on
	SBC	-0.39	0.19	13	Medium	-0.23	0.48	12	Weak	-0.37	0.10	21	Medium	-0.39	0.11	18	Medium
T11LI	SBL	0.21	0.34	23	Weak	0.25	0.27	21	Weak	0.25	0.41	13	Weak	0.06	0.84	12	None
	AS	-0.12	0.75	10	Weak	-0.04	0.90	12	None	0.33	0.32	11	Medium	0.39	0.20	12	Medium
	SBC	-0.08	0.79	13	None	0.00	1.00	12	None	-0.42	0.06	21	Medium	-0.42	0.08	18	Medium
T11RI	SBL	-0.02	0.92	23	None	0.05	0.83	22	None	0.10	0.74	13	Weak	-0.06	0.86	12	None
	AS	0.13	0.71	10	Weak	0.22	0.48	12	Weak	-0.04	0.91	11	None	0.02	0.95	12	None
	SBC	-0.39	0.19	13	Medium	-0.37	0.24	12	Medium	-0.33	0.17	19	Medium	-0.38	0.14	16	Medium
T12LS	SBL	0.19	0.40	21	Weak	0.10	0.68	20	None	0.18	0.55	13	Weak	0.12	0.72	12	Weak
	AS	0.28	0.46	9	Weak	0.33	0.32	11	Medium	0.44	0.17	11	Medium	0.50	0.10	12	Strong
	SBC	-0.39	0.19	13	Medium	-0.15	0.64	12	Weak	-0.52	0.02	19	Strong	-0.56	0.03	16	Strong
T12RS	SBL	0.12	0.61	21	Weak	0.06	0.79	20	None	0.16	0.59	13	Weak	0.03	0.93	12	None
	AS	-0.15	0.69	9	Weak	-0.01	0.97	11	None	0.28	0.41	11	Weak	0.34	0.28	12	Medium
	SBC	-0.10	0.75	13	Weak	0.24	0.45	12	Weak	-0.19	0.44	19	Weak	-0.22	0.41	16	Weak
T12LI	SBL	-0.13	0.57	22	Weak	-0.12	0.59	21	Weak	0.78	0.00	13	Strong	0.39	0.21	12	Medium
	AS	0.39	0.30	9	Medium	0.42	0.19	11	Medium	-0.18	0.62	10	Weak	-0.15	0.67	11	Weak

					М	ale							Fem	ale			
		Le	ft Femora	l Robustio	city	R	ight Fem	oral Rob	usticity	L	eft Femo	oral Rob	ousticity	Rig	ght Femor	al Robu	sticity
Facet	Sample	C Coeffici ent	Ρ	N	Strength of Correlat ion	C Coeffic ient	Р	N	Strength of Correlation	C Coeffi cient	Ρ	N	Strength of Correlation	C Coeffi cient	Р	N	Strength of Correlati on
	SBC	0.11	0.72	13	Weak	0.55	0.07	12	Strong	-0.16	0.53	19	Weak	-0.05	0.85	16	None
T12RI	SBL	-0.17	0.45	22	Weak	-0.16	0.48	21	Weak	0.50	0.08	13	Strong	0.18	0.59	12	Weak
	AS	-0.12	0.76	9	Weak	-0.16	0.67	10	Weak	0.62	0.04	11	Strong	0.65	0.02	12	Strong
	SBC	-0.13	0.68	12	Weak	-0.20	0.53	12	Weak	-0.11	0.64	20	Weak	-0.09	0.72	17	None
L1LS	SBL	-0.21	0.35	23	Weak	-0.19	0.38	23	Weak	0.95	0.00	12	Strong	0.43	0.16	12	Medium
	AS	0.22	0.53	10	Weak	0.00	0.99	13	None	0.01	0.97	11	None	0.17	0.60	12	Weak
	SBC	0.11	0.73	12	Weak	0.38	0.22	12	Medium	-0.23	0.32	20	Weak	-0.26	0.31	17	Weak
L1RS	SBL	-0.13	0.55	22	Weak	-0.10	0.65	22	Weak	0.93	0.00	12	Strong	0.27	0.40	12	Weak
	AS	0.50	0.21	8	Strong	0.09	0.80	11	None	0.15	0.66	11	Weak	0.52	0.09	12	Strong
	SBC	0.31	0.32	12	Medium	-0.02	0.95	12	None	-0.42	0.06	20	Medium	-0.35	0.17	17	Medium
L1LI	SBL	-0.22	0.31	23	Weak	-0.39	0.07	23	Medium	0.02	0.96	12	None	-0.34	0.28	12	Medium
	AS	-0.48	0.19	9	Medium	-0.54	0.07	12	Strong	-0.10	0.75	12	Weak	0.02	0.95	13	None
	SBC	0.19	0.55	12	Weak	-0.39	0.20	12	Medium	-0.43	0.07	19	Medium	-0.37	0.14	17	Medium
L1RI	SBL	-0.09	0.69	22	None	-0.05	0.83	22	None	-0.68	0.01	12	Strong	-0.11	0.73	12	Weak
	AS	-0.32	0.37	10	Medium	-0.26	0.39	13	Weak	0.22	0.50	12	Weak	0.31	0.30	13	Medium

			Left Femoral Robusticity							Female							
		Le	eft Femoral	Robusti	city		Right Fem	oral Robust	icity	l	eft Femo	ral Robust	ticity	Ri	ght Femor	al Robust	icity
Facet	Sample	C Coeffi cient	Ρ	Ν	Strength of Correlat ion	C Coeffi cient	Ρ	Ν	Strength of Correlatio n	C Coeffi cient	Ρ	Ν	Strength of Correlatio n	C Coeffi cient	Ρ	N	Strength of Correlat ion
	SBC	-0.09	0.76	13	None	-0.25	0.43	12	Weak	-0.31	0.17	21	Medium	-0.16	0.52	19	Weak
L2LS	SBL	0.12	0.59	23	Weak	-0.05	0.83	23	None	-0.30	0.35	12	Medium	-0.30	0.34	12	Medium
	AS	-0.09	0.79	11	None	-0.19	0.52	14	Weak	-0.05	0.89	10	None	0.27	0.42	11	Weak
	SBC	-0.10	0.74	13	Weak	-0.73	0.01	12	Strong	-0.19	0.41	21	Weak	-0.29	0.23	19	Weak
L2RS	SBL	-0.13	0.56	21	Weak	-0.18	0.43	21	Weak	-0.60	0.04	12	Strong	-0.29	0.38	11	Weak
	AS	0.00	0.99	11	None	-0.07	0.83	14	None	0.45	0.19	10	Medium	0.53	0.09	11	Strong
	SBC	-0.06	0.84	13	None	-0.23	0.47	12	Weak	-0.33	0.14	21	Medium	-0.23	0.35	19	Weak
L2LI	SBL	0.02	0.91	23	None	0.04	0.85	23	None	-0.65	0.03	11	Strong	-0.32	0.34	11	Medium
	AS	-0.04	0.92	10	None	-0.17	0.57	13	Weak	0.27	0.52	8	Weak	0.38	0.32	9	Medium
	SBC	0.40	0.18	13	Medium	0.23	0.47	12	Weak	-0.20	0.39	21	Weak	-0.46	0.05	19	Medium
L2RI	SBL	0.01	0.97	23	None	-0.20	0.35	23	Weak	-0.54	0.07	12	Strong	-0.25	0.44	12	Weak
	AS	0.02	0.96	10	None	-0.16	0.59	13	Weak	0.54	0.13	9	Strong	0.48	0.16	10	Medium
	SBC	0.31	0.30	13	Medium	0.15	0.65	12	Weak	0.06	0.80	20	None	-0.23	0.37	17	Weak
L3LS	SBL	0.08	0.74	21	None	-0.05	0.82	21	None	-0.40	0.18	13	Medium	-0.03	0.92	11	None
	AS	-0.25	0.44	12	Weak	-0.36	0.21	14	Medium	0.06	0.85	13	None	0.12	0.68	14	Weak

			Male										Female				
			Left Femor	al Robus	sticity	Rig	ht Femc	oral Robusti	city	L	eft Femoral R	obusticity		Right Femoral Rob	ousticity		
Facet	Sample	C Coeffi cient	Ρ	N	Strength of Correlatio n	C Coefficien t	Р	N	Streng th of Correl ation	C Coeffi cient	Ρ	N	Strength of Correlation	C Coefficient	Р	N	Strength of Correlatio n
	SBC	0.19	0.54	13	Weak	0.08	0.80	12	None	0.12	0.61	20	Weak	-0.11	0.66	17	Weak
L3RS	SBL	-0.21	0.36	22	Weak	-0.31	0.17	21	Mediu m	-0.55	0.05	13	Strong	-0.10	0.77	11	Weak
	AS	0.40	0.22	11	Medium	0.09	0.75	14	None	0.03	0.93	13	None	0.14	0.62	14	Weak
	SBC	0.12	0.70	13	Weak	0.52	0.08	12	Strong	-0.26	0.26	20	Weak	0.00	0.99	17	None
L3LI	SBL	-0.14	0.53	21	Weak	0.01	0.95	21	None	-0.42	0.16	13	Medium	0.02	0.94	12	None
	AS	-0.13	0.69	12	Weak	-0.27	0.32	15	Weak	0.20	0.52	13	Weak	0.11	0.71	14	Weak
	SBC	0.56	0.04	13	Strong	0.39	0.21	12	Mediu m	-0.22	0.34	20	Weak	-0.08	0.75	17	None
L3RI	SBL	0.27	0.26	19	Weak	0.11	0.63	20	Weak	-0.50	0.09	13	Strong	-0.27	0.40	12	Weak
	AS	-0.19	0.58	11	Weak	-0.28	0.32	14	Weak	0.18	0.57	12	Weak	0.13	0.66	13	Weak
	SBC	-0.03	0.92	13	None	0.40	0.20	12	Mediu m	-0.21	0.37	21	Weak	-0.09	0.73	18	None
L4LS	SBL	-0.04	0.87	19	None	0.07	0.78	18	None	0.17	0.63	10	Weak	-0.33	0.32	11	Medium
	AS	-0.37	0.27	11	Medium	-0.31	0.29	14	Mediu m	0.16	0.62	12	Weak	0.12	0.69	13	Weak
	SBC	0.24	0.42	13	Weak	0.20	0.54	12	Weak	-0.26	0.26	21	Weak	-0.02	0.92	18	None
L4RS	SBL	-0.19	0.42	20	Weak	-0.29	0.22	19	Weak	-0.18	0.62	10	Weak	0.01	0.98	10	None
	AS	-0.02	0.95	11	None	-0.16	0.57	14	Weak	-0.14	0.66	12	Weak	-0.14	0.65	13	Weak

					Ν	1ale				Female							
		L	eft Femo	oral Rob	usticity	I	Right Fem	ioral Rob	usticity	L	eft Femor	al Robusti	city	R	ight Femoral I	Robustic	ity
Facet	Sample	C Coeffi cient	Ρ	Ν	Strength of Correlation	C Coeffi cient	Ρ	N	Strength of Correlation	C Coeffi cient	Ρ	N	Strength of Correlat ion	C Coeffici ent	Ρ	N	Strength of Correlati on
	SBC	-0.22	0.47	13	Weak	0.13	0.69	12	Weak	-0.51	0.02	21	Strong	-0.11	0.68	18	Weak
L4LI	SBL	-0.10	0.70	19	Weak	-0.04	0.86	18	None	0.27	0.45	10	Weak	-0.01	0.97	11	None
	AS	-0.50	0.12	11	Strong	-0.52	0.05	14	Strong	-0.32	0.33	11	Medium	-0.38	0.22	12	Medium
	SBC	-0.36	0.23	13	Medium	-0.04	0.90	12	None	-0.58	0.01	21	Strong	-0.17	0.51	18	Weak
L4RI	SBL	0.02	0.95	20	None	0.01	0.95	19	None	0.47	0.17	10	Medium	-0.22	0.51	11	Weak
	AS	-0.37	0.26	11	Medium	-0.27	0.37	13	Weak	-0.31	0.39	10	Medium	-0.30	0.37	11	Medium
	SBC	-0.44	0.18	11	Medium	0.15	0.65	11	Weak	-0.53	0.02	18	Strong	-0.06	0.81	16	None
L5LS	SBL	0.11	0.64	22	Weak	0.10	0.65	22	Weak	-0.55	0.07	12	Strong	-0.32	0.33	11	Medium
	AS	0.30	0.48	8	Medium	0.22	0.52	11	Weak	0.05	0.88	11	None	0.04	0.89	12	None
	SBC	0.22	0.51	11	Weak	-0.05	0.87	11	None	-0.46	0.05	18	Medium	-0.20	0.46	16	Weak
L5RS	SBL	0.18	0.46	20	Weak	0.22	0.35	20	Weak	-0.44	0.15	12	Medium	0.12	0.73	11	Weak
	AS	-0.07	0.88	8	None	-0.02	0.95	11	None	0.00	0.99	11	None	-0.09	0.79	12	None
	SBC	0.13	0.70	11	Weak	-0.06	0.86	11	None	0.10	0.68		Weak	0.17	0.54	16	Weak
L5LI	SBL	0.17	0.45	22	Weak	0.17	0.45	22	Weak	-0.31	0.33	12	Medium	0.30	0.37	11	Medium
	AS	-0.59	0.17	7	Strong	-0.45	0.19	10	Medium	-0.07	0.84	10	None	0.03	0.93	11	None

					N	Male							F	emale			
		L	eft Fem	oral Rob	ousticity		Right Fen	noral Rob	ousticity	I	eft Femor	al Robust	icity	R	ight Femoral I	Robustic	ity
Facet	Sample	C Coeffi cient	Ρ	N	Strength of Correlation	C Coeffi cient	Р	N	Strength of Correlation	C Coeffi cient	Ρ	N	Strength of Correlat ion	C Coeffici ent	Р	N	Strength of Correlati on
	SBC	-0.19	0.57	11	Weak	0.15	0.66	11	Weak	0.16	0.53	18	Weak	0.19	0.48	16	Weak
L5RI	SBL	0.10	0.68	21	None	0.05	0.85	21	None	-0.31	0.32	12	Medium	0.30	0.38	11	Medium
	AS	-0.45	0.26	8	Medium	-0.42	0.20	11	Medium	-0.38	0.27	10	Medium	-0.32	0.34	11	Medium

Appendix C-7 Correlation between femoral robusticity and facet angle continued

Facet	Sample		Ma	ale			Fen	nale	
		Continuity Coefficient	Р	Phi	Phi correlation Strength	Continuity Coefficient	Ρ	Phi	Phi correlation Strength
	SBC	1.02	0.31	0.38	Medium	0.00	0.96	-0.09	Weak
C1S	SBL	3.33	0.07	0.56	Strong	0.00	1.00	-0.07	None
	AS	0.31	0.58	0.67	Strong				
	SBC	1.07	0.30	0.45	Medium	0.00	1.00	-0.01	None
C1I	SBL	0.00	1.00	0.09	None	0.00	1.00	0.14	Weak
	AS	0.00	1.00	-0.41	Medium				
	SBC	0.01	0.91	-0.19	Weak	0.00	1.00	-0.07	None
C2S	SBL	0.00	1.00	0.06	None	0.55	0.46	0.33	Medium
	AS	0.00	1.00	0.58	Strong				
	SBC	0.00	1.00	-0.04	None	0.13	0.72	-0.14	Weak
C2I	SBL	0.00	1.00	0.01	None	5.66	0.02	-0.76	Strong
	AS	0.31	0.58	-0.67	Strong				
	SBC	0.00	1.00	0.17	Weak	0.17	0.68	-0.17	Weak
C3S	SBL	0.65	0.42	0.18	Weak	0.25	0.62	-0.25	Weak
	AS	0.00	1.00	0.33	Medium				
	SBC	0.00	1.00	-0.27	Weak	0.05	0.82	0.14	Weak
C3I	SBL	1.32	0.25	-0.36	Medium	0.20	0.66	0.24	Weak
	AS	0.00	1.00	0.33	Medium				
	SBC	0.05	0.83	0.22	Weak	0.44	0.51	-0.26	Weak
C4S	SBL	0.01	0.91	0.16	Weak	0.14	0.71	0.20	Weak
	AS	0.05	0.82	0.61	Strong				
	SBC	0.00	1.00	0.14	Weak	0.00	1.00	0.05	None
CC4I	SBL	1.09	0.60	-0.38	Medium	0.00	1.00	-0.01	None
	AS	0.00	1.00	0.25	Weak				
	SBC	0.00	1.00	0.08	None	2.85	0.09	-0.40	Medium
C5S	SBL	0.00	1.00	-0.03	None	0.00	1.00	0.01	None
	AS	0.05	0.82	0.61	Strong				

Appendix C Facet Angle

			N	/lale			Fe	emale	
Facet	Sample	Continuity Coefficient	Ρ	Phi	Phi correlation Strength	Continuity Coefficient	Ρ	Phi	Phi correlation Strength
	SBC	0.03	0.87	0.16	Weak	0.15	0.70	0.23	Weak
C5I	SBL	0.32	0.57	0.23	Weak	0.00	1.00	-0.14	Weak
	AS	0.00	1.00	0.25	Weak				
	SBC	0.00	1.00	0.07	None	0.03	0.87	0.13	Weak
C6S	SBL	0.00	1.00	0.09	None	0.00	1.00	-0.12	Weak
	AS	0.05	0.82	0.61	Strong				
	SBC	1.24	0.27	-0.45	Medium	0.00	1.00	0.07	None
C6I	SBL	1.06	0.30	-0.36	Medium	0.00	1.00	0.10	None
	AS	0.00	1.00	-0.41	Medium				
	SBC	0.00	1.00	-0.14	Weak	0.01	0.93	-0.10	Weak
C7S	SBL	0.54	0.46	-0.29	Weak	0.00	1.00	0.05	None
	AS	0.00	1.00	-0.41	Medium				
	SBC	3.15	0.08	0.63	Strong	0.00	0.97	0.12	Weak
C7!	SBL	0.01	0.94	0.13	Weak	0.35	0.85	0.18	Weak
	AS	0.00	1.00	0.50	Strong				
	SBC	1.64	0.20	0.47	Medium	0.00	1.00	0.06	None
T1S	SBL	3.00	0.58	-0.22	Weak	1.43	0.23	0.40	Medium
	AS	0.32	0.57	0.47	Medium	0.00	1.00	0.41	Medium
	SBC	0.00	1.00	0.00	None	0.00	0.96	0.11	Weak
T1I	SBL	0.00	1.00	0.08	None	1.82	0.18	0.44	Medium
	AS	0.00	1.00	0.26	Weak				
	SBC	0.62	0.43	0.39	Medium	0.00	1.00	-0.05	None
T2S	SBL	0.00	1.00	0.06	None	0.19	0.67	-0.21	Weak
	AS	0.00	1.00	0.26	Weak				
	SBC	2.12	0.15	0.51	Strong	0.00	1.00	-0.07	None
T2I	SBL	0.00	1.00	0.02	None	0.75	0.39	0.36	Medium
	AS	0.00	1.00	-0.32	Medium	0.00	1.00	0.50	Strong

Male Female Continui Phi Facet Sample Continuity Phi correlatio tv Coefficien Ρ Phi Ρ correlatio Phi Coeffici n n Strength t Strength ent SBC 0.06 0.81 0.22 Weak 0.02 0.89 0.10 None T3S 0.00 1.00 -0.03 0.00 SBL None 1.00 -0.14 Weak AS 0.00 1.00 0.15 Weak 0.00 1.00 -0.58 Strong SBC 0.09 0.77 0.20 Weak T3I SBL 0.00 1.00 -0.15 Weak 1.50 0.22 -0.43 Medium Weak 0.00 1.00 -0.29 0.00 1.00 0.00 AS None Weak SBC 0.83 0.36 0.36 Medium 0.61 0.43 0.23 T4S SBL 0.01 0.92 -0.17 Weak 0.85 0.36 0.42 Medium AS 0.00 1.00 0.07 None 0.03 0.58 0.67 Strong Weak SBC 0.00 1.00 0.24 0.13 0.71 0.22 Weak T4I SBL 0.00 1.00 0.02 None 0.00 0.96 0.18 Weak AS 1.74 0.57 -0.47 Medium 0.00 1.00 0.00 None SBC 0.00 1.00 -0.14 Weak 0.13 0.72 -0.16 Weak T5S 0.00 0.00 0.00 SBL 1.00 0.00 1.00 None None AS 1.47 0.23 -1.00 0.00 1.00 0.50 Strong Strong SBC 0.35 0.56 0.32 Medium 0.00 1.00 0.14 Weak T5I SBL 0.26 -0.37 Medium 0.00 0.07 1.25 1.00 None Medium 0.08 0.78 0.49 Medium 0.00 -0.33 AS 1.00 SBC 0.00 0.96 0.02 0.88 0.13 Weak 0.12 Weak T6S SBL 0.00 1.00 -0.08 None 0.00 1.00 0.12 Weak AS 0.00 1.00 -0.58 0.00 1.00 -0.58 Strong Strong Weak SBC 0.00 1.00 0.13 Weak 0.53 0.47 0.26 T6I SBL 0.00 1.00 0.07 None 0.00 1.00 0.23 Weak AS 0.19 0.67 -1.00 Strong SBC 0.00 0.07 0.09 -0.40 Medium 1.00 None 2.87 T7S SBL 0.01 0.91 0.13 Weak 0.49 0.49 0.31 Medium AS 0.00 1.00 0.00 None 0.00 1.00 -0.41 Medium

Appendix C Facet Angle

			M	ale			Fe	emale	
Facet	Sample	Continuity Coefficien t	Ρ	Phi	Phi correlation Strength	Continui ty Coeffici ent	Ρ	Phi	Phi correlatio n Strength
	SBC	0.02	0.90	-0.29	Weak	0.21	0.65	-0.20	Weak
T7I	SBL					1.07	0.30	-0.44	Medium
	AS	0.00	1.00	-0.17	None	0.00	1.00	0.17	Weak
	SBC	0.00	1.00	0.06	None	0.64	0.42	0.23	Weak
T8S	SBL	1.49	0.22	-0.35	Medium	0.00	1.00	0.00	None
	AS	0.02	0.88	-0.47	Medium				
	SBC					0.00	1.00	0.05	None
Т8І	SBL	1.44	0.23	0.37	Medium	0.00	1.00	0.15	Weak
	AS	0.37	0.55	-0.55	Strong	0.00	1.00	-0.41	Medium
	SBC	1.86	0.17	-0.54	Strong	0.09	0.76	0.13	Weak
T9S	SBL	0.21	0.65	0.21	Weak	0.00	1.00	0.15	Weak
	AS	0.00	1.00	0.45	Medium	0.05	0.82	0.61	Strong
	SBC	0.00	1.00	0.13	Weak	0.00	1.00	0.03	None
T9!	SBL	0.15	0.70	0.19	Weak	2.40	0.12	0.52	Strong
	AS	0.00	1.00	0.00	None	0.00	1.00	-0.41	Medium
	SBC	0.85	0.36	-0.42	Medium	0.62	0.43	-0.23	Weak
T10S	SBL	1.16	0.28	0.36	Medium	0.08	0.78	0.25	Weak
	AS	0.00	1.00	0.00	None	0.00	1.00	-0.33	Medium
	SBC	0.00	1.00	0.16	Weak	0.00	1.00	0.05	None
T10I	SBL	0.00	1.00	-0.10	None	1.38	0.24	0.40	Medium
	AS	0.00	1.00	0.33	Medium	0.00	1.00	0.17	Weak
	SBC	0.00	1.00	0.05	None	0.09	0.77	-0.13	Weak
T11S	SBL	0.00	1.00	-0.11	Weak	0.00	0.95	-0.15	Weak
	AS	2.67	0.10	1.00	Strong	0.00	1.00	0.17	Weak
	SBC	0.00	1.00	-0.07	None	0.17	0.68	-0.15	Weak
T11I	SBL	0.00	1.00	-0.02	None	0.00	1.00	0.06	None
	AS	0.37	0.55	-0.55	Strong	0.00	1.00	-0.41	Medium

Male Female Continui Phi Continuity Facet Sample Phi correlatio tv Ρ Phi Coefficien Ρ Phi correlation Coeffici n Strength t Strength ent SBC 0.02 0.90 0.20 Weak 0.00 1.00 -0.05 None T12S SBL 0.00 0.08 0.02 Weak 1.00 None 0.90 0.14 AS 0.75 0.39 0.71 0.05 0.82 0.61 Strong Strong SBC 5.69 0.02 -0.75 0.97 0.32 0.27 Weak Strong T12I SBL 0.20 -0.38 Medium 0.00 1.00 -0.03 None 1.68 AS 0.00 1.00 0.25 Weak 0.00 1.00 0.41 Medium SBC 0.09 0.77 0.19 0.04 0.84 0.12 Weak Weak L1S SBL 0.11 0.74 0.20 Weak 0.00 1.00 -0.05 None AS 0.11 0.74 -0.42 Medium 0.00 1.00 0.41 Medium SBC 0.70 0.40 0.32 Medium 0.00 1.00 -0.07 None L1I SBL 0.00 1.00 0.06 None 0.02 0.90 0.16 Weak AS 0.11 0.74 -0.42 Medium 0.00 1.00 0.33 Medium SBC 0.02 0.90 0.16 Weak 0.58 0.45 0.23 Weak L2S 0.79 SBL 0.07 0.17 Weak 2.85 0.09 0.53 Strong AS 0.89 0.35 -0.60 0.31 0.58 0.67 Strong Strong SBC 0.00 0.96 0.13 2.18 0.14 -0.35 Medium Weak L2I SBL 0.03 0.85 0.00 Weak -0.15 Weak 1.00 -0.13 Medium 0.67 -0.45 0.00 1.00 -0.17 Weak AS 0.18 SBC -0.52 1.20 -0.29 2.52 0.11 Strong 0.27 Weak L3S SBL 0.01 0.91 -0.15 Weak 0.55 0.46 0.33 Medium AS 0.00 1.00 0.26 Weak 0.00 1.00 -0.41 Medium SBC 0.14 0.71 0.20 0.00 1.00 0.05 Weak None L3I SBL 0.02 0.88 0.15 Weak 0.00 1.00 -0.08 None AS 0.00 1.00 -0.45 Weak 0.05 0.82 -0.61 Strong SBC 0.12 -0.48 Medium 0.97 0.32 0.27 Weak 2.46 L4S SBL 0.00 1.00 0.13 Weak 1.31 0.25 -0.47 Medium AS 0.75 0.39 -0.71 Strong 0.31 0.58 -0.67 Strong

Appendix C Facet Angle

			Ν	ſale			Fe	emale	
Facet	Sample	Continui ty Coeffici ent	Ρ	Phi	Phi correlation Strength	Continuity Coefficien t	Ρ	Phi	Phi correlatio n Strength
	SBC	0.00	0.96	0.18	Weak	0.61	0.43	0.23	Weak
L4I	SBL	0.00	1.00	0.09	None	0.04	0.84	-0.24	Weak
	AS	0.00	1.00	-0.26	Weak	0.00	1.00	-0.17	Weak
	SBC	0.02	0.90	-0.16	Weak	0.00	1.00	0.02	None
L5S	SBL	0.00	1.00	-0.09	None	0.51	0.47	-0.29	Weak
	AS	0.75	0.39	0.71	Strong	0.00	1.00	0.33	Medium
	SBC	0.00	1.00	0.00	None	0.00	1.00	0.02	None
L5I	SBL	0.00	1.00	-0.09	None	0.24	0.27	-0.45	Medium
	AS	0.09	0.76	0.50	Strong				

Male Female Facet Sample Correlation Strength of Correlation Strength of р Ν р Ν Coefficient Correlation Coefficient Correlation SBC 0.39 0.1 19 Medium 21 C1LS SBL 0.09 0.72 19 None 15 AS 0.22 0.54 10 Weak 6 0.66 SBC -0.11 19 Weak 0.15 0.5 Weak 22 C1RS SBL 0.4 0.09 19 Medium 15 Weak AS 0.26 0.42 12 6 Weak SBC 0.05 0.86 18 0.55 21 None -0.14 SBL 0.06 C1LI -0.12 0.62 19 Weak 0.83 15 None AS -0.7 0.04 9 5 Strong 22 SBC 19 C1RI SBL 15 0.2 0.41 19 Weak -0.28 0.31 Weak AS -0.23 0.48 12 Weak 6 SBC -0.15 0.52 20 Weak 0.13 0.56 23 Weak C2LS SBL 0.48 0.03 21 Medium 0 1 14 None AS 0.38 0.29 10 Medium -0.61 0.14 7 Strong SBC -0.32 0.16 21 Medium -0.23 0.29 24 Weak C2RS SBL 0.46 0.04 21 Medium -0.04 0.91 14 None -0.08 7 AS 0.84 10 None SBC 0.35 0.15 18 Medium 0.14 0.53 23 Weak C2LI SBL 0.28 0.23 20 Weak -0.46 0.11 13 Medium AS -0.6 0.09 9 Strong -0.66 0.16 6 Strong SBC 0.51 24 18 -0.14 Weak C2RI SBL 0.34 0.14 21 Medium -0.45 0.11 14 Medium AS -0.18 0.62 10 Weak 0.13 0.81 6 Weak SBC -0.24 0.44 13 Weak -0.1 0.66 23 Weak C3LS SBL 0.01 0.97 22 0.06 0.83 15 None None AS -0.49 0.07 14 Medium -0.12 0.8 7 Weak SBC 0.26 0.4 13 Weak 0.13 0.56 23 Weak C3RS SBL 0.43 0.04 22 Medium 0 1 14 None 7 AS -0.44 0.12 14 Medium -0.22 0.63 Weak SBC 0.53 0.07 13 -0.37 0.1 21 Medium Strong C3LI SBL 0.35 Weak 0.22 0.44 Weak -0.21 21 15 0.92 AS 0.03 14 None -0.12 0.8 7 Weak SBC Weak 13 -0.26 0.24 22 C3RI SBL -0.19 0.4 22 Weak -0.28 0.34 14 Weak AS 0.61 0.27 0.56 7 -0.16 12 Weak Weak

Appendix C-9 Correlation between eburnation and facet angle

Appendix C-9 Correlation	between eburnatio	n and facet angle	continued
Appendix e 5 con clation	Settleen esamatio	in and racet angle	continueu

			Male				Fema	le	
Facet	Sample	Correlation Coefficient	р	N	Strength of Correlation	Correlation Coefficient	р	N	Strength of Correlation
	SBC	0.09	0.76	14	None	0.06	0.8	21	None
C4LS	SBL			24				17	
	AS	0.31	0.32	12	Medium	0.11	0.84	6	Weak
	SBC	-0.39	0.19	13	Medium	-0.08	0.74	20	None
C4RS	SBL	-0.01	0.97	25	None	-0.08	0.77	16	None
	AS	0.15	0.63	12	Weak	0.11	0.84	6	Weak
	SBC	-0.5	0.08	13	Strong	-0.1	0.67	21	Weak
C4LI	SBL	0.17	0.41	25	Weak	0.41	0.1	17	Medium
	AS	0.39	0.24	11	Medium	0.84	0.04	6	Strong
	SBC	-0.31	0.33	12	Medium	-0.1	0.68	21	Weak
C4RI	SBL	0.07	0.73	25	None	0.31	0.24	16	Medium
	AS	0.67	0.02	11	Strong	0.66	0.15	6	Strong
	SBC	-0.02	0.93	20	None	0.07	0.75	26	None
C5LS	SBL	0.48	0.02	24	Medium	0.51	0.02	20	Strong
	AS	0.81	0.02	8	Strong	-0.41	0.49	5	Medium
	SBC	0.12	0.63	20	Weak	0.19	0.36	26	Weak
C5RS	SBL	0.15	0.46	25	Weak	0.34	0.14	20	Medium
	AS	0.26	0.53	8	Weak	-0.78	0.12	5	Strong
	SBC			20		-0.19	0.36	26	Weak
C5LI	SBL	-0.18	0.41	24	Weak			20	
	AS	0.06	0.9	8	None	-0.89	0.04	5	Strong
	SBC	0.09	0.7	20	None			26	
C5RI	SBL	0	0.99	24	None	0.18	0.45	20	Weak
	AS	0.26	0.54	8	Weak	-0.11	0.86	5	Weak
	SBC	0.43	0.06	20	Medium	-0.29	0.17	24	Weak
C6LS	SBL	0.46	0.02	25	Medium	0.42	0.08	18	Medium
	AS	0.32	0.41	9	Medium	-0.22	0.63	7	Weak
	SBC	0.27	0.24	20	Weak	0.27	0.2	25	Weak
C6RS	SBL	0.01	0.97	24	None	0.12	0.64	18	Weak
	AS	0.34	0.38	9	Medium	-0.79	0.04	7	Strong
	SBC	-0.29	0.21	20	Weak			24	
C6LI	SBL	0.2	0.32	26	Weak	0.38	0.12	18	Medium
	AS	0	1	9	None	-0.36	0.43	7	Medium
	SBC			20				25	
C6RI	SBL	-0.19	0.37	25	Weak			19	
	AS	0.46	0.22	9	Medium	-0.62	0.14	7	Strong
	SBC	0.12	0.63	20	Weak	-0.21	0.29	28	Weak
C7LS	SBL	-0.27	0.19	26	Weak	-0.07	0.77	21	None
	AS	0.14	0.72	9	Weak	0.05	0.94	5	None

Appendix C-9 Correlation between eburnation and facet angle continued

Facet	Sample		Male				Female		
		Correlation Coefficient	р	N	Strength of Correlatio n	Correlation Coefficient	р	N	Strength of Correlati on
C7RS	SBC	-0.02	0.93	20	None			27	
	SBL	0	1	27	None	-0.04	0.87	21	None
	AS	0.61	0.06	10	Strong	-0.21	0.74	5	Weak
C7LI	SBC	0.3	0.21	19	Medium	0.31	0.11	28	Medium
	SBL	-0.14	0.46	28	Weak	0.37	0.1	21	Medium
	AS	-0.2	0.58	10	Weak	-0.71	0.18	5	Strong
C7RI	SBC			19				25	
	SBL	-0.27	0.17	28	Weak	0.38	0.2	21	Medium
	AS	0.27	0.45	10	Weak	-0.71	0.18	5	Strong
T1LS	SBC	0.26	0.32	17	Weak	-0.04	0.83	29	None
	SBL	-0.01	0.96	30	None	0.35	0.1	21	Medium
	AS	0.11	0.71	14	Weak	0.34	0.37	9	Medium
T1RS	SBC	0.17	0.49	19	Weak	0.14	0.47	30	Weak
	SBL	0.01	0.95	29	None	0.13	0.58	20	Weak
	AS	0.03	0.93	13	None	0.29	0.42	10	Weak
T1LI	SBC			20		0.09	0.65	31	None
	SBL	0.11	0.55	31	Weak	-0.02	0.94	21	None
	AS	0.25	0.42	13	Weak	0.14	0.7	10	Weak
T1RI	SBC			20		-0.05	0.82	29	None
	SBL			31		-0.07	0.75	21	None
	AS	-0.03	0.93	13	None	-0.23	0.53	10	Weak
T2LS	SBC			23		0.19	0.34	28	Weak
	SBL	-0.1	0.6	32	Weak	-0.32	0.17	20	Medium
	AS	0.36	0.25	12	Medium	-0.39	0.18	13	Medium
T2RS	SBC	0.27	0.21	23	Weak	-0.31	0.1	29	Medium
	SBL	0.09	0.63	33	None			19	
	AS	0.42	0.18	12	Medium	-0.13	0.68	13	Weak
T2LI	SBC			23		0.29	0.12	29	Weak
	SBL			33				20	
	AS	0.6	0.04	12	Strong	-0.17	0.58	13	Weak
T2RI	SBC	0.16	0.46	23	Weak	0.14	0.44	31	Weak
	SBL	0.34	0.06	31	Medium			19	
	AS	0.38	0.22	12	Medium	-0.43	0.15	13	Medium
T3LS	SBC	0.08	0.73	21	None	0.16	0.43	26	Weak
	SBL	0.19	0.4	22	Weak	0.31	0.2	19	Medium
	AS	0.1	0.73	14	Weak	0.07	0.81	14	None

Appendix C-9 Correlation between eburnation and facet angle continued

Facet	Sample	Male				Female			
		Correlation Coefficient	р	N	Strength of Correlatio n	Correlation Coefficient	р	N	Strength of Correlati on
T3RS	SBC	0.16	0.48	22	Weak	0.21	0.3	27	Weak
	SBL	0.11	0.59	25	Weak	0.17	0.46	21	Weak
	AS	0.29	0.3	15	Weak	-0.42	0.14	14	Medium
T3LI	SBC			22		-0.16	0.44	27	Weak
	SBL			25		-0.19	0.42	21	Weak
	AS	0.49	0.09	13	Medium	0.16	0.61	13	Weak
T3RI	SBC	-0.29	0.2	22	Weak	-0.03	0.87	25	None
	SBL	0.14	0.51	25	Weak	0.34	0.14	20	Medium
	AS	0.58	0.03	14	Strong	0.31	0.28	14	Medium
T4LS	SBC	0.08	0.75	19	None	-0.11	0.6	24	Weak
	SBL	0.17	0.41	26	Weak	-0.15	0.56	18	Weak
	AS	0.01	0.98	13	None	-0.11	0.7	14	Weak
T4RS	SBC	-0.21	0.4	19	Weak	0.23	0.29	24	Weak
	SBL	0.27	0.2	25	Weak	0.35	0.14	19	Medium
	AS	0.32	0.28	13	Medium	-0.18	0.53	14	Weak
T4LI	SBC	0.02	0.93	19	None	-0.07	0.74	24	None
	SBL			25		-0.27	0.29	18	Weak
	AS	0.31	0.3	13	Medium	0.12	0.71	13	Weak
T4RI	SBC	0.04	0.86	19	None	0.3	0.14	25	Medium
	SBL	0.26	0.21	25	Weak			16	
	AS	0.36	0.23	13	Medium	0.18	0.55	13	Weak
T5LS	SBC	-0.04	0.86	19	None	0.14	0.49	25	Weak
	SBL	-0.35	0.11	22	Medium	-0.26	0.28	19	Weak
	AS	0	1	14	None	0.33	0.36	10	Medium
T5RS	SBC	-0.37	0.12	19	Medium	-0.34	0.1	24	Medium
	SBL	-0.2	0.39	21	Weak	0.09	0.72	19	None
	AS	0.11	0.71	14	Weak	-0.07	0.84	10	None
T5LI	SBC	-0.15	0.56	17	Weak	0.33	0.14	22	Medium
	SBL			20		-0.22	0.37	19	Weak
	AS	0.5	0.12	11	Strong	0.17	0.64	10	Weak
T5RI	SBC	-0.36	0.15	18	Medium	0.12	0.6	21	Weak
	SBL			23				19	
	AS	0.18	0.61	11	Weak	0.41	0.24	10	Medium
T6LS	SBC	0.34	0.19	17	Medium	-0.07	0.73	25	None
	SBL	0.3	0.16	24	Medium	-0.13	0.6	19	Weak
	AS	0.57	0.07	11	Strong	0.21	0.55	10	Weak
T6RS	SBC	0.05	0.86	17	None	0.01	0.98	27	None
	SBL	0.07	0.74	24	None	0.06	0.81	19	None
	AS	0.1	0.77	12	Weak	0.17	0.64	10	Weak
Appendix C-9 Correlation between eburnation and facet angle continued

Facet	Sample		Ma	le		Female			
		Correlati on Coefficie nt	р	Ν	Strength of Correlati on	Correlati on Coefficie nt	р	N	Strength of Correlati on
T6LI	SBC			19		-0.18	0.38	27	Weak
	SBL			23		-0.33	0.16	20	Medium
	AS	0.28	0.47	9	Weak	0.52	0.12	10	Strong
T6RI	SBC			19				27	
	SBL			23				20	
	AS	-0.06	0.88	10	None	0.1	0.78	10	Weak
T7LS	SBC			20		-0.27	0.18	27	Weak
	SBL	0.28	0.2	23	Weak	0.38	0.12	18	Medium
	AS	-0.06	0.87	12	None	0.4	0.18	13	Medium
T7RS	SBC			20		-0.18	0.39	26	Weak
	SBL	0.33	0.12	24	Medium	0.43	0.1	16	Medium
	AS	0.13	0.68	12	Weak	-0.22	0.47	13	Weak
T7LI	SBC			19		0.36	0.16	17	Medium
	SBL			19		0.36	0.16	17	Medium
	AS	0.37	0.3	10	Medium	-0.08	0.81	12	None
T7RI	SBC			19				25	
	SBL			22				17	
	AS	0.04	0.91	10	None	0.08	0.8	13	None
T8LS	SBC	-0.24	0.29	21	Weak	0	1	24	None
	SBL			25				17	
	AS	0.02	0.94	13	None	0.22	0.48	13	Weak
T8RS	SBC			21		0	1	24	None
	SBL	0.02	0.91	25	None	0.06	0.82	16	None
	AS	-0.14	0.66	13	Weak	0.33	0.27	13	Medium
T8LI	SBC	-0.22	0.33	21	Weak	0.11	0.62	24	Weak
	SBL	-0.03	0.89	24	None	-0.04	0.91	14	None
	AS	-0.17	0.57	13	Weak	-0.23	0.45	13	Weak
T8RI	SBC			22		0.32	0.1	27	Medium
	SBL	0.06	0.78	24	None			15	
	AS	-0.2	0.52	13	Weak	0	1	13	None
T9LS	SBC	0.03	0.91	21	None	0.02	0.91	29	None
	SBL	-0.13	0.54	26	Weak	0.03	0.92	16	None
	AS	-0.14	0.64	14	Weak	-0.57	0.05	12	Strong
T9RS	SBC	0.05	0.82	21	None	-0.35	0.06	29	Medium
	SBL	-0.01	0.97	26	None	-0.45	0.11	14	Medium
	AS	0.32	0.29	13	Medium	-0.15	0.64	12	Weak

Appendix C-9 Correlation between eburnation and facet angle continued

Facet	Sample		M	ale		Female			
		Correlati	р	N	Strength	Correlati	р	N	Strength
		on Coofficio			of Corrolati	on Coofficio			of Corrolati
		nt			on	nt			on
T9LI	SBC			19		0.27	0.17	29	Weak
	SBL	0.01	0.95	25	None	-0.13	0.67	14	Weak
	AS	0.29	0.32	14	Weak	-0.41	0.22	11	Medium
T9RI	SBC			19		-0.05	0.8	29	None
	SBL			24		-0.42	0.14	14	Medium
	AS	-0.22	0.45	14	Weak	-0.5	0.12	11	Strong
T10LS	SBC	-0.05	0.82	20	None	0.06	0.78	28	None
	SBL	0.34	0.09	26	Medium	-0.03	0.93	14	None
	AS	-0.57	0.03	14	Strong	0.29	0.39	11	Weak
T10RS	SBC			20		0.32	0.11	27	Medium
	SBL			27				14	
	AS	-0.11	0.72	13	Weak	-0.07	0.84	11	None
T10LI	SBC			20		0.06	0.76	28	None
	SBL			27		0.14	0.63	14	Weak
	AS	-0.52	0.06	14	Strong	-0.43	0.19	11	Medium
T10RI	SBC			20		0.19	0.33	28	Weak
	SBL			27		-0.08	0.8	13	None
	AS	-0.5	0.07	14	Strong	-0.34	0.31	11	Medium
T11LS	SBC	0.01	0.96	21	None	0.02	0.94	28	None
	SBL	-0.07	0.72	25	None			15	
	AS	-0.23	0.42	14	Weak	0.29	0.39	11	Weak
T11RS	SBC	0.15	0.51	21	Weak	0.05	0.81	28	None
	SBL	-0.14	0.52	24	Weak	-0.31	0.27	14	Medium
	AS	0.22	0.44	14	Weak	-0.05	0.9	11	None
T11LI	SBC	0.38	0.1	20	Medium	-0.03	0.9	27	None
	SBL			23				14	
	AS	-0.37	0.21	13	Medium	0.3	0.37	11	Medium
T11RI	SBC			20		-0.34	0.1	25	Medium
	SBL			25				14	
	AS	-0.34	0.25	13	Medium	0.08	0.82	11	None
T12LS	SBC	-0.12	0.64	18	Weak	-0.23	0.28	25	Weak
	SBL	0.08	0.72	23	None			14	
	AS	0.04	0.91	13	None	0.41	0.21	11	Medium
T12RS	SBC	-0.35	0.15	18	Medium	0.06	0.76	25	None
	SBL			23				14	
	AS	-0.07	0.81	13	None	0.28	0.41	11	Weak
T12LI	SBC	0.07	0.77	19	None	-0.09	0.65	26	None
	SBL	-0.05	0.83	23	None			14	
	AS	0.14	0.66	12	Weak	0.23	0.53	10	Weak

Appendix C-9 Correlation between eburnation and facet angle continued

Facet	Sam		Male	2		Female			
	ple	Correlation Coefficient	р	N	Strength of Correlati on	Correlati on Coefficie nt	р	N	Strength of Correlati on
T12RI	SBC	0.29	0.25	18	Weak			24	
	SBL	-0.1	0.66	23	Weak			14	
	AS	0.07	0.84	12	None	0.32	0.33	11	Medium
L1LS	SBC	-0.01	0.97	19	None			25	
	SBL			24				12	
	AS	-0.07	0.81	14	None	-0.2	0.53	12	Weak
L1RS	SBC	0.42	0.08	19	Medium			24	
	SBL	0.21	0.32	24	Weak			13	
	AS	-0.22	0.49	12	Weak	0.04	0.9	13	None
L1LI	SBC	-0.33	0.17	19	Medium			23	
	SBL	0.07	0.74	25	None			12	
	AS	-0.39	0.19	13	Medium	-0.17	0.56	15	Weak
L1RI	SBC	-0.38	0.11	19	Medium			24	
	SBL	0.23	0.27	25	Weak			13	
	AS	-0.16	0.58	14	Weak	0.37	0.19	14	Medium
L2LS	SBC	-0.19	0.42	21	Weak	0.28	0.17	26	Weak
	SBL			23				13	
	AS	-0.09	0.75	15	None	0.43	0.19	11	Medium
L2RS	SBC			21				25	
	SBL			23				12	
	AS	-0.09	0.75	15	None	0.43	0.19	11	Medium
L2LI	SBC			21				25	
	SBL			24				13	
	AS	-0.12	0.69	14	Weak	0.23	0.52	10	Weak
L2RI	SBC			20				24	
	SBL	0.12	0.57	24	Weak			13	
	AS	-0.36	0.23	13	Medium	0.43	0.22	10	Medium
L3LS	SBC	-0.24	0.31	20	Weak			27	
	SBL	0.24	0.29	21	Weak	0.31	0.28	14	Medium
	AS	-0.15	0.59	16	Weak	-0.07	0.81	15	None
L3RS	SBC	0.2	0.38	21	Weak			26	
	SBL	-0.02	0.93	20	None			13	
	AS	-0.15	0.58	16	Weak	0.03	0.92	15	None
L3LI	SBC	0.15	0.52	21	Weak	-0.16	0.41	28	Weak
	SBL	0.06	0.8	20	None	0.17	0.56	14	Weak
	AS	-0.19	0.46	17	Weak	-0.31	0.26	15	Medium

Appendix C-9 Correlation	between eburnation	and facet angle	continued

Facet	Sample		М	ale		Female				
		Correlati on Coefficie nt	р	N	Strength of Correlati on	Correlati on Coefficie nt	р	N	Strength of Correlati on	
L3RI	SBC	0.42	0.06	21	Medium			29		
	SBL			19		0.17	0.56	14	Weak	
	AS	-0.02	0.95	15	None	-0.22	0.46	13	Weak	
L4LS	SBC			20				25		
	SBL			19				10		
	AS	0.32	0.23	16	Medium	-0.11	0.71	13	Weak	
L4RS	SBC	0.23	0.33	20	Weak	0.06	0.79	25	None	
	SBL			20		0.48	0.19	9	Medium	
	AS	-0.07	0.79	16	None	-0.01	0.98	13	None	
L4LI	SBC	-0.16	0.5	20	Weak			25		
	SBL			18				10		
	AS	-0.17	0.55	15	Weak	0.31	0.33	12	Medium	
L4RI	SBC	0.34	0.14	20	Medium	-0.02	0.94	23	None	
	SBL			20		0.18	0.63	10	Weak	
	AS	0.39	0.18	13	Medium	-0.32	0.34	11	Medium	
L5LS	SBC	0.24	0.35	18	Weak	0.26	0.27	20	Weak	
	SBL	-0.09	0.66	24	None			13		
	AS	0.08	0.82	11	None	0.27	0.42	11	Weak	
L5RS	SBC	0.51	0.03	18	Strong			20		
	SBL	-0.21	0.36	22	Weak	-0.08	0.74	18	None	
	AS	0.4	0.22	11	Medium	-0.5	0.12	11	Strong	
L5LI	SBC	0.11	0.65	18	Weak			18		
	SBL	0.08	0.71	24	None	-0.24	0.33	18	Weak	
	AS	-0.48	0.19	9	Medium	0.35	0.32	10	Medium	
L5RI	SBC	-0.12	0.64	18	Weak			18		
	SBL	0.1	0.66	23	Weak	0.28	0.27	18	Weak	
	AS	-0.65	0.03	11	Strong	-0.33	0.35	10	Medium	

Male Female Facet Correlation Strength of Correlation Strength of Sample р Ν р Ν Coefficient Correlation Coefficient Correlation SBC -0.02 0.95 18 None 0.12 0.6 23 Weak C1LS -0.03 Medium SBL 0.92 19 None 0.31 0.28 14 AS 0 0.99 11 None -0.42 0.35 7 Medium SBC 0.48 0.04 Medium 0.07 18 0.75 23 None C1RS SBL -0.17 0.47 20 Weak 14 AS -0.05 0.88 12 None 0.05 0.92 7 None Medium SBC 0.32 0.2 -0.2 0.36 23 Weak 18 -0.23 C1LI SBL 0.35 19 Weak 0.03 0.91 14 None AS -0.44 0.24 9 0.38 Medium Medium -0.45 6 SBC 0.34 0.16 18 Medium 0.22 0.31 23 Weak SBL -0.18 0.45 0.73 C1RI 20 Weak -0.1 14 Weak AS -0.2 0.53 12 Weak -0.1 0.83 7 Weak SBC -0.07 0.78 20 None -0.41 0.04 25 Medium C2LS SBL -0.28 0.24 20 Weak 0 1 13 None AS 0.24 0.5 10 Weak 0.73 0.04 8 Strong SBC -0.02 0.95 20 -0.13 0.53 25 Weak None C2RS SBL 0.01 0.98 -0.04 0.89 13 21 None None AS -0.17 0.64 10 Weak 0.4 0.33 8 Medium SBC 0.31 0.21 18 -0.07 0.74 25 Medium None C2LI SBL 0.3 0.21 19 Medium 0.1 0.76 13 None AS -0.07 0.86 9 0.61 0.15 7 Strong None SBC 0.36 0.15 18 Medium -0.12 0.56 25 Weak Weak C2RI SBL 0.26 0.26 21 -0.39 0.19 13 Medium -0.09 10 7 AS 0.8 0.67 None 0.1 Strong SBC -0.04 0.89 13 None -0.23 0.3 22 Weak C3LS SBL -0.06 0.81 0.04 0.89 21 None 17 None AS -0.06 0.85 14 None 0.6 0.21 6 Strong SBC -0.42 0.15 13 Medium -0.31 0.17 22 Medium C3RS SBL 0 0.99 21 None -0.27 0.32 16 Weak AS -0.06 0.85 14 -0.52 0.3 6 None Strong SBC 0.14 0.66 13 Weak 0.06 0.79 20 None C3LI SBL 0.09 0.69 20 -0.01 0.96 None 17 None Strong AS 0.22 0.45 14 Weak -0.93 0.01 6 SBC -0.22 0.47 13 Weak -0.07 0.78 20 None C3RI SBL 0.15 0.52 21 Weak -0.12 0.67 16 Weak 0.15 -0.46 6 Medium AS 0.65 12 Weak 0.36

			Μ	lale		Female			
Facat	Commis	Correlati			Strength	Correlati			Strength
Facet	Sample	on Coefficie	р	N	of Correlatio	on Coefficie	р	N	of Correlat
		nt			n	nt			ion
	SBC	-0.44	0.13	13	Medium	-0.04	0.88	20	None
C4LS	SBL	-0.04	0.85	23	None	0.02	0.94	16	None
	AS	0.31	0.32	12	Medium	-0.4	0.43	6	Medium
	SBC	-0.5	0.1	12	Strong	0.13	0.57	21	Weak
C4RS	SBL	-0.08	0.72	24	None	0.26	0.36	15	Weak
	AS	0.25	0.43	12	Weak	-0.75	0.09	6	Strong
	SBC	0.29	0.34	13	Weak	0.17	0.44	23	Weak
C4LI	SBL	-0.44	0.03	24	Medium	0.04	0.89	16	None
	AS	0.26	0.44	11	Weak	0.45	0.37	6	Medium
	SBC	0.23	0.46	12	Weak	-0.13	0.54	23	Weak
C4RI	SBL	-0.02	0.93	24	None	-0.31	0.26	15	Medium
	AS	-0.13	0.72	11	Weak	0.34	0.51	6	Medium
	SBC	0.02	0.95	20	None	-0.13	0.53	27	Weak
C5LS	SBL	-0.26	0.22	24	Weak	-0.09	0.72	20	None
	AS	0.42	0.3	8	Medium	-0.89	0.04	5	Strong
	SBC	0.02	0.94	20	None	0.31	0.12	27	Medium
C5RS	SBL	0.11	0.59	25	Weak	0.03	0.91	17	None
	AS	-0.13	0.75	8	Weak	-0.79	0.11	5	Strong
	SBC	0.5	0.02	20	Strong	-0.05	0.79	27	None
C5LI	SBL	-0.13	0.54	24	Weak	0.1	0.68	20	Weak
	AS	0.1	0.81	8	Weak	-0.35	0.56	5	Medium
	SBC	0.01	0.97	20	None	0.05	0.82	27	None
C5RI	SBL	0.22	0.29	24	Weak	0.06	0.8	20	None
	AS	0.46	0.26	8	Medium	-0.71	0.18	5	Strong
	SBC	0.08	0.76	19	None	-0.05	0.83	25	None
C6LS	SBL	0.12	0.58	23	Weak	0.11	0.66	19	Weak
	AS	0.06	0.88	9	None	-0.13	0.78	7	Weak
	SBC	0.22	0.37	19	Weak	-0.12	0.56	26	Weak
C6RS	SBL	0.52	0.01	22	Strong	0.12	0.63	19	Weak
	AS	0.07	0.86	9	None	-0.16	0.74	7	Weak
	SBC	-0.26	0.29	19	Weak	0.12	0.58	25	Weak
C6LI	SBL	-0.27	0.19	26	Weak	-0.25	0.31	19	Weak
	AS	-0.04	0.92	9	None	0.19	0.69	7	Weak
	SBC	-0.21	0.39	19	Weak	0.23	0.25	26	Weak
C6RI	SBL	-0.16	0.47	22	Weak	-0.28	0.24	19	Weak
	AS	0.02	0.96	9	None	-0.06	0.9	7	None
	SBC	-0.14	0.57	19	Weak	0.12	0.54	28	Weak
C7LS	SBL	-0.31	0.12	26	Medium	-0.21	0.37	20	Weak
	AS	0.16	0.66	10	Weak	0.95	0.01	5	Strong

Female Male Correlati Correlati Strength Strength Facet Sample of of on on Ν Ν р р Coefficie Correlati Coefficie Correlati nt on nt on SBC 0.09 0.72 19 None -0.14 0.47 28 Weak C7RS SBL 0 45 0.03 25 Medium -0 41 0.06 22 Medium -0.03 0.95 8 0.45 0.45 5 Medium AS None SBC 0.58 0.01 18 0.21 0.28 29 Weak Strong C7LI 22 SBL 27 -0.1 0.62 Weak -0.35 0.12 Medium 9 5 AS 0.72 0.03 -0.71 0.18 Strong Strong SBC 0.23 29 0.36 18 Weak 0.05 0.79 None C7RI 22 SBL -0.02 0.91 27 None 0.22 0.32 Weak 9 5 AS 0.37 0.33 Medium 0.71 0.18 Strong SBC 0.41 0.13 15 Medium -0.18 0.32 32 Weak 20 T1LS SBL 0.15 0.45 29 Weak -0.3 0.2 Weak AS 0.4 0.18 13 Medium -0.36 0.31 10 Medium SBC 0.29 0.24 18 Weak -0.2 0.26 32 Weak T1RS Medium SBL 0.05 0.79 29 None -0.3 0.18 21 AS 0.6 0.04 12 Strong 0.35 0.29 11 Medium SBC 0.08 0.74 19 -0.1 32 Weak None 0.6 T1LI SBL -0.06 0.75 30 None 0.02 0.92 22 None 0 12 0.06 0.87 AS 1 None 11 None SBC -0.25 0.3 19 Weak -0.02 0.91 32 None T1RI SBL 0.24 0.21 30 Weak -0.1 0.66 21 Weak AS -0.18 0.58 12 Weak -0.3 0.38 11 Medium SBC 0.04 0.83 29 -0.44 0.07 18 Medium None T2LS 0.83 29 SBL 0.04 None -0.44 0.07 18 Medium 0.44 0.15 12 13 AS Medium 0.21 0.49 Weak SBC -0.1 0.58 34 None 0.41 0.08 19 Medium T2RS SBL -0.1 0.58 34 None 0.41 0.08 19 Medium AS 0.39 0.21 12 Medium -0.11 0.72 13 Weak SBC 0.38 0.09 21 Medium -0.13 0.5 31 Weak T2LI SBL 19 -0.03 0.85 33 None 0.11 0.65 Weak AS 0.03 0.93 12 None 0.05 0.87 13 None SBC -0.36 0.11 21 Medium 0.16 0.41 30 Weak T2RI SBL 0.02 0.92 32 None -0.15 0.52 20 Weak AS 0.36 0.25 12 Medium -0.26 0.39 13 Weak SBC 0.03 0.9 20 0.15 0.44 28 Weak None 20 T3LS SBL 0.07 0.73 24 0.06 0.8 None None AS -0.26 0.39 13 Weak -0.39 0.19 13 Medium

Male Female									
Facat	Comunic	Correlati			Strength	Correlati			Strength
Facet	Sample	on Coofficio	р	N	of Corrolati	on Coofficio	р	N	of Corrolati
		nt			on	nt			on
	SBC	-0.08	0.75	20	None	0.13	0.51	29	Weak
T3RS	SBL	0.15	0.46	26	Weak	0.28	0.24	20	Weak
	AS	-0.14	0.61	15	Weak	0.07	0.83	13	None
	SBC	0.04	0.85	20	None	-0.22	0.25	29	Weak
T3LI	SBL	-0.14	0.5	24	Weak	0.1	0.67	21	Weak
	AS	0.23	0.43	14	Weak	-0.06	0.87	12	None
	SBC	-0.2	0.4	20	Weak	-0.03	0.89	29	None
T3RI	SBL	-0.18	0.38	25	Weak	0.17	0.47	21	Weak
	AS	0.46	0.09	15	Medium	-0.25	0.41	13	Weak
	SBC	0.03	0.9	16	None	-0.04	0.84	28	None
T4LS	SBL	-0.01	0.95	26	None	0.05	0.85	18	None
	AS	0.04	0.9	15	None	-0.28	0.35	13	Weak
	SBC	0	0.99	16	None	0.05	0.81	27	None
T4RS	SBL	0.1	0.65	24	None	0.13	0.61	18	Weak
	AS	0.08	0.78	14	None	-0.23	0.45	13	Weak
	SBC	0.2	0.45	16	Weak	0.19	0.35	27	Weak
T4LI	SBL	0.03	0.88	24	None	0.49	0.05	16	Strong
	AS	-0.27	0.34	14	Weak	-0.33	0.29	12	Medium
	SBC	-0.26	0.32	16	Weak	-0.29	0.14	27	Weak
T4RI	SBL	0.5	0.05	16	Strong	0.45	0.14	12	Medium
	AS	0.01	0.97	15	None	0.07	0.83	12	None
	SBC	-0.22	0.42	16	Weak	0.11	0.58	28	Weak
T5LS	SBL	0.29	0.19	22	Weak	-0.05	0.83	19	None
	AS	0.03	0.92	16	None	-0.33	0.39	9	Medium
	SBC	-0.16	0.54	18	Weak	-0.09	0.66	25	None
T5RS	SBL	0.4	0.07	21	Medium	-0.12	0.63	19	Weak
	AS	0.09	0.75	16	None	-0.56	0.1	10	Strong
	SBC	-0.19	0.49	15	Weak	0.02	0.93	24	None
T5LI	SBL	-0.04	0.86	20	None	0.3	0.19	20	Medium
	AS	0.53	0.06	13	Strong	0.64	0.06	9	Strong
	SBC	0.01	0.97	16	None	0.31	0.14	24	Medium
T5RI	SBL	0.01	0.98	23	None	-0.07	0.78	20	None
	AS	-0.03	0.92	13	None	0.64	0.06	9	Strong
	SBC	0.26	0.31	18	Weak	0.3	0.15	25	Medium
T6LS	SBL	-0.1	0.63	25	Weak	-0.18	0.47	19	Weak
	AS	0.17	0.58	13	Weak	-0.37	0.33	9	Medium
	SBC	-0.42	0.08	18	Medium	-0.32	0.12	25	Medium
T6RS	SBL	0.09	0.67	25	None	-0.14	0.57	19	Weak
	AS	0.26	0.39	13	Weak	-0.6	0.09	9	Strong

Female Male Correlati Strength Correlati Strength Facet Sample of of on on Ν Ν р р Coefficie Coefficie Correlati Correlati nt on nt on SBC 0.47 0.05 19 Medium -0.08 0.71 26 None T6LI SBL 0.05 0.84 22 None 0 18 0.44 20 Weak 0.82 0 11 8 AS Strong SBC -0.17 0.5 19 Weak -0.22 0.28 27 Weak T6RI 17 SBL -0.11 0.61 23 -0.2 Weak 0.43 Weak 8 AS 0.31 0.36 11 Medium -0.23 0.58 Weak SBC 0.09 19 29 0.41 Medium -0.17 0.39 Weak T7LS 25 SBL 0.04 0.85 None 0.25 0.33 18 Weak AS 0 1 13 None 0.3 0.34 12 Medium 0.58 27 SBC 0.13 19 Weak -0.18 0.36 Weak T7RS SBL -0.12 0.59 24 Weak -0.17 0.52 16 Weak AS -0.19 0.55 13 Weak 0.02 0.96 12 None SBC 0.37 0.13 18 Medium 0.15 0.45 28 Weak T7LI SBL Weak -0.04 0.87 22 None -0.2 0.43 17 Weak AS 0.22 0.54 10 Weak -0.1 0.78 11 SBC 0.02 0.95 18 -0.01 0.98 28 None None T7RI SBL -0.11 0.61 23 Weak -0.2 0.43 17 Weak 0.23 0.53 10 Weak -0.23 0.48 12 Weak AS SBC -0.15 0.55 19 Weak -0.15 0.45 28 Weak T8LS 0.04 SBL 0.85 25 None 0.25 0.33 18 Weak AS 0.3 0.38 11 Medium -0.08 0.8 13 None SBC -0.21 0.38 19 0.09 0.64 28 Weak None T8RS -0.16 0.45 25 0.52 SBL Weak 0.16 18 Weak -0.25 0.46 0.25 0.41 13 AS 11 Weak Weak SBC -0.09 0.72 19 None -0.14 0.48 28 Weak T8LI SBL -0.12 0.56 25 Weak 16 AS 0.33 0.32 11 Medium 0.11 0.73 13 Weak SBC -0.2 0.41 20 Weak 0.24 0.22 28 Weak T8RI SBL 0.15 0.49 25 Weak 0.36 0.16 17 Medium AS 0.09 0.79 11 None -0.19 0.54 13 Weak SBC 0.35 0.15 19 Medium 0.04 0.84 30 None T9LS SBL 0.29 0.16 26 Weak -0.09 0.74 17 None AS 0.06 0.85 13 -0.09 0.76 13 None None SBC 0.16 0.53 19 Weak -0.1 0.6 30 Weak T9RS 17 SBL 0.55 0 26 Strong AS 0.16 0.63 11 Weak -0.47 0.11 13 Medium

			M	ale		Female			
Facat	Comple	Correlati			Strength	Correlati			Strength
Facel	Sample	on Coefficie	р	N	of Correlati	on Coefficie	р	N	of Correlati
		nt			on	nt			on
	SBC	0.44	0.07	18	Medium	-0.61	0	29	Strong
T9LI	SBL	-0.08	0.72	25	None			17	
	AS	-0.3	0.34	12	Medium	-0.02	0.95	12	None
	SBC	-0.07	0.79	18	None	0.13	0.49	29	Weak
T9RI	SBL	-0.22	0.32	22	Weak			17	
	AS	-0.61	0.04	12	Strong	-0.28	0.38	12	Weak
	SBC	0.06	0.8	18	None	-0.07	0.71	28	None
T10LS	SBL	0.1	0.65	25	None	0.36	0.17	16	Medium
	AS	0.01	0.99	12	None	0.23	0.47	12	Weak
	SBC	0.26	0.31	18	Weak	0.17	0.39	28	Weak
T10RS	SBL	0.26	0.21	26	Weak			16	
	AS	-0.23	0.48	12	Weak	0.17	0.6	12	Weak
	SBC	0.01	0.98	18	None	0.16	0.4	29	Weak
T10LI	SBL	0.07	0.74	26	None			16	
	AS	-0.13	0.7	11	Weak	0.25	0.43	12	Weak
	SBC	-0.42	0.08	18	Medium	-0.09	0.64	29	None
T10RI	SBL	-0.28	0.17	26	Weak			15	
	AS	-0.28	0.38	12	Weak	-0.18	0.57	12	Weak
	SBC	0.09	0.71	19	None	-0.03	0.87	28	None
T11LS	SBL	-0.04	0.85	23	None	0.15	0.13	16	Weak
	AS	-0.27	0.37	13	Weak	-0.2	0.54	12	Weak
	SBC	-0.52	0.02	19	Strong	0.25	0.19	29	Weak
T11RS	SBL	-0.01	0.96	24	None	-0.29	0.28	16	Weak
	AS	0.08	0.79	13	None	-0.18	0.58	12	Weak
	SBC	-0.22	0.38	18	Weak	-0.23	0.23	28	Weak
T11LI	SBL	0	0.99	23	None	0.12	0.66	15	Weak
	AS	-0.14	0.67	12	Weak	0.41	0.19	12	Medium
	SBC	-0.2	0.43	18	Weak	-0.02	0.92	28	None
T11RI	SBL	0.01	0.98	24	None	0.42	0.11	16	Medium
	AS	-0.54	0.11	10	Strong	-0.07	0.83	12	None
	SBC	-0.18	0.48	17	Weak	0.14	0.53	24	Weak
T12LS	SBL	-0.06	0.8	22	None	-0.39	0.14	16	Medium
	AS	-0.16	0.61	12	Weak	0.28	0.4	11	Weak
	SBC	-0.46	0.06	17	Medium	-0.08	0.71	24	None
T12RS	SBL	-0.04	0.86	22	None	0.2	0.47	16	Weak
	AS	-0.61	0.04	12	Strong	0.59	0.06	11	Strong
	SBC	0.07	0.8	18	None	-0.13	0.54	25	Weak
T12LI	SBL	0.16	0.49	22	Weak	0.23	16	-0.37	Weak
	AS	-0.07	0.84	11	None	0.16	0.65	11	Weak

Female Male Correlati Strength Correlati Strength Facet Sample of of on on Ν Ν р р Coefficie Correlati Coefficie Correlati nt on nt on SBC 0.22 0.4 17 Weak 0 0.99 25 None T12RI SBL 0.32 0 1 5 22 Medium -0.04 0.88 16 None -0.11 0.77 10 0.4 11 AS Weak 0.23 Medium SBC -0.08 0.75 19 None 0.04 0.86 25 None 13 L1LS SBL 0.07 22 -0.31 0.4 Medium 0.31 Medium AS 0.15 0.59 15 Weak 0.01 0.97 13 None SBC 19 24 0.04 0.88 -0.06 0.79 None None L1RS SBL -0.13 0.56 24 Weak 0.05 0.88 12 None AS 0.18 0.58 12 Weak 0.5 0.07 14 Strong 24 SBC -0.12 0.62 19 Weak -0.03 0.89 None L1LI SBL 0.09 0.69 25 None 0.08 0.8 13 None AS -0.3 0.3 14 Medium 0.08 0.78 15 None SBC 0.36 0.13 19 Medium -0.41 0.05 24 Medium L1RI Medium SBL -0.09 0.68 24 None -0.4 0.2 12 AS -0.11 0.7 15 Weak 0.52 0.06 14 Strong SBC -0.05 0.82 21 -0.08 0.69 27 None None L2LS SBL 0.36 0.1 22 Medium 0.31 0.3 13 Medium -0.26 0.33 16 0.33 0.29 12 Medium AS Weak SBC -0.07 0.78 21 0.15 0.45 26 Weak None L2RS SBL 0.09 0.7 23 None 0.1 0.77 11 Weak AS 0.1 0.74 14 Weak 0.32 0.31 12 Medium SBC 0.15 0.52 21 0.13 0.52 26 Weak Weak 0.44 L2LI 0.03 24 SBL Medium 0.56 0.06 12 Strong -0.12 0.69 14 0.29 0.42 10 AS Weak Weak SBC 0 1 20 None 0 0.99 26 None L2RI SBL -0.04 0.86 24 None 0.38 0.22 12 Medium AS -0.03 0.93 13 None 0.1 0.77 11 Weak SBC -0.06 0.81 20 None 0.13 0.53 26 Weak L3LS SBL 0.2 0.4 20 Weak -0.4 0.18 13 Medium AS -0.45 0.08 16 Medium -0.2 0.49 14 Weak SBC -0.01 0.98 21 None 0.06 0.78 26 None L3RS SBL -0.05 0.86 18 -0.29 0.35 13 Weak None AS -0.02 0.95 16 -0.26 0.38 14 Weak None SBC 0.31 0.18 20 Medium -0.33 0.08 28 Medium Weak L3LI SBL -0.14 0.56 20 0.3 0.27 15 Medium AS -0.37 0.14 17 Medium -0.06 0.83 14 None

Appendix C-10 Correlation	between pitting scores and	facet angle continued

			M	ale			Female			
Facet	Sample	Correlati on Coefficie nt	р	N	Strength of Correlati on	Correlati on Coefficie nt	р	N	Strength of Correlati on	
	SBC	-0.01	0.96	20	None	0.01	0.95	27	None	
L3RI	SBL	0.19	0.49	16	Weak	-0.31	0.3	13	Medium	
	AS	-0.37	0.17	15	Medium	-0.67	0.02	12	Strong	
	SBC	0.02	0.94	20	None	0.1	0.63	25	Weak	
L4LS	SBL	-0.07	0.78	19	None			10		
	AS	0.19	0.49	16	Weak	-0.31	0.3	13	Medium	
	SBC	0.35	0.13	20	Medium	0.47	0.02	25	Medium	
L4RS	SBL	0.35	0.13	20	Medium	0.14	0.73	9	Weak	
	AS	0.19	0.49	16	Weak	-0.31	0.3	13	Medium	
	SBC	0.41	0.07	20	Medium	-0.33	0.1	25	Medium	
L4LI	SBL	0.07	0.78	18	None	0.61	0.06	10	Strong	
	AS	-0.39	0.16	15	Medium	-0.06	0.85	12	None	
	SBC	-0.01	0.98	19	None	0.12	0.55	26	Weak	
L4RI	SBL	0.03	0.89	20	None	-0.52	0.12	10	Strong	
	AS	0	1	13	None	-0.09	0.8	11	None	
	SBC	-0.44	0.07	18	Medium	-0.3	0.18	21	Medium	
L5LS	SBL	0.02	0.93	24	None	-0.54	0.06	13	Strong	
	AS	0.51	0.11	11	Strong	0.06	0.87	11	None	
	SBC	0.16	0.54	18	Weak	0	0.99	21	None	
L5RS	SBL	-0.19	0.42	21	Weak	-0.17	0.58	13	Weak	
	AS	0.4	0.22	11	Medium	-0.3	0.37	11	Medium	
	SBC	0.01	0.97	18	None	0.16	0.49	21	Weak	
L5LI	SBL	-0.04	0.84	24	None	-0.17	0.59	13	Weak	
	AS	-0.47	0.17	10	Medium	0.3	0.4	10	Medium	
	SBC	-0.15	0.56	18	Weak	-0.18	0.44	21	Weak	
L5RI	SBL	0.14	0.54	22	Weak	-0.39	0.19	13	Medium	
	AS	-0.56	0.07	11	Strong	-0.21	0.57	10	Weak	

Male Female Facet Sample Correlation Strength of Correlation Strength of р Ν р Ν Coefficient Correlation Coefficient Correlation SBC 18 0.13 0.56 23 Weak C1LS SBL -0.03 0.92 19 0.31 0.28 Medium None 14 AS 11 6 SBC 18 0.24 0.26 23 Weak C1RS SBL -0.17 0.47 20 Weak 14 AS 12 6 SBC 0.28 0.26 23 18 Weak SBL -0.23 0.35 C1LI 19 Weak 0.03 0.91 14 None 9 AS 5 23 SBC 18 C1RI SBL 20 0.73 14 -0.18 0.45 Weak -0.1 Weak AS -0.13 0.68 12 Weak 6 SBC -0.16 0.5 20 Weak 25 C2LS SBL -0.28 0.24 20 Weak 0 1 13 None AS 0.18 0.63 10 Weak 8 SBC 20 -0.17 0.42 25 Weak C2RS SBL 0.01 0.98 21 None -0.04 0.89 13 None 10 8 AS SBC 0.29 0.24 18 Weak 0.01 0.98 25 None C2LI SBL 0.3 0.21 19 Medium 0.1 0.76 13 None 7 AS 9 SBC -0.05 25 0.85 18 None -0.11 0.62 Weak 0.19 SBL 0.26 0.26 21 -0.39 C2RI Weak 13 Medium 7 0 AS 10 1 None SBC -0.06 0.85 13 None -0.01 0.95 22 None -0.06 C3LS SBL 0.81 21 0.04 0.89 17 None None AS 14 -0.66 0.16 6 Strong 22 SBC -0.18 0.56 13 Weak 0.23 0.31 Weak C3RS SBL 0 0.99 21 None -0.27 0.32 16 Weak AS 0.38 0.18 14 Medium -0.68 0.14 6 Strong SBC 0.23 0.45 13 Weak 0.52 0.02 20 Strong C3LI SBL 0.09 0.69 20 None -0.01 0.96 17 None -0.44 Medium AS 0.14 0.64 14 Weak 0.38 6 SBC -0.06 0.84 0.27 0.24 20 Weak 13 None C3RI SBL 0.15 0.52 21 Weak -0.12 0.67 16 Weak AS 12 -0.21 Weak 0.69 6

			М	ale		Female			
Facet	Sample	Correlation Coefficient	р	N	Strength of Correlation	Correlation Coefficient	р	N	Strength of Correlation
	SBC	-0.37	0.22	13	Medium	0.23	0.33	21	Weak
C4LS	SBL	-0.04	0.85	23	None	0.02	0.94	16	None
	AS	0.47	0.12	12	Medium			6	
	SBC	-0.61	0.04	12	Strong	-0.1	0.68	21	Weak
C4RS	SBL	-0.08	0.72	24	None	0.26	0.36	15	Weak
	AS	-0.42	0.17	12	Medium	-0.89	0.02	6	Strong
	SBC	-0.09	0.78	13	None	0.31	0.15	23	Medium
C4LI	SBL	-0.44	0.03	24	Medium	0.04	0.89	16	None
	AS	0.35	0.29	11	Medium	0.69	0.13	6	Strong
	SBC	-0.1	0.77	12	Weak	-0.05	0.81	23	None
C4RI	SBL	-0.02	0.93	24	None	-0.31	0.26	15	Medium
	AS	0.5	0.12	11	Strong	0.67	0.15	6	Strong
	SBC	-0.07	0.77	20	None	0.31	0.11	27	Medium
C5LS	SBL	-0.26	0.22	24	Weak	-0.09	0.72	20	None
	AS	0.23	0.58	8	Weak	-0.71	0.18	5	Strong
	SBC	-0.15	0.52	20	Weak	-0.13	0.51	27	Weak
C5RS	SBL	0.11	0.59	25	Weak	0.03	0.91	17	None
	AS	0.25	0.55	8	Weak	-0.71	0.18	5	Strong
	SBC	0.05	0.82	20	None	-0.12	0.56	27	Weak
C5LI	SBL	-0.13	0.54	24	Weak	0.1	0.68	20	Weak
	AS	-0.08	0.85	8	None			5	
	SBC	-0.18	0.44	20	Weak	-0.08	0.68	27	None
C5RI	SBL	0.22	0.29	24	Weak	0.06	0.8	20	None
	AS	0.25	0.55	8	Weak	-0.71	0.18	5	Strong
	SBC	0.26	0.28	19	Weak	-0.14	0.52	25	Weak
C6LS	SBL	0.12	0.58	23	Weak	0.11	0.66	19	Weak
	AS	-0.55	0.13	9	Strong			7	
	SBC	-0.21	0.4	19	Weak	-0.04	0.86	26	None
C6RS	SBL	0.52	0.01	22	Strong	0.12	0.63	19	Weak
	AS	0	1	9	None	-0.27	0.56	7	Weak
	SBC	-0.18	0.46	19	Weak	0.24	0.25	25	Weak
C6LI	SBL	-0.27	0.19	26	Weak	-0.25	0.31	19	Weak
	AS	0	1	9	None	0.41	0.36	7	Medium
	SBC	-0.1	0.68	19	Weak	0.05	0.81	26	None
C6RI	SBL	-0.16	0.47	22	Weak	-0.28	0.24	19	Weak
	AS			9		-0.1	0.83	7	Weak
	SBC	-0.06	0.81	19	None	-0.02	0.91	29	None
C7LS	SBL	-0.31	0.12	26	Medium	-0.21	0.37	20	Weak
	AS	-0.06	0.87	10	None	0.87	0.06	5	Strong

			М	ale			Ferr	nale	
Facet	Sample	Correlation Coefficient	р	Ν	Strength of Correlation	Correlation Coefficient	р	Ν	Strength of Correlation
	SBC	-0.39	0.1	19	Medium	0.03	0.87	28	None
C7RS	SBL	0.45	0.03	25	Medium	-0.41	0.06	22	Medium
	AS			10		-0.71	0.18	5	Strong
	SBC	0.3	0.23	18	Medium	-0.07	0.73	29	None
C7LI	SBL	-0.1	0.62	27	Weak	-0.35	0.12	22	Medium
	AS	0.46	0.19	10	Medium	-0.71	0.18	5	Strong
	SBC	-0.21	0.41	18	Weak	0.17	0.39	29	Weak
C7RI	SBL	-0.02	0.91	27	None	0.22	0.32	22	Weak
	AS	0.52	0.12	10	Strong	0.71	0.18	5	Strong
	SBC	0.4	0.11	17	Medium	0.2	0.29	30	Weak
T1LS	SBL	0.15	0.45	29	Weak	-0.3	0.2	20	Weak
	AS	0.02	0.94	14	None	0.17	0.67	9	Weak
	SBC	-0.1	0.69	19	Weak	0.07	0.7	31	None
T1RS	SBL	0.05	0.79	29	None	-0.3	0.18	21	Medium
	AS	0.26	0.4	13	Weak	0.23	0.53	10	Weak
	SBC			20		-0.02	0.91	31	None
T1LI	SBL	-0.06	0.75	30	None	0.02	0.92	22	None
	AS			13		-0.23	0.53	10	Weak
	SBC			20		0.16	0.39	31	Weak
T1RI	SBL	0.24	0.21	30	Weak	-0.1	0.66	21	Weak
	AS	-0.46	0.11	13	Medium	0.06	0.87	10	None
	SBC	0.04	0.83	29	None	-0.44	0.07	18	Medium
T2LS	SBL	0.04	0.83	29	None	-0.44	0.07	18	Medium
	AS	0.13	0.69	12	Weak	0.16	0.6	13	Weak
	SBC	-0.1	0.58	34	None	0.41	0.08	19	Medium
T2RS	SBL	-0.1	0.58	34	None	0.41	0.08	19	Medium
	AS	0.18	0.58	12	Weak	-0.74	0	13	Strong
	SBC	-0.13	0.56	23	Weak	0.11	0.57	29	Weak
T2LI	SBL	-0.03	0.85	33	None	0.11	0.65	19	Weak
	AS	0.18	0.59	12	Weak	-0.31	0.3	13	Medium
	SBC	-0.04	0.86	23	None	0.19	0.34	29	Weak
T2RI	SBL	0.02	0.92	32	None	-0.15	0.52	20	Weak
	AS	0.27	0.41	12	Weak	-0.35	0.24	13	Medium
	SBC	-0.02	0.92	21	None	0.13	0.53	27	Weak
T3LS	SBL	0.07	0.73	24	None	0.06	0.8	20	None
	AS	0.04	0.88	14	None	-0.43	0.12	14	Medium
	SBC	0.09	0.7	22	None	0.17	0.41	27	Weak
T3RS	SBL	0.15	0.46	26	Weak	0.28	0.24	20	Weak
	AS			15		-0.02	0.96	14	None

Male Female Facet Sample Strength of Correlation Correlation Strength of Ν Ν р р Coefficient Correlation Coefficient Correlation 0.14 SBC -0.32 22 Medium -0.08 0.7 27 None SBL -0.14 0.5 Weak 24 Weak 0.1 0.67 21 T3LI AS -0.02 0.95 14 None 0.12 0.69 13 Weak SBC -0.29 0.2 22 Weak -0.01 0.98 26 None T3RI SBL -0.18 0.38 25 Weak 0.17 0 47 21 Weak AS 0.06 0.83 15 None -0.29 0.34 13 Weak SBC 0.08 0.74 19 None -0.1 0.65 25 Weak T4LS SBL -0.01 0.95 26 None 0.05 0.85 18 None AS -0.09 0.78 13 0.14 13 None 0.65 Weak SBC 19 0.05 0.83 None 0.08 0.73 24 None T4RS SBL 0.1 0.65 24 None 0.13 0.61 18 Weak 13 -0.3 0.32 Medium AS 13 SBC -0.09 0.71 19 None 0.23 0.27 24 Weak 24 T4LI SBL 0.03 0.49 0.88 None 0.05 16 Strong AS 13 12 SBC 19 -0.61 0.01 Strong 0.16 0.47 24 Weak T4RI SBL 0.5 0.05 16 Strong 0.45 0.14 12 Medium AS 0.22 0.49 12 13 Weak SBC -0.35 0.14 19 Medium 0.06 0.8 25 None T5LS SBL 0.29 0.19 22 Weak -0.05 0.83 19 None 0.1 0.73 14 Weak 0.06 0.86 10 AS None SBC -0.11 0.66 19 Weak 0.56 25 0 Strong T5RS SBL 0.4 0.07 21 Medium -0.12 0.63 19 Weak AS -0.14 0.63 14 Weak 0.28 0.43 10 Weak SBC 22 -0.26 0.32 17 Weak 0.03 0.9 None T5LI SBL -0.04 0.86 20 None 0.3 0.19 20 Medium 9 AS 0.06 0.87 10 None SBC -0.03 18 0.9 None 0.22 0.32 22 Weak T5RI SBL 0.01 0.98 23 None -0.07 0.78 20 None AS -0.55 0.13 9 0.48 0.16 10 Medium Strong SBC 0.17 0.51 17 Weak 0.35 0.09 25 Medium T6LS SBL -0.1 0.63 25 19 Weak -0.18 0.47 Weak AS 12 Strong 0.62 0.03 -0.46 0.18 10 Medium SBC 0.55 0.2 25 -0.16 17 Weak -0.26 Weak T6RS SBL 0.09 0.67 25 0.57 19 None -0.14 Weak AS 0.4 0.2 12 Medium -0.24 0.5 10 Weak SBC 19 -0.11 0.6 26 Weak T6LI SBL 0.05 0.84 22 None 0.18 0.44 20 Weak AS 10 -0.81 0 10 Strong

Male Female Facet Sample Correlation Strength of Correlation Strength of Ν Ν р р Correlation Coefficient Correlation Coefficient 0.75 0.93 27 SBC 0.08 19 Medium 0.02 None SBL -0.11 Weak -0.2 0.43 Weak T6RI 0.61 23 17 Weak AS 10 0.22 0.54 10 SBC -0.2 0.39 20 Weak 0.3 0.12 28 Medium T7LS SBL 0.04 0.85 25 0.25 0.33 18 Weak None AS -0.21 0.51 12 Weak -0.44 0.15 12 Medium SBC -0.13 0.58 0.09 20 Weak 0.68 26 None T7RS SBL -0.12 0.59 24 Weak -0.17 0.52 16 Weak AS 0.48 0.11 12 -0.09 0.78 12 Medium None SBC 18 -0.03 27 0.89 None T7LI SBL -0.04 0.87 22 None -0.2 0.43 17 Weak AS 0.41 0.24 10 Medium -0.1 0.77 11 Weak SBC 19 0.15 0.45 27 Weak T7RI -0.11 SBL 0.61 23 -0.2 17 Weak 0.43 Weak AS 10 0.04 0.89 12 None SBC -0.08 -0.03 0.9 21 None 0.72 24 None T8LS SBL 0.04 0.85 25 None 0.25 0.33 18 Weak AS 0.18 0.58 12 Weak -0.06 0.85 12 None Weak SBC -0.14 0.55 21 0.14 0.52 24 Weak T8RS SBL -0.16 0.45 25 Weak 0.16 0.52 18 Weak -0.06 0.07 AS 0.86 12 0.83 12 None None SBC -0.08 0.75 21 None -0.27 0.2 24 Weak T8LI SBL -0.12 0.56 25 Weak 16 AS 12 -0.22 0.49 12 Weak SBC 22 0.23 0.27 -0.08 0.74 None 24 Weak T8RI SBL 0.15 0.49 25 Weak 0.36 0.16 17 Medium AS 0.48 0.11 12 Medium 0.22 0.49 12 Weak SBC 0.05 0.83 21 0.93 29 None 0.02 None T9LS SBL 0.29 0.16 26 Weak -0.09 0.74 17 None 0.33 0.25 AS 14 Medium -0.17 0.61 11 Weak SBC -0.19 0.41 21 Weak 0.12 0.52 29 Weak 0 T9RS SBL 0.55 26 17 Strong AS -0.07 0.83 11 13 None 0.02 0.94 None SBC 19 -0.34 0.07 29 Medium T9LI SBL 0.72 25 17 -0.08 None AS 0.21 0.48 14 Weak -0.53 0.11 10 Strong SBC 0.26 0.28 19 Weak 0.15 0.45 28 Weak T9RI SBL -0.22 0.32 22 Weak 17 AS 14 0.12 0.73 11 Weak

Male Female Facet Sample Strength of Correlation Correlation Strength of Ν Ν р р Coefficient Correlation Coefficient Correlation 0.37 SBC -0.21 20 Weak -0.2 0.3 28 Weak SBL 0.65 0.36 T10LS 0.1 25 None 0.17 16 Medium AS 0.08 0.78 14 None 0.56 0.07 11 Strong SBC -0.07 0.77 20 None -0.21 0.28 28 Weak T10RS SBL 0.26 0.21 26 Weak 16 AS 0.21 0.48 13 Weak 0.23 0.4 11 Medium SBC 0.72 19 -0.09 0.09 0.65 27 None None T10LI SBL 0.07 0.74 26 None 16 AS 0.4 13 10 0.18 Medium -0.32 0.37 Medium SBC 19 Strong 0.62 0 -0.26 0.18 27 Weak T10RI SBL -0.28 0.17 26 Weak 15 -0.35 0.24 Medium AS 13 -0.56 0.09 10 Strong SBC 0.4 0.09 20 Medium 0.18 0.36 29 Weak T11LS SBL -0.04 0.85 23 None 0.15 0.13 16 Weak AS -0.26 0.38 14 Weak 0.05 0.89 11 None 21 SBC -0.06 29 0.8 None -0.02 0.92 None T11RS SBL -0.01 0.96 24 None -0.29 0.28 16 Weak AS -0.04 0.89 14 0.07 11 None -0.56 Strong SBC -0.01 0.95 20 -0.2 28 None 0.31 Weak T11LI SBL 0 0.99 23 None 0.12 0.66 15 Weak 0.23 0.45 13 Weak 0.26 0.44 Weak AS 11 SBC 0.2 0.41 20 Weak -0.23 0.25 28 Weak T11RI SBL 0.01 0.98 24 None 0.42 0.11 16 Medium AS 13 0.04 0.9 11 None SBC 25 -0.27 0.27 18 Weak -0.06 0.77 None T12LS SBL -0.06 0.8 22 None -0.39 0.14 16 Medium AS 0.45 0.13 13 Medium 0.32 0.34 11 Medium SBC 0 18 25 0.65 Strong -0.43 0.03 Medium T12RS SBL -0.04 0.86 22 None 0.2 0.47 16 Weak 0.36 AS 0.28 13 Weak 0.22 0.52 11 Weak SBC 0.23 0.35 19 Weak 0.1 0.64 26 Weak _ SBL 0.49 22 0.23 T12LI 0.16 Weak 16 Weak 12 AS 0.06 0.87 None 0.24 0.51 10 Weak 0.34 -0.02 0.93 26 SBC 0.17 18 Medium None T12RI SBL 0.32 0.15 22 Medium -0.04 0.88 16 None AS 12 0.5 0.12 11 Strong 25 SBC 0.15 0.53 19 Weak 0.19 0.37 Weak L1LS SBL 0.4 0.07 22 Medium -0.31 0.31 13 Medium AS -0.43 0.12 14 Medium -0.27 0.38 13 Weak

Male Female Facet Sample Correlation Strength of Correlation Strength of Ν Ν р р Coefficient Correlation Coefficient Correlation 0.01 Strong 0.97 SBC 0.59 18 -0.01 24 None 0.05 SBL -0.13 0.56 L1RS 24 Weak 0.88 12 None AS -0.36 0.25 12 Medium 0.17 0.55 14 Weak SBC -0.25 0.3 19 Weak -0.17 0.44 24 Weak 0.08 L1LI SBL 0.09 0.69 25 None 0.8 13 None AS 13 -0.27 0.33 15 Weak SBC 0.32 19 -0.4 0.05 24 Medium -0.24 Weak L1RI SBL -0.09 0.68 24 None -0.4 0.2 12 Medium AS 0.27 0.35 14 0.15 14 Weak 0.6 Weak SBC 27 0.01 0.96 21 None -0.11 0.59 Weak L2LS SBL 0.36 0.1 22 Medium 0.31 0.3 13 Medium AS 0.07 0.8 -0.3 0.35 12 Medium 15 None SBC 0.27 0.23 21 Weak -0.41 0.04 26 Medium L2RS SBL 23 0.1 11 0.09 0.7 0.77 Weak None AS -0.13 0.65 14 Weak -0.22 0.49 12 Weak SBC 26 0.53 0.01 21 Strong -0.45 0.02 Medium Strong L2LI SBL 0.44 0.03 24 Medium 0.56 0.06 12 0.04 0.9 14 0.72 10 AS None -0.13 Weak -0.33 SBC -0.01 0.97 20 None 26 Medium 0.1 L2RI SBL -0.04 0.86 24 None 0.38 0.22 12 Medium AS 0.07 0.81 14 0.51 0.11 11 None Strong SBC -0.19 0.42 20 Weak -0.05 0.79 27 None L3LS SBL 0.2 0.4 20 Weak -0.4 0.18 13 Medium AS -0.16 0.55 16 Weak -0.13 0.66 14 Weak SBC 0.17 20 0.32 Medium -0.01 0.98 26 None L3RS SBL -0.05 0.86 18 None -0.29 0.35 13 Weak AS 0.21 0.45 16 Weak -0.18 0.54 14 Weak SBC 0.03 -0.05 0.79 28 0.47 21 Medium None L3LI SBL -0.14 0.56 20 Weak 0.3 0.27 15 Medium 0.09 AS -0.42 17 Medium 0.22 0.45 14 Weak SBC 0.28 0.23 21 Weak -0.04 0.83 26 None SBL 0.04 0.89 16 -0.4 0.2 12 Medium L3RI None AS -0.39 0.16 15 Medium -0.46 0.16 11 Medium SBC 20 0.93 25 0.1 0.68 Weak -0.02 None SBL -0.07 0.78 10 L4LS 19 None AS 0.04 0.89 16 None -0.4 0.2 12 Medium SBC 0.4 0.08 20 Weak -0.1 0.63 25 Weak L4RS SBL 0.35 0.13 20 Medium 0.14 0.73 9 Weak AS 0.04 0.89 16 None -0.4 0.2 12 Medium

			М	ale			Ferr	nale	
Facet	Sample	Correlation Coefficient	р	Ν	Strength of Correlation	Correlation Coefficient	р	N	Strength of Correlation
	SBC	0.28	0.22	20	Weak	0.18	0.39	25	Weak
L4LI	SBL	0.07	0.78	18	None	0.61	0.06	10	Strong
	AS	0.1	0.73	15	Weak	0.01	0.97	11	None
	SBC	0.22	0.34	20	Weak	0.25	0.23	26	Weak
L4RI	SBL	0.03	0.89	20	None	-0.52	0.12	10	Strong
	AS	0.14	0.65	13	Weak	0.3	0.4	10	Medium
	SBC	0.02	0.94	18	None	0.38	0.09	21	Medium
L5LS	SBL	0.02	0.93	24	None	-0.54	0.06	13	Strong
	AS	0.24	0.48	11	Weak	0.01	0.97	10	None
	SBC	0.47	0.05	18	Medium	-0.39	0.08	21	Medium
L5RS	SBL	-0.19	0.42	21	Weak	-0.17	0.58	13	Weak
	AS	-0.46	0.15	11	Medium	-0.67	0.04	10	Strong
	SBC	-0.25	0.33	18	Weak	-0.11	0.64	21	Weak
L5LI	SBL	-0.04	0.84	24	None	-0.17	0.59	13	Weak
	AS	0.25	0.48	10	Weak	-0.18	0.64	9	Weak
	SBC	0.13	0.61	18	Weak	0.17	0.46	21	Weak
L5RI	SBL	0.14	0.54	22	Weak	-0.39	0.19	13	Medium
	AS	-0.06	0.87	11	None	-0.15	0.71	9	Weak

			Male			Female			
Facet	Sample	Correlation Coefficient	р	N	Strength of Correlation	Correlation Coefficient	р	N	Strength of Correlation
	SBC								
C1LS	SBL								
	AS								
	SBC								
C1RS	SBL								
	AS								
	SBC								
C1LI	SBL								
	AS								
	SBC								
C1RI	SBL								
	AS								
	SBC	-0.076	0.731	23	None	0.155	0.375	35	Weak
C2LS	SBL	0.399	0.032	29	Medium	-0.176	0.459	20	Weak
	AS			9				6	
	SBC	-0.146	0.505	23	Weak	0.173	0.321	35	Weak
C2RS	SBL	-0.028	0.885	29	None	-0.173	0.466	20	Weak
	AS			9				6	
	SBC	-0.373	0.088	22	Medium	0.007	0.968	35	None
C2LI	SBL	0.028	0.887	29	None	0.403	0.078	20	Medium
	AS			9				6	
	SBC	-0.271	0.211	23	Weak	0.081	0.643	35	None
C2RI	SBL	-0.303	0.117	28	Medium	0.29	0.214	20	Weak
	AS			9				5	
	SBC	-0.011	0.966	19	None	-0.095	0.605	32	None
C3LS	SBL	-0.185	0.347	28	Weak	-0.151	0.513	21	Weak
	AS	0.194	0.591	10	Weak			5	
	SBC	0.115	0.629	20	Weak	0.03	0.869	32	None
C3RS	SBL	0.002	0.993	27	None	0.066	0.782	20	None
	AS	-0.055	0.88	10	Weak			5	
	SBC	0.049	0.839	20	None	-0.056	0.766	31	None
C3LI	SBL	0.08	0.684	28	None	0.228	0.319	21	Weak
	AS	-0.225	0.531	10	Weak			5	
	SBC	-0.139	0.56	20	Weak	0.217	0.241	31	Weak
C3RI	SBL	-0.096	0.628	28	None	0.24	0.308	20	Weak
	AS	-0.375	0.325	9	Medium			5	

			Male			Female			
Facet	Sample	Correlation Coefficient	р	N	Strength of Correlation	Correlation Coefficient	р	N	Strength of Correlation
	SBC	0.03	0.903	19	None	0.211	0.28	28	Weak
C4LS	SBL	-0.148	0.445	29	Weak	-0.025	0.911	22	None
	AS			10				7	
	SBC	0.214	0.366	20	Weak	-0.022	0.91	28	None
C4RS	SBL	-0.032	0.868	29	None	0.007	0.975	22	None
	AS			10				7	
	SBC	-0.259	0.271	20	Weak	-0.172	0.372	29	Weak
C4LI	SBL	0.075	0.699	29	None	0.052	0.813	23	None
	AS			10				7	
	SBC	-0.212	0.369	20	Weak	-0.031	0.874	29	None
C4RI	SBL	0.019	0.923	28	None	0.148	0.511	22	Weak
	AS			10				7	
	SBC	-0.127	0.562	23	Weak	0.052	0.771	34	None
C5LS	SBL	-0.126	0.507	30	Weak	0.055	0.797	24	None
	AS	-0.127	0.727	10	Weak			8	
	SBC	-0.263	0.215	24	Weak	-0.094	0.598	34	None
C5RS	SBL	-0.087	0.654	29	None	0.131	0.541	24	Weak
	AS	0.265	0.46	10	Weak			8	
	SBC	0.036	0.866	24	None	-0.195	0.268	34	Weak
C5LI	SBL	0.125	0.518	29	Weak	0.067	0.756	24	None
	AS	0.58	0.079	10	Strong			8	
	SBC	-0.514	0.01	24	Strong	0.039	0.827	34	None
C5RI	SBL	0.134	0.506	27	Weak	-0.122	0.569	24	Weak
	AS	0.628	0.052	10	Strong			8	
	SBC	0.149	0.531	20	Weak	-0.09	0.601	36	None
C6LS	SBL	-0.181	0.347	29	Weak	-0.059	0.789	23	None
	AS	0.4	0.222	11	Medium			7	
	SBC	-0.525	0.017	20	Strong	-0.101	0.558	36	Weak
C6RS	SBL	-0.316	0.083	31	Medium	-0.028	0.895	24	None
	AS	-0.196	0.563	11	Weak			7	
	SBC	-0.165	0.488	20	Weak	0.032	0.852	36	None
C6LI	SBL	0.003	0.988	31	None	-0.082	0.71	23	None
	AS	0.374	0.257	11	Medium			7	
	SBC	-0.355	0.125	20	Medium	0.023	0.893	36	None
C6RI	SBL	-0.027	0.888	30	None	-0.153	0.475	24	Weak
	AS	0.436	0.18	11	Medium			7	
	SBC	-0.233	0.273	24	Weak	-0.128	0.45	37	Weak
C7LS	SBL	-0.245	0.176	32	Weak	-0.009	0.968	23	None
	AS	0.673	0.033	10	Strong			5	

Male Female Facet Sample Correlation Strength of Correlation Strength of Ν р Ν р Coefficient Correlation Coefficient Correlation SBC 0.061 0.776 24 None 0.049 0.777 36 None C7RS SBL 0.025 0.893 32 0.24 0.271 23 Weak None AS 0.469 0.172 10 Medium 5 24 SBC 0.028 0.897 0.785 37 None 0.046 None 0.3 23 Medium C7LI SBL -0.14 0.443 32 0.164 None 5 -0.425 AS 0.192 11 Medium SBC 0.034 24 0.876 37 0.876 None -0.027 None C7RI SBL 0.023 0.075 23 0.9 32 None 0.378 Medium AS -0.003 0.993 11 None 5 SBC 0.007 0.76 18 None 0.347 0.035 37 Medium T1LS SBL -0.036 0.842 34 None 0.226 0.312 22 Weak AS -0.736 0.095 6 Strong 6 SBC 0.003 0.99 21 0.049 0.771 38 None None T1RS SBL -0.285 0.102 34 Weak -0.076 0.736 22 None 7 AS 0.457 0.217 9 Medium SBC -0.24 0.294 21 Weak 0.054 0.474 38 None T1LI SBL 0.096 0.585 35 -0.137 0.543 22 Weak None 7 AS -0.267 0.488 9 Weak 0.972 21 SBC 0.008 0.066 0.694 38 None None T1RI SBL 0.235 0.174 35 Weak -0.267 0.23 22 Weak -0.447 Medium 7 AS 0.227 9 SBC -0.049 0.783 34 0.21 0.347 22 None Weak T2LS -0.049 0.783 34 0.21 0.347 22 SBL None Weak 8 AS -0.597 0.069 10 Strong -0.286 0.493 Weak 22 SBC -0.055 0.757 34 None 0.03 0.894 None T2RS SBL -0.055 0.757 34 None 0.03 0.894 22 None AS -0.471 0.169 10 Medium 0.224 0.593 8 Weak SBC 0.026 0.908 23 None -0.185 0.273 37 Weak T2LI SBL -0.109 0.541 34 None 0.077 0.733 22 None Strong AS -0.494 0.147 10 Medium -0.608 0.11 8 SBC 0.028 0.9 23 None 0.027 0.874 37 None T2RI SBL -0.046 0.801 33 0.272 0.221 22 Weak None 0.667 0.035 0.991 0.001 8 AS 10 Strong Strong SBC 0.077 0.73 22 0.073 0.664 38 None None T3LS SBL -0.408 0.031 28 Medium 0.024 0.919 20 None 9 AS 0.075 0.828 11 None 0.013 0.973 None SBC 0.436 22 0.359 0.027 38 Medium 0.175 Weak T3RS SBL 0.218 0.256 29 Weak 0.018 0.941 20 None 9 AS -0.361 0.275 11 Medium 0.028 0.942 None

			Male				Female	2	
Facet	Sample	Correlation Coefficient	р	N	Strength of Correlation	Correlation Coefficient	р	Ν	Strength of Correlation
	SBC	-0.015	0.948	22	None	-0.021	0.901	38	None
T3LI	SBL	-0.146	0.45	29	Weak	0.116	0.625	20	Weak
	AS	-0.39	0.265	10	Medium	-0.356	0.386	8	Medium
	SBC	0.05	0.826	22	None	0.002	0.99	38	None
T3RI	SBL	-0.157	0.415	29	Weak	-0.021	0.931	20	None
	AS	-0.307	0.359	11	Medium	-0.16	0.681	9	None
	SBC	0.162	0.449	24	Weak	0.027	0.874	37	None
T4LS	SBL	-0.078	0.681	30	None	-0.463	0.046	19	Medium
	AS	-0.439	0.238	9	Medium	0.133	0.733	9	Weak
	SBC	0.036	0.868	24	None	0.179	0.297	36	Weak
T4RS	SBL	-0.027	0.89	29	None	0.188	0.441	19	Weak
	AS	0.696	0.037	9	Strong	-0.374	0.321	9	Medium
	SBC	-0.095	0.659	24	None	0.029	0.868	36	None
T4LI	SBL	-0.026	0.894	29	None	-0.279	0.263	18	Weak
	AS	-0.396	0.292	9	Medium	-0.058	0.892	8	None
	SBC	-0.167	0.435	24	Weak	0.003	0.987	36	None
T4RI	SBL	0.059	0.759	30	None	-0.062	0.807	18	None
	AS	-0.257	0.504	9	Weak	-0.189	0.654	8	Weak
	SBC	0.042	0.85	23	None	0.165	0.343	35	Weak
T5LS	SBL	-0.019	0.923	28	None	0.032	0.895	19	None
	AS	-0.26	0.5	9	Weak	-0.316	0.49	7	Medium
	SBC	0.173	0.43	23	Weak	0.055	0.753	35	None
T5RS	SBL	-0.18	0.359	28	Weak	0.107	0.663	19	Weak
	AS	0.088	0.822	9	None	-0.068	0.885	7	None
	SBC	0.03	0.894	23	None	-0.179	0.304	35	Weak
T5LI	SBL	-0.095	0.645	26	None	0.3	0.212	19	Medium
	AS	0.232	0.616	7	Weak	0.206	0.657	7	Weak
	SBC	-0.031	0.89	23	None	-0.216	0.212	35	Weak
T5RI	SBL	0.26	0.182	28	Weak	0.033	0.894	19	None
	AS	-0.029	0.95	7	None	-0.546	0.205	7	Strong
	SBC	0.187	0.383	24	Weak	-0.17	0.338	34	Weak
T6LS	SBL	-0.08	0.667	31	None	0.069	0.765	21	None
	AS	-0.086	0.826	9	None	-0.196	0.641	8	Weak
	SBC	-0.002	0.994	24	None	0.105	0.556	34	Weak
T6RS	SBL	0.081	0.664	31	None	-0.243	0.288	21	Weak
	AS	0.186	0.632	9	Weak	0.318	0.442	8	Medium
	SBC	-0.197	0.357	24	Weak	0.158	0.358	36	Weak
T6LI	SBL	-0.267	0.162	29	Weak	-0.049	0.833	21	None
	AS	-0.515	0.237	7	strong	0.318	0.443	8	Medium

Male Female Facet Sample Correlation Strength of Correlation Strength of Ν Ν р р Coefficient Correlation Coefficient Correlation SBC 0.108 0.635 24 Weak 0.083 0.631 36 None T6RI SBL 0.142 0.447 31 Weak 0.126 0.596 20 Weak AS -0.411 036 7 None 0.224 0.594 8 Weak 26 SBC 0.042 0.839 -0.008 0.96 37 None None T7LS 32 20 SBL -0.001 0.995 None 0.004 0.988 None 8 9 0.435 0.207 AS 0.323 Medium 0.593 Weak SBC 26 37 -0.016 0.936 None 0.158 0.35 Weak 31 -0.056 T7RS SBL 0.091 0.627 None 0.813 20 None AS -0.027 0.949 8 None 0.223 0.565 9 Weak SBC 0.463 0.019 25 Medium 0.038 0.825 37 None T7LI SBL -0.111 0.558 30 Weak -0.074 0.756 20 None AS -0.092 0.828 8 None -0.233 0.579 8 None SBC 0.193 25 Weak 0.178 0.291 37 Weak 0.269 T7RI SBL 0.142 0.447 31 Weak 0.126 0.596 20 Weak 8 AS -0.321 0.439 Medium 0.032 0.935 9 None SBC 0.188 0.379 24 Weak -0.122 0.485 35 Weak T8LS SBL -0.001 0.995 32 0.004 0.988 20 None None AS 0.444 0.27 8 Medium 0.598 0.089 9 Strong SBC -0.161 0.453 24 Weak -0.018 0.917 35 None T8RS SBL 0.06 0.743 32 -0.042 0.861 20 None None 8 0.252 AS 0.242 0.563 Weak 0.513 9 Weak 0.046 25 SBC -0.402 Medium -0.203 0.243 35 Weak -0.142 T8LI 0.437 32 0.085 0.739 SBL Weak 18 None AS -0.313 0.45 8 Medium 0.492 0.179 9 Medium SBC -0.084 0.69 25 None -0.072 0.681 35 None T8RI SBL 0.155 0.397 32 Weak -0.294 0.221 19 Weak AS 0.088 0.836 8 None 0.744 0.022 9 Strong SBC -0.009 0.966 26 None 0.037 0.826 38 None T9LS SBL -0.106 0.57 31 Weak 0.123 0.605 20 Weak AS -0.259 0.47 10 Medium 0.391 0.298 9 Medium SBC 0.122 0.552 26 Weak 0.179 0.282 38 Weak T9RS SBL -0.165 0.374 31 Weak 0.016 0.945 20 None 0.022 0.951 10 0.264 0.493 9 AS Weak None 25 SBC -0.028 0.869 -0.054 0.75 38 None None T9LI SBL 0.032 0.865 31 -0.034 0.885 20 None None 9 AS -0.508 0.134 10 Strong 0.519 0.152 Strong SBC -0.088 0.675 25 0.006 0.971 None 38 None T9RI SBL 0.422 0.02 30 Medium -0.238 0.312 20 Weak 0.421 AS 0.005 0.989 10 None 0.259 9 Medium

			Male				Female	2	
Facet	Sample	Correlation Coefficient	р	Ν	Strength of Correlation	Correlation Coefficient	р	N	Strength of Correlation
	SBC	-0.013	0.949	25	None	0.087	0.603	38	None
T10LS	SBL	-0.231	0.219	30	Weak	0.242	0.304	20	Weak
	AS	-0.544	0.104	10	Strong	0.184	0.653	9	Weak
	SBC	0.026	0.902	25	None	0.023	0.889	38	None
T10RS	SBL	0.229	0.214	31	Weak	0.231	0.324	20	Weak
	AS	-0.386	0.271	10	None	0.414	0.268	9	Medium
	SBC	-0.015	0.941	26	None	-0.058	0.728	38	None
T10LI	SBL	0.623	0.001	30	Strong	-0.275	0.24	20	Weak
	AS	0.481	0.159	10	Medium	-0.1	0.798	9	Weak
	SBC	-0.023	0.913	26	None	0.167	0.317	38	Weak
T10RI	SBL	-0.406	0.026	30	Medium	-0.19	0.423	20	Weak
	AS	-0.029	0.936	10	None	0.124	0.75	9	Weak
	SBC	0.165	0.421	26	Weak	0.09	0.59	38	None
T11LS	SBL	-0.174	0.366	29	Weak	0.201	0.382	21	Weak
	AS	0.32	0.401	9	Medium	0.013	0.973	9	None
	SBC	0.103	0.618	26	Weak	0.074	0.658	38	None
T11RS	SBL	-0.094	0.626	29	None	-0.142	0.538	21	Weak
	AS	0.014	0.972	9	None	0.455	0.219	9	Medium
	SBC	0.132	0.528	25	Weak	-0.045	0.791	38	None
T11LI	SBL	0.1	0.614	28	Weak	0.234	0.308	21	Weak
	AS	-0.078	0.842	9	None	0.14	0.719	9	Weak
	SBC	0.066	0.753	25	None	0.012	0.943	38	None
T11RI	SBL	0.164	0.396	29	Weak	0.301	0.185	21	Medium
	AS	0.382	0.31	9	Medium	0.113	0.772	9	Weak
	SBC	0.166	0.437	24	Weak	0.149	0.393	35	Weak
T12LS	SBL	0.091	0.646	28	None	0.09	0.697	21	None
	AS	0.084	0.83	9	None	-0.532	0.14	9	Strong
	SBC	-0.003	0.989	24	None	0.143	0.413	35	Weak
T12RS	SBL	0.238	0.222	28	Weak	0.147	0.526	21	Weak
	AS	0.324	0.395	9	Medium	-0.366	0.333	9	Medium
	SBC	-0.047	0.822	25	None	-0.017	0.924	35	None
T12LI	SBL	-0.057	0.769	29	None	-0.149	0.52	21	Weak
	AS	0.18	0.644	9	Weak	0.036	0.932	8	None
	SBC	0.306	0.137	25	Medium	-0.087	0.62	35	None
T12RI	SBL	0.003	0.989	29	None	-0.122	0.598	21	Weak
	AS	0.272	0.479	9	Weak	-0.146	0.708	9	Weak
	SBC	0.249	0.23	25	Weak	-0.018	0.92	35	None
L1LS	SBL	0.02	0.918	29	None	-0.199	0.401	20	Weak
	AS	0.422	0.224	10	Medium	-0.017	0.968	8	None

Male Female Facet Sample Correlation Strength of Correlation Strength of Ν Ν р р Coefficient Correlation Coefficient Correlation SBC -0.066 0.753 25 None -0.085 0.628 35 None L1RS SBL 0.161 0.412 28 Weak -0.152 0.523 20 Weak AS 0.472 02 9 Medium 0.191 0.651 8 Weak SBC -0.011 0.958 25 0.194 None 0.265 35 Weak 29 0.001 L1LI SBL 0.282 0.138 Weak 0.998 20 None 0.591 0.072 10 0.573 9 AS 0.107 Strong Strong 0.302 25 0.037 SBC 0.142 Medium 0.835 34 None 28 0.757 0.779 L1RI SBL -0.061 None 0.067 20 None AS 0.683 0.029 10 Strong 0.291 0.447 9 Weak SBC 0.342 0.088 26 Medium -0.037 0.828 36 None L2LS SBL 0.128 0.499 30 Weak -0.227 0.336 20 Weak AS -0.236 0.512 10 None -0.062 0.884 8 None 0.025 0.903 26 0.026 0.881 SBC None 36 None L2RS SBL 0.466 0.012 28 Medium -0.383 0.105 19 Medium AS 0.075 0.836 10 None -0.069 0.871 8 None SBC 0.364 0.067 26 Medium 0.118 0.493 36 Weak L2LI SBL -0.084 0.659 30 -0.507 0.027 19 None Strong AS 0.047 0.904 9 0.307 0.503 7 Medium None SBC 0.286 0.157 26 Weak 0.206 0.229 36 Weak L2RI SBL 0.279 0.135 30 Weak -0.483 0.036 19 Medium 9 0.032 AS 0.283 0.461 Weak 0.945 7 None 0.036 0.019 SBC 0.439 23 Medium 0.918 33 None L3LS 0.341 25 -0.409 0.131 15 SBL 0.112 Medium Medium AS 0.725 0.042 8 Strong 0.567 0.111 9 Strong 23 SBC 0.369 0.083 Medium -0.075 0.678 33 None L3RS SBL 0.502 0.015 23 -0.599 0.018 15 Strong Strong AS 0.254 0.543 8 Weak 0.262 0.496 9 Weak SBC 0.132 0.549 23 Weak -0.066 0.713 33 None L3LI SBL 0.395 0.062 23 Medium -0.452 0.079 16 Medium AS -0.207 0.623 8 Weak -0.011 0.978 9 None SBC -0.067 0.76 23 None -0.029 0.871 33 None 0.938 L3RI SBL 0.025 12 -0.502 0.169 9 None Strong 0.001 0.999 8 0.411 9 AS 0.272 Medium None SBC 0.157 0 4 4 3 26 Weak 0.16 0 3 4 4 37 Weak 7 L4LS SBL 0.379 0.402 Medium 4 9 AS 0.025 0.938 12 None -0.502 0.169 Strong SBC 0.103 0.615 26 -0.021 0.901 37 Weak None L4RS SBL 0.068 0.885 7 None 3 9 AS 0.025 0.938 12 None -0.502 0.169 Strong

			Male			Female			
Facet	Sample	Correlation Coefficient	р	Ν	Strength of Correlation	Correlation Coefficient	р	Ν	Strength of Correlation
	SBC	0.14	0.497	26	Weak	0.154	0.362	37	Weak
L4LI	SBL	0.636	0.125	7	Strong			4	
	AS	0.578	0.049	12	Strong	0.299	0.434	9	Weak
	SBC	0.076	0.714	26	None	0.149	0.38	37	Weak
L4RI	SBL	0.348	0.444	7	Medium			4	
	AS	0.394	0.231	11	Medium	0.539	0.168	8	Strong
	SBC	0.482	0.017	24	Medium	-0.015	0.934	33	None
L5LS	SBL	0.281	0.139	29	Weak	0.388	0.101	19	Medium
	AS	0.771	0.015	9	Strong	-0.11	0.796	8	None
	SBC	-0.024	0.911	24	None	-0.093	0.595	33	None
L5RS	SBL	0.178	0.374	27	Weak	0.446	0.056	19	Medium
	AS	0.5	0.17	9	Strong	0.09	0.832	8	None
	SBC	-0.077	0.722	24	None	0.002	0.99	33	None
L5LI	SBL	0.343	0.068	29	Medium	-0.225	0.354	19	Weak
	AS	-0.118	0.762	9	None	0.071	0.868	8	None
	SBC	-0.002	0.991	24	None	-0.1	0.581	33	Weak
L5RI	SBL	0.254	0.192	28	Weak	0.053	0.829	19	None
	AS	0.051	0.895	9	None	0.31	0.499	7	Medium

Appendix D. True prevalence rates

Appendix D-1 True prevalence rates for eburnation, pitting, osteophytes and vertebral

osteophytosis

481-508	Sex	True Eburnation	True Pitting	True Osteophytes	True Vertebral Osteophytosis
GC1	F	42.42	52.78	55.56	85.71
GC9	F	54.35	80.43	29.35	0.00
GC14	F	55.43	89.01	55.43	39.13
GC32	F	74.16	95.12	21.35	0.00
GC37	F	42.59	93.65	46.30	69.23
GC81	F	100.00	88.14	26.79	53.85
GC92	F	65.63	79.17	51.04	63.64
GC100	F	44.21	77.66	13.68	0.00
GC103	F	20.43	46.97	25.53	36.36
GC152	F	31.43	60.98	26.47	55.56
B250	F	55.56	88.57	32.14	0.00
B288	F	29.49	86.79	19.48	0.00
B309	F	0.00	79.75	18.64	0.00
B420	F	7.35	78.43	17.65	0.00
A56	F	30.43	86.96	13.04	0.00
A64	F	0.00	52.00	33.33	0.00
A447	F	1.92	70.18	17.24	0.00
A765	F	0.00	27.08	0.00	0.00
A780	F	0.00	57.14	3.70	0.00
A640	F	0.00	44.44	10.00	0.00
A270	F	0.00	10.00	28.26	0.00
SBC8	F	25.00	3.80	7.14	0.00
SBC10	F	27.08	15.63	28.72	0.00
SBC13	F	50.00	37.50	28.13	0.00
SBC17	F	11.96	44.57	38.04	31.82
SCB29	F	7.29	16.67	26.04	56.52
SBC33	F	27.08	13.54	28.13	39.13
SBC43	F	0.00	0.00	4.55	0.00
SBC44	F	4.17	2.94	2.08	0.00
SBC45	F	10.00	5.00	21.25	5.00
SBC52	F	0.00	4.55	2.27	0.00
SBC63	F	11.27	23.21	16.90	0.00
SBC66	F	10.00	15.22	21.05	47.06
SBC67	F	12.50	18.75	15.63	0.00
SBC72	F	5.26	2.78	3.95	0.00
SBC106	F	8.05	27.27	6.90	0.00
SBC117	F	2.27	22.73	3.41	0.00

ID	Sex	True	True Pitting	True	True Vertebral
SBC120	F	3 41	23.96	10.23	5 00
SBC120	F	2 38	10.00	3 57	0.00
SBC122	F	2.50	8 33	15 10	80.00
SBC129	F F	42.30	8.33 E7.0E	13.1 3	80.00
SBC130	F F	42.39	37.35	6.04	45.45
SBC133	г г	0.00	34.72	7.20	10.07
SBC144	r r	1.00	31.23	1.00	0.00
SBC152	F	1.09	22.83	1.09	0.00
SBC153	F	1.09	11.96	6.52	0.00
SBC154	F	7.61	20.65	16.48	0.00
SBC101	F	29.17	20.65	26.09	27.27
SBC98	F	8.33	17.71	11.46	0.00
SBC93	F	2.78	50.00	27.78	44.44
SBC156	F	10.29	26.47	22.06	0.00
SBC113	F	3.41	14.94	8.05	0.00
SBC167	F	4.55	21.59	4.55	19.05
SBC177	F	0.00	0.00	2.08	0.00
SBC178	F	1.04	5.21	10.42	4.35
SBC182	F	21.74	30.43	30.43	0.00
SBC203	F	1.04	0.00	18.75	0.00
SBC209	F	4.05	34.67	17.11	21.05
SBC214	F	1.27	0.00	12.66	0.00
SBC215	F	0.00	6.25	17.50	5.26
SBC219	F	6.25	33.33	11.46	0.00
SBC220	F	5.95	27.38	44.05	100.00
SBC225	F	8.70	14.13	7.61	0.00
FAO901641	F	0.00	45.78	16.87	5.26
FAO901755	F	3.13	30.21	14.58	0.00
FAO901703	F	4.76	46.43	19.05	9.52
FAO901417	F	8.33	43.75	28.42	21.74
FAO901874	F	13.16	24.00	28.00	38.89
FAO901152	F	4.05	29.73	32.43	58.82
FAO901207	F	3.70	17.28	7.41	0.00
FAO901369	F	0.00	15.87	22.22	40.00
FAO901946	F	18.75	37.50	31.25	40.91
FAO901793	F	7.29	27.37	13.68	0.00
FAO901608	F	7.14	10.71	28.57	0.00
FAO901123	F	20.83	47.92	51.04	78.95
FAO901954	F	0.00	13.54	36.46	0.00
FAO901913	F	13.25	10.84	51.81	52.94
FAO901653	F	1.05	11.58	21.05	0.00
FAO901343	F	17.31	36.54	39.22	75.00

ID	Sex	True Eburnation	True Pitting	True Osteophytes	True Vertebral
FAO901376	F	0.00	25.00	5.43	0.00
FAO901809	F	0.00	22.92	17.71	0.00
FAO901887	F	0.00	2.17	15.22	4.55
FAO901799	F	0.00	1.14	21.59	40.91
FAO901409	F	14.58	43.75	54.00	100.00
FAO901278	F	0.00	7.37	1.04	0.00
FAO901519	F	0.00	23.96	21.88	0.00
FAO901893	F	1.30	15.38	10.39	0.00
GC2A	м	13.68	13.83	4.21	0.00
GC5	М	79.69	85.71	62.50	18.75
GC8	М	7.50	44.78	58.33	12.50
GC22	М	94.68	97.78	29.79	26.09
GC80	М	95.00	100.00	36.25	85.00
GC84	М	19.12	95.95	11.76	18.75
GC90	М	95.74	85.42	20.21	21.74
GC117	М	50.00	66.13	10.53	11.11
GC121	М	16.30	79.55	11.96	0.00
GC149	М	18.48	56.84	3.26	0.00
B282B	м	52.83	94.29	7.55	0.00
B311	М	8.00	49.09	45.95	0.00
B338	м	0.00	83.53	30.65	0.00
B385L	м	4.26	72.92	20.21	0.00
A49	м	0.00	93.33	0.00	0.00
A43	м	0.00	83.33	0.00	0.00
A747	м	0.00	20.34	31.51	0.00
A777	м	0.00	40.00	36.36	0.00
A768	м	4.55	16.67	27.27	0.00
A591	М	0.00	57.89	12.50	0.00
A821	М	0.00	65.12	8.82	0.00
A53	М	0.00	88.16	12.96	72.22
A111	М	2.82	69.05	34.55	50.00
A315	М	0.00	0.00	30.77	0.00
A644	М	0.00	45.45	0.00	0.00
A812	М	0.00	10.81	36.36	0.00
A334	М	0.00	0.00	5.56	0.00
A783	М	0.00	34.62	10.53	0.00
A453	М	0.00	26.92	20.00	0.00
A309	М	0.00	12.00	11.11	0.00
A444	М	0.00	5.26	0.00	0.00
A481	М	0.00	17.31	3.57	0.00

ID	Sex	True	True Pitting	True Osteophytes	True Vertebral
SBC2	М	16.30	0.00	20.00	0.00
SBC12	M	44.57	22.83	15.96	0.00
SBC31	M	3.26	7.61	17.39	0.00
SBC58	M	7.61	12.50	22.83	72.73
SBC61	М	13.92	26.83	10.13	0.00
SBC67	М	4.17	9.72	1.39	0.00
SBC75	М	2.38	2.27	11.90	0.00
SBC84	М	5.21	5.26	2.08	13.04
SBC108	М	1.04	4.17	4.17	0.00
SBC110	М	3.13	20.83	11.46	11.76
SBC119	М	1.04	20.45	7.29	0.00
SBC124	М	3.41	10.00	22.73	71.43
SBC131	М	6.25	20.00	8.42	26.09
SBC143	М	8.70	40.22	14.13	22.73
SBC94	М	31.82	13.64	23.86	86.36
SBC158	М	27.17	52.17	39.13	68.18
SBC107	М	39.47	48.05	55.26	70.00
SBC112	М	4.88	12.35	4.88	0.00
SBC170	М	6.52	17.58	5.43	0.00
SBC171	М	8.70	50.00	44.57	5.88
SBC179	М	18.18	30.68	12.50	47.62
SBC196	М	1.06	6.25	6.25	17.39
SBC205	М	5.21	8.33	33.33	95.65
SBC221	М	4.35	39.13	18.95	0.00
SBC227	М	7.87	17.98	25.84	0.00
SBC224	М	0.00	3.13	20.83	4.35
FAO901860	М	0.00	44.94	17.98	0.00
FAO901635	М	15.12	57.47	29.07	54.55
FAO901546	М	8.51	44.68	26.60	33.33
FAO901338	М	3.95	9.21	28.95	25.00
FAO901549	М	8.57	48.57	38.57	93.33
FAO901390	М	3.26	36.96	32.61	17.39
FAO901991	М	17.71	34.38	27.08	61.90
FAO901819	М	1.06	26.60	14.89	0.00
FAO901420	М	0.00	6.25	22.50	0.00
FAO901845	М	23.96	17.71	27.08	0.00
FAO901591	М	6.25	38.54	19.79	42.86
FAO901767	М	1.04	17.71	11.46	0.00
FAO901578	М	1.25	35.00	13.75	0.00
FAO901116	М	0.00	33.78	17.81	5.56

		True		True	True Vortobral
ID	Sex	Eburnation	True Pitting	Osteophytes	Osteophytosis
FAO901925	М	0.00	36.25	22.50	0.00
FAO901454	М	0.00	8.33	4.17	4.55
FAO901932	М	3.57	17.86	28.57	0.00
FAO901606	М	2.08	20.83	46.88	38.10
FAO901155	М	8.93	16.07	28.57	30.77
FAO901589	М	2.27	6.98	30.23	18.18
FAO901721	М	3.13	25.00	22.92	0.00
FAO901745	М	26.32	46.32	56.84	90.91
FAO901827	М	1.64	13.11	8.20	46.67
FAO901885	М	3.49	11.76	17.65	95.00
FAO901785	М	0.00	18.39	14.94	0.00
FAO901251	М	3.33	16.67	6.67	7.14
FAO901200	М	1.32	3.95	9.21	44.44
FAO901751	М	0.00	16.18	14.71	0.00
FAO901825	М	2.13	3.19	23.40	13.64
FAO901345	М	0.00	53.06	30.61	81.82
FAO901521	М	1.61	12.70	7.94	0.00
FAO902001	М	2.17	5.43	30.43	0.00
FAO901449	М	0.00	10.00	1.67	0.00
FAO901500	М	9.47	26.32	29.47	36.36
FAO901868	М	0.00	41.46	17.07	20.00

Appendix D-2 True prevalence rates for osteoarthritis (OA) by individual

ID	Sex	True OA
GC1	F	36.92
GC9	F	79.37
GC14	F	52.75
GC32	F	54.95
GC37	F	42.59
GC81	F	59.1
GC92	F	65.26
GC100	F	51.35
GC103	F	55.56
GC152	F	27.37
B250	F	12.09
B288	F	3.3
B309	F	26.32
B420	F	4.17
A56	F	10
A64	F	7.69
A447	F	14.1
A765	F	8.86
A780	F	3.16
A640	F	2.3
A270	F	1.05
SBC8	F	8.6
SBC10	F	13.83
SBC13	F	7.69
SBC17	F	94.62
SCB29	F	73.86
SBC33	F	94.94
SBC43	F	19.12
SBC44	F	95.7
SBC45	F	44.68
SBC52	F	20.43
SBC63	F	15.22
SBC66	F	18.68
SBC67	F	30.43
SBC72	F	53.85
SBC106	F	29.87
SBC117	F	0
SBC120	F	6.67
SBC122	F	0
SBC129	F	45.05

ID True OA Sex SBC130 F 50.53 SBC133 F 7.37 F 0 SBC144 F SBC152 0 SBC153 F 11.43 SBC154 F 4.23 F 12.63 SBC101 SBC98 F 5.33 SBC93 F 2.41 SBC156 F 5.26 F SBC113 8.05 SBC167 F 1.05 F 0 SBC177 SBC178 F 14.12 F 0 SBC182 F 0 SBC203 F 0 SBC209 SBC214 F 0 SBC215 F 0 F SBC219 16.48 F SBC220 25.35 F SBC225 4.94 F FAO901641 4.6 F 8.7 FAO901755 FAO901703 F 0 FAO901417 F 6.02 FAO901874 F 0 F FAO901152 17.31 FAO901207 F 26.6 F FAO901369 1.67 FAO901946 F 3.53 F FAO901793 0 FAO901608 F 3.33 FAO901123 F 1.32 FAO901954 F 0 F FAO901913 2.15 F FAO901653 2.2 FAO901343 F 0 FAO901376 F 0 FAO901809 F 0 FAO901887 F 0 FAO901799 F 27.47 F FAO901409 38.67 FAO901278 F 3.45

Appendix D-2 True prevalence rates for osteoarthritis (OA) by individual continued

	fan ante a authorite	10 A) L		-
nrovalonco ratos	TOP OCTOORTOPITIC	$(1)\Delta n$	ובווחועותחו ע	CONTINUER
prevalence rates	ioi osceoai cinita		y maiviauai	continucu

ID	Sex	True OA	
FAO901519	F	6.59	
FAO901893	F	1.05	
GC2A	М	18.39	
GC5	М	21.98	
GC8	М	1.05	
GC22	М	1.05	
GC80	М	5.26	
GC84	М	4	
GC90	М	1.27	
GC117	М	0	
GC121	М	6.32	
GC149	М	4.4	
B282B	М	7.95	
B311	М	8.79	
B338	М	2.33	
B385L	М	3.16	
A49	М	0	
A43	М	0	
A747	М	0	
A777	М	0	
A768	М	14.89	
A591	М	0	
A821	М	0	
A53	М	1.64	
A111	М	0	
A315	М	9.57	
A644	М	0	
A812	М	1.3	
A334	М	4.21	
A783	М	7.6	
A453	М	0	
A309	М	0	
A444	М	0	
A481	М	0	
SBC2	М	0	
SBC12	М	0	
SBC31	М	0	
SBC58	М	2.82	
SBC61	М	0	
SBC67	М	0	
SBC75	М	0	
SBC84	М	0	
SBC108	М	2.41	
SBC110	М	42.39	
ID True OA Sex SBC119 6.32 Μ SBC124 Μ 0 8.79 SBC131 Μ SBC143 Μ 7.37 SBC94 1.1 Μ SBC158 Μ 29.47 Μ 8.42 SBC107 SBC112 Μ 32.18 0 SBC170 Μ SBC171 Μ 3.16 SBC179 Μ 4.82 М 8.42 SBC196 4.11 SBC205 Μ SBC221 М 3.7 0 SBC227 Μ 18.95 SBC224 М 4.21 FAO901860 Μ FAO901635 Μ 0 7.14 FAO901546 Μ FAO901338 Μ 1.27 FAO901549 Μ 0 FAO901390 Μ 0 FAO901991 Μ 0 FAO901819 Μ 3.61 FAO901420 Μ 2.11 FAO901845 Μ 4.3 FAO901591 Μ 30.43 FAO901767 Μ 4.55 FAO901578 Μ 0 1.92 FAO901116 Μ FAO901925 0 Μ FAO901454 Μ 20.25 FAO901932 Μ 3.45 FAO901606 1.1 Μ FAO901155 М 6.59 FAO901589 Μ 2.86 FAO901721 М 10.45 FAO901745 Μ 4 FAO901827 8.7 Μ FAO901885 Μ 13.16 FAO901785 Μ 3.3 FAO901251 Μ 17.89

Appendix D-2 True prevalence rates for osteoarthritis (OA) by individual continued

ID	Sex	True OA
FAO901200	М	1.08
FAO901751	М	24.21
FAO901825	М	21.05
FAO901345	М	6.32
FAO901521	М	1.05
FAO902001	М	12.2
FAO901449	М	1.06
FAO901500	М	8.93
FAO901868	М	4.6

Appendix D-2 True prevalence rates for osteoarthritis (OA) by individual continued

Appendix E. Overall summary of extrinsic variables strongly correlated to facet size and angle

Predictor Variable	Sample	Sex	C1 LS	C1 RS	C1 LI	C1 RI	C2 LS	C2 RS	C2 LI	C2 RI	C3 LS	C3 RS	C3 LI	C3 RI	C4 LS	C4 RS	C4 LI	C4 RI	C5 LS	C5 RS	C5 LI	C5 RI	C6 LS	C6 RS	C6 LI	C6 RI	C7 LS	C7 RS	C7 LI	C7 RI
Sex/facet size	SBC																													
	SBL																													
	AS																													
Sex/facet angle	SBC																													
	SBL																													
	AS																													
Age/facet size	SBC	М																												
		F																												
	SBL	М																												
		F																												
	AS	М																												
		F																												
Age/facet angle	SBC	М																												
		F																												
	SBL	М																												
		F																												
	AS	М																												
		F																												

Appendix Table E-1 Statistically significant results from Mann-Witney U test for sex and age with facet size and angle, cervical region

Predictor Variable	Sample	Sex	C1	C1	C1	C1	C2	C2	C2	C2	C3	C3	С3	С3	C4	C4	C4	C4	C5	C5	C5	C5	C6	C6	C6	C6	C7	C7	C7	C7
			LS	RS	LI	RI																								
femoral	SBC	М																												
robusticity/facet		F																L				R								
size	SBL	М																												
		F															L												Í	
	AS	М																LR												
		F								LR	R	LR			LR			L			LR	LR	L	LR		R	LR	LR	LR	
femoral	SBC	М					LR		LR																					L
robusticity/facet		F	L			L	L	L	LR	R			R				LR	LR			LR				L				LR	
angie	SBL	М												LR																
		F													L	L		L	R	L				L						
	AS	М					LR		LR																					L
		F		L		L	L	L	LR	R			R				LR	LR			LR				L				LR	

Appendix Table E-2 Strong correlation between femoral robusticity and facet size and angle, cervical region

Predictor Variable	Sample	Sex	C1 LS	C1 RS	C1 LI	C1 RI	C2 LS	C2 RS	C2 LI	C2 RI	C3 LS	C3 RS	C3 LI	C3 RI	C4 LS	C4 RS	C4 LI	C4 RI	C5 LS	C5 RS	C5 LI	C5 RI	C6 LS	C6 RS	C6 LI	C6 RI	C7 LS	C7 RS	C7 LI	C7 RI
Humeral/facet	SBC	М																												
size asymmetry		F																												
	SBL	М																												
		F																												
	AS	М																												
		F																												
Humeral/facet	SBC	М																												
angle		F																												
asymmetry	SBL	М																												
		F																												
	AS	М																												
		F																												

Appendix Table E-3 Facets with strong correlation between humeral and facet size and angle asymmetry, cervical region

Predictor Variable	Sample	Sex	C1 LS	C1 RS	C1 LI	C1 RI	C2 LS	C2 RS	C2 LI	C2 RI	C3 LS	C3 RS	C3 LI	C3 RI	C4 LS	C4 RS	C4 LI	C4 RI	C5 LS	C5 RS	C5 LI	C5 RI	C6 LS	C6 RS	C6 LI	C6 RI	C7 LS	C7 RS	C7 LI	C7 RI
Ebtot/facet size	SBC	М																												
		F																												
	SBL	М																												
		F																												
	AS	М																												
		F																												
Ebtot/facet	SBC	М											1																	
angle		F																												
	SBL	М																												
		F																												
	AS	М																												
		F																												

Appendix Table E-4 Facets with strong correlation between ebtot and facet size and angle, cervical region

Predictor Variable	Sample	Sex	C1 LS	C1 RS	C1 LI	C1 RI	C2 LS	C2 RS	C2 LI	C2 RI	C3 LS	C3 RS	C3 LI	C3 RI	C4 LS	C4 RS	C4 LI	C4 RI	C5 LS	C5 RS	C5 LI	C5 RI	C6 LS	C6 RS	C6 LI	C6 RI	C7 LS	C7 RS	C7 LI	C7 RI
Pitot/	SBC	М																												
facet size		F																												
	SBL	М																												
		F																												
	AS	М																												
		F																												
Pitot/	SBC	М																												
facet angle		F																												
	SBL	М																												
		F																												
	AS	М																												
		F																												

Appendix Table E-5 Facets with strong correlation between pitot and facet size and angle, cervical region

Predictor Variable	Sample	Sex	C1 LS	C1 RS	C1 LI	C1 RI	C2 LS	C2 RS	C2 LI	C2 RI	C3 LS	C3 RS	C3 LI	C3 RI	C4 LS	C4 RS	C4 LI	C4 RI	C5 LS	C5 RS	C5 LI	C5 RI	C6 LS	C6 RS	C6 LI	C6 RI	C7 LS	C7 RS	C7 LI	C7 RI
Ostot/	SBC	М																												
facet size		F																												
	SBL	М																												
		F																												
	AS	М																												
		F																												
Ostot/	SBC	М																												
facet angle		F																												
	SBL	М																												
		F																												
	AS	М																												
		F																												

Appendix Table E-6 Facets with strong correlation between ostot and facet size and angle, cervical region

Predictor Variable	Sample	Sex	C1 LS	C1 RS	C1 LI	C1 RI	C2 LS	C2 RS	C2 LI	C2 RI	C3 LS	C3 RS	C3 LI	C3 RI	C4 LS	C4 RS	C4 LI	C4 RI	C5 LS	C5 RS	C5 LI	C5 RI	C6 LS	C6 RS	C6 LI	C6 RI	C7 LS	C7 RS	C7 LI	C7 RI
Cvostot/	SBC	М																												
Facet size		F																												
	SBL	М																												
		F																												
	AS	М																												
		F																												
Cvostot/	SBC	М																												
facet angle		F																												
	SBL	М																												
		F																												
	AS	М																												
		F																												

Appendix Table E-7 Facets with strong correlation between cvostot and facet size and angle, cervical region

Predictor Variable	Sample	T1	T1	T1	T1	T2	T2	T2	T2	Т3	Т3	Т3	Т3	T4	T4	T4	T4	T5	T5	T5	T5	T6	T6	Т6	Т6
		LS	RS	LI	RI																				
Sex/facet size	SBC																								
	SBL																								
	AS																								
Sex/facet angle	SBC																								
	SBL																								
	AS																								
Age/facet size	SBC																								
	SBL																								
	AS																								
Age/facet angle	SBC																								
	SBL																								
	AS																								

Appendix Table E-8 Facets with statistically significant results from Mann-Whitney U test for sex and age with facet size and angle, thoracic region

Predictor Variable	Sample	Т7	Т7	Т7	Т7	Т8	Т8	Т8	Т8	Т9	Т9	Т9	Т9	T10	T10	T10	T10	T11	T11	T11	T11	T12	T12	T12	T12
		LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI
Sex/facet size	SBC																								
	SBL																								
	AS																								
Sex/facet angle	SBC																								
	SBL																								
	AS																								
Age/facet size	SBC																								
	SBL																								
	AS																								
Age/facet angle	SBC																								
	SBL																								
	AS																								

Appendix Table E-8 Facets with statistically significant results from Mann-Whitney U test for sex and age with facet size and angle, thoracic region continued

Predictor Variable	Sample	Sex	T1	T1	T1	T1	T2	T2	T2	T2	Т3	Т3	Т3	Т3	T4	T4	T4	T4	T5	T5	T5	T5	Т6	Т6	Т6	Т6
			LS	RS	LI	RI																				
Femoral	SBC	М														R										
robusticity/facet		F						R										R								
Size	SBL	М																								
		F							L	L	L			L		L		R	L							
	AS	М								L												L		L		
		F																LR								
Femoral	SBC	М					LR	LR																	L	L
robusticity/facet		F																				R		L		L
angle	SBL	М																								
		F			LR	R		R	R			R	R		L		R	R		R					L	
	AS	М					LR	LR																	L	L
		F	LR																			R		L		L

Appendix Table E-9 Facets with strong correlation between femoral robusticity and facet size and angle, thoracic region

Predictor Variable	Sample	Sex	Т7	T7	T7	Т7	Т8	Т8	Т8	Т8	Т9	Т9	Т9	Т9	T10	T10	T10	T10	T11	T11	T11	T11	T12	T12	T12	T12
			LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI
Femoral	SBC	М						R				R														R
robusticity/facet		F			L	L																				
size	SBL	М																								
		F																		R			L			
	AS	М				LR																				
		F																								
Femoral	SBC	М				LR																				
robusticity/facet		F			L	L																	R			LR
angie	SBL	М												LR				LR								
		F	R				L	L		R									R						L	L
	AS	М																								
		F																					R			LR

Appendix Table E-9 Facets with strong correlation between femoral robusticity and facet size and angle, thoracic region continued

Predictor Variable	Sample	Sex	T1 LS	T1 RS	T1 LI	T1 RI	T2 LS	T2 RS	T2 LI	T2 RI	T3 LS	T3 RS	T3 LI	T3 RI	T4 LS	T4 RS	T4 LI	T4 RI	T5 LS	T5 RS	T5 LI	T5 RI	T6 LS	T6 RS	T6 LI	T6 RI
humeral/facet size	SBC	М																								
asymmetry		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								
humeral/facet	SBC	М																								
angle asymmetry		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								

Appendix Table E-10 Facets showing strong correlation between humeral and facet size and angle, thoracic region

Predictor Variable	Sample	Sex	Т7	Τ7	Τ7	Т7	Т8	Т8	Т8	Т8	Т9	Т9	Т9	Т9	T10	T10	T10	T10	T11	T11	T11	T11	T12	T12	T12	T12
			LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI
humeral/facet size	SBC	М																								
asymmetry		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								
humeral/facet	SBC	М																								
angle asymmetry		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								

Appendix Table E-10 Facets showing strong correlation between humeral and facet size and angle, thoracic region

Predictor Variable	Sample	Sex	T1	T1	T1	T1	T2	T2	T2	T2	T3	T3	Т3	T3	T4	T4	T4	T4	T5	T5	T5	T5	T6	T6	T6	T6
			LS	RS	LI	RI																				
Ebtot/facet size	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								
Ebtot/facet angle	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								

Appendix Table E-11 Facets with strong correlation between ebtot and facet size and angle, thoracic region

Predictor Variable	Sample	Sex	Т7	Т7	Т7	Т7	Т8	Т8	Т8	Т8	Т9	Т9	Т9	Т9	T10	T10	T10	T10	T11	T11	T11	T11	T12	T12	T12	T12
			LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI
Ebtot/facet size	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								
Ebtot/facet angle	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								

Appendix Table E-11 Facets with strong correlation between ebtot and facet size and angle, thoracic region continued

Predictor Variable	Sample	Sex	T1	T1	T1	T1	T2	T2	T2	T2	T3	T3	T3	T3	T4	T4	T4	T4	T5	T5	T5	T5	T6	T6 PS	T6	T6
			LS	кэ	LI	RI	LS	кs	LI	RI	LS	кэ	LI	RI	LS	кэ	LI	RI	LS	кэ	LI	RI	LS	кs	LI	RI
Pitot/facet size	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								
Pitot/facet angle	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								

Appendix Table E-12 Facets with strong correlation between pitot and facet size and angle, thoracic region

Predictor Variable	Sample	Sex	Т7	T7	T7	T7	Т8	Т8	Т8	Т8	Т9	Т9	Т9	Т9	T10	T10	T10	T10	T11	T11	T11	T11	T12	T12	T12	T12
			LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI
Pitot/facet size	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								
Pitot/facet angle	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								

Appendix Table E-12 Facets with strong correlation between pitot and facet size and angle, thoracic region continued

Predictor Variable	Sample	Sex	T1	T1	T1	T1	T2	T2	T2	T2	Т3	Т3	Т3	Т3	T4	T4	T4	T4	T5	T5	T5	T5	T6	T6	T6	Т6
			LS	RS	LI	RI																				
Ostot/facet size	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								
Ostot/facet angle	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								

Appendix Table E-13 Facets with strong correlation between ostot and facet size and angle, thoracic region

Predictor Variable	Sample	Sex	Т7	T7	Т7	Т7	Т8	Т8	Т8	Т8	Т9	Т9	Т9	Т9	T10	T10	T10	T10	T11	T11	T11	T11	T12	T12	T12	T12
			LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI
Ostot/facet size	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								
Ostot/facet angle	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								

Appendix Table E-13 Facets with strong correlation between ostot and facet size and angle, thoracic region continued

Appendix Table E-14 Facets with strong correlation between cvostot and facet size and angle, thoracic region

Predictor Variable	Sample	Sex	T1	T1	T1	T1	Т2	T2	T2	T2	Т3	Т3	Т3	Т3	T4	T4	T4	T4	T5	T5	T5	T5	Т6	T6	Т6	Т6
			LS	RS	LI	RI																				
Cvostot/facet size	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								
Cvostot/facet	SBC	М																								
angle		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								

Predictor Variable	Sample	Sex	Т7	Т7	Т7	Т7	Т8	Т8	Т8	Т8	Т9	Т9	Т9	Т9	T10	T10	T10	T10	T11	T11	T11	T11	T12	T12	T12	T12
			LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI	LS	RS	LI	RI
Cvostot/facet size	SBC	М																								
		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								
Cvostot/facet	SBC	М																								
angle		F																								
	SBL	М																								
		F																								
	AS	М																								
		F																								

Appendix Table E-14 Facets with strong correlation between cvostot and facet size and angle, thoracic region continued

Appendix Table E-15 Facets with statistically significant results from Mann-Whitney U test for sex and age with facet size and angle, lumbar region

Predictor Variable	Sample	L1	L1	L1	L1	L2	L2	L2	L2	L3	L3	L3	L3	L4	L4	L4	L4	L5	L5	L5	L5
		LS	RS	LI	RI																
Sex/facet size	SBC																				
	SBL																				
	AS																				
Sex/facet angle	SBC																				
	SBL																				
	AS																				
Age/facet size	SBC																				
	SBL																				
	AS																				
Age/facet angle	SBC																				
	SBL																				
	AS																				

Predictor Variable	Sample	Sex	L1	L1	L1	L1	L2	L2	L2	L2	L3	L3	L3	L3	L4	L4	L4	L4	L5	L5	L5	L5
			LS	RS	LI	RI																
Femoral	SBC	М									R											
robusticity/facet size		F																				
	SBL	М																				
		F													L	L	L					
	AS	М	R		L		L					R										
		F																				
Femoral	SBC	М		L	R												LR				L	
robusticity/facet		F		R				R		L												
angle	SBL	М																				
		F	L	L		L		L	L	L		L		L					L			
	AS	М		L	R												LR				L	
		F		R				R		R												

Appendix Table E-16 Facets with strong correlation between femoral robusticity and facet size and angle, lumbar region

		0			-					0				-0	-							
Predictor Variable	Sample	Sex	L1	L1	L1	L1	L2	L2	L2	L2	L3	L3	L3	L3	L4	L4	L4	L4	L5	L5	L5	L5
			LS	RS	LI	RI																
Humeral/facet size	SBC	М																				
asymmetry		F																				
	SBL	М																				
		F																				
	AS	м																				
		F																				
Humeral/facet angle	SBC	м																				
asymmetry		F																				
	SBL	М																				
		F																				
	AS	М																				
		F																				

Appendix Table E-17 Facets with strong correlation between humeral and facet size and angle asymmetry, lumbar region

Predictor Variable	Sample	Sex	L1	L1	L1	L1	L2	L2	L2	L2	L3	L3	L3	L3	L4	L4	L4	L4	L5	L5	L5	L5
			LS	RS	LI	RI																
Ebtot/facet size	SBC	М																				
		F																				
	SBL	М																				
		F																				
	AS	М																				
		F																				
Ebtot/facet angle	SBC	М																				
		F																				
	SBL	М																				
		F																				
	AS	М																				
		F																				

Appendix Table E-18 Facets with strong correlation between ebtot and facet size and angle, lumbar region

Appendix Table E-19 F	acets with strong	correlation between	pitot and facet si	ze and angle. I	umbar region

Predictor Variable	Sample	Sex	L1	L1	L1	L1	L2	L2	L2	L2	L3	L3	L3	L3	L4	L4	L4	L4	L5	L5	L5	L5
			LS	RS	LI	RI																
Pitot/facet size	SBC	М																				
		F																				
	SBL	м																				
		F																				
	AS	М																				
		F																				
Pittot/facet angle	SBC	М																				
		F																				
	SBL	М																				
		F																				
	AS	М																				
		F																				

Predictor Variable	Sample	Sex	L1	L1	L1	L1	L2	L2	L2	L2	L3	L3	L3	L3	L4	L4	L4	L4	L5	L5	L5	L5
			LS	RS	LI	RI																
Ostot/facet size	SBC	М																				
		F																				
	SBL	М																				
		F																				
	AS	М																				
		F																				
Ostot/facet angle	SBC	М																				
		F																				
	SBL	М																				
		F																				
	AS	М																				
		F																				

Appendix Table E-20 Facets with strong correlation between ostot and facet size and angle, lumbar region

Predictor Variable	Sample	Sex	L1	L1	L1	L1	L2	L2	L2	L2	L3	L3	L3	L3	L4	L4	L4	L4	L5	L5	L5	L5
			LS	RS	LI	RI																
Cvostot/facet size	SBC	М	•																			
		F																				
	SBL	М																				
		F																				
	AS	М																				
		F																				
Cvostot/facet angle	SBC	М																				
		F																				
	SBL	М																				
		F																				
	AS	М																				
		F																				

R

R

Appendix F. Sample Data Sheet

Site Context Age Data: Pubic symphysis R **Auricular surface** L L Todd (1-10) (1-8)Suchey-Brooks (1-6) Sex Data R Skull Pelvis L Μ L Ventral arc (1-3) Nuchal crest (1-5) Sub-pubic cavity (1-3) Mastoid process (1-5) Ischiopubic ramus bridge (1-3) Supra-orbital margin (1-5) Greater sciatic notch (1-5) Supra-orbital ridge (1-5) L Osteometrics R Average Femoral head Diam XLF LCT XLH XLR XLU XLG Femur A/P Femur M/L Humerus A/P Humerus M/L PUM SPU DCOX IIMT

Appendix Table F-1 Sample Data Sheet

ISMM SCOX SS SA SIS VEAC

.

Cemetery ID																								
Level	C1	C2	C3	C4	C5	C6	С7	T1	Т2	Т3	Т4	Т5	Т6	T7	Т8	Т9	T10	T11	T12	L1	L2	L3	L4	L5
L sup facet width																								
R sup facet width																								
L sup facet height																							 	
R sup facet height																								
L inf facet width																								
R inf facet width																								
L inf facet height																								
R inf facet height																								
L sup eburnation S/E																								
R sup eburnation S/E																								
L inf eburnation S/E																								
R inf eburnation																								

Level	C1	C2	СЗ	C4	C5	C6	С7	т1	т2	тз	Т4	Т5	т6	т7	т8	Т9	T10	T11	T12	L1	L2	L3	L4	L5
L sup pitting																								
R sup pitting																								
L inf pitting																								
R inf pitting																								
L sup osteophyte																								
R sup osteophyte																								
L inf osteophyte																								
R inf osteophyte																								
L sup sag angle																								
R sup sag angle																								
L inf sagittal angle																								
R inf sagittal angle																								
VOP sup																								
VOP Inf																								

List of References

List of References

Abelson, R. P. (1995). Statistics as Principled Argument. Hillsdale, NJ: Lawrence Erlbaum.

- Adams, B. J., and Byrd, J. E. (2002). Inter-observer variation of selected postcranial skeletal measurements. *Journal of Forensic Science* 47(6) pp. 1193-1202.
- Adams, M., Bogduk, N., Burton, K., and Dolan, P. (2006). *The Biomechanics of Back Pain* (2nd Edition). New York: Elsevier Churchill Livingstone.
- Adams, M., and Roughly, P. (2006). What is intervertebral disc degeneration, and what causes it? *Spine* 31(18), pp. 2151-2161.
- Adams, M. A., and Hutton, W. C. (1980). The effect of posture on the role of the apophyseal joints in resting inter vertebral forces. *Journal of Bone Joint Surgery (Br) 62* pp. 358-362.
- Aeby, C. (1879). Die Alterverscheidenheiten der menslichen Wirbelsaule. Archiv für Anatomie Physiologie. Leipzig pp. 77-137.
- Agarwal, S. C. (2008). Light and broken bones: Examining and interpreting bone loss and osteoporosis in past populations. In Katzenberg, M. A., and Saunders, S. R. (eds).
 Biological Anthropology of the Human Skeleton. New York: Wiley-Liss pp. 387-442.
- Agarwal, S. C. (2012). The past of sex, gender, and health: Bioarchaeology of the aging skeleton. American Anthropologist 114(2) pp. 332-335.
- Agarwal, S. C., and Beauchesne, P. (2011). Development and plasticity of the aged skeleton. In Agarwal, S. C., and Glencross B. A. (eds). *Social Bioarchaeology*. Chichester West Sussex: Wiley -Blackwell Publishing pp. 312-332.
- Agarwal, S. C., and Stuart-Macadam, P. (2004). Reproduction and the female skeleton. In Agarwal S. C., and Stout, S. D. (eds). *Bone Loss and Osteoporosis*. New York: Klewer Plenum Academic Press.
- al-Oumaoui, I., Jimenez-Brobeil, S., and du Souich, P. (2004). Markers of activity patterns in some populations of the Iberian Peninsula. *International Journal of Osteoarchaeology* 14 pp. 343-359.
- Albanese, J. (2003). A metric method of sex determination using the hip bone and femur. *Journal* of Forensic Sciences 48 pp. 263-273.

- Albanese, J., Eklics, G., and Tuck, A. (2008). A metric method for sex determination using the proximal femur and fragmentary hip bone. *Journal of Forensic Sciences* 53(6) pp. 1283-1288.
- Allan, D. A. (1998). Structure and physiology of joints and their relationship to repetitive strain injuries. *Clinical Orthopaedics 352* pp. 32-40.
- Allen, M., Burr, B. (2008) Skeletal microdamage: less about biomechanics and more about remodeling. *Clinical Review of Bone and Mineral Metabolism* 6 pp. 24-30.
- Amores, A., Botella, M., and Aleman, I. (2014). Sexual dimorphism in the 7th cervical and 12th thoracic vertebrae from a Mediterranean population. *Journal of Forensic Sciences 59*(2) pp. 310-305.
- Anderson, J. E. (1963). The People of Fairty: An Osteological Analysis of an Iraquois Ossuary. *Contributions to Anthropology 1961-2*: National Museum of Canada.
- Anderson, J. J., and Felson, D. T. (1988). Factors associated with osteoarthritis of the knee in the first National Health and Nutrition Examination Survey (HANES1): Evidence for an association with overweight, race and physical demands of work. *American Journal of Epidemiology* 128 pp. 179-189.
- Anderson, R. J. (1883). Observations on the diameters of human vertebrae in different regions. *Journal of Anatomy and Physiology* 17, pp. 341-344.
- Anderson, S. (1990). The human skeletal remains from Staunch Meadow, Brandon, Suffolk. Ancient Monuments Laboratory Report (Vol. 99/90).
- Andrianakos, A., Kontelis, L., Karamitsos, D., Aslanidis, S, Georountzos, A., Kaziolas, G., and Dantis,
 P. (2006). Prevalence of symptomatic knee, hand and hip osteoarthritis in Greece: the
 ESORDIG study. *The Journal of Rheumatology 33*(12) pp. 2507-2513.
- Angel, J. L. (1966). Early skeletons from Tranquility, California. *Smithsonian Institute Contributions* to Anthropology 2 pp. 1-19.
- Angel, J. L., Kelley, J. O., Parrington, M., and Pinter, S. (1987). Life stresses of the free Black community as represented by the First African Baptist Church, Philadelphia, 1823-1841.
 American Journal of Physical Anthropology 74(2) pp. 213-229.
- Arcini, C. (1999). Health and disease in early Lund, osteopathological studies of 3,305 individuals buried in the first cemetery area of Lund, Lund: Lund University Publications. Available at:
http://lup.lub.lu.se/record/27698 (Accessed: 5th June 2011).

- Arden, N., and Nevitt, M. C. (2006). Osteoarthritis: epidemiology. *Best Practice and Research Clinical Rheumatology* 20 pp. 3-25.
- Armelagos, G. J. (2008). Biocultural anthropology at its origins. In J. Kelso (ed). *The Tao of Anthropology.* Gainesville: University Press of Florida pp. 269-282.
- Armelagos, G. J., Carlson, D. S., and van Gerven, D. P. (1982). The theoretical foundations and development of skeletal biology. In F. Spencer (ed). A History of American Physical Anthropology 1930-1980. New York: Academic Press pp. 305-328.
- Armelagos, G. J., and van Gerven, D. P. (2003). A century of skeletal biology and paleopathology: contrasts, contradictions and conflicts. *American Anthropologist 105*(1) pp. 54-64.
- Armstrong, R. R. (2014). When to use the Bonferroni correction. *Ophthalmic and Physiological Optics*. 34 pp. 502-508.
- Arsuaga, J. L., and Carretero, J. M. (1994). Multivariate analysis of the sexual dimorphism of the hip bone in a modern population and in early hominids. *American Journal of Physical Anthropology* 93 pp. 241-257.
- Asher, R. J., Lin, K. H., Kardjilov, N., and Hautier, L. 92011). Variability and constraint in the mammalian vertebral column. *Journal of Evolutionary Biology* 24 pp. 1080-1090.
- Askádi, G., and Nemeréski, J. (1970). *History of Human Life Span and Mortality*. Budapest: Akademiai Kiado.
- Atchley, W. R., Gaskins, C. T., and Anderson, D. (1976). Statistical properties of ratios: I. Empirical results. *Systematic Zoology* 25 pp. 137-148.
- Atkinson, S. J. (1985). Variations in Rates of Age-Related Change in the Skeletal Tissue in a Romano-British Population. PhD. Thesis. Durham University. Available at: <u>http://etheses.dur.ac.uk/7594</u> (Accessed 16th February 2016).
- Auerbach, B. M. (2011). Methods for estimating missing skeletal element osteometric dimensions employed in the revised Fully technique for estimating stature. *American Journal of Physical Anthropology* 145 pp. 67-80.
- Auerbach, B. M., and Raxter, M. H. (2008). Patterns of clavicular bilateral asymmetry in relation to the humerus: variation among humans. *Journal of Human Evolution* 54 pp. 663-674.

- Auerbach, B. M., and Ruff, C. B. (2004). Human body mass estimation: a comparison of "morphometric" and "mechanical" methods. *American Journal of Physical Anthropology* 125 pp. 331-342.
- Auerbach, B. M., and Ruff, C. B. (2006). Limb bone bilateral asymmetry: variability and commonality among modern humans. *Journal of Human Evolution* 50 pp. 203-218.
- Aufderhiede, A. C., and Rodriguez-Martin, C. (1998). *The Cambridge Encyclopedia of Human Pathology*. Cambridge: Cambridge University Press.
- Baker, P. T. (1984). The adaptive limits of human populations. Man 19, pp. 1-14.
- Barak, M. M., Lieberman, D. E., and Hublin, J. J. (2011). A Wolf in sheep's clothing: Trabecular bone adaptation in response to changes in joint loading orientation. *Bone* 49 pp. 1141-1151.
- Barnes, E. (2012). Atlas of Developmental Field Anomalies of the Human Skeleton: A Paleopathological Perspective. New Jersey: Wiley-Blackwell.
- Barnes, E. (2008) Congenital anomalies. In: Pinhasi, R., and Mays, S. (eds). *Advances in Human Paleopathology*. New York: Wiley-Liss.
- Barrier, I. L., and Abbe, E. N. (2008). Sex determination from the radius and ulna in a modern South African sample. *Forensic Science International* 179(1) pp. 85-89.
- Barrio, P. A., Trancho, G. J., and Sanchez, J. A. (2006). Metacarpal sex determination in a Spanish population. *Journal of Forensic Science* 51(5) pp. 990-995.
- Bašić, Ž., Anterić, I., Vilović, k., Petaros, A., Bosnar, A., and Madžar, T. (2013). Sex determination in skeletal remains from the medieval Eastern Adriatic coast – discriminant function analysis of humeri. *Croatian Medical Journal*. 54 pp. 272-8.
- Bass, W. M. (1995). *Human Osteology: A Laboratory and Field Manual*. Columbia: Missouri Archaeological Society Special Publication No. 2
- Bastir, M., Higuero, A., Rios, L., and Martinez, D. G. (2014). Three-dimensional analysis of sexual dimorphism in human thoracic vertebrae: implications for the respiratory system and spine morphology. *American Journal of Physical Anthropology* 155 pp. 513-521.
- Beall, C., and Steegman, A. T. (2000). Human adaptation to climate: Temperate, ultraviolet radiation and altitude. In Stinson S., Bogin, B., Huss-Ashmore, R. (eds). *Human Biology: An Evolutionary And Biocultural Perspective*. New York: Wiley-Liss.

- Bendall, S. (2014). To write a distick upon it: Busks and the language of courtship and sexual desire in sixteenth and seventeenth century England. *Gender and History*, 26(2), pp. 199-222.
- Bedford, M. E., Russell, K. F., Lovejoy, C. O., Meindl, R. S., Simpson, S. W., and Stuart-Macadam, P.
 L. (1993). Test of the Multifactorial Aging Method using skeletons with known age-atdeath from the Grant Collection. *American Journal of Physical Anthropology* 91 pp. 287-297.
- Bellamare, F., Jeanneret, A., and Couture, J. (2003). Sex differences in thoracic dimensions and configuration. *American Journal of Respiratory Critical Care Medicine* 168 pp. 305-312.
- Benoist, M. (2003). Natural history of the aging spine. *European Spine Journal* 12 (Suppl 2), pp. S86-89.
- Bergink, A. P., van Meurs, J. B., Loughlin, J., Arp, P. P., Fang, Y., Hofman, A., and Pol, H. A. P.
 (2003). Estrogen receptor α-gene halotype is associated with radiographic osteoarthritis of the knee in elderly men and women. *Arthritis and Rheumatism* 48 pp. 1913-1922.

Berkow, R. (ed). (1992). The Merck Manual. Rahway: N. J.: Merck Research Laboratories.

- Bertram, J. E. A., and Schwartz, S. M. (1991). The law of bone transformation: a case of crying Wolff? *Biological Reviews* 66 pp. 245-273.
- Bethard, J. D., and Seet, B. L. (2013). Sex determination of the second cervical vertebra: a test of Wescott's method on a modern American sample. *Journal of Forensic Science* 58(1), pp. 101-103.
- Bick, E. M. (1948). Source Book Of Orthopaedics. Baltimore: Williams and Wilkins.
- Bidmos, M. A., and Dayal, M. R. (2004). Further evidence to show population specificity of discriminant function equations for sex determination using the talus of South African Blacks. *Journal of Forensic Science* 49 pp. 1165-1170.

Binford, L. R. (1962). Archaeology as Anthropology. American Antiquity 28(2) pp. 217-225.

Bird, H. (1990). When the body takes the strain. *New Scientist 7* July pp. 49-52. Available from: <u>https://www.newscientist.com/article/mg12717244-400-when-the-body-takes-the-</u> <u>strain-working-may-be-bad-for-your-health-especially-if-it-involves-making-repetitive-</u> <u>movements-for-hours-on-end</u>. (Accessed 24th May 2012).

Bird, H. A., Tribe, C. R, and Bacon, P. A. (1978). Joint hypermobility leading to osteoarthritis and

chondrocalcinosis. Annals of the Rheumatic Disease 37 pp. 203-211.

- Black, B., Dooley, J., Pyper, A., and Reed, M. (1993). Multiple hereditary exostoses: An epidemiologic study of an isolated community in Manitoba. *Clinical Orthopaedics and Related Research* (287) pp. 212-217.
- Black, D. M., Cummings, S. R., Stone, K., Hudes, E., Palermo, L., and Steiger, P. (1991). A new approach to defining normal vertebral dimensions. *Journal of Bone and Mineral Research* 16 pp. 883-892.
- Blackburn, A., and Knüsel, C. (2006). Hand dominance and bilateral asymmetry of the epicondylar breadth of the humerus: A test in a living sample. *Current Anthropology* 47(2) pp. 377-382.
- Bliziotse, M., Sibonga, J., Turner, R., Orwell, E. (2006). Periosteal remodeling at the femoral neck in nonhuman primates. *Journal of Bone and Mineral Research* 21 pp. 1060-1067.
- Blom, B. E., Buikstra, J. E., Keng, L., Tomczak, P. D., Shoreman, E., and Stevens-Tuttle, D. (2005).
 Anemia and childhood mortality: latitudinal patterning along the coast of pre-Columbian
 Peru. American Journal of Physical Anthropology 127 pp. 152-169.
- Bogduk, N. (1997). *Clinical Anatomy Of The Lumbar Spine And Sacrum*. New York Churchill Livingstone.
- Bogin, B. (1995). Plasticity in the growth of Mayan refugee children living in the United States. In
 Mascie-Taylor, C. G. N., and Bogin, B. (eds). *Human Variability and Plasticity*. Cambridge:
 Cambridge University Press pp. 46-74

Bogin, B. (1999). Patterns of Human Growth. Cambridge: Cambridge University Press.

- Bolam, K. A., van Uffelen, J. G., and Taaffe, D. R. (2013) The effect of physical exercise on bone density in middle-aged and older men: a systematic review. *Osteoporosis International* 24 pp.2749-2762.
- Boldsen, J. L., Milner, G. R., Konigsberg, L. W., and Wood, J. W. (2002). Transition analysis: a new method for estimating age from skeletons, In Hoppa R.D., and Vaupel, J. W. (eds).
 Paleodemography. Cambridge: Cambridge University Press pp. 73-106.
- Bonivitch, A., Francis, L., Moravits, D., Thacker, B., Pintar, F., and Yogandan, N. (2006). Cervical spine geometry: female vertebrae cannot be scaled from male vertebrae. *Proceedings of the American Society of Biomechanics Annual Meeting* Sept 6-9 Blackburg VA: Available

from: http://www.asweb.org/conferences/2006/pdfs/93.pdf. (Accessed 15/08/18).

- Bonjour, J.P., Theintz, G., Buchs, B., Slosman, B., and Rizzoli, R. (1991). Critical years and stages of puberty for spinal and femoral bone mass accumulation during adolescence. *Journal of Clinical Endocrinology Metabolism* 73 pp. 555-563.
- Bonneau, N., Simonis, C., Seringe, R., and Tardieu, C. (2011). Study of femoral torsion during prenatal growth: Interpretations associated with the effects of intrauterine pressure.
 American Journal of Physical Anthropology 145 pp. 438-445.
 https://doi.org/10.1002/ajpa.21521. (accessed 24th June 2018).
- Bornstein, P. E., and Peterson, R. R. (1996). Numerical variation of the pre-sacral column in three population groups in North America. *American Journal of Physical Anthropology* 25 pp. 139-146.
- Bouquet-Appel, J. P., and Masset, C. (1982). Farewell to paleodemography. *Journal of Human Evolution* 11 pp. 321-333.
- Bourke, J. B. (1967). A Review of the Paleopathology of Arthritic Disease. In Brothwell, D., and Sandison, A. T. (eds). *Diseases in Antiquity*. Springfield II.: Charles C Thomas pp. 352-370.
- Boyle, J. J., Singer, K. P., and Milne, N. (1996). Morphological survey of the cervico-thoracic junctional region. *Spine* 21(5) pp. 544-548.
- Brace, C. L. (1973). Sexual Dimorphism in Human Evolution. In Brace, C. L., and Metress, J. (eds). *Man in Evolutionary Perspective*. New York: John Wiley and Sons pp. 238-254.
- Brace, C. L., & Ryan, A. S. (1980). Sexual dimorphism and human tooth sizes. *Journal of Human Evolution* 9 pp. 417-435.
- Brain, Lord. (1967). Cervical Spondylosis. London: William Heinemann Medical Books.
- Brauer, G. W. (1982). Size, sexual dimorphism and secular trend: Indicators of subclinical malnutrition. In Hall, R. L. (ed). Sexual Dimorphism In Homo Sapiens: A Question Of Size. New York: Praeger Publishers pp. 245-262.
- Brauer, G. (1988). Osteometrie. In Knussman, K. R. (ed). *Anthropologie*. Stuttgart: Gustav Fischer Gerlag pp. 160-231.
- Brickley, M., and Agarwal, S. C. (2003). Techniques for the investigation of age-related bone loss and osteoporosis in archaeological bone. In Agarwal, S. C. and Stout, S.D. (eds). *Bone Loss and Osteoporosis: An Anthropological Perspective*. New York: Kluwer Academic Press pp.

List of References

157-172.

- Brickley, M., and Ives, R. (2006). Skeletal manifestations of infantile scurvy. *American Journal of Physical Anthropology* 129 163-172.
- Brickley, M., and Ives, R. (2008). *The Bioarchaeology Of Metabolic Bone Disease*. San Diego: Elsevier.
- Brickley, M., and McKinley, J. (2004). Guidelines to the Standards for Recording Human Remains *IFA Paper no 7*. BABAO University of Southampton, IFA University of Reading. Available from: <u>https://www.archaeologists.net/sites/default/files/ifa_paper_7.pdf</u>. (Accessed 4th May 2010).
- Bridges, P. S. (1989). Changes in activities with the shift to agriculture in southeastern United States. *Current Archaeology.* 30: pp. 385-394.
- Bridges, P. S. (1991). Degenerative joint disease in hunter-gatherers and agriculturalists from the south-eastern United States. *American Journal of Physical Anthropology* 85 pp. 379-391.
- Bridges, P. S. (1992.) Prehistoric arthritis in the Americas. *Annual Review of Anthropology* 21 pp. 67-91.
- Bridges, P. S. (1993). Reply to Dr. Knüsel. *American Journal of Physical Anthropology* 91 pp. 526-527.
- Bridges, P. S. (1994). Vertebral arthritis and physical activity in the prehistoric southeastern United States. *American Journal of Physical Anthropology* 93 pp. 83-93.
- Bridges, P. S. (1996). Skeletal biology and behavior in ancient humans. *Evolutionary Anthropology* 4 pp. 112-120.
- Bridges, P. S., Blitz, J. H., and Solano, M. C. (2000). Changes in long bone diaphyseal strength with horticultural intensification in west-central Illinois. *American Journal of Physical Anthropology* 112(2) pp. 217-238.
- Brinson, J. G. S. (1963) Great Chesterford. In Hull M. R. (ed). "Roman Gazetteer" The Victoria History Of The Counties Of England:" A History Of Essex, Hull Vol. 3, pp. 72-88.
- Brocher, J. E. W., and Willert H. G. (1980). Die nicht-tuberkulöse Spondylitis. In: Brocher, J. E. W., and Willert, H. G. (eds). *Differentialdiagnose der Wirbelsälenkrankungen*. Thieme, Stuttgart.

- Bron, J. L., van Royen B. J., and Wuisman, P. I. (2007). The clinical significance of lumbosacral transitional anomalies. *Acta Orthopaedica Belgica* 73 pp. 687-95.
- Brooks, S., and Suchey, J. M. (1990). Skeletal age determination based on the os pubis: a comparison of the Ascádi-Nemereski and Suchey Brooks Method. *Human Evolution* 5 pp. 227-238.
- Brothwell, D. R. (1981). *Digging up Bones*. (3rd ed). London: British Museum (Natural History).
- Bruzek, J. (2002). A method for visual determination of sex using the hip bone. *American Journal* of *Physical Anthropology* 117 pp. 157-168.
- Bruzek, J., and Murail, P. (2006) Methodology and reliability of sex diagnosis from the skeleton. In
 Schmitt, A., Cuhna, E., and Pinheiro, J. (eds). *Forensic Anthropology And Medicine: Complementary Sciences From Recovery To Cause Of Death.* Humana Inc., New Jersey pp. 225-242.
- Buckwalter, J. A, and Lane, N. E. (1997). Athletics and osteoarthritis. *The American Journal of* Sports Medicine 25 pp. 873-881.
- Buckwalter, J. A., and Lappin, D. R. (2000). The disproportionate impact of chronic arthralgia and arthritis among women. *Clinical Orthopaedic Related Research* 372 pp. 159-168.
- Budinoff, L. C., and Tague, R. G. (1990). Anatomical and developmental bases for the ventral arc of the human pubis. *American Journal of Physical Anthropology* 82 pp. 73-79.
- Buikstra, J. (2006). Emerging Specialties. In Buikstra, J. and Beck, L. (eds). *Bioarchaeology: The Contextual Analysis Of Human Remains*. Amsterdam: Elsevier.
- Buikstra, J., and Beck, L. (2006). *Bioarchaeology: The Contextual Study Of Human Remains*. Amsterdam: Elsevier.
- Buikstra, J., and Uberlaker, D. (1994). Standards for data collection from human skeletal remains. Archaeological Survey, Research Series 44, Fayetteville, Arkansas.
- Buikstra, J. E. (1977). Biocultural dimensions of archaeological study. In Blakely, R. L. (ed).
 Biocultural Adaptation In Prehistoric America, Athens: Southern Anthropological Society
 Proceedings, No. 11, University of Georgia pp. 67-84.
- Buikstra, J. E. (1991). Out of the appendix and into the dirt: comments on thirteen years of
 bioarchaeological research. In Powell, M., Bridges P., and Mires, A. (eds). What Mean
 These Bones: Studies In Southeastern Bioarchaeology. Tuscaloosa, AL: The University of

Alabama Press pp. 172-189.

Buikstra, J. E., and Meilke, J. H. (1985). Demography, diet and health. In Gilbert, R.I., and Meilke, J.H. (eds). *The Analysis of Prehistoric Diets*. New York: Academic Press pp. 359-422.

Bullough, P. (2004). Orthopaedic pathology. Edinburgh: Mosby.

- Bullough, P., and Jaggannath, A. (1983). The morphology of the calcification front in articular cartilage. *Journal of Bone and Joint Surgery British Volume* 65 pp. 72-78.
- Bullough, P. G. (1981). The geometry of diarthrodial joints, its maintenance and the possible significance of age-related changes in geometry-to-load distribution and the development of osteoarthritis. *Clinical Orthopaedic Related Research* 156 pp. 61-66.
- Burris, B. G., and Harris, E. F. (1998). Identification of race and sex from palate dimensions. Journal of Forensic Science 43(5) pp. 959-963.
- Busija, L., Bridgett, L., Williams, S. R. M., Osborne, R. H., Buchbinder, R., March, L., and Fransen,
 M. (2010). Osteoarthritis. *Best Practice and Research Clinical Rheumatology* 24 pp. 757-768.
- Butler, B., Tramiflow, J. H., Andersson, G. B. J., McNeill, T. W., and Huckman, M. S. (1990). Discs degenerate before facets. *Spine* 15 pp. 111-113.
- Calce, S. E. (2012). A new method to estimate adult age-at-death using the acetabulum. *American Journal of Physical Anthropology* 148 pp. 11-23.
- Calgnago, J. M. (1981). On the applicability of sexing human skeletal material by discriminant function analysis. *Journal of Human Evolution* 10, pp. 189-198.
- Campanacho, V., Santos, A. L., and Carduso, H. F. V. (2012). Assessing the influence of occupational and physical activity on the rate of degenerative change of the pubic symphysis in Portuguese males from the 19th to 20th century. *American Journal of Physical Anthropology* 148(3) pp. 371-378.
- Carduso, F. Alves, and Henderson, C. Y. (2010). Enthesopathy formation in the humerus: data from known age at death and known occupation skeletal collections. *American Journal of Physical Anthropology* 141 pp. 550-560.
- Carlson, K. J., Grine, F. E., and Pearson, O. M. (2007). Robusticity and sexual dimorphism in the post-cranium of modern hunter-gatherers from Australia. *American Journal of Physical Anthropology* 134 pp. 9-23.

- Carter, D. R., van der Meulen, M. C., and Beaupre, G. S. (1996). Skeletal development: mechanical consequences of growth, aging and disease. In Marcus, R., Feldmen, D., and Kelsey, J. (eds). *Osteoporosis.* San Diego: Academic Press pp. 333-350.
- Carter, D. R., Wong, M., and Orr, T. E. (1991). Musculoskeletal ontogeny, phylogeny and functional adaptation. *Journal of Biomechanics* 24 (Suppl1), pp. 3-16.
- Case, D. T., and Ross, A. H. (2007). Sex determination from hand and foot bone lengths. *Journal of Forensic Science* 52, pp. 264-270.
- Cauley, J., Kwoh, C., and Egeland, G. (1993). Serum sex hormones and severity of osteoarthritis of the hand. *The Journal of Rheumatology* 20 pp. 1165-1170.
- Chaisson, C. E., Zhang, Y., McAlindon, T. E., Hannan, M. T., Aliabadi, P., Naimark, A., and Felson, D.
 T. (1997). Radiographic hand osteoarthritis: incidence, patterns and influence of preexisting disease in a population-based sample. *Journal of Rheumatology* 24(7) pp. 1337-1343.

Chamberlain, A. (2006). Demography in Archaeology. Cambridge: Cambridge University Press.

- Chapman, F. H. (1972). Vertebral osteophytosis in prehistoric populations. *American Journal of Physical Anthropology* 36, pp. 31-38.
- Charisi, D., Eliopoulos, C., Vanna, V., Koilias, C.G., and Manolis, S. K. (2011). Sexual dimorphism of the arm bones in a modern Greek population. *Journal of Forensic Science* 56 (1) pp. 10-18.
- Chen, C. S. (2008). Mechanotransduction a field pulling together. *Journal of Cell Science* 121 pp. 3285-3292.
- Chevereud, J. M. (1988). A comparison of genetic and phenotypic correlations. *Evolution* 42 pp. 958-68.
- Chopra, A. (2013). The COPCORD world of musculoskeletal pain and arthritis. *Rheumatology* (*Oxford*) 52 pp.1925-8.
- Clark, J. G. D. (1972). Starr Carr: A Case Study In Bioarchaeology: Modular Publications 10. London: Addison Wesley.
- Clarke, J. G. D. (1973). Bioarchaeology: Some extracts on a theme. *Current Archaeology* 14(4) pp. 464-470.

Clutton-Brock, T. H. (1985). Size, sexual dimorphism and polygeny in primates. in Jungers. W. L.

(ed). Size And Scaling In Primate Biology. New York: Plenum Press pp. 51-60.

- Cocilovo, J. L., Fuchs, M. L., O'Brien, T. G., and Varela, H. H. (2013). Sexual dimorphism in Prehispanic populations of the Cochabamba Valleys, Bolivia. *Advances in Anthropology*, 3(1) pp. 10-14.
- Coggon, D., Croft, P., and Kellingray, K. S. (2000). Occupational physical activity and osteoarthritis of the knee. *Arthritis and Rheumatism* 43(7) pp. 1443-1449.
- Cohen, M., and Crane-Kramer, G. M. (2007). *Ancient Health: Skeletal Indicators Of Agricultural And Economic Intensification*. Florida: University Press Florida.
- Cohen, S. P., and Raja, S. N. (2007). Pathogenesis, diagnosis and treatment of lumbar zygapophyseal (facet) joint pain. *Anesthesiology* 106(3) pp. 591-614.
- Cole, S. (2010). The Story Of Men's Underwear. New York: Parkstone International pp. 46-52.
- Cole, T. M. (1994). Size and shape of the femur and tibia in Northern Plains Indians. In Ousley, D.
 W., and Jantz, R. L. (eds). *Skeletal Biology In The Great Plains: Migration, Health And Subsistence*. Washington D C: Smithsonian Institution Press.
- Conaghan, P. G., Vanharanta, H., and Dieppe, P. (2005). Is progressive osteoarthritis an atheromatous vascular disease? *Annals of Rheumatic Diseases* 64 pp. 1539-1541.
- Cooley, H., Stankovich, J., and Jones, G. (2003). The association between hormonal and reproductive factors and hand osteoarthritis. *Maturitas* 45 pp. 257-265.
- Coolidge, F. L. (2013) Statistics, A Gentle Introduction. California: Sage Publishing.
- Cooper, C. (1998). Osteoarthritis and Related Disorders. In Klippel, J., and Dieppe, P. (eds). *Rheumatology*. London: Mosby pp. 1-8.
- Cooper, C., McAlindon, T., and Coggon, D. (1994). Occupational activity and osteoarthritis of the knee. *Annals of Rheumatic Diseases* 53 pp. 90-93.
- Cowin, S. C. (2001). Mechanics of materials. In Cowin, S.C. (ed). *Bone Biomechanics Handbook* (2nd ed.), Boca Raton: CRC Press pp. 6-24.
- Cowin, T. (2004). Tissue growth and remodeling. *Annual Review of Biomedical Engineering* 6 pp. 77-107.
- Cox, M., and Mays, S. (2000). *Human Osteology In Archaeology And Forensic Science*. London: Greenwich Media.

- Croft, P., Cooper, C., Wickham, C., and Coggon, D. (1992). Osteoarthritis of the hip and occupational activity. *Scandinavian Journal of Work, Environmental Health,* 18, pp. 59-63.
- Cross, M., Smith, E., Hoy, D., Nolte, S., Ackerman, I., and Fransen, M. (2014). The global burden of hip and knee osteoarthritis: estimates from the global burden of the disease 2010 study. *Annals of Rheumatic Disease* 73 pp. 1323-30
- Crubézy, E. (1988). Interactions Entre Facteurs Biocultures, Pathologie Et Catacteres Discrets: Exemple D'une Population Medieval. (PhD), Université de Montpellier, Canac (Aveyron).
- Cubuk, R., Kozakcioglu, M., Tasali, N., Atatley, A., and Celik, L. (2009). Lumbar disc and facet degeneration: correlation with age and facet orientation. *Balkan Medical Journal* 26(1) pp. 36-42.
- Cubuk, R., Tasali, N., Kozakcioglu, M., and Guney, S. (2009). Is bone mineral density measurement correlated with facet joint orientation? A dual energy absorptiometry study. *Prague Medical Report* 110(4) pp. 343-349.

Cunningham, J. (1886). Cunningham Memoir. No. 2 Royal Irish Academy pp. 1-116.

- Currey, J. D. (1984). The Mechanical Adaptation Of Bones. Princeton: Princeton University Press.
- Cyriax, E. F. (1920). On certain absolute and relative measurements of human vertebrae. *Journal* of Anatomy and Physiology 54 pp. 305-308.
- Dabbs, G., and Moore-Jansen, P. H. (2010). A method for estimating sex using metric analysis of the scapula. *Journal of Forensic Science* 55(1) pp. 63-71.
- Daly, R. M., Saxon, L., and Turner, C. H. (2004). The relationship between muscle size and bone geometry during growth and in response to exercise. *Bone* 34 pp. 281-287.
- Darwin, C. (1871). The Descent Of Man And Selection In Relation To Sex. New York: D Appleton.
- Davis, P. R. (1955). The thoraco- lumbar mortice joint. Journal of Anatomy 89 pp. 370-377.
- Davivong, V. (1963). The pelvic girdle of Australian Aboriginals, sex differences and sex determination. *American Journal of Physical Anthropology* 21 pp. 443-455.
- DeAngelis, G., and Chen, Y. (2013). Obesity among women may increase the risk of arthritis: observations from the Canadian Community Health survey, 2007-2008. *Rheumatology International* 33 pp. 2249-53.

Dempster, D. (1997) Exploiting and bypassing the bone remodeling cycle to optimize treatment of

osteoporosis. Journal of Bone and Mineral Research 12 pp. 1152-1154.

- Dempster, D. (2002). Bone remodeling . In Coe, F., Favus, M. (eds). *Disorders of Bone and Mineral Metabolism.* Baltimore MD: Lippincott, Williams and Wilkins. pp. 315-343
- Dempster, D. (2006). Anatomy and functions of the adult skeleton. In Favus, M, (ed). *Primer on the Metabolic Bone Diseases and Disorders of mineral Metabolism.* 6th Edn. Washington DC: American Society for Bone and Mineral Research. pp. 7-11.
- Dequeker, J., Verstraeten, A., Collee, G., and Gielen, J. (1980). Confrontation osteoarthrosisosteoporosis: A cross-sectional and longitudinal study. In Peyron, J. G. (ed). *Epidemiology* of Osteoarthritis. Manchester: Giegy Publications.
- DeSilva, J. M. (2009). Functional morphology of the ankle and likelihood of climbing in early hominins. *Proceedings of the National Academy of Science U. S. A.* 195 pp. 6567-6572.
- Dhall, V. (1984). Bilateral asymmetry in the area of the articular surface of the human ankle joint. Journal of the Anatomical Society of India 33(15) pp. 18.
- DiBennardo, R., and Bruce, M. F. (1983). Multiple discriminant function analysis of sex and race in the post cranial skeleton. *American Journal of Physical Anthropology* 61 pp. 305-314.
- DiBennardo, R., and Taylor, J. V. (1982). Classification and misclassification in sexing the black femur by discriminant function analysis. *American Journal of Physical Anthropology* 58 pp. 145-151.
- Dieppe, P. (1987). Osteoarthritis and related disorders. In Wetherall, D. J., and Ledingham, J.G. G. (eds). *Oxford Textbook Of Medicine*, New York: Oxford University Press pp. 160.76-160.84.
- DiGangi, E. A., and Moore, M. K. (2013). Introduction to skeletal biology. In DiGangi, E. A., and Moore, M. K. (eds). *Research Methods In Human Skeletal Biology*. Amsterdam: Elsevier.
- Dittrick, J., and Suchey, J. M. (1986). Sex determination of prehistoric central California skeletal remains using discriminant analysis of the femur and humerus. *American Journal of Physical Anthropology* 70(1) pp. 3-9.
- Dixit, S. G., Kakar, S., Agarwal, S., and Choudhry, R. (2007). Sexing of human hip bones of Indian origin by discriminant function analysis. *Journal of Forensic and Legal Medicine* 14 pp. 429-435.
- Doherty, M., Watt, I., and Dieppe, P. (1983). Influence of primary generalised osteoarthritis on development of secondary osteoarthritis. *Lancet* 322(8340) pp. 8-11.

- Donlon, D. (2000). The value of postcranial nonmetric variation in studies of global populations in modern Homo sapiens. *American Journal of Physical Anthropology* 113 (3) pp. 349-368.
- Dornan, J. L. (2002). Agency and archaeology: past, present and future directions. *Journal of Archaeological Method and Theory* 9 pp. 303-329.
- Drew, R. (2013). A review of the ischium-pubic index: accuracy, reliability and common errors. Human Biology 85(4) pp. 579-596.
- Dufour, D L. (2006). Biocultural approaches in human biology. *American Journal of Human Biology* 18(1) pp. 1-9.
- Duncan, H. (1979). Osteoarthritis. Henry Ford Hospital Medical Journal 27 pp. 6-9.
- Dutour, O. (2008). Archaeology of human pathogens: Paleopathological appraisal of
 Paleoepidemiology. In Raoult, D., and Drancourt, M. (eds). *Paleomicrobiology: Past Human Infections* Berlin: Springer-Verlag Berlin Heidelberg pp. 124-144.
- Dwight, T., McMurrich, J. P., Harman, C. A., Piersch, G. A., and White, J. W. (1907). Human anatomy including structure and development and practical considerations. In Piersol G.
 A. (ed). *Human Anatomy*. Philadelphia: J B Lippincott and Co.
- Ebraheim, N. A., Patil, V., Liu, J., Haman, S. P., and Yeasting, R. A. (2008). Morphometric analyses of the cervical superior facets and implications for facet dislocation. *International Orthopaedics* 32(1) pp. 97-101
- Einhorn, T. A. (1992). Bone strength: the bottom line. *Calcified Tissue International* 51 pp. 333-339.
- Eisenstein, S. M., and Parry, C. R. (1987). The lumbar facet arthrosis syndrome. *Journal of Bone* and Joint Surgery (British Volume) 69 pp. 3-70.
- Ekanem, T., Udongwu, A., and Singh, S. (2013). Radiographic determination of sex differences in ischiopubic index of a Nigerian population. *The Internet Journal of Biological Anthropology* 3(2) pp. 6-10.
- Ekenman, I., Eriksson, S. A., and Lindgren, J. U. (1995). Bone density in the medieval skeleton. *Calcified Tissue International* 56 pp. 355-358.
- Elliot-Wood, G., and Wood-Jones, F. (1910). Report on the human bones. In Reisner, G. A. (ed). *The Archaeological Survey Of Nubia: Report for 1907-8*. Cairo, National Printing Department.

- Elster, A. D. (1989). Berlotti's syndrome revisited: transitional vertebrae of the lumbar spine. *Spine* 14 pp. 1373-77.
- Englund, M., and Lohmander, L. S. (2004). Risk factors for symptomatic knee osteoarthritis fifteen to twenty-two years after meniscectomy. *Arthritis and Rheumatism* 50 pp. 2811-2819.
- Eriksen, M. F. (1976). Some aspects of aging in the lumbar spine. *American Journal of Physical Anthropology* 45 pp. 575-580.
- Erikson, M. F. (1980). Patterns of microscopic bone remodeling in three Aboriginal American populations. In Brownman, D. L. (ed). *E. A. T. Early Native Americans: Prehistoric Demography.* The Hague: De Gruyter Mouton.
- Eubanks, J. D., Lee, M. J., Cassinelli, E., and Ahn, N. U. (2007). Does lumbar facet arthrosis precede disc degeneration? A postmortem study. *Clinical Orthopaedic and Related Research 464* pp. 184-189
- Eveleth, P. B., and Tanner, J. M. (1990). *Worldwide Variation In Human Growth*. Cambridge: Cambridge University Press.
- Evison, V. (1994). An Anglo-Saxon cemetery at Great Chesterford, Essex. *Council for British* Archaeology Research Report 91.

Falotico, R., and Quatto, P. (2014). "Fleiss' kappa statistic without paradoxes." *Quality and Quantity* 49(3) pp. 463-470.

- Fell, C. (1984). Women in Anglo-Saxon England. British Museum Publications: London.
- Felson, D., Zhang, Y., and Hannan, M. T. (1995). The incidence and natural history of knee osteoarthritis in the elderly. *Arthritis and Rheumatism* 38 pp. 1500-1505.
- Felson, D. T. (2004). Risk factors for osteoarthritis: understanding joint vulnerability. *Clinical Orthopaedics And Related Research* 427S pp. S16-S21.
- Felson, D. T., and Chaisson, C. E. (1997). Understanding the relationship between body weight and osteoarthritis. *Balliere's Clinical Rheumatology* 11(4) pp. 671-681.
- Felson, D. T., Hannan, M. T., and Naimark, A. (1991). Occupational physical demands, knee bending and knee osteoarthritis: results from the Framingham study. *The Journal of Rheumatology* 18 pp. 1587-1592.
- Felson, D. T., Lawrence, R. C., and Dieppe, P. A. (2000). Osteoarthritis: new insights. Part 1: The

disease and its risk factors. Annals of Internal Medicine 133 pp. 635-646.

- Felson, D. T., and Zhang, Y. (1998). An update on the epidemiology of knee and hip osteoarthritis with a view to prevention. *Arthritis and Rheumatism* 41 pp. 1343-1355.
- Felson, D. T., Zhang, Y., and Anthony, J. M. (1992). Weight loss reduces the risk of symptomatic knee osteoarthritis in women. *Annals of Internal Medicine* 116 pp. 535-539.
- Fergusson, C. M. (1987). The aetiology of osteoarthritis. *Postgraduate Medical Journal* 63 pp. 439-445.
- Field, A. (2013). Discovering Statistics using SPSS (4th ed). London: Sage.
- Fisher, D. J. V. (1995). The Anglo-Saxon Age c 400-1042. Harlow: Longman Group Ltd.
- Fisher, N. M., and Prendergast, D. R. (1997). Reduced muscle function in patients with osteoarthritis. *Scandinavian Journal of Rehabilitation Medicine 29* pp. 213-231.
- Forwood, M. R., and Turner, C. H. (2005). Skeletal adaptations to mechanical usage: results from tibial loading studies in rats. *Bone* 17 pp. 197S-205S.
- France, D. L. (1988). Observational and metric analysis of sex in the skeleton. In Reichs K. J. (ed). Forensic Osteology: Advances In The Identification Of Human Remains. Springfield IL: Charles C Thomas.
- Francis, C. C. (1955a). Dimensions of the cervical vertebrae. *Anatomical Record*, 122(4) pp. 603-609.
- Francis, C. C. (1955b). Variations in the articular facets of the cervical vertebrae. *Anatomical Record* 122(4) pp. 589-602.
- Frayer, D. W. (1980). Sexual dimorphism and cultural evolution in the late Pleistocene and Holocene of Europe. *Journal of Human Evolution* 9 pp. 399-415.
- Frayer, D. W., and Wolpoff, M. H. (1985). Sexual Dimorphism. *Annual Review of Anthropology* 14 pp. 429-473.
- Frisancho, A. R. (1977). Developmental adaptation to high altitude hypoxia. *International Journal of Biometeorology* 21 pp. 135-146.

Frost, H. M. (1979). A chondral theory of modeling. Calcified Tissue International 28 pp. 181-200.

Frost, H. M. (1987). Bone "mass" and the "mechanostat": a proposal. Anatomical Record 219 pp.

1-9.

- Frost, H. M. (1990a). Skeletal structural adaptations to mechanical usage (SATMU): 1. Redefining Wolff's Law: the bone modeling problems. *Anatomical Record* 4 pp. 403-413.
- Frost, H. M. (1990b). Skeletal structural adaptations to mechanical usage (SATMU): 2. Redefining Wolff's law: the remodeling problem. *Anatomical Record* 4 pp. 414-422.
- Frost, H. M. (1994). Perspectives: a vital biomechanical model of synovial joint design. *Anatomical Record* 240 pp. 1-18.
- Frost, H. M. (1999). Joint anatomy, design and arthroses: insights of the Utah paradigm. *Anatomical Record* 255 pp. 162-174.
- Frost, H. M. (2003). Bone's mechanostat: a 2003 update. Anatomical Record 275A pp. 1081-1101.
- Frutos, L. R. (2005). Determination of sex from the clavicle and scapula in a Guatemalan contemporary rural indigenous population. *American Journal of Forensic Medicine and Pathology* 23 pp. 284-288.

Frutos L. R. (2005). Metric determination of sex from the humerus in a Guatemalan forensic sample. *Forensic Science International* 147 pp. 153-7.

- Fujiwara, A., Kazuya, T., Howard, S., Lond, T. H., Yoshida, H., Kurihashi, A., and Saotome, K. (2001).
 Orientation and osteoarthritis of the lumbar facet joints. *Clinical Orthopaedics and Related Research* 385 pp. 8-94.
- Fully, G. (1956). Determination de la stature au moyen du squelette. *Annals de Medecine Legale,*40 pp. 145-154.
- Gaillard, J. (1960). Détermination sexuelle d'un os coxal fragmentaire. *Bulletins et Mémoires de la Société d'Anthropologie de Paris, XIe série* 2 pp. 255-267.
- Gama, I., Navega, D., and Cunha, E. (2015). Sex estimation of the second cervical vertebra: a morphometric analysis in a documented Portuguese skeletal sample. *International Journal of Legal Medicine* 129 pp. 365-372.
- Gapert, R., Black, S., and Last, J. (2009). Sex determination from the foramen magnum:
 discriminant function analysis in an eighteenth and nineteenth century British sample.
 International Journal of Legal Medicine 123(1) pp. 25-33.
- Garvin, H. M., Sholts, S. B., and Mosca, L. A. (2014). Sexual dimorphism in human cranial trait

scores: Effects of population, age and body size. *American Journal of Physical Anthropology* 154 pp. 259-269.

- van Gerven, D. P., and Oakland, G. B. (1973). Univariate and multivariate statistical models in the analysis of human sexual dimorphism. *Statistician* 22 pp. 256-268.
- Gibson, R. (2015). Effects of long term corseting on the female skeleton: A preliminary
 morphological examination. *The Canadian Student Journal of Anthropology* 23(2) pp. 45-60.
- Giles, E. (1970). Discriminant Function Sexing of the Human Skeleton. In Stewart , T. D. (ed).
 Personal Identification In Mass Disasters. Washington DC: National Museum of Natural
 History, Smithsonian Institution pp. 99-107.
- Gilsanz, V., Boechat, M. I., Roe, T. F., Loro, M. L., Sayre, J. W., and Goodman, W. G. (1994). Gender differences in vertebral body size in children and adolescents. *Radiology* 190 pp. 673-677.
- Glynn, J. H., Sutherland, I. D., Walker, G. F., and Young, A. C. (1966). Low incidence of osteoarthritis in hip and knee after anterior poliomyelitis. *British Medical Journal ii* pp. 739-742.
- Goldberg, C., and Dowling, F. E. (1990). Handedness and scoliosis convexity: a reappraisal. *Spine* 15 pp. 61-64.
- Golightly, Y. M., Allen, K. D., Helmick, C. G., Renner, J. B., and Jordan, J. M. (2008). Symptoms of the knee and hip in individuals with and without limb length inequality. *Osteoarthritis Cartilage* 17(5) pp. 595-600.
- Gonzales-Reimers, E. J., Velasco-Vasquez, M., Arney-de-la-Rosa, F., Santolaria-Fernandez, M. A.,
 Rodriguez, G., and Machado-Calvo, M. (2002). Double-energy X-ray absorptiometry in the
 diagnosis of osteopenia in ancient skeletal remains. *American Journal Physical Anthropology* 118(2) pp. 134-145.
- Goodfellow, J., and Mitsou, A. (1977). Joint surface incongruity and its maintenance. *Journal of Bone and Joint Surgery (British Volume)* 59 pp. 446-451.
- Goodman, A. H., and Leatherman, T. L. (1998). *Building A New Biocultural Synthesis: Political-Economic Perspectives On Human Biology*. Ann Arbor: University of Michigan Press.
- Goodman, A. H., and Martin, D. L. (2002). Reconstructing health profiles from skeletal remains. In: Steckel, R. H., and Rose, J. C. (eds). *The Backbone of History: Health and Nutrition in the*

Western Hemisphere. Cambridge: Cambridge University Press pp. 11-60.

- Goodship, A. E., and Cunningham, J. L. (2001). Pathophysiology of functional adaptation of bone in remodeling and repair in vitro. In Cowin, S.C.(ed). *Bone Mechanics Handbook* (2nd Edition). Boca Raton: CRC Press pp. 26-31.
- Gosling, J.A., Harris, P. F., Humpherson, J. R., Whitmore, I., and Willan, P. L. (1996). *Human Anatomy Colour Atlas And Text*. London: Mosby-Wolfe.
- Grauer, A. L., and Stuart-Macadam, P. (1998). *Sex And Gender In Paleopathological Perspective*. New York: Cambridge University Press.
- Gray, J. P., and Wolfe, L. P. (2005). Height and sexual dimorphism of stature among human societies. *American Journal of Physical Anthropology* 53(3) pp. 441-456.
- Grave, B., Brown, T., and Townsend, G. (1999). Comparison of cervico-vertebral dimensions in Australian Aborigines and Caucasians. *European Journal of Orthodontics* 21 pp. 127-35
- Greenberg, D., Stoker, A., Kane, S., Cockrell, M., and Cook, J. L. (2006). Biochemical effects of two different hyaluronic acid products in a co-culture model of osteoarthritis. *Osteoarthritis Cartilage* 14 pp. 814-822.
- Greenland, S., Senn, S. J., Rothman, K. J., Carlin, J.B., Poole, C., Goodman, S. N., & Altman, D. G.
 (2016). Statistical tests, P values, confidence intervals and power, a guide to
 misinterpretations. *European Journal of Epidemiology*. 31 pp. 337-350.
- Grewar, D. (1965). Infantile scurvy. *Clinical Pediatrics* 4 pp. 82-89.
- Griffin, T. M., and Guilak, F. (2005). The role of mechanical loading in the onset and progression of osteoarthritis. *Exercise and Sports Science Review* 33(4) pp. 195-200.

Grimes, W. F. G. (1968). The Excavation Of Roman And Medieval London. New York: Praeger.

- Grine, F. E., Jungers, W. L., Tobias, P. V., and Pearson, O. M. (1995). Fossil Homo femur from Berg Aukas, northern Namibia. *American Journal of Physical Anthropology* 26 pp. 67-78.
- Grotle, M., Hagen, K. B., Natvig, B., Dahl, F. A., and Kvien, T. K. (2008). Obesity and osteoarthritis on knee, hip and/or hand: an epidemiological study in the general population with 10 years follow up. *BMC Musculoskeletal Disorders* 9 pp. 132-136.
- Gualdi-Russo, E. (2007). Sex determination from the talus and calcaneus measurements. *Forensic Science International* 171(2-3) pp. 151-161.

Guebert, G. M., Yochum, T. R., and Rowe, L. J. Congenital anomalies and normal skeleton variants. In: *Essentials of Skeletal Radiology*. Baltimore: Williams and Wilkins pp. 197-306.

Gunn, C. (1992). Bones and Joints (2nd ed.). London: Churchill Livingstone.

- Gustaffson, A., Werdelin, L., Tullberg, B., and Lindefors, P. (2007). Stature and sexual dimorphism in Sweden, from the 10th to the end of the 20th century. *American Journal of Human Biology* 19 pp. 861-870.
- Hadler, N. M., Gillings, D. B., Imbus, H. R., Levitin, P. M., MaKuc, D., and Utsinger, P. D. (1978).
 Hand structure and function in an industrial setting: influence of three patterns of stereotyped, repetitive usage. *Arthritis and Rheumatism* 21 pp. 210-220.
- Hagen, A. (2010). A Handbook of Anglo-Saxon Food: Processing and Consumption. Ely: Anglo-Saxon books
- Hall, B. K. (2015) *Bones And Cartilage: Developmental And Evolutionary Skeletal Biology,* (2nd Ed). London, UK: Elsevier/AP.
- Hamrick, M. W. (1999). A chondral modeling theory revisited. *Journal of Theoretical Biology* 201 pp. 201-208.
- Hanihara, K. (1952). Age changes in the male Japanese pubic bone. *Journal of the Anthropological Society of Nippon* 62 pp. 245-260.

Hanna, R. E., and Washburn, S. L. (1953). Determination of sex of a skeleton as illustrated by a study of Eskimo pelvis. *Human Biology* 25 pp. 21-27.

- Hart, D. J., Cronin, C., and Daniels, M. (2002). The relationship of bone density and fracture to incident and progressive radiographic osteoarthritis of the knee. *Arthritis and Rheumatism* 46 pp. 92-99.
- Hashimoto, I., and Tak, Y. K. (1977). The true sagittal diameter of the cervical spinal canal and its diagnostic significance in cervical myelopathy. *Journal of Neurosurgery* 47 pp. 912-916.

 Haugen, G. M. (1994). The Cervical Spine: Race And Sex Differences. (MA Thesis), University of Tennessee Available from: <u>https://trace.tennessee.edu/cgi/viewcontent.cgi?referer=https://www.google.com/&htt</u> <u>psredir=1&article=5645&context=utk_gradthes</u> (Accessed 26th April 2010).

Haeusler, M., Frater, N., and Bonneau, N. (2014). The transition from thoracic to lumbar facet joint orientation at T11: functional implications of a more cranially positioned transitional vertebra in early hominids. *Yearbook of the American Journal of Physical Anthropology* pp. S58-133

- Havelkova, M., Vilotte, S., Veleminsky, P., Polacek, L., and Dobiskova, M. (2011). Enthesopathies and activity patterns in the early Medieval Great Moravian population: evidence of division of labour. *International Journal of Osteoarchaeology* 21(4) pp. 487-504.
- Hawkey, D. (1998). Disability, compassion and the skeletal record: using musculoskeletal stress markers (MSM) to construct an osteobiography from early New Mexico. *International Journal of Osteoarchaeology* 8 pp. 326-340.
- Hawkey, D. (1988). Use Of Upper Extremity Enthesopathies To Indicate Habitual Activity Patterns. Arizona: Arizona State University.
- Hay, O., Dar, G., Abbas, J., Stein, D., May, H., Marashawi, Y., Peled, N., and Hershkovitz, I. (2015).
 The lumbar lordosis in males and females, revisited. *PLoS One*, 10(8), e0133685,
 doi:10,1371/journal.pone.0133685
- Hayashi, K., and Yabuki, T. (1985). Origin of the uncus and of Luschka's joint in the cervical spine. Journal of Bone and Joint Surgery 67-A (5) pp. 788-791.
- Hernandez, C. J., Beaupré, G. S., and Carter, D. R. (2000). A model of mechanobiologic and metabolic influences on bone adaptation. *Journal of Rehabilitation Research and Development* 37 (2) pp. 235-244.
- Herrero-Beaumont, G., Roman-Blas, J. A., Castaneda, S., and Jimenez, S. A. (2009). Primary osteoarthritis no longer primary: three subsets with distinct etiology, clinical and therapeutic characteristics. *Seminars Arthritis Rheumatism* 39 pp. 71-80.
- Hinck, V. C., Hopkins, C. E., and Savara, B. S. (1962). The size of the atlantal spinal canal: a sex difference. *Human Biology* 34 pp. 197-205.
- Hines, J. (2003). Society, community and identity. In *After Rome*, Charles-Edwards, T. (ed). Oxford University Press: Oxford pp. 61-102.
- Hirst, S.M. (1985). An Anglo-Saxon inhumation cemetery at Sewerby, East Yorkshire. *York University Archaeological Publications* (Vol. 4). York.
- Holdsworth, F. W. (1970). Review Article: Fracture, dislocations and fracture dislocations of the spine. *Journal of Bone and Joint Surgery* 52 pp. 1534-1551.

Holliday, T. W., and Ruff, C. B. (1997). Ecographic patterning and stature prediction in fossil

hominids: comment of Feldesman and Fountain. *American Journal of Physical Anthropology* 103 pp. 137-140.

- Holman, D. J., and Bennett, K. A. (1991). Determination of sex from arm measurements. *American Journal of Physical Anthropology* 84 pp. 421-426.
- Hooten, E. A. (1930). The Indians of Pecos Pablos. New Haven: Yale University Press.
- Hoppa, R. D., and Vaupel, J. W. (2002). *Paleodemography: Age Distributions From Skeletal Samples*. Cambridge: Cambridge University Press.
- Hora, M., and Sládek, V. (2018) Population specificity of sex estimation from vertebrae. *Forensic Science International.* 291 pp. 291.e1-297.e12.
- Horwitz, T., and Smith, R. M. (1940). An anatomical, pathological and roentgenological study of the intervertebral joints of the lumbar spine and of the sacroiliac joints. *American Journal* of Roentgenology Radium Therapy 43 pp. 173-186.
- Hou, W. B., Cheng, L. K. L., Tian, S. Y., Lu, Y. Q., Han, Y. Y., Lai, Y., and Li, Y. Q. (2012). Metric method for sex determination based on the 12th thoracic vertebra in contemporary North-Easterners in China. *Journal of Forensic and Legal Medicine* 19 pp. 137-143.
- Hough, A. J. (1993). Pathology of osteoarthritis. In McCarty, D. J. and Coopman, W. J. (eds). Arthritis And Allied Conditions. Philadelphia: Lea and Febiger 12th Ed. pp. 699-721.
- Hrdlička, A. (1920). Anthropometry. Philadelphia: The Wistar Institute of Anatomy and Biology.
- Hrdlička, A. (1934) Shapes of the femur: Additional data. Anthropologie 12 pp. 27-40.
- Hukuda, S., and Kojima, Y. (2002). Sex discrepancy in the canal/body ratio of the cervical spine implicating the prevalence of cervical myelopathy. *Spine* 27(3) pp. 250-253.
- Humphreys, S. C., Hodges, S. D., Patwardhan, A., Eck, J. C., Covington, L. A., & Sartori, M. (1998).
 The natural history of the cervical foramen in symptomatic and asymptomatic individuals aged 20-60 years as measured by magnetic resonance imaging: A descriptive approach.
 Spine 23(20) pp. 2180-2184.
- Igbigbi, P. S., and Msamati, B. C. (2000). Ischiopubic index in black Malawians. *East African Medical Journal* 77(9) pp. 514-516.
- Ingvarsson, T., Steffansson, S. E., and Hallgrimsdottir, I. B. (2000). The inheritance of hip osteoarthritis in Iceland. *Arthritis and Rheumatism* 43 pp. 2785-2792.

- Introna, F., DiVella, G., and Campobasso, C. P. (1998). Sex determination by discriminant analysis of patella measurements. *Forensic Science International* 95 pp. 39-45.
- Ipiña, S. L., and Durand, A. I. (2010). Assessment of sexual dimorphism: a critical discussion in a (paleo)- anthropological context. *Human Biology* 82(2) pp. 199-220.
- Isçan, M. Y. (1988) Rise of forensic anthropology. *Yearbook of Physical Anthropology*. 31 pp. 203-230.
- Isçan, M. Y. (2005). Forensic anthropology of sex and body size. *Forensic Science International* 147 pp. 107-112.
- Isçan, M. Y., and Helmer, R. P. (1993). Morphologic and osteometric assessment of age, sex and race from the skull. In Isçan, M. Y., and Helmer, R.P. (eds). *Forensic Analysis Of The Skull* New York: John Wiley and Sons Inc., pp. 81-83.
- Isçan, M. Y., and Loth, S. (1986). Estimation of age and determination of sex from the sternal rib. In Reichs, K. J. (ed). *Forensic Osteology: Advances In The Identification Of Human Remains* Springfield IL: Thomas, pp. 68-89.
- Isçan, M. Y., and Steyn, M. (2013). *The Human Skeleton In Forensic Anthropology*. Springfield IL: Charles C. Thomas.
- Isçan, M. Y., Yasar, S. R., King, C. A., Shihai, D., and Yoshino, M. (1998). Sexual dimorphism in the humerus: A comparative study of Chinese, Japanese and Thais. *Forensic Sciences International* 98 pp. 17-21.
- Jackson, D. A., Harvey, H. H., and Somers, K. M. (1990). Ratios in aquatic sciences: statistical shortcomings with mean depth and the morphoedaphic index. *Canadian Journal of Fish and Aquatic Science* 47 pp. 1788-1795.
- Jankauskas, R. (1994). Variability of vertebral column measurements in a Lithuanian population. International Journal of Anthropology 19 pp. 137-151.
- Jantz, R. L., Hunt, D. R., and Meadows, L. (1995). The estimation and mis-measure of the tibia: implications for stature estimation. *Journal of Forensic Sciences* 40(5) pp. 758-761.
- Jantz, R. L., and Ousley, S. D. (2005). FORDISC 3.0 Personal Computer Forensic Discriminate Functions. Knoxville: Forensic Anthropology Centre: University of Tennessee.
- Jaskulska, E. (2009). Skeletal bilateral asymmetry in a medieval population from Deir an Naqlun (Nekroni), Egypt. *Bioarchaeology of the Near East* 3 pp. 17-26.

- Jentsch, T., Geiger, J., Zimmermann, S. M., Slankamenac, K., Nguyen-Kim, T. D. L., and Werner, C.
 M. L. (2013). Lumbar joint arthritis is associated with a more coronal orientation of the facet joints of the upper lumbar spine. *Radiology Research and Practice* 90 pp. 246-249.
- Johnson, L. C. (1962). Joint remodeling as a basis for osteoarthritis. *Journal of the American Veterinary Medical Association* 141 pp. 1237-1241.
- Johnson, V. L., and Hunter, D. J. (2014). The epidemiology of osteoarthritis. *Best Practice in Clinical Rheumatology* 28 pp. 5-15.
- Jones, G., Cooley, H., and Stankovich, J. (2002). A cross-sectional study of the association between sex, smoking and other lifestyle factors and osteoarthritis of the hand. *The Journal of Rheumatology*, 29(8) pp. 1719-1724.
- Jones, H. H., Priest, J. D., Hayes, W. C., Tichenor, C. C., and Nagel, D. A. (1977). Humeral hypertrophy in response to exercise. *Journal of Bone and Joint Surgery* 59A pp. 204-208.
- Jones, W. T. (ed). (1980). *Early Saxon Cemeteries in Essex*: Council for British Archaeology Research Report No 34.
- Jordan, J. M., Kingston, R. S., Lane, N. E., Nevitt, M. C., Zhang, Y., and Sowers, M. F. (2000). *Systemic risk factors for osteoarthritis* Paper presented at the Osteoarthritis: New Insights Part 1: The disease and its risk factors. *Annals of Internal Medicine* 133 pp. 637-639.
- Judd, M. A., and Redfern, R. (2012). Trauma. In Grauer, A. L. (ed). *A Companion To Paleopathology* Chichester: Wiley Blackwell pp. 359-379.
- Judd, M. A., and Roberts, C. A. (1999). Fracture trauma in a medieval British farming village. American Journal of Physical Anthropology 109(2) pp. 229-243.
- Jungers, W. L., and Minns, R. J. (1979). Computed tomography and biomechanical analysis of fossil long bones. *American Journal of Physical Anthropology* 50(2) pp. 285-290.
- Jurmain, R. (1977). Stress and the etiology of osteoarthritis. *American Journal of Physical Anthropology* 46 pp. 353-366.
- Jurmain, R. (1990). Paleoepidemiology of a central California prehistoric population from CA-ALA-329:11. American Journal of Physical Anthropology 83 pp. 83-94.

Jurmain, R. (2012). Essentials Of Physical Anthropology. New York: Wadsworth Publishing Co Inc.

Jurmain, R., Carduso, F. Alves., Henderson, C., and Vilotte, S. (2012). Bioarchaeology's Holy Grail:

The reconstruction of activity. In Grauer, A. L. (ed). *A Companion To Paleopathology*. New York: Wiley-Blackwell.

- Jurmain, R., and Kilgore, K. (1995). Skeletal evidence of osteoarthritis: a paleopathological perspective. *Annals of the Rheumatic Diseases* 54 pp. 443-450.
- Jurmain, R. D. (1999). *Stories From The Skeleton. Behavioural Reconstruction In Human Osteology*. London: Taylor and Francis.
- Junno, J. A., Niskanen, M., Nieminen, M. T., Maijanen, H., Niinikäki, J., Bloigu, R., and Tuukkanen,
 J. (2009). Temporal trends in vertebral size and shape from Medieval to Modern-day.
 Available from:
 https://journals.plos.org/plosone/article?id=10.1371/journal.pone.0004836 (Accessed
 15th March 2018).
- Kacar, E., Unlu, E., Beker-Acay, M., Balcik, C., Gultekin, M. A., Kocak, U., Eroglu, S., and Yucel, A.
 (2017). Age estimation by assessing the vertebral osteophytes with the aid of 3D CT imaging. *Australian Journal of Forensic Science* 49 (4) pp. 449-458.
- Kajanoja, P. (1966). Sex diagnosis of Finnish crania by discriminant function analysis. *American Journal of Physical Anthropology* 24 pp. 29-34.
- Kalichman, L., and Hunter, D. J. (2007). Lumbar facet joint osteoarthritis: a review. *Seminar Arthritis Rheumatism* 37(2) pp. 69-80.
- Kalichman, L., Kim, D. H., Li, L., Guermazi, A., Berkin, V., and Hunter, D. J. (2009). Spondylolysis and spondylolisthesis: prevalence and association with low back pain in the adult community-based population. *Spine* 34(2) pp. 199-205.
- Kannus, P., Haapsalo, H., Sankelo, M., Sievanen, H., Pasanen, M., Heinonen, A., and Viouri, I.
 (1995). Effects of starting age of physical activity on bone mass in the dominant arm of tennis and squash players. *Annals of Internal Medicine* 123 pp. 27-31.
- Kaplan, K. M., Spivak, J. M., and Bendo, J. A. (2005). Embryology of the spine and associated congenital abnormalities. *The Spine Journal* 5 pp. 564-576.
- Kaplowitz, P. (2006). Pubertal development of girls: secular trends. *Current Opinion in Obstetrics* and Gynaecology 18(5) pp. 487-491.
- Katz, P. R., Reynolds, H. M., Foust, D. R., and Baum, J. K. (1973). Mid-sagittal dimensions of cervical vertebral bodies. *American Journal of Physical Anthropology* 43 pp. 319-326.

- Katzenberg, M., and Saunders, S. (2000). *Biological Anthropology Of The Human Skeleton*. New York Wiley.
- Kellgren, J. H. (1961). Osteoarthrosis in patients and populations. *British Medical Journal* 2(5234) pp. 1-6.
- Kellgren, J. H., and Moore, R. (1952). Generalised osteoarthritis and Heberden's nodes. *British Medical Journal* 1(4751) pp. 181-187.
- Kemkes-Gotthaler, A. (2005). Sex determination by discriminant analysis: an evaluation of the reliability of patella measurements. *Forensic Science International* 147(2-3) pp. 129-133.
- Kennedy, K. (1989). Skeletal markers of occupational stress. In Isçan, M., and Kennedy, K. (eds). *Reconstruction Of Life From The Skeleton*. New York: Wiley-Liss pp. 129-160.
- Kerrigan, D. C., Todd, M. K., Riley, P. O. (1998). Knee osteoarthritis and high-heeled shoes. Lancet, 351 pp. 1399-1401.
- Kilgore, L. (1984). *Degenerative Joint Disease In A Medieval Nubian Population*. (PhD), University of Columbo, Boulder Co.
- Kim, J., and Beang, H. (2016). Three common misuses of P values. *Dental Hypotheses* 7(3) pp. 73-80.
- Kim, K. H., Park, J. Y., Kuh, S. U., Chin, D. K., Kim, K. S., and Cho, Y. E. (2013). Changes in spinal canal diameter and vertebral body height with age. *Yonsei Medical Journal* 54(6) pp. 1498-1504.
- Kimmerle, E. H., Ross, A., and Slice, D. (2008). Sexual dimorphism in America: geometric morphometric analysis of the cranio-facial region. *Journal of Forensic Sciences* 53(1) pp. 54-57.
- Kirkaldy-Willis, W. H. (1985). Presidential symposium on instability of the lumbar spine. *Spine* 10 pp. 254-256.
- Klales, A. R., Ousley, S.D., and Vollner, J. J. (2012). A revised method of sexing the human innominate using Phenice's nonmetric traits and statistical methods. *American Journal of Physical Anthropology* 149 pp. 104-14.
- Klaus, H. D. (2014) Frontiers in the bioarchaeology of stress and disease: Cross-disciplinary perspectives from pathophysiology, human biology, and epidemiology. *American Journal* of Physical Anthropology 155 pp. 294-308. <u>https://doi.org/10.1002/ajpa.22574</u>.

- Klaus, H. D., Larsen, C. S., and Tam, M. E. (2009). Economic intensification and degenerative joint disease: Life and labour on the post-contact north coast of Peru. *American Journal of Physical Anthropology* 139 pp. 204-221.
- Klein-Nulend, J., Bakker, A. D., Bacabac, R. G., Vatsa, A., and Weinbaum, S. (2013).Mechanosensation and transduction in osteocytes. *Bone* 54 pp. 182-190.
- Knüsel, C. (2000). Bone adaptation and its relationship to physical activity in the past. In Cox, M and Mays, S. (eds). *Human Osteology In Archaeology And Forensic Science* London:
 Greenwich Medical Media pp. 381-401.
- Knüsel, C. J., Goggel, S., and Lucy, D. (1997). Comparative degenerative joint disease of the vertebral column in the Medieval monastic cemetery of the Gilbertine Priory of St Andrew, Fishergate, York, England. *American Journal of Physical Anthropology* 103 pp. 481-495.
- Knüsel, C. J., Roberts, C. A., and Boylston, A. (1996). Brief communication: When Adam delved: An activity related lesion in three human skeletal populations. *American Journal of Physical Anthropology* 100(3) pp. 427-433.
- Konigsberg, L. W., and Frankenberg, S. R. (1992). Estimation of age structure in anthropological demography. *American Journal of Physical Anthropology* 89 pp. 235-256.
- Konigsberg, L. W., and Frankenberg, S. R. (1994). Paleodemography: not quite dead. *Evolutionary Anthropology* 3 pp. 92-105.
- Konigsberg, L. W., and Frankenberg, S. R. (2013). Demography. In DiGangi, E. and Moore, M. K. (eds). *Research Methods In Human Skeletal Biology*. New York Academic Press.
- Konigsberg, L. W., and Hens, S. M. (1998). Use of ordinal categorical variables in skeletal assessment of sex from the cranium. *American Journal of Physical Anthropology* 107 pp. 97-112.
- Kontulainen. S., Sievanen, H., Kannus, P., Pasanen, M., and Vuori, I. (2003). Effect of long term impact-loading on mass, size and estimated strength of humerus and radii of female racquet-sport players: a peripheral quantitative computed tomography study between young and old starters. *Journal of Bone and Mineral Research* 28 pp. 251-267.
- Kotěrová, A., Velemínská, J., Dupej, J., Brzobohatá, A., Pilný, A., Bružek, J. (2017) Disregarding population specificity: its influence on the sex assessment methods from the tibia. *International Journal of Legal Medicine* 131(1) pp. 251-261.

- Krahl, H., Michealis, U., Pieper, H. G., Quack, G., and Montag, M. (1994). Stimulation of bonegrowth through sports - a radiologic investigation of the upper extremities in professional tennis players. *American Journal of Sports Medicine* 22 pp. 751-757.
- Kranioti, E. F., Isçan, M. Y., and Michalodimitrakis, M. (2008). Craniometric analysis of the modern Cretan population. *Forensic Science International* 180(2-3) pp. 110.e1-5
- Krantz, G. S. (1985). The fossil record of sex. In Hall, R.L. (ed). *Sexual Dimorphism In Homo Sapiens* New York: Praeger Publishers pp. 85-105.
- Krogman, W. M., and Isçan, M. Y. (1986). *The human skeleton in forensic medicine* (2nd Ed). Springfield IL.: Charles C Thomas.
- Kujala, U. M., Kaprio, K. J., and Sarna, S. (1994). Osteoarthritis of weight bearing joints of the lower limbs in former elite male athletes. *British Medical Journal* 308 pp. 231-234.
- Kujala, U. M., Kettunen, J., and Paananen, H. (1995). Knee osteoarthritis in former runners, soccer players, weight lifters and shooters. *Arthritis and Rheumatism* 38 pp. 539-546.
- Kulkarni, V., and Ramesh, B. R. (2012) A spectrum of vertebral synostosis. International Journal of Basic and Applied Medical Sciences 2(2) pp.71-77. Available from: http://www.cibtech.org/jms.htm (Accessed 17th December 2018).
- Kuzawa, C. W., and Bragg, J. M. (2012). Plasticity in human life history strategy: Implications for contemporary human variation and the evolution of the Genus *Homo. Current Anthropology* 53(S6) pp. S369-S382.
- Kwon, B. K., Song, F., Morrison, W. B., Grauer, J. M., Beiner, J. M., Vaccaro, A. R., and Albert, T. J.
 (2004). Morphologic evaluation of cervical spine anatomy with computed tomography.
 Journal of Spinal Discord Technology 17(2) pp. 102-107.
- Lai, P., and Lovell, N. (1992). Skeletal markers of occupational stress: a case study from Hudson's Bay Company fur trading post. *International Journal of Osteoarchaeology* 2(3) pp. 221-234.
- Lajeunesse, D. (2002). Altered subchondral osteoblast cellular metabolism in osteoarthritis: cytokines, eicosanoids and growth factors. *Journal of Musculoskeletal and Neuronal Interaction* (6) pp. 504-506.
- Lande, R. (1980). Sexual dimorphism, sexual selection and adaptation in polygenic traits. *Evolution* 34 pp. 292-305.

541

- Landis, J. R., and Koch, G. G. (1977). The measurement of observer agreement for categorical data. *Biometrics* 33(1) pp. 159-174.
- Lane, W. A. (1887). The causation of several variations and congenital abnormalities in the human skeleton. *Journal of Anatomy and Physiology* 21(4) pp. 586-610.
- Langdahl, B., Ferrari, S., Dempster, D. (2016). Bone modeling and remodeling: potential as therapeutic targets for the treatment of osteoporosis. *Therapeutic Advances in Musculoskeletal Disease* 8(6) pp.225-235.
- Langley-Shirley, N., Jantz, R.L. (2010). A Bayesian approach to age estimation in modern Americans from the clavicle. *Journal of Forensic Sciences* 55(3) pp. 571-583.
- Lanyon, P., Muir, K., Doherty, S., and Doherty, M. (2000). Assessment of a genetic contribution to osteoarthritis of the hip: sibling study. *British Medical Journal* 321(727) pp. 1179-1183.
- Larsen, C. S. (1995). Biological changes in human populations with agriculture. *Annual Review of Anthropology* 24 pp. 185-213.
- Larsen, C. S. (1997). *Bioarchaeology: Interpreting Behavior From The Human Skeleton*. Cambridge: Cambridge University Press.
- Larsen, C. S. (2003). Equality for the sexes in human evolution? Early hominid sexual dimorphism and implications for mating systems and social behaviour. *PNAS*, *100*(16), 9103-9104.
 Available from: http://www.pnas.org/content/100/16/9103 (Accessed 9th March 2012).
- Larsen, C. S. (2003). Animal source foods and human health during evolution. *Journal of Nutrition* 11(S2) pp. 3893S-3897S.
- Larsen, C. S. (2006). The changing face of bioarchaeology: An interdisciplinary science. In Buikstra,J. E., and Beck, L.A. (eds). *Bioarchaeology: The Contextual Analysis Of Human Remains*.San Diego: Academic Press pp. 359-374.
- Larsen, C. S., Kelly, R. L., Ruff, C. B., Schoeninger, M. J., Hutchinson, D. L., and Hemphill, B. E.
 (2008). Living on the margins: Biobehavioural adaptations in the Western Great Basin. In:
 Reitz, E. J., Scudder, S. J., and Scarry, C. M. (eds). Case Studies in Environmental
 Archaeology. New York, Springer New York pp. 161-189.
- Lasker, G. W. (1976). Physical Anthropology (2nd ed.). New York: Holt Rhinehart and Winston.
- Lazenby, R. A. (2002). Skeletal biology, functional asymmetry and the origins of "handedness". *Journal of Theoretical Biology* 218 pp. 129-138.

- Lees, B., Molleson, T., Arnett, T. R., and Stevenson, J. C. (1993). Differences in proximal femur bone density over two centuries. *Lancet* 341 pp. 673-675.
- Lessa, A. (2011). Spondylosis and lifestyle among the prehistoric coastal groups from Brazil. International Journal of Osteoarchaeology 21 pp. 660-668.
- Leutenegger, W. (1978). Scaling of sexual dimorphism in body size and breeding systems. *Nature*, 272 pp. 610-611.
- Leutenegger, K. W., and Cheveraud, J. (1982). Correlates of sexual dimorphism in primates: ecological and size variables. *International Journal of Primatology* 3 pp. 387-402.
- Leutenegger K. W., and Cheveraud J. M. (1985) Sexual dimorphism in primates. In Jungers W. L. (ed). *Size And Scaling In Primate Biology.* New York: Plenum Press pp. 33-50.
- Levangie, P. K., and Norkin, C. C. (2005). *Joint Structure and Function* (4th ed.) Philadelphia: F A Davis.
- Levins, R., and Lewontin, R. C. (1985). *The Dialectical Biologist*. New York: Harvard University Press.
- Lieberman, D. E. (1996). How humans grow thin skulls: experimental evidence for systemic cortical robusticity. *American Journal of Physical Anthropology* 101 pp. 217-236.
- Lieberman, D. E., Devlin, M. J., and Pearson, O. M. (2001). Articular area responses to mechanical loading: effects of exercise, age and skeletal location. *American Journal of Physical Anthropology* 116 pp. 266-277.
- Lieberman, L. S. (1982). Normal and Abnormal Sexual Dimorphic Patterns of Growth and Development. In R. L. Hall (ed.), *Sexual Dimorphism Of Homo Sapiens: A Question Of Size* New York: Praeger Publishing pp. 263-316.
- Lieverse, A. R., Stock, J. T., Katzenberg, M. A., and Haverkort, C. M. (2011). The bioarchaeology of habitual activity and dietary change in the Siberian Middle Holocene. In Pinhasi, R., and Stock, J. (eds.), *Human Bioarchaeology of the Transition To Agriculture* Chichester: John Wiley and Sons pp. 265-291.
- Lieverse, A. R., Weber, A. W., Bazaliiskiy, V. I., Goriunova, O. I., and Savel'ev, N. A. (2007).
 Osteoarthritis in Siberia's Cis-Baikal: skeletal indicators of hunter-gatherer adaptation and cultural change. *American Journal of Physical Anthropology* 132 pp. 1-16.

Liguoro, D., Vandermeersch, B., and Guerin, O. (1994). Dimensions of cervical vertebral bodies

according to age and sex. Surgical and Radiological Anatomy 16(2) pp. 149-150.

- LIm, J., and Wong, H. (2004). Variation of the cervical spinal torg ration with gender and ethnicity. *The Spine Journal* 4 pp. 396-401.
- Lindsay, R., Cosman, F., Zhou, H., Bostrom, M., Shen, V., Cruz, J. (2006). A novel tetracycline labeling schedule for longitudinal evaluation of short-term effects of anabolic therapy with a single iliac crest bone biopsy: early actions of teriparatide. *Journal of Bone Mineral Research* 21 pp.366-373
- Listi, G. A., and Mannheim, M. H. (2012). The use of vertebral osteoarthritis and osteophytosis in age estimation. *Journal of Forensic Science* 57(6) pp. 1537-1540.
- Loeser, R. F., and Shakkor, N. (2003). Aging or osteoarthritic: which is the problem? *Rheumatic Diseases Clinic Of North America* 25 pp. 653-673.
- Love, T. W., Fagan, A. B., & Fraser, R. D. (1999). Degenerative spondylolisthesis. Developmental or acquired? *Journal of Bone and Joint Surgery (British Volume)* 81(4) pp. 670-674.
- Lovejoy, C. O., Bernstein, A. H., and Heiple, K. G. (1976). The biomechanical analysis of bone strength: a method and its application to Platycnemia. *American Journal of Physical Anthropology* 44 pp. 489-506.
- Lovejoy, C. O., Meindl, R. S., Pryzbeck, T. R., and Mensforth, R. P. (1985). Chronological metamorphosis of the auricular surface of the ilium: a new method for the determination of age at death. *American Journal of Physical Anthropology* 68 pp.15-28.
- Lovejoy. C.O., Mensforth, R.P., and Armelagos, G. J. (1982). Five decades of skeletal biology as reflected in the American Journal of Physical Anthropology. In Spencer, F. (ed). *A History of American Physical Anthropology: 1930-1980*. New York: Academic Press pp. 329-336.
- Lovell, N. C. (1994). Spinal arthritis and physical stress at Bronze Age Harappa. *American Journal* of Physical Anthropology 93(2) pp. 149-184.
- Lozada, C. J., Diamond, H. S., Agnew, S., Bal, B.S., Basu, P.A., Chansky, H. (2011). Osteoarthritis. *Medscape Reference Drugs, Diseases and Procedures*. Available from: <u>https://link.springer.com/content/pdf/10.1007%2F978-1-4614-4839-6.pdf</u>. Accessed 12th October 2012.
- Luttmann, A., Jager, M., Laurig, W., and Schlegel, K. F. (1988). Orthopaedic diseases among transport workers. *International Archives of Occupational and Environmental Health* 61(3)

pp. 197-205.

- Maat, J. R., Mastwijk, R. W., and Vandervelde, E. (1995). Skeletal distribution of degenerative changes in vertebral osteophytosis, vertebral osteoarthritis and DISH. *International Journal of Osteoarchaeology* 5(3) pp. 288-298.
- MacLauglin, S. M., and Bruce, M.F. (1986). Population variation in sexual dimorphism in the human innominate bone. *Human Evolution* 1 pp. 221-231.
- MacLauglin, S. M., and Oldale, K. N. M. (1992). Vertebral body diameters and sex prediction. Annals of Human Biology 19(3) pp. 285-292.
- Mahato, N. K. (2011). Facet dimensions, orientation and symmetry at L5-S1 junction in lumbosacral transition states. *Spine* 36(9) pp. E569-E573.
- Mahfouz, M., Badawi, A., Meerkl, B., Fatah, E. E., Pritchard, E., Kesler, K., and Jantz, R. (2007). Patella sex determination by 3D statistical shape models and nonlinear classifiers. *Forensic Science International* 173(2-3) pp. 161-170.
- Mall, G., Gehring, K., Graw, M., and Hubig M. (2000). Determination of sex from femora. *Forensic Science International* 113 pp. 315-21.
- Mall, G., Hubig, M., Buttner, A., Kuznik, J., Penning, R., and Graw, M. (2001). Sex determination and estimation of stature from the long bones of the arm. *Forensic Science International* 117 pp. 23-40
- Manolagas, B. (1995). Mathematical model for repair of fatigue damage and stress fracture in osteonal bone. *Endocrine Review* 21 pp. 115-137.
- Manaster, B. J., and Osborn, A. G. (1987). CT patterns of facet fracture dislocations in the thoracolumbar region. *American Journal of Roentgenolology* 148(2) pp. 335-340.
- Mannoni, A., Briganti, M. P., DiBari, M., Ferruci, L., Constanzo, S., and Serni, U. (2005). Epidemiological profile of symptomatic osteoarthritis in older adults. A population study based in Dicomano, Italy. *Annals of Rheumatic Diseases* 62 pp. 576-587.
- Marchant, L. F., McGrew, W. C., and Eibl-Eibesfeldt, I. (1985). Is human handedness universal? Ethological analysis from three traditional cultures *Ethology* 101 pp. 239-258.
- Marchi, D., Facchini, F., and Belcastro, M. G. (2006). Biomechanical approach to the reconstruction of activity patterns in Neolithic Western Liguria, Italy. *American Journal of Physical Anthropology* 131 pp. 447-455.

- Marini, E., Racugno, W., and Borgognini-Tarli, S M. (1999). Univariate estimates of sexual dimorphism: the effects of intra-sexual variability. *American Journal of Physical Anthropology* 109 pp. 501-508.
- Marino, E.A. (1993). Sex estimation using the first cervical vertebra. *American Journal of Physical Anthropology* 97(2) pp. 127-133.
- Marino, E. A. (1993). Estimation Of Race And Sex Using Multiple-Regression And Discriminant Function Analysis Of The First Cervical Vertebra Of Homo Sapiens. (Master's Thesis). Department of Anthropology, University of Missouri-Columbia.
- Marino, E .A. (1997). A pilot study using the first cervical vertebra as an indicator of race. *Journal of Forensic Science* 42(6) pp. 1114-1118.
- Martin, R. B. (2007). Age and sex-related changes in the structure and strength of the human femoral shaft. *Journal of Musculoskeletal and Neuronal Interactions* 4 pp. 48-53.
- Marlow, E. J., and Pastor, P. F. (2011). Sex determination using the second cervical vertebra a test of method. *Journal of Forensic Science* 56(1) pp. 165-169.
- Martin, D. L. (2013). Bioarchaeology of individuals: Identity, social theory and skeletal analysis. In Martin, D. L., Harrod, R. P., and Perez, V. P (eds.), *Bioarchaeology: An Integrated Approach To Working With Human Remains*. New York: Springer Science and Business Media.

Martin, R., and Saller, K. (1957). Lehrbuch der Anthropologie. Stuttgart: Gustav Fischer Verlag.

- Martyn-St James, M., and Carroll, S. (2010). Effects of different impact exercise modalities on bone mineral density in premenopausal women: a meta-analysis. *Journal of Bone and Mineral Research* 28 pp. 251-267.
- Masharawi, Y. M., Smagar, P., Albert, H. B., Dar, G., Steilberg, N., Medlej, B., and Peled, N. (2008). Facet asymmetry in normal vertebral growth: characterisation and etiologic theory of scoliosis. *Spine* 33(8) pp. 898-902.
- Masharawi, Y., Rothschild, B., Dar, G., Peleg, S., Robinson, D., Been, E., and Hershkovitz, I. (2004). Facet orientation in the thoracolumbar spine: three-dimensional anatomic and biomechanical analysis. *Spine* 29 (16) pp. 1755-1763.
- Masharawi, Y., Rothschild, B., Salame, K., Dar, G., Peleg, S., and Hershkovitz, I. (2005). Facet tropism and inter-facet shape in the thoracolumbar vertebrae. *Spine* 30(11) pp. E281-E292.

- Masharawi, Y. M., Alperovitch-Najenson, D., Steinberg, N., Dar, G., Peleg, S., Rothschild, B., and Hershkovitz, I. (2007). Lumbar facet orientation in spondylolysis: a skeletal study. *Spine* 32(6) pp. E176-180.
- Masharawi, Y. M., Peleg, S., Albert, H. B., Dar, G., Steinberg, N., Medlej, B., and Hershkovitz, I. (2008). Facet asymmetry in normal vertebral growth: characterization and etiologic theory of scoliosis. *Spine* 33(8) pp. 898-902.
- Masharawi Y., Dar G., Peleg S., Steinberg N., Medlej B., May H., Abbas J., and Hershkovitz I. (2010) A morphological adaptation of the thoracic and lumbar vertebrae to lumbar hyperlordosis in young and adult females. *European Spine Journal* 19 pp. 768-773.
- Mays, S. (1999). A biomechanical study of activity patterns in a medieval human skeletal assemblage. *International Journal of Osteoarchaeology* 9 pp. 68-73.
- Mays, S. (2002). The Archaeology of Human Bones. London: Routledge.
- Mays, S. (2002). Asymmetry in metacarpal cortical bone in a collection of British post-medieval human skeletons. *Journal of Archaeological Science* 29 pp. 435-441.
- Mays, S. (2007). Part Three: The Human Remains. In Mays, S., Harding, C., and Heighway, C. (eds). Wharram Churchyard V.11 (Wharram, a Study of Settlement on the Yorkshire Wolds). London: English Heritage pp. 77-193.
- Mays, S., Brickley, M., and Ives, R. (2006). Skeletal manifestations of rickets in infants and young children in a historic population from England. *American Journal of Physical Anthropology* 129(3) pp. 362-374.
- Mays, S., and Cox, M. (2000). Sex determination in skeletal remains. In Cox, M., and Mays, S.
 (eds). *Human Osteology In Archaeology And Forensic Science*. Cambridge: Cambridge University Press pp. 117-130.
- Mays, S., Lees, B., and Stevenson, J. C. (1998). Age-dependent bone loss in the femur in a medieval population. *International Journal of Osteoarchaeology* 8(2) pp. 97-106.
- McCann, P. (1962). The incidence and radiologic value of signs of scurvy. *British Journal of Radiology* pp. 683-686.
- McHenry, H. M. (1992). Body size and proportions in early Hominids. *American Journal of Physical Anthropology* 87 pp. 407-431.

McKeown, A. H., and Schmidt, R. W. (2013). Geometric Morphometrics. In DiGangi, E. A., and

Moore, M. K. (eds). Research Methods In Human Skeletal Biology. London: Elsevier.

- McMillan, G., and Nichols, L. (2005). Osteoarthritis and meniscal disorders of the knee as occupational diseases of miners. *Occupational and Environmental Medicine* 62(8) pp. 567-576.
- McMinn, R. M. H., Pegington, J., and Abrahams, P. (1993). *A Colour Atlas of Human Anatomy* (3rd ed). London: Wolfe Publishing Ltd.
- Medina, S.C. (2011). Dimorfismo Sexual En Columna Vertebral. Universidad de Granada, PhD Dissertation. Available at : <u>http://digibug.ugr.es/bitstream/handle/10481/19973/20157459.pdf?sequence=1&is</u> Allowed=y (Accessed 19th December 2018).
- Medland, S. E., Perelle, I. B., DeMonte, D., and Ehrman, L (2004). Effects of culture, sex and age on the distribution of handedness: an evaluation of the sensitivity of three measures of handedness. *Laterality 9 pp.* 297-297.
- Meiklejohn, C., and Babb, J. (2011). Long bone length, stature and time in the European Late
 Pleistocene and Early Holocene. In Pinhasi, R., and Stock, J. T. (eds). *Human Bioarchaeology And The Transition To Agriculture.* Chichester: Wiley-Blackwell pp. 153-202.
- Meiklejohn, C., Schentag, C., Venema, A., and Key, P. (1984). Socioeconomic change and patterns of pathology and variation in the Mesolithic and Neolithic of Western Europe: Some suggestions. In Cohen, M. N., and Armelagos G. J. (eds). *Paleopathology At The Origins Of Agriculture*. London: Academic Press pp. 75-100.
- Meindl, R. S., and Lovejoy, C. O. (1985). Ectocranial suture closure: A revised method for the determination of skeletal age at death based on the latero-anterior sutures. *American Journal of Physical Anthropology* 68 pp. 57-66.
- Meindl, R. S., Lovejoy, C. O., Mensforth, R. P., and Carlos, L. D. (1985). Accuracy and direction of error in sexing of the skeleton: Implications for paleodemography. *American Journal of Physical Anthropology* 68(1) pp. 79-85.
- Meindl, R. S., and Russell, K. F. (1998). Recent advances in method and theory in Paleodemography. *Annual Review of Anthropology* 27 pp. 375-399.
- Merbs, C. F. (1983). Patterns of activity-produced pathology in a Canadian Inuit population. In Ottowa, N. (ed). *Archaeological Survey Of Canada, Paper no: 119*. Ottawa.

- Merbs, C. F. (2001). Degenerative spondylolisthesis in ancient and historic skeletons from New Mexico pueblo sites. *American Journal of Physical Anthropology* 116 pp. 285-295.
- Mercer, C. S., and Bogduk, N. (1993). Intra-articular inclusions of the cervical synovial joints. British Journal of Rheumatology 32 pp. 705-711.
- Meyer, C., Nicklisch, N., Held, P., Fritsch, B., and Alt, K. W. (2011). Tracing patterns of activity in the human skeleton: An overview of methods, problems, and limits of interpretation. *Homo* 62 pp. 202-217.
- Milella, M., Belcastro, M., Zollikoffer, C. P. E., and Mariotti, V. (2012). The effects of age, sex and physical activity on entheseal morphology in a contemporary Italian skeletal collection. *American Journal of Physical Anthropology* 148 pp. 379-388.
- Miles, A., and Conheeney, J. (2005). *A Post-medieval population from London: Excavations in the St Bride's Lower Churchyard, 75-82 Farringdon Street, City of London EC4*. MoLAS Studies Series. (unpublished).
- Miles, A. E. W. (1963). The dentition in the assessment of individual age in skeletal material. In Brothwell, D.(ed) .Dental anthropology: Symposia of the Society for the Study of Human Biology (Vol. 5). Oxford: Pergamon Press.
- Milgram, J. W. (1983). Morphological alterations of the subchondral bone in advanced degenerative arthritis. *Clinical and Orthopaedic Related Research* 173 pp. 293-312.
- Milne, G., and Reynolds, A. (1994). The Medieval Lady Chapel at St Bride's Church. *London Archaeologist* 7(6) pp. 143-149.
- Milne, N. (1990). Sexing of human hip bones. Journal of Anatomy 172 pp. 221-226.
- Milne, N. (1991). The role of zygapophyseal joint orientation and uncinate processes in controlling motion in the cervical spine. *Journal of Anatomy* 178 pp. 189-201.
- Milner, G. R., and Boldsen, J. L. (2012a). Estimating age and sex from the skeleton, a paleopathological perspective. In Grauer, A. L. (ed). A Companion To Paleopathology.
 Chichester: Blackwell Publishing Ltd.
- Milner, G. R., and Boldsen, J. L. (2012b). Humeral and femoral head diameters in recent White American skeletons. *Journal of Forensic Science* 57(1) pp. 35-40.
- Mitchell, N. S., and Cruess, R. L. (1977). Classification of degenerative arthritis. *Canadian Medical Society Journal* 117(7) pp. 763-765.

- Molleson, T. (2007). A method for the study of activity related skeletal morphologies. Bioarchaeology of the Near East 1 pp. 5-33.
- Molleson, T., and Cox, M. (1993). The Spitalfields Project: Volume II The Anthropology: The Middling Sort. London: CBA Report 86.
- Molnar, P. (2010). Patterns of physical activity and material culture on Gotland Sweden, during the middle Neolithic. *International Journal of Osteoarchaeology* 20(1) pp. 1-14.
- Molnar, P., Ahlstrom, T. P., and Leden, I. (2011). Osteoarthritis and activity An analysis of the relationship between eburnation, musculoskeletal stress markers (MSM) and age in two Neolithic hunter-gatherer populations from Gotland Sweden. *International Journal of Osteoarchaeology* 21(3), pp. 283-291.
- Moore, A. P., and Petty, N. J. (2005). Function of the spine. In Trew, M. and Everett, T. (eds). *Human Movement* (5th ed) London: Elsevier Churchill Livingstone pp. 207-230.
- Moore, J., and Buckberry, J. (2016). The use of corsetry to treat Pott's disease of the spine from 19th century Wolverhampton, England. *International Journal of Paleopathology* 14 pp. 74-80.
- Moore K. L., Dalley A. F., and Agur, A. M. R. (2006). *Clinically Orientated Anatomy* (5th Ed). Philadelphia: Lippincott Williams and Wilkins.
- Moore, M. K. (2013). Sex estimation and assessment. In DiGangi, E. A., and Moore, M. K. (eds). *Research Methods In Human Skeletal Biology* Oxford: Elsevier pp. 93-102.
- Moore, M. K., and Persaud, T. V. N. (2003). *The Developing Human: Clinically Orientated Embryology*. 7th ed. Philadelphia: W. B. Saunders Company
- Morris, A. G. (2010). Racial identification of single skulls in forensic cases: When myth becomes reality. In Strkalj, G. (ed). *Teaching Human Variation*. New York: Nova Science Publishers Inc.
- Mosekilde, L. (2000). Age-related changes in bone mass, structure and strength-effects of loading. Zeitschrift für Rheumatologie 59 pp. 1-9.
- Mosekilde, L., and Mosekilde, L. (1990). Sex differences in age-related changes in vertebral body size, density and biomechanical competence in normal individuals. *Bone* 11(2) pp. 67-73.
- Moskowitz, R. W., Kelly, M. A., and Lewallen, D. G. (2004). Understanding osteoarthritis of the knee. *American Journal of Orthopaedics* 33(2) pp. S5-S9.
- Mueller, U., and Mazur, A. (2001). Evidence of unconstrained directional selection for male tallness. *Behavioural and Ecological Sociobiology* 50 pp. 302-311.
- Mummert, A., Esche, E., Robinson, J., and Armelagos, G. J. (2011). Stature and robusticity during the agricultural transition: evidence from the bioarchaeological record. *Economics and Human Biology* 9(3) pp. 284-301.
- Murail, P., Bruzek, J., Houet, F., and Cuhna, E. (2005). DSP: A tool for probabilistic sex diagnosis using worldwide variability in hip-bone measurements. *Bulletins et Memoires de la Société d'Anthropologie de Paris* 17(3-4) pp. 167-176.
- Murphy, A. (2002). Articular surfaces of the pectoral girdle: sex assessment of prehistoric New Zealand Polynesian skeletal remains. *Forensic Science International* 125(2-3) pp. 134-136.
- Murrie, V. L., Dixon, A. K., Hollingsworth, W., Wilson, H., and Doyle, T. A. C. (2003). Lumbar lordosis: Study of patients with and without back pain. *Clinical Anatomy* 16 pp. 144-147.
- Nagesha, K. R., Kanchana, T., and Bastiab, B. K. (2007). Sexual dimorphism of acetabulum-pubis index length in South-Indian population. *Legal Medicine* 9(6) pp. 305-308.
- Nakatsukasa, M., Hayama, S., and Preuschoft, H. (1995). Post-cranial skeleton of a macaque trained for bipedal standing and walking and implications for functional adaptation. *Folia Primatologica* 64 (1-2) pp. 1-29.
- Nathan, H. (1962). Osteophytes of the vertebral column. *Journal of Bone and Joint Surgery* (*American Volume*) 44 pp. 243-268.
- Nathan, M., Pope, M. H., and Grobler, L. J. (1994). Osteophyte formation in the vertebral column: a review of the etiologic factors – Part 1. *Contemporary Orthopaedics* 29 pp. 113-119.
- Nikita, E. (2014). Age-associated variation and sexual dimorphism in adult cranial morphology: implications in anthropological studies. *International Journal of Osteoarchaeology* 24(5) pp. 557-569
- Nilsson, M., Sundh, D., Ohisson, C., Karlsson, M., Mellstrom, D., and Lorentzon, M. (2014).
 Exercise during growth and young adulthood is independently associated with cortical bone size and strength in old Swedish men. *Journal of Bone and Mineral Research* 29(8) pp. 1795-1804.
- Novak, M., and Slaus, M. (2011). Vertebral pathologies in two early modern (16th -19th century) populations from Croatia. *American Journal of Physical Anthropology* 145(2) pp. 270-281.

- Nuki, G. (1999). Osteoarthritis: a problem of joint failure. *Journal of Rheumatology* 58 pp. 142-147.
- O'Neill, T. W., McCloskey, E. V., Kanis, J. A., Bhalla, A. K., Reeve, J., Reid, D. M., Todd, C., Woolf,
 A.D., and Silman, A. J. (1999). The distribution, determinants and clinical correlates of
 vertebral osteophytosis: a population based survey. *Journal of Rheumatology* 26 pp. 842-848.
- O'Rahilly, R. (1986). *Gardner-Gray-O'Rahilly Anatomy. A Regional Study of Human Structures* (5th edition) Philadelphia: Saunders.
- O'Rahilly, R. (1996). Human Embryology And Teratology. New York: John Wiley and Sons
- Oliver, J., and Middleditch, A. (1991). *Functional Anatomy Of The Spine*. Oxford: Butterworth Heinemann.
- Oliveria, S. A., Felson, D. T., and Reed, J. I. (1995). Incidence of symptomatic hand, hip and knee osteoarthritis among patients in a health maintenance organisation. *Arthritis and Rheumatism* 38 pp. 1134-1141.
- Ominsky, M., Libanati, C., Nui, Q., Boyce, R., Kostenuik, P., Wagman, R. (2015). Sustained modeling-based bone formation during adulthood in cynomolgus monkeys may contribute to continuous BMD gains with denosumab. *Journal of Bone Mineral Research* 30 pp. 1280-1289.
- Ominsky, M., Nui, Q., Li, c., Li, X., Ke, H. (2014). Tissue-level mechanisms responsible for the increase of bone formation and bone volume by sclerostin antibody. *Journal of Bone Mineral Research* 29 pp. 1424-1430.
- Orlove, B. (1980). Ecological anthropology. Annual Review of Anthropology 131 pp. 334-342.
- Ortner, D J. (1998). Male/female immune reactivity and its implications in interpreting evidence in human skeletal paleopathology. In Grauer, A. L., and Stuart-Macadam, P. (eds). *Sex And Gender In Paleopathological Perspective*. Cambridge: Cambridge University Press.
- Ortner, D. J. (2003). *Identification Of Pathological Conditions In Human Skeletal Remains*. New York: Academic Press.
- Ortner, D. J. (2012). Differential diagnosis and issues in disease classification. In Grauer, A. L (ed). *A Companion To Paleopathology*. Chichester: Blackwell Publishing Ltd.

Ostuka, Y., An, H. S., Ochia, R. S., Andersson, G. B. J., Orias, A. A., and Inoue, N. (2010). In vivo

measurement of lumbar facet joint area in asymptomatic and chronic low back pain subjects. *Spine* 35(8) pp. 924-928.

- Osborne, D. L., Simmons, T. L., and Nawrocki, S. P. (2004). Reconsidering the auricular surface as an indicator of age at death. *Journal of Forensic Science* 49 pp. 905-911.
- Ostendorf Smith, M. (2013). Paleopathology. In DiGangi, E. A., and Moore, M. K. (eds). *Research Methods In Human Skeletal Biology*. New York: Elsevier pp. 181-207.
- Ostrofsky, K. R., and Churchill, S. E. (2015). Sex determination by discriminant function analysis of lumbar vertebrae. *Journal of Forensic Science* 60 (1) pp. 21-28.
- Oura P., Paananen, M., and Ninimaki, J. (2016). Effects of leisure-time physical activity on vertebral dimensions in the North Finland Birth Cohort 1966. *Scientific Reports* 6 p. 27844
- Oura P., Paananen, M., and Ninimaki, J. (2016) Effects of occupational physical activities on vertebral dimensions in midlife in the Northern Finland Birth Cohort 1966. *Occupational and Environmental Medicine* 74 (5) pp. 351-356.
- Overton, C. M., and Grossman, J. W. (1952). Anatomical variations in the articulations between the second and third cervical vertebrae. *Journal of Bone and Joint Surgery (American Volume)* 34(A-1) pp. 155-161.
- Özener, B. (2010). Fluctuating and directional asymmetry in young human males: effect of heavy working condition and socioeconomic status. *American Journal of Physical Anthropology* 143 pp. 112-120.
- Pal, G .P., and Routal, R. V. (1999). Mechanism of change in the orientation of the articular process of the zygapophyseal joint at the thoracolumbar junction. *Journal of Anatomy* 195 pp. 119-209.
- Pal, G. P, and Routal, R. V. (2001). The orientation of articular facets of the zygapophyseal joints at the cervical and upper thoracic region. *Journal of Anatomy* 198 pp. 431-441.
- Panjabi, M. M., Goel, V., Oxland, T., Takata, K., Duranceau, J., Krag, M., and Price, M. (1993).
 Human lumbar vertebrae: quantitative three-dimensional anatomy. *Spine* 17(8) pp. 299-306.
- Panjabi, M. M., Oxland, T., Takata, K., Goel, V., Duanceau, J., and Krag, M. (1993). Articular facets of the human spine. Quantitative three-dimensional anatomy. *Spine* 18 pp. 1298-1310.

Parfitt, A. M. (1998). Osteomalacia and Related Disorders. In Alvioli, L., and Krane, S. (eds).

Metabolic Bone Disease And Clinically Related Disorders. San Diego: Academic Press pp. 327-386.

- Parfitt, A. M. (2002). Targeted and non-targeted bone remodeling: relationship to basic multicellular unit organisation and progression. *Bone* 30 pp. 5-7.
- Parfitt, A. M. (2003). Renal bone disease: a new conceptual framework for the implications for the interpretation of bone histomorphometry. *Current Opinion in Nephrology and Hypertension* 12 pp. 387-403.
- Parfitt, K. (2002). A prehistoric site off Green Lane, Whitfield, near Dover Archaeologica Cantiana Vol. cxxii, pp. 373-396.
- Parfitt, K., and Anderson, T. (2012). Buckland Anglo-Saxon cemetery, Dover excavations 1994 *The Archaeology of Canterbury* (Vol. VI).
- Pastor, F. (2005). Sexual dimorphism in vertebral dimensions at the T12/L1 junction. *Proceedings on the American Academy of Forensic Sciences;* New Orleans, Publication Printers 11 pp. 302-303.
- Patel, M. M., Gohil, D. V., and Singel, T. C. (2004). Orientation of the superior facets from C3 to S1 vertebrae. *Journal of the Anatomical Society of India* 53(2) pp. 35-39.
- Patel, M. M., Singel, T. C., Gohil, D. V., and Pandaya, A. M. (2007). Study of osteometric measurements of articular facets from C3 to S1. *Journal of the Anatomical Society of India* 56(1) pp. 1-6.
- Pauwels, F. (1976). Biomechanics Of The Normal And Diseased Hip. Berlin: Springer-Verlag.
- Pawlowski, B., Dunbar, R. I. M., and Lipowicz, A. (2000). Tall men have more reproductive success. *Nature* 403 pp. 156-158.
- Payne, E. E., and Spillane, J. D. (1957). The cervical spine: an anatomic-pathological study of 70 specimens (using a special technique) with particular reference to the problem of cervical spondylosis. *Brain* 80 pp. 571-596.
- Pearson, O., and Buikstra, J. (2006). Behaviour and the bones. In Buikstra, J., and Beck, L. (eds). Bioarchaeology: The Contextual Analysis Of Human Remains. New York: Elsevier pp. 207-225.
- Pearson, O. M., and Lieberman, D. E. (2004). The aging of Wolff's "law": ontogeny and responses to mechanical loading in cortical bone. *American Journal of Physical Anthropology* Suppl

39 pp. 63-99.

- Peat, J. (2001). *Health Science Research: A Handbook Of Quantitative Methods*. Sydney: Allen and Unwin.
- Penton-Voak, J. S., Little, A. C., Jones, A. C., and Burt, D. M. (2003). Female condition influences preferences for sexual dimorphism in faces of male humans. *Journal of Comparative Psychology* 117 pp. 264-271.
- Perneger, T. V. (1998). What's wrong with Bonferroni adjustments. *British Medical Journal.* 316 pp. 1236-1238.
- Pettifor, J., and Daniels, E. (1997). Vitamin D deficiency and nutritional rickets in children. In
 Feldman, D., Glorieux, F. and Pike, J. (eds). *Vitamin D*. San Diego: Academic Press pp. 663-678.
- Phenice, T. W. (1969). A newly developed visual method of sexing the os pubis. *American Journal* of Physical Anthropology 30(2) pp. 297-301.
- Philp, B. J. (1973). Saxon gold ring found at Dover Kent Archaeological Review Vol. 31 pp. 10-11.
- Philp, B. J. (1989). The Roman house with Bacchic murals at Dover *Fifth Research Report in the Kent Monograph Series*. Dover.
- Plato, C. C., Fox, K. M., and Tobin, J. D. (1994). Skeletal changes in human aging. In Crews, D. E., and Garruto, R. M. (eds). *Biological Anthropology and Human Aging* Oxford: Oxford University Press pp. 272-300.
- Plavcan, J. (2001). Sexual dimorphism in primate evolution. *Yearbook of Physical Anthropology*. 44 pp. 25-53.
- Plavcan, J. M., van Schaik C. P. (1994) Canine dimorphism. *Evolutionary Anthropology* 2 pp. 208-214.
- Plochocki, J. H., and Organ, J. (2003). The effects of differential mechanical loading on articular surface area in miniature swine. *American Journal Physical Anthropology* 36 (Suppl) pp. 169-173.
- Plomp, K. A., Viðarsdóttir, U. S., Weston, D. A., Dobney, K., and Collard, M. (2015). The ancestral shape hypothesis: an evolutionary explanation for the occurrence of intervertebral disc herniation in humans. *BMC Evolutionary Biology* 15 pp. 68-72.

- Pomeroy, E., and Zakrzewski, S. K. (2009). Sexual dimorphism in diaphyseal cross-sectional shape in the medieval Muslim population of Éjica, Spain and Anglo-Saxon Great Chesterford, UK. *International Journal of Osteoarchaeology* 19 pp. 50-65.
- Ponce, P. V. (2010). A Comparative Study Of Activity Related Skeletal Changes In 3rd -2nd Millennium BC Coastal Fishers And 1st Millennium AD Inland Agriculturalists In Chile, South America. (Doctoral Thesis Doctoral Thesis), Durham University. Available from: http://etheses.dur.ac.uk/546/. (Accessed 7th November 2013).
- Poole, A. R. (1999). An introduction to the pathophysiology of osteoarthritis. *Frontiers in Bioscience* 4 pp. 662-670.
- Poss, R. (1984). Functional adaptation of the human locomotor system to normal and abnormal loading patterns. *Calcified Tissue Research* 36 pp. 151-161.
- Prescher, A. (1998). Anatomy and pathology of the aging spine. *European Journal of Radiology* 27 pp. 181-195.
- Preuschoft, H. (2004). Mechanisms for the acquisition of habitual bipedality: Are there
 biomechanical reasons for the acquisition of upright bipedal posture? *Journal of Anatomy* 204(5) pp. 363-380.
- Prieto-Alhambra, D., Judge, A., Javaid, M. K., Cooper, C., Diez-Perez, A., and Arden, N. K. (2014).
 Incidence and risk factors for clinically diagnosed knee, hip and hand osteoarthritis:
 influence of age, gender and osteoarthritis affecting other joints. *Annals of Rheumatic Disease* 73 pp. 1659-64.
- Purkait, R. (2005). Triangle identified at the proximal end of femur: a new sex determinant. Forensic Science International 147(2-3) pp. 135-130.
- Putti, V. (1927a). Lady Jones Lecture. On new conceptions in the pathogenesis of sciatic pain. Lancet 2 pp. 53-60.
- Putti, V. (1927b). New conceptions in the pathogenesis of sciatic pain. Lancet 9 pp. 53-60.
- Radin, E. L., Martin, R. B., Burr, D. B., Caterson, B., Boyd, R. D., and Goodwin, C. (1984). Effects of mechanical loading on the tissues of the rabbit knee. *Journal of Orthopaedic Research* 1 pp. 221-234.
- Radin, E. L., Orr, R. B., Kelman, J. L., Paul, I. L., and Rose, R. M. (1982). Effect of prolonged walking on concrete on the knees of sheep. *Journal of Biomechanics* 15 pp. 487-492.

- Rafferty, K. L., and Ruff, C. B. (1994). Articular structure and function in Hylobates, Colobus and Papio. *American Journal of Physical Anthropology* 94 pp. 395-408.
- Ramsthaler, F., Kreutz, K., and Verhoff, M. A. (2007). Accuracy of metric sex analysis remains using Fordisc based on a recent skull collection. *International Journal of Legal Medicine* 121(6) pp. 477-482.
- Rauch, F., and Schonau, E. (2001). The developing bone: slave or master of its cells and molecules. *Pediatric Research* 50 pp. 309-314.
- Raxtor, M. H., Auerbach, B. M., and Ruff, C. B. (2006). Revision of the Fully technique for estimating statures. *American Journal of Physical Anthropology* 130(3) pp. 374-384.
- Reid, a., Schneider-Kolsky, M.E., O'Donnell, C. J. (2008). Comparison of computed radiography and multi-detector-computed tomography in the detection of post-mortem metacarpal index. *Forensic Science International* 177 pp. 192-198.
- Reijman, M., Pols, H. A., Bergink, A. P., Hazes, J. M., Belo, J. N., and Lieverse, A. M. (2007). Body mass index associated with onset and progression of osteoarthritis of the knee but not the hip: the Rotterdam Study. *Annals of the Rheumatic Diseases* 66 pp. 158-162.
- Reno, P. L., Meindl, R. S., McCollum M .A., Owen Lovejoy C. (2003) Sexual dimorphism in *Australopithecus Afarensis* was similar to that of modern humans. *PNAS* 100(16), 9404-9409. Available from: <u>http://www.pnas.org/content/100/16/9404</u>. (Accessed 03/06/18).
- Resnick, D. (2002). Degenerative diseases of the spine. In Resnick, D. (ed). *Diagnosis Of Bone And Joint Disorders* (4th Ed.). Philadelphia: W B Saunders.
- Reynolds, E. L. (1947) The bony pelvis in pre-pubertal children. *American Journal of Physical Anthropology* 5 pp. 165-200.
- Richards, J. (1999). Statistics of sexual size dimorphism. *Journal of Human Evolution* 26 pp. 423-458.
- Rickland, D. E. and Tobias, P. V. (1986). Unusually low sexual dimorphism of endocranial capacity in a Zulu cranial series. *American Journal of Physical Anthropology* 71 pp. 285-293).
- Rios, L. (2003). Brief Communication: sex determination accuracy of the minimum supero-inferior femoral neck diameter in a contemporary Guatemalan population. *American Journal of Physical Anthropology* 122 pp. 123-126.

Rios-Frutos, L. (2005). Metric determination of sex from the humerus in a Guatemalan forensic

sample. Forensic Science International 147 pp. 153-157.

- Robb, J. (1994). Skeletal signs of activity in the Italian Metal Ages: methodological and interpretive notes. *Human Evolution* 9 pp. 215-229.
- Robb, J. (2002). Time and biography: Osteobiography of the Italian Neolithic Lifespan. In Hamilakis, Y., Pluchiennik, M., and Tarlow, S. (eds). *Thinking Through The Body: Archaeologies of corporeality*. New York: Springer.
- Roberts, C., and Cox, M. (2003). *Health And Disease In Britain: From Prehistory To The Present Day*. Stroud: Sutton Publishing.
- Roberts, C., and Manchester, K. (2005). The Archaeology Of Disease. Stroud: Sutton Publishing.
- Roberts, C. A. (2006). A View From afar: Bioarchaeology in Britain. In Buikstra, J. E., and Beck, L. A. (eds). *Bioarchaeology: The Contextual Analysis Of Human Remains*. New York: Elsevier.
- Roberts, C. A., and Connell, B. (2004). Guidance on recording palaeopathology. In Brickley, M., and McKinley, J. I. (eds). *Guidelines To The Standards For Recording Human Remains.* IFA Paper No 7, Reading and Southampton: British Association for Biological Anthropology and Osteoarchaeology and Institute of Field Archaeologists pp. 34-79.
- Roberts, D. F. (1995). The pervasiveness of plasticity. In Mascie-Taylor, C. G. N., and Bogin, B. (eds). *Human Variability And Plasticity*. Cambridge: Cambridge University Press.
- Robling, A. G., Burr, D. B., and Turner, C. H. (2001). Partitioning a daily mechanical stimulus into discrete loading bouts improves the osteogenic response to loading. *Journal of Bone and Mineral Research* 15 pp. 1596-1602.
- Robling, A. G., Burr, D. B., and Turner, C. H. (2001). Recovery periods restore mechanosensitivity to dynamically loaded bone. *Journal of Experimental Biology* 204 pp. 3389-3399.
- Robling, A. G., Burr, D. B., and Turner, C. H. (2006). Biomechanical and molecular regulation of bone remodeling. *Annual Review of Biomedical Engineering* 8 pp. 455-498.
- Rogers, J., and Waldron, T. (1995). A Field Guide To Joint Disease In Archaeology. New York: John Wiley.
- Rogers, J., Waldron, T., and Dieppe, P. (1987). Arthropathies in paleopathology: the basis of classification according to the most probable cause. *Journal of Archaeological Science* 14 pp. 179-193.

- Rogers, J., Watt, I., and Dieppe, P. (1990). Comparison of visual and radiographic detection of bony changes in the knee joint. *British Medical Journal* 300 pp. 367-368.
- Rogers, T. L. (1999). A visual method of determining the sex of skeletal remains using the distal humerus. *Journal of Forensic Science* 44(1) pp. 57-60.
- Rojas-Sepulveda, C., Ardagna, Y., and Dutour, O. (2008). Paleoepidemiology of vertebral degenerative disease in a Pre-Columbian Muisca series from Colombia. *American Journal of Physical Anthropology* 135(4) pp. 416-430.
- Rose, J. C., Burnett, B. A., Nassaney, M., and Blaeur, M W. (1984). Paleopathology and the origins of maize agriculture in the Lower Mississippi Valley and Caddoan Culture Areas. In Cohen, M. N., and Armelagos, G. J. (eds). *Paleopathology at the Origins of Agriculture* Orlando: Academic Press pp. 393-424.
- Rosing, F. W., Graw, M., Marre, B., Ritz-Timee, S., Rothschild, M.A., and Rotzscher, K. (2007). Recommendations for the forensic diagnosis of sex and age from skeletons. *Homo* 58(1) pp. 75-79
- Ross, P. D., Davis, J. W., Epstein, R. S., and Wasnich, R. D. (1991). Pre-existing fractures and bone mass predict vertebral fracture incidence in women. *Annals of Internal Medicine* 114(11) pp. 919-923.
- Ross, W. D., and Ward R. (1982) Human proportionality and sexual dimorphism. In Hall, R. L. (ed). Sexual Dimorphism In Homo Sapiens. New York: Praeger pp. 317-362
- Rothschild, B. (1997). Porosity: a curiosity without diagnostic significance. *American Journal of Physical Anthropology* 104 pp. 529-533.
- Rubin, C. T., and Lanyon, L. E. (1984). Regulation on bone formation by applied dynamic loads. *Journal of Bone and Joint Surgery* 66 pp. 367-402.
- Rubin, C. T., McLeod, K. J., and Bain, S. D. (1990). Functional strains and cortical bone adaptation: epigenetic assurance of skeletal integrity. *Journal of Biomechanics* 23 pp. 43-54.
- Ruff, C. B. (1984). Allometry between length and cross-sectional dimensions of the femur and tibia in *Homo sapiens*. *American Journal of Physical Anthropology 65 pp. 347-358*.
- Ruff, C. B. (1987). Sexual dimorphism in human lower limb bone structure: relationship to subsistence strategy and sexual division of labour. *Journal of Human Evolution* 16 pp. 391-416.

- Ruff, C. B. (1992). Biomechanical analyses of archaeological human skeletal samples. In Katzenberg, M.A., and Saunders, S.R. (eds). Skeletal Biology Of Past People: Research Methods (pp. 37-58).
- Ruff, C. B. (2000a). Biomechanical analyses of archaeological human skeletons. In Katzenberg,
 M.A., and Saunders, S.R. (eds). *Biological Anthropology Of The Human Skeleton* New York:
 Wiley pp. 71-102.
- Ruff, C. B. (2000b). Body size, body shape and long bone strength in modern humans. *Journal of Human Evolution.* 38 pp. 269-290.
- Ruff, C. B. (2002). Variations in human body size and shape. *Annual Review of Anthropology* 31 pp. 211-232.
- Ruff, C B. (2003a). Long bone articular and diaphyseal structure in Old World monkeys and apes. American Journal of Physical Anthropology 120 pp. 305-342.
- Ruff, C. B. (2003). Ontogenetic adaptation to bipedalism: age changes in femoral to humeral length and strength proportions in humans, with a comparison *Journal of Human Evolution* 45(4) pp. 317-49.
- Ruff, C B. (2005). Mechanical determinants of bone form: insights from skeletal remains. *Journal* of Musculoskeletal and Neuronal Interactions 5 pp. 202-212.
- Ruff, C. B. (2007). Body size prediction from juvenile skeletal remains. *American Journal of Physical Anthropology* 133 pp. 698-716.
- Ruff, C. B. (2008) Biomechanical analyses of archaeological human skeletons. In: Katzenberg, M.
 A., and Saunders, S. R. (eds). Anthropology of the human skeleton, Second Edition.
 Hoboken, NJ: Wiley-Liss pp. 183-206.
- Ruff, C., Holt, B., and Trinkaus, E. (2006). Who's afraid of the big bad Wollf?" Wolff's Law" and bone functional adaptation. *American Journal of Physical Anthropology* 129 pp. 484-498.
- Ruff, C. B., Holt, B. M., Niskanen, M., Spadek, V., Berner, M., Garafulo, E., Garvin, H.M, Hora, M.,
 Maijanen, H., Nijnijmäki, S. and Tompkins, D. (2012). Stature and body mass estimation
 from skeletal remains in the European Holocene. *American Journal of Physical Anthropology* 148 pp. 601-617.
- Ruff, C. B., Junno. J-A., and Jamison, P. (2005). Body mass prediction form stature and bi-iliac breadth in two high latitude populations, with application to earlier high latitude humans.

Journal of Human Evolution 48 (4) pp. 381-392.

- Ruff, C. B., and Larsen, C. S. (2001). Reconstructing behavior in Spanish Florida: The biomechanical evidence. In Larsen, C. S. (ed). *Bioarchaeology Of Spanish Florida: The Impact Of Colonialism* Gainesville: University Press of Florida pp. 113-145.
- Ruff, C. B., Scott, W., and Liu, A. (1991). Articular and diaphyseal remodeling of the proximal femur with changes in body mass in adults. *American Journal of Physical Anthropology* 86 pp. 397-413.
- Ruff, C. B., Trinkhaus, A. W., and Larsen, C. S. (1993). Postcranial robusticity in Homo I: Temporal trends and mechanical interpretation. *American Journal of Physical Anthropology* 91 pp. 21-53.
- Rühli, F. J., and Henneberg, M. (2004) Clinical perspectives on secular trends of intervertebral foramen diameters in an industrialized European society. *European Spine Journal* 13(8) pp. 733-739.
- Rühli, F. J., Muntener, M., and Henneberg, M. (2005). Age-dependent changes of the normal human spine during adulthood. *American Journal of Physical Anthropology* 17 pp. 460-469.
- Rühli, F. J., Muntener, M., and Henneberg, M. (2006). Human osseous intervertebral foramen width. *American Journal of Physical Anthropology* 129 pp. 177-188.
- Saini, V., Srivastava, R., Rai, R. K., Shamal, S. M., Singh, T. B., and Tripathi, S. K. (2011) An osteometric study of northern Indian populations for sexual dimorphism in craniofacial region. *Journal of Forensic Science* 56 pp. 700-705.
- Sakaue, K. (1998). Bilateral asymmetry of the humerus in Jomon people and modern Japanese. Anthropological Science 105 pp. 231-246.
- Santos, A L. (1996). *How Old Is This Pelvis? A Comparison Of Age And Death Estimation Using The Auricular Surface Of The Ilium And Os Pubis*. Paper presented at the Aspects of African Archaeology. Papers from the 10th Congress of the Pan African Association for Prehistory and Related Studies, Harare. Available from:

https://core.ac.uk/download/pdf/19132685.pdf. (Accessed 23rd February 2011).

Santos, A. L., Alves-Carduso, F., Assis, S., and Vilotte, S. (2011). The Coimbra Workshop in musculoskeletal stress markers (MSM): an annotated review. *Anthropologia Portuguesa*, 28 pp. 135-161.

- Saunders, S. R., and Hoppa, R. D. (1993). Growth deficit in survivors and non-survivors: biological mortality bias in subadult skeletal samples. *Yearbook of Physical Anthropology* 36 pp. 127-151.
- Schaffler, M. B., Alson, M. D., Heller, J. G., and Garfin, S. R. (1992). Morphology of the dens. A quantitative study. *Spine* 17(7) pp. 738-743.
- Scheuer, L. (2002). Application of osteology to forensic medicine. *Clinical Anatomy* 15 pp. 297-312.
- Scheuer, L., and Black, S. M. (2004). The Juvenile Skeleton. San Diego: Elsevier Academic Press.
- Scheuer, L., and Bowman, J. E. (1995). Correlation of documentary and skeletal evidence in the St Bride's crypt population. In Saunders, S., and Herring, A. (eds). *Grave Reflections: Portraying The Past Through Cemetery Studies* Toronto: Canadian Scholars Press Inc pp. 49-70.
- Schmorl, G., and Junghanns, H. (1973). *The Human Spine In Health And Disease*. New York: Grune and Stratton.
- Schouten, J., de Bie, R. A., and Swaen, G. (2002). An update on the relationship between occupational factors and osteoarthritis of the hip and knee. *Current Opinion in Rheumatology* 14 pp. 89-92.
- Schrader, S A. (2012). Activity patterns in New Kingdom Nubia: An examination of entheseal modelling and osteoarthritis at Tombos. *American Journal of Physical Anthropology* 149 pp. 60-70.
- Schulter-Ellis, F. P., Schmidt, D. J., Hayek, L. C., and Craig, J. (1983). Determination of sex with a discriminant analysis of new pelvic bone measurements: Part I. *Journal of Forensic Sciences* 28(1) pp. 169-180.
- Schwartz, G. S. (1979). Society, physicians and the corset. *Bulletin of the New York Academy of Medicine* 55 pp. 35-66.
- Schwartz, J. H. (2006). *Skeleton Keys. An introduction to human skeletal morphology, development and analysis.* Oxford: Oxford University Press.
- Segal, D. A., and Yanagisako, S. J. (2005). Unwrapping The Sacred Bundle: Reflections On The Disciplining Of Anthropology. Durham: Duke University Press.

Segebarth-Orban, R. (1980). An evaluation of the sexual dimorphism of the human innominate

bone. Journal of Human Evolution 9 pp. 601-617.

- Selander, R. K. (1972). Sexual selection and dimorphism. In Campbell, B. (ed). *Sexual Selection And The Descent Of Man.* Chicago: Aldine Publishing pp. 180-230.
- Seeman, E. (2013). Age- and menopause-related bone loss compromise cortical and trabecular microstructure. *Journals of Gerontology. Series A, Biological and Medical Sciences.* 68 pp.1218-1225.
- Shapiro, R., and Heaney, R. P. (2003). Co-dependence of calcium and phosphorus for growth and development under conditions of varying deficiency. *Bone* 32 pp. 532-540.
- Sharma, L. (2001). The epidemiology of osteoarthritis. In Moskowitz, R. W., Howell, D. S., Altman,
 R. D., Buckwalter, J. A., and Goldberg, V. N. (eds). *Osteoarthritis: Diagnosis and Medical/Surgical Management* Philadelphia: W B Saunders pp. 3-28.
- Shaw, C. (2011). Is 'hand preference' coded in the hominin skeleton? An in-vivo study of bilateral morphological variation. *Journal of Human Evolution* 61(4) pp. 480-487.
- Shaw, C. N., and Stock, J. T. (2009a). Intensity, repetitiveness and directionality of habitual adolescent mobility patterns influence the tibial diaphysis morphology in athletes. *American Journal of Physical Anthropology* 140 pp. 149-159.
- Shaw, C. N., and Stock, J. T. (2009b). Habitual throwing and swimming correspond with upper limb diaphyseal strength and shape in modern human athletes. *American Journal of Physical Anthropology* 140 pp. 160-172.
- Shinohara, H. (1997). Changes in the surface of the superior articular joint from the lower thoracic to the upper lumbar vertebrae. *Journal of Anatomy* 190 pp. 461-465.
- Silberstein, C. E. (1965). The evolution of degenerative changes in the cervical spine and an investigation into the "Joints of Luschka". *Clinical Orthopaedics and Related Research* 40 pp. 184-204.
- Sindermann, A. and Flohr, S. (2007). Degenerative Veränderungen und Fehlbildungen an den Wirbelsäulen einer merowingergerzeitlichen Population aus Rhens, Kreis Mayen-Koblenz. Beiträge zur Archäozoologie und Prähistorischen Anthropologie 6, Beier & Beran: Langenweißbach pp. 76-83.
- Sim, J., and Wright, C. C. (2005). The kappa statistic in reliability studies: use, interpretation and sample size requirements. *Physical Therapy* 85(3) pp. 257-268.

- Simmons, T., and Haglund, W. D. (2005). Anthropology in a forensic context. In Hunter, J., and Cox, M. (ed). *Forensic Archaeology: Advances In Theory And Practice* London: Routledge pp. 159-176.
- Singer, M. (1996) Farewell to adaptionism: Unnatural selection and the politics of biology. *Medical* Anthropology Quarterly 10(4) pp. 496-515
- Singer, K. P., Bricdahl, P. D., and Day, R. E. (1998). Variations in the zygapophyseal joint orientation and the level of transition at the thoracolumbar junction. *Surgery and Radiologic Anatomy* 10 pp. 291-295.
- Skerry, T. (2006). One mechanostat or many? Modifications of the site-specific response of bone to mechanical loading by nature or nurture. *Journal of Musculoskeletal and Neuronal Interaction* 6 pp. 122-127.
- Slatkin, M. (1984). Ecological causes of sexual dimorphism. Evolution 38 pp. 622-630.
- Slaus, M., Strinovic, D., Skavic, K. J., and Petrovecki, V. (2003). Discriminant function sexing of fragmentary and complete femora: standards for contemporary Croatia. *Journal of Forensic Science* 48 pp. 509-512.
- Sládek, v., Machácek, J., Ruff, C. B., Schuplerová, E., Prichystalová, R., and Hora, M. (2015).
 Population-specific stature estimation from long bones in the early Medieval human population at Pohansko (Czech Republic). *American Journal of Physical Anthropology* 158 pp. 312-324.
- De Smet, A. A. (1985). Radiology Of Spinal Curvature. St Louis: Mosby.
- Smith, A. (1903). Anglo-Saxon remains. In Doubleday, A., and Page, W. (eds). *Victoria County History Vol 1: Early man, Ancient Earthworks, Anglo-Saxon Remains And The Domesday Book.*
- Smith, R. J. (1999). Statistics of sexual size dimorphism. *Journal of Human Evolution* 36 pp. 423-459.
- Snell, R. S. (2004). Clinical Anatomy (7th ed.). Baltimore: Lippincott, Williams and Wilkins.
- Snodgrass, J. (2004). Sex differences and aging of the vertebral column. *Journal of Forensic Science* 49(3) pp. 1-6.
- Snow, C. M., Shaw, J. M., and Matkin, C. C. (1996). Physical activity and the risk of osteoporosis. In Feldman, M. R., and Kelsey, J. (eds). *Osteoporosis*. New York: Academic Press pp. 511-528.

- Sofaer Deverenski, J. R. (2000). Sex differences in activity-related osseous change in the spine and the gendered division of labor at Ensay and Wharram Percy, UK. *American Journal of Physical Anthropology* 111(3) pp. 333-354.
- Sofaer, J. (2006). *The Body As Material Culture: A Theoretical Osteoarchaeology*. Cambridge: Cambridge University Press.
- Sokal, R. R., and Rohlf, F. J. (1995) *Biometry* (3rd ed) New York: W. H. Freeman.
- Sokoloff, L. (1987). Osteoarthritis as a remodeling process. Journal of Rheumatology 14 pp. 7-10.
- Solano, M. (2002). Activity related pathology in the Albany county almshouse, Albany, NY. American Journal of Physical Anthropology 34S pp. 145-149.
- Solomon, L. (1984). Geographical and anatomical patterns of osteoarthritis. *British Journal of Rheumatology* 23 pp. 177-180.
- Sowers, M. (2001). Epidemiology of risk factors for osteoarthritis: systemic factors. *Current Opinions in Rheumatology* 13(5) pp. 447-451.
- Sowers, M., and Karvonen-Gutierrez, C. A. (2012). Osteoarthritis. In Schenk-Gustafsson, K., DeCola, P.R., Pfaff, D., and Pisetsky, D.S. (eds). *Handbook Of Clinical Gender Medicine*. Basel: Karger pp. 358-366.
- Sowers, M. F., Hochenberg, M., and Crabbe, J. P. (1996). Association of bone mineral density and sex hormone levels with osteoarthritis of the hand and knee in premenopausal women. *American Journal of Epidemiology* 143 pp. 38-47.
- Spector, T. D., Cicuttini, F., and Baker, J. (1996). Genetic influences on osteoarthritis in women: a twin study. *British Medical Journal* 312 (7036) pp. 940-943.
- Spector, T. D., Harris, P. A., and Hart, D. J. (1996). Risk of osteoarthritis associated with long-term weight-bearing sports: a radiologic survey of the hips and knees of female ex-athletes and population controls. *Arthritis and Rheumatism* 39 pp. 988-995.
- Spector, T. D., and MacGregor, A. J. (2004). Risk factors for osteoarthritis: genetics. *Osteoarthritis Cartilage* 12 Suppl A pp. S39-44.
- Sperduti, A. (1997). Life conditions of a Roman Imperial Age population: Occupational skeletal markers and working activities in Lucas Feroniae (Rome, 1st-2nd cent. AD). *Human Evolution* 12(4) pp. 253-267.

- Spradley, M., Jantz, R., Robinson, A., and Peccerelli, B. (2008). Demographic change and forensic identification: problems in metric identification of Hispanic skeletons. *Journal of Forensic Science* 53(1) pp. 21-28.
- Spradley, M K, & Jantz, R L. (2011). Sex estimation in forensic archaeology: skull versus postcranial elements. *Journal of Forensic Sciences, 56*(2), 289-296.
- Spurrell, H. (1882). Antiquarian notes on the District of Eastbourne Natural History Society *Transactions of the Eastbourne Natural History Society* Vol. 1 pp. 30-31.
- Srikanth, V. K., Fryer, J. L., Zhai, G., Winzenberg, T. M., Hosmer, D., & Jones, G. (2005). A metaanalysis of sex differences prevalence, incidence and severity of osteoarthritis. *Osteoarthritis Cartilage* 13(9) pp. 769-781
- Steckel, R. H., and Rose, J. C. (2002). *The Backbone Of History: Health And Nutrition In The Western Hemisphere*. Cambridge: Cambridge University Press.
- Steele, D. G., and Bramblett, C. A. (1988). *The Anatomy And Biology Of The Human Skeleton* Texas: A & M University Press.
- Steele, J. (2000). Skeletal indicators of handedness. In Cox, M., and Mays, S. (eds). Human osteology in archaeology and forensic science Cambridge: Cambridge University Press pp. 307-323.
- Steele, J., and Mays, S. (1995). Handedness and directional asymmetry in the long bones of the human upper limb. *International Journal of Osteoarchaeology* 5 pp. 39-49.
- Steele, V. (2001). *Fashion And Eroticism: The Ideals Of Feminine Beauty From The Victorian Era To The Jazz Age.* New York: Oxford University Press.
- Stefanović, S., and Porčić, M. (2011). Between-group differences in the patterning of musculoskeletal stress markers: avoiding confounding factors by focusing on qualitative aspects of physical activity. *International Journal of Osteoarchaeology* 23 pp. 94-105
- Stensland, S. H., and Margolis, S. (1990). Simplifying the calculation of body mass index for quick reference. *Journal of the American Dietetic Association* 90(6) pp. 858-861.
- Stewart, T. D. (1947). Racial patterns in vertebral osteoarthritis. *American Journal of Physical Anthropology* 5 pp. 230-231.
- Stewart, T. D. (1952). Hardlicka's Practical Anthropometry. Instruments 4th Ed Philadelphia pp. 25-40.

- Stewart, T. D. (1954). Determination of the skeleton by guess and measurement. *American Journal of Physical Anthropology* 12 pp. 385-392.
- Stewart, T D. (1966). Some problems in human paleopathology. In Jarcho, S. (ed). *Human Paleopathology*. New Haven: Yale University Press.
- Stewart, T. D. (1979). Essentials Of Forensic Anthropology, Especially As Developed In The United States. Springfield IL: Charles C Thomas.
- Steyn M., Işcan M. (1999). Osteometric variation in the humerus: sexual dimorphism in South Africans. *Forensic Science International* 106(2) pp. 77-85.
- Stini, W. A. (1982a). Adaptive strategies of human populations under nutritional stress. In Watts,
 S.E., Johnson, F. E., and Lasker, G. W. (eds). *Biosocial Interrelations In Population Adaptation* The Hague: Mouton pp. 19-41.
- Stini, W A. (1982b). Sexual dimorphism and nutrient reserves. In Hall, R. L. (ed). Sexual Dimorphism In Homo Sapiens: A Question Of Size. New York: Praeger Publishing.
- Stinson, S. (1985). Sex differences in environmental sensitivity during growth and development. *Yearbook of Physical Anthropology* 28 pp. 123-147.
- Stinson, S., Bogin, B., Huss-Ashmore, R., and O'Rourke, D. (2012). *Human Biology: An Evolutionary* And Biocultural Perspective. Hoboken, N. J.: Wiley-Blackwell.
- Stirland, A. (1991). Diagnosis of occupation related paleopathology. Can it be done? In Ortner, D.
 J., and Aufderheide, A. C. (eds). *Human Paleopathology: Current Syntheses And Future Options*. Washington DC: Smithsonian Institution Press.
- Stirland, A. (1993). Asymmetry and activity-related change in the male humerus. *International Journal of Osteoarchaeology* 3 pp. 105-113.
- Stirland, A., and Waldron, T. (1997). Evidence for activity related markers in the vertebrae of the crew of the Mary Rose. *Journal of Archaeological Science* 24 pp. 327-335.
- Stirland, A. J. (1993). Asymmetry and activity-related change in the male humerus. *International Journal of Osteoarchaeology* 3 pp. 105-113.
- Stock, J. (2002). A test of two methods of radiographically deriving long bone cross sectional properties compared to direct sectioning of the diaphysis. *International Journal of Osteoarchaeology* 12 pp. 335-342.

- Stock, J. (2006). Hunter-gatherer postcranial robusticity relative to patterns of mobility, climatic adaptation, and selection for tissue economy. *American Journal of Physical Anthropology* 131 pp. 194-204.
- Stock, J. O'Neill, M. C., Ruff, C. B., Zabecki, M., Shackleford, L., and Rose, J. C. (2011). Body size, skeletal biomechanics, mobility and habitual activity from the Late Paleolithic to the Middle-Dynastic Nile Valley. In Pinhasi, R., and Stock, J. T. (eds). *Human Bioarchaeology Of The Transition To Agriculture*. Chichester: John Wiley and Sons Ltd.
- Stock, J., and Pfeiffer, S. (2001). Linking structural variability in long bone diaphyses to habitual behaviours: foragers from the Southern African Later Stone Age and the Andaman Islands. *American Journal of Physical Anthropology* 115 pp. 337-348.
- Stock, J. T. (2006). Hunter-gatherer postcranial robusticity relative to patterns of mobility, climatic adaptation, and selection for tissue economy. *American Journal of Physical Anthropology* 131 pp. 194-204.
- Stock, J. T., and Pfeiffer, S. K. (2004). Long bone robusticity and subsistence behaviour among Later Stone Age foragers of the forest and fynbos biomes of South Africa. *Journal of Archaeological Science* 31 pp. 999-1013.
- Stock, J. T., and Shaw, C. N. (2007). Which measures of diaphyseal robusticity are robust? A comparison of external methods of quantifying the strength of long bone diaphyses to cross-sectional geometric properties. *American Journal of Physical Anthropology* 134 pp. 412-423.
- Stone, K. L., Seeley, D. G., Lui, L-Y., Cauley, J. A., Ensrund, K., Browner, W. S., and Cummings, S. R.
 (2003). BMD at multiple sites and risk of fracture of multiple-types: Long term results
 from the study of osteoporotic fractures. *Journal of Bone and Mineral Research* 18(11) pp. 1947-1954.
- Stoodley, N. (2011). Childhood to old age. In Hammerow, H., Hinton, D.A., and Crawford, C. (eds).
 The Oxford handbook of Anglo-Saxon archaeology. Oxford: Oxford University Press pp. 641-666.
- Su, A., Wallace, I. J., and Nakatsukasa, M. (2013). Trabecular bone anistropy and orientation in an Early Pleistocene hominin talus from East Turkana, Kenya. *Journal of Human Evolution* 64 pp. 667-677.

Suchey, J. M., Brooks, S. T., and Katz, D. M. (1988). Instructions for the use of the Suchey-Brooks

system for age determination for the female os pubis. *Instructional Materials Accompanying Female Pubic Symphyseal Models*. Suchey-Brooks system. Distributed by France Casting, Fort Collins, Colorado.

- Sumati, V. V. G., and Phatak, A. (2010). Determination of sex from the mastoid process. *Journal of the Anatomical Society of India* 59(2) pp. 222-228.
- Suri, P., Katz, J. N., Rainville, J., Kalichman, L., Guermazi, A., and Hunter, D. J. (2010). Vascular disease is associated with facet joint osteoarthritis. *Osteoarthritis Cartilage* 18 pp. 1127-1132.
- Swedenborg, I. (1974). *Degenerative Changes In The Human Spine*. Stockholm: Osteological Research Laboratory, University of Stockholm.
- Swedlund, A. C., and Herring, D. H. (2003). Human biologists in the archives: Demography, health, nutrition and genetics in historical populations. In Herring, D., and Swedland, A. (eds).
 Human Biologists In The Archives: Demography, Health, Nutrition And Genetics In Historical Populations Cambridge: Cambridge University Press pp. 1-10.
- Swenson, V. M. (2013). Ancestral And Sex Estimation Using E. A. Marino's Analysis Of The First Cervical Vertebra Applied To Three Modern Groups. (Masters thesis), University of Montana, Missoula, MT. Available from: http://scholarworks.umt.edu/cgi/viewcontent.cgi?article=1164&xontext=edu. (Accessed 24th June 2014).
- Taddei, F., Balestri, M., Rimondi, E., Viceconti, M., Manfrini, M. (2009). Tibia adaptation after fibula harvesting: an *in vivo* quantitative study. *Clinical Orthopaedics and Related Research.* 467 pp. 2149-2158.
- Tait, T. J., and Bird, H. A. (1994). Asymmetrical osteoarthritis in a patient with Ehrlers-Danlos syndrome and poliomyelitis. *Clinical and Experimental Rheumatology* 12 pp. 425-427.
- Tanzer, M., and Noiseux, N. (2004). Osseous abnormalities and early osteoarthritis: the role of hip impingement. *Clinical Orthopaedics and Related Research* 429 pp. 170-177.
- Tarli, S. M., and Repetto, E. (1997). Sex differences in Human Populations: Change through time.
 In Morbeck, M.E., Galloway, A., and Zihlman, A. (eds). *Evolving Female* Princeton NJ:
 Princeton University Press pp. 198-208.
- Tatarek, N. E. (2001). Variation in the lumbar neural canal. *American Journal of Physical Anthropology* 10 pp. 4613-4514.

Taylor, E. J. (ed). (1988). Dorland's Medical Dictionary. Philadelphia: W B Saunders.

Taylor, J. R. (1983). Scoliosis and growth. Acta Orthopaedica Scandinavica 54 pp. 596-602.

- Taylor, J. R. (1986). Vascular causes of vertebral asymmetry. *Medical Journal of Australia* 144 pp. 533-535.
- Tepper, S., and Hochberg, M. C. (1993). Factors associated with hip osteoarthritis: Data from the First National and Nutrition Examination survey (NHANES - 1). American Journal of Epidemiology 137 pp. 1081-1088.
- Thelin, A. (1990). Hip joint arthrosis: an occupational disorder among farmers. *American Journal of Industrial Medicine* 18 pp. 339-343.
- Thieme, F., and Schull, W. (1957). Sexual determination from the skeleton. *Human Biology* 29(3) pp. 242-273.
- Thomas, A. (2014). Bioarchaeology of the middle Neolithic: evidence for archery among early European farmers. *American Journal of Physical Anthropology* 154 pp. 279-290.
- Thorstensson, C. A., Petersson, I. F., Jacobsson, L. T., Boegard, T. L., and Roos, E. M. (2004). Reduced functional performance in the lower extremity predicted radiographic knee osteoarthritis five years later. *Annals of the Rheumatic Diseases* 63 pp. 402-407.
- Todd, T. W. (1920). Age changes in the pubic bone: I The male White pubis. *American Journal of Physical Anthropology* 3 pp. 285-334.
- Trew, M., and Everett, T. (2005). *Human Movement* (5th Ed.) London: Elsevier Churchill Livingstone.
- Trinkaus, E., Churchill, S. E., and Ruff, C. B. (1994). Postcranial robusticity in homo. II Human
 bilateral asymmetry and bone plasticity. *American Journal of Physical Anthropology* 93 pp.
 1-34.
- Trotter, M., and Gleser, G. C. (1952). Estimation of stature from long bones of American Whites and Negroes. *American Journal of Physical Anthropology* 10 pp. 463-514.
- Trotter, M., and Gleser, G. C. (1958). A re-revaluation of estimation of stature based on measurements of stature taken during life and of long bones after death. *American Journal of Physical Anthropology* 16 pp. 79-124.

Tucker, C. (1990). The Mechanics Of Sports Injuries: An Osteopathic Approach. Oxford: Blackwell

Scientific Publications.

- Tulsi, R. S., and Hermanis, G. M. (1993). A study of the angle of inclination and facet curvature of superior lumbar zygapophyseal facets. *Spine* 18(10) pp. 1311-1317.
- Turner, C. H. (1992). Functional determinants of bone structure: beyond Wolff's law of bone transformation. *Bone* 13 pp. 403-409.
- Turner, C. H., and Robling, A.G. (2003). Designing exercise regimes to increase bone strength. *Exercise and Sports Review* 31 pp. 45-50.
- Turner, C. H., Owan, I., and Takano, Y. (1995). Mechanotransduction in bone: do bone cells act as sensors of fluid flow. *Biomechanics* 32 pp. 437-441.
- Ubelaker, D. H. (1989). Human Skeletal Remains. Washington DC: Taraxacum Press.
- Ubelaker, D. H, and Volt, C. G. (2002). A test of the Phenice method for the estimation of sex. Journal of Forensic Science 47 pp. 19-24.
- Uesugi, Y., Taguchi, O., Noumura, T., and Iguchi, T. (1992). Effects of sex steroids on the development of sexual dimorphism in mouse innominate bone. *The Anatomical Record,* 234 pp. 541-548.
- Uhl, N. M. (2013). Age-at-Death Estimation. In DiGangi, E. A., and Moore, M. K. (Eds.), *Research Methods In Human Skeletal Biology*. New York: Academic Press pp. 63-90.
- Uhrenholt, L., Hauge, E., Charles, A. V., and Gregersen, M. (2008). Degenerative and traumatic changes in the lower cervical spine facet joints. *Scandinavian Journal of Rheumatology* 37(5) pp. 375-384.
- Underdown, S. (2012). Fitzroy's fighting Fuegians: Comparison of paleopathological and ethnographical indications of fighting behaviour in an extinct human population. *International Journal of Osteoarchaeology* 22(3) pp. 373-378.
- Usher, B. (2002). Reference samples: The first step in linking biology and age in the human In Hoppa, R. D., and Vaupel, J. W. (eds). *Paleodemography: Age Distributions From Skeletal Samples* Cambridge: Cambridge University Press pp. 29-47.
- Valdes, A. M., and Spector, T. D. (2009). The contribution of genes to osteoarthritis. *Medical Clinics of North America* 93 pp. 45-66.

Van de Merwe, A. E., Isçan, M. Y. and Abbel, E. N. (2006). The pattern of vertebral osteophyte

development in a south African population. *International Journal of Osteoarchaeology* 16 pp. 459-464.

- Van Roy, P., Caboor, D., De Boelpaep, S., Barbaix, E., and Clarys, J. P. (1997). Left-right asymmetries and other common anatomical variants of the first cervical vertebra. *Manual Therapy* 2(1) pp. 24-36.
- van Saase, J. L., Vandenbroucke, J. P., van Romunde, L. K., and Valkenburg, H. A. (1988). Osteoarthritis and obesity in the general population. A relationship calling for an explanation. *Journal of Rheumatology* 15(7) pp. 1152-1158.
- Vignon, E., Valat, J., Rossignol, M., Avouac, B., Rozenberg, S., Thoumie, P., and Hilliquin, P. (2006).
 Osteoarthritis of the knee and hip and activity; a systematic review and synthesis (OASIS).
 Joint Bone Spine 73 pp. 442-455.
- Vilotte, S., Castex, D., Couallier, V., Dutour, O., Knüsel, C. J., and Henry-Gambier, D. (2010). Enthesopathies as occupational stress markers: evidence from the upper limb. *American Journal of Physical Anthropology* 142 pp. 224-234.
- Vingard, E., Alfredsson, L., Goldie, I., and Hogstedt, C. (1993). Sports and osteoarthritis of the hip. An epidemiologic study. *The American Journal of Sports Medicine* 21 pp. 195-200.
- Waldron, T. (1987). The relative survival of human skeleton: implications for paleopathology. In
 Boddington, A., Garland, A. N., and Janaway, R. C. (eds). *Death, Decay and Reconstruction: Approaches to Archaeology and Forensic Science.* Manchester University Press:
 Manchester pp. 55-64.
- Waldron, T. (1991). Prevalence and distribution of osteoarthritis in a population from Georgian and early Victorian London. *Annals of Rheumatic Diseases* 50 pp. 301-357.
- Waldron, T. (1991). The prevalence of, and the relationship between some spinal diseases in a human skeletal population from London. *International Journal of Osteoarchaeology* 1 pp. 103-110.
- Waldron, T. (1992). Osteoarthritis in a Black Death cemetery in London. *International Journal of Osteoarchaeology* 2 pp. 235-240.
- Waldron, T. (1994). *Counting the Dead, the Epidemiology of Skeletal Populations*. New York: Wiley.
- Waldron, T. (1994b). The human remains. In Evison, V. (ed). An Anglo-Saxon Cemetery At Great

Chesterford, Essex. Council for British Archaeology Report 91 pp. 52-66.

- Waldron, T. (1995). Changes in the distribution of osteoarthritis over historical time. *International Journal of Osteoarchaeology* 5 pp. 385-389.
- Waldron, T. (1996). Osteoarthritis of the hands in early populations. *British Journal of Rheumatology* 35(12) pp. 1292-1298.
- Waldron, T. (1997). Osteoarthritis of the hip in past populations. *International Journal of Osteoarchaeology* 7 pp. 186-189.
- Waldron, T. (2007). *Paleoepidemiology. The measure of disease in the human past*. California: Left Coast Press
- Waldron, T. (2009). Palaeopathology. Cambridge: Cambridge University Press.
- Waldron, T. (2012). Joint Disease. In Grauer. A. (ed). *A Companion Guide To Paleopathology*. Chichester Wiley Blackwell.
- Walker, P. L. (1995). Problems of preservation and sexism in sexing: Some lessons from historical collections for Paleodemographers. In Saunders, S.R. and Herring, A. (eds). *Portraying the Past Through Skeletal Studies.* Toronto: Canadian Scholars' Press pp. 31-47.
- Walker, P. L. (2008). Sexing skulls using discriminant function analysis of visually assessed traits. *American Journal of Physical Anthropology* 136 pp. 39-50.
- Walker, P. L., Bathurst, P. R., Richman, R., Gjerdrum, T., and Andrushko, V. A. (2009). The causes of porotic hyperostosis and cribra orbitalia: a reappraisal of the iron-deficiency hypothesis. *American Journal of Physical Anthropology* 139 pp. 109-125.
- Walker, P. L., Johnson, J. R., and Lambert, P. M. (1988). Age and sex biases in the preservation of human skeletal remains. *American Journal of Physical Anthropology* 76 pp. 183-188.
- Walker, P. L., and Lambert, P M. (1991). Human Skeletal Remains From The Historic Cemetery At Seccombe Lake Park, San Bernardino, California. Report prepared for City of San Bernardino Parks, Recreation and Community Services Department.
- Walrath, D. E., Turner, P., and Bruzek, J. (2004). Reliability test of the visual assessment of cranial traits for sex determination. *American Journal of Physical Anthropology* 125 pp. 132-137.
- Wang, J., and Yang, X. (2009). Age-related changes in the orientation of lumbar facet joints. *Spine* 34(17) pp. E596-598.

- Wanner, I. S., Sosa, T. S., Alt, K. W., and Blas, V. T. (2007). Lifestyle, occupation and whole bone morphology of the pre-Hispanic Maya coastal populations from Xcambo, Yucatan, Mexico. *International Journal of Osteoarchaeology* 17(3) pp. 253-268.
- Ward, J. A., Lord, S. R., Williams, P., Anstey, K., and Zivanovic, E. (1995). Physiologic, health and lifestyle factors associated with femoral neck and bone density in older women. *Bone* 16(4) pp. 373S-378S.
- Warden, S. J., Hurst, J. A., and Sanders, M. S. (2005). Bone adaptation to mechanical loading program significantly increases skeletal fatigue resistance. *Journal of Bone and Mineral Research* 20 pp. 809-816.
- Washburn, S. L. (1948). Sex differences in the pubic bone. *American Journal of Physical Anthropology* 6(2) pp. 199-208.
- Washburn, S. L. (1949) Sex differences in the pubic bone of Bantu and Bushmen. *American Journal* of Physical Anthropology 7 pp. 425-432.
- Watanabe, S., and Terazawa, K. (2006). Age estimation from the degree of osteophyte formation of vertebral columns in Japanese. *Legal Medicine* 8 pp. 156-160.
- Waters-Rist, A., Bozaliiskii, V. I., Weber, A., Goriunova, O., and Katzenberg, M. A. (2011). Activity induced dental modification in Holocene Siberian Hunter-Fisher-Gatherers. *American Journal of Physical Anthropology* 20(2) pp. 266-278.
- Watts, R. (2010). Non-specific indicators of stress and their association with age at death in Medieval York: Using stature and vertebral neural canal size to examine the effects of stress occurring during different periods of development. *International Journal of Osteoarchaeology* 21(5) pp. 568-576.
- Weiss, E. (2003). Effect of rowing on humeral strength. *American Journal of Physical Anthropology* 121 pp. 293-302.
- Weiss, E. (2005a). Understanding osteoarthritis patterns: an examination of aggregate osteoarthritis. *Journal of Paleopathology* 16 pp. 88-98.
- Weiss, E. (2005b). Osteoarthritis and body mass. *Journal of Archaeological Science* 33(5) pp. 690-695.
- Weiss, E., Corona, L., and Schultz, B. (2012). Sex differences in musculoskeletal markers: problems with activity pattern reconstruction. *International Journal of Osteoarchaeology* 22 pp. 70-

80.

- Weiss, E., and Jurmain, R. (2007). Osteoarthritis revisited: a contemporary review of aetiology. International Journal of Osteoarchaeology 17(5) pp. 437-450.
- Weiss, K. (1998). Coming to terms with human variation. *Annual Review of Anthropology* 27 pp. 273-300.
- Welch, M. (2000). Anglo-Saxon England. B. T. Batsford Ltd.: London.
- Wells, J. C. K., and Stock, J. T. (2007). The biology of the colonizing ape. *Yearbook of Physical Anthropology* 50 pp. 191-222.
- Wescott, D. J. (2000). Sex variation in the second cervical vertebra. *Journal of Forensic Science* 45(2) pp. 462-466.
- Wescott, D. J. (2001). Structural Variation in the Humerus and Femur in the American Great Plains and Adjacent Regions: Differences in Subsistence Strategy and Physical Terrain. (PhD thesis), University of Tennessee, Knoxville. Available from: <u>https://www.academia.edu/295997/Structural Variation In the Humerus and Femur I</u> <u>n the American Great Plains and Adjacent Regions Differences In Subsistence Strat</u> <u>egy and Physical Terrain</u>. (Accessed 19th April 2009).
- Wescott, D. J. (2006). Effect of mobility on femur midshaft external shape and robusticity. *American Journal of Physical Anthropology* 130 pp. 201-203.
- Whitcombe, K. K., Shapiro, L. J., and Lieberman, D. E. (2007). Fetal load and the evolution of lumbar lordosis in bipedal hominins. *Nature* 450(3) pp. 1075-1078.
- White, T. D., Black, M. T., and Folkens, M. T. (2012). *Human Osteology*. San Diego: Academic Press.
- White, T. D., and Folkens, M. T. (2005). *The Human Bone Manual*. New York: Elsevier Academic Press.
- Whitmarsh, T., Barquero, L. M. Del Rio, Gregorio, S. D., Sierra, J. M., Humbert, L., and Frangi, A. F.
 (2012). Age-related changes in vertebral morphology by statistical shape analysis. In
 Levine, J. A., Paulson, R. R. and Zhang, Y. (eds). *MeshMed 2012, LNCS 7599* Heidelberg:
 Springer-Verlag pp. 30-39.
- Wilder, F. V., Barrett, J. P., and Farina, E. J. (2006). Joint-specific prevalence of osteoarthritis of the hand. *Osteoarthritis Cartilage* 14(9) pp. 953-957.

- Williams, B. A., and Rogers, T. (2006). Evaluating the accuracy and precision of cranial morphological traits for sex determination. *Journal of Forensic Science* 51(4) pp. 729-735.
- Williams, M. M. (1987). Sex determination of fragmentary crania by analysis of the cranial base: Applications for the study of an Arikara skeletal sample. (Masters Thesis), University of Tennessee, Knoxville.
- Williams, P. (2000). Gray's Anatomy (38th ed.). Edinburgh, London: Churchill Livingstone.
- Wilson, J. D., George, F. W., and Griffin, J. E. (1981). The hormonal control of sexual development. *Science* 11 pp. 1278-1284.
- Wilson, J. D., Griffin, J. E., George, F. W., and Leshin, M. (1981). The role of gonadal steroids in sexual differentiation. *Recent Progress in Hormone Research* 37 pp. 1-39.
- Wolfe, J. P., and Gray, L. D. (1982). Height and sexual dimorphism of stature among human societies. *American Journal of Physical Anthropology* 53(3) pp. 441-456.
- Wolff, J. (1986). *The Law of Bone Remodeling (translation of the German 1892 edition)*. Berlin: Springer.
- Woo, E. J., and Sciulli, P. W. (2011). Degenerative joint disease and social status in the terminal late Archaic period (1000-500 b.c.) of Ohio. *International Journal of Osteoarchaeology* 23(5) pp. 529-544.
- Wood, R. J. (1995). Biomechanics Of Osteoarthritis Of The Knee. (PhD thesis), Columbus OH.
 Available from: <u>https://etd.ohiolink.edu/!etd.send_file?accession=osu1487931512620656&disposition=in</u> <u>line</u>. (Accessed 19th April 2012).
- Wright, V. (1990). Post-traumatic osteoarthritis a medico-legal minefield. *Rheumatology* 29(6) pp. 474-478.
- Xavier, M. J. (2003). A practical guide to analyzing survey data [web page] Available from: <u>http://www.scribd.com/doc/62237715/Guide-Xavier</u>. (Accessed 23rd June 2016).
- Yang, K. H., and King, A. I. (1984). Mechanism of facet load transmission as a hypothesis for lowback pain. *Spine* 9 pp. 557-565.
- Yogandan, N., Knowles, S. A., Maiman, D. J., and Pintar, F. A. (2003). Anatomic study of the morphology of human cervical facet joint. *Spine* 28(20) pp. 2317-2323.

- Yu, S. B., Lee, U. Y., Kwak, D. S., Ahn, Y. W., Jin, C. Z., Zhao, J., and Han, S. H. (2008). Determination of sex for the 12th thoracic vertebra by morphometry of three-dimensional reconstructed vertebral models. *Journal of Forensic Science* 53(3) pp. 620-625.
- Zakrzewski, S. R. (2003). Variation in ancient Egyptian stature and body proportions. *American Journal of Physical Anthropology* 121 pp. 219-229.
- Zheng, W. X., Cheng, F. B., Cheng, K. L., Tian, Y., Lai, Y., Zhang, W. S., and Li, Y. Q. (2012). Sex assessment using measurements of the first lumbar vertebrae. *Forensic Science International* 10(219) pp. 1-3.
- Zuckerman, M. K, and Armelagos, G. J. (2011). The origins of biocultural dimensions. In Agarwal, S.A., and Glencross, B.A. (eds) *Social Bioarchaeology*. Chichester: Wiley Blackwell pp. 1-43.
- Zukowski, L. A., Falsetti, A. B., and Tillman, M. D. (2012). The influence of sex, age and BMI on the degeneration of the lumbar spine. *Journal of Anatomy* 220(1) pp. 57-66.